



建物

ProQuest Number: 10694068

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10694068

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

# Accounting Information System in the Water Industry: The Case of Cost Management

## Yasser Kamal El-Sayed Abd-El-Naby

A thesis submitted in partial fulfilment of the requirements of Sheffield Hallam University for the degree of Doctor of Philosophy

**July 2002** 



### **Table of Contents**

Abstract	16
Acknowledgements	17
List of Abbreviations	

## **Chapter One: Introduction**

1.0 Introduction	
1.1 Background	20
1.2 Objectives and research questions	22
1.3 The motivation of the study	23
1.4 Theoretical framework	24
1.5 Research Methodology and Method	26
1.6 Structure of the thesis	28

## **Chapter Two: Institutional theory**

2.0 Introduction	
2.1 The rationale for choosing institutional theory	
2.2 History of institutional theory	
2.3 Institutional theory and Accounting	43
2.4 Conclusion	55

## Chapter Three: Institutional theory and the role of management

### accounting

3.0 Introduction	58
3.1 The role of accounting in organisation	59
3.1.1 Different views of accounting and its roles	59
3.1.2 Accounting systems and organisational practice	62
3.1.3 Accounting and Social Practice	67
3.2 The role of Management Accounting	71
3.2.1 Economic perspective	71
3.2.2 The problem of management accounting in organisation services	77
3.3 Theory of management accounting	80
3.3.1 Management Accounting from a neo-classical economic perspective	80

3.3.2 Management Accounting	g from an Institutional perspective	83
3.4 Conclusion		92

## **Chapter Four: Models of cost management**

4.0 Introduction	94
4.1 Efficiency	95
4.2 Institutional theory and efficiency	
4.3 Activity Based Costing (ABC)	
4.4 Activity Based Management	
4.3 Conclusion	

## Chapter Five: Strategic management accounting and cost management

5.0 Introduction	
5.1 Cost accounting	
5.1.1 Changes in cost accounting practices textbook	
5.1.2 Cost accounting in the past and now	
5.1.3 Cost accounting in service organisations	
5.2 Strategic Management Accounting (SMA)	
5.2.1 The first viewpoint of SMA.	134
5.2.2 The second viewpoint of SMA	
5.2.3 The third viewpoint of SMA	144
5.2.4 The fourth viewpoint of SMA	
5.3 Cost accounting and corporate strategy	150
5.4 Conclusion	

## Chapter Six: Methodology and method

6.0 Introduction	
6.1 Research design and methodology	
6.1.1 Epistemology	
6.1.2 Ontology	
6.1.3 Human nature	
6.2 Research method	
6.2.1 Case study	
6.2.2 Qualitative approach	
6.3 Conclusion	

I

## Chapter Seven: The Water industry, Cost management and

## Institutional theory

7.0 Introduction	191
7.1 Public utility	192
7.1.1 Characteristics	192
7.1.2 Types of services	192
7.1.3 A schematic Diagram of the public utilities	193
7.1.4 Importance, demand for water and the responsibility of government	196
7.2 Privatisation	199
7.2.1 The aim of privatisation in UK	199
7.2.2 The effect of privatisation	202
7.2.3 The system of water in France	211
7.2.4 The water system in Egypt	213
7.2.5 Water industry and institutional perspective	218
7.3 Issues of costs in water industry	222
7.3.1 Cost system in water industry and some issues	222
7.3.2 How do the people pay for the water?	227
7.3.3 How to calculate the cost of water?	231
7.4 Conclusion	234

## **Chapter Eight: Cost management in the GOGCWS**

8.0 Introduction	236
8.1 The Legal framework for public utilities in Egypt	238
8.2 The description of the GOGCWS	240
<ul><li>8.2.1 A brief history of the GOGCWS</li><li>8.2.2 The role (activities) of the GOGCWS</li></ul>	240 241
8.3 The organisation structure of the GOGCWS	242
8.4 The supervision and control body on the GOGCWS	246
<ul><li>8.4.1 The internal control</li><li>8.4.2 The external control</li></ul>	246 249
8.5 The Accounting system in the GOGCWS	251
8.5.1 Objectives, notion and contents of the Unified Accounting System	251
8.6 Cost accounting System	254
8.6.1 Cost accounting development and the needs	255

8.6.2 Changes in cost management	257
8.6.3 Measuring and estimating product cost	259
8.6.4 Water Funding	
8.6.5 Cost information, reporting and decision-making	
8.6.6 Financial management system and cost accounting system	
8.6.7 Distinguishing between direct and indirect costs	
8.7 Cost management in the GOGCWS	266
8.7.1 Throughput accounting model	
8.7.2 Benchmarking	
8.7.3 Budgeting	
8.7.4 Activity based costing	275
8.7.5 Activity based management	276
8.7.6 Strategic Management Accounting (SMA)	277
8.7.7 The Consultants	278
8.8 Conclusion	

## Chapter Nine: Cost management and external environment

9.0 Introduction	
9.1 Data	
9.2 Data analysis one	
9.2.1 Cost information	
9.2.2 Company environment	296
9.3 Reliability and validity of construct	
9.3.1 Reliability	
9.3.2 Factor analysis	
9.4 Restructured data	322
9.4.1 Cost management	
9.4.2 The GOGCWS environment	
9.4.3 New factors of cost management and external environment	324
9.5 Data analysis two	
9.6 Discussion	
9.7 Conclusion	

## Chapter Ten: External and Internal Environments of the GOGCWS

10.0 Introduction	
10.1 The External environment	
10.1.1 Government policy	
10.1.2 Supply and Demand of water in Egypt	

10.2 Internal environment	350
10.2.1 Current cost practice in the GOGCWS	
10.2.2 Managers' perceptions of cost management	
10.3 Conclusion	

## **Chapter Eleven: Discussion**

11.0 Introduction	355
11.1 Cost Management	356
11.2 Explanation of the findings	359
11.2.1 Role of accounting in the GOGCWS	
11.2.2 The Unified Accounting System and Institutional theory	
11.2.3 Institutional imprinting	
11.2.4 The lack of change at the GOGCWS	
11.2.5 Rationale for choosing the Budget incremental model	
11.3 Conclusion	

## **Chapter Twelve: Conclusion**

.

.

12.0 Introduction	
12.1 An overview	
12.2 Contribution to knowledge	
12.2.1 The role of Cost management and Management accounting	
12.3 Limitations and future research	
12.3.1 Limitations 12.3.2 Future research	379 380
12.4 Reflections on the learning process	
• References	

## Appendix

## Appendix One

A1.1 Letters of the access for the GOGCWS	433
A1.2 The GOGCWS	
A1.2.1 The GOGCWS' law and decrees	
A1.2.2 Stages of production of fresh water	
A1.2.3 Organisation structure of the GOGCWS	
A1.2.4 The Unified Accounting System	

# Appendix Two

A2.1 The Questionnaire and Interview g	uide463
A2.2 Analysis of the Questionnaire	

.

## List of Tables

Table 2.1: Institutional Pillars and varying levels: Illustrative theorists	53
Table 3.1: Different views of Accounting and impact on its role	62
Table 3.2: The historical development of management accounting	73
Table 3.3: The Basic Components of the Transaction Analysis Framework	88
Table 4.1: Prior studies to ABC implementation	112
Table 5.1: Major characteristics in three phases	128
Table 6.1: Differences between quantitative and qualitative research	181
Table 7.1: Key differences between Egypt, U.K and France	221
Table 7.2: Accounting and economic costs in the water industry	227
Table 8.1: Perceived organisation of the GOGCWS	243
Table 8.2: Perceived control structure of the GOGCWS	244
Table 8.3: Perception of how control structure fits within the organisation structure	245
Table 8.4: Development methods of the cost accounting system	255
Table 8.5: Standards used for cost accounting system	256
Table 8.6: Adequacy of the cost accounting system	257
Table 8.7: Any revision of methods used for estimating and measuring product cost.	257
Table 8.8: Perception of measuring product cost	259
Table 8.9: Perception of competitive intensity of capital and financing in the	
GOGCWS	260
Table 8.10: Using data from cost accounting	262
Table 8.11: Using data from cost accounting in budget formulation	263
Table 8.12: Reporting purposes of cost accounting data	263
Table 8.13: Users of cost accounting reports	264
Table 8.14: Use of cost accounting reports in decision-making	264
Table 8.15: Integration between financial management system and cost accounting	
system	265
Table 8.16: Use of throughput accounting in the GOGCWS	267
Table 8.17: Use of benchmarking in the GOGCWS	267
Table 8.18: Use of Operational and Capital budgeting in the GOGCWS	
	268
Table 8.19: Description of the budgeting process in the GOGCWS	268 271
Table 8.19: Description of the budgeting process in the GOGCWSTable 8.20: Methods used for measuring product cost	268 271 275

Table 8.22: Use of ABM in the GOGCWS	276
Table 8.23: Description of the GOGCWS's strategy	277
Table 9.1: Descriptive analysis of the use of cost information for system efficiency	288
Table 9.1a: Descriptive analysis of the use of cost information for system efficiency	/ 289
Table 9.1b: Reported significance of cost information for system efficiency used at	the
GOGCWS	290
Table 9.2: Descriptive analysis of the use of cost information for system optimisation	on
	291
Table 9.2a: Descriptive analysis of the use of cost information for system optimisat	ion
Table 0.2h. Departed significance of cost information for system entimisation used	292 t
the COCCWS	ai 202
Table 0.2: Descriptive analysis of the use of cost information for system strategy	295 204
Table 9.5. Descriptive analysis of the use of cost information for system strategy	<i>2</i> 97
GOGCWS	 295
Table 9.4: P (CI) of the product market in the GOGCWS	296
Table 9.4a: P (Un) of the product market in the GOGCWS	297
Table 9.5: Measure of factor market	297
Table 9.6: P (CI) of raw materials and supply in the GOGCWS	298
Table 9.6a: P (Un) of raw materials and supply in the GOGCWS	298
Table 9.7: P (CI) of the labour in the GOGCWS	299
Table 9.7a: P (Un) of the labour in the GOGCWS	299
Table 9.8: P (CI) of the management in the GOGCWS	300
Table 9.8a: P (Un) of the management in the GOGCWS	301
Table 9.9: P (CI) of Capital and financing in the GOGCWS.	301
Table 9.9a: P (Un) of capital and finance in the GOGCWS	302
Table 9.10: P (CI) of Technology in the GOGCWS	303
Table 9.10a: P (Un) of technology in the GOGCWS	304
Table 9.11: P (CI) of the legislation in the GOGCWS	305
Table 9.11a: P (Un) of the legislation in the GOGCWS	305
Table 9.12: Alpha coefficients of Reliability for the items of cost management	307
Table 9.12a: Alpha coefficients of Reliability for factor analysis of P (CI) and P (Un	n) of
the GOGCWS environment	308
Table 9.13: Total variance explained for Using Cost Information for Efficiency	310

9
legislation) related to P (CI) and P (Un)
and perception of GOGCWS environment (product market, factor market and
Table 9.27a: The significance levels of the Pearson correlations between Cost practice
(Un)
environment (product market, factor market and legislation) related to P (CI) and P
Table 9.27: Pearson correlations between Cost practice and perception of GOGCWS
and perception of the GOGCWS environment related to P (CI)
Table 9.26a: The significance levels of the Pearson correlations between Cost practice
environment related to P (Un)
Table 9.26: Pearson correlations between Cost practice and perception of the GOGCWS
and perception of the GOGCWS environment related to P (CI)
Table 9.25a: The significance levels of the Pearson correlations between Cost practice
environment related to P (CI)
Table 9.25: Pearson correlations between Cost practice and perception of the GOGCWS
factor market and legislation
Table 9.24: New factors of cost management and P (CI) and P (Un) of product market,
factor market and legislation
Table 9.23: Factors of cost management and P (CI) and P (Un) of the product market,
Table 9.22a: Factor of P (Un) of legislation 321
Table 9.22: Factors of P (CI) of legislation
Table 9.21a: Factors of P (Un) of technology
Table 9.21: Factors of P (CI) of technology 319
Table 9.20a: Factor of P (Un) of capital and financing
Table 9.20: Factors of P (CI) of capital and financing
Table 9.19a: Factor of P (Un) of management
Table 9.19: Factors of P (CI) of management
Table 9.18a: Factors market of P (Un) of labour 317
Table 9.18: Factors of P (Cl) of labour     317       317     317
Table 9.1/a: Factors of P (Un) of raw materials and supply
Table 9.17: Factors of P (U) of raw materials and supply     310       Table 9.17: Factors of P (U) of raw materials and supply     310
Table 9.16: Factor of P (CI) in the product market
Table 9.15: Factors of Use of Cost Information for Strategy     314       Table 0.16: Easter of D (CL) in the product monitor     315
Table 9.14: Factors of Use of Cost Information for Optimisation
Table 9.13a: Factors of Use of Cost Information for Efficiency  312
$T_{-1} = 0.12$ as Eastern of Use of Cost Information for Efficiency (212)

Table 9.28: Pearson correlations between Cost practice and perception of GOGCWS
environment (product market, factor market and legislation) related to P (CI) and P
(Un)/high level
Table 9.28a: The significance levels of the Pearson correlations between Cost practice
and perception of GOGCWS environment (product market, factor market and
legislation) related to P (CI) and P (Un)/high level
Table 9.29: Pearson correlations between Cost practice and perception of GOGCWS
environment (product market, factor market and legislation) related to P (CI) and P
(Un)/medium level
Table 9.29a: The significance levels of the Pearson correlations between Cost practice
and perception of GOGCWS environment (product market, factor market and
legislation) related to P (CI) and P (Un)/medium level
Table 9.30: Pearson correlations between Cost practice and perception of GOGCWS
environment (product market, factor market and legislation) related to P (CI) and P
(Un)/low level
Table 9.30a: The significance levels of the Pearson correlations between Cost practice
and perception of GOGCWS environment (product market, factor market and
legislation) related to P (CI) and P (Un)/low level
Table 9.31: Pearson correlations between Cost practice and perception of the Chief of
Cost Department of GOGCWS environment (product market, factor market and
legislation) related to P (CI) and P (Un)/ Mean score - Accounting
Table 9.32: Pearson correlations between Cost practice and perception of the Vice
Chairman for Technical Affairs of GOGCWS environment (product market, factor
market and legislation) related to P (CI) and P (Un)/ (Mean score) - Technical337
Table 9.33: Pearson correlations between Cost practice and perception of the Chief of
General Chief of Meters Administration of GOGCWS environment (product
market, factor market and legislation) related to P (CI) and P (Un)/ (Mean scored) -
Administrative
Table 10.1: Perception of privatisation or regulation of the water industry in the
GOGCWS
Table 10.2: Purposes of water and the quantities for the year 1999/2000
Table 10.3: Managers Perception of Cost Information in the GOGCWS
Table A1.1: Accounts code of the level of general accounts
Table A2.1: The respondents of the questionnaire

Table A2.2: Component Matrix for Uses of Cost Information for Efficiency.	
Table A2.2a: Pattern Matrix for Uses of Cost Information for Efficiency	
Table A2.2b: Structure Matrix for Uses of Cost Information for Efficiency	494
Table A2.3: Rotated Component Matrix for Uses of Cost Information for Op	ximisation 495
Table A2.4: Structure Matrix for Uses of Cost Information for Strategy	
Table A2.5: Structure Matrix for the P (Un) of Product market	
Table A2.6: Structure Matrix for the P (CI) of Raw Material and Supply	
Table A2.6a: Structure Matrix for the P (Un) of Raw Material and Supply	
Table A2.7: Structure Matrix for P (CI) of Labour	498
Table A2.7a: Structure Matrix for P (Un) of Labour	498
Table A2.8: Structure Matrix for P (CI) of Management	
Table A2.8a: Structure Matrix for P (Un) of Management	499
Table A2.9: Structure Matrix for P (CI) of Capital and Financing	500
Table A2.9a: Structure Matrix for P (Un) of Capital and Financing	500
Table A2.10: Structure Matrix for P (CI) of Technology	
Table A2.10a: Structure Matrix for P (Un) of Technology	
Table A2.11: Structure Matrix for P (CI) of Legislation	502
Table A2.11a:Structure Matrix for P (Un) of legislation	502
Table A2.12: Pearson correlations -Level Matrixes for Uses of Cost Information	tion for
Efficiency	503
Table A2.12a: The significance levels of the Pearson correlations -Level Mar	trixes for
Uses of Cost Information for Efficiency	504
Table A2.13: Pearson correlations -Level Matrixes for Uses of Cost Information	tion for
Optimisation	
Table A2.13a: The significance levels of the Pearson correlations -Level Mat	trixes for
Uses of Cost Information for Optimisation	506
Table A2.14: Pearson correlations -Level Matrixes for Uses of Cost Information	tion for
Strategy	
Table A2.14a: The significance levels of the Pearson correlations -Level Mat	trixes for
Uses of Cost Information for Strategy	508
Table A2.15: Pearson Correlation of Level Matrixes for P (Un) of Product M	1arket 509
Table A2.15a: The Significance levels of the Pearson correlations of Level M	Aatrixes for
P (Un) of Product Market	

,	Table A2.16: Pearson Correlation of Level Matrixes for P (CI) of Raw Material and
	Supply510
,	Table A2.16a: The Significance levels of the Pearson correlations of Level Matrixes for
	P (CI) of Raw Material and Supply510
,	Table A2.17: Pearson Correlation of Level Matrixes for P (Un) of Raw Material and
	Supply
,	Table A2.17a: The Significance levels of the Pearson correlations of Level Matrixes for
	P (Un) of Raw Material and Supply511
,	Table A2.18: Pearson Correlation of Level Matrixes for P (CI) of Labour
,	Table A2.18a: The Significance levels of the Pearson correlations of Level Matrixes for
	P (CI) of Labour
,	Table A2.19: Pearson Correlation of Level Matrixes for P (Un) of Labour
-	Table A2.19a: The Significance levels of the Pearson correlations of Level Matrixes for
	P (Un) of Labour
•	Table A2.20: Pearson Correlation of Level Matrixes for P (CI) of Management514
•	Table A2.20a: The Significance levels of the Pearson correlations of Level Matrixes for
	P (CI) of Management
r	Table A2.21: Pearson Correlation of Level Matrixes for P (Un) of Management515
-	Table A2.21a: The Significance levels of the Pearson correlations of Level Matrixes for
	P (Un) of Management515
•	Fable A2.22: Pearson Correlation of Level Matrixes for P (CI) of Capital and Financing
•	Table A2.22a: The Significance levels of the Pearson correlations of Level Matrixes for
	P (CI) of Capital and Financing
5	Table A2.23: Pearson Correlation of Level Matrixes for P (Un) of Capital and
	Financing517
5	Table A2.23a: The Significance levels of the Pearson correlations of Level Matrixes for
	P (Un) of Capital and Financing517
	Table A2.24: Pearson Correlation of Level Matrixes for P (CI) of Technology
	Table A2.24a: The Significance levels of the Pearson correlations of Level Matrixes for
	P (CI) of Technology
-	Table A2.25: Pearson Correlation of Level Matrixes for P (Un) of Technology
5	Table A2.25a: The Significance levels of the Pearson correlations of Level Matrixes for
	P (Un) of Technology

Table A2.26: Pearson Correlation of Level Matrixes for P (CI) of Legislation
Table A2.26a: The Significance levels of the Pearson correlations of Level Matrixes for
P (CI) of Legislation
Table A2.27: Pearson Correlation of Level Matrixes for P (Un) of Legislation
Table A2.27a: The Significance levels of the Pearson correlations of Level Matrixes for
P (Un) of Legislation

.

# List of Figures

Figure 1.1: Flowchart for The structure of the thesis
Figure 2.1: Institutional and Giddens' theory
Figure 2.2: Institutional Pillars and varying levels: Illustrative schools
Figure 3.1: Decision-making and location of organisational uncertainty
Figure 3.2: The different roles of Management Accounting74
Figure 3.3: The relationship between rules and routines
Figure 3.4: The process of Institutionalisation
Figure 4.1: Bounding domains for purpose, effectiveness and efficiency
Figure 4.2: Activity-Based Management
Figure 4.3: Implementation steps of the process view of AMB
Figure 4.4: Models (tools) of cost management123
Figure 5.1: The relationship between cost management and the external environment156
Figure 6.1: Inductive approach159
Figure 6.2: Deductive approach160
Figure 6.3: Understanding case study research172
Figure 6.4: The hermeneutic process
Figure 7.1: The public utility and its related constituencies
Figure 7.2: Economic Circular Flow195
Figure 7.3: Current Payment Method228
Figure 7.4: A fairer alternative
Figure 8.1: The Structure of the GOGCWS242
Figure 8.2: Water Quality Control
Figure 8.3: Water quality control247
Figure 8.4: Contents of the Unified Accounting System
Figure 8.5: Organisational model of stages of budget preparation
Figure 8.6: Stages of operational budget preparation271
Figure 9.1: The relationship between cost management and the external environment283
Figure A1.1: Stages of production of fresh water446
Figure A1.2: Filtration stage
Figure A1.3: Refining stage
Figure A1.4: Organisation Structure of the GOGCWS450
Figure A1.5: Summary of the suggested solutions for The GOGCWS

Figure A1.5.1: Scenario of the Greater Cairo4	55
Figure A1.5.2: The plan of the organisational structure at the re-organisation stage 45	56
Figure A1.6: The environment of the GOGCWS45	57
Figure A1.7: The rules of transactions of the Unified Accounting System	58

#### Abstract

Water is a primary human need. The supply of water has not increased in the face of population growth. At present severe pressure exists on water supplies in California and along the river system both in the developed and developing world (especially in the case of the Nile in Egypt). Whatever our policy stance towards the economics of water supply (the liberal market, social democracy and democratic socialism), cost management and processes are central to the water sector.

The control or governing ideas within which cost management takes place are the two major concerns of this thesis. These two issues were explored in a case study of the General Organisation for the Greater Cairo Water Supply, focusing specifically upon government policy, administrative controls, the influence of public sector bodies and other customers. Control procedures, the perceptions and experiences of managers (regarding policy choice), cost management policies and practices, environmental demands (both market and physical), were all sought. Data was collected from multiple sources (triangulation) involving interviews, questionnaires, documentation, direct observation and participant observation.

The research findings showed that the practice of cost management has developed. Reasons were multi-fold: (a) To keep costs well above revenue (b) To engender a factor-resources cost and technical approach to cost management (for efficiency and optimisation) (c) To deflect any attention away from strategic cost management (d) To engender an organisational belief that ambiguity and unpredictability in the environmental market is impossible to handle (e) To deny the relevance of customer cost/usage efficiency as a matter of systemic significance.

This 'evolution of control' is explained through the theoretical context of institutional theory; demonstrating how values, beliefs and modes of regulation have produced a technical and passive non-reactive control system.

### Acknowledgements

The student is grateful to Allah for allowing him the opportunity to improve his academic knowledge and providing him with the strength and patience to accomplish it. I would like to acknowledge a deep debt of gratitude to my supervisor Professor Anthony Berry, who has at all times shown me great patience and generosity in his intellectual guidance and encouragement throughout this period of study. It is true that without Dr. Berry's invaluable feedback, this work would never have reached its present standard. I was fortunate lucky to be his student.

I acknowledge my appreciation to Professor John Cullen as my second supervisor. Professor Cullen assisted me with constructive suggestions and feedback which have improved my research. I am also grateful to Professors Scapens and Coad for their insightful comments which have enhanced this thesis.

I have benefited from discussions with participants in internal seminars within Sheffield Hallam University led by Professor Berry, also at the conferences and workshops I attended with the Management Control Association. I would like to thank all my lecturers during my MA and Ph.D. courses at Sheffield Hallam University for providing me with a good academic background on which to build. I am especially obliged to my MA supervisor, Professor Anthony Rosie for his advice which led me to embark on this Ph.D. study. Special thanks are also due to the staff in The General Organisation for the Greater Cairo Water Supply who participated in the initial fieldwork. I should like to thank all my friends who provided strong and continuous support while I undertook this research.

I gratefully acknowledge the financial support given to me by the Institute of National Planning in Egypt and the support of the Cultural Attaché in London. Finally this acknowledgement would be incomplete without the thanks owed to my dear parents, my wife, my sisters and brothers for their considerable patience and consistent encouragement. I owe a great deal to my parents and indeed I feel very proud in dedicating this dissertation to them.

### List of Abbreviations

ABC	Activity-Based Costing
ABM	Activity-Based Management
AHD	Aswan High Dam
BPR	Business Process Re-engineering
CVF	Causal Variability Factors
CIGS	Central Institution for General Statistical
CMS	Cost Management System
DTI	Department of Trade and Industry
E. P.	Egyptian Pound
FAC or AC	Full Absorption Costing
FDCi	Fully Distributed Cost of Service I
GOA	Government of Egypt
GOGAWS	The General Organisation for the Greater Alexandria Water Supply
GOGCWS	The General Organisation for the Greater Cairo water Supply
IBRD	International Bank for Reconstruction and Development
ICWP	Inter-Ministerial Committee on Water Planning
JIT	Just In Time
MACS	Management Accounting Control System
MAS	Management Accounting System
MHPU	Ministry of Housing and Public Utilities
MPWWR	The Ministry of Public Works and Water Resources
MTSS	Management and a Training and System Strengthening
OFWAT	Office of Water Service
OIE	Old Institutional Economic theory
P (CI)	Perception of Competitive Intensity
P (Un)	Perception of Unpredictability
PA	Principal-Agent Theory
SBU	Strategic Business Unit
SCM	Strategic Cost Management
SMA	Strategic Management Accounting
ТА	Throughput Accounting

·

•

TFA	Total Factor Cost
TQM	Total Quality Management
UAS	The Unified Accounting System
USAID	The U.S. Agency for International Development
VC	Value Chain
<b>V. E.</b>	Value Engineering
ZBB	Zero-Based Budgeting

·

.

## Chapter 1 Introduction

### **1.0 Introduction**

This thesis offers an explanation for the role of cost management in the Egyptian water industry. Chapter one outlines the contributory value of this thesis and links it to wider research in this area (see sections 1.1 and 1.2). It also explains reasons for the author's interest in this topic, and the rationale for choosing it (see section 1.3).

The chapter outlines the theoretical framework (see section 1.4) and the research methodology and method (see section 1.5) used in the research. It gives a brief synopsis of the contents of chapters in this thesis (see section 1.6).

### 1.1 Background

This research investigates the role of cost management in the water industry. There are several definitions of cost management. Brimson (1996: 104) argues that cost management is the "management and control of activities to determine an accurate product cost, improve business processes, eliminate waste, identify cost drivers, plan operations, and set business strategies". Horngren, et al., (1999: 875) argue that cost management is the "actions of managers undertaken to satisfy customers while continuously reducing and controlling costs". The Chartered Institute of Management Accountants defines cost management as follows: "Cost management is the application of management is the information required to enable costs to be planned, monitored and controlled" (CIMA, 2000: 27).

Scapens (1991) notes that whilst there is a large and ever-growing 'tool-chest' of modern management accounting techniques, (including cost-volume-profit analysis, activity-based costing etc) "little can be said at the theoretical level about the techniques (tools) which should generally be used in practice" (p. 219). He goes on to argue that before the gap between theory and practice can be closed, "researchers must examine the various roles which management accounting fulfils within the organisation" (p.

221). He goes on to argue that there is little' evidence of how different costing methods are being used. There is much less research on why only one method is used and why one method is preferred over another. Similarly there is a dearth of research on how accounting practices have developed over time and on the sources (institutions) that have influenced that development. In addition Simons (1992: 44) raised the question "What role can accounting play in stimulating emergent strategies?" Scapens (1991: 10) contends that "the terms 'cost accounting' and 'management accounting' now tend to be used synonymously in textbook titles".

In cost management literature, evidence of renewed interest was witnessed from the reaction to Johnson and Kaplan (1987) who argued that the systematic structures of cost accounting did not make much of a connection with the systematic problems of managing an enterprise. They also contend that the cost management accounting systems (which do support financial reporting requirements) are difficult to justify economically. A great deal of recent cost management literature has centred around activity-based costing (Banker and Hughes, 1994; Cooper and Kaplan, 1992; and Datar and Gupta, 1994). However other modern cost management tools such as 'benchmarking' (Elnathan and Kim, 1995) and just-in-time inventory systems (Alles, et al., 1995) have also recently been examined.

Cost management has three dimensions to it: Efficiency, optimisation and strategy (Al-Hazami, 1995). Most researchers concentrate on efficiency (Thompson, 1967; Miller, 1989; Williams and Giardina, 1993; Ezzamel, et al., 1994; Dugdale, et al., 1996; Horngren, et al., 1999)<sup>2</sup>. Others focus on optimisation<sup>3</sup> (Turney, 1991b; Anotos, 1992; Cooper, et al., 1992; Harmmer and Champy, 1993; Brimson, 1996; Smith, 1997; Trussel and Binter, 1998; 1998; Rouse and Putterill, 2000) and strategy<sup>4</sup> (Ansoff, 1965; Anthony, 1965; Mintzberg, 1978; Miles and Snow, 1978; Simmonds, 1981; 1982; 1985; 1986; Simons, 1987; 1990; 1995; Bromwich, 1988; 1990; Dent, 1990 and Johnson and Scholes, 1993).

 $<sup>^{1}</sup>$  With the exception of a few studies (Ask and Ax, 1992; Drury and Tayles, 1994; 2000; Innes and Mitchell, 1995).

<sup>&</sup>lt;sup>2</sup> See section 4.1.

<sup>&</sup>lt;sup>3</sup> See section 4.4.

<sup>&</sup>lt;sup>4</sup> See section 5.2.

Strategic management accounting has attracted attention from numerous authors to its different aspects. For instance, Johnson and Kaplan, 1987; Simmonds, 1981; 1982; 1985; 1986; Bromwich, 1988; Allen, 1985; and Taylor and Graham, 1992; all emphasise the extension beyond management accounting's internal focus to include, for example external information about competitors, see section 5.2.1). There has been extensive and rapidly growing literature on strategic management accounting (Simmonds, 1981; Shank, 1989; Bromwich, 1990; Dent, 1990; Wilson, 1991; Hartman, 1993).

Some researchers focused on business strategy, - identifying the relationship between the strategic position chosen by a firm and the expected emphasis on management accounting<sup>5</sup> (see section 5.2.2). Others advocate an analysis of ways to decrease costs and/or enhance differentiation of firm products, through exploiting linkages in the value chain and optimising cost drivers<sup>6</sup> (see section 5.2.3).

### **1.2 Objectives and research questions**

The main objectives of this study are to understand, describe and explain the role of cost management in the water industry. The focus of this study is the service water industry in Egypt. This research will study the problems of cost management which have arisen in the Egyptian water system. Cost management will be examined as a whole system, from different viewpoints, including accounting (managerial and economic perspective), and Government policy (social and political perspective).

Taking the main objective and the meaning of the term 'cost management', questions to be addressed in this research include:

<sup>&</sup>lt;sup>5</sup> Ansoff, 1965; Anthony, 1965; Miles and Snow, 1978; Andrew, 1971; Kaplan and Norton, 1992; 1996; Mintzberg, 1973; Dent, 1990; Goold and Quinn, 1990; Govindarajan and Gupta, 1985; Simons, 1987; Shank, 1989; Shank and Govindarajan, 1989; Porter, 1980; 1985; 1990; Gray, 1990; and Hartmann, 1999.

<sup>&</sup>lt;sup>6</sup> Hargert and Morris, 1989; Shank and Govindarajan, 1992b; Nanni, et al., 1992; Harland, 1996; Ellram, 1991; Williamson, 1985; Berry, et al., 1997; Seal, et al., 1999; Slack, et al., 1998; and Rajagopal and Bernard, 1993.

- What is the current cost structure for the provision of Egyptian water services? (descriptive and illustrative)<sup>7</sup>
- What is the rationale of the companies for the cost structure? (explanatory)<sup>6</sup>
- What methods of cost accounting were/are used in the Egyptian water industry? (descriptive and illustrative)<sup>6</sup>
- What is the role of the accounting (its nature and function) in controlling the water industry? (descriptive and explanatory)<sup>6</sup>

### 1.3 The motivation of the study

The motivation to research this area of study arose from a desire to understand the contribution of cost management to the complex task of decision-making. Many researchers draw attention to the importance of cost management. Rajaopal and Bernard (1993) claim that cost management draws on a number of functions but concentrates on total quality management, negotiating, and supplier alliances. Horngren (1995) argues that the focus of cost management should be on decision making, while Cooper (1996) argues that the growing importance of cost management is significantly changing the practice of management accounting. Seal, et al., (1999) argue that cost information not only plays a role in the strategic sourcing decision but will also influence the ongoing management of partnerships. In a strategic approach, the ongoing management of a strategic partnership should be on cost rather than a price basis. This thesis explains cost management in the Egyptian Water industry.

Young and Selto (1991: 288) in their review of research in new manufacturing practices and cost management stated that "In order to have a significant impact on either the academic or practitioner community, cost management research will have to assess the economic, behavioural and social consequences of new manufacturing practices and new cost management methods". Durden and Mak (1999) argue that the focus of a contemporary costing system should be on cost management, not extended reporting. They argue that if cost management is the primary purpose of a costing system, then it is appropriate to report overhead variances as a period cost, rather than pro-rating these variances to various accounts. The researcher's view is that the future of the water

<sup>&</sup>lt;sup>7</sup> A term identified by Scapens (1990).

industry in Egypt will potentially change focus from cost management of supply to cost management of demand.

A second motivation was that the future looks bleak if Egypt does not succeed in formulating and implementing a water management policy which can match limited fresh supply with increasing demand. Per capita water resources were expected to drop from about 922  $m^3$  per year (in 1990) to about 337  $m^3$  per year in the year 2025. This will help change the focus from cost management of supply to cost management of demand. Hvidt (1998) in his research identified eight characteristics from his analysis of water resource planning in Egypt. They were: (a) a shift from water abundance to water deficit, (b) the importance of international co-operation (c) supply bias (d) environmental concern (e) lack of data (f) established priority to non-agricultural uses of water (g) delayed implementation and (h) an uncertain administrative framework in water resource planning.

### **1.4 Theoretical framework**

Institutional theory is adopted as a framework for explaining the role of cost management in the Egyptian Water Industry. There are differing views of institutional theory; from the economic, political and sociological perspectives. Each of these dimensions has been used by previous studies to examine different management accounting issues. The core idea of each view is discussed later in the thesis (see section 2.2).

This thesis uses neo-institutional economic theory to explain the operation of cost management in the GOGCWS (see section 11.2). The study examines how government policy (a coercive institutional pressure) influenced the choice of accounting system in the GOGCWS. The justification for the choice of neo-institutional economic theory is fully discussed later in the thesis. However, the theory offers a better explanation for cost management practices in the GOGCWS, which operated in an unchanging and relatively static environment. The thesis also offers alternative governance structures for the water industry in different countries from a neo-institutional economic<sup>8</sup> theory perspective (such as the UK where water is provided by a highly regulated market,

<sup>&</sup>lt;sup>8</sup> According to neo-institutional economic theory (transaction cost), there are three models of governance: (1) markets (2) hybrids and (3) hierarchies or internalisation (Williamson, 1996).

France where water services are a hybrid structure and Egypt where water services are a hierarchical (see sections 7.2.5 and 10.1.1).

There are significant differences between the neo-classical economic theory<sup>9</sup> and the neo-institutional economic theory. Perhaps the most significant differences for this research is related to the first element of cost management (efficiency). Neo-classical economic theory defines cost management in three ways: allocative efficiency or pareto efficiency, productive or technical efficiency and dynamic or interemporal efficiency. In contrast, neo-institutional economic theorists have expressed strong reservations about using the pareto efficiency criterion to justify government intervention. They also argue that by comparing real-world arrangements against the ideal of allocative efficiency rather than feasible institutional alternatives, policy-makers have become far too inclined to prescribe government intervention.

There are two views of management accounting within institutional research. The first view sees management accounting as an institution within the organisation (Burns and Scapens, 1998; 2000). The second view shows accounting as primarily concerned with the effects of the external or macro-institutional, (i.e. effect of the social, economic and political on accounting practices of organisations, Covaleski, et al., 1993; 1996; Carruthers, 1995).

This research takes into consideration the two views of management accounting within institutional research (see sections 11.2.2 and 11.2.3). Thus the accounting system (the unified accounting system) is an institution within the GOGCWS (see section 8.5) and also institutional pressures exerted by the Egyptian government shapes accounting practice (the UAS) in the GOGCWS (see chapters 9, 10 and 11).

The explanations of this research are centred around two propositions: (a) Government policy (coercive isomorphism) shapes the limited use of cost management for efficiency, optimisation and strategy (b) Organisations are the 'theatre' in which institutions are visible. In other words, the explanation of this research takes into consideration the institutional functionalist view (first proposition) and the institutional interpretive view (second proposition)

<sup>&</sup>lt;sup>9</sup> For the criticism of neo-classical economic theory (see sections 3.1 and 3.3.1).

Institutional theory adds a broader (extra-organisational) institutional dimension to the analysis and discussion of this thesis. Burns and Scapens (2000) argue that in order to understand micro-processes it is necessary to recognise the institutional context both within the organisation itself, (i.e. the organisation's rules, routines and institutions) and outside, (i.e. the broader social, economic and political institutions of the organisational field and the society in which the organisation operates).

As mentioned in section 1.1, cost management has three elements to it: efficiency, optimisation and strategy (Al-Hazami, 1995). Most researchers concentrate on efficiency (see sections 4.1, 4.2 and 4.3). Others focus on optimisation (see section 4.4) and strategy (see section 5.2). This research incorporates all three elements of cost management. Pfeffer and Salancik (1978) argue that a good deal of organisational behaviour can be understood by knowing something about the organisation's environment<sup>10</sup> and the problems it faces in obtaining resources. What happens in an organisation is not only a function of its internal structure, leadership, procedures, or goals, it is also a consequence of the environment, and the particular contingencies and constraints from that environment. Therefore, this research focuses on institutional context within and outside the GOGCWS.

#### **1.5 Research Methodology and Method**

Methodology refers to the methods and ways used to conduct research. Blaikie (1995) states that methodology is essentially epistemology. He argues that epistemology is a theory of knowledge. It presents a view and justification for what should be regarded as 'knowledge' - (i.e. what can be known, and what criteria such knowledge must satisfy in order to be termed 'knowledge' rather than 'beliefs'). In short, epistemology is concerned with the nature and forms of knowledge; how knowledge is acquired and the relationship between ideas about the world and the world itself. Epistemological debates are concerned with the appropriateness of procedures and methods in the process of knowledge acquisition. A typical epistemological debate would focus on induction and deduction<sup>11</sup>.

<sup>&</sup>lt;sup>10</sup> This research argues that the environment of the GOGCWS could explain three main factors: product market, factor market and legislation.

<sup>&</sup>lt;sup>11</sup> For definitions of 'deductive' and 'inductive' approach, see section 6.1.1.

This study adopts a hybrid approach, focusing more on the inductive strand. The rationale for choosing a hybrid approach is because the objectives of this research are to describe what is going on (i.e. to describe the current cost structure for the provision of water in Egypt); and to understand and explain why it has been put into effect, (i.e. the companies rationale for choice of cost structure). Explanation therefore involves induction and deduction. DeVaus (1996: 11) claims that inductive and deductive approaches are not alternative ways of arriving at good theories; rather they represent two stages with different starting points. The research also takes a 'voluntaristic'<sup>12</sup> view of human behaviour rather than one which guides contingency research (Child, 1972; Schreyogg, 1980). The voluntaristic view is noticeably absent in early neo-institutional research (DiMaggio, 1988 and Oliver, 1991). This approach followed from more recent advances in institutional theory which call for more-in-depth, interpretive analyses of the dialectic between stability and change (Czarniawska and Servon, 1996).

A case study method is used in this research. Strengths and weaknesses of the case study method (see Miles, 1979) have been addressed since the late 1970s. There is an increasing body of literature on how to conduct and analyse case-study data (Campbell, 1975; McClintock, et al., 1979; Yin, 1981; 1989; Eisenhardt, 1989; Scapens, 1990; Gummesson, 1991; Rose, 1991; Ryan, et al., 1992; Otley and Berry, 1994; Hartely, 1995).

This case study research will focus on three objectives: descriptive, illustrative, and explanatory (Scapens, 1990). These objectives will assist the researcher in answering his research questions (see section 1.2).

This research will be based on a case study of one company in Cairo, - the General Organisation for the Greater Cairo Water Supply. Reasons for choosing this company were several. Firstly, it was the first company to be established in the water industry in Egypt. It therefore helps describe and explain the role of cost management in the Egyptian Water Industry (because it represents the past, present and future experience of Water Service in Egypt). Secondly it provides water for 25% of the population in

<sup>&</sup>lt;sup>12</sup> 'Voluntaristic' view assumes that human beings are free-willed, act independently of external stimuli and exercise freedom of choice (Burrel and Morgan, 1979).

Egypt. It also has a unique<sup>13</sup> laboratory for analysing water in the Middle East (The cost of this laboratory was 25 million Egyptian pounds<sup>14</sup>).

Thirdly there are discrepancies between the expenses of the GOGCWS and its income. The cost per  $m^3 = 0.46$  pence (wages 8.03 + Power and electricity 6.99 + raw materials 4.47 + other current expenses 7.00 + interest "local and international"  $11.29 + Depreciation 8.19)^{15}$ . By contrast, average income was per  $m^3 = 0.16$  pence, so this price does not encourage efficiency. Finally it is estimated that the General Organisation for the Greater Cairo Water Supply loses 40% of water through leakages each year. This results in an estimated wasteful expenditure of 250 million Egyptian pounds every year. Efficiency in the company is therefore low.

Data will be collected from multiple sources (triangulation) including interviews (qualitative), questionnaires (quantitative) and company documentation. The rationale behind this integration is that any limitations of one method will be counter-balanced by the strength of another (Hoque and Hopper, 1994).

### **1.6 Structure of the thesis**

This thesis has eleven chapters. The aim of chapter two is to outline neo-institutional economic theory as the theoretical basis of the research. The chapter argues that both contingency theory and institutional theory are important for explaining the design and use of cost accounting systems. It also argues that while technical (or contingent) explanations of cost management are not rejected, they are clearly viewed as incomplete (Scott, 1987). Institutional theory adds the social and political elements, which are typically absent or de-emphasised in the rational instrumental approach (which guides most of the recent research on cost management). The chapter examines the theories used in accounting research, and offers an overview of management accounting. The

<sup>&</sup>lt;sup>13</sup> The GOGCWS, (no date source given, c) - "Greater Cairo Water", The Administration of Public Relations.

<sup>&</sup>lt;sup>14</sup> N.B: English pound  $\pounds = 6.00$  Egyptian pounds. American Dollar \$ = 5.00 Egyptian pounds.

<sup>&</sup>lt;sup>15</sup> The GOGCWS (no date source given, c) - "Greater Cairo Water", The Administration of Public Relations.

chapter also clarifies the level (organisation level) and approach (a process approach) which this study has adopted.

Chapter three aims to explain the role of management accounting by examining the theory behind it. It discusses the role accounting has come to occupy in micro-level (organisations) and macro-level (societies). This chapter gives an overview of management accounting theory. It criticises and shows the limitations of management accounting from a neo-classical economic stance. The chapter outlines some alternative approaches to management accounting, (namely neo-institutional economic theory, old institutional economic theory and neo-institutional sociology theory). Furthermore it has argued that the institutional theory approach has the potential to provide a useful theoretical framework for understanding management accounting (cost management) practices. The chapter argues that institutional theory offers a useful theoretical framework for understanding management accounting (and cost management) practice.

Chapters four and five focus on the model of cost management which this research has adopted. This research is based on a model of cost management which has three elements: efficiency, optimisation and strategy. Chapter four focuses on the first two elements of cost management: efficiency and optimisation. It outlines the problem and discusses the development of cost management. The chapter then identifies some models of cost management which indicate how to increase efficiency (i.e. throughput accounting, benchmarking, target costing and ABC) and others to increase optimisation (such as, ABM). The chapter identifies links between institutional theory, transaction cost and efficiency. It examines how the efficiency framework is favoured by transaction cost theories. Chapter five identifies strategic management accounting as one of the models of cost management which considers both the internal and external environment (the third element of cost management). The chapter focuses on cost accounting in service organisations and discusses how it can support corporate strategy.

The aim of chapter six is to outline the research design and methodology. The methodological position adopted in this research is a hybrid approach (i.e. between inductive and deductive), but mainly an inductive approach (for the rationale for choosing a hybrid approach, see section 6.1.1). The research also took a nominalist view (ontology - the rationale for which can be found in section 6.1.2). It also takes a

'voluntaristic' view of human behaviour rather than one which guides contingency research (Child, 1972; Schreyogg, 1980). The chapter also clarifies the methods of data collection and the rationale for choosing them. A case study approach is adopted in this research. The case study is both a process of induction and deduction. The methodological position led the researcher to use multiple sources for collecting data (including semi-structured interviews, questionnaires and company documents of the GOGCWS).

Chapter seven offers alternative governance structures from the neo-institutional economic perspective for the water industry in different countries (such as the UK where water is provided by a highly regulated market, France where water services are hybrid and Egypt where water services are hierarchical structure). The chapter discusses the importance of water in order to explore the phenomenon of privatisation in the water industry. It looks at its effect on the role of accounting in general, and cost management in particular. The chapter considers the pros and cons of public versus private ownership of water. Finally, the issue of costs in the water industry is explored (i.e. how do people pay for their water requirements, how is the cost of water calculated, is the current system fair or is a new system required?).

Chapters eight and nine focus wholly on the case study of the General Organisation for the Greater Cairo Water Supply (the GOGCWS). Chapter eight describes the accounting system used at the GOGCWS and examines its supervision and control. It explores the questions: "Which methods of cost accounting were used in the Egyptian water industry?" What model of cost management was followed in the GOGCWS?" "Are there any similarities or differences between rules and routines<sup>16</sup> of the cost accounting system at the GOGCWS?" The chapter also discusses alternatives to cost management as suggested by consultants Black and Veatch International. It also discusses perceptions of the GOGCWS managers' towards the issue of cost management. Chapter nine explores the question: "Was there a relationship between cost management<sup>17</sup> and P (C.I)<sup>18</sup> and P (Un)<sup>19</sup> which relates to the three areas of the external environment

<sup>&</sup>lt;sup>16</sup> Rules and Routines used by Burns and Scapens (2000) see rules as comprising of accounting systems as set out in procedure manuals. 'Routines' are the actual accounting practices in use.

<sup>&</sup>lt;sup>17</sup> The objectives of cost management are efficiency, optimisation and strategy

(product market, factor market and legislation) from the neo-institutional economic theory perspective?

Chapter ten aims to discuss the external and internal environment of the GOGCWS. It also examines the perceptions of the GOGCWS managers of the issue of privatisation in the water industry and cost management. The chapter highlights the importance of neoinstitutional economic theory as theoretical framework for this research because it focuses on both the external (Covaleski, et al., 1993; 1996; Mezias, 1990) and internal environment (Burns and Scapens, 2000).

Chapter eleven aims to discuss the GOGCWS research findings in the light of the theoretical framework which was adopted. It also relates the findings back to the literature review. The chapter explains why there was no change in cost management at the GOGCWS by adopting two propositions from the neo-institutional perspective. Firstly, government policy (coercive isomorphism)<sup>20</sup> shaped the limited use of cost management for efficiency, optimisation and strategy. The second proposition assumed organisations are the 'theatre' in which institutions are visible. It examines the accounting system of the GOGCWS (the UAS) from an institutional perspective (the three pillars of institutional theory: cognitive, normative and regulative, Scott, 1995).

The final chapter aims to summarise the results of the research and identify its contributions to knowledge. Some contributions relate to cost management and management accounting while others relate to institutional theory and the case study approach. The chapter discusses the limitations of this study and offers suggestions for future research. It concludes with a discussion on the research process and the learning experience.

The structure of the thesis is shown in figure 1.1 as below:

 $<sup>^{18}</sup>$  P (C.I) refers to the perception of competitive intensity, (i.e. the general level of market conditions which make profitability difficult).

<sup>&</sup>lt;sup>19</sup> P (Un) refers to the perception of unpredictability, (The overall degree of variability from period to period, which make forecasting and planning difficult).

<sup>&</sup>lt;sup>20</sup> Although, the term isomorphism is generally associated with the neo-institutional economic sociology, it is equally applicable to any analysis of neo-institutional they in general (such as neo-institutional economic theory).



### **Figure 1.1: Flowchart for The structure of the thesis**
# Chapter 2 Institutional theory

# 2.0 Introduction

The aim of this chapter is to outline the theoretical basis of the research and the rationale for choosing it. For this purpose, the chapter is divided into four sections.

The first section gives the reasoning behind institutional theory as a choice of framework. It highlights the differences between contingency theory and institutional theory. It shows the different approaches to institutional theory and accounting research.

The second section introduces questions arising out of institutional theory. It outlines the core ideas of those schools and models which study institutions. It then gives a brief review of early institutionalist and neo-institutional theory in economic, political and sociology. It goes on to introduce the theories used in accounting research and offers an overview of management accounting within institutional theory.

The third section explains concepts which have been applied to institutions and isolates management accounting within these concepts. It highlights the similarity between structuration theory and institutional theory. It also illustrates levels of analysis in institutional theory (i.e. the two approaches in order to explain institutions). It then clarifies the level and approach which this study has adopted.

The final (fourth) section outlines the main conclusions of this chapter.

# 2.1 The rationale for choosing institutional theory

Why is a theoretical framework important? Coase<sup>21</sup> (1983: 230) argued that "without a theory there is nothing to pass on except a mass of descriptive material, waiting for a theory, or a fire". Institutions often play important roles in the diffusion of ideas (Bjornenak, 1996). In some cases they act as propagators, in others as moderators of change. It is therefore reasonable to expect cost practices to be influenced by those institutions involved in setting standards for the profession (e.g. professional bodies, academic institutions and major companies).

As part of the growing pluralism in the study of organisations, new sociological paradigms have emerged since the 1970s. Like economics, they offer explanations of organisational structure, - additional to those available in structural contingency theory - (Penning 1992; Davis and Powell 1992). These include the theories of 'resource dependence' (Pfeffer and Salancik 1977), 'Institutional' (Powell and DiMaggio, 1991), 'population ecology' (Hannan and Freeman, 1989), 'agency' (Jensen and Meckling, 1976) and 'transaction cost economics' (Williamson, 1985).

Subsequently, many new approaches have arisen including institutional theory in the US (Meyer and Scott 1983a) and action theory in the UK (Silverman 1970). Drury and Tayles (1995) called for different approaches and perspectives, rather than a wholly economic perspective of management accounting. They advocate "explaining observed practices by examining their role within the broader organisational, social, political and cultural dimensions in which accounting information is used" (p. 278). Brignal (1997) argues that service costing systems must be seen in their wider context (as part of management information systems which embrace non-financial information).

Geiger and Ittner (1996) claim that contingency and institutional theories can explain the design and use of cost accounting systems in government agencies. This research has adopted institutional theory as a framework for explaining the role of cost management in the Egyptian Water Industry. Despite the many advantages of

<sup>&</sup>lt;sup>21</sup> The 'godfather' of the new institutional economic theory.

contingency theory<sup>22</sup>, it was not adopted in this research. Reasons for this are two-fold: theoretical and practical. At the theoretical level, Bourgeois (1984) criticises contingency research for failing to consider reverse causation (where the presumed contingency factor results from the structure). The positive correlation between strategy and structure may have arisen through structure causing strategy.

Merchant (1984) contends that contingency theory can be criticised on the grounds that organisational context is often a construct of the researcher rather than a description of perceived reality as used by the organisations under study. Spekle (2001) claims that a contingency theory approach is more a vague idea than an actual theory. He contends that contingency theory has no a priori intuition of its own as to what the pertinent factors are and to their likely consequences. He goes on to argue that contingency modes tend to be partial, - focusing as they do on elements of the control system (e.g. budgeting systems), rather than directly addressing the full configuration of control devices.

At the practical level, under contingency theory the type of management accounting system (MAS) varies according to the specific circumstances or situation in which the organisation operates (see section 5.2.2). Hence, choice of MAS design is constrained by conditions and depends on the ability of management to find the best "fit". This is not applicable to the General Organisation for the Greater Cairo Water Supply because the GOGCWS is an economic organisation within the public utilities sector. The GOGCWS must therefore apply the Unified Accounting System which includes a management accounting system. In other words, the GOGCWS has no alternative in its choice of management accounting system.

<sup>&</sup>lt;sup>22</sup> Otley (1995: 191) claim that "the design of any planning and control system is situationally specific. When designing a system, a distinction is made between controllable and non-controllable factors. Controllable factors of the organisation are not considered contingent variables, rather they are part of a package of organisational controls selected for use. The organisation is expected to adapt to the contingency it faces, by arranging these suitable factors. It controls appropriate configuration that it hopes will lead to an effective control system. Interdependence between accounting information systems and components of contingency elements are still possible".

Berry et al., (1995: 22) argue that "contingency theory provides a picture for the organisation through its environment. The environment is crucial in shaping the organisation, and the survival of an organisation depends on its 'fitness' for the changing environment". Donaldson (1995) argues that while organisational theories can supplement contingency theory, the latter nevertheless remains the core explanatory theory of organisational structure.

While technical (or contingent) explanations of cost management are not rejected, they are clearly viewed as incomplete (Scott, 1987). Institutional theory adds the social and political elements which are typically absent or de-emphasised in the rational instrumental approach (which guides most of the recent research on cost management). Bjornenak (1997) argues that there are several explanations underlying characteristics of cost practice, and that an institutional frame of reference may be important for understanding how and why changes in cost practice occur. Contribution to knowledge in the field of cost management (building on contingency theory) remains fragmented. However more inclusive approaches are required before we can fully understand cost management (Spekle, 2001).

In recent years there has been increasing interest in institutional theories within the social sciences (Scott, 1995). Three such theories have been used in accounting literature namely neo-institutional/transaction cost economics<sup>23</sup> (derived largely from the work Williamson, 1975), old institutional economics<sup>24</sup> (Scapens, 1994), and neo-institutional sociology (Miller, 1994). These will be discussed in detail in section 3.3).

Previous governmental accounting research (based on economic theory) generally ignores how institutional and organisational pressures constrain accounting choice (i.e. the method of cost management in the water industry, Carpenter and Feroz, 2001). This research argues that institutional theory offers the best explanation for the role of cost management in the Egyptian Water industry.

## **2.2 History of institutional theory**

Institutional theory has been used here as the framework for explaining the role of cost management. The theory has raised some contentious questions about organisational 'worlds' (Scott, 1995). These include:

<sup>&</sup>lt;sup>23</sup> New institutional economics focuses primarily on micro-analytic questions (e.g. the comparative efficacy with which alternative generic forms of governance-market, hybrids and hierarchies economise on transaction costs). It avoids the broader questions of origins and change in the "institutional rules of the game: customers, laws, politics" (Williamson, 1991: 269).

<sup>&</sup>lt;sup>24</sup> There is similarity between Giddens' theory and old institutional economic theory (see figure 2.1). Burns and Scapens (2000) argue that structuration theory is not particularly helpful for exploring processes of change. Archer (1995) supports this view arguing that structuration theory does not incorporate historical time. She contends that "structure and agency can only be linked by explaining the interplay between them over time. Without the attention paid to the time factor, the problem of structure and agency can never be satisfactory resolved" (p. 65).

(a) Why organisations of the same type, (located in scattered localities) closely resemble one another (e.g. schools and hospitals).

(b) How should we regard behaviour in an organisational setting? Does it reflect the pursuit of rational interests and the exercising of conscious choice; or is behaviour primarily shaped by conventions, routines, and habits?

(c) Why does observed behaviour of organisational participants often depart from the formal rules and stated goals of the organisation?

(d) Why and how do laws, rules, and other types of regulative and normative systems arise? Do individuals voluntarily construct rule systems which then influences their own behaviour?

(e) Where do interests come from? Do they stem from human nature, or are they culturally constructed?

(f) Why do specific structures and practices diffuse through organisations in ways not predicted by the adopting organisations?

(g) How do differences in cultural belief shape the nature and operation of organisations?

(h) Why do organisations and individuals conform to institutions? Is it because they are rewarded for doing so, because they are morally obliged to obey, or because they can conceive of no other way of behaving?

There are several models which attempt to study institutional theory. These include The Columbia school, Parson's approach, The Carnegie school and The Cognitive school. It is useful here to highlight the key points of each model in turn.

In the Columbia school Merton (1936, 1957) and Selznick (1948) laid the groundwork for a process model of institutions. Merton described 'process-operating' in bureaucratic organisations as encouraging officials toward over-conformity. Selznick focused on processes within particular organisations as giving rise to a distinctive set of valued commitments. Stinchcombe (1968) claims the mechanisms used by powerful actors perpetuate their own interests and commitments. Stinchcombe<sup>25</sup> (1968: 107) adds that an institution can be defined as "a structure in which powerful people are committed to certain values or interest".

<sup>&</sup>lt;sup>25</sup> Stinchcombe is a scholar of the Columbia school and a student of Selznick's.

Parsons (1960a: 20) develops his "cultural-institutional" argument by examining the relationship between an organisation and its environment. He looks at the ways in which the value system of the organisation is legitimated by its links to "main institutional patterns" in "different functional contexts". Parsons (1960b) offers a different definition of the term 'institution'. He argues that organisations become vertically differentiated into three distinct levels or 'layers': (1) Technical (production activities) (2) the Managerial (concerned with control and co-ordination activities, the procurement of resources and the disposal of products) and (3) Institutional (relating the organisation to the norms and conventions of the wider community and society). Parsons concludes that every organisation is a subsystem of "a wider social system which is the source of meaning, legitimation, or higher-level support which makes implementation of the organisation's goals possible" (pp. 63-64).

As a member of the Carnegie school Simon (1957) was one of the first theorists to link individual cognitive capacity with organisational structure. He notes how organisational structures work to simplify and support decision-making in organisations, - allowing them to achieve higher levels of consistent "boundedly rational" behaviour.

March and Simon (1958: 141-142) argued that in many circumstances, "search and choice processes are very much abridged. Much of the behaviour in organisations is governed by performance programs". Hence value assumptions, cognitive frames, rules and routines are all ingredients conducive to individuals behaving 'rationally'. The "rational individual is organised and institutionalised" (Simon, 1957: 102).

In the Cognitive school early research concentrated on "human organisms". Markus and Zajonc (1985: 141) found that "the idea of the human organism as an information-processor became popular". The mind came to be viewed by many as a computer-like apparatus that registered incoming information and then subjected it to a variety of transformations before ordering a response". Recent cognitive theory and research has emphasised individual shortcomings as information processors and decision-makers (Tversky and Kahneman 1974; Nisbett and Ross 1980).

There are differing views of institutional theory from an economic, political and sociological perspective. There are salient differences between institutional theory and

neo-institutional theory from an economic perspective. Similarly, between institutional theory and neo-institutional theory in political science. Likewise, between old institutional theory and neo-institutional theory in sociology. The core ideas of each perspective will be examined in turn.

Institutional economists emphasised the importance of change. Veblen (1898; 1919) embraced an evolutionary perspective and claimed that valid economics could highlight the role of technological change and trace the changing phases of the economy. Commons (1924) stressed the centrality of change, - viewing the economy as "a moving changing process" (p. 376).

Neo-institutional economic theories are concerned with rule and governance systems which develop to regulate or manage economic exchanges. The first theorist to advocate this new approach was Coase<sup>26</sup> (1937). His article "The Nature of the Firm" asks why some economic exchanges are carried out within firms under a governance structure (involving rules and hierarchical enforcement mechanisms), rather than being directly subject to price mechanisms in markets. Coase (1937: 389) concludes the reason must be that "there is a cost of using the price mechanism", (i.e. the cost of negotiating and concluding a separate contract for each exchange transaction which takes place in a market). It is because of these "transaction costs"<sup>27</sup> that companies emerge.

Jaccoby (1990) argues that the approaches offered by early institutionalists depart from those of their neo-classical colleagues in four important aspects.

<sup>&</sup>lt;sup>26</sup> Coase in his article 1937, posed an important question. He asks

<sup>&</sup>quot;What is a firm and what determines its boundaries"?

The importance of Coase's contribution was the recognition that conventional neo-classical economics cannot explain why firms exist. Coase hypothesised that firms exist because they are able to operate under certain aspects of economic activity more cheaply than the market. This valuable insight has been further developed by economists under the general heading of "transaction costs economics". Two particularly influential contributions were from Williamson (1975) and Klein et al., (1978). More recently Hart (1995) has introduced "A Property Right Approach" to the theory of the firm (offering a significant extension of the transactions costs approach).

<sup>&</sup>lt;sup>27</sup> The basis of transaction costs is to explain the existence of different organisational forms as choices from a limited menu of "workable" contracting structures (Walker, 1998).

(a) Indeterminancy versus determinancy: where the orthodox model assumes "perfect competition and unique equilibria, the institutionalists point to pervasive market power and to indeterminancy even under competition" (Jaccoby, 1990: 318).

(b) Endogenous versus exogenous determination of preferences: Neo-classicalists stress the importance of individual wants or preferences, whereas institutionalists argue that such preferences are shaped by social institutions, whose operation should be the actual subject of economic analysis.

(c) Behavioural realism versus simplifying assumptions: Institutional theorists argue that economists should use pragmatic and 'psychologically-realistic' models of economic motivation, rather than subscribe to naive utilitarian assumptions.

(d) Diachronic versus synchronic analysis: Rather than accepting the "timeless and placeless" assumptions of neo-classical theorists, institutionalists recommend that economists should ascertain "how the economy acquired its features and conditions and what causes these features to vary over time and place" (Jaccoby, 1990: 320).

Institutional approaches dominate political science in both Europe and America. Eckstein (1963) claims the early institutionalists ushered in the first crude form of positivism in political science. Beginning during the mid-1930s and continuing through the 1960s, the institutionalist perspective was challenged and largely supplanted by the behavioralist approach (not to be confused with "behaviourism" in psychology). This was an attempt to sever the tie to moral philosophy and rebuild political science as a theoretically-guided empirical science (Easton, 1965). The behavioralist persuasion diverts attention away from institutional structure to political behaviour.

Neo-institutionalists within political behaviour have grouped themselves into two camps. Historical theorists and rational choice theorists. Historical institutionalists devote themselves to the detailed analysis of regimes and governance mechanisms (e.g. March and Olsen, 1984; 1989; Hall, 1986; and Skocpol, 1985; 1992). Institutions are viewed as including "both formal structures and informal rules and procedures that structure conduct" (Thelen and Steinmo, 1992: 2). Historical theorists take a social-constructionist position which assumes that "capabilities and preferences (i.e. the vary

nature of the actors) cannot be understood except as part of some larger institutional framework" (Krasner, 1988: 72).

The social-political constitution of accounting practice and regulation has gained increasing attention in research literature (e.g. Hopwood, 1984; Hopper et al., 1987; Miller et al., 1991). Theorists recommend that within the governmental sector, researchers should direct their efforts towards understanding the emergence of political institutions; particularly administrative institutions which have been influential in providing order and change in American politics, (in terms of the regulations they issue) (e.g. March and Olsen, 1989: 17).

Rational choice theorists see institutions as governance or rule systems. This approach represents an extension of neo-institutional work in economics, to the study of political systems. Moe (1984: 750) dissects the major elements making up this paradigm as including: "the contractual nature of organisations, markets vs. hierarchies, transaction costs, the rationality of structure, individualistic explanation, and economic methods of analysis. Standard neo-classical notions, optimisation, marginality, equilibrium are often central to this new tradition".

Thelen and Steinmo (1992: 9) see no conflict between historical institutionalists and rational choice theorists. They examine the historical stance: "Historical institutionalists would not have trouble with the rational choice idea that political actors are acting strategically to achieve their ends. But clearly it is not very useful simply to leave it at that. We need a historically-based analysis to tell us what they are trying to maximise and why they emphasis certain goals over others".

The importance of politics, (specifically in relation to the issue of 'change') is welldocumented in management literature (Buchanan and Badham, 1999; Dawson, 1994; 1996; Pettigrew, 1973). This research highlights the key role politics can play in driving and shaping change processes; (though politics is not the only factor). Although it is not intended here to get drawn into a debate of what politics represents, most writers see politics as "attaining interest-based demands" (Pettigrew, 1973) and "getting things done your way" (Buchanan and Badham, 1999). Pfeffer's (1981: 7) definition of politics aptly captures the generally-accepted view amongst most academics that: "Organisational politics involves those activities taken within organisations to acquire, develop and use power. It uses other resources to obtain one's preferred outcome in a situation in which there is uncertainty or discussion about choice"

Sociologists gives more attention to the institution itself than economists and political scientists. Cooley (1956) for example, emphasised the interdependence of individuals and institutions, of self and social structure. Hughes (1936; 1939) shares and elaborates upon this interdependent view. Weber however is caught between the three major debates; between those who view social science as a 'natural' science, and those who argue that it is a cultural science.

Weber argued that the social sciences differ fundamentally from the 'natural' sciences in that in the former both researcher and object of study attach meaning to events. For Weber (1968), action is social, and individuals do not automatically respond to stimuli. Researchers cannot expect to understand social behaviour without taking into account the meanings that mediate social action.

Between the idealist arguments of Durkheim and the materialist emphasis of Marx, Weber employed an interpretative approach to attempt a synthesis between the two. Although the material conditions and interests stressed by Marx constrain choice and action, the idealist interpretations of normative values, motivate and 'activate' action (Alexander, 1983).

Between the institutionalist-historical model of economics and classical interest in developing general theoretical principles, Weber embraced institutionalist arguments. He argued that economics should be historically informed and comparative in its approach. At the same time he agreed with Manger and classicists in supporting the value of theoretical models, (which allow us to derive information from 'historically-embedded' systems in order to formulate and evaluate general arguments. Weber claims by abstracting from the specificity and complexity of concrete events, researchers can create "ideal types" to guide and inform comparative studies<sup>28</sup>.

<sup>&</sup>lt;sup>28</sup> Researchers should not insist that individuals under all conditions behave as economists). Such models could then prove useful as guides for analysis, and increase understanding of the real world.

Neo-institutional theory in sociology focuses on legitimisation of organisational forms and processes. Silverman (1971) proposes an 'action' theory of organisations. Silverman attacks prevailing models of organisations (including contingency arguments and Parsons-Selznick's structural-functional view), as being overly concerned with stability, order, and technical efficiency. Silverman proposes a phenomenological view of organisations which focuses attention on meaning systems, (and the ways in which they are constructed and reconstructed in social action).

Meyer and Rowan (1977) embrace the view of institutions as "complexes of cultural rules". These rules are increasingly being rationalised through the actions of academic professions, nation-states, and the mass media. This has therefore engendered the development of more types of organisations. Meyer and Rowan draw attention to the impact of organisational forms of change on the wider institutional environment.

## **2.3 Institutional theory and Accounting**

There is no universal definition of an 'institution'. Scott (1995: 533) defines an institution as consisting of "cognitive, normative, and regulative structures and activities which provide stability and meaning to social behaviour. Institutions are driven by various carriers (cultures, structures, and routines) and they operate at multiple levels jurisdiction. In this conceptualisation, institutions are multifaceted systems incorporating symbolic systems, (cognitive constructions and normative rules) - i.e. the regulative processes carried out which shape social behaviour. As such meaning systems, monitoring processes, and actions are all interwoven".

#### Cognitive pillar

Cognitive elements of institutions are rules that constitute the nature of reality and a frame through which meaning<sup>29</sup> is made. Symbols - words, signs, and gestures - have their effect by shaping the meaning we attribute to objects and activities. Meanings arise in interaction and are employed to make sense of the ongoing stream of events.

<sup>&</sup>lt;sup>29</sup> In terms of the organisation's wider institutional context, its members conceptualise and give meanings to terms like "budget", "profit", and "costs" (Preston, 1995). The meanings of such terms are reproduced over time, and become embedded in the rules and routines which define their calculation. As such, accounting practices (e.g., budgeting) and accounting information (e.g., the return on capital employed) represents more than is commonly perceived. They include action and thought, and are more than simple decision-making aids (Swieringa and Weick, 1987). Perceptions and meanings are both cognitively and theoretically bound (Hodgson, 1988). Hence, the term "profit", for example would differ in interpretation for different professions.

Accounting routines facilitate interpretation of the work place, and can help individuals understand how their organisation is performing (March, 1988).

Social scientists have long recognised the importance of symbolic systems and shared meanings. However their early work treats these systems and meanings as internalised and subjective. Beyond the traditional view of institutional theory, formal control systems serve as symbolic displays, leaving actual work activities to be promoted by social means (Meyer and Rowan, 1977). Neo-institutional theory in sociology treats symbolic systems and cultural rules as objective, and external to individual actors. A cognitive perspective directs us to pay attention to the symbolic aspects of social life, however it would be a mistake not to attend to the activities also associated with these belief systems.

Previous research has found accounting to be a symbol of legitimacy (Carpenter and Feroz, 1992; Covaleski and Dirsmith, 1988; DiMaggio and Powell, 1983; Meyer and Rowan, 1977). Meyer and Rowan argue that certain elements of the formal structure in bureaucracies, function as myths. They cite accounting as an example. They argue that "Institutional techniques are not based on efficiency but are used to establish an organisation as appropriate, rational, and modern. The use of these techniques "display responsibility and avoid claims of negligence" (p. 344). They show that bureaucratisation is caused, in part, by the ubiquity of these rationalised myths in society. By designing a formal structure which adheres to the prescription of myths in the institutional environment, an organisation demonstrates that it is acting in a 'proper' and adequate manner.

There are a variety of cognitive elements (D'Andrade, 1984). Of these, 'constitutive rules' are the most important (Searle, 1969). These rules or "processes" involve the creation of categories and the construction of typifications (Berger and Luckmann, 1967: 39). Such processes are variously applied to things, ideas, events, and actors. Constitutive processes offer an explanation for much social behaviour and differ from lay interpretations, and from those found in social science. As Meyer, et al., (1987: 13) argue: "Most social theory takes actors (from individuals to states) and their actions, as real a priori, elements. Conversely we see the "existence" and characteristics of actors as socially constructed and highly problematic. The action is the enactment of broad

institutional scripts rather than a matter of internally-generated and autonomous choice, motivation and purpose".

#### Normative pillar

The normative pillar focuses on normative rules which introduce a perspective, - an evaluative and obligatory dimension into social life. Normative systems include both values and norms. Values are conceptions of the desirable, (which along with construction of standards, existing structures or behaviour), can be compared and assessed. Norms specify how things should be done. They define legitimate means to pursue valued ends. Normative systems define goals or objectives (e.g. making a profit), but they also designate appropriate ways to pursue them (e.g. conceptions of fair business practices). The normative conception of institutions is embraced by sociologists, because it focuses attention on institutions where common beliefs and values are more likely to exist.

The accounting norm system consists of the institutional environment of the accounting action system<sup>30</sup>. It is characterised by "rules and requirements to which individual organisations must conform if they are to receive legitimacy from the environment" (Scott and Meyer, 1983: 140). The accounting norm system should not however be perceived as a system acting in 'perfect harmony'.

Different interests are represented within the norm system. These often consist of many individuals and organisational actors (Mezias, 1990). The norm system comprises 'the multiple institutional environment' of the accounting action system. The accounting norm system is involved in many activities (most of which can be designated as "talk",both process and output). The rhetoric of accounting is one of its main activities.

The rhetoric of accounting plays a role in reducing uncertainty in decision-making and accountability (Mellemvik, et al., 1988; Miller and O'Leary, 1990). It is also the basis for most normative and positive accounting theory (American Accounting Association, 1977; Watts and Zimmerman, 1986). Rhetoric is fundamental to accounting as an organisational and societal activity (Birnberg, 1980; Jonsson, 1988).

<sup>&</sup>lt;sup>30</sup> The accounting action system comprises instrumental accounting activities. It operates within single organisations and can be divided into separate but interrelated procedures (e.g. the daily registration of transactions and annual reporting, Bergevarn, et al., 1995).

Rhetoric is important to the accounting norm system in the process of regulating norms (Puxty, et al., 1987; Johnson, 1988). It also generates and legitimises norms. Auditing is also involved in the accounting norm system (i.e. the evaluation of accounting procedures in relation to existing norms). From this perspective of the organisation, there are two types of norm systems - external and internal. Whereas an external norm system acts in order to co-ordinate the accounting action of organisations within its domain, an internal norm system acts mainly in order to adapt accounting action to its local context.

In theory, accounting norms should control accounting action. Although a strong link exists between norms and action, the latter may generate more organisational and societal qualities than are included at present in accounting rhetoric (Bergevarn, et al., 1995). This is because an implemented accounting procedure always confronts its context (Birnberg, et al., 1983; Boland and Pondy, 1983; Hopwood, 1983; 1987; Roberts and Scapens, 1985; Swieringa and Wieck, 1987; Nahapiet, 1988). The use of accounting in any environment can only be regulated by norms to a certain extent. The consequences of norms will therefore be different to those from intentions, (even though accounting action may be wholly controlled by norms).

As mentioned, a normative pillar includes values and beliefs. Definitions of value and belief differ in accounting. 'Values' are concerned with the legitimacy of the ends to which accounting is directed, while 'beliefs' are the confidence in the means of pursuing them.

In much of conventional accounting value is attached to capital (the maximising of shareholders wealth). This has a legitimate claim of being a universal value (Stewartt III, 1994). Also important are the values of those who propose "social responsibility" or "green accounting" (Jones and Dugdale, 2001). An accountant's overriding duty is to be "a friend of the Earth" (Gray, 1990). Disagreement over the term 'value' can also be detected in management accounting's treatment of costs. Whether labour should be treated as a "variable" or "fixed" cost may be merely a technical issue. Advocates of the latter view suggest there are important humanitarian values at stake (Jones and Dugdale, 1998). These concern whether employees are to be treated as costs (that must be

reduced or eliminated), or whether organisations should (on moral grounds) provide secure employment for their members (Goldratt, 1994).

Disagreement over which path to pursue is accompanied by differing beliefs as to which means should be adopted. The last few decades have seen successive views of "management philosophy" with its familiar acronyms (JIT, TQM). They all propose differing suggestions. Often "new" philosophy runs counter to 'conventional' wisdom (Jones and Dugdale, 2001). To move from one perspective to another requires a fundamental shift in viewpoint. Hence, belief has to be established and should not be taken for granted). Since the 1980s, fundamental changes in views ("paradigm shifts") have been associated with the activities of "management gurus". Observers of this phenomenon have noted that 'beliefs' and 'faith' are important and in this respect not dissimilar to spiritual beliefs (Clark and Salaman, 1995; Wilson, 1996; Oliver, 1997).

Marsh and Olsen (1989: 21) developed a primarily normative conception of institutions: "The proposition that organisations follow rules (where much of the behaviour is specified by standard operating procedures), is a common one in bureaucratic and organisational literature. It can be extended to 'political institutions'. Much of the behaviour we observe in political institutions reflects a routine way in which people do what they are supposed to do".

Very little institutional accounting research examines the internal processes of an organisation. However, Burns and Scapens (1998; 2000) focus on the internal processes of routinisation and institutionalisation. They look at how accounting routines provide the normative basis (i.e. legitimation) for organisational action. Accounting routines have 'productive potential' (i.e. instrumental value). They can also embody norms, values, etc., which favour particular vested interests (ceremonial value).

For cognitive theorists, compliance occurs in many circumstances, because other types of behaviour are inconceivable. Routines are followed because they are taken for granted as "the way to do things". Whereas normative theorists emphasise the power of roles (normative expectation guiding behaviour), the cognitive framework stresses the importance of social identities, (i.e. our conceptions of who we are and what action make sense for us in a given situation).

#### Regulative pillar

Regulative processes involve establishing rules. They help in the review of others' conformity and can manipulate sanctions (i.e. rewards or punishments) in an attempt to influence future behaviour. These processes may operate through diffuse, informal mechanisms, involving shaming or shunning activities (Scott, 1995: 35).

Economists view institutions as resting primarily on the regulative pillar. North (1990: 4) argues that "[institutions] are perfectly analogous to the rules of a game in a competitive team sport. That is, they consist of formal written rules as well as unwritten codes of conduct which underlie and complement formal rules. Rules and informal codes are sometimes violated and punishment enacted. Hence, an essential part of institutional functioning is the cost of ascertaining 'violations'. Economists focus their attention on the behaviour of individuals, firms in markets and other competitive situations".

Force, fear and expedience are central ingredients of the regulative pillar. They are tempered by the existence of rules, whether in the guise of informal mores or formal rules and laws. Weber (1968) notes that few (if any) rulers are content to base their regime on force alone. All attempt to cultivate a belief in its legitimacy. Economics emphasises the cost(s) of regulation. Economic historians view this as an important function of the state. North (1990: 64) argues: "Because ultimately a third party must always involve the state (as a source of coercion), a theory of institutions also inevitably involves an analysis of the political structure of a society, (and the degree to which that political structure provides a framework of effective enforcement)".

Some institutional accounting research (e.g., Covaleski et al., 1993; 1996; Mezias, 1990) focuses on the impact of external legitimation in shaping accounting practice. For example, Covaleski, et al., (1993) suggest that pressures exerted by the US Federal Government were crucial in the development of a case-mix accounting system in the health industry. Mezias (1990) describes how various institutional factors (e.g. laws, expectations, professionalism) influenced the formulation of US accounting policies.

Cultures rely primarily on interpretative structures - (codified patterns of meanings and rule systems). The emphasised aspects of culture will vary according to which institutional elements are given prominence. Cognitive theorists stress the importance of categories, distinctions, and typifications. Normative theorists point to shared values and normative expectations; while regulative theorists look to conventions, rules and laws.

Simons (1995) introduces the idea of belief systems (i.e. organisation culture) into his 'levers of control'. He links belief systems to boundary systems, interactive control systems, diagnostic control systems and business strategy. The levers of control can be described as follows:

(a) Belief systems - to inspire, and direct the search for new opportunities.

(b) Boundary systems - to set limits on opportunity-seeking behaviour.

(c) Diagnostic control systems - to motivate, monitor and reward achievement of specified goals.

(d) Interactive control systems - to stimulate organisational learning and the emergence of new ideas and strategies.

Social structures rely on expectations associated with networks of social positions (role systems). These structures constrain and empower the behaviour of actors, while at the same time they are reproduced and transformed by the same. This aspect of structure, depends on which elements are put into effect. Cognitive theorists stress structural isomorphism. Cognitive typifications are often coded into organisational structures as differentiated departments and roles. Normative and regulative theorists are apt to view structures as governance systems, emphasising either the normative (authority) or the coercive (power) aspects of these structures.

Routines rely on actions which reflect the tacit knowledge of actors - (deeply ingrained habits and procedures based on prior knowledge and beliefs). Early institutionalists viewed habitualised actions, routines, standard operating procedures, and patterned activities as central features of institutions. More recently, evolutionary theorists such as Nelson and Winter (1982), point to the stabilising role played by participants' skills and organisational routines. These are activities involving little or no conscious choice, and

behaviour is governed by tacit knowledge and constrained by rules of which the actor may be unaware.

Nelson and Winter, (1982: 410-411) describe accounting as "a highly structured set of routines". Their work on evolutionary economics is useful in developing an institutional approach to management accounting research. An institutional approach recognises that accounting can both shape and be shaped by institutions which govern organisational activity. Managers now appear to be using their accounting systems and routine financial reports more flexibly, in conjunction with a range of other performance measures, (both financial and non-financial) (Kaplan and Norton, 1992; 1996; Miller and O'Leary, 1993; Ezzamel, et al., 1994; Burns and Scapens, 2000).

During the 1990s, a growing number of scholars characterised their contributions (in the interpretation of accounting systems) as 'organisational routines' (Roperts and Scapens, 1990; Dent, 1991; Scapens, 1994; Burns and Scapens, 2000, Burns 2000). In his institutional perspective on accounting, Burns (2000a) classifies these contributions into three different groupings: Neo institutional economics (or transaction cost economics), Neo institutional sociology and finally Old institutional economics.

There is a two-way relationship in both directions between rules and routines (Burns and Scapens, 1998; 2000). In effect, rules are formalised statements of procedures, whereas routines are the actual procedures used. Rules are normally only modified at intervals, whereas routines are in a cumulative process of change. Rules may be imposed and become implemented through routines, or they can emerge out of established routines.

In the context of management accounting, rules are made up of accounting systems (as set out in procedure manuals), whereas routines are accounting practices in use (Burns and Scapens, 2000). Clearly, there is a relationship between rules and routines, however it is important not to confuse the two. Any given management accounting system in use, may not accurately mirror the procedures set out in the procedure manuals (Robert and Scapens, 1985).

Structuration theory<sup>31</sup> is the relationship between activities of knowledgeable human actors and the structuring of social systems. Giddens distinguishes between systems, (comprising discernibly similar social practices), which are reproduced across time and space through human actions, and structures, which transform those social practices into systems. For Giddens, systems are not structures, but are structures drawn upon in action (Burns and Scapens, 1998; 2000; Jones and Dugdale, 2001).

Giddens set out to develop a theory of the production, transformation, reproduction and dissolution of social institutions by incorporating both structure and agency. These are the structural properties of social institutions (codes, rules and blueprints that influence and are influenced by social actions in day-to-day activities across time and space (Giddens 1984: 17). Agency is strategic actions of knowledgeable, reflexive social actors which illustrate their working lives.

Old institutional economics has defined 'institutions' as "a way of thought or action of some prevalence and permanence, which is embedded in the habits of a group, or the customs of a people" (Hamilton, 1932: 84; Scapens, 1994). This definition embodies both systems (thoughts and actions of prevalence and permanence) and structure (embedded habits and customs). Macintosh and Scapens (1990) argue that management accounting can be understood by drawing upon Giddens' three dimensions of structuration: signification<sup>32</sup>, domination<sup>33</sup> and legitimation<sup>34</sup>.

<sup>&</sup>lt;sup>31</sup> Macintosh (1994) summarises: "Structuration theory is the interplay between agent action and social structure, in the production, reproduction, and regulation of social order. Structures existing in virtual time and space, are drawn upon by agents as they act and interact in specific time-space. Agents are setting themselves the outcomes of these actions and interactions" (Macintosh, 1994: 172).

Giddens (1992) does not accord priority to either structure or action. Rather he advocates a study of social practices which both socialises (constitute) individuals as actors, and realise (embody) structures. Thus structure and action are "different ways of looking at the same thing" (Craib, 1992: 34). The production and reproduction of society is seen as a skilled performance for its members who draw upon both practical and discursive consciousness. 'Practical consciousness' is knowledgability about the world, while 'discursive consciousness' is our reflexive monitoring of action (Jones and Dugdale, 2001). The challenge in empirical work (based on structuration theory) is to keep in mind concepts of structure and agency and relate this duality to the inter-relationship of dis-embedded and embedded levels over time and space.

<sup>&</sup>lt;sup>32</sup> Signification creates meaning in social interaction.

<sup>&</sup>lt;sup>33</sup> Domination produces power.

<sup>&</sup>lt;sup>34</sup> Legitimation provides a system's morality.

These three dimensions<sup>35</sup> also appear within institutional literature. Figure 2.1 illustrates the link between them.

Institutional theory		Giddens' theory	
Old institutional economics	focus on	Signification	(meaning)
Neo institutional economics	focus on	Domination	(governance structures) reflects a concern for control
Neo institutional sociology	concerned with	Legitimation	of organisational forms and process

Figure 2.1: Institutional and Giddens' theory

Source: Adapted from Scott, W. (1995); and Burns and Scapens (1998)

For institutions, levels may be operationalised as a range of jurisdictions taken from the institutional form. Scott (1995) identifies six categories: (a) levels of world system (b) societal (c) organisational field (d) organisational population (e) organisation and (f) organisational subsystem

This research has focused on the 'organisation level'. This means it focuses on the micro-processes, but does not ignore the broader (macro-institutional) dimension. To understand these micro-processes it is necessary to understand the wider institutional context. This involves both within the organisation (i.e. the organisation's specific milieu of rules, routines and institutions), and outside (i.e. the broader social, economic and political institutions of the organisational field, and the society in which the organisation is situated). Table 2.1 refers to those theorists whose work has pursued on institutional element at each of the specified levels.

<sup>&</sup>lt;sup>35</sup> Macintosh (1994: 172) argues that the three layers (outlined in the abstract for analytical purposes) are closely entwined in reality. Accounting systems consist of signification codes and the interpretative schemes for applying them. They also include domination codes and resources (rights) used to draw upon them, and also legitimation codes and also norms for using them (Jones and Dugdale, 2001).

	Pillars			
Levels	Regulative	Normative	Cognitive	
World system	North and Thomas	Krasner 1983	Meyer 1994	
	1973			
Societal	Skocpol 1979	Parsons 1953, 1960a	Dobbin 1994	
Organisational field	Campbell and	Mezias 1990	DiMaggio 1991	
	Lindberg 1990,			
	Schmitter 1990,			
	Ezzamel 1993, 1996			
Organisational	Barnett and Carroll	Singh, Tucker, and	Carrol and Hannan	
population	1993a	House 1986	1989	
Organisation	Williamson 1975,	Selznick 1949	Clark 1970, Burns	
	1985, 1991, Burns		and Scapens 1998,	
	1997, Robert and		2000.	
	Westin, 1999			
Organisational	Shepsle and	Roy 1952, Buroway	Zimmerman 1969.	
subsystem	Weingast 1987	1979		

# Table 2.1: Institutional Pillars and varying levels: Illustrative theorists

Source: Adapted from Scott, W. (1995)

It is possible to associate various 'schools' or 'types' of work with different locations in the property space, (created by the cross-classification of emphasis and level, see figure 2.2).

.

	Pillars		
Levels	Regulative. Normative.		Cognitive.
World-system	Economic history		Neo- institutional
Societal			sociology
Organisational field	Historical institutionalism in political science	Traditional institutional sociology	
Organisational population			Population ecology
Organisation	Neo- institutionalism in economics		Evolutionary theory in economics
Organisational subsystem			Ethno- methodology

Figure 2.2: Institutional Pillars and varying levels: Illustrative schools

Source: Scott, W. (1995)- "Institutions and Organisation", p. 59

Scott (1995) discovered salient differences among schools aligned with neoinstitutionalism. Sociologists emphasise a cognitive conception, cultural carriers, and macro level forces. By contrast, neo-institutional economists stress a regulative conception, structural carriers, and a micro focus. Different ideological perspectives share the same view.

Zucker (1977: 728) observed that "institutionalisation is both a process and a property variable". Mohr (1982) differentiates between variance and process theories. Variance theories (associated with viewing institutions as entities), focus on abstract variables, (both independent and dependent), and attempt to establish their causal relations.

Variance approaches identify the factors causally associated with the phenomena of interest. Variance theories address the question: why did the observed effect happen?

A process approach addresses the question: how did the observed effect happen? This approach assumes that "history matters". It believes how things occur, influences things that subsequently happen (Scott and Meyer, 1983). Burns and Scapens (2000) have used a process approach to explore the complex and ongoing relationship between actions and institutions. They demonstrate the importance of organisational routines and institutions in shaping the processes of change in management accounting.

This research adopts a process approach. The rationale for choosing this approach is because it allowed the researcher to observe the state of management accounting and control in the Egyptian water industry. Questions to be addressed include: How has the water industry changed? What needs changing? There have long been recommendations for a 'process' approach to be adopted in accounting research; - an approach which highlights the characteristics of accounting as a process, rather than as an outcome (Hopwood, 1987; Laughlin, 1991; Covaleski, et al., 1993; Scapens, 1994; Burns, 2000).

Management accounting systems and practices constitute organisational rules and routines. In order to understand the complex processes of the Egyptian Water Industry, it is necessary to examine both the internal and external institutional context of the organisation. An understanding of management accounting as organisational rules and routines, should enable the researcher to focus his energies on developing management accounting concepts, techniques, systems, etc. These are likely to be more useful in practice than the so-called 'optimal techniques'- designed for abstract rational decision-makers (Burns and Scapens, 2000).

## **2.4 Conclusion**

This chapter has shown the importance of both contingency theory and institutional theory for explaining the design and use of a cost accounting system. The chapter has argued that while technical (or contingent) explanations of cost management were not rejected, they were clearly viewed as incomplete (Scott, 1987). Institutional theory adds the social and political elements, which were typically absent or de-emphasised in the

rational instrumental approach (which guides most of the recent research on cost management).

This chapter works with a definition of 'institutional theory' and its core elements: cognitive, normative and regulative. It states the researcher view of management accounting. This research contends that accounting in organisation is primarily concerned with external or macro institutional effects (social, economic and political) and management accounting as an institution within the organisation.

The chapter looks at previous accounting research and notes that very little institutional accounting research focuses on the internal processes of an organisation. Hitherto, most institutional accounting research has focused on the impact of external legitimisation in shaping accounting practice. This led the researcher to look at the institutional context within the GOGCWS (for example, which models of cost management were followed in the GOGCWS, and what methods of cost accounting were used in the GOGCWS, see sections 8.5, 8.6 and 8.7). It also led the researcher to look at the institutional context outside (i.e. the GOGCWS the legal framework for water industry in Egypt, the external supervision and control body of the GOGCWS and the suggestions by the consultants Black and Veatch International which related to cost management, see sections 8.1, 8.3 and 8.7).

The chapter links the three pillars of institutional theory (cognitive, normative and regulative) with accounting research. It also examines the differences among schools aligned with neo-institutionalism. For example, sociologists emphasise a cognitive conception, cultural carriers, and macro level forces. By contrast neo-institutional economists stress a regulative conception, structural carriers and a micro focus. As mentioned in section 1.4, this research adopts neo-institutional economic theory, this led the researcher to focus on regulative pillar of institutional theory.

The chapter points out the similarity between Giddens' three dimensions (signification, domination and legitimation) - and institutional theory. It focuses on the 'organisational level' of analysis. It gives the chosen approach of analysis, a process approach and offers a rationale for choosing it (see section 2.3). Furthermore, there have long been recommendations for a 'process' approach to be adopted in accounting research. Having

determined the theoretical framework of this research, the next chapter will discuss the role of management accounting from the institutional theory perspective.

# Chapter 3

# Institutional theory and the role of management accounting

# **3.0 Introduction**

Management accounting (cost management) research is explained largely by rational economic theory, and neo-classical theory (the normative view). A major criticism of rational economic theory is that it fails to consider the behaviour of the firm. It also neglects the social and political context. Rational economic theory assumes the imperfect market does not exist and therefore neglects the monopoly market, uncertainty, and unpredictability.

In order to address these issues, this chapter examines alternative institutional approaches, - namely old institutional economic theory, neo-institutional economic theory and neo-institutional sociological theory. The chapter begins with a rejection of the core assumptions of neo-classical economics theory. It is argued that institutional theory offers a more useful theoretical framework for understanding management accounting (cost management) practices.

The aim of this chapter is to illustrate the role of management accounting, by examining the theory behind it. For this purpose this chapter is divided into four sections.

The aim of the first section is to illustrate the role accounting has come to serve in both micro-level (organisations) and macro-level (societies). There then follow three subsections: the first part explores several views on accounting, explains its role and examines the relationship between the state and accounting. The second sub-section is devoted to the role of accounting in micro-level organisations. It reviews the relationship between accounting and decision-making and its problems as outlined by Thompson and Tuden (1959). The third sub-section looks at the role of accounting in macro-level societies. It focuses on rational decision-making, economic rationality, differences between formal and substantive rationality, while also examining the process of accounting regulation. The second section reviews the role of management accounting within organisations, from an economic perspective. Problems of management accounting within the organisational setting are discussed.

The third section reviews the different perceptive on management accounting. It begins with criticism of management accounting from a neo-classical economic perspective. There then follows a review of management accounting from an institutional perspective, - (incorporating old institutional economic theory, neo-institutional economic theory and neo-institutional sociology theory).

The final section sums up the main conclusions of the chapter.

#### 3.1 The role of accounting in organisation

#### 3.1.1 Different views of accounting and its roles

Previously accounting has suffered from the misconceptions of being too "narrow" or "technical". Accounting is portrayed as a simple technique, - merely a set of instructions for generating, processing and evaluating information (Carruthers, 1995).

Social science has demonstrated the richness and significance of accounting. Establishing links between accounting research and other paradigms has been an important part of this project. For example, theories of class conflict inform Bougen's (1989) investigation of accounting and industrial relations. Similarly, Miller and O'Leary's (1987) historical analysis of standard costing and budgeting employs Michel Foucault's conception of power. Likewise Baecker's (1992) discussion of accounting draws upon mathematical logic and literary criticism. Such examples serve to reinforce the relevance of accounting for a variety of issues (power, conflict, rationality, governance, and organisational behaviour).

Accounting has functional claims and pretentions<sup>36</sup> at its core. We are led to believe it is essentially concerned with the provision of "relevant information for decision-making". It has a "rational allocation of resources" and maintains institutional "accountability"

<sup>&</sup>lt;sup>36</sup> Littleton (1953: 18) in his search for a rationale, argues that: "There must be some basic concept which makes accounting different from all other methods of quantitative analysis. There must be a central idea which expresses better than any other the objectives, effects and results characteristic of accounting. A 'centre of gravity' if you will''.

and "stewardship". Accounting systems are seen as mirrors of both societies and the organisations in which they are utilised. At a societal level, accounting reflects on organisation's 'social' relationships. The organisational level has recently been influenced by the emergence of contingency schools of thought in the study of organisational behaviour (Bruns and Waterhouse, 1975; Hopwood, 1974; Sathe, 1975; Waterhouse and Tiessen, 1978; Watson, 1975).

Accounting systems may be viewed as a product of the complexity, uncertainty of the organisational environment (Galbraith, 1973; Khandwalla, 1972). Accounting may also be seen as 'technology of the enterprise' (Daft and MacIntosh, 1978) or a strategy of corporate management (Chandler, 1962). Accounting is a discipline which can encompass a body of ideas, conventions, variable tools/techniques and a variety of actual practices (Boyns and Edwards, 1997). Hoskin and Macve (1992) note that "the power of accounting" is associated with the need to co-ordinate activities of departmentalised large-scale business organisations (Williamson, 1980; Johnson and Kaplan, 1987).

The varied role of accounting can tell us how it influences accounting discourse and the way people have sought to influence accounting. The various roles of accounting have changed its practice. Argyris and Schon (1974) point out a mere theory on paper is very different from theory in practice. The roles of accounting and its actual practice appear to have a rather equivocal relationship.

Functional claims for accounting stem mainly from professional institutes, accounting regulation bodies, state agencies and the academy itself. They very often reflect the pressures on those bodies and a need for a public legitimacy and rationale for action. Formulated in the context of particular institutional needs and actions, functional claims attempt to provide novel interpretations of the accounting mission (Burchell, et al., 1980).

The roles of accounting provide a normative structure for accounting thought. The roles define what accounting is, and ought to be about. These roles therefore facilitate the appraisal of accounting practice. Accounting has changed in terms of the roles it serves. Those who have sought to 'extend' accounting promote "corporate accountability" and

further "rational decision making". Others point to the challenges which social change necessarily creates for accounting practice (Gilling, 1976).

The State exerts influence on accounting in terms of accountability and the furtherance of organisational and societal efficiency (Searle, 1971). Professional institutes and agencies concerned with accounting regulation adopt a similar stance. They emphasise the role accounting could play in improving the flow of information (useful for investment decisions of shareholders). Those practising accounting within organisations, point to its relevance in improving organisational efficiency and the maintenance of organisational control.

Shaoul (1997) claims that since the 1960s, mainstream financial accounting research analyses problems and issues in positivist or technical ways (in which the basic values of the model are not questioned). Issues are therefore conceived as purely technical problems. In financial accounting research, the issue was largely how to make public sector accounting more like that of the private sector. The debate therefore revolved around the value of *accrual* accounting.

Throughout the 1970s, emerging accounting research opposed the technical approach. This research re-examines the theoretical underpinnings of the discipline, - critically analysing the basic concepts of profit, value, efficiency, etc. Other research traces the history of accounting discourses and its pliability according to the social context in question. Other research has looked at accounting's role as ideological in the distribution conflict over wealth. Some researchers have analysed accounting as rhetoric, examining its epistemological basis.

Critical accounting research has turned away from analysing contemporary events or outcomes of public policy decisions in ways that are useful or helpful to the broader public. This has happened at the very time when the public sector has faced interventions (in which accounting was to play a crucial role). A recent study by Edwards and Shaoul (1996) found that relatively few academic researchers have examined contemporary corporate events. There are also few accounting studies of corporate performance using primary sources (see Adcroft, et al., 1991; Berry, et al., 1985; Cooper and Hopper, 1988). Briloff and Chatov were the foremost accounting critics of corporate behaviour in the USA. In the UK there was Stamp (see Stamp and Marley, 1970; Stamp, et al., 1980) and more recently, Sikka (see Mitchell, et al., 1992; Mitchell and Sikka, 1993; Russell, 1991).

The different views and roles of accounting are summarised in table 3.1 below:

Views of accounting	Role of accounting	
Accounting is seen to have a core of	Provision of relevant information for	
functional claims	decision-making, with the achievement of	
	rational allocation resources.	
Accounting is seen as a mirror of society,	Reflects the organisation of social	
or a mirror of organisations	relationships (contingency theory).	
Accounting is a discipline which	Changes the practice of accounting.	
encompasses a body of ideas, a set of		
variable techniques and a variety of		
practices.		

Table 3.1: Different views of Accounting and impact on its role

Source: Adapted from Burchell, et al., (1980)

#### 3.1.2 Accounting systems and organisational practice

The relationship between accounting and organisational decision-making has been an influential basis for the analysis, development and articulation of normative accounting roles and "solutions". Many writers have noted the multiple roles which accounting systems can and do play in providing relevant information for decision-making. They help improve the rationality of the decision-making process and maintain the organisation in 'a state of control'. A problem however is that the relationship between accounting information and decision-making has not been critically examined. Burchell, et al. (1980) refer to the work of Thompson and Tuden (1959). Their study distinguishes between uncertainty over the objectives for organisational action, and uncertainty over the patterns of causation that determine the consequences of action (see figure 3.1).

		Uncertainty of objectives	
		Low	High
Uncertainty of	Low	Decision by computation	Decision by compromise
cause and effect	High	Decision by judgement	Decision by inspiration

Figure 3.1: Decision-making and location of organisational uncertainty.

Source: Burchell, et al., (1980)- "The Roles of Accounting in Organisations and Society", p. 13

When objectives are clear and undisputed, the consequences of action are presumed to be known. Thompson and Tuden highlight the potential for decision-making by computation. This situation might represent what Simon (1960) called 'structured decision-making', where the intelligence, design and choice phases are all programmable. Accounting systems serve as "answer machines", providing the simple investment appraisal methods, stock control systems and credit control routines which feature in management accounting texts.

With clear objectives but uncertain causation, the situation becomes more complex and the potential for computation diminishes. Thompson and Tuden see decisions being made in a judgmental manner. Accounting systems might be expected to provide assistance through what Gorry and Scott-Morton (1971) refer to as 'decision support systems' and Churchman (1971) calls 'inquiry systems'.

There are two reasons behind the extending of computational practice into the realms of the judgmental. The first reason stems from the increasing formalisation and objectification of management knowledge. The second from the extent to which accounting practices have become implicated in the development of neo-management and other more complex forms of organisational segmentation and management.

Accounting involves the implementation of changes in management practice. The increasing formalisation of investment appraisals and planning processes has extended the sphere of financial calculation. Developments in accounting practice have enabled 'operationalisation' of organisational efficiency and performance, - allowing objectives for action to be stated in less ambiguous terms.

Given the uncertainty over objectives and relative certainty over causation, values, principles, perspectives and interests will conflict. Standards for appraisal and the criteria for guiding the organisational task are problematic. Here political processes are important in the decision-making process. Modes of accounting can act as "ammunition machines" through which different parties seek to promote their own vested interests. In order to selectively channel distribution of information, certain parties implicated in organised action, can introduce neo-mechanisms for organisational control, and management of information-flow (Pettigew, 1973).

Where objectives are uncertain, accounting systems are often used to promote particular values. Mechanisms for organisational control are now arising out of the political and conflictive nature of organisational life. As Pfeffer (1978) states: "Structure, it would appear, is not just the outcome of a managerial process in which (organisational) designs are selected to ensure higher profit. Structure, rather, is itself the outcome of a process in which conflicting interests are mediated so that decisions emerge as to what criteria the organisation will seek to satisfy. Organisational structures can be viewed as the outcome of contests-for-control and influence occurring within the organisation. Organisational structural arrangements are as likely to be the outcomes of political processes as are organisational resource allocation decisions".

The design of information systems and accounting systems are often implicated in the management of political processes. For organisational actions and their consequences, accounting systems can influence those which become relatively more visible (Becker and Neuheuser, 1975) (particularly to senior management groups). The 'visibility' that is established is very often an asymmetric one. The powerful observe the less powerful, but not vice versa. It is a unique mode of surveillance that is established. The centralised co-ordination of activities maybe established. Budgeting, planning and reporting can together provide a framework within which a measured and observed delegation of authority can take place.

With uncertain objectives and causation, Thompson and Tuden point out that decisionmaking tends to be 'inspirational' in its nature. As such, the accounting system can be created to serve as an organisational "rationalisation machine". The accounting mission focuses exclusively on those accounting roles which precede decision-making. Accounting systems are often implicated in organisational processes. As Bower (1970) discovered, the widespread use of capital budgeting procedures has resulted in the availability of justification devices for proposals of organisational action. Accounting has gained widespread support, and has a simple provision of information for decision-making (for proposals which remain problematic). Similarly budgets can be built around what is likely to happen. Burchell, et al., (1980: 28) refer to the work of Meyer and Rowan (1978) who claim that: "Much of the irrationality of life in modern organisations arises because the organisation itself must maintain a 'rational corporate persona'. We find planners and economists who will waste their time legitimising plans already made, accounts to justify our prices, and human relations professionals to defect blame from our conflicts. Life in modern organisations is a constant interplay between the activities that we need to carry on and the organisational accounts we need to give".

Some authors have looked at how accounting is 'implicated' in organisational practices (Burchell, et al., 1980; Hopwood, 1983; Neimark and Tinker, 1986; Hopwood, 1987; Knights and Collinson, 1987; Lehman and Tinker, 1987; Miller and O'Leary, 1987; 1990; Hines, 1988; Loft, 1986; Nahapiet, 1989). This 'constitutive' role of accounting to create particular conceptions of organisational reality, is especially potent in processes of organisational change. The role of accounting in actively constructing organisational purposes and rendering visible the issues, projects, and criteria which accrue significance in the pursuit of those purposes, has been considered by a number of authors (Hines, 1988; Nahapiet, 1989; Loft, 1986; Hopwood, 1983; 1987; Miller and O'Leary, 1987; Neimark and Tinker, 1986; Burchell, et al., 1980).

Accounting researchers agree that accounting techniques are an effective means of facilitating the process of allocating and utilising resources in pursuit of desired objectives. These techniques also produce a variety of other effects within organisations. Covaleski and Dirsmith (1986) suggest that budgetary systems can be used for representing vested interests in political bargaining processes, and in maintaining power-relationships within organisations. Likewise Bariff and Galbraith (1978) concluded that by manipulating the method by which information is processed, control systems can sustain or alter power-relationships within organisations. Nahapiet (1989) found in particular circumstances, accounting can be seen as a set of rules for

negotiating an understanding of organisational reality. This is contrary to being viewed simply as a mechanism for establishing an objective economic reality. Markus and Pfeffer (1983) have illustrated cases of accounting system failures in organisations (following the implementation of modified accounting and control mechanisms to achieve desired ends).

It is increasingly being acknowledged that accounting systems do not simply allow their intended objectives to be realised. The 'effects' of accounting are not necessarily confined to their designated use. As such, "the consequences of accounting do not necessarily have a close and automatic relationship with the aims in the name of which it was introduced" (Hopwood, 1984: 185).

Espeland and Hirsch (1990) in their case study of a conglomerate merger, illustrate the rhetorical power of accounting as a symbolic system for legitimating new corporate forms and practices. Similarly Jones (1992) charted the changing perceptions and use of accounting control systems to improve efficiency following a management buy-out. These studies have been central to the work which has explored the role of accounting in recent public sector organisational reform, (particularly the impact of new forms of financial control), (Broadbent and Guthrie, 1992; Dent, 1991; Humphrey and Scapens, 1990; Humphrey, 1991; Bourn and Ezzamel, 1986). The role of accounting changed from its previous concerns with probity, compliance and control, to one which focused on efficiency, effectiveness and cost saving (Richardson and Cullen, 2000).

Recent studies of organisational change in the private sector have shown pro-active accounting to be an important feature (Espeland and Hirsch, 1990; Jones, 1992) and in the public sector (Bourn and Ezzamel, 1986; Espeland and Bourn, 1990; Broadbent, et al., 1991; Hood, 1991; Broadbent and Guthrie, 1992; Humphrey and Scapens, 1992). It has been especially evident in studies of organisations earmarked for privatisation, (such as the Regional Water Authorities of England and Wales which supply water and sewerage services, Ogden, 1995a).

In the case of water, the constitutive role of accounting (in actively constructing organisational purposes and shaping new images of the organisation and its relationship with its environment), is especially evident. In response to new imperatives to cut costs

and improve financial performance, senior managers sought to re-focus and improve the ways in which the organisation was governed and controlled. As such, more attention was given to economic implications, particularly in term of costs and management decision-making. Subsequently considerable organisational resources have been given over to the development of more elaborate management information systems. These include: (a) developing, planning and control mechanisms (b) Ensuring detailed specification of objectives (c) Generating more accurate assessment and allocation of costs to each unit of operational activity and (d) Introducing performance indicators and output measures (to evaluate and compare achievements). Managers were given new training in management skills and were subject to a much more numerate and performance-based type of reporting and accountability (Ogden, 1995a: 198).

#### **3.1.3 Accounting and Social Practice**

Accounting cannot, however, be conceived as purely an organisational phenomenon. Whilst arising out of organisational and institutional pressures, it is also a prevalent feature of the societies in which we live. Certainly the development of accounting itself has paralleled the emergence of numerous other specialised mechanisms for information-processing and social and economic calculation. These include statistics, the compilation of information for social and economic administration, instruments for social and economic categorisation in medicine, psychiatry, education, law and business and economic life (Baritz, 1960; Cullen, 1975; Kamin, 1974; Kendrick, 1970; Sutherland, 1977).

Marx saw accounting as an important tool in decision-making for capitalists. His ideas were taken up by Most (1963) and Bailey (1978). Most (1963) was critical of Marx for not discarding his ideological "baggage", and rather concentrating wholly on management accounting. However it is important to note that Marx stressed the 'social rationality' of accounting. Social rationality encompasses 'mystification' of social relationships. Burchell, et. al., (1980: 19) refers to a Marxist text; "The way in which surplus value is transformed into profit by the rate of profit is a further development of the inversion of a subject which takes place in the process of production. In the latter we have seen the subjective productive forces of labour appear as productive forces of capital. The value (or past labour), which dominates living labour, is incarnated in the capitalist. The labourer appears as mere labour-power – (a commodity). Even in the

simple relations of production, this inverted relationship produces correspondingly inverted conceptions..."

Weber (1969: 86) stressed economic rationality: "From a purely technical point of view, money is the most "perfect" means of economic calculation. That is, it is the most rational means of orienting economic activity". Weber saw rationality in terms of the calculative means which might bring it about. He defined the 'formal rationality of economic action' as "the extent of quantitative calculation or accounting, technically possible, which is actually applied" (1969: 85). Weber (1969) distinguished between 'formal rationality' and 'substantive rationality'. He thought formal rationality provided an adequate means to achieve substantive rationality.

Hirst (1976) takes this argument further. He states: "Only formal rationality can adjust a means to an ends in terms of efficiency, (since it provides a quantitative measure of efficiency). A qualitative measure of the efficient use of resources is logically impossible. All economic action therefore requires formal rationality, and indeed, is modelled on formal rationality. Resources cannot be "rationally oriented" to economic ends without quantitative calculation. The definition of 'economic action' is viewed in terms of formal rationality. Formal and substantive rationality are not alternative and equally "rational" calculations. End-rational action (in the economic sphere) requires formal calculation.

Some commentators have stressed the role which accounting plays in allowing the devolution and decentralisation of economic decision-making. Others have pointed to the different internal pressures which should be taken into account when decision-making is centralised, (either in the hands of the monolithic enterprise, Chandler, 1962) or the state (Bettleheim, 1976). Accounting is seen as both reflecting and enabling the construction of society. Institutional forms and modes of social action are intertwined in its emergence and development.

In terms of technical elucidation and standardisation, attention has been given to the institutional and political components of the regulatory endeavour. In the case of inflation-accounting in the United Kingdom, it has been seen that the institutional mechanisms for accounting regulation arose at the interface between a critical media,
concerned agencies of the state and a profession concerned with preserving its powers of self-regulation and control (Zeff, 1972).

The interests of the state are changing. Initially inflation-accounting was concerned with the taxation of corporate income and the formulation of policies for macro-economic management. However other agencies of the state began to devote attention to the role which inflation-accounting might play in the implementation and furtherance of microeconomic policies for industrial recovery and growth. Others have looked at the role accounting plays in coping with both an economic crisis and threat of a changing social power structure.

In a regulatory environment, the conditions for accounting change are complex. Both technical and conceptual developments are required. However to be influential they have to root themselves in a dynamic constellation of issues which constitutes the accounting context. Privatisation led to many changes in different areas. Central to these changes was an all-pervasive managerial concern with introducing and operating more stringent forms of accounting controls over organisational activities. Initiatives in cost-conscious resource management, cost improvement programmes, performance indicators, delegated budgets, cost centres and new management information system have all been variously deployed. Managers have tried to give effect to politically - determined notions of efficiency, cost effectiveness, value for money, and greater financial accountability (Gray and Jenkins, 1986; Fry, 1988; Clarke and Cochrane, 1989; Dent, 1991; Humphrey and Scapens, 1990; Humphrey, 1991; Broadbent, et al., 1991).

Accounting is central to the processes of organisational change. Privatisation has changed accounting information systems and the uses made of accounting information. In the case of the ten Regional Water Authorities of England and Wales, Ogden (1995a) argued that there should be an increased emphasis attached to costs and cost-reduction to meet Government - determined performance aims prior to privatisation. He also stressed the need to enhance their efficiency profiles in the run-up to privatisation. This resulted in a management initiative that not only served to achieve the targeted improvements in organisational efficiency, but also to communicate new organisational

realities and promote cost consciousness as a mobilising rationale for organisational action.

Accounting has come to be seen as a social and institutional practice that acts upon individuals, entities and processes, and transforms them in an attempt to achieve a specific end (Miller, 1994). Accounting is widely seen as a powerful force, influencing both individuals and society (Jones and Dugdale, 2001: 35-36). Miller (1994:1) argues that development of interests in the social and institutional practice of accounting "entailed a move by researchers beyond those organisations at the executive level of research".

Accounting operates more broadly and deeply than appears apparent on the surface of specific practices (Jones and Dugdale, 2001). Accounting is embedded in (and constitutes) social relations, and the strong occupational organisation of accounting (Armstrong, 1987a: Anderson et al., 1997).

Accounting as a process constitutes, and is embedded in social and system relations<sup>37</sup>. The re-embedding of accounting is shaped by social relations of local contexts, (whilst simultaneously shaping those contexts, Jones and Dugdale, 2001). Time is organised around accounting periods - for example, it becomes structured around annual, quarterly, and monthly reporting. Accounting generates chronological cycles of planning and budgeting. Accounting thus regulates relations between individuals, the wider organisation, and parts of the actual organisation.

The claims of accounting to represent corporate activities faithfully, are in part, supported and reinforced by the institutional commitments of regulators, academics and politicians. They are all keen on the idea of "free markets". They characterise markets as "fair" distributive mechanisms. In other words, unimpeded markets (i.e. unregulated), are seen to reward the 'productive' with pecuniary compensation, and punish the "non-productive" with failure. Markets reward productivity and punish non-productivity in an even-handed manner. Hence, intervention into the operation of markets is considered

<sup>&</sup>lt;sup>37</sup> 'Social' integration refers to the ways in which actors relate to each other. 'System' integration is how institutionalised parts of systems relate to each other (Lockwood, 1964).

inappropriate. A reluctance exists among certain economists, regulators, and politicians to intervene directly into the operation of markets.

Despite the belief in the fairness of markets, there remains a need to monitor market operations at a macro<sup>38</sup> level, (as well as to monitor the success or failure of individual entities as they function in a market economy). Regulators often call upon accounting (with its representational claims) to make the effects of the economic, "visible" and measurable. Accounting is thus seen to act as a scorecard by which we identify productive and non-productive entities. The representational claims of accounting are reinforced when its practices are called upon to provide a window through which to view specific entities operating in the market. This in turn allows us to view operations of the market.

The representational claims of accounting are reinforced by a commitment to the efficiency and effectiveness of markets. They also help maintain a commitment to the non-intervention in the functioning of such markets. As accounting is seen to provide a window into markets, questions about the "fair", "effective" or "efficient" functioning of markets may be re-framed as questions about existing accounting practices.

## **3.2** The role of Management Accounting

### **3.2.1 Economic perspective**

Ward (1992) discussed the role of management accounting and claimed that in most businesses it is normal to separate the financial function into three main roles. These are:

(a) Recording the financial transactions of the business and externally reporting to shareholders the historic financial results of these transactions (i.e. financial accounting).

(b) Raising the funds required by the business in the most appropriate manner (i.e. financial management or corporate finance).

(c) Supporting the managers of the business in the financial decision-making process (and being part of that management team, i.e. management accounting).

<sup>&</sup>lt;sup>38</sup> See the role of OFWAT in the UK water industry (section 7.3.3).

The separation of management accounting from financial accounting is fundamental in highlighting the fact that management accounting does not concentrate wholly on recording past events or on presenting externally published financial statements.

Bromwich and Bhimani (1989) in their study of the development of management accounting in Britain highlighted some key issues. These include: (a) the role of the management accounting system for purposes of strategic decision-making (compared to those of everyday control) (b) The extent to which the system influenced, or was influenced by, organisational structure (c) The extent to which non-routine accounting information was used. Table 3.2 shows the development in management accounting.

Development	Date
Cost and management control information	1890
Scientific management	1902
Break-even charts	1903
Standard costing and variance analysis	1908
Centralised accounting systems with decentralised functional organisation	1900
Capital and operating budgets	1910
Centralised control and decentralised responsibility	1920
Separation of financial and cost accounting	1923
Residual income method	1940
Simplex method for linear programming	1944
Discounted cash flow	1950
Total quality management	1950s
Cusum charts	1954
Optimum transfer pricing	1957
Computer technology	1960
Opportunity cost budgeting	1966
Zero-base budgeting	1969
Information economics and agency theory	1970s
Just-in-time scheduling	1970s
Activity-based costing	1980s
Target costing	1980s
Value-added management	1980s
Activity-based management	1990s
Benchmarking	1990s
Strategic management accounting	1990s
Supply chain management	1990s

# Table 3.2: The historical development of management accounting

Source: Adapted from Smith, M. (1995)- " New Tools For Management Accounting", pp. 6-8

The investigation into the role of management-accounting information (in complex operations) has focused on the interaction of disciplines in actual practice and the

formalisation and correction of deficiencies. The pioneering work of Robin Cooper, Tom Johnson and Robert Kaplan has been central to such developments since the mid-1980s.

Scapens (1991) found that while there is a large and growing tool-chest of modern management accounting techniques, (including cost-volume-profit analysis, activitybased costing etc.), "little can be said at the theoretical level about the techniques (tools) which should be used in practice" (p. 219). He goes on to argue that before the gap between theory and practice can be closed, "researchers must examine the various roles which management accounting fulfils within the organisation" (p. 221). Scapens (1991: 10) notes that "the terms cost accounting and management accounting are now used synonymously in textbook titles".

Dietrich (2001) outlines different economic perspectives on the firm and examines implications for the role and functioning of management accounting systems. He summarises the different approaches to management accounting in a simple 2 \* 2 matrix set out in figure 3.2

Figure 3.2: The different roles of Management Accounting Knowledge

	That	How
Constraining	(a) Passive optimiser	(d) Active regulator
Authority		
<b>Co-ordinating</b>	(b) Passive co-ordinator	(c) Active co-ordinator

Source: Dietrich, M. (2001)- "Accounting for the Economic firm", p. 17

Before the role of management accounting can be discussed, certain terminology needs to be clarified (e.g. 'knowledge that', 'knowledge how', 'authority constraining' and 'authority co-ordination'). Loasby (1998) distinguishes between "knowledge that" and "knowledge how". "Knowing that" is knowledge of facts and relationships and is the result of formal education and the news. According to Loasby, traditional production functions are based on "knowledge that". Dietrich (2001) argues that if the use of institutions is based on "knowledge that", the necessary predictability will be generated by rules that operate as general constraints. This occurs in an optimising world where

rules act as parameters to decisions (i.e. institutions are part of a non-economic background to economic activity).

"Knowing how" is the ability to carry out appropriate actions to achieve desired results. This includes knowledge of a particular skill and of when and where skills should be applied. Dietrich (2001) claims that we can view "knowledge that" and "knowledge how" as respectively 'passive' and 'active', use of information. If institutions are underpinned by "knowledge how", spatial and temporal specific learning is necessary. This learning is a basic requirement for institutions to involve active participation in a social context, (rather than just operate rule-defined behaviour).

Dietrich (2001) argues that different perspectives on the firm, view authority as either being necessary for co-ordination purposes, or see it as a constraint on economic activity. If authority exists for co-ordination purposes, it is viewed as having a positive role in the perspective concerned. For instance, clearly defined authority (and the coordination it provides), might be argued to have efficiency advantages. Authority as a constraint, is viewed (in general terms) as influencing relationships to promote the interests of particular economic agents. This constraining effect might involve ex-ante intentions (as with monopoly control over resource allocation).

There are four different roles of management accounting. (a) Management accounting is a passive optimisation tool, where authority is viewed as a constraint, and informationuse described in a "knowledge that" manner. This is the world of textbook accounting. Hence the requirement for efficient and effective monitoring fails to recognise the importance of *'bounded rationality'* (Dietrich, 1993). The (implicit) complete contracting is based on a universal "knowledge that" rather than learning-based "knowledge how".

(b) Much of the recent literature in accounting appears to have adopted a transactioncost logic (neo-institutional economic theory). This is accounting as 'passive coordinator'. Accounting systems are viewed here as efficient responses to particular environmental conditions. Authority is one means of co-ordinating resource allocation and is necessary to economise on market-based transaction costs. The "knowledge that" basis of an accounting system is more problematic. We can refer to site and human asset specificities here (see section 3.3.2.2). Any management accounting system involves set-up costs and 'learning-by-doing', (i.e. the system itself will have firm specific characteristics).

(c) The 'competence' theory of the firm<sup>39</sup> has been shown to involve a more active role in management accounting (but this is still one of co-ordination). This active role involves recognising accounting as informal and routinised, as well as a formal activity. Authority is seen as co-ordinating, and decision-making is based on "knowledge how" rather than "knowledge that". The competence approach emphasises the rationale of the firm, and its long-running competitive advantage. This theory sees institutional characteristics as derived from individual decisions and activities, (i.e. it is based on methodological individualism).

(d) Management accounting prevents what might be called an 'active regulator' (when accountants use constraining authority and "knowledge how" information). "Knowledge how" and learning are linked to co-ordination (described in terms of direction and routines, and hence institutional development). A constraining authority - ("knowledge how") has, what might be called a 'reserve logic'. Institutions are given an existence that logically precedes individual activity, (i.e. it is methodologically holist). Dietrich (2001) argues that the limitations in a constraining authority - ("knowledge how"), should be viewed as leading to more than just exchange-based economy. Dietrich (2001) claims that implications of this general framework affect accounting in two ways: (a) By viewing accounting as a professionalised activity (B) by incorporating a symbolic as well as instrumental significance.

Dietrich and Roberts (1997, 1999) suggest that traditional economic approaches to professionalisation are incomplete as they adopt either efficiency or power. Dietrich (1997), (using institutional analyses of the role of accounting - e.g. Armstrong, 1985; 1987; Fligstein, 1987), shows professional accounting control within an organisational

<sup>&</sup>lt;sup>39</sup> The basic difference between the contracting and competence perspective can be understood (in game theoretic terms) as the difference between the prisoner's dilemma and co-ordination problems (Foss, 1993; Longlois and Robertson, 1995). Contracting (prisoner's dilemma) analysis is oriented towards issues of motivation and control, (given the divergent incentives of the factors involved). The emphasis is on issues of organisation and co-ordinating behaviour rather than production. Conversely the competence perspective of the firm sees its basic nature in terms of the co-ordination of divergent capabilities. The way in which this co-ordination problem is solved, influences a firm's productive capabilities.

context, can become "locked-in". As Perkin (1989) found, this organisational dominance of professionals reflects a wider social dominance. It follows that accounting can channel organisational activity partly by formal rule-setting, but also by management of organisational context. This perspective is consistent with that suggested by Roberts and Scapens (1990) in which accounting is viewed as enhancing the position of particular organisational actors relative to others.

The second context is that organisational activity (specifically accounting activity), can be viewed as having a symbolic, as well as instrumental significance. For instance, Carruthers (1995) suggests that management accounting is a legitimation tool for a firm's 'outside' world. If accounting practices are viewed as having purely instrumental significance, (whether this is contracting or competence-based), such legitimation is irrelevant. It only becomes significant if institutionalisation processes are based on "participation in a social context".

#### **3.2.2** The problem of management accounting in organisation services

During the 1980s, criticism of management accounting practices began to appear in professional and academic accounting literature. The most prominent critic was Robert Kaplan, who, in a series of articles in the mid-1980s, questioned the relevance of current practices (Johnson and Kaplan, 1987). The principal criticisms of current management accounting practices can be summarised as follows: (a) Conventional management accounting does not meet the needs of today's manufacturing and competitive environment (b) Traditional product costing systems provide misleading information for decision-making purposes (c) Management focuses almost entirely on internal activities and relatively little attention is given to the external environment in which the business operates

Bromwich and Bhimani (1989) identify five persistent weaknesses in management accounting: (1) The subservience of management accounting to external financial accounting requirements (2) The lack of strategic considerations in management accounting and project appraisal (3) The reliance of management accounting on redundant assumptions concerning manufacturing processes (4) The maintenance of traditional assumptions in performance evaluation (5) The continued short-term orientation of performance measurement Their findings are useful in providing indications of the likely direction of future research. In particular they identify: (a) the need to identify cost drivers which link processes to cost of output (b) the need to measure activity-based costs where these are meaningful and can lead to significant benefits (c) the benefits of non-financial accounting information in different manufacturing environments (d) the opportunity to incorporate both qualitative and non-financial, quantitative information into management accounting information systems (e) the increasing relevance of a strategic approach to management accounting.

Bjornenak and Olson (1999: 237) argue that conventional management accounting has focused primarily on the scope dimension<sup>40</sup> (i.e. what is accounted for, and for what period of time). The scope dimension includes sub-dimensions such as cost items, cost objects and allocation bases. Bjornenak and Olson (1999) argue that the system dimension<sup>41</sup> is almost non-existent in conventional management accounting (p. 328) and that the choice of time perspective plays no important role in management accounting.

Bjornenak and Olson (1999: 333) argue that the conventional view of a management accounting system includes a continuous view of the system itself. This means that the system is assumed to have longevity. This idea of the 'continuous system' was undermined when Madsen (1963) introduced the variability-based accounting system<sup>42</sup>. Bjornenak and Olson (1999) argue that the conventional management view of a management accounting system (one which uses different costs for different purposes), is challenged by the "different system for different purposes perspective". The "different systems perspective" operates at different levels in the organisation. They argue that the conventional wisdom of management accounting shows information symmetry as a way

<sup>&</sup>lt;sup>40</sup> The scope system includes three elements: (a) descriptive variability factors or descriptive objects (DO), (its purpose is to describe the variations in costs related to the object) (b) Causal variability factors (CVF) (describes the cause of variation) (c) Time. (The period of time accounted for, p. 328).

<sup>&</sup>lt;sup>41</sup> The dimension system is the link between the users of the system and how the system is designed. There are two aspects in the system dimension: (a) the lifetime of the system (b) the integration of user aspects in management accounting (Bjornenak and Olson, 1999: 328).

<sup>&</sup>lt;sup>42</sup> In this system, data collection was considered a continuous task, while reporting was made on both a continuous and ad hoc basis (Bjornenak and Olson, 1999: 333).

of achieving goal congruence through aggregation of cost data. Systems are designed in a top-down fashion because the centre is assumed to know which information has no value.

Management accounting faced a "crisis of confidence" (Dent, 1990) following "an acute failure of managerial expertise, and of calculative expertise in particular" (Miller and O'leary, 1993: 188). Following the "lost relevance" of management accounting (Johnson and Kaplan, 1987) there was development of a new market for new accounting - (e.g. activity-based costing, throughput accounting, target costing, economic value added). Traditional systems were condemned as producing irrelevant (mis)information (Jones and Dugdale, 2001).

Fitzgerald, et al., (1991) have argued that conventional standard costing and variance analysis do not provide useful information for cost control. The majority of costs are fixed with respect to changes in unit production volume, (which is often the case in service industries). Management accountants (Johnson, 1992) and operations managers (Nanni, et al., 1992) both agree that such techniques are irrelevant, untimely and cause manipulation of operations to meet the standard/budget.

Some authors have suggested ways of making management accounting more relevant via either activity-based costing (Cooper and Kaplan, 1988) or throughput costing (Galloway and Waldron, 1988). However, these initiatives would only address part of the problem. Service organisations<sup>43</sup> today compete on a range of dimensions, (not just cost and price), so a good planning and control system cannot focus on the financial dimension alone. Service costing systems must be seen in their wider context, (i.e. as part of management information systems<sup>44</sup> which embrace non-financial information).

<sup>&</sup>lt;sup>43</sup> Service organisations have five characteristics (Fitzgerald, et al., 1991). (1) The customer is present during the service delivery process, (which poses threats and offers opportunities to serve managers). (2) Many aspects of services are intangible, making it difficult to know what the customer values. (3) Services are heterogeneous in two ways: (a) The frequently high labour content means that service standards vary from person to person or from day-to-day. (b) Customer expectations may also vary. (4) Production and consumption of most services is simultaneous, so they cannot be counted or tested prior to delivery. Finally, most services cannot be stored because they are perishable.

<sup>&</sup>lt;sup>44</sup> Management information systems are defined as "a system that enables both internal and external information to be reported at operational, tactical and strategic levels". Management information systems cover such dimensions as quality, flexibility and innovation, as well as finance, within an appropriate control model (Brignal, 1997).

It is commonly agreed that businesses today are competing across a range of variables, (not just cost and price), therefore their management information systems should reflect this (Kaplan, 1983; 1984; Sink, 1985). Fitzgerald et al., (1991) have argued that the measurement of service business performance across six dimensions, is a vital part of feed forward, (feedback control at the strategic business unit level). These six dimensions have been divided into categories. They measure the results of competitive success. "Competitiveness" includes market share, and financial measures include unit cost, profit and value added. The other four dimensions measure aspects of the determinants of that success (quality, flexibility, resource utilisation and innovation). Cost leaders will tend to focus on measures of resource utilisation and cost. Those following some form of "differentiation" strategy will slant the mix and weighting of measures of determinants towards their differentiating aspect, (such as quality). Measures of competitiveness will be necessary under either strategy. There will be interactions and 'trade-offs' made between the six dimensions as priorities change through the life cycle. These trade-offs may be of such importance to competitive success that they must be included in an organisation's reward system.

## **3.3 Theory of management accounting**

The academic production of accounting theory is closely related to economics (Jones and Dugdale, 2001) - (in terms of the theoretical approaches adopted, the disciplinary background of the individuals involved, and the organisational setting of university accounting departments). The issues addressed by those who construct and promote accounting theory may be very different from the concerns of practitioners - (shown by the long-running discussion of "the gap between theory and practice" - Edwards and Emmanuel, 1990). Similarly, Johnson and Kaplan (1987) cite the use of simple (abstracted) economic models by accounting academics in US business schools, as one of the key factors for management accounting losing its relevance for management at local level.

#### 3.3.1 Management Accounting from a neo-classical economic perspective

The neo-classical theory of the firm seeks to represent the choice behaviour of the firm as a rational solution to the single-person-decision problem (Walker, 1998). Neoclassical theory has been criticised for its inability to address fundamental economic realities. Nelson and Winter (1982: 5) criticised "the inability of the prevailing theory to come to grips with uncertainty, bounded rationality, the presence of large corporations, institutional complexity, or the dynamics of actual adjustment processes". In recent years various heterodox schools have emerged within economics. These include radial economics, institutional economics, post Keynesian economics, neo-Marxist economics. At the heart of many of these alternative approaches is a concern about the fundamental assumptions of (neo-classical) economic theory (Scapens, 1994). In particular, the assumptions of economic rationality and market equilibrium.

The two core assumptions of the neo-classical economic<sup>45</sup> stance are: (a) the rational maximising economic agent (b) the general equilibrium achieved by market forces (these have been increasingly challenged by institutional and other heterodox economists, Scapens, 1994). Institutional economists argue that these core assumptions are inadequate for such theoretical purposes as understanding economic change and analysing the process of transformation in a modern economy. They see both individual behaviour and the working of the market as influenced by institutions (which are social phenomena in need of theoretical explanation). Machlup (1967) claim that neo-classical theory of the firm was developed for market and industry analyses, not as description of what managers actually do. Scapens and Arnold (1986) argued that neo-classical theory was not intended as an explanation for the behaviour of managers within firms. The neo-classical theory of the firm does not analyse institutional arrangements which govern economic activity within the firm (Scapens, 1994). Walker (1998) argues that the neo-classical model was not designed for understanding the complex issues that arise in choosing an optimal organisational form.

Simon's concept of "bounded rationality" (in relation to Rational maximising economic agents), has had a significant impact on institutional economics (Scapens, 1994). Even the new institutional economists, (e.g. Williamson) claim to be influenced by Simon. They link Simon's concept of bounded rationality with economising on transaction costs. They thereby retain the concept of optimising economic rationality, (although rational choices are bounded by available information and information-processing ability). Simon argues that individuals must "satisfy, because they cannot maximise".

<sup>&</sup>lt;sup>45</sup> Scapens (1994) argues that old institutional economic theory has developed in response to neo-classical analysis. Old institutional economic theory makes the "institution" the unit of analysis, rather than the rational maximising behaviour of individual decision-makers. Neo-institutional economic theory has developed as extension to neo-classical analysis.

He links bounded rationality with 'satisfying', rather than optimising behaviour. In line with this view, old institutional economists argue that satisfying can be seen in rule-based behaviour.

Economic-based approaches to theory of organisations assumes self-seeking rational optimisation on behalf of all economic agents. In the specific context of management accounting, there are two potential problems with this approach (Walker, 1998). The first is that by ruling out bounded rationality on the part of decision-makers, one automatically rules out any explanation of a demand for accounting reports (which help decision-makers to understand their decisions). If all individuals are super-rational then they have no need for decision aids of any kind.

The second problem with super-rationality is that it implicitly carries the assumption that all individuals share a common understanding of the world in which they operate. This is an understanding which they all appear to share. The core problem of this approach is that it fails to consider a socially constructed reality. The influence of social and cultural factors on organisational structures cannot be easily captured by models which assume economic agents are adept at calculation.

The economics-based approach sees management accounting as "the provision of information which is designed to enable decision-makers to make optimal decisions" (Scapens and Arnold, 1986). Accounting research which adopts agency theory or transaction cost economics, focuses on equilibrium and optimal solutions. This form of research may introduce new techniques, but it does not assist our understanding of how such techniques become an established part of accounting practice in organisations. For example, it may be possible to demonstrate that the use of activity-based costing (ABC) is more 'rational', than cost allocations based on labour hours. However such 'rational' analysis cannot explain why ABC techniques come to be used in some organisations, but not in others. It also fails to anticipate potential problems or difficulties faced in the implementation process, (Burns and Scapens, 2000).

When viewed from a perspective of management accounting, (neo-classical economic theory can be seen to adopt a conventional wisdom's view of management accounting) for the provision of information in decision-making. This is associated with the design

of optimal organisational structures. Neo-classical theory of the firm provides an inadequate theoretical framework. The standard theory of the firm is based on the assumption of absolute certainty. This assumption is problematic. It cannot explain organisational structures, because there is no rationale for the existence of firms in a world of 'perfect' certainty. In other words, the conventional theory of the firm is inconsistent. Economic uncertainty *must* form an essential part of any attempt to explain why organisational structures are what they are. Conventional wisdom sees accounting as an outcome of "rational" behaviour, - aiming to maximise company profits through the provision of new "optimal" information (Hopwood, 1987; Burns, 2000). This "technical" view is grounded in neo-classical economic theory of the firm (Scapens and Arnold, 1986).

## **3.3.2 Management Accounting from an Institutional perspective**

As mentioned in section 2.1, there are three theories used in accounting literature namely: neo-institutional economics (transaction cost), old institutional economics and neo-institutional sociology.

## 3.3.2.1 Old institutional economic theory and Management accounting

Accounting from an Old institutional economic perspective can be understood in terms of the change in accounting routines that may be part of the assumptions and beliefs of an organisation (Scapens, 1994; Burns, 1997; Burns and Scapens, 2000). These assumptions and beliefs, (common to organisational members) are called "institutions".

Old institutional economics was introduced by Veblen (1898; 1919: 239) who defined an institution as the "settled habits of thought common to the generality of man". OIE is undergoing somewhat of a revival in research (Hodgson, 1988; 1993; Tool, 1993). It emerged in opposition to 'static' rational-actor economic theorising, and should not to be confused with neo-institutional economics.

Old institutional economic theory is "a thought or action of some prevalence and permanence, which is embedded in the habits of a group or the customs of people" (Hamilton, 1932: 84). Burns and Scapens (2000) apply old institutional economic theory concepts to show how accounting practices can become routinised. They explain

the nature of organisational rules and routines in the relationship between action and institution, (from old institutional economic theory).

Burns and Scapens (2000) argue that habits are personal. Routines may involve groups rather than individuals. Therefore, routines represent patterns of thought and action which are habitually adopted by groups. Organisational routines play an important role in the relationship between actions and institutions.

Habits and routines are be particularly useful in dealing with the uncertainty and complexity of everyday life. In a complex and uncertain organisational world (i.e. the GOGCWS), it is impossible for an individual to make choices which are 'assumed' in rational economic models (Simon, 1957). Given that routines may involve groups, they become the components of institutions. Institutions may enforce formal and social coherence upon human activity, (partly through continuing production and reproduction of habits of thought and action, Scapens, 1994). Parsons (1940: 190) argued that "institutions are 'normative patterns' which define what it is to be part of a given group or society. Institutions are proper, legitimate, and expected modes of action or social relationship".

Scapens (1994) highlighted the importance of rules when he argued that "rules are necessary to co-ordinate and give coherence to the actions of groups of individuals". He argued that individuals are able to give reasons for following rules, and the rules themselves may be the result of earlier action. In this way, although rules and routines may give structure and coherence to individual action, the rules themselves will have emerged through actions (Scapens, 1994: 309). Institutional economists see the relationship between past actions and current rule-based behaviour, as complex and socially-constructed.

Burns and Scapens (2000) found a two-way relationship between rules and routines as illustrated in figure 3.3.



## Figure 3.3: The relationship between rules and routines

Arrow A refers to the first direction between rules<sup>46</sup> and routines. It is here that rules are established (e.g. budgeting procedures), and through their implementation, routine will emerge. Arrow B refers to routines. These routines either deviate from original rules, or were never originally set out in the form of rules. In such cases, it may be decided to formalise established routines in a set of rules. Hence the process moves from routines to rules.

In order to explore the role of rules and routines in the relationship between actions and institutions<sup>47</sup>, it is useful to begin with Giddens theory. Giddens (1984) looked at the relationship between actions of 'knowledgeable' actors and the structuring of social systems. He distinguished between systems (which comprise discernibly similar social practices, - reproduced across time through individual action), and structures which bind those social practices into systems. His theory assumes systems are not structures, but contain structures which are drawn upon in action.

Burns and Scapens (2000) argue that both Old institutional economic theory and Giddens theory contain systems. These are thoughts and actions of some prevalence, permanence and structure, (i.e. embeddedness in habits and customs). According to Old institutional economic theory, it is important to view rules and routines as interacting between the realm of action and realm of the institutional (Burns and Scapens, 2000). Figure 3.4. illustrates this.

Source: Adapted from Burns and Scapens (2000), "Conceptualising Management Accounting Change: An Institutional Framework"

<sup>&</sup>lt;sup>46</sup> Formulated rules may become modified when a group finds mutually acceptable ways of implementing them (Burns and Scapens, 2000).

<sup>&</sup>lt;sup>47</sup> Scapens (1994: 309) found there is no simple, one-way direction of causality between institutions and actions. There is a duality in the relationship between them. Institutions which influence actions, are themselves often the outcome of those actions. Hence, institutions do not determine actions, nor do actions create institutions. There is a duality between institutions and actions (Scapens, 1994).

### Figure 3.4: The process of Institutionalisation

#### **Institutional Realm**



**Realm of Action Key :** a = encoding b = enacting c = reproduction d = institutionalisation Source: Burns and Scapens (2000), p. 9.

The framework combines both synchronic<sup>48</sup> (arrows a and b) and diachronic (arrows c and d) elements. Burns and Scapens (2000) found that there are four processes.

The first process (arrow a) indicates the encoding of institutional principles into rules and routines. Existing routines will encode prevailing institutional principles, (which in turn will lead to the formulation of routines). This encoding process draws on assumptions which comprise institutional principles and follows the process of change into existing meanings, values and power.

The second process (arrow b) involves actors enacting the routines (and rules) which encode institutional principles. The third process (arrow c) occurs when repeated behaviour leads to a reproduction of routines.

The fourth process (arrow d) is the institutionalisation of rules and routines which have been reproduced through the behaviour of individual actors. This involves a disassociation of patterns of behaviour from particular historical circumstances. Rules and routines therefore take on a normative and factual quality, which obscures their

<sup>&</sup>lt;sup>48</sup> 'Institutions' constrain and shape action at specific points in time. Actions produce and reproduce institutions diachronically, (i.e. through their cumulative influence over time).

relationship with the interests of different actors. In this way, rules and routines become 'the way things are done'. In this way, routines themselves can be institutionalised. In other words, it becomes an established way of behaving. Routines become an unquestioned way of doing things. The institution assumes that these routines represent appropriate behaviour for a particular social group. The routines themselves are the local environment of the institution. If the institution is accepted in its external environment, the more likely it is to influence action and to resist change.

Management accounting in institutional theory (from an old institutional perspective) provides the basis for decision-making. It also assists in the formation of expectations and beliefs. The extent to which accounting practices give social coherence and meaning to organisational behaviour, and allows individuals, and groups within firms to give meaning to their day-to-day activities.

Burns and Scapens (2000) argue that Old institutional economic theory provides a focus for organisational routines and institutionalisation. In management accounting, OIE focuses on change in organisational routines. Burns and Scapens (2000) see management accounting as a routine, and therefore potentially institutionalised. They believe that management accounting can come to underpin the 'ways of thinking and doing' in organisations (Mouritsen, 1994). This means they view management accounting as an institution within an organisation.

### 3.3.2.2 Neo-institutional economic theory and Management accounting

Neo-institutional economic theory has been studied by many researchers (e.g. Williamson, 1975; 1985; 1996; and North, 1990). Neo-institutional economic theory has been employed in accounting research (Spicer and Ballew, 1983; Johnson, 1983; Spicer, 1988; Colbert and Spicer, 1995). Scapens (1994) argues that neo-institutional economic theory remains firmly rooted in "static" neo-classical economic theory. On the other hand, Old institutional economic theory attempts to explain phenomena in "processual" terms, (examining why and how things become what they are (or are not), over time.

As Burns (1998) points out, the "markets and hierarchies" theory has probably exerted more influence on management accounting research, than any other branch of newinstitutional economics<sup>49</sup>. The discussion of this sub-section will restrict itself to the link between management accounting and transaction cost theory. The basic components of the transaction analysis framework are illustrated in table 3.3.

Transaction	The cost of running the system, and include ex-ante	
	costs (e.g. drafting and negotiating contracts), and ex-	
	post costs (e.g. monitoring and enforcing agreements)	
Governance Structures		
Markets versus Hierarchy	- Markets - where transactions are conducted between	
	firms, (e.g. manufacturer and wholesaler)	
	- Hierarchies - (where transactions are conducted within	
	a firm, (e.g. vertically-integrated supply chains) are	
	compared for their relative efficiency in sustaining	
	recurrent trading relationships	
Assumptions regarding human nature		
Bounded rationality	Decision-makers intend to be rational, but are	
	constrained by their limited information-processing and	
~	communication abilities	
Opportunism	Decision-makers may unscrupulously seek to serve their	
	own self-interests, and it is difficult to determine who is	
	trustworthy and who is not	
Degree of asset specificity	Problematic, when an inter-firm relationship is	
	supported by specific assets. (Degree of asset	
	specificity, whose values are limited to outside the focal	
	relationship)	

Table 3.3: The Basic Components of the Transaction Analysis Framework

Source: Standifird, S. and Marshall, R. (2000)- "The Transaction Cost Advantage of Guanxi-Based Business Practices", p. 17:18

Three factors determine the extent of transaction costs: bounded rationality, opportunism and asset specificity (Dietrich, 2001). In the absence of any of these

<sup>&</sup>lt;sup>49</sup> The application of neo-institutional theory in accounting is not restricted to perspectives derived from economics. It is also evident that neo-institutional sociology plays an important role, (for example: Covaleski and Dirsmith, 1998; Covaleski et al., 1993).

factors, transaction costs will be zero, and asset specificity is given a pivotal role. Two types of specificity are particularly relevant for the current discussion. "Site" and "human"<sup>50</sup> specificity arise from 'learning by doing'. These asset specificities lead to ex post lock in between contracting partners. Hence, they aggravate bargaining and policing problems if 'arms-length' contracting is adopted. Such problems can be readily managed by using an authority relationship which is accepted by all contracting parties.

An existing management accounting system is consistent with a transaction cost logic. The development and use of accounting systems involve resource expenditures which transactions cost theorists deem as important. Systems provide an important means of codifying rights and responsibilities. Without this codification, the policing and economising of behaviour would be difficult (if not impossible). Changes in accounting system can be explained as response to change in the factors which determine transaction costs (e.g. bounded rationality and asset specificity). The historical development of management accounting has been analysed (Johnson, 1983; Johnson and Kaplan, 1987), likewise the internal organisation (Spicer and Ballew, 1983), transfer pricing (Spicer, 1988; Colbert and Spicer, 1995), and financial control (Ezzamel, 1992).

Zimmerman (1997) approaches management accounting from a neo-institutional economic perspective. He argues that management accounting cannot be properly understood without a theory of the nature of an organisation. Zimmerman's (1997) contribution shows how the transaction costs approach can provide a general framework to which all standard management accounting techniques can be applied. Zimmerman sees management accounting as a general organisational design problem. The design of an organisation involves three key issues: (1) the optimum division of decision rights (2) the establishment of systems for measuring and evaluating the performance of decision centres (3) the choice of system for linking rewards and promotions to measured performance.

<sup>&</sup>lt;sup>50</sup> In addition to site and human asset specificity, Williamson (1985) speaks of "physical" and "dedicated" asset specificity. The former involves specialised equipment, the latter, - generalised investments which produce outputs for specific customers. From a transaction cost perspective, human and site specificity lead to vertical integration, (i.e. firm-based organisation), whereas physical and dedicated specificity lead to quasi integration (Dietrich, 1994).

Transaction cost economics has been criticised for failing to recognise the importance of the institutional environment (Robins, 1987). Williamson's (1991) discussion of the hybrid form is a direct response to this criticism. Predictions concerning the hybrid form revolve around the existence of contract law (Williamson, 1991). The hybrid structure is one exception (Granovetter, 1985). Contract law characterises institutional structures. Lack of contract law is assumed to represent a lack of institutional structure (Nee, 1992; Xin and Pearce, 1994).

The main weaknesses of Zimmerman's (1997) theory is that he fails to discuss any alternative approaches, (and their respective strengths and weaknesses). He also chooses to ignore all sociology-based literature on management accounting.

Hopper and Armstong (1991) identify fundamental problems in the theoretical framework of neo-institutional economic theory (transaction costs economics). These are (1) the term "transaction costs" has never been precisely defined (2) There is a failure to consider links between product markets, power and organisational forms (3) There is a failure to take into account the influence of social and political processes on the development of firms (4) A failure to consider the potentially exploitative nature between capitalist firms and their employees. In relation to point one, Walker (1998) argues that 'labour process theory' is also guilty of using terms which are difficult to precisely define, (e.g. "class", "power"). "It is difficult to see why product market conditions should influence organisational form (at least as a first order effect"), (his response to the second point). The problem with considering political and social forces is the implication that one cannot study any economic phenomena in isolation from everything else", (his response to the third criticism).

## 3.3.2.3 Neo-institutional sociology theory and Management accounting

Scott (1987) distinguishes four sociological research directions and offers a definition of institutional theory which has been utilised in organisational analysis. The first approach sees the structure of the organisation as adaptive, and the result of individual and environmental influences/constraints (Selznick, 1957). 'Institutionalisation' refers to this adaptive process, (Scott, 1987: 493).

The second approach sees institutionalisation as a process of 'creating reality', and is based on the idea of a 'shared reality', (i.e. a construction created in interaction between people), (Berger and Luckman, 1967). 'Institutionalisation' is viewed here as the process whereby actions are repeated over time, and are assigned similar meanings by the subject and by others. In this process, individuals come to accept and share a definition of reality (Scott, 1987: 495).

The third approach has its roots in the work of Berger and Luckman (1967). Institutionalisation is seen not as a process here, but as a set of elements (Scott, 1987: 497). Meyer and Rowan (1977) suggest that organisations adapt to the wider environment in order to gain legitimacy, resources, stability and survival. An organisation can gain legitimacy by turning to experts in a specialist field, (e.g. accounting). The external symbolic, rhetorical or political order is a major influence on the internal structural order. DiMaggio and Powell (1983) propose three ways in which an organisation conforms to symbolic order: (1) coercive (2) mimetic and (3) normative processes.

The fourth approach sees institutions as distinct societal 'spheres'. There are 'symbolic' and 'behavioural' systems, but these differ from institutional 'spheres' (Scott, 1987: 499).

Sociological theory shares a concern with the normative institutionalist approach. Sociology theory looks at the role values play in shaping behaviour within a structure. Selznick (1957) distinguished between organisations as 'structural expressions of rational action' and organisations as 'normative and adaptive'. The latter view is similar to the March and Olsen perspective (1989, 1995).

What appears to differentiate the sociological perspective from the normative institutionalist perspective is the relationship with the environment. This view, is common with one major strand of organisational theory. It looks at organisations as both dependent upon the resources of their environment, in turn, shaped by that environment. March and Oslen (1989, 1995) see institutions as largely autonomous 'actors', (at least after their initial formation). The sociological perspective places an institution on the periphery.

Neo-institutional sociology (DiMaggio and Powell, 1991; Scott, 1995) has previously been employed in accounting research (Covaleski et al., 1993; 1996; Carruthers, 1995). Although there is a significant overlap between Old institutional economic theory and neo-institutional sociology, the former deals directly with the emergence, continuity and change of institutions through time (Scapens, 1994). Neo-institutional sociology focuses more on "macro-institutions"<sup>51</sup> while old institutional economic theory examines micro-institutions within organisations (Scapens, 1994: Burns, 1996; 1997). Neo-institutional sociology research in accounting, focuses predominately on extra-organisational institutions (their social, economic and political effect) (Covaleski et al., 1993; 1996; Carruthers, 1995; Fligstein, 1998).

Miller (1994: 2) argues that "Whilst accounting shares with statistics the ability to translate qualities into quantities, it does so largely by translating these qualitative differences into financial values which need no further referent".

## **3.4 Conclusion**

This chapter has shown that there are different views of accounting. It has highlighted the relationship between accounting and organisational decision-making. This led the researcher to address the question "What role has accounting played in the GOGCWS?" (see chapter 11.2.1). This chapter has stressed the importance of the social aspect of management accounting. It concludes that accounting cannot be conceived as purely an organisational phenomenon. This led the researcher to examine the external environment of the GOGCWS (see chapter 9). It also led him to explore the social aspect of privatisation in the Egyptian water industry (see section 10.1).

The chapter has reviewed different views of management accounting and its different roles (passive optimiser, passive co-ordinator, active regulator and active co-ordinator). It highlights the problem of management accounting generally and in organisation services particularly.

This chapter has argued that the institutional theory approach has the potential to provide a useful theoretical framework for understanding management accounting (cost

<sup>&</sup>lt;sup>51</sup> For example a generally-accepted accounting principle at society level (Mezias, 1990).

management) practices. This led the researcher to adopt two propositions from institutional theory in order to explain the findings of this research. These two propositions are: firstly, government policy (coercive isomorphism) shaped the limited use of cost management for efficiency, optimisation and strategy; secondly organisations are the 'theatre' in which institutions are visible.

This chapter has given an overview of management accounting theory. It has criticised and shown the limitations of management accounting from a neo-classical economic stance. The chapter has outlined some alternative approaches to management accounting, (namely neo-institutional economic theory, old institutional economic theory and neo-institutional sociology theory). The next two chapters are going to discuss the elements of cost management (chapter four focuses on the first two elements: efficiency and optimisation and chapter five indicates the third element which is strategy).

## **Chapter 4**

## Models of cost management

## **4.0 Introduction**

The aim of this chapter is to illustrate literature relating to a model of cost management. For this purpose, the chapter is divided into six sections.

The aim of the first section is to examine the concept of efficiency (i.e., Pareto efficiency, technical efficiency, and economic efficiency) and the ways of using cost management to increase efficiency.

The second section illustrates the link between institutional theory, transaction cost and efficiency. It argues that efficiency framework is favoured by transaction cost theories.

The third section reviews literature relating to activity based cost (ABC), emerging ideas and its core idea, the difference between ABC and traditional accounting systems, the adoption of ABC and its criticisms, the limitations of ABC and alternative approaches. It also highlights the relation between ABC and institutional theory.

The fourth section discusses activity-based management (ABM) as ABC cannot totally eliminate arbitrary allocation of overheads. The key extension of ABC is into ABM. This section discusses emerging literature on ABM and the similarity and differences between AMB and ABC, the difference between the views of traditional cost and ABM, and its cost and importance.

The final (fifth) section outlines the main conclusion of this chapter.

## 4.1 Efficiency

The aim of this section is to review literature relating to the struggle for efficiency. Control has two problems:

(a) The regulation of the processes of the formulation of purpose

(b) The regulation of the processes of purpose achievement

Thompson (1967) suggested these problems could be further understood through three major themes; firstly, the establishment of purpose; secondly, the pursuit of effectiveness; and finally, the struggle for efficiency.

The control of efficiency via the loop of the budgeting and reporting process, in connection with standard costing as a part of design and planning process, leads to the emphasis given to calculation and analysis of variances. For example Ezzamel, et al., (1994) stated that companies are focusing on improving the technical capabilities of their management accounting system in order to improve their costing calculations and therefore enhance their efficiency. But control of efficiency and standard costing might have become obscured by the attention paid to the observation and analysis of variances.

Pareto improvements<sup>52</sup> have come to be labelled improvements in efficiency, and, Pareto optimal situations have come to be labelled efficient situations, and in the process the special assumptions that "lie behind this persuasive shorthand have tended to become obscured" (Williams and Giardina, 1993: 66). It is quite difficult to resist a proposal which has apparently been demonstrated to improve efficiency, yet there may be very good reasons for doing so in certain circumstances.

Efficiency is the relative amount of inputs used to achieve a given level of output (Horngren, et al., 1999: 539). Efficiency is the relationship of outputs to given sets of inputs (Berry, et al., 1995). This relationship can be expressed in many ways. Accountants find themselves relating the value of outputs in the market-place to the value of inputs in the factor market place: raw material and supply, labour, management, capital and financing, and technology, concluding that efficiency gains

<sup>&</sup>lt;sup>52</sup> The general rule is that, if such a change means at least one person is better off, and leaves nobody worse off, that change is "Pareto improvement", and if the system reaches a situation in which it is impossible to make someone else worse off, that situation is "Pareto optimal" (Williams and Giardina, 1993: 66).

might occur if either the value of the outputs rises per unit of input, or the cost of inputs fall per unit of output. It is helpful to notice that even technical efficiency gains measured in this way might be confusing in relative price changes and gains in technical efficiency during the transformation process.

Thompson (1967) argued that technical efficiency can only be discussed in a bounded system where the boundaries are closed for analysis, and that effectiveness can only be discussed in an equally, but different, bounded system. Purpose will tend to be unbounded and to be the product of social interaction.

Figure 4.1: Bounding domains for purpose, effectiveness and efficiency



Source: Berry, et al., (1995)- "Management control", p. 6.

Does efficiency variance occur in fixed manufacturing overheads? Horngren, et al., (1999) argue that the efficiency variance<sup>53</sup> pertains only to variable manufacturing overhead. There can be no notion of efficiency for fixed manufacturing overhead because this does not alter with changes in output level. With fixed capital equipment, the most important issue is not "what you have but how you use it". Against this, management accounting texts assume that capacity and efficiency are fixed at the level set by "currently variable standards based on efficient operating conditions" (Ricketts and Grays, 1988).

Miller (1989) argues that the vocabulary of cost provides the basis for an economic language of organisational motive, which subsequently informs the construction of new operational concepts of accountability and performance measurement. Priority objectives were seen in terms of "reviews of operating costs", "cost saving investment programmes" and "close examination of the potential for further cost saving". Better

<sup>&</sup>lt;sup>53</sup> Efficiency variance: the difference between the actual quantity of input used and the budgeted quantity of input that should have been used multiplied by the budgeted price. Also called input efficiency variance or usage variance (Horgnern, et al., 1999: 523).

management information systems, new performance measures, cost base reviews, new methods of budgeting, more rigorous business planning, the introduction of new technology and reductions in manpower were all extolled as a means of securing reductions in costs. Achievements and measures of performance were reported in terms of "cost saving".

Al-Hazami (1995) in his literature review on management accounting and the accounting and control paradigm concludes that three ways of using cost management to increase efficiency are discussed: throughput costing (Waldron and Galloway, 1988; 1989a, b; Darlington, et al., 1992; Dugdale and Jones, 1994; Tayles and Walley, 1994), benchmarking and target costing<sup>54</sup> (Monden and Hamada, 1991; Tani, et al., 1994), and cost leadership<sup>55</sup> (Porter, 1985; Simons, 1990). These ideas on cost management are driven by the economics of internal organisation and transaction cost analysis (Coase, 1937; Alchian and Demsetz, 1972; Williamson, 1975; Jensen and Meckling, 1976). A brief discussion on each of them - three ways of using cost management to increase efficiency - is given below:

The first method of using cost management to increase efficiency is throughput accounting (TA). Galloway and Waldron, published a series of articles in Management Accounting in late 1988 and early 1989 explaining the ideas of throughput accounting. They identify three concepts which should be taken into consideration:

(1) Manufacturing units are an integrated whole whose operating costs in the short term are largely predetermined. It is more useful and infinitely simpler to consider the entire cost, excluding material, as fixed and to call the cost the "Total Factor Cost" (TFC). Dugdale, et al., (1996) illustrate the core idea of this concept and say Galloway and Waldron point out that any approach to allocation of overhead, particularly depreciation, is likely to generate misleading information (for example, cost per hour rate) and so lead to poor decisions.

<sup>&</sup>lt;sup>54</sup> For target costing (see section 5.5).

<sup>&</sup>lt;sup>55</sup> See section 5.2.2.

(2) For all business, profit is a function of the time taken for manufacturing to respond to the needs of the market. This in turn means that profitability is inversely proportional to the level of inventory in the system, since the response time is itself a function of all inventory.

(3) It is the rate at which a product contributes money that determines relative product profitability. And it is the rate at which a product contributes money compared to the rate at which the factory spends it that determines absolute contribution as sales revenue less material costs. If the total of this contribution from all products per unit time exceeds total expenditure per unit time then an overall profit will be generated (Dugdale, et al., 1996).

Galloway and Waldron's first concept indicates that throughput accounting presumes that all factory costs (except materials) are fixed in the short/medium term - the opposite of activity based costing, which presumes all costs to be variable! Throughput accounting is therefore closely allied with traditional contribution analysis which emphasises the importance of maximising contribution in order to maximise short-term profits.

The second concept is rather contentious. It is hardly self-evident that profitability is inversely related to manufacturing response time and Galloway and Waldron's corollary, in order to maximise, is hardly easier to accept.

The third concept is very similar to the traditional idea of "contribution per unit of limiting factor". Galloway and Waldron argue that profitability will be maximised by maximising the rate at which cash is generated (cash being the difference between selling price and material cost for each product), and they propose that products be ranked according to the throughput accounting (TA) ratio which is defined as:

$$TA = \frac{Return \ per \ factory \ hour}{Cost \ per \ factory \ hour}$$

Where: -

 $Return \ per \ factory \ hour = \frac{Sales \ price - Material \ cost}{Time \ on \ key \ resources}$ 

 $Cost per factory hour = \frac{Total factory cost}{Total time available on key resource}$ 

Dugdale, et al., (1996) argue that aspects of throughput accounting can be integrated into conventional theory. The particular usefulness of the TA ratio appears to be that it indicates whether, if one product only were manufactured, the result would be a profit or a loss (a TA ratio of more than 1.0 indicates profit, less than 1.0 indicates a loss).

In conclusion, cost derived by activity costing and by throughput accounting serve very different purposes. Activity costing aims to identify all overhead costs with a product as accurately as possible. A medium to long term time-scale is envisaged so that product strategy decisions can be reflected in changed levels of overhead costs. Throughput accounting costs focus on short term with overhead attributed to product accounting and usage of existing manufacturing bottlenecks. In short, the view of Dugdale, et al., (1996) is that the idea of building up product cost on a "throughput accounting" principle is probably impracticable.

The second way for using cost management to increase efficiency is benchmarking and target costing. The following section will discuss benchmarking and the fifth chapter will discuss target costing.

American Productivity and Quality Centre defines benchmarking as "a systematic and continuous measurement process, a process of continuously measuring and comparing an organisation's business processes against business process leaders anywhere in the world to gain information which helps the organisation to improve its performance" (Beretta, et al., 1998: 11).

Benchmarking was seen as a management technique to be used only by specialists and to be applied only when large breakthrough improvements were needed, often in a crisis situation. Benchmarking has since become an important tool in the management process. It represents ongoing, structured comparisons with the goal of promoting continuous improvement from organisational solutions adopted by each company in managing its business processes. Clarke and Marton (1997) argue that benchmarking is a tool for change management.

There is no doubt that organisational aspects of the benchmarking process are crucial to the success of the project. Benchmarking attempts to help companies by providing reliable answers to two basic questions: how much can the performance of process be improved and which practices lead to best performance. Beretta, et al., (1998) argue that there are at least four methodological issues that are extremely critical to the success of benchmarking projects. These four methodological issues are: (a) how to define the performance measures (b) how to achieve comparability of performances (c) how to identify best practices (d) how to evaluate the transferability of best practices. The core idea of each element is discussed in turn

The first methodological issue is how to define performance measurement. Benchmarking is aimed at answering two basic questions connected to breakthrough performance improvement: (1) how much can a company improve its performance? (2) how to dramatically improve performance? Connecting performance measurement of accounting activities to value generation implies two logical shifts in performance measurement methodology: (1) changing the point of view: from organisational units (the accounting department) to organisational processes (the accounting processes). (2) expanding the traditional cost based perspective: from a uni-dimensional perspective (typically cost-based), to a multi-dimensional perspective (considering the process cycle, time and quality of services as well as cost).

The problem in defining performance measures is how to compose a well-balanced multi-dimensional vector (integrating cost, time and quality) for target setting and evaluation of an accounting process. The performance should include three performance measures: cost minimisation, cycle time minimisation and improvement of quality.

The second issue is to achieve comparability of performance. This issue relates to comparing company performances for benchmarking purposes. The boundaries or parameters of the process context need to be established for comparability of performance targets derived from external benchmarks. The performance of an accounting process is dependent on the way resources (mainly human resources) are organised, the way technology (mainly information technology) is applied to the process, and the peculiarities of the process environment (or context).

In short, comparability of performance data has to be assured through the identification of the structural performance drivers that mostly influence the context in which the process is managed. For accounting processes, structural performance drivers are process specific more than industry specific.

The third methodological issue is to identify best practice. Benchmarking should not be limited to gap measurement. It should help management identify the drivers or practices underlying the best performance (Beretta et al., 1998). The starting point is the task of defining what is best practice. Although the idea of best practice implies universality, universality has two invincible enemies in benchmarking. The first is the fact that there is no practical possibility to include in a benchmarking study all the best practitioners. So, best practices are those that, having applied certain criteria of performance evaluation, have proved to work better than others among all that we benchmarked. The second enemy is the fact that in operations it is very difficult to prove that there is one best way of doing something.

In summary, for identifying best practices in accounting processes, an effective benchmarking methodology should clearly show the different ways accounting activities can be carried out and how they are connected with each other (practices). Also, it should highlight the functionality induced on accounting processes by nonaccounting processes and practices (enablers).

The final issue is how to evaluate the transferability of best practice. Benchmarking is a methodology intended to facilitate learning from outside. However, sound practices developed by one company often cannot simply be adopted by another company. Identification of best practices should be followed by the test of transferability in different contexts. What is needed is a methodology for the specific to a well defined context.

## 4.2 Institutional theory and efficiency

Granovetter (1985) claims that transaction cost theory provides an under-socialised account whereas institutional theory offers an over-socialised perspective. More specifically, "economic approaches to the study of organisation, transaction cost analysis included, generally focus on efficiency" (Williamson, 1981a: 549). Whereas "institutional theorists place particular emphasis on legitimation processes and the tendency for institutionalised organisational structures and procedures to be taken for granted" (Oliver, 1992: 563), regardless of their efficiency.

Peter and Royston (1997) argue that transaction cost and institutional perspectives are not necessarily in conflict, but are complementary elements of a constrained-efficiency framework. The constrained-efficiency framework illuminates the adoption of organisational designs by viewing organisations as efficiency seeking (Nelson and Winter, 1982) under cognitive and institutional constraints, as opposed to efficiency optimising.

Transaction cost theory has evolved from the initial question, why do organisations exist?, to address questions of more immediate concern to organisation theory. In particular, what designs will organisations adopt under various circumstances (Williamson, 1992)? Thus, Williamson (1981a: 568) suggested that the reason "there are so many kinds of organisation [is] because transactions differ so greatly and efficiency is realised only if governance structures are tailored to the specific needs of each type of transaction". Transaction cost explanations have been employed to examine hierarchies, franchises, multidivisional, conglomerates, holding companies (Williamson, 1975), clans (Ouchi, 1980), networks (Jarillo, 1988), and market-hierarchy hybrids (Williamson, 1991b). In each of these explanations, the rationale for existence of any given organisational design is its efficiency compared to the set of available alternatives, including markets (Winter, 1991). In this sense, the transaction cost explanation rests on a comparative-efficiency account of design selection (Klein and Shelanski, 1994).

Peter and Royston (1997) argue that the conceptual foundation for the constrainedefficiency framework is cognitive<sup>56</sup> constraints, competitive<sup>57</sup> pressures and institutional influences. For the purpose of this research the concentration will be on institutional pressures. Williamson (1992: 41) is explicit about the role of institutional environment suggesting that "the institutions of governance (firms, markets, hybrids, bureaus) are embedded in the institutional environment". However, the institutional environment envisioned by transaction cost economists consists only of the "set of fundamental political, social and legal ground rules that establishes the basis for production, exchange and distribution" (Williamson, 1993: 111). The institutional environment envisioned by institutional theorists includes its more cognitive and sociological elements, such as patterns of social relations (Burns and Wholey, 1993) and embeddedness interpretations (Meyer and Rowan, 1977).

Scott (1987: 493) pointed out, "the beginning of wisdom in approaching institutional theory is to recognise at the outset that there is not one but several variants". The critical consideration is the contrast between transaction cost theory, which addresses an organisation's concerns with efficiency, and institutional theory, which "demonstrates how non-choice behaviours can occur through the exercise of habit, convenience, or social obligation" (Oliver, 1991: 151). In its strongest form, the institutional approach rejects the premise that organisational phenomena are the products of rational choice based on technical considerations (Westney, 1993). Rather, emphasis is placed either on the decisions made, or on the pressures to secure legitimacy that operate on organisations, or both.

In general, institutional theorists have been more interested in explaining uniformity than diversity (DiMaggio and Powell, 1983). Isomorphism, a central concept within institutional theory (Westney, 1993), captures the extent to which the organisational designs adopted within organisational fields tend toward increasing homogeneity over

<sup>&</sup>lt;sup>56</sup> March and Simon (1958) recognised that decision-makers operate under cognitive constraints that make strict optimising theories untenable. In their framework, and that of Cyert and March (1963), decision-makers tend to conduct more-or-less limited searches among available alternatives to obtain satisfying, rather than optimising solutions.

<sup>&</sup>lt;sup>57</sup> Although individuals are cognitively constrained in their pursuit of efficient organisation designs, one mught still argue that efficency-based competitive pressures should ensure that observed designs will be those in which organisational efficiency is optimised. However, Robins (1987) stressed that in the absence of perfect competition, optimal exchange efficiency need not always result.

time. This emphasis on uniformity naturally leads institutional theorists to focus on the organisational field, or even society at large, as the primary unit of analysis (DiMaggio and Powell, 1991).

Echoing Granovetter's (1985) concerns about over-socialised theory, Powell (1991: 194) suggested that "much of the imagery of institutional theory portrays organisations too passively and depicts environments as overly constraining." Finally, DiMaggio and Powell (1983) and Tolbert and Zucker (1984) recognised that institutional forces are not always primary, noting the tendency for early adoption to be driven by technical, as opposed to legitimacy considerations. These shortcomings are addressed by integrating elements of institutional theory into the comparative-efficiency approach offered by transaction cost theorists. The various institutionalisms (Scott, 1987, 1995) may be grouped into two broad categories of institutional effects: pre-conscious and post-conscious.

According to preconscious institutionalisation, organisations operate and make choices in environments where much is taken for granted. These elements, which are either infused with excess value (Selznick, 1957) or constructed in the process of social interaction (Berger and Luckmann, 1966), serve as powerful 'frames', shaping the decision-making process by influencing what is or is not perceived by decision-makers. Preconscious institutionalisation, in other words, corresponds to DiMaggio's (1988: 4-5) factors "that make actors unlikely to recognise or act on their interests". According to post-conscious institutionalisation, tangible forces in an organisation's environment directly or indirectly divert design adoption away from the proposed dynamic in transaction cost economics (i.e., comparative efficiency) and toward the dynamic of legitimacy.

Although the terminology may be new, these two constructs find support elsewhere within the institutional theory literature. Jepperson (1991: 147), for example, stresses that "one may take for granted some pattern because one does not perceive it, or think about it [preconscious]; alternatively, one may subject the pattern to substantial scrutiny, but still take it for granted . . . as an external objective constraint [post-conscious]".
Pre-conscious and post-conscious institutionalisation may be discussed in relation to Scott's (1995) normative and cognitive pillars of institutional theory. According to Scott (1995: 38-39), the normative pillar refers to sets of expectations within particular organisational contexts of what constitutes appropriate, and thus legitimate behaviour. Rational action is always grounded in social context that specifies appropriate means to particular ends; action acquires its very reasonableness in terms of these social rules and guidelines for behaviour. Here, choices are structured by socially mediated values and normative frameworks.

In other words, Scott's (1995) normative pillar is grounded in the "logic of appropriateness" (March, 1981); that is, what is expected of organisations. Scott's (1995) normative pillar may be viewed as either pre- or post-conscious institutionalism. Much of the writing on normative constraints emphasises how the normative expectations assume a taken-for-granted form; the ways of organising become unquestioned, and alternatives become unthinkable (Zucker, 1983). In this sense, normative expectations correspond to the current use of the term preconscious; organisational actors may simply be unaware of possible alternatives. However, as soon as actors deliberately consider alternatives, the normative expectation is no longer preconscious. Post-conscious institutionalism is evident when organisational actors are aware of the need to change, but actively consider only a limited range of alternatives, each of which is acceptable within the prevailing institutional context.

Scott's (1995) cognitive pillar of institutional theory was focused on the frameworks of meaning by which actors interpret and make sense of their world. Unlike bounded rationality within transaction cost theory, the cognitive influences emphasised by institutional theorists (e.g., Fligstein, 1991: 315; Greenwood and Hinings, 1996) relate to "the internalised symbolic representations of the world" (Scott, 1995: 40). The role of these symbolic representations has been examined through various terms (see Walsh, 1995) and there have been calls for a more careful synthesis of the cognitive and normative perspectives (see Scott, 1994). It is important to note that the cognitive pillar (as with the normative pillar) may be unwitting and therefore preconscious or it may be post-conscious.

## **4.3 Activity Based Costing (ABC)**

This section aims to review literature relating to the cost of product generally and activity-based cost especially.

Cost accounting research has been concentrated on two broad areas (Vollmers, 1994): (1) The study of modern cost methods and techniques<sup>58</sup>, (e.g.; ABC<sup>59</sup>)

(2) The study of the origin and development of cost accounting (Johnson and Kaplan, 1987; Ahmed, 1992), for example how budgeting and standard costing resulted in increased control over labour.

Horngren (1995) says in the U.S. during the past 30 years there were three noteworthy developments in management accounting:

(a) In the early 1960s, emphasis on variable costing and contribution reporting; these ideas influenced designing and changing of the basic management accounting system(b) In the late 1970s, emphasis on Zero-based budgeting (ZBB), these ideas gave welcome attention to how to improve the budgeting process; but ZBB has practically

vanished from actual practice<sup>60</sup>

(c) In the late 1980s, activity based costing (ABC) dominated U.S. literature on management accounting. Furthermore, Johnson (1990) describes ABC, variance analysis, and return on investment as three of the most important management accounting theories in the twentieth century.

Activity-based costing (ABC) has been established for the best part of thirty years (e.g. Staubus, 1971) and its current popularity and form can be traced to the development and implementation in the U.S. (Cooper and Kaplan, 1988; Cooper, 1988). Use of ABC has spread quickly, to varying degrees, in the US, Canada, and the UK. One might expect the use of ABC to spread to other countries, particularly countries that share a common language and have similar capitalistic business cultures.

<sup>&</sup>lt;sup>58</sup> This involves theoretical development, e.g.; new cost drivers, or implementation of these techniques, e.g.; implementation of ABC.

<sup>&</sup>lt;sup>59</sup> Johnson, 1988; Cooper and Kaplan, 1988; Jeans and Morrow, 1989; Noreen, 1990; Johnson, 1990; Innes and Mitchell, 1990; Bjornenak and Mitchell, 2000; Daniels, 1999; Malmi, 1999; Innes, et al., 2000

<sup>&</sup>lt;sup>60</sup> ZBB has vanished because it has no history.

Some factors have contributed to the dissatisfaction with conventional costing systems. The problem was caused by three compounding factors (Innes and Mitchell, 1995):

(a) The structure of product cost<sup>61</sup> in many contemporary businesses had changed substantially, with production and non-production overhead costs growing in relative size and importance while direct labour had shrunk dramatically.

(b) Direct labour and machine hours have persisted as major bases of production overhead absorption products. Both of these bases relate fairly closely to production volume and their use therefore rests on the assumption that overhead incurrence is output driven.

(c) There is a shift in emphasis from production overhead costs driven mainly by output to production overhead costs driven by the diversity and complexity of production. Production overheads of this type have been concisely summarised by Miller and Volluman (1985) as falling within four transaction type categories: (a) Logistical transactions (b) Balancing transactions (c) Quality transactions (d) Change transactions.

Kaplan, (1994) argues that traditional cost allocation systems allocated overhead costs on simplistic measures such as direct labour, or sales dollars or direct costing systems that ignored overhead costs entirely for calculating the costs of products. Johnson and Kaplan (1987); Cooper and Kaplan (1988); Miller and Volluman (1985); all suggested Activity based costing – a two stage cost allocation model - to trace overhead costs to specific activities (cost pools) and then from these activities to individual products or groups of products using cost drivers. If we assume the previous argument regarding the traditional cost is right, it is less clear what new approaches could overcome these limitations.

ABC goes for greater complexity by expanding the notion of cost variability implicit in the second half of the procedure. ABC uses multiple "cost drivers" linked to functional activities in all parts of the organisation's value chain, assuming that only activities cause costs. These factors, together with the lagged relationship between resource consumption and cash spending, mean that ABC is not a perfect cost control or prediction model (Piper and Walley, 1990).

<sup>&</sup>lt;sup>61</sup> Product cost: The cost of a finished product built up from its cost element (CIMA, 2000).

ABC tries to solve the "allocation problem" identified by Thomas (1974), but Brignall (1997) argues that it is focusing on the wrong area: accurate product cost rather than cost engineering to gain a target market share at a predetermined price with acceptable profit, with planned cost reductions through the life cycle as in target costing (Hiromoto, 1988; Kato, 1993).

ABC may have some merits as a product costing system for those mass services (for example, Water services) which compete via a "cost leadership" strategy. As they have significant amounts of short-term fixed, indirect costs, it appears to be a cumbersome and ineffective method of cost management.

ABC systems could be valuable not only for analysing manufacturing overhead costs and influencing product design decisions, but also for assigning marketing, selling and distribution costs to customers, business segments and distribution channels. The costing field shifted from arguments on how to allocate costs (the traditional domain of cost accounting) to identifying the cost flows from organisational spending to supply resources that create the capability to perform organisational activities.

ABC systems came from two theoretical developments (Kaplan, 1994) that elevated the approach from deductive assertions to scientifically testable hypotheses:

(1) Discovery of the cost hierarchy of indirect and support expenses

(2) The distinction between costs of resources supplied and the costs of resources used i.e. the role for unused capacity costs.

In comparison with the conventional costing model, (full absorption costing 'FAC'), two major points are made by Kaplan and his colleagues:

(1) They move away from the notions of short-term fixed and variable costs which are an important feature of management accounting's conventional wisdom, and focus on the variability of cost in the longer term

(2) They propose a hierarchical resource consumption costing model which assumes that the costs which are shared at one level will be variable at a higher level (Scapens, 1991; Brignall, et. al., 1991).

ABC will generate product (or service) line costing which will differ from those produced by a traditional system primarily in respect of their indirect cost content. This is because costs will be attached to products in a way which reflects how these products have created a demand for, and effected the consumption of, the resources which comprise the organisation's overheads. Where this element of cost is material and where traditional methods of overhead costing (e.g. labour hour/cost rates) do not adequately capture resource consumption, then the use of ABC will improve the product cost information which is produced (Innes and Mitchell, 1995).

Turney, 1991a; Kaplan, 1995; and Cooper, et al., 1992 argue that ABC is important because it allows management accountants to become part of the organisation's valueadded team by working in conjunction with other organisational and administrative functions. The Accounting function in firms which choose to adopt ABC moves from a narrow recording role to a broader management decision-support role. For example, it is argued that ABC provides more precise costing which leads to more informed pricing and more informal cost control. Furthermore, the information provided by an ABC system can also lead to changes in the operational design of the firm. Firms which choose not to consider ABC may risk having cost systems that possess computational accuracy but lack conceptual integrity (Emore and Ness, 1991)<sup>62</sup>. This is because the traditional accounting system merely collects and aggregates data but does not reflect or encourage the changes necessary in today's manufacturing climate (e.g. computeraided-manufacturing, just-in-time production processes, or inventory management techniques such as material requirement planning). In addition, decision-makers tend to exclude accountants from the management team when they could benefit from their insight and expertise.

Bjornenak and Olson (1999: 326) argue that most studies examining the new models (techniques or ideas) focus on a limited set of characteristics, for example the cost function assumptions in the ABC model (Noreen, 1991; Christensen and Demski,

<sup>&</sup>lt;sup>62</sup> This does not suggest that ABC is appropriate for all firms. Contingency theory research in accounting argues that factors such as the environment, technology, and organisational size and structure influence the choice of accounting and information systems (Hayes, 1977; Bruns and Waterhouse, 1975; Waterhouse and Tiessen, 1978; Otley, 1980) and therefore, there is no "universally appropriate accounting system which applies equally to all organisations in all circumstances". In practice, Cobb, et al., (1993) also note that implementation issues (particularly lack of resources) continue to hinder the use of ABC in UK firms.

1995). Furthermore, they argue that the ABC literature identifies the need for different objects<sup>63</sup>.

Activity-based costing can be considered an administrative innovation. Evan (1996) distinguishes between technical and administrative innovations. For example, technical innovations include new products or processes while administrative innovations include such changes as new organisational structures and internal reporting systems. Hopwood (1974) and Merchant (1981) classify accounting systems as administrative innovation.

Damanpour and Evan (1984) suggest that administrative innovations more often lead technical innovations than the reverse. This implies that firms with innovative accounting (administrative) systems will be more likely to adopt technical innovations. Further, Dunk (1989) suggests that non-innovative accounting control systems can act as barriers to technical innovation.

Dunk (1989) also discusses the difference between technical and administrative innovations and focuses on the lag between administrative (accounting) innovations and technical innovations. Dunk suggests that one reason that organisations are loath to adopt accounting innovations (e.g. ABC) is because the perceived economic benefits of a change in accounting system are difficult to quantify. Kaplan (1984) and Johnson and Kaplan (1987) also contend that the cost of management accounting systems (which do not support financial reporting requirements) are difficult to justify economically.

Several surveys have examined the adoption of ABC in companies in the US and report a range of results. For example, Green and Amenkhienan (1992) state that 45% of responding manufacturing firms using advanced technologies have implemented ABC to some degree in their firms. Shim and Sudit (1995) state that 27% of the manufacturing firms surveyed had fully or partially implemented ABC. A 1993 study conducted by the Cost Management Group of the Institute of Management Accountants (1993) found that 36% of responding US firms had implemented ABC and a 1995

<sup>&</sup>lt;sup>63</sup> Bjornenak and Olson (1999: 330) identify three major developments regarding descriptive objects in the new techniques: (a) the traditional focus of responsibility centres and products extended to a larger number of factors (b) financial data is combined with non-financial data to extend the description of variability (c) external descriptive objects (e.g. competitors) and external descriptions of variability (e.g. customer satisfaction) are introduced in the techniques

survey by the same group (Cost Management Group of the Institute of Management Accountants, 1996) showed that the percentage of ABC adopters had increased to 41%.

In the UK, Innes and Mitchell's (1991) research on the implementation and use of ABC systems reveals that only 6% of surveyed U.K. firms had begun to implement ABC by 1990. Drury and Tayles (1994, 2000) who report that 13% of UK manufacturing firms had adopted ABC by 1991 found somewhat higher rates of adoption. Recent research indicates that the upward trend in adoption of ABC continues as Innes and Mitchell (1995) report that 20% of the top 1,000 firms in the UK have adopted ABC. Innes, et. al., (2000) review the changes that have occurred in the ABC adoption status of companies over a recent 5-year period and their result reveals that the use and interest in ABC has not increased in this time.

Although the result of Innes, et al., (2000) highlighted that the use of ABC has not increased over the past 5 years. Some changes have been observed in the purposes<sup>64</sup> of ABC. For example, there is an increase in the use of ABC for cost reduction, product/service pricing, cost modelling, customer profitability analysis, output decisions, new product/service design and decreased use of ABC for performance measurement/improvement, budgeting and stock valuation. Comparatively less research on adoption of ABC in Canada exists. However, a survey sponsored by the Society of Management Accountants of Canada found that 23% of responding manufacturing firms have adopted ABC (Armitage and Nicholson, 1993).

Shields and Young (1989) developed a theoretical model about the implementation of cost management systems (CMS) that is applicable to ABC. They identified seven behavioural and organisational factors as being critical to the implementation of CMS:

(a) Top management support

- (b) Linkage of CMS to competitive strategies
- (c) Linkage to performance and evaluation systems
- (d) Adequate internal resources
- (e) Training
- (f) Non accounting ownership

<sup>&</sup>lt;sup>64</sup> The researcher acquired this knowledge from the comparison between the survey in 1994 and 1999 which was highlighted by Innes, et al., (2000).

#### (g) General consensus about objectives of CMS

This approach is in line with the tool approach where researchers survey managers to determine factors that appear to be important, and correlate these factors with success and failure (Shields, 1995; Gosselin, 1997). The table 4.1 summarises prior studies relating to ABC implementation.

Study	Method	Implementation Model	
Cooper et al. (1990)	Field study of eight	Analysis and action	
	companies		
Anderson (1995)	Field study of one company	Initiation, Adoption,	
· ·		Adaptation, and Acceptance	
Coderre (1995)	Theoretical model	No stages	
Anderson and Young	Surveys and multiple case	No stages	
(1997)	studies of 21 ABC projects		
Mackley and Thomas	Field study of one company	Goal attainment, Adaptation,	
(1993)		Integration and Long term	
		development	
Norkiewicz (1994)	Field study of one company	No stages	
Shields (1995)	Survey of 143 firms that had	No stages	
	implementation of ABC		
McGowan and	Survey of 53 employees	No stages	
Klammer (1997)	from 4 target sites		
Gosselin (1997)	Mail survey of 161	Adoption and Implementation.	
	Canadian manufacturing		
	SBUs		

 Table 4.1: Prior studies to ABC implementation

Lemond (1992) predicted that ABC would become a major force in Australia. Yet, Joye and Blaney, 1990; Dean, et al., 1991, Corrigan, 1996 and Booth and Giacobbe (1995) report little use of ABC in Australia. In fact Dean, et al., (1991) reported that only 8% of Australian firms had adopted ABC by 1990 and Booth and Giacobbe found a similarly low adoption rate of 12% in 1995.

Surveys of ABC systems in non-English speaking countries, for example, Sweden (Ask and Ax, 1992), Finland (Lukka and Granlund, 1994). France (Bescos and Mendoza, 1995), Spain (Sales, et al., 1995) and Denmark (Sorensen and Israelsen, 1996) suggested adoption rates below those recently found in the Anglo-American countries. Malmi (1997) reported low but increasing emphasis on ABC in Finland.

Kaplan, 1988; and Hartnett, et al., 1994 argue that firms are less likely to change management practices (e.g. ABC) when costs, especially those relating to inventory valuation, may have to be restated under financial accounting reporting requirements. In addition, successful adoption of management accounting practices will not occur without support from top level management (Shields, 1995).

In theory, it is possible to trace all overheads, but, in practice, it is quite difficult to trace some overhead costs such as head office administrative costs or joint cost to individual products, and hence there is still some room for arbitrary cost allocation. For example, Scapens (1991) described an allocation problem where there might be a temptation to terminate production of individual products when their joint production is still worthwhile. In addition, the ABC model has been criticised for being based on the simple economic analysis which underlies management accounting's conventional wisdom, and on the assumption that there is a sense of harmony and co-operation among operating units, and between operating units and service functions; it does not acknowledge the possibility of conflict over the allocation process (Ahmed and Scapens, 1991; Gietzmann, 1991; Bhimani and Pigott, 1992).

Most of the few studies which have addressed ABC success and/or failure have been the so-called factors studies<sup>65</sup> (Cobb, et al., 1992; Shields, 1995; Anderson, 1995). The focus has been on identifying factors which ABC success or failure. Shields (1995), for example, found top management support, linkage to competitive strategies, linkage to performance evaluation and compensation, training in implementing ABC, non-accounting ownership and adequate resources all positively correlated with ABC

<sup>&</sup>lt;sup>65</sup> The term "factors study" is borrowed from information technology (IT) implementation literature (Kwon and Zmud, 1987; Lucas, 1990; Cooper and Zmud, 1990), where it refers to studies which "try to identify those "factors" most related to IS implementation success and failure (Kwon and Zumd, 1987: 228). Most of these studies are typically cross-sectional seeking statistical correlation between factors hypothesised to influence implementation outcome and some measure of that outcome.

success. Cobb, et al., (1992), in turn, suggested that the major problems experienced with ABC related to the lack of adequate internal resources, particularly staff time and computer resources. Anderson (1995) identified 21 factors, related to the individuals involved, the organisation structure, the task, the technology employed and the external environment, which influenced ABC implementation at General Motors. One problem with the factor approach is that there is hardly any limit to the number of possible factors affecting the relative implementation outcome.

Argyris and Kaplan (1994) seek an alternative way of explaining ABC failure by presenting a behaviour model of why and how employees resist ABC. They build on Argyris (1985, 1990: 38), noting that "...barriers to change arise from the defensive routines that participants trigger to protect themselves from experiencing embarrassment and threat from the new idea' (see also Cooper, et al., 1992). Apart from the ABC context Markus (1983) argued that resistance to new information systems can be understood in terms of organisational power and politics. In a similar vein, Scapens and Roberts (1993) illustrated how division's attempts to increase unit accountability led unit management to resist a new accounting system. Malmi (1997) proposed that claims on ABC failure result, in part, from assessing the use and value of ABC's from the decision-making perspective.

Innes and Mitchell (1995) said ABC will not solve all of the complexities of product costing. Some arbitrariness will certainly remain. For example, some overheads are incurred at a level divorced from that of any individual product. Cost drivers may suffer from the problem of "jointness". If the number of orders is being used as a driver for procurement overhead and the direct material for a number of products is made on each individual order then the association of the cost driver with the individual products becomes problematic. Indeed the identification of sets of cost drivers, which explain 100 percent of the relationship between products and activity costs, is unlikely. ABC product costing has some technical difficulties.

Horngren (1995) said the ABC movement, in the late 1980s has been stimulating and beneficial. Nevertheless, the following aspects are troublesome:

(1) Excessive of the past developments and excessive claims regarding what is new

(2) The lack of respect for history. For example, although ABC had another name, its basic ideas were exposited in Longman and Schiff (1955: 70) as practical distribution cost analysis. Many scholars and consultants have neither time nor the inclination to explore the history of thought

(3) Outrageous claims. These exaggerations included such statements as all costs are variable or all costs are fixed

(4) Proneness toward regarding product costs as the major influence on decisions regarding pricing, product mix, and make or buy decisions

(5) The proclivity to not practice what is preached

(6) The lack of ABC in Japan. American literature abounds with descriptions of changes in management theory and practice inspired by developments in Japan

(7) The frequent failure of ABC systems to distinguish between resources provided and resources consumed

ABC has maintained a high profile status as an important management accounting innovation for well over a decade (Bjornenak and Mitchell, 1999). However, despite a strong and durable advocacy (Cooper, 1988; Cooper and Kaplan, 1991; 1992; 1998; Kaplan, 1992), several reservations have been expressed concerning

(a) The substance of its practical attraction (Bjornenak, 1997; Gosselin, 1997; Malmi, 1999), i.e. that it may be a fad or fashion, engendering a bandwagon effect rather than a genuine and useful technical enhancement

(b) Its decision-making relevance (Noreen, 1991; Bromwich and Hong, 1999), i.e. that several restrictive (and practically unlikely) conditions must apply before the ABC information can legitimately be used to generate relevant costs for decisions

(c) The problematic and costly design, implementation and operation of the systems required for ABC in an organisation context (Cobb, et al., 1992; Malmi, 1997).

Horngren, 1995; and Innes and Mitchell, 1995 summarised the advantages from ABC as follows: ABC has generated enormous enthusiasm about its basic ideas. It links causes (cost drivers) with effects (changes in costs). ABC entreats us to choose cost allocation bases with great care. In particular, we should be aware of the fact that costs are driven by many factors other than the volume of units produced or sold. ABC has correctly advocated the use of multiple cost allocation bases that are appropriate cost drivers. It criticised the overuse of direct labour as a cost allocation base, particularly when it is

used as the lone base for applying indirect costs to products. ABC has stressed that executives manage costs by overseeing activities instead of products. The accounting for costs by activities highlights the interdependencies among activities in many departments or functional areas. ABC has emphasised that product costs are affected by all functions in the value-chain, not just by manufacturing alone. It has alerted managers to the existence of cross-subsidisation among the product cost where there is a wide range of operating and a wide range of products.

ABC advocates have started to admit that the complexities of a detailed ABC system may inhibit its use as an ongoing cost accounting system. ABC information may give a good indication of how a product has consumed resources in the recent past; this may bear little relationship to future incremental cost (from expansion) or future avoidable cost (from dropping the product). The use of ABC will generate information of relevance to cost control. Overheads are made more visible through pooling by activity and their underlying cause, to the extent it is reflected in their cost driver, is also indicated. ABC has given a greater understanding of cost behaviour, improved communication of cost information due to the logic of ABC. It is also directly related to how costs are incurred, particularly indirect labour costs, and the gathering of information to run the system can be a matter of some sensitivity. ABC will result in a new set of information, which can be used to measure aspects of operational performance within the organisation, i.e. cost drivers and cost driver rates. These may provide useful and quick regular feedback to line management and may help to foster cost consciousness and efficiency (Johnson, 1988).

ABC possesses several compelling positive attributes. It has the potential to contribute to strategic decision-making, cost control and management and performance measurement. But both system designers and users, in order to realise the benefits of ABC and minimise the costs, must exercise some care. Implementation success will depend particularly on attention being given to the following three factors:

(a) A recognition of the nature and variation in ABC

(b) A recognition that ABC is not a complete general purpose system which will meet all cost information needs

(c) A recognition that any ABC system will have some limitations and if not carefully designed and monitored will have the potential for dysfunctional consequences for the adopting organisations

Activity based costing has been seen to offer important advantages over traditional costing techniques including: enhanced product cost accuracy (Brimson 1991; Cooper and Kaplan, 1987; Roth and Borthick, 1989), more comprehensive cost data for performance evaluation (Berliner and Brimson, 1988), more relevant information for managerial decision making (Cooper and Kaplan, 1988), greater potential for sensitivity analysis (Shank and Govindaragin, 1989), and generally, in providing a novel outlook on value-adding organisational transactions and activities (Johnson, 1988).

Bhimani and Pigott (1992) argue that although research concerns over such consequences in relation to organisations implementing ABC have remained largely unaddressed. It has been acknowledged that behavioural, organisational and social factors are likely to be implicated in replacing conventional costing systems with more complex and innovative accounting techniques (Bromwich and Bhimani, 1989).

ABC is the basis of the cost assignment view in which resources are assigned to activities and then to cost objects (products). While this process can provide more accurate costing of products or services, it does not in itself manage costs (Rouse and Putterill, 2000: 368).

Regarding to the relationship between ABC and institutional theory, Soin et al., (2002) used old institutional economic theory to interpret the role of management accounting in organisational change. They present an interpretative study of the introduction and implementation of an Activity Based Costing (ABC) system in the clearing department of a UK-based multinational bank. Soin et al., (2002) was concerned with the interplay between management accounting and other agents of organisational change that drew on rationalities and legitimating for spheres such as scientific management and human resource management. Their paper drew on institutional theory in order to understand the extent of change, as a way of evaluating the relative roles of other institutionalised practices apart from management accounting and to assess the relative impact of the new management systems.

Although it is easy to find technical justifications for ABC in literature, sociological institutional theory does not automatically assume that particular techniques are chosen purely for their instrumental efficiency (Sion, et al., 2002). Actors often have a hazy view on the nature of the problem and have a socially constructed view of reality (Berger and Luckmann, 1966).

## **4.4 Activity Based Management**

This section reviews literature relating to the cost of product in general and activity based management in particular.

Williams and Ashford (1994) highlighted four of the changes in control systems and product costing systems that may be necessitated by new manufacturing technologies:

(1) Competitive pressures necessitating shorter product life cycles and faster introduction of new products and services. Management Accounting Control System (MACS) must respond with flexible management structures, project teams and new performance measures (NFIS)

(2) Emphasis on activity analysis and the supply chain. MACS must respond with much closer attention to long-term supplier alliances and investigate activity-based management systems (ABM) and customers' profitability analysis

(3) Adoption of Total Quality Management (TQM). MACS must respond with an increased emphasis on NFIS, to observe internal and external failure costs, to monitor prevention costs and to measure the costs of quality

(4) Adoption of Just in Time management (JIT). MACS must respond by making wholesale modifications to traditional systems based on labour productivity machine efficiencies, rejection and wastage rates, and inventory holdings. These will no longer be appropriate and new NFIs must place the emphasis on service to customers and speed of delivery

The evidence suggests that it will be impossible to eliminate arbitrary allocations of overheads totally, even under an ABC system, so we must look beyond product costing to a more appropriate emphasis on process management. ABC may be a useful starting point.

The key to the extension of ABC into Activity Based Management (ABM) is a wider appreciation of the concept of "drivers". We can no longer focus on cost drivers alone but must investigate the manner in which resources are consumed in non-dollar areas (Smith, 1997). Customers have perceived need in at least four areas, all of which require simultaneous satisfaction: lower costs, higher quality, faster response times, and greater innovation.

Activity Based Management (ABM) reshapes how companies manage costs. By understanding its activities, a company can expose opportunities for performance improvement that conventional cost accounting systems seldom detect. Cost management is improved by identifying what the organisation does and providing a benchmark to judge how much better a company's performance might be (Brimson, 1996). Activity based management is a system of management which uses activitybased cost information for a variety of purposes including cost reduction, cost modelling and customer profitability analysis (CIMA, 2000). Activity-based management (ABM) is a system that incorporates many of the concepts of strategic management re-engineering and applies them to cost management (Trussel and Bitner, 1998). Activity based management is a modern cost accounting and management model that is consistent with the concepts of strategic management and re-engineering. ABM is both an accurate cost accounting system and a performance improvement tool (Turney, 1991b).

Business process re-engineering (BPR) is a management tool for redesigning business processes in order to obtain dramatic improvements in performance measures, such as cost and quality. The idea behind BPR is to fundamentally revise all aspects of performing activities, from a revision of strategic goals and operating objectives to an alternation of work methods. Under BPR, work should be focused on processes and not functional tasks (Harmmer and Champy, 1993). Activity-based management (ABM) is similar in nature to BPR but adds the analysis of the cost management.

ABM consists of two viewpoints: a cost view and a process view. Under the cost view, ABM is a cost accounting system (called activity-based costing). The value of the cost view is to determine the cost of a product in service. Under the process view, ABM is used to develop financial and non-financial performance indicators for the output of each activity centre. The goal of the process is to measure performance (Trussel and Binter, 1998). The two viewpoints of ABM are summarised in figure 4.2.



Figure 4.2: Activity-Based Management

Source: Trussel and Bitner, (1998)- "Strategic Cost Management: An ABM approach", p. 442

Rouse and Putterill (2000: 368) argue that cost management is pursued via ABM, which is portrayed in a process revealing the causes (cost drivers) of work or activities, and focusing attention through performance measurement on how activities are conducted. The process by which activities are triggered by cost drivers is shown separately to recognise that the mere occurrence of a cost driver may not in itself initiate an activity and that management authorisation may be required. Trussel and Binter (1998) determined ten steps for designing and implementing the process view of activity-based management. The steps are based upon procedures espoused by Harmmer and Champy, 1993 and Cooper, et al., 1992. The steps of ABM are summarised in figure 4.3.



Figure 4.3: Implementation steps of the process view of AMB

Source: Adapted from Harmmer and Champy, 1993; Copper, et al., 1992; and Trussel and Bitner, 1998.

Cooper, et al., (1992) discuss the implementation strategies of ABM systems for eight companies: five manufacturing entities, one distribution company, one financial services provider, and one energy company. Anotos (1992) discusses ABM for a not-for-profit organisation. Harmmer and Champy (1993) discuss the implementation of business process re-engineering for several companies.

The steps necessary for designing and implementing the cost view of ABM follow (Cooper et al., 1992) the cost view of ABM called activity-based costing (ABC):

(a) Determine the activity centre

(b) Assign costs to the activity centres

(c) Define the cost objects

(d) Link activity costs to the cost objects

The traditional view is that costs are best controlled by department managers who are responsible for minimising the variance between budgeted and actual costs by cost element. The emphasis is on efficient use of resources. The ABM view is that costs are best controlled by managing the workload, eliminating non-value added activities, managing the factors that drive costs, continuously improving value added activities, and streamlining management. The emphasis is on effective use of resources. (Brimson, 1996). In addition to that the initial literature on ABC essentially viewed it as a product costing system (Cooper and Kaplan, 1988). Whereas ABM was viewed as a broader management system incorporating activity-analysis, analysis of business processes and analysis of value and non-value activities.

The output from the ABM system can be utilised for many purposes. The costs of all activities and all programs are more accurately determined. The information from ABM enables better decisions to be made relating to resource allocation, programme retention, marketing strategies, programme returns and like (Trussel and Binter, 1998). Activity information allows managers to identify and eliminate waste. It also confirms progress in removing waste from operating activities (Brimson, 1996).

Managers need activity information to help them achieve enterprise excellence. Activity accounting identifies what the organisation does. In order to improve profitability and performance, it is critical to understand where the enterprise's precious time goes and, in detail, what the enterprise does and how it does it. Activity-based management is a powerful tool for managing the complex operations of a business through a detailed assessment of its activities. Activity accounting attributes cost and performance data to activities. Activity cost and performance data provide management with information needed to determine an accurate product cost, improve the business process, eliminate waste (non-value added activities) identify cost drivers, plan operations, and set business strategies. Activity-based management generates cost and production information in a manner that drives continuous improvement and total quality. Continuous improvement and total quality control are facilitated by treating each

activity as a process and identifying the source of cost rather than focusing on the symptoms. In focusing attention on the source of problems, management must assign responsibility to those departmental activities that drive costs and monitor their execution to see if the planned results are achieved.

# **4.3 Conclusion**

This chapter has highlighted the concept of efficiency<sup>66</sup> and the way<sup>67</sup> of using cost management to increase this. It has been argued that some techniques/tools of cost management relate to traditional management accounting such as throughput accounting and others relate to contemporary management accounting (i.e. ABC). The chapter has argued that institutional theory (sociology institutional theory) does not automatically assume that particular techniques of cost management are chosen purely for their instrumental efficiency.

The chapter has highlighted the focus of transaction cost theory (under-socialised) and institutional theory (over-socialised). It has been argued that transaction costs and institutional perspectives are not necessarily in conflict. It shows the transaction cost explanations rests on a comparative-efficiency account. The chapter has also highlighted the importance of institutional environment and shown that isomorphism is a central concept within institutional theory (DiMaggio and Powell, 1983). Scott (1987; 1995) grouped institutional effects into two board categories: pre-conscious and post-conscious.

Figure 4.4 will help to demonstrate that some of the new tools of cost management - ABC/ ABM - contain aspects of the same idea used in past tools of cost management.



<sup>&</sup>lt;sup>66</sup> Efficiency is the relationship of outputs to given sets of inputs. It can only be discussed in a bounded system where the boundaries are closed for analysis.

<sup>&</sup>lt;sup>67</sup> There are three ways of using cost management to increase efficiency; throughput costing, benchmark and target costing, and cost leadership.

Berry, et al., (1995: 55) say that Johnson and Kaplan (1987), in their critique of management accounting and control, argued that the systematic structures of cost accounting did not make much of a connection with the systematic problems of managing an enterprise. To recapitulate a little, the structures of traditional cost accounting, especially methods such as full absorption costing, fit beautifully with the notion of functional bureaucracy, in that they are a classic case of decomposition and disagregation of hierarchic elements to lower-order activities.

Managers need an analysis to enable them to see the relationship between the way in which they were using resources and the way in which they were allocating costs. Even if we accept the economic managerial resource vs. managerial cost analysis, it should be noted that the use of full absorption costing to reflect opportunity cost was unhelpful and inadequate. What Kaplan argued was that it was important to break this systematic notion of full absorption costing, and to actually create the idea of activity pools to which overheads could be allocated. Then these activity pools would be rigorously criticised (in other words, there would be the emergence of the ideas of Zero-based budgeting, i.e. a new form) to decide upon appropriate cost structures, then the costs in the activity pools would be allocated to products on some reasoned basis.

In the late 1980s and early 1990s ABC has become widespread. It is claimed that ABC can be used as part of the strategic management process through understanding cost behaviour and analysing the profitability of customers and the newly created products (Brignall, 1997). In ABC and the traditional literature, cost causality is related to internal factors like number of set-ups, number of machine hours or number of products (Bjornenak and Olson, 1999: 331). The ABM literature expands the view on cost causality in activities by identifying a set of different drivers, e.g. initial drivers, resources drivers and cost drivers (AX and Ask, 1995).

In conclusion, it was observed that most models of cost management previously mentioned such as; FAC, ZBB, ABC, and ABM focus on internal factors and neglects the external factors. This leads to choose institutional theory as the framework because it helps to recognise the institutional context, both inside and outside of the organisation. Furthermore, this leads the researcher to look for another model of cost

management that considers internal and external environment (see strategic management accounting).

# Chapter 5

# Strategic management accounting and cost management

# **5.0 Introduction**

The aim of this chapter is to focus on cost accounting in service organisations and illustrate strategic management accounting as one of the models of cost management which considers the internal and external environment. Furthermore, it focuses on how cost accounting can support corporate strategy. For these purposes, this chapter is divided into four sections.

The first section illustrates the gap between changes in cost accounting practices and changes in textbooks and it links the idea of cost accounting in the past and the present. It also highlights the cost accounting in a service organisation.

The second section examines strategic management accounting (SMA) from four viewpoints (Lord, 1996). The first viewpoint emphasises the extension of traditional management accounting's internal focus to include external information about competitors. The second viewpoint has been developed in business strategy literature and has examined the relationship between the strategic position chosen by a firm and the expected emphasis on management accounting. The third viewpoint advocates the analysis of ways to decrease costs and/or enhance differentiation of a firm's products through exploiting linkages in the value chain and optimising cost drivers. The fourth viewpoint challenges the basis of the other viewpoints. It points out that they assume that managers are able to deliberately plan what strategy the firm will follow.

The third section aims to explain how costing techniques can support strategy. It gives a brief discussion on six cost management techniques that support strategy and illustrates how these six models can help to manage the cost of existing products and future products.

The final (fourth) section outlines the main conclusion of this chapter.

## **5.1** Cost accounting

#### 5.1.1 Changes in cost accounting practices textbook

There is a gap between the content in changes costing in textbooks and the methods used (Scapens, 1991). The conventional wisdom of management accounting textbooks suggests that in many situations absorption costing cannot provide relevant costs. In practice, however, absorption costing seems to be the dominating technique in most Western countries (e.g. U.S.: Howell, et al., 1987; U.K.: Drury and Tayles, 1994; 2000; Sweden: Ask and Ax, 1992).

The gap between costing practice and textbooks and the current criticisms of both costing practices and conventional wisdom (e.g. Johnson and Kaplan, 1987) are based on a limited number of studies of costing practices. Scapens (1994) argues that more attention should be given to management accounting practices, and less to the comparisons of management accounting practices against theoretical "ideas". There is little<sup>68</sup> evidence of how different costing methods are being used. There is less research on why one method is being used, and even less research on why one method is preferred to another, and on how accounting practices have developed over time or on the sources (institutions) that have influenced the development.

This research does not aim to highlight the gap but rather the clear link between changes in practices and textbooks. Table 5.1 shows this link.

<sup>&</sup>lt;sup>68</sup> With the exception of few studies (e.g. Ask and Ax, 1992; Drury and Tayles, 1994; 2000; Innes and Mitchell, 1995).

	Source of	Textbooks major (new)	Costing practice major
	inspiration	topics	characteristics
1936-55	German tradition	Advanced allocation	Advanced allocation
		How to implement	absorption costing
		Absorption costing	
1955-70	U.S./ Danish	Contribution margin	Contribution costing
	inspiration	Standard costing	
		Variability accounting	
1970-95	U.S. inspiration	Decision orientation	Different methods for
		Optimal production	different situations
		Inventory valuation	

 Table 5.1: Major characteristics in three phases

Source: Bjornenak, T. (1997)-" Conventional Wisdom and Costing Practice", p. 378.

The link between changes in costing practices and changes in textbooks seems to continue. The major new topic today is Activity Based Costing. Most textbooks adopted the conceptual framework related to ABC around 1990. A gap between costing practices and conventional wisdom in textbooks is found in all periods.

Bjornenak (1997) concluded that no one can assume that textbooks have a major influence on costing practices. More likely, both costing practices and textbooks are the results of the same institutional influences, for example academic institutions or consultants introducing the contribution margin approach in the 1960s or ABC in more recent years.

### 5.1.2 Cost accounting in the past and now

Clark (1923) coined the phrase "different costs for different purposes: stock valuation, planning control and decision-making (Johnson and Kaplan, 1987; Drury and Dugdale, 1992). Chandler (1977: 109) argued that the key to subsequent developments is found in change. "Did cost accounting become a basic tool of railroad managers" (Chandler, 1977: 116).

Cost accounting is not embedded in accounting theory. German literature has emphasised that accounting systems transform cash flow into flows of periodic surplus. German literature identifies four basic systems for measuring monetary flows and hence income, namely:

(1) Net cash flows, which are defined as changes in liquid funds. They are used as the underlying income concept to perform capital budgeting

(2) Changes in the sum of liquid funds plus accounts receivable minus liabilities. Kloock and Schiller (1997) define flows as being the basis to Riebel's "costing" system (Riebel, 1974)

(3) Earning, defined as revenues minus expenses

(4) Operating revenues<sup>69</sup> minus costs<sup>70</sup>

In Anglo-American literature the term "cost" does not often seem to be well defined and is frequently confused with other concepts of surplus determination. Costs always include a quantity components (consumption of input factors) and a value component. The associated value component can be measured in two ways. The first involves only the assignment of payments in the form of actual, past or future market prices to the consumed input factors. This is the concept of "pagatory" costs<sup>71</sup>. The second approach adds opportunity costs to pagatory costs, and from optimisation theory, opportunity costs can only be computed after optimal decisions have determined (Hax, 1965: 145). Nevertheless, the system of marginal costing allows for the inclusion of opportunity costs. Lucker's Theorem is important for costing purposes. It demonstrates how to compute the capital given the decision-maker has decided to measure income by (revenues - costs). Lucker's Theorem bridges the gap between long-term and short-term decision-making.

Cost accounting has not been central to WCM ideas, though activity-based costing (ABC) is sometimes incorporated within it. However, WCM with its emphasis on non-financial measures, continuous improvement, computerised information systems, and

<sup>&</sup>lt;sup>69</sup> Operating revenues are defined as the periodic creation of goods (or services) in the pursuit of the substantive goal of the corporation.

 $<sup>^{70}</sup>$  Costs are defined as the value of the periodic consumption of goods (or services) in pursuing the substantive goal of the corporation.

<sup>&</sup>lt;sup>71</sup> This expression is German which means "consisting of (cash) payments".

changed management structures and roles is likely to impact upon the role of management accounting.

In the pre-WCM era it was thought that production could be managed "by numbers". The numbers would show what to make, what to buy, who to blame. If, for example, the latest cost report shows a negative cost variance in welding, the onus is on the welding supervisor to cut costs. But how? There is no data on the causes of the cost average - WCM mandates simplification and direct action: Do it, measure it, diagnose it, fix it, manage it on the factory floor. Don't wait to find out about it by reading a report later (Schonberger, 1986).

Schonberger's position is not dissimilar to Johnson's thesis in Relevance Regained (1992). Johnson argues that accounting relevance was not lost but rather it was never relevant to the efficient running of businesses: "Certainly I do not believe the answer is to reform how we do management accounting. In that regard I refute the advice. You hear from people who advocate activity based costing, or ABC, as the panacea to what ails American business... Improving how companies trace overhead costs may be important for some things, but in the guise of cost drivers, ABC does not necessarily stimulate continuous improvement nor does it mark a pathway to competitive excellence" (Johnson, 1992: 8).

Schonberger (1991) passed similar comments indicating that whilst ABC may be relevant to product line decisions it was unsuitable for performance assessment, adding that the cost reporting role would diminish as managers spend more time on physical customer-centred measures of continuous improvement. Clark is often given the credit for introducing the concept of "different costs for different purposes" (Frank, 1990). Schmalenbach introduced the importance of dividing fixed and variable costs for decision purposes as early as 1899 (Solomons, 1952).

Coward (1944: 287) argues that "the costing schemes should be designed in a way that enables one to separate direct costs, variable overhead and fixed overhead". Shark (1943) and Coward (1944) presented alternative classifications of cost pools, called "kostnadssted". The normal choice of "kostnadssted" would be department, but Coward argued for other alternatives: "if a manufacturing company follows the conversion of

materials in different steps through the factory, then it is natural to divide the production process into sub-processes" (Coward, 1944: 49). The meaning of the concept "sub-process" is the same as what we today might call an activity. Whether using activities, functions or departments, the importance of using many cost pools is pointed out, and little attention is given to the use of only one plant wide overhead cost pool.

Henzel (1931) argued that the overhead costs should be allocated on a casual basis. Coward (1944) discussed how to allocate under-absorbed cost due to sub-normal activity. Coward calls these costs "cost of unused capacity" and discusses eight different approaches to this problem. The same concept is used in the contemporary debate about spending versus consumption (Cooper and Kaplan, 1991; 1992).

# 5.1.3 Cost accounting in service organisations

Silversto, et al., (1992)<sup>72</sup> identified three different service delivery process types, professional, shop and mass, whose costing systems also varied systematically (Fitzgerald, et al., 1991). Brignall, et al., (1991) discovered that cost traceability varied systematically, being greatest in professional services and least in mass services. Mass services (for example Water services) use a complicated mix of equipment, facilities and mainly back-office staff at each strategic business unit to deliver a standardised service. Professional services are labour intensive, with their cost structure dominated by front-office staff delivering a customised service.

Traditional full product costing systems trace input costs into the conversion process, separating them between those which can then be traced directly to outputs (products) and those which cannot (indirect costs): these indirect costs are then allocated to products through a two-stage procedure. The indirect costs are first collected into cost pools within the conversion process, and then attached to products by some method based on unit volume of production such as direct labour hours.

Johnson and Kaplan (1987) argue that this second step may cause product costs to fail to reflect the resources they consume. ABC<sup>73</sup> and Throughput Accounting<sup>74</sup> aim to improve product costing system by attacking different parts of the two stage allocation.

<sup>&</sup>lt;sup>72</sup> From a study of operations management literature and their research in 11 service organisations.

<sup>&</sup>lt;sup>73</sup> See section 4.3.

The life cycle model (Koutsoyiannis, 1982) argued that a business unit's/product's life has four basic stages (launch, growth, maturity, decline) with attendant changes to various contingent variables such as its external environment, mission and chosen generic strategy (Porter, 1980). Business risk is high in the first phase (where the future is uncertain and the mission is to build) and low in the final phase, but financial risk is low in the first phase yet high by the final one. As businesses in the launch phase are cash-negative, and businesses in the mature phase cash positive, appropriate capital structure moves from equity (perhaps venture capital) at the start to a mix of debt and equity by maturity and debt in the final phase (Ward, 1993). Cost structure and cost behaviour will therefore vary systematically across the business life cycle. Cost tracebility may also vary if fixed, indirect costs are kept to a minimum unit at the mature stage.

In consequence, management information systems (and the costing and performance management systems within them) may also need to change through time so as to maintain the visibility of the changing factors critical to competitive success at each stage of the life cycle. Ideas drawn from Target costing and life cycle management may be more helpful for cost management here than conventional ABC or Throughput Accounting. For example, "Target costing is not actually a form of costing. Target costing is an activity which is aimed at reducing the life-cycle costs of new products, while ensuring quality, reliability and other consumer requirements, by examining all possible ideas for cost reduction at the product planning, research and development, and the prototyping phases of production. But it is not just a cost reduction technique, it is part of a comprehensive strategic profit management system" (Kato, 1993: 36).

Some authors (Govindarajan and Shank, 1992; Adamany and Gonsalves, 1994) argue that the type of cost and other performance management information to be monitored should vary at different organisational levels and stages of the business/product life cycle.

<sup>&</sup>lt;sup>74</sup> Throughput Accounting attacks the first half of the two-stage costing model by eliminating the distinction between direct and indirect, treating all costs as facility factory costs (Goldratt and Fox, 1989). Throughput Accounting is similar to the contribution per unit of scarce resource approach and assumes that all non-material costs are fixed in the short-run (for Throughput Accounting, see section 4.1).

## 5.2 Strategic Management Accounting (SMA)

The aim of this section is to review literature relating to strategy and management accounting. Strategy is hotly debated and elusive (Dent, 1990). The term is employed in literature of many disciplines. Penning (1985) notes that economists, social psychologists, anthropologists, sociologists and political scientists use it, and in addition, literature abounds with definitions ranging from the general to the specific (Dent, 1990). The term "strategy" has been borrowed from the military. It is defined as the "art of so moving or disposing troops or ship or aircraft as to impose upon the enemy the place and time and conditions for fighting preferred by oneself" (The Concise Oxford Dictionary, 1983).

The theorists have attempted to provide taxonomies of "generic strategy" that are generalisable across firms and industries. There was considerable agreement in the literature that the scope of strategic choice includes: (1) Corporate strategy that deals with the allocation of resources among various business or division of an enterprise (2) Business strategy that concerns primarily the question of the competitive position of a particular business or division of an enterprise (3) functional (operational) strategy that is limited to specific functional areas (e.g., marketing and distribution) within particular businesses. Each of these strategies is concerned with specific sets of strategic characteristics and choices, which are related to different levels of an enterprise (Dent, 1990; Johnson and Scholes, 1993).

Literature on strategic management accounting has several views. These views could be divided into four viewpoints (Lord, 1996). The first viewpoint emphasises the extension of traditional management accounting's internal focus to include external information about competitors. The second viewpoint has been developed in the business strategy literature, identifying the relationship between the strategic position chosen by a firm and the expected emphasis on management accounting. The third viewpoint advocates analysis of ways to decrease costs and/or enhance differentiation of firm products, through exploiting linkages in the value chain and optimising cost drivers. The fourth viewpoint challenges the basis of the other viewpoints. It points out that they assume managers are able to deliberately plan what strategy the firm will follow.

#### 5.2.1 The first viewpoint of SMA.

Johnson and Kaplan (1987: 33-34) wrote that, "Carnegie relied on financial information which concerned his competitors' direct production costs. Carnegie's operating strategy [of pushing] his own direct costs below his competitors'... promoted him to require frequent information showing his direct costs in relation to those of his competitors".

Simmonds' (1981; 1982) definition and description of strategic management accounting focused on comparison of the firm with its competitors. The contribution of Simmonds (1981; 1982; 1985; 1986) was most significant from the viewpoint of showing how strategic management accounting might be undertaken (Wilson 1991: 100-104). He suggested adding more relevant market data to management records and a regular reporting system. Simmonds advised accountants to collect market share details, costs, sales and volume data against those of competitors (market leader, close competitors and laggards). He emphasised in particular the need for the relative competitive data on levels and trends of prices, costs, profit and volume changes for the firm and its competitors. For example, Simmonds (1981) pointed out that management can gain strategic advantage via its pricing policy and low cost strategy (e.g. the market leader can control the competitive situation with the highest volume and profits, plus the lowest unit cost and price).

In addition Simmonds drew some attention to time and long run strategic reconfiguration in relation to cost structure "profit is stemming from the pattern of competition over time in the competition configuration 'rather than from' internal efficiency. For example, increased profit can be reflection of decay in competitive position as a result of higher prices, decreased quality, reduction in advertising or any other reduction in spending that reduces market share" (Simmonds, 1981: 26).

Bromwich (1988: 26) intends that the focus of the firm should be on external matters, as "it is in the firm's markets that profits are made and where competitors challenge the enterprise". He said that it is necessary to go even further than Simmonds' suggestions, not only comparing the firm with its competitors, but also evaluating "the benefits of the enterprise's products both from the customer's point of view and the firm's perspective" (p. 27).

Allen (1985) stated that information for strategic decision-making must be forward looking not based on past costs and the concept of capital maintenance, but being concerned with values, investments and cash flows over the long-term continuum. Taylor and Graham (1992) promoted the inclusion of non-financial information crucial to strategic planning and control.

#### **5.2.2** The second viewpoint of SMA

Before the relationship between strategic management and management accounting is examined, one question should be addressed: Why do we need to be concerned with the relationship between strategic management and management accounting at all?

One reason is that it seems as if every organisation is supposed to have both a goal and an idea of how to reach this goal. In addition, it must be efficient in its actions towards achieving the goal before we will consider calling it an organisation. A more practical reason for looking at the relationship between the two kinds of management is the fact that many organisations over the years use a considerable amount of resources, measured in employees' time and company money, on making budgets, accounts and strategies. Furthermore, when both managers and researchers are asked what the difference is between strategic management and management accounting, they cannot give a clear answer, although none of them would reduce one discipline or practice at the expense of the other.

The relationship between strategic management and management accounting has been described from several perspectives such as a hierarchical, contingency theory, Miles and Snow (1978) or new economic view. The relationship between strategic management and management accounting has been described as a hierarchical relation within much of the literature that takes a rational and objectivistic perspective on organisations and their environment.

Ansoff (1965) and Anthony (1965)<sup>75</sup> saw Strategic Management as a holistic interdisciplinary device that was engaged in determining the relationship between the organisation and its environment. In comparison, Management Accounting was seen as an instrument for calculation, implementation, and control that belonged to the tactical and administrative level as only one out of several functional disciplines within the organisation. Strategic Management is supposed to deal with the future of the whole organisation in a creative and innovative form, whereas Management Accounting is expected to handle the history of the company and all its every day details and routines.

Andrew (1971) took this view even further in a formal planning perspective. He saw strategic management as the first part of the rational decision-making process concerned with goal setting and development of strategic alternatives. Management accounting, on the contrary, was concerned with evaluating alternative strategies and translating these into operational and measurable budgets for securing and controlling the implementation of the strategy.

During the 1980s and 1990s, the planning and control tradition extended the use of financial measures to include non-financial representations as strategic indicators, organised in strategic control concepts like Critical Success Factors (Bruno and Leidecker 1984), Benchmarking (Spendolini 1992) and the Balanced Scorecard (Kaplan and Norton 1992; 1996).

<sup>&</sup>lt;sup>75</sup> Anthony (1965) prescribed three normative and universalistic control activities to be the core functions of managing: strategic planning, management control and operation (task) control. Strategic planning was viewed as being concerned with setting and changing of overall corporate strategies and objectives and the establishment of policies to govern the acquisitions and uses of resources. Management control is involved with the manner in which managers monitor activities and take action to ensure that resources are being achieved as stipulated in strategic planning. Task control (physical production) was concerned with the monitoring of specific day-to-day activities that call for little judgmental activity (due to differences in tasks and the nature of the activity itself, Anthony, et al., 1989).

Anthony had no doubts, "structure is given- systems are therefore designed to meet the needs of a given structure" (Anthony, 1965). The strategies were taken as given and control systems were used to implement these strategies two levels: management level and task level. It should be noted that Anthony's model assumed a hierarchical relationship (institutional hierarchy) between three separated layers of management in a formal stable world: strategic planning, management control and operational control. This categorisation explicitly avoided addressing the issues of organisational goals and objectives formulating process (Berry, 1983; Berry, et al., 1991; Otley and Berry, 1980) and adoption to environmental contingencies (Otley, 1980). However, this perspective of strategy and control procedures of cost practices may have been suited to organisations that have a wide geographical dispersion of sub-units where reliance on control procedures as designed is greater (Emmanuel, et al., 1992).

What are the problems with applying this hierarchical perspective on the relationship between strategic management and management accounting?

First of all, when we take a rationalistic view, it is not clear whether strategy or economics are the goal or the means of the organisation. This depends on the situation and the actors speaking about strategy and economics. Secondly, these perspectives look at the organisation from an objectivistic and rationalistic point of view, which presumes that a manager's task is limited to increase efficiency in the organisation and to secure optimisation and control. In real life, top managers often need to take care of many other tasks besides efficiency, such as securing the legitimisation of the organisations (Mintzberg, 1983) and participating in many symbolic activities as well as non-formal actions (Mintzberg, 1973). Third, the efficiency perspective on planning and control is not adequate to explain the processes that take place in organisations between strategic planning, budgeting and the follow-up on budgets by utilisation of accounting and strategic control devices.

Goold and Quinn (1990) point to four problems of strategic control, namely: to overcome uncertainty and flexibility at the same time, to motivate rather than control managers, to assist rather than replace management judgement and to secure rather than destroy mutual confidence between management levels. From a management accounting view, Dent (1990) suggested that we should at least look at the relationship between strategic management and management accounting from three different perspectives, namely, as already suggested above, from a control system perspective. from a decision-making perspective and from a perspective legitimising strategic change. By looking at the role of accounting in decision-making and how accounting provides a language for justification of strategic changes, Dent argued that it was possible to get much closer to what is going on in organisations' everyday life. Finally, it can be argued that leaders do not only conceive and justify their visions of the future organisation by creating images of the development of the surrounding environments, they also mobilise the past as an argument for present and future actions. Therefore, leaders also need to look at the capabilities of management accounting to legitimise the present and future by reconstructing the past.

Based on the above arguments, it seems as if the relationship between strategic management and management accounting is much more complex than just a typical hierarchical relationship. It appears that the concepts of both strategy and accounting play many roles within organisational life other than generating efficiency and profit.

The contingency approach is based on the premise that there is no universally acceptable model of organisation that explains the diversity of organisational design. It simply states that the most efficient form of organisation is contingency theory and the type of conditions relevant to the situation. In particular, under contingency theory, the type of management accounting system (MAS) varies according to specific circumstances or the situation in which the organisation operates. Hence, the choice of MAS design is constrained by these conditions and depends on the ability of management to find the best solution.

Within the contingency perspective, strategic management is more comprehensive than management accounting. Both Govindarajan and Gupta (1985) and Simons (1987) suggested how control systems could be best suited to a strategic business that follows a specific strategy. The major focus of Simons was on the link between business strategy and management control systems. He explicitly used the contingency framework as a dominant logic for research on control systems design (Simons, 1987: 358). His first attempt (1987) was to extend Miles and Snow's (1978) analysis to the financial control attributes system. It was based on the premise that control systems should be modified in accordance with the business strategy of a firm. However, his result was inconsistent with Miles and Snow's (and also Ouchi 1979; Hirst, 1983; Govindarajan, 1984) evidence that defenders tend to use more detailed cost analysis techniques and cost control oriented towards efficiency in task performance (Dent 1990). Simons (1987) found that prospectors tend to use their financial control system more intensively than defender with tighter budget goals and more frequent reporting on forecast and outputs than cost control.

Porter (1980; 1985) detailed two specific ways in which managers can position their firms so they have a strategic advantage over their competitors: firms need to either differentiate their product(s) or achieve a position of cost leadership. To differentiate its product, competitive advantage is attained by being able to charge high prices, by being able to sell more at the given price, or by achieving increased customer loyalty. For cost leaders, having lower costs than all competitors attains competitive advantage.

Shank (1989) and Shank and Govindarajan (1989) analysed the relative importance of several management accounting methods depending on whether the firm was pursuing cost leadership or differentiation. They suggested that companies choosing cost leadership would put the most emphasis on traditional cost accounting applications. They would use standard costs to assess performance, product costs as an input to pricing decisions, and flexible budgeting for manufacturing cost control. They would perceive meeting budgets and analysis of competitors' costs to be of great importance. On the other hand, companies differentiating their products as a way of achieving competitive advantage would consider marketing cost analysis to be critical to their success. They would consider flexible budgeting and meeting budgets to be of only moderate importance, and rank standard costing for performance assessment, product costing for pricing decisions and competitor cost analysis of little importance.

Shank (1989), Shank and Goverdarajan (1989) focused on the question of how cost information could be used to assist managers in formulating and implementing strategy. They developed a conceptual framework based on a four stage process of the strategic management cycle; formulation, communication, implementation and action. They defined strategic cost management (SCM) as "the managerial use of accounting information explicitly directed at one of the four stages of the strategic management cycle" (Shank, 1989: 50). The proposed framework of SCM simply was strategic control model concerned with the financial analysis of the value chain and cost driver in different strategic positions (Porter, 1985; Shank, 1989). To some extent, they addressed the literature debate on how the use of cost information is based on different costing models. For example, full absorption costing, activity based costing and contribution costing led to different scenarios on comparisons of competitors' cost structure and how competitors could attack firms who use each of these models. Their suggested solution to the problem of variance analysis was that companies should develop a strategic management accounting system rooted in Simmonds' approach.

Bromwich (1990) used two economic theories in an effort to provide theoretical support for strategic management accounting, linking the theories with Porter's taxonomy. Their theory attributed analysis supports accountants "costing attributes and monitoring the performance of these attributes over time" (p. 28), which in turn contributes to Porter's differentiation strategy. The theory of contestable markets "requires that the accountant extends cost analysis beyond the firm and reports on the cost structure of rival enterprises" (p. 29), which contributes to Porter's cost leadership strategy.

Bromwich (1990) extended this perspective to include the differentiation strategies on management accounting reports. He tried to differentiate between two types of cost information in relation to strategic position. The first was related to differentiation strategy and was concerned with "costing the characteristics provided by goods and in monitoring and reporting on these costs regularly" (Bromwich, 1990: 44). The second was related to low cost strategy and was concerned with the analysis of "competitors" costs and cost structures and the monitoring of the enterprise's strategy and those of its competitors in these markets over a number of periods" (Bromwich, 1990: 28).

Bromwich's work was based on economic theory on the value of product attributes in differentiation (Lancaster, 1966; 1979) and more recent work on the theory of contestable market (Baumol, 1982; Baumol, et al., 1988). He drew on these two economic theories to provide a theoretical basis for the involvement of accountants in strategic cost management. Bromwich used these two theories to establish the need for accountants to compare the cost structures of their own firms with those of actual and potential competitors in relation to demand factors. In particular, Bromwich drew attention to both (1) the role of sunk cost as an important feature of cost structures in creating barriers to entry and then to determining the sustainability of enterprise strategies (2) The cost of joint products to be of critical nature when compared to the cost of producing these products individually (economic of scope).

Govindarajan and Shank (1992) added dimension to the expected management accounting emphasis, by linking the firm's "mission" (i.e. build, hold or harvest) to its strategic position. They concluded that accounting implications for strategic planning, budgeting and incentive compensation would be similar under both 'harvest' mission and cost leadership strategic position. Similarly, the accounting emphasis suggested for the differentiation strategy would also fit a 'build' mission.
Simons (1990: 141) added the suggestion that defenders "need only focus on strategic uncertainties often related to product or technological changes that could undermine current low cost positions". Simons focused on the management process as it related to management control and strategy. Basically he considered the management control system as tools used for both strategy implementation and strategy formulation. He provided a "dynamic process model of strategy and control in which top management regulates strategic (environmental) uncertainties through its choice of strategy and the institutionalisation of management control system" (Gray, 1990: 145).

Simons (1990) provided empirical evidence of the way in which management control systems were not only important for strategy implementation, but also for strategic formulation. His research was based on a two-year field study and it provided an illustration of how top managers used formal systems to guide the emergence of new strategies and ensure competitive advantage.

Simons introduced the idea of interactive management control as opposed to programmed control: "Management controls become interactive when business managers use planning and control procedures to actively monitor and intervene in on-going decision activities of subordinates. Since this intervention provides an opportunity for top management to debate and challenge underlying data, assumptions and action plans, interactive management controls demand regular attention from operating subordinates at all levels of the company. Programmed controls, by contrast, rely heavily on staff specialists in preparing and interpreting information. Data is transmitted through formal reporting procedures and operating managers are involved infrequently and on an exception basis" (Simons, 1990: 136).

The focus of attention in Simon's work was business strategy and the way in which top managers use interactive control systems to monitor the progress of the business unit towards its business goals. He argued that the top managers make selected control systems interactive in order to monitor the strategic uncertainties that they believe are critical to achieving the organisational goals. It is also suggested that these control systems can be linked to the use of subjective reward systems which are not formula based. Company B, for instance, in Simons (1990) had a reward system based on effort. As a result of debate

and dialogue that surrounds the interactive management control process, new strategic and tactics emerge over time.

Although the view of management control presented in his paper (1990) differed from the traditional approach (management control is an important tool for strategy implementation). His methodological approach was linear, based upon a functionalist approach "the control system refers to formalised producers and systems that use information to maintain or alter patterns in organisational activity" (Simons, 1990: 623). His model essentially was normative and was based on the premise that top managers are always able to alleviate the effects of environmental uncertainties by designing an appropriate control system (contingency view). Simons (1991) gave some attention to the complex nature of strategy formulation and implementation. Presumably, Simons viewed strategies as organisational learning and as an emergent process (Hartman, 1993).

The analysis thus far has been based on contingency theory approach, which has been criticised on the grounds that it suffers from two fundamental problems related to conceptual deficiencies and methodological assumptions. Firstly, the contingency approach is epistemologically rooted in functionalism (Otley, 1980; Berry, 1983). These critics argue that contingency theory is based on linear deductive model of causality and relationship, and assumes a highly technical view of organisational choice. Hence, studies based on this theory tend to ignore the wider context of organisation as well as the critical and dynamic issues such as organisational goals and effectiveness (Otley, 1980; Otley and Berry, 1980; Lowe and Puxty, 1989) and authority and accountability (Berry, et al., 1985). Secondly, the conceptualisation, definition and measurement of key variables within a contingency theory have not been adequately elucidated which has led to an emphasis on statistical analysis at the expense of theoretical and empirical attention (Otley, 1989; Dunk, 1989). Despite the difficulty in employing a contingency theory, it was commonly agreed that this approach provided useful insight for understanding how environmental factors were related to management accounting systems (Berry et al, 1991; Hoque, 1993).

Miles and Snow (1978: 29) classified organisations by their way of responding to the environment, and according to their "particular configuration of technology structure, and process". Miles and Snow's typology (1978) attempted to study strategy at business level in four industries. They identified four major viable strategies: Defenders, Prospectors,

Analysers and Reactors. The dimension underlying the typology is the rate at which an organisation changes its products or markets. Defenders are organisations (strategic business unit, SBU) that engage in little or no new product development or new market development. Often they control relatively secure niches within their industries, competing primarily by price, quality, delivery or service. Prospectors attempt to pioneer in product/market development. They tend to offer a frequently changing product line and compete primarily by stimulating and meeting new market opportunities. Analysers are an intermediate type. They make fewer and slower product/market changes than prospectors, and are less committed to stability and efficiency than defenders. Reactors behave with a passive strategic adaptation.

Miles and Snow's view formed the basis for a range of research concerning the best fit of strategy and MAS design (e.g. Simons, 1987; 1990). This body of research emphasised the relationship between the firm and changing market and focuses on how an optimal accounting system can be designed in accordance with a specific strategy the firm had decided to follow.

Within the "new economic view" (Hartmann, 1999) we found, among other perspectives, transaction cost theory, evolution theory and resource based perspective. The transaction cost perspective explained the existence of market and companies, vertical integration, product diversification, and divisionalisation of companies, etc. by referring to the cost of co-ordination in each transaction. The evolutionary theory explained the company as a transmission and variation mechanism, whereas the market function was called a selecting mechanism selecting, choosing the most economic efficient companies. The resource-based perspective saw the company as a bundle of resources and competencies that competitors could not easily imitate due to the complexity of the competencies, their historical nature, and because the competencies sometimes existed as tacit knowledge. Even though these perspectives differed extensively in their perception of rationality and in their explanation of the firm and competition, they used economic efficiency as an explanatory factor, thus reproducing a hierarchical view of management.

#### 5.2.3 The third viewpoint of SMA

Cost can be reduced by reducing activities that incur costs without increasing value. They can also be reduced by exploiting linkages in the value chain. The value chain is "the linked set of value-creating activities all the way from basic raw material sources... through to the ultimate end-use product" (Shank, 1989: 50).

Porter's model (1985) provided a potential alternative to ABC. He proposed a valuechain (VC) model which breaks down the firm's activities into nine strategic units of interrelated activities and collects cost information about a set of value-creating activities all the way from basic raw material suppliers to the ultimate end-users (see Wilson, 1991 for summary). Porter (1985) listed twelve factors, including learning, economies of scale, interrelationships with suppliers, and discretionary policies, that should be examined with a view to reducing costs. He identified three strategic postures to manage pricing in relationship to the VC cost<sup>76</sup>. So while some companies tended to construct efficient scale facilities and reduce cost through accumulated experience, others gained strategic advantage via the cost of differentiation (e.g.; services and image) in the market place (Brignall, et al., 1991). Porter's value-chain approach becomes difficult to use in practice (Hargert and Morris, 1989; Hartman, 1993), because firms define their business activities around responsibility centres rather than strategic business units and value chain activities as defined by Porter.

Porter (1985) presented value chain analysis to gaining competitive advantage. The aim of value chain analysis is to find linkages between value creating activities, which result in lower costs and/or enhanced differentiation. This linkage may be within the firm or between the firm and its suppliers, channels and customers.

Shank and Govindarajan, (1988; 1992a; b; c) have promoted accounting inputs to chain analysis under the name "strategic cost management". They gave several examples of how value chain analysis using accounting figures would result in different decisions than using traditional management techniques.

<sup>&</sup>lt;sup>76</sup> These strategic postures are: (1) cost leadership (competition on price) (2) product/service differentiation (3) focus strategy (a closely defined target segment of the market [niches])

Hargert and Morris (1989) asserted that much accounting is not in a suitable form to be used in analysing the value chain. They pointed out that a traditional management accounting system does not adequately quantify the costs and benefits of joint optimisation and co-ordination between parts of the firm, and between the firm and suppliers and buyers. Hargert and Morries, (1989) considered that the process itself provided a useful insight, even if it was impossible to estimate precise numerical output. "One of the strengths of value chain analysis is that it forces managers to think about which activities created profits, to choose a generic strategy for each product and to ask of each item of expenditure "how does this add value to buyers"? (p.187).

Shank and Govindarajan (1992b) recognised that there were problems involved in calculating the value chain, but reiterated that "Even the process of performing the value chain analysis, can be quite instructive" (p. 184).

Shank (1989) grouped cost drivers into two types: structural and executional. Structural cost drivers are scale, scope, experience, technology and complexity. Increasing these structural drivers does not necessarily decrease costs. Executional drivers include work force involvement, total quality management, capacity utilisation, plant layout efficiency, product configuration effectiveness and exploitation of linkages. Shank (1989) pointed out that the cost drivers analysis suggested by Porter (1985) was a much broader concept than "activity based costing", which focuses primarily on complexity.

Nanni, et al., (1992) recognised the need for performance measurement both as a guide to putting strategy into action and evaluation of actions taken. However, they suggested that both financial and non-financial measures be used.

As mentioned in the introduction, this viewpoint advocates analysis of ways to decrease cost and/or enhance differentiation of firm products, through exploiting linkages in the value chain and optimising cost drivers. This view leads to the issue of supply chain management. According to Harland (1996), the term "Supply Chain Management" (SCM) seems to have originated in the early 1980s in discussion on the integration of the internal business functions of purchasing, manufacturing, sales and distribution. Slack, et al., (1998) defined SCM as a holistic approach to managing across company boundaries.

Ellram (1991) saw SCM as a competitive tool that share similar advantages and disadvantages with vertical integration<sup>77</sup> and obligation contracting<sup>78</sup> form of governance. The advantages of vertical integration as opposed to obligation contracting were that integration increased control, reduced the probability of opportunism, improved communication and avoided the replication of activities. The disadvantages of integration included problems in achieving economies of scale, problems of span control, the loss of high powered market incentives and the possible perpetuation of obsolete processes (Williamson, 1985).

The DTI (Department of Trade and Industry) (1997) argued that supply chain management is "the strategic management process, unifying the systematic planning and control of all technologies, materials and services, from identification of need by the ultimate customer. It encompasses planning, designing, purchasing, production, inventory control, storage handling, distribution, logistics and quality. The objectives are to optimise performance in meeting agreed customer service requirements, minimising cost, whilst optimising the use of all resources throughout the entire supply chain".

Berry, et al., (1997) argue that the above definition outlines four main uses of the term "supply chain management". The internal supply chain which integrates business functions had new materials and derived information in a "inbound" "outbound" way. Secondly, the management of dyadic or two party relationships with immediate suppliers. Thirdly, the management of a chain of businesses including a supplier, a supplier's suppliers, a customer and a customer's customer. Fourthly, the management of a network of interconnected businesses involved in the ultimate provision of product and service packages required by end customers.

Berry, et al., (1997) argue that total cost control is the idea of open book costing. Open book costing means the supplier opens his books to the customer and this supports the idea of active collaboration and partnership. Lamming (1993) talked about cost

<sup>&</sup>lt;sup>77</sup> Vertical integration internalises the whole activity (Williamson, 1985).

<sup>&</sup>lt;sup>78</sup> Obligation contracting whereby firms recognise their interdependence and ongoing nature of relationships (Williamson, 1985).

transparency and the sharing of cost information between customer and supplier in a way which allowed customer and supplier to work together to reduce costs. This was analysed into constituent parts and could support an exercise based on target costing or continuous improvement (Kaizen costing).

Seal, et al., (1999) argue that open book accounting could enable both greater trust as well as self-enforcing agreements. From an economics perspective, the relation between information and trust ran in two (opposite) directions. First, trust between partners could reduce costs of negotiating and monitoring agreements by avoiding the need for complex contingency agreements. Second, if the costs and rewards of agreements could be verified at low cost, then the need for trust was reduced.

Seal, et al., (1999) argue that there were a number of areas where accounting and, in particular, management accounting was involved in supply chain management. These areas included the make-or-buy decision leading to the choice of the partnership mode, the role of accounting in managing a partnership and the accountability of the partners to their internal constituencies. The researcher's view was that the supply chain had four important dimensions (reliability, availability, efficiency/effectiveness, and flexibility) and it has three major criteria. These were as follows:

(a) Cost

(b) Response of the market

(c) Strategic management of the competition

The researcher saw cost as the main element and the core idea of supply chain management was to eliminate non-value activity (cost effectiveness). For the purpose of this research it would be useful to focus on cost related supply chain management.

Slack, et al., (1998) demonstrated the gearing effect on profits of reducing material costs. Seal, et al., (1999) argue that traditional manufacturing was characterised by poor quality cost accounting information which could not identify drivers in the assemblers' own operations. The prescribed roles of management accounting emphasised reporting and controlling functions within a bureaucratic hierarchy. Management accounting might also be used in simple "make-or-buy" decisions with the danger here (as in other areas of decision-making) being that cost data from the reporting function may influence costs for decision-making (Johnson and Kaplan, 1987).

Lere and Saraph, (1995) highlighted the problem of supply chain management caused by cost and suggested a solution for it. They argued that most suppliers would tend only to provide information based on traditional costing. Lere and Saraph, (1995) suggested that one way round this problem was to deduce a supplier's cost by asking for a process flow chart that may indicate any wasteful, non-value adding activities and a breakdown of overheads to indicate activities that were performed specifically on his or her product.

Rajagopal and Bernard, (1993) emphasised the importance of focusing on "all-in-cost"<sup>79</sup> rather than purchase price. They argued that cost management drew on a number of functions but focused on total quality management, negotiating, and supplier alliances.

Seal, et al., (1999) argue that cost information not only played a role in the strategic sourcing decision but would also influence the ongoing management of partnerships. In a strategic approach, the ongoing management of a strategic partnership should be on a cost rather than a price basis.

## **5.2.4** The fourth viewpoint of SMA

The concept of strategic planning and positioning only covers part of business strategy. Strategies emerge from interaction between management, employees and the environment.

Mintzberg (1978) was one of the first to point out that strategic planning literature ignored other types of strategy formation. He made a distinction between strategy formulation and formation. He defined strategy formulation as long-range planning by leaders of organisations. Strategy formation, on the other hand, he defined as the result of interplay between the environment, the organisational operating system and the organisation's leadership. Mintzberg (1978: 935) defined the strategies that formed, "the realised strategies", as "a pattern in a stream of decisions". He considered a strategy to have formed "when a sequence of decisions in some area exhibits a consistency over time.

<sup>&</sup>lt;sup>79</sup> The term of "all-in-cost" refers to the sum of the purchase price plus all in-house costs involved in receiving and converting the purchased materials into finished products.

Strategy is not formulated and implemented solely by top management. There are likely to be several interest groups within an organisation; each having "its own set of stakeholders with whom it knowledge's relationships and whose exceptions it considers" (Dermer and Lucas, 1986: 473).

Simons (1992: 44) raised the question, "what role can accounting … play in… stimulating emergent strategies?" He suggested that accounting plays a role in interactive control systems. In a more recent piece of work, Simons (1995) talked about levers of control. In his 1995 work, Simons referred to interactive control systems and diagnostic control systems. Diagnostic control systems are in fact just another name for programmed control systems. What was interesting about Simons (1995) was the way in which he introduced the idea of belief systems (organisation culture) into his levers of control systems to boundary systems, interactive control systems, diagnostic control systems and business strategy.

Simons (1995) suggested that these four levers of control work in opposite ways to ensure effective strategy implementation and development. Two of the control levers (belief systems and interactive control systems) created positive and inspirational forces. The two other levers (boundary systems and diagnostic control systems) created constraints and ensured compliance with orders. Simons suggested that effective top managers used the levers of control to inspire commitment to the organisation; to stake out the territory for experimentation and competition; to co-ordinate and monitor the execution of today's strategies; and to stimulate and guide the search for strategies of the future.

Dermer (1990: 74) proposed that accounting has three roles in shaping strategy: it may be used as "a language of discourse", as an authority establishing and maintaining credibility, and as a provider of an historical context for strategy. Dermer also argued

<sup>&</sup>lt;sup>80</sup> The levers of control can be described as follows:

<sup>\*</sup> Belief systems - used to inspire and direct the search for new opportunities.

<sup>\*</sup> Boundary systems - used to set limits on opportunity seeking behaviour.

<sup>\*</sup> Diagnostic control systems - used to motivate, monitor and reward achievement of specified goals.

<sup>\*</sup> Interactive control systems - used to stimulate organisational learning and the emergence of new ideas and strategies.

that research on the relationship between accounting and strategy has been biased towards attempting to make accounting more useful to managers. He pointed out that stakeholders will access and use "the strengths and weakness of conventional accounting ... in ways not anticipated by accountants" (p. 75).

# **5.3 Cost accounting and corporate strategy**

The main aim of this section is to explain how costing techniques can support corporate strategy. In order to explain this, the section will firstly highlight how companies have to compete and what elements are successful. Secondly, it will give a brief discussion of six cost management techniques and how they support strategy. Finally, this section will explain how firms manage costs.

Sustainable competitive advantages play a critical role in determining the way firms compete. They do so by enabling firms to avoid competition by adopting the generic strategies of either cost leadership or differentiation (Porter, 1990). When firms cannot create sustainable competitive advantages, they are forced to compete by repeatedly creating temporary ones.

Three elements<sup>81</sup> play a critical role in determining the success of firms. These three elements comprise the product price, quality<sup>82</sup>, and functionality<sup>83</sup>. Competition is based upon firms performing better than rivals in one or more characteristics. In markets with perfect information and only strictly economic rational customers, the specific customer trade-offs between price, quality and functionality would be clearly visible. Therefore, well specified functional relationships between values for the three elements could be set down as the basis for determining a starting point.

When firms adopt a confrontation strategy, a highly effective cost management programme is a necessity not a luxury. Cooper (1996) argued that these programmes consisted of six products and production process related techniques. Three of them were

<sup>&</sup>lt;sup>81</sup> These three elements - price, quality, and functionality - are called survival triplet by Cooper, in (1995).

<sup>&</sup>lt;sup>82</sup> Quality is defined as conformance with product specification.

<sup>&</sup>lt;sup>83</sup> Functionality refers to the degree of success in designing the product to meet specifications that customer require.

forward oriented and designed to help manage the costs of future products. These techniques were:

(a) Target costing

(b) Value engineering

(c) Inter-organisational cost management systems

The other three techniques were feedback oriented and designed to help manage the cost of existing products. These techniques were:

(a) Product costing<sup>84</sup>

(b) Operational control

(c) Kaizen costing

The core idea of each technique will be explained in turn.

# Target costing.

Kato (1993: 33) defined target costing as "an activity which is aimed at reducing the life cycle costs of new products, while ensuring quality, reliability, and other consumer requirements, by examining all possible ideas for cost reduction at the product planning, research and development, and the prototyping phases of production. Kato emphasised that it was not just a cost reduction technique, rather it was part of a comprehensive strategic profit management system. Target costing was the discipline which ensured that new products were profitable when they were launched. There were two major steps in target costing. The first was to determine a product's target price and target margin so that its target cost could be determined. The second was to break down that target cost down to the component and raw material level so that the purchase prices of those items could be determined. A product's target cost was determined by subtracting its target profit margin from its target selling price. That is:

# **Target cost = Target price - Target margin**

The target price of a new product is determined primarily from market analysis. The target margin is set based upon corporate profit expectations, historical results, competitive analysis, and sometimes, computer simulations. The new product's target

<sup>&</sup>lt;sup>84</sup> See full absorption costing, activity based costing.

price can be used as the basis for determining the purchase price of components, and raw materials acquired externally (Cooper, 1996: 237).

The critical factor that identifies target costing (as opposed to other approaches to managing the costs of new products) is the intensity with which the rule, "The target cost can never be exceeded" is applied. Without the application of such a rule, target costing systems typically lose their effectiveness. Target cost challenges the calendartime perspective of costs (Monden and Hamada, 1991). Bjornenak and Olson, (1999) argued that target costing took a normative perspective. It said something about how it should be, and not how it has been.

# Value engineering.

Value engineering (VE) is a systematic inter-disciplinary examination of factors affecting the cost of a product in order to devise means of achieving the specified purpose at the required standard of quality and reliability at the target cost. VE, like target costing, is applied during the design phase of product development. VE is a multi-disciplinary team based approach. Teams are typically drawn from multiple function areas, including design engineering, applications engineering, manufacturing, purchasing, and, sometimes, even the firm's suppliers and subcontractors. VE plays a critical role in the management of product costs by helping the firm manage the trade-off between functionality and cost.

VE requires that each product's basic and secondary functions be identified and their values analysed. A basic function is the principle reason for the existence of the product. The secondary functions are outcomes of the way the designers chose to achieve the basic functions.

#### Inter-organisational cost management systems.

For many firms, the pressure to become more efficient has caused them to try to increase the efficiency of the firms that supply them with raw materials and components by developing inter-organisational cost management systems. For this it is necessary to be part of the most efficient supply chain<sup>85</sup>. There are many ways to achieve this

<sup>&</sup>lt;sup>85</sup> Seal, et al., (1999a) argue that inter-firm supply chains were embedded in institutional environments offering resources that may be drawn on by firms that face difficulties in regulation cost and quality issues.

objective such as: the company can opt to blur their organisational boundaries in numerous ways (organisational blurring typically occurs when information that is critical to one firm is possessed by another firm either further up or down the supplier chain); create relationships that share organisational resources, including information that helps improve efficiency of the inter-firm<sup>86</sup> activities.

#### *Operational control.*

Operational control requires holding individuals responsible for the costs they control and for determining how well they control them. The two primary techniques of operational control are the establishment of responsibility centres and variance analysis.

As regards to responsibility centres, for an individual to be held responsible for a cost, that cost must be assigned directly to the centre over which the individual has control. If indirect cost assignments are used, then it is impossible to hold the individual responsible for any apparent changes in the level of resource consumption. They cannot be held responsible because there is no way of knowing if apparent changes in resource consumption are due to distortions in the indirect assignment process or to an actual change in the level of consumption. In other words, the first technique aims to change the nature of cost management to focus on different objects and process objects.

As regards variance analysis, the traditional use of variance analysis is to monitor how well the responsible individual is keeping control over costs. The aim is to ensure that the budget is achieved.

#### Kaizen costing.

Kaizen stands for continuous improvement. Kaizen costing is the application of kaizen techniques to reduce the costs of components and products by a pre-specified amount. Kaizen, unlike Kaizen costing, typically does not focus on individual products, instead it focuses on making the production processes themselves more efficient. As a kaizen programme achieves its objectives, the overall cost of production and hence reported

<sup>&</sup>lt;sup>86</sup> Inter-firm relations were partly the outcome of technology, transaction characteristics and industry structure (Ellram, 1991; Slack, et al., 1998). Seal, et al., (1999a) added to that, inter-firm relations were also influenced by interaction between the firms' strategies and institutional environment of the industry.

product cost falls. Kaizen does not concern itself with product design, which is considered as given.

The difference between target and kaizen costing focuses on the point in the life cycle when the techniques are applied, (and what their cost reduction objective is). Target costing is applied during the design stage of the product life cycle. It achieves its cost reduction objective primarily through improvements in product design. In contrast, kaizen costing is applied during the manufacturing stage of the product life cycle. It achieves its cost reduction objectives primarily through increased efficiency of the production process.

Firms can manage costs in three ways. Firstly, they determine the mix (both present and future) of products that the firm sells. Secondly, they manage the costs of future products, and thirdly, they manage the costs of existing products.

The effectiveness of the various techniques for a given firm appears to be dependent upon several factors including the competitive environment, the maturity of the technologies used in the products, and the length of the product life cycle. The role of these three factors in determining the effectiveness of the six cost management techniques is primarily shaped by how the firm is competing using the following three elements: price, quality, and functionality. Under confrontation, firms compete on all three dimensions, but one of the three characteristics typically dominates the way the firm competes. The feed-forward techniques are particularly effective when firms are competing on functionality. The feedback techniques are particularly effective when firms are competing on price.

# **5.4 Conclusion**

This chapter has focused on cost accounting in service organisations. It has identified three different service delivery types, professional, shop and mass and related water industry as a mass service. The study sees strategic management accounting as the sole model of cost management which considers both the internal and external environment. Furthermore, it has centred attention on how cost accounting - six models - can support corporate strategy.

In ABC and traditional literature, cost causality is related to internal factors like number of set-ups, number of machine hours or number of products (Bjornenak and Olson, 1999: 331). ABM literature expands the view on cost causality in activities by identifying a set of different drivers, e.g. initial drivers, resources drivers and cost drivers (AX and Ask, 1995). In SMA literature the cost variability factor is inspired by the broader set of causal variability factors inspired by strategy literature (Bjornenak and Olson, 1999: 331). Porter (1985) included external factors (institutional factors) in his set of structural cost drivers. Thus, the cost driver concept in SMA is much broader than in ABC, which primarily focuses on complexity (Shank, 1989).

Chapters 2, 3, 4 and 5 have helped to build the framework for this research. Chapter two has shown the importance of both contingency and institutional theories for explaining the design and use of a cost accounting system. The chapter has argued that while technical (or contingent) explanations of cost management were not rejected, they were viewed as incomplete (Scott, 1987). This led the researcher to look at the institutional context within (e.g. which models of cost management were followed in the GOGCWS, and what methods of cost accounting were used in the GOGCWS, see sections 8.5, 8.6 and 8.7) and outside (i.e. the legal framework for water industry in Egypt, the external supervision and control body of the GOGCWS and the suggestions by the consultants Black and Veatch International which related to cost management, see sections 8.1, 8.3 and 8.7) the GOGCWS. The discussion of both the external and internal environments of the GOGCWS will be discuss according to neo-institutional economic theory in chapters ten and eleven (see figure 5.1).

Chapter three highlighted the different roles which accounting can play in organisational practice and social practice. It also illustrated that three institutional theories have been used in the accounting literature: old institutional economic theory which focus on micro-institutions within organisations; neo-institutional sociology which focuses predominately on extra-organisational institutions and neo-institutional economic theory which focuses on both micro and macro institutional contexts. As mentioned in section 1.4, this research adopts neo-institutional economic theory, this led the researcher to look at different kinds of environment in the water industry such as: (privatised water services in the UK, hybrid water services in France and hierarchical water services in Egypt, see section 7.2).

Chapters four and five focus on the model of cost management which this research has adopted. This research is based on a model of cost management which has three elements: efficiency, optimisation and strategy (see figure 5.1). Chapter four focuses on the first two elements of cost management: efficiency and optimisation. It identifies some models of cost management which indicate how to increase efficiency (i.e. throughput accounting, benchmarking, target costing and ABC) and others to increase optimisation (such as, ABM)<sup>87</sup>. The chapter identifies links between institutional theory, transaction cost and efficiency. It examines how the efficiency framework is favoured by transaction cost theories. Chapter five identifies strategic management accounting as one of the models of cost management which considers both the internal and external environment (the third element of cost management). This led the researcher to explore the question "is there any relationship between cost management (efficiency, optimisation and strategy) and the perception of competitive intensity and the perception of unpredictability related to three areas of the external environment (product market, factor market and legislation? See figure 5.1 and chapter 9.



Figure 5.1: The relationship between cost management and the external

<sup>&</sup>lt;sup>87</sup> For the discussion of these models within the GOGCWS see section 8.7.

In order to understand and explain the dynamic relationship between cost management and external environment, a case study method has been adopted (see chapter 6). The data is collected from multiple sources (triangulation) including company documents and questionnaires as a basis of the semi-structured interviews.

# Chapter 6 Methodology and method

# **6.0 Introduction**

The aim of this chapter is to outline the research design and methodology, and methods of data collection. For this purpose, this chapter is divided into three main sections, - each further divided into three sub-sections.

Section one outlines the methodology that the research intends to utilise and the rationale for choosing it. Section two clarifies the method followed in this research and the rationale for using it. Section two is divided into three sub-sections. The first explains case study research, case study theory, case study (qualitative). It gives the rationale for using case studies in management accounting research and examines some accounting case studies. Sub-section two looks at the strengths, weaknesses, and problems of the case study method. It suggests ways to understand case study research (in the preparation, collection and assessment of evidence, the analysis of data, theory development, and report writing).

Sub-section three opens the debate between qualitative and quantitative research, and examines three methods of qualitative data collection. It explains why the researcher combined qualitative research interviews and questionnaires. It discusses the different types of research interview and the strengths and limitations of each. It then explores the advantages and disadvantages of qualitative research interviews. Next it explains the different forms of company documentation, its advantages, and why it is not commonly used organisational research. Finally it describes the method which will be used in the analysis of organisational documents (the hermeneutic method) and the rationale for choosing this method.

The final section outlines the main conclusions of the previous sections (and subsections) of this chapter.

# 6.1 Research design and methodology

Methodology refers to the methods and ways used to conduct research. A methodological position consists of three elements (epistemology, ontology and human nature, Burrel and Morgan, 1979: 1).

## 6.1.1 Epistemology

Epistemology is the nature and forms of knowledge. It defines what 'knowledge' is, and the relationship between our ideas about what is going on in the world and the nature of the world itself. Epistemology explains the assumptions about knowledge – (i.e. about how we might begin to understand the world and communicate this knowledge) (Burrel and Morgan, 1979). It is the study of criteria, by which we determine what constitutes 'valid' knowledge.

Epistemological debates are concerned with the appropriateness of procedures and methods in the process of knowledge acquisition. A central debate focuses on induction and deduction. The inductive/deductive debate arises from whether knowledge is acquired by observation, or through theory.

The inductive approach<sup>88</sup> assumes that true knowledge begins with observations, from which theory can then be inferred. It is a process which begins with a set of observations (i.e. descriptions), and moves on to develop theories of these observations (see figure 6.1).



Source: DeVaus, D. (1996) - "Survey in Social Research", p. 12.

<sup>&</sup>lt;sup>88</sup> Theory construction / Grounded theory / Post Factum Theory / Ex post facto theory.

The deductive approach (theory testing), begins by formulating tentative theories (hypothesis), and goes on to develop procedures in order to test these theories. The deductive approach is sometimes referred to as 'hypothetico- deductive' or the 'falsificationist' approach, or "the method of conjecture and refutation" (Blaikie, 1995: 143-144). The deductive approach begins with a question or a problem that needs to be understood or explained<sup>89</sup>. The first stage involves producing a possible answer to a question, or an explanation of the problem itself. The deductive approach begins with a theory. This theory then allows us to predict how things will be in the "real" world (see figure 6.2).

The literature review gave the researcher certain expectations, as to how institutional pressures work to influence the role of cost management in the Egyptian Water industry.

**Figure 6.2: Deductive approach** 



Source: DeVaus, D. (1996)- "Survey in Social Research", p. 12.

This study adopts a hybrid approach (focusing on the inductive method). The rationale for choosing a hybrid approach is because objectives of this research are to describe what is going on (i.e. what is the current cost structure for the provision of water in the Egyptian water industry), and to understand and explain why it is going on (i.e. what is the rationale of the water companies for their cost structure?). This therefore involves induction and deduction. DeVaus (1996: 11) argues that inductive and deductive approaches are not alternative ways of arriving at theories, rather they represent two stages with different starting points.

<sup>&</sup>lt;sup>89</sup> The main objectives of this research are to understand, describe and explain the role of cost management in the Egyptian Water Industry.

#### 6.1.2 Ontology

Ontology is the nature of reality and "being". Ontology refers to the claims or assumptions which social enquiry makes about the nature of social reality. This involves claims about what exists, what it looks like, what units make it up, and how these units interact with each other (Blaikie, 1995). Ontological assumptions are at the core of the phenomena (cost management) (Burrel and Morgan, 1979: 1).

The core debates in ontology relate to "nominalism" and "realism". Nominalism refers to assumptions that the social world (external to individual cognition) and made up of names, concepts and labels which are used to structure reality. The nominalist does not admit to there being any "real" structure to the world (Burrel and Morgan, 1979: 4). Nominalism assumes that situations and objects are not independent of prior knowledge and the perceiving mind. Researchers' from differ ideological backgrounds can arrive at *different* conclusions about the same thing.

Realism<sup>90</sup> claims that the social world, (external to individual cognition) is a 'real' world made up of hard, tangible and relatively immutable structures. Whether or not we label and perceive these structures. They still exist as empirical entities (Burrel and Morgan, 1979: 4). Realism assumes that situations and objects exist in their own right, (external and independent of the observer's mind). It assumes that the researcher's view of the world is unaffected by his/her beliefs and prior knowledge. Researcher therefore arrive at the *same* conclusions about a given problem in a similar situation.

This research has adopted a nomalist view. The rationale for choosing a nominal approach was largely due to the debate between Macintosh and Scapens<sup>91</sup>(1990) and Boland<sup>92</sup> (1993). Boland (1993: 126) claims that Macintosh and Scapens exaggerate the operation of management accounting systems when they describe it "as a monolithic set

<sup>&</sup>lt;sup>90</sup> Realism encompasses "metaphysical realism" and "epistemological realism". "Metaphysical realism" perceives reality existing independently of the cognitive structure of observers. Epistemological realism perceives reality as cognitively accessible to the observer. Much of realism encompasses both views, although some realists would claim that while reality does exist independently of our efforts to understand it, it is not cognitively accessible (Gill and Johnson, 1991: 166).

<sup>&</sup>lt;sup>91</sup> Macintosh and Scapens (1990) employ structuration theory to frame their analysis of the budget at the University of Wisconsin (as reported by Covaleski and Dirsmith , 1988).

<sup>&</sup>lt;sup>92</sup> Boland (1993) draws upon existing empirical work. He re-interprets Milne's (1980) study of MBA students, through the use of accounting reports and utilising structure theory.

of structuring priorities". Boland stress "the actor's creative open possibilities for making meaning (p. 140). Macintosh and Scapens discovered that the 'power of accounting' "signifies, dominates and legitimises action". While Boland, however, stresses the 'power' of the actor in interpreting accounting as an open text.

Jones and Dugdale (2001) claim that these differences in opinion relate to methodological differences<sup>93</sup>. Boland (1993) characterises Macintosh and Scapens's approach as "a view from a distance", in contrast to his own which is, "up close". Scapens and Macintosh (1996: 681) define their own approach as "institutional analysis", and Boland's as "analysis of strategic conduct". This study has adopted institutional analysis. Scapens and Macintosh (1996: 683) argue that "in institutional analysis, the knowledge, skill and awareness of actors are kept in suspension, while institutions are treated as chronologically reproduced rules and resources. In the analysis of strategic conduct, the institutional properties of interaction are assumed to be given methodologically. Attention is focused on the processes through which social activities are produced and reproduced.

## 6.1.3 Human nature

Human nature concerns assumptions about the relationship between human beings and their environment, (i.e., human life as the subject enquiry, Burrel and Morgan, 1979: 2). The issue of human nature is that of what model of man is reflected in any given social-scientific theory (Burrel and Morgan, 1979: 6). In other words, they select a position which sees human beings as either 'deterministic' or 'voluntaristic', when they act.

The deterministic view regards human beings and their activities as being "completely determined by the "environment" in which they are situated" (Burrel and Morgan, 1979: 6). The deterministic position assumes that human action can be understood as a direct response to external stimuli. It seeks to establish a "cause and effect" relationship between stimulus and action. The voluntaristic view assumes that human beings are free-willed, they act independently of external stimuli, and exercise choice (Burrel and Morgan, 1979: 6).

<sup>&</sup>lt;sup>93</sup> Researchers with different viewpoints and differing ideological backgrounds can come to different conclusions about the same subject.

Although accounting offers rules and guidelines which companies should follow, (especially public sector companies or public utilities which have specific laws – e.g. the GOGCWS, deterministic view). This study adopted a voluntaristic view.

Jones and Dugdale (2001) stress the importance of the empirical content of a study by comparing Macintosh and Scapens' (1990) study with that of Boland (1993). Macintosh and Scapens (1990) studied the University Wisconsin's relationship with the state of Wisconsin. Boland (1993) studied managers in organisations confronting a fictional case study. (Where no indication was given of organisational value, or meaning systems that might be the basis for sharing). Jones and Dugdale (2001) argue that if we are concerned with shared values and meanings constructed at the concrete, embedded level of the organisation, then it is studies such as that by Macintosh and Scapens that might reveal them. Conversely, if we are concerned with values and meanings constructed at the abstract dis-embedded level, we might then expect them to be discovered in studies such as that by Boland.

This study focused on shared values and meanings constructed at the concrete embedded level of the GOGCWS. Therefore, this study adopts a more 'voluntaristic' view of human behaviour than that which guides contingency<sup>94</sup> and neo-institutional<sup>95</sup> research. This approach can be compared with more recent advances in institutional theory, which calls for more in-depth, interpretative analyses of dialectic between stability and change (Czarniawska and Sevon, 1996).

## 6.2 Research method

#### 6.2.1 Case study

#### 6.2.1.1 Case study and the rationale for using it

The argument about methods depends on two factors. (1) The relationship between theory and method (2) how the researcher attends to the potential weaknesses of the method (Hartley, 1995).

The first of these factors (the relationship between theory and method) is discussed initially. In the 1990s there was significant interest in both the practice and theory of

<sup>&</sup>lt;sup>94</sup> See Child, 1972; Schreyogg, 1980.

<sup>&</sup>lt;sup>95</sup> See DiMaggio, 1988; Oliver, 1991.

cost management and management accounting. Many researchers were interested in studying practice to document and understand the many roles (economic, political and social) that management accounting systems can play in the functioning of organisations (Lowe and Machin, 1983; Berry et al., 1985; Chua et al., 1988). A major objective of these studies was to examine organisational practice for the development of a sociology of accounting practice (Scapens, 1988). Many of these case studies (of activity-based accounting systems) explored the issue of how systems could be used to assist managers in formulating and implementing strategy (Spicer, 1992).

One reason for studying accounting practice (the case study approach) is to develop a better understanding of the role a company's management accounting, control system and cost management can play in the formulation and implementation of strategy. Several factors distinguish case study research from other empirical research methods. These are:

(a) A case study is one of a number of different research methods that can be applied to cost management and management accounting research

(b) Case studies involve an empirical inquiry where multiple sources of evidence may be used (e.g. questionnaires, interviews, document analysis and observation)

(c) Case study research design can focus on a single organisation, (i.e. the GOGCWS, Yin, 1989)

(d) Case study research is likely to be valuable where existing theories are inadequate or incomplete

Case studies can give an analysis of the context and processes, of the phenomena under study - (usually involving inductive analysis, data collected over a period of time, and of one or one or more organisations or groups (Hartley, 1995). Case study is a research design which focuses on understanding the dynamic relationships present within a single setting (Eisenhardt, 1989). Case studies must be constructed to be sensitive to the context in which management behaviour takes place. A case study approach is not a method, but rather a research strategy. As Yin (1981) notes, "because the context is deliberately part of the design, there will always be many 'variables' for observation made". A case study (defined in terms of its theoretical orientation) gives an understanding of processes alongside their contexts (Hartley, 1995). Scapens, 1990 and Ryan, et. al., 1992 note that the term "fieldwork" and "case study" have been used to refer to studies of accounting in its actual setting. Fieldwork means studies of social practice in the 'field of activity' in which they take place. This can be a study of a single company, or a number of companies. A case study implies a single unit of analysis. This might be a company or a more aggregated unit of analysis.

Case studies need to develop theoretical frameworks in order to inform the data, provide a sense of general relevance and interest, and offer wider meaning than mere description. Case studies have been widely used to study culture and change in organisations. Case studies are distinguished by their approach to theory-building, which is usually (but not exclusively) inductive. The opportunity to explore issues in depth, (and in their actual context), means that theory development can occur through the systematic piecing together of detailed evidence to generate (or replicate) theories of general interest. Yin (1981) suggests that the case study researcher is someone akin to that of a detective who must sift evidence in order to "build a picture of motive, opportunity and method". In case study research, this 'detective' work is undertaken, not only to understand the unique features of the case, but also to draw out an analysis which is widely applicable.

Within a broader strategy, several methods may be used. These may be either quantitative, qualitative or both. The emphasis is generally more on qualitative methods, (because of the kinds of questions which are best addressed through the case study method). Participant observation is an alternative research method (Burawoy, 1979). Interviews are another research technique. These may range from semi-structured to unstructured (King, 1995). A case study may involve the use of questionnaires (with members of management or the workforce), in addition to observations and interviews.

The distinction between interviews (carried out as part of a case study) and interviews (carried out as part of a survey), is a matter of degree. As part of a case study, interviews are used to explore and probe in-depth, the particular circumstances of the organisation. They can also examine the relation between organisational behaviour and its specific context. In surveys, the emphasis is likely to be on comparisons of the organisation.

#### 6.2.1.2 Case study and accounting

Case studies have become increasingly popular in management accounting research. Little has been written on the methods of case study research in accounting, (although there is a growing literature on case study methods in other areas of social sciences).

What constitutes case study research? How do we evaluate a management accounting case study? When are case study methods more appropriate than other research methods? Otley and Berry, (1994; 1998) argue that one reason for adopting a case-based approach (in the study of accounting control), is because the accounting role and other controls cannot be fully understood in isolation. A more contextual approach is required. Case studies can be used in a variety of ways in accounting research (Scapens, 1990; Ryan, et. al., 1992; and Otley and Berry, 1994; 1998) as follows:

(a) Descriptive case studies. These are case studies which describe accounting systems techniques and procedures currently used in practice. The research objective is to describe methods of accounting practice. Descriptive case studies may be useful in determining the extent of the 'gap' between accounting theory and practice (i.e. what constitutes 'best' practice and "successful" companies)?

(b) Illustrative case studies. These are case studies which utilise the innovative practices developed by different companies and organisations (Ryan et al., 1992). Kaplan (1984) claims that accounting researchers have much to learn from studying innovative companies. These case studies are an illustration of what has been achieved in practice. Illustrative case study research begins with a clear theoretical standpoint. It then takes a 'real-world' situation and interprets it (in light of the theoretical standpoint adopted). Theories are used to explain observations (Otley and Berry, 1994).

(c) Experimental case studies. These are case studies, which examine the difficulties involved in implementing new proposals. Accounting researchers have developed new accounting procedures and techniques, - which intended to be helpful to accounting practitioners. These procedures and techniques have developed from existing theoretical perspectives.

(*d*) *Exploratory case studies*. These are case studies which explore the reasons for certain accounting practices. They enable the researcher to generate hypotheses about the reasons for particular practices (Ryan, et al., 1992). The research objective is to produce generalisations about accounting practices (Scapens, 1990). An exploratory case goes beyond mere description<sup>96</sup> towards 'explanation'. Research is a process between theory (explanation) and data (description) (Otley and Berry, 1994; 1998).

(e) Explanatory case studies. These are case studies which explain the reasons for observed accounting practices. The focus of the research is on a specific case. Theory is used in order to understand and explain specifics, rather than offer generalisations. The research objective is to generate theories which offer valid explanations.

(f) Critical case. A critical case can be used to disprove a theory by generating a set of phenomena which is inconsistent with established theories.

(g) Accidental case. An accidental case arises by accident, because "access" to the company is not pre-planned. It also restricts a number of aspects of organisational life.

Scapens (1990) and Ryan, et. al., (1992) found that these case studies have many areas of overlap. For example, the distinction between exploration and explanation, is rather ambiguous. An exploratory study can generate initial ideas, which will then form the basis of a theory of accounting practice. The use of case study research methods will depend on the nature of the research and the methodology of the researcher.

The researcher pursued three objectives in carrying out this case study research (descriptive, illustrative and explanatory). The rationale for choosing three objectives was to assist the researcher in his attempt to answer the research questions include:

- What is the current cost structure for the provision of Egyptian water services?
- What methods of cost accounting were and are used in the Egyptian water industry? (descriptive and illustrative)
- What is company rationale for choice of cost structure? (explanatory)
- What is the role and function of accounting in controlling the water industry? (descriptive and explanatory).

<sup>&</sup>lt;sup>96</sup> It is difficult to distinguish 'descriptive' case studies from 'exploratory' case studies.

It was Spicer (1992) who classified case studies into (a) Descriptive/explanatory and (b) Informing/explanatory. Group (a) descriptive and/or explanatory, case studies seek to describe/explore reasons for particular accounting practice(s). Group (b) informing and/or explanatory are case studies used indirectly to help inform non-case empirical research, or used directly to explain observed practice.

The primary objective of this research was to "understand, describe and explain the role of cost management in the Egyptian water industry". A group (a) case study was therefore considered most useful for this research. Scapens (1990) argues that the distinction between exploration and explanation is ambiguous. The rationale for choosing a descriptive/explanatory (exploratory) case study was because knowledge of cost management is insufficient to allow the development of good theoretical statements. In other words, the aim of this research was to develop theories for practice, which may then assist managers with the design of cost management systems in future.

Several authors have discussed the criteria with which to judge case-study, (e.g. Yin<sup>97</sup>, 1981; 1989; Kazdin, 1982; Bruns and Kaplan<sup>98</sup>, 1987; Eisbenhardt, 1989; Young and Selto, 1990; Atkinson, 1990; Scapens, 1990; Spicer, 1992). Perhaps the most significant contribution was that made by Scapens (1990). The case methods, (when used for explanatory purposes), relied on *analytical* not *statistical* generalisations. Therefore this research does not draw on a larger population (based on sample evidence), but rather generalises back to theory.

Case study research which describes, explains and explores cost management, can be grouped into three categories, according to the objectives of cost management (efficiency, optimisation and strategy<sup>99</sup>):

<sup>&</sup>lt;sup>97</sup> Yin explored case studies general.

<sup>&</sup>lt;sup>98</sup> Bruns and Kaplan examined case studies in accounting.

<sup>&</sup>lt;sup>99</sup> The three objectives of cost management (efficiency, optimisation and strategy) was adopted from Al-Hazami (1995). The models of cost management relating to each objective, generates originally from this thesis.

(1) Research which studies throughput costing, benchmarking, target costing, cost leadership, full absorption cost and activity based costing, all aim to achieve efficiency

(2) Research which studies activity-based management, re-engineering and economic costs focus on optimisation

(3) Related research which studies Strategic management accounting, focuses on strategy. This research study incorporated all three objectives of cost management.

## 6.2.1.3 Understanding case study research

Researchers have identified the strengths and weaknesses of the case study method (see Miles, 1979). Since the 1970s, an increasingly sizeable literature has emerged on how to conduct case-study research and analyse its data (Campbell, 1975; Eisenhardt, 1989; Gummesson, 1991; McClintock, et al., 1979; Rose, 1991; Yin, 1981; 1989; Scapens, 1990).

A strength of the case study method lies in its capacity to explore social processes as they unfold in organisations. By using multiple (often qualitative methods, e.g. observation), a researcher can learn much more about processes than is possible with other techniques (e.g. surveys). Social studies are also useful in understanding social processes in their organisational and environmental context. Behaviour is only understandable in the context of wider forces operating within the organisation, (whether these are contemporary or historical, Hartley, 1995).

Case studies have an important function in generating hypotheses and theory-building. They can generate new theory and the emergent theory can be tested with constructs that can be measured, and hypotheses that can be disproved (Eisenhardt, 1989). Case study research provides a 'vehicle' by which theories can be modified from newly generated. Case study research is valuable when existing theories are inadequate or incomplete. Case studies can explain a sub-set of the topic under investigation. Case studies provide a more holistic approach.

Case studies can be useful in 'capturing' the intrinsic properties of organisational life. A case study is a useful technique where an insight into organisational life is required (especially if this is informal, unusual or even clandestine). In detailed case studies,

what is required is an understanding of what certain (organisational) concepts and behaviours mean.

It is useful at this point to give an overview of the problems of the case study approach. Several researchers have highlighted problems with the case study approach, and what future researchers should do to avoid potential problems (Goode and Hatt, 1952; Yin 1989; Scapens, 1990; Ryan et al., 1992; and Otley and Berry, 1994). These problems are summarised into five areas:

(a) Wholeness. Ryan et al. (1992) argue there is a difficulty of drawing boundaries around the subject matter (the role of cost management in the water industry) of the case. Case-study researchers must place certain limits on the on the area of study, (and make those limits quite explicit). Another possibility is to try to study every thing in 'superficial' way. Goode and Hatt (1952) posed the question "how can the wholeness of cases be preserved?" They identify four points: (a) the breadth of data<sup>100</sup> (b) levels of data<sup>101</sup> (c) formation of indexes and types<sup>102</sup> and (d) interaction in a time dimension<sup>103</sup>

(b) 'Certainty of the researcher' and the failure to test reliability. A weakness in the case study method is the *response of the researcher*<sup>104</sup>. The researcher may feel a false sense of certainty about his/her conclusions.

<sup>&</sup>lt;sup>100</sup> The collection of a large amount of data allows a unit to be studied as a 'totality'.

<sup>&</sup>lt;sup>101</sup> The case-study method is distinguished by its use of data from abstract levels (other than purely sociological). Sociologists incorporate economic, political and psychodynamic data into their analyses, yet social research has (in the main), been moving towards a clearer definition of 'the sociological', It advocates better researchers techniques in order to abstract it from the complexity of other relationships.

<sup>&</sup>lt;sup>102</sup> An important technique in preserving the 'wholeness' of the unit is the development of typologies and indexes. This technique is not confined to the case-study method and is used in other qualitative analysis.

<sup>&</sup>lt;sup>103</sup> The case-study method also relies heavily on process and time.

<sup>&</sup>lt;sup>104</sup> The 'problem' is created by the observer himself. Consequences of 'this feeling of certainty' are many, but most of them can be grouped under the heading 'a temptation to ignore basic principles of research design'. The researcher feels confident in his own specialist subject area and therefore feels no need to check the overall design of proof. The researcher has a strong conviction that his selection is "representative". He did not follow an established mode of sampling and therefore is certain there was no bias present in the study. Researchers are not categorised into different "types". An investigator who absorbs facts from a large number of cases will believe he has a statistical 'factory' sample, (no matter how knowledgeable he is about sampling design).

When an investigator feels "at home" with a given case, s/he may be tempted to "feel" the right explanation even though other researchers may arrive at an entirely different conclusion. This is what is known as *ad hoc* theorising. 'The feeling of certainty' a researcher has, might lead him/her to fail to test reliability of the data, the classifications used, or the analysis of the data.

(c) Costs (time and money). It is clear that the case-study approach is a costly one in terms of both time and money. Goode and Hatt (1952) argue that each case becomes a research project in itself. Therefore, even several adequately-documented cases may easily consume two years of a researcher's time. Therefore, for certain types of analysis, the use of case study research, can prove fruitful. Other types of analysis may involve a simple polling study, or mailed questionnaires. However, the analysis of detailed processes of social interaction requires some preliminary resource to case study (even where there is not complete reliance upon the case-study method).

A case-study investigator may inadvertently allow evidence or biased views to influence the direction of his findings and conclusions. As a result, each case-study investigator must work hard to avoid such a situation. Yin (1989) supports this view and claims that researcher-bias may influence the findings (Rosenthal, 1966). This problem may also emerge with other research strategies, (such as designing questionnaires for surveys, Sudman and Bradburn, 1982), or conducing historical research (Gottschalk, 1968). Yin (1989) claims that case studies are too lengthy and result in overly-complex documents. This is a valid criticism but a well-planned case study need not prove problematic.

(d) Generalisation. Case studies provide very little basis for scientific generalisations. A frequently-heard question is "How can you generalise from a single case"? Yin (1989) claims that theoretical generalisations are possible from case studies but cannot be applied to wider populations.

(e) Skills involved in effective case study research. These skills have not yet been defined. Case studies are difficult to carry out effectively. We have little in the way of screening or testing for an investigator's ability to carry out effective case study research. Rules for carrying out effective case studies have not been established. As a result, "most people feel they can carry out case-study research, and nearly all of us

believe we can understand them. Neither view is well-founded, and the case studies generally receive a great deal of credit they do not necessarily deserve" (Hoaglin et al., 1982: 134). Otley and Berry (1994) argue that case studies (in the accounting and control tradition), have been "varied" in their focus. Otley and Berry (1994) claim there is currently not much theory-building or cumulation of knowledge. A major criticism of accounting and control research case studies is that researchers adopt invariably one epistemological stance, and ignore the nostrums of another.

There are certain steps a researcher should follow in order to understand case study research. These were suggested by Scapens, 1990; Ryan, et al., 1992 and Hartley, 1995. (For an explanation of these steps, see figure 6.3).



Figure 6.3: Understanding case study research

Source: adapted from Hartley (1995); Scapens (1990); and Ryan. et. al. (1992).

#### Preparation

At the preparation stage, a researcher should review a number of available theories. This will determine the way in which the researcher examines a case. Preparation can involve new theories. It is useful to divide the preparation stage into three sub-stages as follows:

*Choosing a case study company (s).* Single case studies can yield valuable information about the topic under investigation. A research study may become more valid by company two or more cases, or indeed even carrying out a comparative study within the case organisation itself (e.g. perceptions of employees in different departments).

This research was based on one case study on the General Organisation for Greater Cairo Water Supply (the GOGCWS). Reasons for choosing this company were several: (1) It was the first company established in the Egyptian water industry. It could therefore help the researcher achieve his aim of describing and explaining the role of cost management in the Egyptian Water industry. The GOGCWS represents "the total experience of Egyptian Water"

(2) The GOGCWS provides water for 25% of the population in Egypt. It has a unique<sup>105</sup> laboratory - ideal for analysing water in the Middle East (The cost of this laboratory was 25 million Egyptian pounds<sup>106</sup>)

(3) There are discrepancies between the expenses of the GOGCWS and its income. The cost per  $m^3 = 0.46$  pence (wages 8.03 + Power and electricity 6.99 + raw materials 4.47 + other current expenses 7.00 + interest "local and international" 11.29 + Depreciation 8.19)<sup>107</sup>. By contrast, average income was per  $m^3 = 0.16$  pence, so this price does not encourage efficiency

<sup>&</sup>lt;sup>105</sup> The GOGCWS, (no date source given, c) - "Greater Cairo Water", The Administration of Public Relations.

<sup>&</sup>lt;sup>106</sup> N.B: English pound  $\pounds = 6.00$  Egyptian pounds. American Dollar \$ = 5.00 Egyptian pounds.

<sup>&</sup>lt;sup>107</sup> The GOGCWS (no date source given, c) - "Greater Cairo Water", The Administration of Public Relations.

(4) It is estimated that the General Organisation for the Greater Cairo Water Supply loses 40% of water through leakages each year. This results in an estimated wasteful expenditure of 250 million Egyptian pounds every year. Efficiency in the company is therefore low

*Gaining and maintaining access.* Gaining access to an organisation is complex. Maintaining this access requires continual attention. Access is affected by changes in personnel or the fortunes of the company. It is also influenced by how much organisational time the researcher is taking and the perceived contribution of the research. "Reporting back" discussions with the principal sponsor in the organisation can be useful (Buchanan, et al., 1988).

This research aimed to explain the role of cost management in the Egyptian Water Industry. The General Organisation for the Greater Cairo Water Supply (GOGCWS) was chosen as a case study. Egypt has a bureaucratic state. Government bureaucrats often create unnecessary red-tape. As Cook (1998) found his research in Egypt, some topics have 'universal' sensitivity in developing countries. These include military matters, political power structures, government decision-making processes and religious and social problems which reflect the national image. The water industry is a good example of the latter. In March 2000 the researcher prepared his preliminary research for access into the GOGCWS and in June 2000 was granted access into the company. The Fieldwork place as outlined below:

On 13 March 2000, my research supervisor wrote to the Chairman of the Educational and Cultural Bureau of Egypt (my sponsor) to ask his assistance in gaining access to the GOGCWS (for no later than May 2000, see section A1.1). On 20 April 2000, my supervisor sent my fieldwork plan to the Chairman of the Educational and Cultural Bureau of Egypt (see section A1.1). The Chairman granted permission for my fieldwork. On 9 June 2000, I arrived in Cairo and the following day I made my way to the central government building in downtown Cairo the "Mugamma". I was advised there to go to the Central Institution for General Statistical (the CIGS). On 12 June, I went to the CIGS and filled in a form requesting permission to carry out interviews and questionnaires. They asked me to return in seven days. The following week, I returned to the CIGS, only to be told to return a week later. I made it clear then that I only had a

short time in Egypt and asked them to make sure my papers were ready the following week. The next week I return to the CIGS. They informed that a decision had been made and would be sent to the Mission department in Mugamma. I requested permission to take the documents myself to the Mission department.

On 21 June 2000, I took the decision papers and a covering letter and the following day I took these documents (see section A1.1) to the Mission department. They asked me to return in 48 hours. On 24 June I received a letter from the Mission department addressed to the Chairman of the General Organisation for the Greater Cairo Water Supply (the GOGCWS). On the same day I went to see the Chairman of the GOGCWS who asked me to return the next day. On 25 June, I returned to the GOGCWS who passed the letter to the Vice Chairman of Technical Affairs, who in turn transferred it to the Vice Chairman of Financial and Administrative Affairs. From that day I had access to the GOGCWS.

*Choosing initial framework.* Literature is used to focus the study to avoid producing a narrative, rather than theory-based study. In order to avoid being overwhelmed by data, theory is used to examine, support or contest.

# **Collecting evidence**

The researcher should allow issues and theories to emerge out of the case, rather than imposing their own view(s) on it. Typical sources of evidence include interviews, documentation, direct observation and participant observation. In most cases it is important to use multiple sources of evidence. This might include two elements: collecting data and managing data.

*Collecting data.* The data should give the researcher an overview of structure and the functioning of the organisation. It should give information as to which people to talk to, what to observe, research methods, and systematic questioning (not ad hoc data collection). The GOGCWS research utilised company documents, questionnaires and interviews. The questionnaire<sup>108</sup> was used as the basis for interviews<sup>109</sup>. The researcher

<sup>&</sup>lt;sup>108</sup> For employee position (name and job of people who filled in the questionnaire, see table A2.1).

 $<sup>^{109}</sup>$  The people who were interviewed were the same people who filled in the questionnaire (see table A2.1) except Nos. 20, 38, 39, 41- 46, 48-50.

spent time with each respondent, clarifying the questionnaire in order to make it easier and understandable. If a respondent became tired, they were asked to complete it another time. Most questionnaires were completed by the third interview. The interviews were semi-structured because I was 'open' to any new issues (such as privatisation in the Egyptian water industry). In addition, if any new issues were raised, I would ask all respondents about it.

*Managing data.* The researcher should observe and record data about the organisation and its people. Glaser and Strauss (1967) make a distinction between observation notes, method notes and theoretical notes.

# Assessing evidence

The validity of each piece of evidence should be assessed by comparing it with other evidence on the same issue (triangulation). In assessing evidence there are two crucial issues for the case researcher to resolve: (1) The "rules of evidence" which should be applied to the case concerned and (2) The "standard of proof" which is sought at each stage of the analysis. In order to assess validity, this research used questionnaires, interviews and documentation analysis. The questionnaire had different types of questions to measure a single idea. These questions checked the validity and reliability of respondents. The researcher compared the information obtained from questionnaires, with the information from interviews and documentation.

#### Analysing data

Data collection and analysis is an iterative process which allows theory to develop, grounded in empirical evidence. Yin, (1989) argued that to avoid the danger of 'premature closure' - what is required is careful description of data and the development of categories (i.e. key themes, topics and questions). It is sometimes helpful to prepare models (diagrams, flow charts, etc.) which attempt to link the various themes and issues. In the light of these suggestions, the researcher began his own analysis of data by devoting a chapter to the current accounting system of the GOGCWS. The subsequent chapters dealt with the analysis of data.
#### **Theory development**

The researcher should compare existing literature with expert opinion. The researcher should ascertain that the case is an accurate rendition of facts, producing plausible explanations and conclusions based on best explanation.

# **Report writing**

The report should draw out theoretical implications, which will carry over other case studies. Many researchers experience a common problem. How should a researcher select a suitable case study for his/her research? Ryan, et al., (1992) illustrate five different uses of case study research, namely experiment case<sup>110</sup>, critical case<sup>111</sup>, extreme case<sup>112</sup>, exploratory case<sup>113</sup>, and multiple cases<sup>114</sup>. There are no universal rules for selecting a suitable case study but the selection depends on the research question (s). How does the case study help the researcher answer his/her research question (s)?

### 6.2.2 Qualitative approach

## 6.2.2.1 The Debate between qualitative and quantitative research

Bryman (1988) draws out some differences between quantitative and qualitative research. These differences can be summarised as follows:

## (a) The role of qualitative research

Quantitative researchers rarely deny the usefulness of qualitative research, but have tended to view it as an essentially exploratory way of conducting social investigations. Consequently, they have typically seen it as useful at the preparatory stage of a research

<sup>&</sup>lt;sup>110</sup> The 'experiment case' relies on theoretical generalisations. Hence, the issues involved in selecting cases, should be similar to those considered in selecting topics for experiment, (rather than those used for selecting a statistical sample).

<sup>&</sup>lt;sup>111</sup> A 'Critical case' is used where there is a well formulated theory and the major research issues are clearly defined. The objective of such a study would be to determine that the theory provides good explanations, otherwise alternative theories need to be developed.

<sup>&</sup>lt;sup>112</sup> An 'Extreme case' is used where the researcher wants to extend a theory to cover a wide(r) range of circumstances.

<sup>&</sup>lt;sup>113</sup> An 'Exploratory case' could be used to begin the process of theory development. The selection of a particular case for study, is relatively unimportant. What is needed is a case within the relevant area, which will then enable the researcher to begin the process of theory development.

<sup>&</sup>lt;sup>114</sup> 'Multiple cases' can be used for two purposes. The objective of multiple cases is to develop a rich theoretical framework, capable of explaining a wide range of circumstances.

project. Qualitative research is often depicted as useful as a means of hunches and hypotheses which can be tested more rigorously by quantitative research (Bryman, 1988). This research study has combined<sup>115</sup> methods of quantitative research (questionnaires), and qualitative research (interviews).

## (b) Relationship between researcher and subject

In quantitative research, the researcher's contact with the people under study is fairly fleeting, or even non-existent. This was not the case with the GOGCWS research study because an example amount of time was spent with each respondent who completed the questionnaire. By contrast, qualitative research requires much more sustained contact, especially when participant observation is the central method.

## (c) The researcher's stance in relation to the subject

Quantitative researchers adopt the role of an outsider looking in on the social world. He/she applies a pre-ordinate framework on the subjects being investigated and are involved as little as possible in that world. This was not the case with the GOGCWS research study. Among qualitative researchers there is strong urge to "get close" to the subject being investigated. For qualitative researchers, it is only by getting close to their subjects and becoming an "insider", that they can view the world as a participant in that setting. Adler (1985: 11) in his research, found that "the only way I could get close enough to [upper-level drug dealers and smugglers] to discover what they were doing and understand the world from their perspectives was to take a membership role in the setting".

### (d) Relationship between theory/concepts and research

The model of quantitative research, is that theories and concepts are the starting point for investigations carried out within its framework. However the extent to which quantitative research is explicitly guided by theory has been questioned by many commentators. Some argue that theoretical reasoning often occurs towards the *end* of the research process (Cicourel, 1982). Qualitative researchers often reject the idea of using theory as a pre-cursor to investigation (except perhaps as a means of providing an initial orientation to the situation as in "ground theory"), since it may not reflect subjects' views about what is going on and what is important. Consequently, as one

<sup>&</sup>lt;sup>115</sup> Questionnaires are used as a basis for interviews.

advocate of qualitative research has put it: "It is marked by a concern with the discovery of theory rather than the verification of theory" (Filstead, 1979, p. 38). The GOGCWS research had an uncompleted theoretical framework before the fieldwork began. The framework was completed after the fieldwork. Hence, the researcher deemed a combination of quantitative and qualitative research as most appropriate.

## (e) Research strategy.

Quantitative research tends to adopt a structured approach to the study of society. To a large extent, this tendency is a product of the methods with which it is associated-both surveys and experiments. By contrast, qualitative research tends to be more open. Many ethnographers advocate the delineation of a research focus be deferred as long as possible (Cohen, 1978). Consequently, many qualitative researchers feel overwhelmed during their early days in the field, since everything they observe is potentially "data". Whyte (1984) for example, sees ethnographic research as deriving much of its strength from its flexibility, which allows new leads to be followed up or additional data to be gathered in response to changes in ideas. But he also notes a limitation of such flexibility since "you may find so many interesting things to study that you are at a loss to delimit the scope of your project and focus on specific problems' (p. 225). For the purpose of the research and the method used in this study (a combination between quantitative and qualitative), the questionnaire had several open-ended questions and was used as a basis for the semi-structured interviews.

# (f) Scope of findings.

It is common to conceive of the quantitative/qualitative dichotomy in terms of respective commitments to nomothetic and ideographic modes of reasoning (Halfpenny, 1979). This distinction effectively refers to the scope of the findings, which derive from a piece of research. A nomothetic approach seeks to establish general law-like findings, which are deemed to hold irrespective of time and place. An ideographic approach locates its findings in specific time-periods and locations. The findings of this research were done in a specific time-period, and in one location (the GOGCWS) because the researcher focused on only one case study of an Egyptian water company. This means the research followed mainly a nomothetic view. However, the findings can

generalise<sup>110</sup> across the Egyptian water industry, because there is only one law and one accounting system applied in all water companies in Egypt.

Quantitative research exhibits a nomothetic approach because of the investigator's ability to extrapolate findings to larger populations. By contrast, the qualitative researcher frequently conducts research in a specific milieu (a case study), whose representatives are unknown, so the potential for generalisation is also unknown.

### (g) Image of social reality.

Quantitative research conveys a view of social reality which is static, (in that it does not consider the impact and role of change in social life). Qualitative research examines connections between variables. The proponents of qualitative research argue that quantitative research rarely examines the processes which link to qualitative research (Blumer, 1956; 1969). They also claim that the independent and dependent variables fail to take into account the flow of events in which these variables are located. The qualitative researcher is in a better position to view the connections between events and activities, and to explore people's interpretations of the factors which produce such connections. This stance affords qualitative researchers a much greater opportunity to study processes in social life.

### (h) Nature of data.

The data arising from quantitative studies is often depicted as hard, rigorous, and reliable. This suggests that such data exhibits considerable precision, is collected systematically and may be checked by another investigator. These factors mean that quantitative data is more persuasive, and hence more likely to gain the support of policy makers.

Qualitative researchers routinely describe the data derived from ethnographic work as "rich" and "deep", thereby drawing a contrast with quantitative data (which tends to be depicted as superficial). The adjective "rich" suggests an attention to (intricate) detail which many qualitative researchers often provide. For many qualitative researchers, quantitative research produces superficial data. For example, qualitative researches see

<sup>&</sup>lt;sup>116</sup> The objective of the study is not to provide generalisations or explanations at abstract levels. Rather it is to provide theoretical explanations of observations in a particular rather than a general setting.

survey research as merely a source of 'surface' information, which relates to abstract categories. By contrast, the quantitative researcher may consider a limited ample research studied in one location as somewhat lacking (Liebow, 1967). The data collected may be heavily influenced by the particular emphases and predisposition of the researcher.

For a summary of the above arguments see table 6.1

	Quantitative.	Qualitative.
1- Role of qualitative research	Preparatory	Means to exploration of
		actors' interpretation
2- Relationship between	Distant	Close
researcher and subject		
3- Researcher's stance in	Outsider	Insider
relation to subject		
4- Relationship between	Confirmation	Emergent
theory/concepts and research		
5- Research strategy	Structured.	Unstructured
6- Scope of findings	Nomothetic	Ideographic
7- Image of social reality	Static and external to actor	Processual and socially
		constructed by actor
8- Nature of data	Hard, reliable	Rich, deep

#### Table 6.1: Differences between quantitative and qualitative research

Source: Bryman, (1988)- "Quantity and Quality in Social Research", p. 94.

Both qualitative and quantitative research involves the development of methodological skills, e.g. (gaining access, conducting different types of interview, different approaches to participant observation, work histories, case studies, and ethnographic research). However qualitative research is more than a series of methodological skills. It involves reflection on such skills, understanding different approaches to analysis, dealing with issues of generalisation and specificity, and an engagement with theory and method.

Confusion only arises with qualitative research over the way such data must be handled, and the use to which such data can be put. This confusion is caused by a failure to differentiate between several orientations in qualitative data. The difficulty of qualitative research methods is how to treat the data in an appropriate way, and how to choose methodological issues.

Within the positivistic paradigm, qualitative researchers tend to look for cause and effect on the assumption there is a 'real' world which has determinate characteristics. These theories will show how certain variables interrelate - especially how they relate to each other in a cause and effect fashion. Some writers disagree with this view of positivism. The idea is that of an underlying unequivocal reality, consisting of a web of relationships between specifiable variables, and scientific theories are testable models of that reality. It is true that positivism favours quantitative methods, but it is incorrectly thought that qualitative research data cannot fall within a positivist philosophy of science. Ashworth (1997) considers "the ground theory" of Glaser Strauss and colleagues (Strauss and Corbin 1990), as having many of the hallmarks of positivism, despite the fact that qualitative techniques are seen as appropriate to the whole research process, not merely to the early "discovery" phase.

It is clear from the above argument that although qualitative research has several advantages, it also has disadvantages. In order to limit the effect of these disadvantages, the researcher combined both qualitative research (interviews) and quantitative research (questionnaire).

## 6.2.2.2 Qualitative research interview

The most widely used qualitative method in organisational research is the interview method. It is easy to see why. It is a highly flexible method, it can be used almost anywhere and is capable of producing data of great depth (King, 1995). Above all, it is a method with which most research participants feel comfortable. When a researcher tells interviewees "I would like to interview you about..." most people have a reasonable idea of what to expect. This is not necessarily the case with other qualitative methods, such as participant observation or the Twenty Statement test.

King (1995) argues that qualitative research interviews have "depth", are "exploratory", semi-structured, or unstructured. Kvale (1983: 174) defines the qualitative research interview as "an interview, whose purpose is to gather descriptions of the meaning of the described phenomena". The goal of any qualitative research interview is therefore to

see the research topic from the perspective of the interviewee, and to understand how and why he or she comes to have this particular perspective. Hence, what is required for any interview is a low degree of structure imposed by the interviewer; open questions, and a focus on "specific situations and action sequences in the world of the interviewee", rather than abstractions and general opinions (Kvale, 1983: 1976).

There are three kinds of interview: qualitative research interview, quantitative research interview<sup>117</sup>, and structured open-response interviews<sup>118</sup> (King, 1995).

Qualitative research interviews vary in their focus - focusing on an individual's whole life-world. Fetterman (1989) looks at particular topics and how they are perceived and understood by interviewees. Though never highly structured, qualitative research interviews also vary in the degree of structured imposed.

The research used semi-structured interviews (structured open-response interviews). The rationale for choosing this type of interview is that it lies somewhere between the qualitative research interview and the structured interview, (in terms of its degree of imposed structure and the balance of open and closed questions). King (1995) refers to them as "structured open-response interviews". They tend to focus on factual information and general evaluative comments, without exploring deeper layers of meaning. King (1995) argues that the structured open-response interview is most

<sup>&</sup>lt;sup>117</sup> The antithesis to the qualitative research interview is the structured interview. In this, the interviewer uses detailed a schedule with questions asked in specific order. Questions are mostly closed, and will use numerical rating scales, and/or tick- boxes. A small number of open-ended questions may be used to allow the interviewee to expand upon particular points or make general comments about the research topic. The interview will not normally taped; instead the interviewer will record responses by hand. The qualitative research interview is most appropriate when: (a) a study focuses on the meaning of a particular phenomena to participants (b) to study individual perceptions of processes (within a social unit are to be studied prospectively, using a series of interviews) (c) where individual historical accounts are required of how a particular phenomena developed - (for instance, a new shift system) (d) where exploratory work is required before a quantitative study can be carried out (e) where a quantitative study has been carried out and qualitative data is required to validate particular measures or to clarify and illustrate the meaning of the findings.

<sup>&</sup>lt;sup>118</sup> This type of research interview lies somewhere between the qualitative research interview and the structured interview in terms of its degree of imposed structure and balance of open and closed questions. King (1995) refers to them as "structured open-response interviews". They tend to focus on factual information, and general evaluative comments without exploring deeper layers of meaning. The structured open-response interview is most appropriate: (a) where a quick, descriptive account of a topic is required, without formal hypothesis-testing (b) where factual information is to be collected, but there is uncertainty about what, and how much information participants will be able to provide (c) where the nature and range of participants' likely opinions about the research topic are not well known in advance, and cannot easily be quantified.

appropriate (a) where a quick, descriptive account of a topic is required, without formal hypothesis- testing (b) where factual information is to be collected, but there is uncertainty about what and how much information participants will be able to provide (c) where the nature and range of participant opinion about the research topic is not well known in advance and cannot easily be quantified

Qualitative research interviews have many advantages. For example, different type of qualitative research interview can be used to tackle different types of research questions in organisations, making them one of the most flexible methods available. They can address quite focused questions about aspects of organisational life, (for instance, specific decision processes such as selection decisions, or decisions about innovation adoption). Similarly, they can focus on experiences of a particular training or development programme, perhaps as part of a wider assessment process. At the other end of the scale, qualitative research interviews can be used to examine much broader issues, in such areas as gender, organisational culture and the effects of unemployment.

The key feature between the quantitative and qualitative research interview is the nature of the relationship between interviewer and interviewee. In a quantitative study using structured interviews, the interviewee is seen as a research "subject", in much the same way as he or she would if completing a questionnaire or taking part in an experiment. The researcher's concern is to obtain accurate information from the interviewee, untainted by relationship factors. The interviewer therefore tries to minimise the impact of interpersonal processes on the course of the interview. By contrast, qualitative researchers believe there can be no such thing as a "relationship-free" interviewe. Indeed, the relationship is *part* of the research process, not external to it. The interviewee is seen as a "participant" in the research, actively shaping the course of the interview rather than passively responding to the interviewer's pre-set questions.

The qualitative research interview is ideally suited to examining topics in which different levels of meaning need to be explored. This is something that is very difficult to do with quantitative methods, and problematic for many other qualitative techniques. Finally, the qualitative research interview is a method which most research participants accept readily. Some researchers have suggested that people like talking about their

work, whether to share enthusiasm or to air complaints, but rarely have the opportunity to do so with interested outsiders (King, 1995).

Without doubt, there is no one-research method which has advantages without having its disadvantages. Qualitative research interviews have several disadvantages. The process is very time consuming, (in developing an interview guide, carrying out interviews, and analysing transcripts). It is essential that the researcher does not attempt to take on more interviews than he or she has time for in a study. Qualitative research interviews are also demanding to conduct, as they involve considerable concentration from the interviewer. Interviews are also time-consuming for interviewees, and this may cause problems in recruiting participants in some organisations and occupations. The best recruitment strategy is probably to send a letter with basic details of the study's aims and what will be required of the interviewee, with a follow-up phone call in which the researcher can explain his or her aims in more depth and answer any queries.

Occasionally, the researcher will experience "difficult" interviewees, defensive, hostile or unable/unwilling to focus on the research topic(s). To be told directly "that was a ridiculous question!" or I can't see why you are asking me these things' is much more uncomfortable than receiving a questionnaire back from a survey participant with "stupid question!" scrawled across it. The researcher may be able to use interpersonal skills (above all, patience) to defuse awkward situations, and must ensure that his or her research questions are likely to be relevant to the people being interviewed.

A difficulty faced by many researchers using qualitative research interviews is the feeling of data overload as a result of the huge volume of rich data produced by even a moderate-sized study. It is easy for the researcher, particularly if inexperienced in the method, to feel that he or she is lost and sinking in a quagmire. This is most likely to happen when the researcher is utilising solely qualitative methods in an institution dominated by traditional quantitative techniques. In these circumstances, King (1995) suggests three directions where the researcher can turn for a 'rope' to pull him or herself out of the swamp. First, there are the original aims of the study. If beginning to feel lost in a particular line of exploration, the researcher should ask: 'Is this adding to my understanding of the topic(s) I set out to study? If not, is it raising new and related topics which are of interest?' If the answer to both questions is "no", then the researcher

should change the direction of the analysis. Second, the inexperienced researcher can turn to literature describing other studies using qualitative research interviews, to provide examples of how problems in data analysis were tackled, including material outside his or her own area. Third, personal networking is of great importance. If there is no one even sympathetic to qualitative methods in the researcher's own department, he can try other departments.

#### **6.2.2.3 Company documents**

Organisational documentation comes in many forms: company annual reports; public relations material and press releases; accounts statements; corporate mission statements; policies on marketing strategy; formal charters and legal documents; policies on rules and procedures; human resources management strategies; policies directive on training, career management; job mobility and relocation management; formal memos between different groups and departments; and informal and private correspondence between staff and correspondence between respondent and researchers (Forster, 1995).

There are many advantages to using organisational documents such as:

(a) Organisational documentation and the history of the company. Documentary records data is often more comprehensive than the kind of material which the researcher who is new to an organisation could obtain from either interviews or questionnaires. They are often contemporary records of events in organisations. This can help the researcher to look more closely at historical processes and developments in organisations, and can help in interpreting informants' "rewriting" of history. Most of it is within company boundaries and even those documents which are explicitly public relations in tone, can tell the researcher a great deal about the kind of "image" and the culture of a company.

(b) Organisational documentation and understanding. Company documents are (con)textual paradigms which are an integral part of other systems and structures in organisations. They can highlight different understandings and interpretations of life amongst different groups within companies. (c) Organisational documentation and time and validity. Company documents save time in research. They enable the researcher to access detailed information on, for example, organisational structure, succession planning systems, career development policies, and so forth. Organisational documentation provides another means of triangulating data.

If company documents have all these advantages why are they not used more in organisational research? Forster (1995) argues that company documentation may not be an accurate record of actual events and processes. So they should never be taken at "face-value". Company documents need to be carefully checked, interpreted and triangulated with other data sources. There is no universal method of analysis of organisational documents. This research will use the hermeneutic method to analyse company documents. The hermeneutic method is concerned with the study of historical texts, with understanding rather than hypothesis-testing, and analysing the meaning of individual texts. The rationale for choosing the hermeneutic approach is that it describes the process of developing an inductive understanding of clusters of company documents, through to a deductive understanding of the whole.

Forster (1995) identifies seven stages in the hermeneutic process and the researcher has broken them down into six stages<sup>119</sup> (see figure 6.4).

<sup>&</sup>lt;sup>119</sup> The researcher sees the third stage of Forster's 'seven stages' as identifying thematic clusters, which developed from the second stage which identified (sub-) themes. The researcher combined the two stages in the GOGCWS.





Source: Forster, N. (1995)- " The analysis of company documentation", p. 151

Understanding the meanings of individual texts. This stage involves a continuous back and forth process between the parts and the whole. The initial task of the researcher is to search for themes within each document and then within clusters of the documents. At this stage the focus of attention is on meaning rather than on analysis. From this, "ground knowledge", themes and sub-themes will emerge, and the researcher can then determine what is important to extend our understanding of textual themes and subthemes.

*Identifying (sub-) themes and thematic clusters.* At this stage the researcher tries to understand contradictory sub-themes, so it then becomes possible to cluster documents.

*Triangulating documentary data.* At this stage, the analysis of company documents is critical. A serious problem arises here, because whilst a segment of text may be analysed in the researcher's own right, the true meaning of these can only be evaluated with reference to other texts and other forms of data. There are two solutions for these problems. The first solution is to recognise that it impossible to do full justice to the range and quantity of documents in any organisation. The best that we can hope to

achieve is to provide slices of organisational life in academic journals. The second solution is to check reliability and validity. The GOGCWS research study opted for the second solution (see section 9.3).

*Employing reliability and validity checks.* The aim of this stage is to enable the researcher to verify the findings.

(*Re*)contextualising documentary data. Company documents can only be fully understood within broader organisational contexts and processes, and with reference to other forms of data. The GOGCWS research study used company documents, questionnaires and interviews.

*Using representative case material.* At this stage, a researcher selects which documents are to be sampled and used as the case materials in company reports, conference presentations and research publications.

# **6.3 Conclusion**

This chapter has focused on the methodology and method used in the GOGCWS research. The research combined inductive and deductive approaches, focusing more on the inductive strand. It adopts both nominal and voluntraistic views. This research is characterised as "institutional analysis", similar to that of Scapens and Macintosh (1996). Theory is used in order to understand and explain the role of cost management in the Egyptian water industry. The research objective is to generate theories which provide good explanations (explanatory case study, Scapens, 1990).

The case study method was adopted in this research. The argument about methods depends on two factors; firstly, the relationship between theory and method, and secondly, how the researcher attends to the potential weaknesses of the methods (see sections 6.2.1.1 and 6.2.1.3).

This research is based on one case study - The General Organisation for the Greater Cairo Water Supply (GOGCWS) (the rationale for choice of company, can be found in section 6.2.1.3). The case study method was used for analysis of processes of cost management at the GOGCWS (inductive approach). The case study focused on

understanding the dynamic relationship between cost management and the external environment (see chapters 9 and 10).

This chapter has illustrated different types of accounting case studies (descriptive, illustrative, explanatory, exploratory, etc). It also explained how a case study is conducted, - beginning with its preparation and ending with report writing. The case study in this research had a descriptive and explanatory purpose.

The research combined quantitative research (i.e. questionnaires) and qualitative research (interview methods). The questionnaire was used as a basis for the semistructured interviews. The data was collected from multiple sources (triangulation), including company documents and questionnaires.

# Chapter 7

# The Water industry, Cost management and Institutional theory

# 7.0 Introduction

The aim of this chapter is to illustrate the importance of water, to explore the phenomenon of privatisation in the water industry and the effect it has on the role of accounting, considering cost management in particular. The chapter will also consider the pros and cons of public versus private-owned water systems. Finally the issue of costs in the water industry will be explored (i.e. how do people pay for their water requirements, how to calculate the cost of water, and is this system fair or is a new system required).

The research focuses on England, France and Egypt. The rationale for choosing these three countries is England has experience of privatisation and problems in the water industry. Before privatisation, England experienced similar problems to those currently experienced in Egypt. France was selected because in the researcher's view, it has one of the 'best'<sup>120</sup> water industries in the world (in terms of the management of water). The French system is a state-regulated system. Egypt is currently experiencing many problems with its public utilities (especially the water industry).

This chapter is divided into four sections. Section one offers a definition of a public utility and indicates that the water industry is one of these. The chapter also highlights the importance of water, the responsibility of government (in relation to) water, and how to meet the growing demand for water.

Section two examines the aim of privatisation and why the UK government chose to change from a regulated water industry to a privatised one. It also examines the effect of privatisation on the customer, on efficiency, on costs, on competition, and on

<sup>&</sup>lt;sup>120</sup> The researcher has built his view from many examples such as; in the global water industry, the two largest companies are both French; Suez Lyonnaise des Eaux and Vivendi. These two giants dwarf the UK water companies- Suez Lyonnaise has a market capitalisation of £15.6 billion and Vivendi £22.9 billion. Their mission is also clear. When the chairman of Suez Lyonnaise was asked about this strategy, he replied: "In water and waste water services, we want to be world leaders". These companies are competing aggressively in the global market and are respectively four and six times larger than the largest UK water company, Thames Water, which has a market capitalisation of £3.9 billion.

performance. Finally, a brief assessment of the system of water in France and Egypt is conducted from an institutional perspective.

The third section illustrates cost systems and problems in the water industry; i.e. how people pay for their water requirements, how the cost of water is calculated and whether this system is fair or a new system is needed.

The final (fourth) section outlines the main conclusion of this chapter.

## 7.1 Public utility

## 7.1.1 Characteristics

No single characteristic distinguishes a public utility from other firms. However, several conditions that follow Howe and Rasmussen (1986) when considered in combination do serve to identify a company.

(1) A public utility frequently has natural monopolies, in a similar way to electric, gas, water and telephone companies

(2) The prices these industries charge for services are determined by government regulation

(3) The utility must supply all who require the service, when they want it and at the regulated price. The utility cannot turn away customers who are willing to pay the prescribed price

(4) Utilities are given franchises as exclusive suppliers within the franchise area

(5) The utility is subject to additional regulations not required of other firms, including regulation of accounting practices, procedures of security issuance, plus extensive reporting of business activities

(6) A utility must provide a service that is essential in some sense to modern living but this condition is difficult to define in practice

### **7.1.2 Types of services**

Another way of identifying a public utility is by specifying the services it provides. As the well-known public utility economist (Bonbright, 1961) put it, there are basically two categories of public utilities: (a) Those enterprises which supply, directly, continuous or repeated services through more or less permanent physical connection between the plant of the suppliers and the premises of the consumer

(b) The public transportation industries. Important among this first group are utilities supplying electricity, gas, telephone, telegraph and water. The second group includes airlines, motor freight carriers, railway, water carriers, petroleum pipelines, and local transit systems.

An interesting alternative viewpoint is expressed by the economist Primeaux, (1979) who argues that the electric utility currently does not possess the prerequisite conditions for a natural monopoly and therefore its monopoly status, as a public utility, ought to be re-examined.

# 7.1.3 A schematic Diagram of the public utilities

The political and economic institutions that are important in public utility economics are shown in figure 7.1.





Source: Howe and Rasmussen, (1986)- "Public Utility Economics and Finance", p. 9

A brief explanation of the relationships among the various participants is given as follows:

A- Creditors and owners provide capital to finance the rate of base-working capital, land, and plant and equipment; and in turn, receive interest payments and dividends.

B- Regulatory commissions regulate and control; they set rates so as to provide a reasonable return to the capital suppliers of the utility and ensure adequate service. Public utility managers provide information and request permission for changes in service, corporate policy, and rates.

C- Legislatures give the mandate to monitor and set the general provisions of the regulation.

D- The acts of state and federal legislatures and administration are subject to review by the court.

E- Public utility firms provide a service to various classes of customers and receive payment.

F- Industrial and commercial consumers and residential users render service complaints and suggestions for rate level and structure change.

G- Potential competition is largely limited, although specific types of competition are allowed in some areas by the regulators.

These institutional relationships may be compared and contrasted with the economic circular flow diagram presented in figure 7.2.





Source: Howe and Rasmussen, (1986)- "Public Utility Economics and Finance", p. 10

As with economic circular flow, firms provide services in exchange for revenues and incur costs by purchasing resources, thereby providing income to resource suppliers. Since competition does not work well in many utility markets, the balance between revenues, costs, utility services and resources is distorted.

The public therefore regulates the rates and services of public utilities and attempts to restore the imbalance by eliminating monopoly profits and the underproduction of services. (Tax revenues, raised from general revenues or from excise taxes on utility services finance regulatory commissions and other governmental bodies). The administration of the regulatory commission will also affect each regulatory jurisdiction via public employment and governmental budget policies.

Public utilities are ultimately public only by law. For example, oil refineries, though in many ways a public utility, cannot said to be so without legislation establishing regulation. Fundamentally, a public utility is a creature of law, a creation of property rights more limited than those permitted by businesses and consumers in general.

To summarise, the water system is increasingly being distanced from other utilities. The boundary between the competitive market and regulated activities has been drawn in the water industry so, that competition is almost excluded. With the implication that the industry as a whole is a natural monopoly, their activities will be subject indefinitely to prospective regulation.

### 7.1.4 Importance, demand for water and the responsibility of government

Byatt - The Director of Office of Water Services in England - (1997) set some objectives which he wanted to achieve generally in public utilities but especially in the water industry. These aims are as follows:

(a) Price review<sup>121</sup>

- (b) Customer service standards<sup>122</sup>
- (c) Competition<sup>123</sup>
- (d) Charging policy<sup>124</sup>
- (e) Open and transparent regulatory regime<sup>125</sup>

Water is the most important natural resource in the world, as without it life cannot exist and most industries could not operate. Although human life can exist for many days without food, the absence of water for only a few days has fatal consequences. The presence of a safe and reliable source of water is thus an essential prerequisite for the establishment of a stable community.

In the absence of such a source, a nomadic life-style becomes necessary and communities must move from one area to another as demands for water exceed its

<sup>&</sup>lt;sup>121</sup> A price review aims to reset water and sewerage price limits and to reduce prices in real terms for customers through significant reductions in bills. It also aims to maintain incentives for companies to increase their efficiency and reduce costs, while also ensuring an adequate level on investment in capital assets in order to arrive at the price limits via an open, transparent, fair and responsible process. This includes a high level of consultation with the industry and its customers.

<sup>&</sup>lt;sup>122</sup> Customer service standards protect the interests of customers by ensuring a good quality service. The comparative competition between water companies to increase standards and reduce costs, protects the interests of customers by maintaining and improving the efficiency and effectiveness of the system of customer representation.

<sup>&</sup>lt;sup>123</sup> Competition develops an effective policy to maintain the role of comparative competition as this is important where direct competition is not possible.

<sup>&</sup>lt;sup>124</sup> A charging policy is necessary to secure a cost reflective tariff structure that encourages efficient use of water and takes into account the needs of customers.

<sup>&</sup>lt;sup>125</sup> An open and transparent regulatory regime provides information about the industry and its regulation.

availability. Therefore it is not surprising that sources of water are often jealously guarded and over the centuries many skirmishes have taken place over water rights.

History shows many occasions where agricultural development has been hindered by interference with water supplies as part of the conflict between landowners and settlers. This has occurred in numerous parts of the world. Other conflicts in relation to water supplies can arise because of the effects which human and industrial waste can have on the environment. This means that the importance of water as a natural resource, which requires careful management and conservation, must be universally recognised. Although nature often has great ability to recover from environmental damage, the growing demands on water resources necessitate the professional application of fundamental knowledge about the water cycle to ensure the maintenance of quality and quantity.

Demand for water is growing into the 21st century, particularly demand for public water supplies. The most important factor affecting the demand of water is the rise in population, for example; Egypt's population increased from 55 million in 1994 to 63 million by the year 2000 and is expected to increase to 86 million by 2025. The demand for water is also expected to increase.

There are some approaches to meet the growing demand for water such as

(1) Raising awareness of the importance of water through public education campaigns

(2) Encouraging people/ business/ industry not to waste water

(3) Installing water meters on all domestic properties, which could reduce demand by about 10% (as water becomes scarce should we pay more for it?)<sup>126</sup>

(4) Increasing the price of water on a graded scale of charges

(5) Restricting water use during periods of drought

(6) Managing water pressure to reduce leaks (the General Organisation for Greater Cairo Water supply, which provides water to 25% of population's Egypt, loses 40% of water through leakage each year)<sup>127</sup>

<sup>&</sup>lt;sup>126</sup> Information source: http://www.Oxfordshire.gov.uk- "Water- where are we going? Agenda 21".

<sup>&</sup>lt;sup>127</sup> The GOGCWS, (no date source given, c)- "Greater Cairo Water", The Administration of Public Relations.

Recent approaches to meet the growing demand for water have focused on voluntary mechanisms to conserve water and reallocate it among competing users rather than on developing new supply resources. At the same time, governments are seeking alternative ways to finance the operation and management of existing infrastructure and the construction of new infrastructure. These alternative approaches may be divided into three categories (Thobani, 1997):

- (a) Pricing policies
- (b) Informal water market
- (c) Formal water market

In many developing countries, governments consider water to be too precious a resource or too difficult a commodity to be left to the market. Decisions about who gets water, at what price and for what use have thus been entrusted to public officials. Although the state retains ownership of this asset, it typically endows private and public entities, such as farmers, industrial users, and power and water companies with the right to use surface water or groundwater for a particular purpose.

The allocation of water rights is typically the responsibility of the government, as is the construction, ownership and operation of the infrastructure such as dams, reservoirs, and canals. In some countries the government even installs and operates wells. Recently many countries have transferred operations and management responsibility to user association; enforcement is the responsibility of public authorities or water user associations, or both (Thobani, 1997).

User associations are typically informal entities that play a role in distributing the water from a canal or river. In some countries, however, such as Chile, Mexico, and Peru, user associations are legally recognised bodies which set and collect fees and who operate and maintain the hydraulic infrastructure. Publicly administered systems of water rights have all too often resulted in inefficient use and supply of water.

Government control has not been effective at ensuring that the poor have access to water. In many cities in developing countries, the poorest are not served by piped municipal water and must resort to buying water from private vendors at prices that are several times higher of those paid by more wealthy residents. Also, farmers who are politically influential manage to get easier access to water rights, which are obtained without charge and for whose use farmers typically pay only a nominal fee.

# 7.2 Privatisation

## 7.2.1 The aim of privatisation in UK

In 1967, a White Paper (Cmnd 3437, 1967) introduced more formal criteria for pricing and investment decisions, and public sector industries in the United Kingdom had been "steadily pushed towards behaving according to commercial rather than social criteria" (Whitehead, 1988: 6).

In the early 1980s, the Conservative Government sought radical changes in the management of public enterprises, inspired by its belief in the innate inefficiency of traditional public sector provision of goods and services. Alarmed at increasing demands on the Exchequer to make good operation losses and finance capital programmes, the government determined to remedy the lack of any efficiency checks on industries' performance.

More generally, difficulties in measuring efficiency and productivity and the problematic nature of making comparisons between public and private sector enterprises, have led to debate about the government's claim that private sector provision of goods and services is necessarily better (Kay, et al., 1986; Utton, 1986; Vickers and Wright, 1989). Some academic commentators have been especially critical over the limited extent to which privatisation has actually promoted more competition (Heald and Steel, 1986; Bishop and Kay, 1988; Kay and Thompson, 1988; Vickers and Yarrow, 1988). Opportunities to break up the old public sector monopolies and create greater competition have been largely ignored, not least because of the influence of the treasury and senior management within the industries concerned. The treasury has been primarily concerned with raising revenue from the sales of public assets and was very aware that "monopolies are worth more than competitive industry" (Kay and Thompson, 1986). Senior management had also opposed any re-structing of their industries. In the early phases of privatisation policy, some chairmen of nationalised industries were openly critical of the government's policy (Thomas, 1986), they were generally more supportive of the policy, however, when concessions from government were forthcoming in terms of competition, debt write-off, and regulation (Hills, 1986; Utton, 1986; Heald, 1989; Vickers and Wright, 1989; Mitchell, 1990).

Ogden and Anderson (1999) say the aim of a move from public sector to private sector for a public service is to achieve "profit and creating shareholder value". Ogden (1995) says the main public justification for pursuing privatisation had been the government's claim that it would bring greater efficiency and improvements in the production of goods and services through more competition and easier access to capital markets. This basis for policy has been informed by the government's belief that private sector provision of goods and services is intrinsically more efficient than public sector provision. Management in the public sector, with little market incentive to improve performance, is viewed as "operating bureaucracies and not businesses" which leads to rigidity and inefficiency, and an inability to respond effectively to technological change or changes in consumer demand (Moore, 1985). For example, the white paper on Water privatisation (Cmnd 9734, 1986) argued that removal of government intervention and political interference in the day-to-day management of enterprises would improve management performance and access to private capital markets would make it easier for enterprises. It concluded:

"Freeing the authorities from the constraints imposed by state ownership will help them to carry out their tasks with vigour and imagination" (para. 38). The Government's claim as for example, argued in the 1986 White Paper on water privatisation (HMSO, 1986), that private sector provision of goods and services, subject to the disciplines of market forces, is more efficient than public sector provision.

The aim of Privatisation (Ogden and Anderson, 1999: 96) is that "private enterprise is both more flexible and readier to pursue energetic and innovative approaches than the public sector. The demands of the market will give management and staff the impetus they need to secure greater efficiency". This aim can be achieved if the privatised enterprise is more competitive. In the case of privatised public utilities, this has not happened, attracting critical attention (Heald and Steel, 1986; Kay and Thomposon, 1986; Bishop and Kay, 1988; Vickers and Yarrow, 1988). Consequently the monopoly character of the industry has remained unchanged with privatisation. Moore (1985) says that privatisation can achieve great efficiency and many of benefits. The privatisation of the utilities has resulted in generous dividends to the shareholders and remuneration to top management, an unsuccessful spending spree in search of future earning growth, and a 25 percent fall in employment for the workforce (Froud, et al., 1996). In the UK, Water Authorities were privatised in 1989 and other major state industries which have been sold off include British Telecom in 1984, British Gas in 1986, and Electricity distribution and generation in 1990 and 1991.

In 1989, the ten Regional Water Authorities of England and Wales were privatised. The Regional Water Authorities were a product of a major re-organisation of the Water industry in 1973. Ten Regional water authorities consists of nine in England (Anglia, Northumbria, NorthWest, Severn Trent, Southern, South West, Thames, Wessex and Yorkshire) and one in Wales (Welsh Water). They vary in size from Thames, the largest with a turnover in 2000 of £1,367,8 million and workforce of 9,9995 to SouthWest, the smallest with a turnover in 2000 of £467 million and a workforce of 3,500 (Annual and Summery Reports, 2000).

From an institutional perspective, the researcher can argue that the policy of the UK's water industry has been informed by the government's belief that one external-factor is expected to motivate government managers to acquire and use cost accounting systems if the presence of competition and funding is uncertain (Scott, 1987). Government reform literature argues that agencies that are required to be self-supporting and compete for funds face many of the same competitive pressures encountered in the private sector, and are similar to private sector firms with zero profit targets (Brown and Sproghe, 1987; Osborne and Gaebler, 1992).

As a result, reform advocates claim that agencies faced with market pressures and requirements to recover all expenditures with revenues will become more businesslike and cost conscious, and will make greater use of cost accounting data in order to control costs and prevent funding shortfalls and associated sanctions (Osborne and Gaebler, 1992; Gore, 1993).

### 7.2.2 The effect of privatisation

The aim of any market, private or public, is to satisfy the customer. The customer is satisfied if he/she receives a good product, properly delivered, at a fair price. These elements lead us to look at efficiency, competition, and performance.

### 7.2.2.1 The effect of privatisation on efficiency

Efficiency is the main aim of privatisation through competition. In other words, competition is seen as the tool to achieve efficiency. Why do we need efficiency in the water industry? The Confederation of British Industry (1991) answers this question and sets some comments for this point such as; "the cost of water depends not just on the unit price but also on the volumes used and wasted". Minimising water wastage on site can mean reduced capital input and operating costs for on site water treatment and help to reduce the pressures and demands on the service network. Efficient water use can mean energy savings and improved water management can increase operational efficiency. The financial advantages of water saving and reduced wastewater discharge can offset the cost of re-circulating, recycling and re-using water. This is due to water being the main constituent of many effluents. Through the minimisation of water use and the reduction of losses, the effluent volume may be reduced and on-going effluent treatment costs cut, thereby reducing the demand on water supply and helping to safeguard supplies for the future.

The main issues of how to procure investment and how to drive efficiency improvement exist under both public and private regimes. The problem economists have called it "principal/agent<sup>128</sup>", refer to differences in information and incentives between players. Companies know best about their own business but do not have the same motivation as

<sup>&</sup>lt;sup>128</sup> The basic insight of principal-agent theory (PA) is, well known. One party (the principle, P) hires another (the agent, A) to undertake a particular task, but the former party suffers from an information asymmetry that introduces a problem in terms of motivating the agent.

In both economics literature (Arrow, 1985; Rees, 1985; Laffont, 1989; and essays in Diamond and Rothschild, 1985) and the accounting extensions (Jensen and Meckling, 1976; Watts and Zimmerman, 1986) the solution to a PA problem involves the development of an optimal incentive package that furthers the objectives of the principal and constrains the actions of agents. To investigate the difficulties we can identify two types of information asymmetry: hidden information (or adverse selection) and hidden action (or moral hazard). The former asymmetry exists ex ante but the information becomes available after completing the transaction. Hidden action asymmetries exist ex post in that an agent's actions can affect the outcomes that occur but these are not observed by the principal.

politicians. Therefore regulators are needed to protect consumers from excessive prices and to ensure quality standards, investment and efficiency.

Does the efficiency saving benefit customers? (OFWAT, 1996). If yes, how? The role of Office of Water Services (OFWAT) in England is to ensure that the prices paid by customers are fair and justified. The regime must also ensure that companies are efficient. A price control system ensures that the benefits of efficiency improvements are shared between customers and shareholders. The best way for the companies to do this is to reduce or defer the take up of their price limits. If profits are to be directly controlled, the regulator must stipulate the maximum acceptable return on assets that can be made in any period. This "rate of return" regulation can involve the regulator adjusting prices every year on the basis of the profit made by the company. The excess return could be clawed back by the regulator which could lead to companies boosting their asset base and providing "gold plated" solutions.

The Director of Office of Water Services must encourage companies to operate efficiently. He monitors standards, comparing performance between companies, against each company's targets and with other sectors. The Director must give companies incentives to reduce operating costs and the cost of sustaining their assets, raise money cheaply and complete environmental projects at low cost.

The regulator in the U.K. is forced to employ very unsatisfactory methods, such as 'comparative' (yardstick) competition which has come to occupy a central place in efficiency comparisons in the water industry. Comparing different companies and trying to bring the less efficient up to the level of the more efficient might appear to be a superficial way of simulating real competition. But, in practice, this does not happen.

Bryan, (1998) identifies three issues related to competition in water. Firstly, there is a big difference in principle between dynamic competition in a real market and static comparisons between companies in a non-competitive market. Where there is no competition, the incentive for improvement which exists in competitive markets is absent. In tightly regulated markets, companies are not so much trying to better their rivals by innovating and reducing costs, they are just trying to impress their regulator. Secondly, making meaningful comparisons in a comparative competition regime is

extremely difficult. In principle, the problem is that there are many variables which affect the costs of any given company. Some means of standardisation for these has to be found if useful efficiency comparisons are to be made. Thirdly, concentration on comparative competition distracts attention from introducing the real thing. It has also produced side effects in water which verge on the ridiculous. There is much concern to avoid "losing" comparators.

Privatisation is best regarded as part of a continuing process of change that has affected the industry over the decade of the 1980s, rather than a singular event. In the early 1980s the water authorities were subject to increasing pressure from new government financial controls and performance aims to become "more efficient". Once the Government publicly announced its interest in privatising them in 1985, and published its intention to do so in its White Paper in 1986, these pressures intensified. Like other privatised enterprises, the water authorities were strongly encouraged to improve their profitability and efficiency profiles to make their successful flotation more viable (Aylen, 1988; Chambers, 1988; Rickard, 1988). They had an extended opportunity to do so, as they experienced an untypical prolonged period of run-up to privatisation, owing to a number of technical and political difficulties the Government encountered with its privatisation proposal (Ogden, 1991).

The water authorities had to transform themselves from public sector bodies concerned with containing costs to private sector PLCS pursuing profitability. The vocabulary of profit replaced the previous one of costs as senior managers deployed a new economic language for describing organisational motives. Managers in the new water PLCS have had to respond to new expectations and assessments of corporate performance that will be measured by shareholders, investors and financial analysts in term of profitability. Although many have been persuaded, by the government's rationale for privatisation by 1989, there remained some managers and employees who retained doubts about the legitimacy of profit as the primary objective for the new Water PLCS. Southern Water's group finance director said in a speech to the Institution of Water Officers in 1992, found that it was still necessary, three years after privatisation, to argue that: "We have to make everybody think profit. Profit is not inconsistent with this industry. It's not a dirty word" (Water Bulletin, 1992).

#### 7.2.2.2 The effect of privatisation on competition

This section considers competition in the UK water industry in a global context. The importance of competition is examined and its implications for efficiency are investigated.

A competitive force in the utilities generally has brought benefits to customers by reducing costs and improving the quality of the service. In the water industry, the chances of market competition were extremely limited, a view shared unequivocally both within and outside the industry (Littlechild, 1986; Harper, 1988). Nevertheless, although it did not entail any significant developments in competitive market forces, privatisation has brought into operation a new economic regulatory frame-work for the provision of water services which provides considerable stimulus to achieve better performance through the operation of a pricing formulae and "yardstick" competition.

During the 1980s the water authorities were subjected by the Government to increased scrutiny and rigorous demands in order to improve their financial performance, improve their efficiency and profitability in preparation for successful flotation as privatised Water PLCS and to submit themselves to yardstick competition and comparative judgements by the financial markets. The ways in which the water authorities adopted new strategic postures as they responded to these successive changes, provided a fertile research site for exploring the roles accounting may play in processes of organisational change.

The international market for water and waste water services is estimated to grow to between \$370 billion and \$380billion per annum by 2010 (West, 1999). Clearly, this represents a substantial opportunity for UK water companies. Indeed, UK companies are already establishing a presence on the global water stage. Thames Water, the UK's largest water company, has responsibility for supplying water of the city of Lzmir in Turkey. Anglia Water has recently won a contract to supply one and a half million people in Valporaiso in Chile. Yorkshire Water has announced its intention to buy Aquarion, a US water company. The UK, the US and France currently have water companies of sufficient experience and critical mass to take advantage of this opportunity. In France, there are two large companies: Suez Lyonnaise des Eaux and Vivendi. Suez Lyonnaise has a market capitalisation of £15.6 billion and Vivendi has a market capitalisation of £22.9 billion. Their mission is clear. When the Chairman of Suez Lyonnaise was asked about his strategy, he replied: "In water and waste services, we want to be world leaders" (Rider, 1999).

Competition in the water sector is very valuable; the UK water regulator (OFWAT's) prospects for price reductions between (10% to 20%) could extract in excess of £2 billion out of the industry over the five-year period of the review<sup>129</sup>. This will be a blow to the water companies and leave them with fewer funds for international expansion. Even with substantial investment to date, it is estimated that meeting the quality of standards will still cost in excess of £8 billion.

The Department of the Environment produced a consultation paper considering ways of extending opportunities for competition in the water industry which was published on 1 April 1996, with a consultation period open until 1 July 1996. The consultation paper proposed seeking legislative changes for inset appointments, cross-boundary supplies, common carriage and liberalising connections between companies. The Director of Water Services has been involved in the preparation of this paper and supports these proposals.

Summerton (1999) says no degree of regulation can match genuine competition and in the next few years we should expect to see far greater effort and imagination going into how to introduce real competition in certain aspects of the water industry.

Before we leave the subject of competition in the water industry we should look at the view of the UK water regulator on mergers (who has prevented mergers between UK water companies). Byatt (1997) has argued that mergers, which reduce the number of comparators, are against the public interest and should only be allowed to proceed if they result in the emergence of a new better comparator who is able to deliver a better service at a lower price to customers. The regulator believes that consolidation would

<sup>&</sup>lt;sup>129</sup> The estimated £2 bn revenue shortfall is derived from the estimated Po and K factor projection for the Water Services Companies delivered in the prospects for prices consultation paper (October 1998).

reduce the number of companies and therefore it will be harder to compare their performance to determine which companies are efficient (and should be rewarded) and which are inefficient (and need to improve).

Indeed, the regulatory regime in water seems inconsistent with the present U.K government's commitment to protect consumers via competitive markets wherever possible (Robinson, 2000). A price review is even more important to the companies concerned than elsewhere in the utility sector because the scope of regulation is so wide. On the other hand utility companies are becoming more entrepreneurial and more occupied with ways to match and, where possible, beat the actions of their competitors. In water, there is a preoccupation with comparative competition. Real competition is marginal: regulation, both "economic" and "social", is the big issue because it bears such weight.

Another symptom of the absence of competition in the water industry is the tendency for government intervention in the industry to continue. Politicians evidently feel the need to prompt the regulator into setting targets (e.g. for leakage) rather than using market incentives (The Financial Times, 1997).

Consolidation could provide significant opportunities for price reduction for the consumer. There are two reasons: firstly, the most efficient companies will prosper and less efficient companies will be taken over and have standards raised. Secondly, companies will share significant overheads (such as call centres and billing systems) and this will reduce the unit cost of serving individual customers. Consolidation will create companies of a size which can compete on a level playing field with the small number of international giants. For example, the combination of Thames Water (£3.9 billion) and Severn Trent (£3.4 billion) would create a £7.3 billion company which could begin to take on the major US and French players on their own terms (Rider, 1999).

### 7.2.2.3 The effect of privatisation on company performance

Prior to privatisation, the Water Authorities were already subject to Government imposed financial targets and performance aims on operating costs since 1981; they

were strongly encouraged to improve their profitability and efficiency profiles to make their shares more attractive to buyers when flotation occurred (Ogden 1995).

After privatisation, The Director General of the Office of Water Service (OFWAT) had responsibility for ensuring that

(1) The water PLCS are able to finance the proper responsibilities of their functions

(2) Charges are regulated through a pricing formula based on the retail price index PRI+K; the K factor is assessed separately for each water company. The pricing formula is expected to promote efficiency and reduce costs as a means of enhancing profitability

(3) Developing competition through a review in "K" factor in the pricing formula. In the interim, OFWAT has published annually comparable information on issues such as comparisons of unit costs on water delivered and sewerage collected (OFWAT, 1993b), capital investment (OFWAT, 1992a) and levels of service (OFWAT, 1992b). The latter have been developed to assess company performance to customers and are measured by seven levels of service indicators: (DG1 and DG4); adequacy of Water Distribution System (DG2 and DG3); adequacy of the sewerage (DG5); and response to customers (DG6 and DG7).

In conclusion, the Director General is not the only person interested in efficiency. In their assessment of the water PLCS current and future profitability, the financial markets and city investment analysts are equally keen to scrutinise management performance in achieving cost savings and making other improvements in efficiency. While financial indicators of corporate performance are clearly the key indicator for city analysts, they also take careful account of management performance on levels of service to customers (Ogden, 1994).

To summarise, the UK water industry moved from the public to the private sector because the government believes that private sector provision of services is more efficient than public sector provision. This move affected the role of accounting<sup>130</sup> and this is the case of the water environment that transferred from the public to the private

<sup>&</sup>lt;sup>130</sup> For example, prior to privatisation, the water authorities were concerned with containing costs, but after privatisation, in pursuing profitability. In other words, vocabulary of profit replaced the previous vocabulary of costs.

sector. The researcher's view is that it is of more value to look at other water environments before we look at the main case study of the thesis, the public water environment in Egypt. E.g. France, whose water industry has been mixed between public and private sector (see section 7.2.3).

### 7.2.2.4 Institutional Change and Privatisation

In numerous text and articles researches identify differences between public and private sectors in terms of their environments, resourcing and management practices (Alford, 1993; Harrow and Willcocks, 1990; Ring and Perry, 1985). Gerry, (2000) argues that privatisation can be seen as coercive institutional change. Furthermore, he argues that during the transitional process of privatisation, two institutional templates coexist and compete for legitimacy<sup>131</sup>. The public sector template may be deeply embedded, taken for granted and largely mindlessly enacted. However, as privatisation proceeds, the private sector institutional template and its implications are mindfully anticipated.

Institutional theory can, however, help us to understand processes of privatisation. Its proponents point to the importance of realising that privatisation constitutes institutional change and suggest mechanisms by which resulting archetypes of organisations will emerge, not only through coercive mechanisms but through normative and mimetic processes (DiMaggio and Powell, 1983). There is also a growing body of literature on deinstitutionalisation and change from an institutional point of view, the conclusions of which can be extended to the context of privatisation.

Fligstein (1991) underscores the need of shocks to the organisational field while acknowledging the role of organisational elites. Oliver (1991) uses institutional and resources dependency theories to explain strategic responses to institutional processes in the context of differing institutional factors. Greenwood and Hinings (1996) argue that radical change is more likely if there is loose coupling within institutions and less well-defined means for preserving dominant templates. Kondra and Hinings (1998) also predict that change is more likely to occur in institutional environments where there is more, rather than less, diversity or higher influence from other organisational fields.

<sup>&</sup>lt;sup>131</sup> In the actors' eyes.

As mentioned in section 2.3, this research adopted a process approach and focused on the micro level. The discussion of this section aims to provide a better understanding of the role of actors in institutional change. Institutions are defined by their rules, this is so both in terms of what is meant by institutionalisation and in terms of how individuals identify with those institutions by "learning to act in a particular way" (March, 1994: 42). Changes in institutional templates, as in privatisation, involve a period of flux (Greenwood and Hinings, 1993). In such circumstances, both from an institutional and individual point of view there is a search for identity (Dutton and Dukerich, 1991), which may be strongest when external cohesion or constraints weaken (March, 1994), as in the context of privatisation.

In the context of institutional change in privatisation, actors are moving from a wellestablished, familiar institutionalised context to a less familiar and, therefore, uncertain private sector context. Thus, we would expect increased conscious information processing because some tasks will be familiar and some unfamiliar and inconsistent with prior experience (Lord and Smith, 1983). Gerry (2000) argues that processes of privatisation may be different in terms of sequencing and procedures, for example there could be a direct sale to a single private sector company, a break up and sale to a number of companies, or management buyout. This shift from public to private sector could be at a particular point in time, or there may be a period of transition.

Processes of privatisation also may affect the extent to which actors have opportunities to reflect on such experiences and are again important in promoting conscious processing (Lord and Smith, 1983). Furthermore, the involvement of actors in changing processes increases not only the likelihood of their acceptance but also their commitment to change (Coch and French, 1948; Nutt, 1998; Watson, 1976). Gerry (2000) argues that the greater the extent of actors' personal involvement in privatisation processes, the greater their direct experience of the private sector. The more actors identify personally with the outcome of privatisation, the more conscious their information processing.

Institutionalists have long understood the significance of organisational structures and systems (Meyer and Rowan, 1977), not only as functional entities but also as symbols "invested with socially shared meaning" (Tolbert and Zucker, 1986: 117). Gioia et al.,

(1994: 365) show more broadly how "symbolic action is central to the institutional legitimacy" of change. In the context of privatisation, such symbolic reinforcement might include a structural form but also actors' experience and involvement regarding organisational rites (Trice and Beyer, 1984) and narratives (Barry and Elmes, 1997). The extent to which actors relate to such symbols as signifying public or private sector templates will be a likely influence on their preference with one or the other.

This section has argued that privatisation can be seen as a process of deinstitutionalising public sector templates and institutionalising private sector templates. Viewing privatisation through an institutional lens highlights the inherent changing institutional templates. This section has conceptualised privatisation as institutional change.

# 7.2.3 The system of water in France

The aim of this section is to illustrate the management of water in France because in the researcher's view it has one of the 'best' water industries in the world in terms of management. It begins with a brief overview of who is responsible for water and expose two options for water management.

Water has become a major political point for local communities since the management of water resources has a direct effect on fundamental issues such as public health and economic development (Anonymous, 1994). Water supply and sanitation drainage is also receiving increased attention as a result, due to heightened concern for the environment. Water authorities are all too aware that the cost of improved water quality and uninterrupted service must remain affordable to the user. Full costs are almost always borne by consumers on the basis of their actual consumption.

In France, the national government is responsible for creating basic rules and specifying essential sanitary control measures. Local authorities are responsible for water services (i.e. drinking water supply, wastewater and rainwater drainage). For a total of 36,763 communes (local authorities), there are 15,244 separate supply systems and 11,992 drainage systems. Regional government authorities only step in on matters concerning investment scheduling and funding. France is divided into six-water catchment areas. Each catchment area has its own water authority, which receives taxes on all water

drawn and discharged in the catchment area in question. These taxes are re-distributed to regional authorities in the form of subsidies for investments aimed at bringing improvements in drinking water quality, supply lines, and wastewater treatment.

Local authorities have no actual legal obligations governing the supply of drinking water supply, but all of them do in practice. The main regulations governing the supply of drinking water are contained in the Public Health Regulations and the Water Act of 1992. Local authority obligations also cover wastewater treatment, removal of residual sludge and supervision of non-collective treatment systems. The local water authorities are required to take responsibility for organising their own water services. French law on decentralisation also sets up an independent organisation to govern financial and budgetary aspects of local authority spending.

Local authorities can choose either to run their water services directly or to commission this task from a private sector. If the direct management option is chosen, the local authority takes on full responsibility for the water services. Staff working for the municipal water services will have public sector employment contracts with local authority status. About 40% of French communes choose the direct management option for drinking water supply systems. This option allows the local authority to develop inhouse technical competence, which is useful when developing partnerships with private operators. Furthermore, direct management is also cheaper for the consumer, though this is gained only the expense of flexibility, especially with regard to staffing.

If the direct management option is chosen to commission all or some part of the tasks involved from a private operator, this sort of management is covered by long term contracts that also specify the cost of water supply and permissible range of variation. This system has been practised in France since the 20th. The system works in one of two ways: leasing and concession.

With a leasing arrangement, the local authority bears the initial investment costs and commissions management tasks from a private operator over a period of 5-20 years. The operator is paid for water supplied but must also pay the local authority rental on the plant it uses. The local authority uses this rental revenue to repay the technical and financial costs on the initial investment.

. ′
With a concession arrangement, the private operator builds and runs the plant itself. Concession contracts are between 25-50 years and when the concession terminates, the plant becomes the property of the local authority. In France in 1994, concession arrangements represent 75% in water supply and 35% in wastewater treatment. To finance long term investments for concession contracts, private operators often sign "BOT" (build, operate, transfer) partnership agreements with local authorities. This enables private and public operators to share financial risks and mobilise substantial bank loans.

#### 7.2.4 The water system in Egypt

The aim of this section is to expose the current situation in Egypt, to explain who is responsible for the management of water and to explain the history of water planning and to highlight some characteristics of water resource planning in Egypt.

Dr. Abu-Zeid, chairman of Egypt's Water Research Centre points out that "satisfying future demands in Egypt depends on better utilisation and efficient use of present water resources. Optimal water management is an essential prerequisite for sustainable development of Egypt".

The future looks bleak if Egypt does not succeed in formulating and implementing a water policy, which can match the limited fresh supply with the increasing demand. Per capita water resources are expected to drop from a current value of about  $922 m^3$  per year (1990) to about  $337 m^3$  per year in 2025. If the present management practice and cropping patterns prevail, this could mean that up to 60 % of agricultural land will not be irrigated.

The system of irrigation in Egypt is tremendous in its size and complexity. It consists of the Aswan High Dam, eight main barrages, approximately 30,000 KM. of public canals, 17,000 KM. of public drains, 80000 KM. of private canals (mesqas) and farm drains, 450,000 private water-lifting devices (sakias or pumps), 22,000 public water-control structures, and 670 large public pumping stations for irrigation. Throughout this system, approximately 59 billion  $m^3$  of water is distributed annually, not only for cultivated

land, but also for municipal and industrial use, for generation of hydro-electricity and for the navigation of freighters and tourist boats on the Nile (Hvidt, 1998).

The Ministry of Irrigation, formerly called the Ministry of Public Works and Water Resources (MPWWR), is responsible for national water resources and is the only body to authorise use of water from the Nile, canals, drains, and groundwater resources. The ministry also has control over works built to discharge water into canals, drain, and the Nile. MPWWR is authorised to impose penalties if its orders are not obeyed. A number of other ministries are involved in water management and use, including agriculture and land reclamation, health, tourism, power, transportation, industry, housing and reconstruction.

The Ministry of Agriculture and Land Reclamation has special responsibilities because agriculture consumes around 85% of water produced. Prior to 1992, when cropping patterns were liberalised, the Ministry of Agriculture and Land Reclamation decided, in consultation with the Ministry of Industry, which crops were to be grown in which localities. Such planning was undertaken a year in advance. From this exercise, the Ministry requested specific volumes of water to deliver to each canal and each branch canal. Following the liberalisation of the cropping pattern, however, it is not known precisely how the water allocation takes place. The Ministry of Health, which also holds special responsibilities, is authorised to close portable water supply works if the water produced does not meet the required standards. The Ministry of Health is further responsible for drafting quality standards for various water uses and for discharges of waste water.

Three committees have been formed to ensure co-ordination among agencies involved in water resources. Two of them, the Supreme Committee of the Nile, headed by the minister of the MPWWR, and the Co-ordinating Committee for Land Reclamation meet monthly to direct and review different developments plans, as well as to resolve conflicts between ministries. The third committee is called the Inter-Ministerial Committee on Water Planning (ICWP) and was established in 1977 as a part of the Master Water Plan project. ICWP is, as the name indicates, a cross-ministerial committee with a strict focus on planning. It has been given the responsibility to set planning assumptions and review development plans. In Egypt, water planning is said to have started in 1933 when a policy was formulated to use the additional storage capacity made available by the second heightening of the old Aswan High Dam (AHD) and the Gabal El-Awlia Dam in Sudan. This plan introduced programmes for land-reclamation, conversion of some basin irrigation to perennial irrigation and increases in the areas under rice cultivation. This policy was first revised in 1974 and again in 1975 when a new plan was drafted to accommodate the extra volumes of water resulting from the erection of the Aswan High Dam (AHD).

In 1981, the first attempt was made to create a master plan for all water use in Egypt. It was carried out in the early 1980s under the auspices of UNDP and the International Bank for Reconstruction and Development (IBRD). The Minister of Irrigation at the time pointed out the objective to be achieved by this effort: "Because of this increasing competition for water and limited availability, it was imperative to introduce new scientific techniques, and to use mathematical models to design future plans for water development, and to ensure efficient use of this resource." The resulting plan, the "Arab Republic of Egypt Master Plan for Water Resources Development and Use" is, however, not the plan as such, but a first step in a process which is intended to lead to improved planning capabilities within the sector. The main objective of the plan is to implement planning tools (i.e., to establish data bases and build flow models) which will make it possible to plan the development and use of water resources with greater precision in the future and, thus, to guide investment decisions.

The aims of this planning effort were: establishment of a data base; the setting-up of specific planning tools including six models which were built and implemented; establishment of three planning scenarios for new land development and determination of an economic rate of return and returns to water for ranking land-development projects. Hvidt (1998) points out that the results obtained from these planning tools are to be seen more as trials than final results since, due to a lack of data, the assumptions built into the models need to be replaced with real values when they are determined. The plan established three scenarios for future water demand and supply. Three main variables were analysed in detail: the supply, through implementation of water conservation projects, i.e., the Jonglei I and II project in Sudan, the demand for all user-groups; and the resulting potential expansion of agriculture on new land. The plan also

established priorities for the satisfaction of water needs. Firstly, the plan stated that water demand for municipal, industrial, navigation and spills should be satisfied and secondly water use on the old agricultural land should be satisfied, and thirdly the remaining volumes of water could be used to satisfy water demand for land reclamation.

The latest update of Egyptian water policy dates from June 1990. It is a ten-year plan covering the period 1990 to 2000. This plan was made in response to several events which, in the late 1980s; altered previous assumptions and had a direct impact on water planning in Egypt. The main events are as follows:

(1) The 1979-88 drought period during which the Nile flows yielded 99 billion  $m^3$  less water than expected into Lake Nasser and reduced the reservoir to a critical minimum of 6.8 billion  $m^3$  by July 1988

(2) Cessation in 1983 of construction works on the Jonglei Canal because of the civil war in southern Sudan, a project which would have provided Egypt with extra 2 billion  $m^3$  per year

(3) Revitalisation of the land-reclamation programme planned to reclaim 60700 hectares annually, requiring one million  $m^3$  of additional water each year. In other words, less water than anticipated would been available to satisfy needs.

The very critical water situation in 1988 opened the eyes of Egypt's water planners. Despite the AHD, Egypt was still vulnerable to the Nile flows and thus the "need for rationalisation and reductions in water use emerged as common agreed possibilities to face the coming unknown" (Hvidt, 1998: 3).

Hvidt, (1998) in his research illustrates eight characteristics from his analysis of water resource planning in Egypt as follows:

(a) Shift from water abundance to water deficit

(b) Importance of international co-operation

(c) Supply bias

(d) Environmental concern

(e) Lack of data

(f) Established priority to non-agriculture uses of water

(g) Delayed implementation

(h) An uncertain administrative framework for water resource planning

The general problems of the water industry in Egypt and the GOGCWS<sup>132</sup> can be divided into three kinds: (a) compliance and management of demand for water, (b) the management and (c) the finance and financial resources<sup>133</sup>. The problem of compliance and management of demand for water arose from different reasons such as:

(a) The population: the number of residents in Cairo is 13 million, and in addition, there will be three million visitors per day (16 million) by 2005. The following example should be enough to imagine the huge increase in demand for water. In 1968 the population in Giza was 610,000 and today it is one of the 50 biggest in the world, bigger than Paris and Berlin (the GOGCWS, no date source given, a: 17). The population in Shoubra El-Khama city was 185,000 in 1968, in 1986 was 700,000 and today it is the fourth biggest city in Egypt (Egyptian Geographic Association, no date source given: 2). This will potentially change the focus from cost management of supply to cost management of demand

(b) Reinvestment for renewal and replacement of the old dragnets (which will eliminate of water losses valued at 250 million E. P., Ministry of Planning: 10) should be equivalent with establishing new stations

(c) Problems with water meters i.e. there are not meters enough due to breakdowns (The GOGCWS, no date source given, a: 42)

With regard to the problem of management, in 1968, 85% of the population (i.e. customers) were centralised alongside the east of the River Nile. Between 1976-1996 the density of the population decreased in the centre of Cairo by 8%. In 1996 the customer percentage in the centre of Cairo decreased to 67%. This means the population who will be provided with water are a great distance from the River Nile. In the past, the GOGCWS faced technical problems relating to the connection of the services, however today the GOGCWS faces the planning of capital problems, finance problems and customer services. The question raised here is how can opportunity cost play a role in the water industry? (see section 10.1.2).

<sup>&</sup>lt;sup>132</sup> The case study of this research.

<sup>&</sup>lt;sup>133</sup> For the problem of finance and financial resources (see section 8.6.4).

#### 7.2.5 Water industry and institutional perspective

According to old institutional economic theory, there are three different types of change processes (Burns and Scapens, 2000) as follows:

- (a) Formal versus informal change
- (b) Revolutionary versus evolutionary change
- (c) Regressive versus progressive change

(a) Formal change occurs by conscious design, usually through the introduction of new rules and/or through the actions of a powerful individual or group (Rutherford, 1994) as in the case of the UK water industry. In 1989, the ten Regional Water Authorities of England and Wales were privatised. The Regional Water Authorities were a product of a major re-organisation of the water industry in 1973.

Informal change occurs at a more tacit level; for example, as new routines adapt over time to changing operating conditions. Also, informal change has occurred in the UK water industry (Yorkshire Water Company). In 2000, after 11 years of privatisation in the UK water industry, the Finance and Commercial Manager of The Water Business Unit of the Yorkshire Water company stated that " we follow full absorption costing - we are not using activity based costing but we consider some ABC principles".

It would probably be reasonable to expect that formal cost management change (e.g. the implementation of new techniques such as ABC, ABM, etc) will be more straight forward than attempting to change the ways of thinking which are embedded in existing cost management routines. This expectation was confirmed when the Chief of Commitment (Engagement) Department (Balance Sheet) of the GOGCWS in Egypt stated that " There have been no significant changes in the cost method and if there were any small changes, these changes were built on the comments from The Accounting Central Institution. I think if there are any plans for change, it is necessary to get support for this plan from the state".

At a practical level, it seems reasonable to expect that hierarchical (i.e. imposed) cost management change will have its initial and most direct impact on the formal rules (i.e. the technical aspects) on cost management systems, but only an indirect impact on the informal processes which underpin cost management routines (for example, the

changing of the of the economic forms and legal structure of the GOGCWS during their life, see section 8.2.1). Whereas, bottom-up change (initiated by organisational members who use cost management from one day to the next) is more likely to have an impact at a tacit level and to shape informal as well as formal cost management processes.

(b) Revolutionary change involves a fundamental disruption to existing routines and institutions (for example, the privatisation of the water industry in UK, - see sections 7.2.1 and 7.2.2). Evolutionary change is incremental with only minor disruption to existing routines and institutions (for example the GOGCWS in Egypt). It was clear from section 7.2.1, a brief history of the GOGCWS, that the economic forms and legal structure of the company were changed during their life starting as a "foreign company", changing to a general institution and then transferring to a stock company and finally an economic general organisation. The GOGCWS was an economic organisation but it aimed to achieve balance between expenditures and revenues and not to make capital profit like the private sector.

(c) Regressive and progressive. Tool (1993) distinguished between what he called "Ceremonial" behaviour and "instrumental" behaviour. Ceremonial behaviour emerges from a values system which discriminates between human beings and preserves an existing power structure; whereas instrumental behaviour emerges from a values system which applies the best available knowledge and technology to problems and seeks to enhance relationships.

Institutional change in the wider economy, especially deregulation and increased competition in the UK water industry set a series of changes in organisation and cost management practices that are still being used by worker. Privatisation of the UK water industry in 1989 and increased competition in the 1990s had forced UK water companies to become more explicitly profitable and cost conscious (Ogden, 1995).

The literature advocates that government agencies faced with market pressures and requirements to recover all expenditures with revenues will become more business like and cost conscious, and will make greater use of cost accounting data in order to control costs and prevent funding shortfalls and associated sanctions (Osborne and Gaebler, 1992; Gore, 1993).

The contingency literature provides considerable theoretical and empirical support for this claim. Zimmerman (1976), for example, develops an analytical model which suggests that non-profit organisations facing funding uncertainty will place greater emphasis on budgets and costs to avoid the sanctions associated with deficit spending. Cooper's (1995) theory of cost system choice argues that intensified competition increases the benefits from more complex cost accounting systems since competitors will be more likely to take advantage of poor decisions arising from inadequate cost information. Similarly, the contingency framework developed by Gordon and Miller (1976) maintains that organisations facing greater environmental uncertainty, due to factors such as competition, require more extensive, frequent and detailed financial and non-financial information and relatively sophisticated cost accounting and control systems.

Empirical studies provide support for the institutional theory and empirical theory. Khandwalla (1972), for example, found that the level of competition was positively associated with the use of more sophisticated cost accounting practices such as standard costing, incremental costing and flexible budgeting, while Gordon and Narayanan (1984) and others supported the hypothesis that greater environmental uncertainty increased the demand for decision-making information.

With an institutional realm<sup>134</sup>, it seems that a highly regulated set of institutions outside the water industry in Egypt were mirrored by institutions and routines within the industry that made these organisations bywords for stability and conservatism.

According to neo-institutional economic theory (transaction cost)<sup>135</sup>, there are three models of governance: (1) markets (2) hybrids and (3) hierarchies or internalisation

<sup>&</sup>lt;sup>134</sup> The Burns and Scapens (2000) model invites its users to identify three elements: (a) organisational routines and habits (b) an institutional realm and (c) a realm of action

<sup>&</sup>lt;sup>135</sup> The central aim of the studies of transaction costs is to explain why some transactions are more likely to be executed within one form of organisation, whereas other transactions tend to be associated with different organisational models. The main thrust of transaction cost is that a specific institutional arrangement is chosen to govern a specific transaction because that arrangements offer some distinctive set of control devices to the control needs of that transaction (Spekle', 2001).

(Williamson, 1975; 1979; 1985; 1996). These alternative governance structures differ in the control mechanisms they employ to safeguard contract execution and to achieve successful adaptation. Hierarchical governance attains control primarily by means of authority, internal incentive structures and monitoring, for example, the management of the water industry in Egypt (see section 7.2.4). Market governance derives control from free competition, for example the management of the water industry in the UK (see sections 7.2.1 and 7.2.2). The hybrid form of governance is typically based on explicit, long-term contracts in conjunction with additional safeguards to assure compliance, for example, the management of the water industry in France (see section 7.2.3).

In conclusion, the researcher can see that there are differences between Egypt, the U.K. and France - from a neo-institutional economic theory (transaction cost) perspective - as shown in table 7.1.

The key differences	Egypt	England (U.K.)	France	
Privatisation or	is still public	transferred from	a mix of privatisation and	
regulation		public to private <sup>136</sup>	regulation	
From institutional	Egypt water is	UK water is	French water is hybrid	
theory perspective	hierarchical	provided by highly		
		regulated market		

Table 7.1: Key differences between Egypt, U.K and France

Furthermore, in institutional theory literature, in terms identified by DiMaggio and Powell (1983), the researcher can relate Egypt to institutional isomorphism and the UK to competitive isomorphism. Isomorphism is a process that causes one unit in a population to resemble other units in the population which face the same set of environmental conditions. Because of isomorphic pressures, organisations will become increasingly homogeneous within given domains and conform to expectations of the wider institutional environment (Carpenter and Feroz, 2001).

DiMaggio and Powell (1983) identify two types of isomorphism: competitive and institutional. Competitive isomorphism is not of interest for this study because it relates primarily to free and open market competition scenarios, and therefore, is not applicable

<sup>&</sup>lt;sup>136</sup> In the researcher's view, the UK water industry is mixed between privatisation and regulation in real terms. The researcher based this view on the role of the OFWAT (see the role of OFWAT in 7.2.2.3).

to the analysis of public utility (sector) organisations. In this study, the focus on the concept of institutional isomorphism which relates to organisational competition for political power, social fitness, and institutional legitimacy. DiMaggio and Powell (1983: 150) state that the concept of institutional isomorphism is a useful tool for understanding the politics and ceremony that pervade modern organisations and contain a number of exogenous pressures that influence their structure and practices (this is the case of the GOGCWS in Egypt, see section 8.6).

Moreover, in terms identified by Brunsson (1994), the researcher observed that the water industry in Egypt relates to "political-ization" (see section 11.2.4) and the water industry in the UK to "company-ization". Company-ization means that an organisation which originally possessed a strong political element, assumes certain features deriving from the company's institution. For example, a state or local government authority begins to regard the people in its environment as customers, or these people begin to consider themselves as such (Brunsson, 1994).

## 7.3 Issues of costs in water industry

The aim of this section is to illustrate cost systems and some problems in water industry, e.g. how people pay for their water requirements, how the cost of water is calculated and whether this system is fair or a new system needed.

#### 7.3.1 Cost system in water industry and some issues

In the U.K., the development of effective cost systems have been handicapped by the absence of suitable output measures for most of the industry's activities. Very detailed analysis of expenditure is provided but comparison is normally with the expenditure budget rather than on a unit cost basis.

The industry has put a great deal of effort into attempting to overcome this problem and internally a number of comparisons are made between broadly similar areas of activities on the basis of the most suitable output measure available. Collectively, regional water authorities have commissioned a comparison of revenue collection costs, using for the most part the number of consumers as the "output measure". Similar studies in other areas are under way.

An alternative approach is to direct greater attention to budget preparation. The objectives of each area of activity are defined, the procedures determined and necessary resources identified on a "Zero based" approach. This type of exercise also provides a basis for determining priorities between different areas of activity.

It is likely that regional water authorities will continue to invest substantial resources in the improvement of this system. Apart from the obvious need to use resources as effectively as possible, regional water authorities are under pressure to open up more of their activities to competition, as local authorities have been required to do for their direct labour organisation. There is, therefore, a need to make proper comparison between external and internal provision.

In the U. K., in 1974, regional water authorities faced the problem of changing from the local government type of accounting operated by their predecessors to a more commercial style. The main changes (Meredith, 1985) were:

(a) Accrual of both capital and revenue expenditure in annual accounts

(b) The use of a single pool of finance in place of separate capital and revenue finance

(c) The use of straight-line depreciation in place of loan redemption provisions

It was necessary to construct the accounting system in such a way as to enable appropriate charges to be levied to cover the costs incurred on each service, avoiding cross-subsidisation between services.

In 1974 costs of fixed assets were usually available only to the extent that loan debt was transferred and there was no complete information regarding assets financed from revenue or where loans had been fully repaid. This was not a problem of major significance - while depreciation was calculated on historical costs only, some approximation to the normal straight-line depreciation charge could be calculated. With the introduction of current cost accounting in 1981, the matter became of far greater significance. Inventories of above ground assets could be prepared but the majority of water authorities' expenditure is on below ground assets (water mains and sewers) and the best estimates of these had to be prepared according to whatever information was available. Valuations were generally on the basis of nationally agreed formulae. Work will be necessary for some years in order to improve the accuracy of fixed asset valuations.

Regional water authority internal accounts, therefore, are mainly designed to measure performance against an annual expenditure budget. The budget is prepared on a basis of cost centres, each representing an area of activity for which an identifiable person is responsible, and will be broken down into accounting periods against which actual expenditure can be compared. The original budget is prepared on the basis of price rulings at a given date, and a separate estimate is made for price changes. There are also procedures for amending budgets during the year where necessary.

Authorities also carry out a substantial "contracting" business usually described as "rechargeable work". Charges for regular work, for instance providing connections to new properties, are usually charged according to fixed price schedules, based on distance and other factors affecting costs, but other work is charged at actual cost. The accounting system must therefore include provision for "job costing" of these activities.

There are lots of issues in cost systems in the water industry such as, cost allocation. Cost allocation is used for many purposes, e.g. requirements in manufacturing for external reporting. Why? Because the allocation of depreciation cost is of major importance, cost allocation acts as a guide to decision taking within the firm, for example pricing decisions. Cost allocation is demanded

(a) To categorise the firm's overall production control problems into more manageable chunks

(b) To sustain a system of incentives to encourage efficient production

(c) To co-ordinate the activities of separate units

(d) To provide information relevant to pricing

The cost allocation problem would be trivial in a world with the following properties: (a) no joint or common cost (b) marginal and average cost equal (c) instantaneous adjustment of all inputs (d) complete information. In practice, however, none of these conditions applies. Most utilities are based upon substantial joint and common costs.

A firm freed from competitive pressure may be able or be required to choose other objectives than simple profit maximisation. These may include the setting of "fair" prices and attainment of aims such as universal service. Fairness can be defined in

various ways, several of which may involve the sharing of some particular cost or benefit among competing products according to particular rules. Universal service will typically involve elements of cross-subsidy and an averaging of charges across consumers with different costs.

The questions which costing procedures are intended to answer include:

How should prices be related to costs, in order to attain aims not related to efficiency while achieving a target level of overall revenue or profitability?

What degree of cross-subsidy is present in the existing set of prices? But this assumes that the need for regulation is clearly established. For the regulator, the decision typically comes in two stages: (a) whether competition should be permitted (b) what price controls need to be imposed to ensure that it is conducted fairly. This situation raises a new set of issues;

Do the cost characteristics of particular products or subsets of products make them natural monopolies?

Will competition jeopardise universal services through "cream skimming"? If competition is fostered and emerges in markets for some products but not in others How can the incidence of cross-subsidisation of products be identified and prevented, in cases where a firm produces both types of outputs? (Cave and Mills, 1992).

In the context of cost allocation used for regulation, one of the main dimensions of competitive rivalry is the degree of joint supply, or bundling of services, which often occurs. There are many cost allocation procedures one of which is fully distributed costs<sup>137</sup>. Beyond this simple approach the four main candidates for establishing the cost share are in accordance with

(a) Relative output

(b) Gross revenue

(c) Net revenue

(d) Attributed costs

<sup>&</sup>lt;sup>137</sup> To determine fully the cost of each service. Each service takes attributes cost, which is assigned to it, and a share of the common costs. The fully distributed cost of service I (FDCi) can be written as: Fully distributed cost service of I = Attribute cost of I + Share of common cost of service of I.

Unfortunately, the various fully distributed cost methods give different results, so the economist's criticisms of fully distributed cost have been strong; Kahn and Shew (1987: 207) forcefully attack the use of conventional rather than a causal basis for cost allocation: "The only costs that have objective reality are ones that describe a causal relationship between the act of purchase and their incurrence. Cost allocations that are not grounded in causality have no basis in objective reality - they have no independent meaning to the prices they are suppose to justify, except in some ritualistic, incantation sense".

There is no attempt to incorporate marginal cost or promote economic efficiency. As Baumol writes: "There is obviously not the slightest reason to expect that the prices emerging from a full-costing process will bear the slightest resemblance to those known to be necessary for efficiency in resource utilisation" (Baumol 1983: 181).

One of the regulatory purposes of cost allocation is to establish a fair set of charges which can be imposed upon customers in a way which is consistent with satisfactory incentives for the water industry.

The Director General of the UK's water service stated in a discussion paper entitled "Paying for water" (OFWAT, 1990) "If charges are to be broadly related to costs, in order to avoid under discrimination and provide proper incentives, the basis for the allocation of costs needs to be determined. In setting in their tariffs, companies will need to decide how to allocate cost between services and customers, and the proportion of costs should be recovered by means of capital contributions standing charges, where charges are related directly to use".

In the same paper, the Director General talked about problems facing the water industry in England and Wales. The water industry has faced major difficulties because most domestic customers pay for their water on a tariff which is related to rateable value of their dwellings while the majority of business customers are metered. Cairo (the capital of Egypt) has a similar problem. OFWAT's view on the accounting cost should be matched as far as possible to economic costs (for example, via current cost accounting), and suggests that the following structure might provide an appropriate balance (see table 7.2).

Capital cost of connection to system	Capital contributions connection plus				
and enhancement of local network	charges plus requisitions				
Average capital cost of enhancement	Use- related charge				
of general infrastructure					
of general infrastructure					
Capital cost of enhancement of general	Capital contributions				
infrastructure is above average	(infrastructure charges)				
Operating costs related to number of	Standing charge				
consumers (billing, meter reading)					
	Liss related shows				
Other operating costs of water supply	Use- related charge				
and fuel drainage					
Other operating costs of surface	Use- Related charge or (banded) standing				
drainage	charge				
-					

Table 7.2: Accounting and economic costs in the water industry

(OFWAT, 1990, Annexe 4, page 8).

In the case of water tariffs for metered domestic customers, OFWAT argued that the correct principle was to distinguish between costs which related to the number of customers connected, and those which related to the volume of water supplied. Costs such as billing and meter reading should be recovered through a service charge, while capital-related and operating costs of reservoirs, etc, which are generally related to the amount of water used, should be recovered through the volumetric charge.

#### 7.3.2 How do the people pay for the water?

The basis on which water bills are estimated in England and Wales is "rateable value". The growing minority who have had water meters installed pay on a very different basis. Their bills have two elements: a small and uniform fixed charge, plus a variable amount proportional to their water usage (the "volumetric" charge). Water metering can be justified on both economic and environmental grounds, as it introduces financial incentives for efficiency among customers and companies. This is illustrated in Figure 7.3.







Objections of the currently favoured system of water charges based on rateable value (Kenway, 1999) include;

(a) Rateable values were last set in 1973, so are more than a quarter of a century out of date. As they are no longer used for local taxation, it would be prohibitively expensive to carry out a new valuation for water alone.

(b) A charging system that is completely independent of the amount of water used makes it impossible to encourage efficiency or economy via financial incentives;

(c) Bills based on rateable values are at odds with the way the water industry regulator has chosen to interpret "fairness" in water charging. The view of regulator is that fairness means "charges should broadly reflect the cost of providing water and sewerage services to different classes of customers". From this point of view, two households using the same amount of water should pay the same for it. Under rateable value, the two households pay different amounts if they live in different-sized houses.

Cost reflectiveness has much to be said for it. It is the basis on which almost all prices are set, regardless of whether a customer is rich or poor. Yet it is not the basis on which we pay for our public services via taxation, and an attempt to introduce such a principle into local government ten years ago was decisively defeated. Most of us still automatically think of water as the public service it was until privatisation.

How much should be paid for each extra unit? The economist's answer would be the "forward-looking long-run marginal cost" of providing that extra unit. Short-run marginal costs would be the cost of providing the extra unit on the assumption that the necessary physical infrastructure (reservoirs, pumping stations, the network of distribution pipes and so on) is already in place. Long-run marginal costs do not take this infrastructure as given but include an amount that reflects the cost of providing it. Forward-looking long-run marginal costs assume that providing additional water may be more costly than providing the current supply if, for example, new reservoirs have to be built in less advantageous locations. Long-run marginal cost is a financial incentive for the customer and the company. The customer, in deciding whether to consume more water, pays a price that reflects the additional costs of supplying it. The company, benefiting from the extra revenue generated by higher water usage, will make the right investment decisions.

(

It is not as easy to get agreement on what the numerical value should be. But while there is genuine room for disagreement, there is one reason why we should err on the side of caution when setting the unit price of water: if the price is too high, the water companies will make undue profits from each extra unit sold. It would no longer be in their interests to encourage the rest of us to be economical with water (for example, via promotions of new more water-efficient appliances such as washing machines), since this would cause their profits to fall. But such a switch would be politically difficult, chiefly because it would create many losers at all income levels, whose opposition might be weightier politically than the support of many winners.

229

Hitherto the debate about how to price water intended to polarise the question of water metering, with the strongest opposition coming from those predominantly concerned about social welfare and justice. Their concerns are compelling: for reasons both of public health and private well-being people should not face financial incentives to economise on water for essential needs such as drinking, preparing and cooking foods, washing and bathing, and flushing the toilet.

Kenway (1999) suggests new system for charging for water. This new system focuses on metering, in the sense of measuring how much each household uses. But what really matters is the tariff, which is then built around the information that metering yields, to determine how much each household has to pay. The heart of objection to metering is that it creates a financial distinctive for people to use any amount of water, including the water they ought to be using for health and hygiene. Figure 7.4 shows the new proposed system.



Figure 7.4: A fairer alternative

Source: Kenway, (1999)- "New Stateman Privatised Water: Who is paying the price?, p. vii

Here, the amount a household pays rises above the fixed charge once consumption exceeds a predetermined level (often, though inaccurately, called the "tranche of free water"). Only then would any financial disincentive to use water kick in. The volume of

water covered by a fixed charge could vary between households, so those with more people would have a proportionally larger tranche of free water. And the amount payable in fixed charge could vary between households according to their liability for council tax, thereby preserving an important element of payment according to means. This certainly is a more complicated system, but it satisfies economic, social justice and environmental objectives.

#### 7.3.3 How to calculate the cost of water?

In England, between 1990 and 1995 average prices rose in real terms by 5.5% a year and by 1.5% a year between 1995 and 2000. The main drivers were the need to improve drinking water quality and environmental protection in response to European legislation, and the need to eliminate capital maintenance backlogs that had built up before 1989.

The regulator fixes a price-increase cap (known as the K factor) separately for each of the 18 water companies and the ten combined water and sewerage companies 'RPI +  $K^{138}$ '(Summerton, 1999; OFWAT, 1991). Initial K values were set for 10 years by the secretaries of State for the Environment and for Wales in 1989. The Licence requires Ks to be completely reset after 10 years at a "periodic review". A review may also take place after only five years if either company or the Director feels it is necessary. The Director took the earliest opportunity available to him to reset K values in July 1994. The new price limits take effect for 10 years from 1 April 1995.

The question arising here is, does K cover all company charges? No. K is the price limit for a "basket of charges" (OFWAT, 1991). These are; charges for unmeasured water supply, charges for measured water supply, charges for unmeasured sewerage services, charges for measured sewerage services, charges for reception, the treatment and disposal of trade effluent. Charges for any of these particular items may be more or less than RPI + K. However, several charges are not covered by the K formula, for example

<sup>&</sup>lt;sup>138</sup> PRI "The Retail Price": to be used in the formula is the PRI in the November prior to the start of the April 1 charging year.

K is a charging limit. It represents the amount by which average charges can rise in any one year. It is set individually for each company year. It reflects the provision services to customers, including financing its capital expenditure programme.

Example: if K is 1 and the RPI increase in the 12 months to previous November is 1.5% then the average charges increase can be no more than 2.5%

infrastructure charges. Limits on other charges, e.g. connection charges, are not laid down, but OFWAT would expect these charges to be reasonably related to costs.

The K factor indicates the percentage figures by which the water companies may raise their prices in each of the five years. For the customer, the K factor is modified by two other factors (Summerton, 1999): Firstly, inflation or deflation: for any given year the change in the retail price index is added. Secondly, there may be changes resulting from the rebalancing of charges between different types of customer.

OFWAT is responsible for ensuring that each company's K continues to reflect the assumptions made when Ks were first set. Under certain limited circumstances the Director can adjust K without reference to the Secretary of State. Condition B of the Licence allows the Director to make, in any year, adjustments to a K factor for certain "relevant changes of circumstance" where they exceed, in total, 10% of turnover. For example: changes in the obligations placed on the companies; failure to achieve a legal obligation or a service standard; non-receipt of sale of surplus land assumed at the periodic review (OFWAT, 1991). Any interim reference notice must be issued by 1 October and a determination made by OFWAT within three months. If a company is not happy with the outcome it can appeal to the Monopolies and Mergers Commission.

Individual K factors are related to what the Director considers companies would need to spend to fulfil their statutory obligation. They are ceilings on what companies can charge, and are based on the assumption that companies can achieve substantial increases in efficiency (OFWAT, 1992). There are some factors, which have a crucial effect on the K factor (Summerton, 1999):

(a) The cost of capital: the water regulator has become increasingly tough about the rate of return he will assume in making his decision, given the low-risk nature of the water business

(b) Efficiency: the water regulator believes improvements of 3 to 4 % a year are feasible between 2000 and 2005, with all the benefit enjoyed in the form of lower prices

(c) Maintenance and replacement of networks and equipment: the water regulator believes existing levels of expenditure are sufficient to preserve the status quo. Some companies' assessments indicate that higher rates of capital maintenance and infrastructure renewals are needed to prevent a return to the deterioration of the 1970s and 1980s

(d) Comparative quality of service: the regulator proposes to reward the most effective companies by giving them more scope to increase their prices and, conversely, to penalise the least effective with less scope. The principle is to try to ape a competitive market in which profit gravitates to the companies that customers perceive to be offering the best value for money. The "least effective" companies argue that the regulator's ways of measuring quality of service are too blunt for the job

(e) Efforts to establish customers' views and to consult interest groups have been much more extensive. But there is a question of whether the regulator prefers the views of the OFWAT Customer Service Committees (which he appoints) to market-research evidence that customers would be content with a smaller price cut or more stable prices, as long as the extra income is spent on service and environmental improvements

(f) There is the interaction between metering and company income. The UK's government has decided to make water meters voluntary and prohibit disconnection as the ultimate deterrent against "won't pays"

OFWAT ensures customers get a fair deal via the companies which have limited scope for increasing their dividends, unless they are able to increase their efficiency above what is allowed for in price limits (OFWAT, 1992). OFWAT (1994) divides the role of The Director of Water Services into two categories, primary duties and secondary duties as follows:

(a) The Director's primary duties: He is required to act in a way he considers is best calculated to ensure that: the financial of a water and sewerage company, as specified in the Act, are properly carried out. Companies are able to finance their functions, in particular by securing a reasonable rate of return on their capital.

(b) The Director's Secondary duties:

(i) Protecting customers: "customer's bills should, in general terms, reflect the costs which the customer imposes on water and sewerage systems for a supply of clean water, disposal of dirty water and draining of surface water. Other aspects of customer's interests are protected, including quality of service and benefits from the sale of land transferred to the companies at privatisation, or acquired since then"

(ii) Promoting economy and efficiency: "the Director must encourage companies to operate efficiently. He monitors standards, compares performance between companies, against each company's targets and with other sectors. The Director must give companies incentives to reduce operating costs and the cost of sustaining their assets, raise money cheaply, and complete environmental projects at low cost"

(iii) Competition: "the Director has a duty to facilitate competition between suppliers and potential suppliers, ensuring that a framework exists in which competitions can develop. He encouraged the Government to change the law, widening the scope for competition"

(iv) General Environmental duty: "the Director is required to exercise his power subject to achieve (among other things) "the future of conservation, enhancement of flora, fauna and geological or physiographical (landscape) features of special interest"

(v) Limiting charges: "the Director sets price caps, which allow the companies to finance their functions. This gives companies the incentive for greater efficiency. The Director does not control profits or dividends. The Director limits the annual price increase, or "K" factor, for each company to reflect what it needs to charge to finance the provision of services to customers"

# 7.4 Conclusion

This chapter highlights the importance of water as one of the most important public utilities. Furthermore, it ensures that the water industry is an extreme monopoly even if it is privatised. It raises the argument of privatisation or regulation in the water industry and it agrees that no one can give a clear answer on privatisation or regulation being the right decision for the water industry by looking to three countries having three different systems (England is a private "market", Egypt is a public "hierarchy" and France a mixture of privatisation and regulation "hybrids").

This chapter has given attention to the new tool of cost management in an indirect way, which is that of privatisation<sup>139</sup>. This is clear when the vocabulary of profit replaced the previous vocabulary of cost in the UK Water industry<sup>140</sup>. Furthermore, it has been stressed that management of water is a major political point. In addition, this chapter

<sup>&</sup>lt;sup>139</sup> See section 7.2.

<sup>&</sup>lt;sup>140</sup> When UK Water Authorities had to transform themselves from public sector to private sector.

has raised the issue of centralisation vs. decentralisation and this become clear by comparison between the management of water in France<sup>141</sup> and Egypt<sup>142</sup>.

This chapter raises some essential points about deficiencies in water utilities such as paying for water on a rateable value for domestic use and on metered value in the majority of business use in Egypt and England. The independence between price and cost prevents efficiency, in other words, there is a gap between accounting and economic theory and this leads the researcher to ask what is the role of accounting and why would accounting be useful in this situation? Furthermore, it highlights the dimension of social elements in the water industry, when explaining the objection to the new system which depends on meters. It exposes the dry future in Egypt and gives a brief history of water planning in Egypt (the case study of this research) which will be discussed in the next chapter.

<sup>&</sup>lt;sup>141</sup> In France, local authorities can choose either to run their water services directly or to commission this task from a private sector.

<sup>&</sup>lt;sup>142</sup> In Egypt, The Ministry of Public Works and Water Resources is responsible for national water resources.

# **Chapter 8**

# Cost management in the GOGCWS

# **8.0 Introduction**

As mentioned in sections 5.4 and 2.1, while technical (or contingent) explanations of cost management were not rejected, they were viewed as incomplete (Scott, 1987). This led the researcher to look at the institutional context within and outside the GOGCWS (see figure 5.1). The aim of this chapter is to describe the accounting system used in the "General Organisation for Greater Cairo Water Supply" (GOGCWS), and its supervision and control body. It explores the question "which model of cost management was followed in the GOGCWS" and " what methods of cost accounting were used in the Egyptian water industry". The chapter examines the perceptions of the GOGCWS's managers of the organisation structure, cost accounting and cost management in the GOGCWS. The rationale for investigating the perceptions of the GOGCWS's managers is largely due to the argument in section 6.2.1.3. This argument is based on the premises that a single case study (such as the study of the GOGCWS) can yield valuable information about the topic under investigation by carrying out comparative study within the case organisation itself (e.g. the perceptions of employees in different departments). The chapter also explores the question "is there any difference or similarity between the rules and routines<sup>143</sup> related to the cost accounting system that is followed by the GOGCWS".

For this purpose, this chapter is divided into eight sections. The first section gives a brief discussion of the legal framework for public utilities in Egypt generally, with emphasis on the water industry. The second section describes the General Organisation for Greater Cairo Water Supply. It is opened by a brief history of the GOGCWS to highlight the change in the economic forms and the legal structure of the organisation. It also looks at the roles and the structure of the GOGCWS.

<sup>&</sup>lt;sup>143</sup> The term rules and routines used in the context as Burns and Scapens (2000) identified rules comprise the accounting systems as set out in the procedure manuals, whereas routines are the accounting practices in use.

The third section examines the organisational structure of the GOGCWS in reality and identifies it with institutional theory. The fourth section gives a brief discussion of the supervision and control body of the "General Organisation for Greater Cairo Water Supply" whether internal (Department of the inspection, financial and managerial control, etc), external (the Cairo region and the Accounting Central Institution, etc) and financial or technical control.

The fifth section discusses the accounting system in the GOGCWS. It highlights the objectives, the notions and the contents of The Unified Accounting System<sup>144</sup>. It studies the general standard followed in preparing the Unified Accounting System and its accounting transactions. This section outlines the role of the cost accounting system in the GOGCWS.

The sixth section aims to explain the methods of cost accounting used in the Egyptian water industry. In other words, this section gives a brief summary of the cost accounting system's background in the GOGCWS; its description, cost recognition and measurement and reporting procedures. Furthermore, this section has looked at whether there are any differences or similarities between the rules (documentations) and routines (practices) in this cost accounting system.

The seventh section aims to explore the question "which model of cost management was followed by the GOGCWS". In other words, this section looks at "whether there was any use or potential use of throughput accounting, benchmarking, programme planning and budgetary systems, activity based costing, activity based management, strategic management accounting, etc in the GOGCWS. The section also discusses alternatives to cost management as suggested by consultants Black and Veatch International. It also discusses perceptions of the GOGCWS managers' towards the issue of cost management.

The final section (eight) highlights the conclusion of this chapter.

<sup>&</sup>lt;sup>144</sup> The accounting system which is followed in GOGCWS.

# 8.1 The Legal framework for public utilities in Egypt

The water public utility in Egypt was centrally controlled<sup>145</sup>, but varies widely in institutional organisations. The largest city, Cairo, was designated through presidential decree as "General Organisation" with a corporate structure and regional responsibility. It was, however, subject to the control of both the governor and the Ministry of Housing and Public Utilities (MHPU). Most local utility organisations were administered through regions.

While public utilities in Egypt were managed, staffed and operated on a local or regional level, there was an excessive amount of central control (see section 8.4: the supervision and control body on the GOGCWS), lack of qualified technical support and inadequate budgeting by the GOE (Government of Egypt) ministries. Local utility organisations lack sufficient powers to make management, financial and personnel decisions (Cooper, 1992). Local authority and popular councils frequently complain about poor communication and poor service from the GOGCWS organisation and therefore often object to proposals for tariff increases or payment for central agency services (Cooper, 1992).

The water sector in Egypt was a nationalised programme. All policy making, planning, budgeting and administration for utilities in the country was carried out under the Ministry of Housing and Public Utilities. Each utility was required to use a centrally-mandated cost accounting method (see section 8.6) which was manual rather than computerised. The Ministry of Finance also seconded accountants to GOGCWS to ensure compliance with the government requirements.

Civil service rules prevented managers from discharging employees except for criminal acts. Wage and salary levels were too low<sup>146</sup> to attract and keep good operators. No career development programmes had been developed. There was little incentive for operators to be productive or to work efficiently.

<sup>&</sup>lt;sup>145</sup> See section 8.4.

<sup>&</sup>lt;sup>146</sup> Comparing with the private sector.

Public utilities in Egypt were generally established as public authorities. The President of the Republic issued the decisions necessary for creating, empowering and organising public utilities and public departments (Act 146 of the Constitution). Under Law 43 (the law of Local Government System) units of local administration establish and manage all utilities under their jurisdiction with the exception of national public utilities or those of a special nature as designated by the President. Executive regulations of the Ministry of Local Administration defined local utilities which Governors might establish and operate. Governors had authority over all public utilities within their region. A Governor had the authority of a Minister regarding review and approval of all actions by a local utility and its board of directors. This was also a provision of article 1 of law 43/1979 regarding juristic powers of local government.

A public utility, by definition, was an enterprise created by a government agency, for the implementation of a common interest/purpose such as satisfaction of the needs of the people in compliance with the principle of equality of users (Cooper Report, 1992)<sup>147</sup>.

Public authorities were established by Presidential decree under law No. 61/1963. The General Authority had the right to contract, undertake execution of work, and act in order to achieve the goals for which it was established. The authority was represented by the Chairman of its Board of Directors in all relationships with other entities and in court. A General Authority could be merged with other authorities or be cancelled by Presidential decree only.

A General Authority was usually established to manage a public utility. All their assets, monies and properties were considered to be public. All laws and regulations (see section A1.2.1) concerning public domain applied to them, unless stated otherwise in the enabling decree. A General Authority was permitted its own budget, based on any conditions in the Presidential decree. The budget was prepared by the Chairman and

<sup>&</sup>lt;sup>147</sup> The researcher does not have this resource but found two pages inside the GOGCWS discussing the organisations legal background and used information from the end of these two pages (source: Cooper Report- 15 October 1992).

approved by the Board of Directors. The central agency for accounting supervised<sup>148</sup> all budget and accounting activities.

An authority could establish its own internal regulations without having to abide by government regulations, and could write regulations concerning the hiring of its staff, its workers, their promotion, dismissal, salaries, benefits and pensions. If delegation of these matters was not included in the presidential decree, the authority must abide by government regulations concerning employment. It should be noted that ministerial review and supervision was limited to general and decision-making affairs, and not to the daily workings of a General Authority.

The water utility in Cairo was created as a public authority under law No. 61/1963. It operated directly under the local authority and had autonomy to operate independently, except that tariffs were approved by the ministries. Annual funding was through the State budget and personnel policies were strictly regulated by the region (local authority).

# 8.2 The description of the GOGCWS

# 8.2.1 A brief history of the GOGCWS

- On 17 May 1865, the joint-stock company<sup>149</sup> of Cairo Water was given a mandate for providing water to Cairo City for 99 years.
- On the first of July 1957 a law was introduced (No. 145/ 1957) to nationalise the joint-stock company of Cairo Water and establish Cairo Water Service which answered to the municipal council of Cairo City.
- In January 1958, Gizera water and Helwan joined the Cairo Water Service.
- In January 1964, Amada water was brought under the management of the Cairo Water Service.
- On 28/11/1965 the president of Egypt issued the order (No. 4417/ 1965) which transferred the management of the Cairo Water Service to an Egyptian stock company called "Greater Cairo Water" which answered to The General Egyptian Institution for Services Work.

<sup>&</sup>lt;sup>148</sup> The Accounting Central Institution was responsible for controlling and auditing the revenues and expenses (see section 8.4.2.6).

<sup>&</sup>lt;sup>149</sup> This company was managed by England (foreign company).

- On 30/11/1968, the decision of the Egyptian president was issued (No. 1683) to establish " General Organisation for Greater Cairo Water Supply" which answered to the Ministry of Housing.
- On 17/10/1971 the president of Egypt issued order (No. 2420) which declared that the General Organisation for Greater Cairo Water Supply had to answer to the city's region.
- On 31/10/1991, the Prime Minister of Egypt issued an order (1655/1991) to include Maser Elgededa water in the General Organisation for Greater Cairo Water Supply.

From the previous brief history, it was clear that the economic forms and legal structure of the GOGCWS were changed during their life starting as a foreign company, becoming a general institution, being transferred to a stock company and becoming finally an economic general organisation according to the order of the Prime Minister (No. 1039/1979). The GOGCWS was an economic organisation but it aimed to achieve a balance between expenditures and revenues, and not to make capital profit like the private sector.

## 8.2.2 The role (activities) of the GOGCWS

The General Organisation for Greater Cairo Water Supply was responsible for managing water utilities in the cities of Cairo, Kalubia and Giza and its related organisations (and was also responsible for operation, and maintenance for utilities which were answerable to the organisation). The Presidential Decree No. 1938 of 1968; concerning establishing the General Organisation for Greater Cairo Water Supply, stated that the organisation was responsible to establish, operate, and manage the water station in order to achieve its aims. Furthermore, the GOGCWS produced the following types of water:

(a) Fresh water<sup>150</sup> (see section A1.2.2)

(b) Industrial water<sup>151</sup>

<sup>&</sup>lt;sup>150</sup> The GOGCWS established water stations which were necessary to provide fresh water for Greater Cairo (around 25% of the population in Egypt) and any new areas in the scope of Greater Cairo whether for domestic use or service use. These included worship centres, charitable associations, bakeries, popularity squares, young peoples care centres, embassies, sports, clubs, unions and party centres. In addition it provided water to the governmental sector, public business sector, companies, commercial shops and different institutions and organisations.

<sup>&</sup>lt;sup>151</sup> The GOGCWS produced water which was not suitable for drinking and used it in industry. This water was less clean than fresh water and the production of it depended on the clarification method. There were several kinds of industrial water, which depended on the amount of chlorinate added.

## (c) Irrigation waters<sup>152</sup>

The main structure of the GOGCWS could be summarised in figure 8.1.



Figure 8.1: The Structure of the GOGCWS

Source: Based on the GOGCWS (no date source given, c), p. 7

# 8.3 The organisation structure of the GOGCWS

The organisation structure consisted of a Chairman and two Vice Chairmen. Each of these were answerable to their central and general administrations - and a group of department were answerable directly to the chairman (see the organisation structure of the GOGCWS<sup>153</sup> in section A1.2.3).

The organisation had a hierarchical structure. The base of the organisation consisted of departments, then administrations, general administrations, central administrations (for the function of the board of directors and central administration of the GOGCWS, see section A1.2.3), etc.

<sup>&</sup>lt;sup>152</sup> The least clarified water was used in garden irrigation.

<sup>&</sup>lt;sup>153</sup> The organisation structure was taken from the administration of the organisation and management at the GOGCWS.

The investigation of this research has focused on the perceptions of the GOGCWS managers<sup>154</sup>. The sample study included all managerial levels (high or senior managerial level "H", medium or middle managerial level "M" and low or junior managerial level "L"). The basis for this classification is the job of the respondents<sup>155</sup>. The Chairman, the two Vice Chairmen and each chief of administration are classified as high level managers. This represents 20% of the sample (10 respondents). Each chief of a department and member of staff working in administration are classified as middle level managers. This represents 26% of the sample (13 respondents). The staff who work at the operating levels are classified as low level managers. This represents 54% of the sample (27 respondents)

For an illustration of *how the GOGCWS is organised*: see table 8.1

	Managerial level			
Organisation of the GOGCWS	High	Medium	Low	Total
According to the nature of the production process	10%	12%	16%	38%
According to the nature of geographic areas		2%	14%	16%
According to the nature of the market served		2%		2%
Others	10%	10%	24%	44%
Total	20%	26%	54%	100%

Table 8.1: Perceived organisation of the GOGCWS (N = 50)

Source: Analysis of questionnaire.

The researcher saw these different answers were due to the complexity of the water environment (see the GOGCWS's law and decrease in section A1.2.1.1 and figure A1.6), which created different perceptions between different managerial levels of the GOGCWS. For example, staff who work in the calculations of revenues and costs for each region of Cairo tended to choose "the nature of geographic areas" as the basis of the existing organisation structure of the GOGCWS. However, there is even a different perception among the same managerial level of the GOGCWS. For example, half of the high managerial level chose "according to the production process" and the other half chose "others" which is a combination of production process and geographic areas.

<sup>&</sup>lt;sup>154</sup> For the rationale for investigating the managers perceptions of the GOGCWS, see sections 6.2.13 and 8.0.

<sup>&</sup>lt;sup>155</sup> For employee position (name and job of people who filled in the questionnaire) see table A2.1.

Similarly, differences exist among the medium managerial level, 12% chose "nature of production process", while 10% chose "other" with 2% choosing "nature of geographic areas" and "nature of market served" respectively.

For a description of the control structure at the GOGCWS how do you describe the control structure at your company/division? see table 8.2

	Managerial level				
<b>Control structure</b>	High	Medium	Low	Total	
Investment centre		4%		4%	
Profit centre			10%	10%	
Contribution centre		2%		2%	
Cost centre	16%	8%	30%	54%	
Others	4%	12%	14%	30%	
Total	20%	26%	54%	100%	

 Table 8.2: Perceived control structure of the GOGCWS (N = 50)

Source: Analysis of questionnaire.

This was not unexpected result - just over half of the respondents (54%) saw the control structure as the cost centre - because the Unified Accounting System divided the production into Production Centre, Productive Services Centres, Marketing Services Centres, and Administrative and Financial Services Centres. So this caused the control structure to be focused on cost. There were two kinds of control structure at the GOGCWS; external control (95% of respondents mentioned this control - observation from the interviews) and internal control<sup>156</sup> (only 73% of respondents mentioned it - observation from the interviews). This came from high pressure for the GOGCWS to satisfy legislative requirements and led to the question "is the accounting system designed to achieve efficiency and optimisation for the company primarily or to satisfy legislative requirements primarily"? (see chapter 9). Furthermore, the researcher did not observe economic efficiency or 'pareto optimal' but there were efficiency controls or effective controls and pareto improvements in the organisation<sup>157</sup>.

<sup>&</sup>lt;sup>156</sup> The role of internal control is too limited, it is checked only on signature, but this research would like the internal control to be able to discover the mistakes which the external control did not discover (Accountant in Central Accounts Department).

<sup>&</sup>lt;sup>157</sup> Accountant in Participating Accounts Administration said there was a procedure of control and no practice control at the same time.

Regarding the question how does the control structure fit within the organisation structure? see table 8.3

How Control structure fits within	Managerial level			Total
the Organisation structure	High	Medium	Low	-
A good fit	2%	4%	14%	20%
Reasonable fit	10%	12%	30%	52%
Problematic fit	8%	10%	10%	28%
Total	20%	26%	54%	100%

Table 8.3: Perception of how control structure fits within the organisationstructure (N = 50)

Source: Analysis of questionnaire.

It can be observed from table 8.3 above, that 72% of respondents indicated that there was a "reasonable or good" fit between the control structure and organisation structure. However, just over half of the respondent's - observation from the interview - said it was not reasonable. This findings<sup>158</sup> led the researcher to ask the interviewees why the control structure did not fit with the organisation structure of the GOGCWS. The interviewees stated that there were several explanations for the unsuitability of the control structure, such as; "the structure is built on favouritism"<sup>159</sup>, "is unchangeable"<sup>160</sup>, "executive command is bad"<sup>161</sup>, "the structure is bureaucratic"<sup>162</sup>, "is centralised"<sup>163</sup> and "like a pyramid"<sup>164</sup>, "the problem is abuse of power"<sup>165</sup>.

Regarding "change in the organisation structure", the majority of respondents (90%) - observation from the interviews - stated there had been no substantial changes in the

<sup>161</sup>Chief of Bills Auditing and Stamp Department.

- <sup>164</sup> Chief of Documentary Credit Department.
- <sup>165</sup> Auditing (Maturities Department).

<sup>&</sup>lt;sup>158</sup> The answer from the interviewees.

<sup>&</sup>lt;sup>159</sup> Accountant in participating Accountants Administration.

<sup>&</sup>lt;sup>160</sup> Accountant (Central Accounts Department).

<sup>&</sup>lt;sup>162</sup> Accountant in Cost Department.

<sup>&</sup>lt;sup>163</sup> Accountant (Documentary Credit Department)

organisation structure but there had been developments in the technical and legal aspect, although no developments regarding the financial and administration aspect except for very small changes such as; establishing new administrations and raising new managers.

There was a plan to change the organisation structure in the GOGCWS but nothing materialised<sup>166</sup>. The GOGCWS suggested a new organisation structure<sup>167</sup> and sent it to the Accounting Central Institution but the Accounting Central Institution did not acknowledge it<sup>168</sup>. One accountant in the Central Department stated "if we want to change we should change the people not the structure". The researcher saw this answer as meaning it was extremely important to change the belief (attitude) of the worker as well as the structure. Furthermore, this raised the issue that organisation structural arrangements were as likely to be the outcomes of political processes as were organisational resource allocation decisions" (Pfeffer, 1978).

## 8.4 The supervision and control body on the GOGCWS

The General Organisation for Greater Cairo Water Supply followed several control procedures. These procedures could be internal or external and also within these either technical or financial.

## 8.4.1 The internal control

## **8.4.1.1** The technical control (examination of production quality)

The general administration for laboratories and research of GOGCWS supervised the technical aspect of all the organisation and its workers. Each laboratory in the stations supervised and controlled the water at all stages of water purification; the subsidiary stations and also each laboratory controlled the dragnets, which were provided by the station.

On that basis, the laboratory's station was responsible for controlling water treatment operations to ensure that the water produced was suitable for drinking, and following it to dragnets until reached the customer, according to health standards.

<sup>&</sup>lt;sup>166</sup> Accountant (participating accounts Administration).

<sup>&</sup>lt;sup>167</sup> See the new organisation structure of the GOGCWS in figure A1.5.2.

<sup>&</sup>lt;sup>168</sup> Chief of Final Statement of Accounts and Follow-up.

The control water treatment operations included chemical and biological examination of raw, filtered and treated water (every two hours). In addition, several searches and studies, which had no fixed system or programme, were carried out according to the circumstances of the work.

Regarding water quality control, each of the GOGCWS's stations had a laboratory which carried out examinations of water treatment. These examinations are represented in figure 8.2



**Figure 8.2: Water Quality Control** 

Source: Based on the GOGCWS, (no date source given, c), p. 12

In addition, the GOGCWS had the biggest central laboratory in the Middle East - the capital cost of which was 250 million Egyptian pounds. The laboratory consisted of three sub-laboratories to measure all kinds of pollution and to provide immediate treatment.

Figure 8.3: Water quality control



#### 8.4.1.2 The financial control

The inspection of the financial and managerial administration was considered the main system for internal control of the GOGCWS. It gave direction and reduced the variances<sup>169</sup>. Its work included all the GOGCWS's activities (technical, financial and managerial). In general the functions of the inspection of financial and managerial administration were as follows (Accounting Central Institution, no date source given):

- Setting internal audit rules for financial and administrative research and follow up
  procedures to ensure the efficiency of workers' performance in production and
  services, follow up of achievements, discovery of mistakes and the difficulties
  which face the GOGCWS. Setting internal audit rules show incapabilities in
  carrying out the legal regulations, suggest solutions and show the different variance
  to reduce these immediately.
- Making random periodical inspections of activities and works' administrations of the GOGCWS and preparing the necessary reports (which include the results of research) to send to the chiefs of these administrations. After approval of these reports from the chief of the committee, the recommendations and suggestions in the reports will be carried out.

# 8.4.1.2.1 The performance of the inspection of financial and managerial administration's functions

Although the Department of the Inspection of Financial and Managerial Control was considered the fundamental control system inside the GOGCWS, and was responsible for ad-hoc inspection in all areas and administration of the GOGCWS, it had not received due recognition from the GOGCWS' administration (Accounting Central Institution, 1994: p. 61). This led to a lack of factors could have helped the administration to achieve its aims. In addition, there were some obstructions and difficulties which faced the inspection of financial and managerial administration to achieve internal control. These difficulties could be summarised as follows (Accounting Central Institution, 1994: 62):

- The lack of technical human ability in administrative areas (the inspection of financial and managerial administration).

- The absence of due care by the GOGCWS's administration.

- The lack of suitable means of transportation for administration workers.

<sup>&</sup>lt;sup>169</sup> In term of activity not the variance occurred from procedures.
- The absence of collaboration between the GOGCWS administrations.

#### 8.4.2 The external control

The GOGCWS followed technical supervision and control from the general administration for central laboratories within the Health Ministry, and it also followed financial control from several institutions. The style of external control may be illustrated as follows:

## 8.4.2.1 The technical control (examination of quality production)

The external control on water production in the GOGCWS was through the central administration within the Health Ministry which did chemical analysis and bacteriological analysis. The health supervisors took and analysed samples of treated water.

#### 8.4.2.2 The financial control

There were several institutions which were responsible for financial control on the General Organisation for Greater Cairo Water Supply namely:

- The Cairo region.
- Finance Ministry (treasury department).
- Ministry of planning.
- The Accounting Central Institution.

Every institution had its role and worked separately from the others. The role of each can be summarised as follows (Accounting Central Institution, no date source given):

### 8.4.2.2.1 The Cairo region

The General Organisation for Greater Cairo Water Supply was one of many organisations which had to answer to the Cairo region in the administrative affairs and workers affairs of the organisation. In addition, the Governor of Cairo City determined the price<sup>170</sup> of water sold by the GOGCWS.

<sup>&</sup>lt;sup>170</sup> The researcher made great effort to get an appointment with the Governor of Cairo City in order to obtain information of how the price of water was set. However, he was unable to gain access which limited the research.

#### **8.4.2.2.2** The Finance Ministry (before buying)

A Finance Ministry delegate<sup>171</sup> was responsible for controlling purchasing before buying (allowing or not allowing to spend), control and supervision of the budget of GOGCWS, controlling all the accounting and calculation procedures and ensuring that it was identical to the laws, rules and procedures of the GOGCWS. The GOGCWS gave accounts to the Finance Ministry - within a time period determined by the Treasury Ministry, which was monthly, quarterly, and annually according to the rules, leaflets, and periodic publications which were issued by the Finance Ministry.

#### 8.4.2.2.3 Ministry of Planning

The Ministry of Planning confirmed the medium term (five years) plan and the annual plan for the GOGCWS. In addition, it gave approval to the annual assignment of funds which the GOGCWS asked for in relation to all items in the budget.

## 8.4.2.2.4 The Accounting Central Institution

The Accounting Central Institution was responsible for controlling purchasing after buying supplies for the GOGCWS. It was responsible for controlling and auditing the revenues and expenses. The most important elements that were audited are as follows:

- The control of salaries, bonus and wages<sup>172</sup>.

- The control of general expenditures<sup>173</sup>.

- The control of investment expenses<sup>174</sup>.

<sup>173</sup> By seeing the projected budget and its distribution to the branches, different administrations examined the system used to distribute these funds. The internal system and rules which determined the expenditure were checked to ensure that all expenditure was according to the rules. Ensured that all expenditure documents relating to rents, travelling, petty expenses were correct.

<sup>174</sup> Ensuring that the procedures relating to expenditures were correct and documentation relating to these was available by looking at the plan. Ensuring that expenditures were made according to the GOGCWS's plan and furthermore that expenditures were limited to the approved expenditure and not exceeded. Finding out why the limit was exceeded if this did happen.

<sup>&</sup>lt;sup>171</sup> This person works for the GOGCWS but reports to the Finance Ministry.

<sup>&</sup>lt;sup>172</sup> The control of salaries, bonus and wages were controlled through seeing the balance sheet of the GOGCWS and also identifying the number and level of seniority by personnel records kept and compared between expenditure documents - salaries, bonus and wages - with personnel and salary records. Auditing of the internal control for wages and salaries was carried out. Employees files were checked and decisions made relating to appointment, bonus, leave without salaries and suggestions for avoiding mistakes in the future.

- The control of the organisation's revenues<sup>1/2</sup>.

- The control of financial statements<sup>176</sup>.

# 8.5 The Accounting system in the GOGCWS

The main development of accounting in Egypt can be divided into two stages: bookkeeping and accounting stages. In the bookkeeping stage, the double entry notion was followed in records, which were represented in an original "journal" connection with the subsidiary journal. Different kinds of business were recorded and carried over to the ledgers general and subsidiary which included classification for these entries to personal and named accounts. Thompson (1994: 51) argues that towards the end of the 15th century, double-entry book-keeping was part of a "larger project to re-emphasise a belief in order sanctified by god". Then financial statements and the balance sheet was derived from the personal and named accounts.

The Unified Accounting System - the system which was followed in the GOGCWS - was established at this stage. In the accounting stage, the accounting information - which the economic organisation<sup>177</sup> used to meet their aims - was connected with the requirements of external institutions.

#### 8.5.1 Objectives, notion and contents of the Unified Accounting System

In spite of the Unified Accounting System meeting certain purposes within the organisation, it was not conceivable<sup>178</sup> that it could meet all the requirements of

<sup>&</sup>lt;sup>175</sup> Examining the internal auditing system, studying laws, ministerial decisions, rules relating to the revenues of the GOGCWS. Examining the collection system, evaluating it and suggesting any modifications. Comparing the actual collected revenues with the revenues which should be collected and studying the reasons for failure and suggesting solutions for that.

<sup>&</sup>lt;sup>176</sup> Ensuring that financial statements and balance sheets were prepared according to the rules issued by the Finance Ministry.

<sup>&</sup>lt;sup>177</sup> The Unified Accounting System was applied on

<sup>(1)</sup> The economic organisation within the public sector (with exception, by law, of banks, credit companies and insurance companies). The definition of the economic organisation, which had to apply the Unified Accounting System, was that "the unity which is practising industrial activity, business activity, agriculture activity, landed activity or any activity had economic characteristics"

<sup>(2)</sup> The general organisation or institution which had to, by law, prepare the balance sheet as a business balance sheet even if it did not have any economic activity

The economic organisation in the public sector took different kinds of forms such as (a) general institution or organisation (b) public sector company (c) co-operative society or (d) subsidiary company

<sup>&</sup>lt;sup>178</sup> See section 8.6.

planning and following up the execution of financial, economic and technical control (The GOGCWS, no date source given, b: 4). The objectives of the Unified Accounting System could be summarised as follows: (a) Providing the essential information and analytic tools which were a requirement for planning, execution, and controlling at different levels<sup>179</sup> (b) It provided the link between the accounts of the economic organisation and the national accounts. The Unified Accounting System combined the terminology and concepts between the account of the economic organisation and the result (c) It simplified the collection of accounting information and its classification and therefore recording it was easier.

The notion and contents of the Unified Accounting System are illustrated in figure 8.4.

<sup>&</sup>lt;sup>179</sup> The level of organisation: The Unified Accounting System provided information, which helped in control, supervision and direction. In addition it participated in planning.

The level of external institution: The Unified Accounting System helped the Ministry of planning to achieve integration between the budget in kind (the budget in kind contained the goods requirements, the service requirements and workers. It was called the budget in kind because it expressed by quantities before translating it to cash values), the financial budget, and the cash budget with the period horizons (short, medium, and long term) which were consistent with national plans.

#### Figure 8.4: Contents of the Unified Accounting System

**Unified Accounting System** 



The Unified Accounting System was built on general standards. These were:

(a) The simplicity<sup>181</sup>, clarity<sup>182</sup> and flexibility<sup>183</sup> standards

(b) Honouring of the accounting rules and principles standards<sup>184</sup> (see section A1.2.4.1)

<sup>180</sup> See section 8.6

<sup>181</sup> Simplicity. It concentrated on financial accounting.

<sup>182</sup> Clarity. It gave a detailed explanation of the Accounting index (see section A1.2.4.2) which was considered as the main resource information.

<sup>183</sup> Flexibility. It used special records to record any details which were work requirements.

<sup>184</sup> These were considered the basis of records in the assets evaluation and goods inventory (the cost of fixed assets were represented by all the capital expenditures for purchasing and preparing the fixed asset to work. There was a difference between the new fixed assets purchased and the old fixed assets purchased) and depreciation (the depreciation was calculated annually whether the result of the activity was loss or profit. The straight-line method was used in the cost accounting system in the GOGCWS with the exception that the usage related method was used for stationary, etc. in the end of each financial year). These principles and rules were approved by the convention. So, the preparation of the Unified Accounting System was built on existing rules and fundamentals.

(c) The verified standards of implementation<sup>103</sup>

(d) Satisfied all the requirements which were created from inside or outside of the economic organisation.

The Unified Accounting System also gave freedom of choice regarding the manner in which documents were circulated within the organisation. It depended on several conditions such as the nature of activity, the size of the organisation and the kind of information required. In addition, the Unified Accounting System gave some examples of records or guidance which were not obligatory.

## **8.6 Cost accounting System**

The aims of cost accounting in the GOGCWS could be summarised from the documentation of the organisation as follows (the GOGCWS, no date source given, c): (1) Perfect control of production costs and the uses of production factors

(2) Providing the right basis to evaluate the finished production, uncompleted production and the work under implementation at the end of financial period

(3) Providing accounting information according to the planning budget (responsibility centres and cost centres)

(4) Helping to set policy and decision-making procedures at all levels (from the economic organisation to the national level)

(5) The cost accounting system recorded the actual  $\cos^{186}$  relating to production and costs according to the centres and stages.

The aims of cost accounting in the GOGCWS were compared with the aims of cost accounting which were mentioned in section 5.1.2. It was observed that the aims of cost accounting (from a theoretical viewpoint) could be summarised in three ways: (a) stock valuation (b) planning control and (c) decision-making (Clark, 1923; Drury and

<sup>&</sup>lt;sup>185</sup> It gave a chance for the economic organisation to study the system project before it was issued. All economic organisations were obliged to apply the Unified Accounting System from the beginning of year 67/68 with the possibility of a one year delay for any economic organisation that requested it, but on condition it had the permission of The Accounting Central Institution.

<sup>&</sup>lt;sup>186</sup> Actual cost means in this context historical cost. The calculation of the cost of water per cubic meter was as follows: costs were calculated for each centres separately - The Unified Accounting System in the GOGCWS divided the production into: (a) Production Centres (b) Productive Services Centres (c) Marketing Services Centres (d) Administrative and Financial Services Centres - which depended on the actual (historical) cost. The cost of each centre was added together in order to get the total cost and finally the total cost on the water production was divided in order to obtain the cost per cubic meter.

Dugdale, 1992; Coward, 1944; Frank, 1990; Cooper and Kaplan 1991; 1992). These were similar to the aims of cost accounting in the GOGCWS. For example, the heart of the first, third and fifth aims of cost accounting in the GOGCWS were planning and control. The focal point of the second and fifth aims were stock valuation and the core idea of the fourth aim was decision-making. On the other hand, it was observed there was a limitation in the aims of cost accounting in the GOGCWS because it depended on historical costs relating to traditional management accounting (for further explanation of this point, see section 9.2.1.1). Furthermore, this suggested that cost information had no role or a very limited role in shaping the strategy of the GOGCWS (see section 9.2.1.3).

Although the records of cost accounting systems in the GOGCWS depended on historical costs, which were useful for knowing the direct and indirect actual costs for the production unit and controlling and monitoring of performance. Furthermore, the record of GOGCWS was divided into four main groups as follows: (1) Records for financial accounts (2) Records data for financial accounts (3) Records value group relating to inventory control (4) Records related to costs. This research concentrated on the records related to cost (a brief summary of its contents as given in section A1.2.4.3). This section is divided into seven sub-sections as follows:

### 8.6.1 Cost accounting development and the needs

For an illustration of the development of cost accounting - *how was the cost accounting system developed?* - see table 8.4

Development of cost	N			
accounting system	High	Medium	Low	Total
In-house	14%	22%	22%	58%
Contractor	6%	2%	26%	34%
Software package				
Customer prepared				
Borrowed from another agency				
Others		2%	6%	8%
Total	20%	26%	54%	100%

Table 8.4: Development methods of the cost accounting system (N =50)

Source: Analysis of questionnaire.

From table 8.4 above, it could be observed that 58% of the respondents chose "inhouse" as a method of development of the cost accounting system. This result is consistent with the analysis of the documentation of the GOGCWS. The UAS (which includes the cost accounting system, see figure 8.4) was imposed on the GOGCWS by government regulation (see section 8.5).

On the topic of what standards are used for your cost accounting activity?, see table 8.5

	Ma			
Standards of cost accounting	High	Medium	Low	Total
Agency standards	2%	2%	8%	12%
Government cost accounting standards	12%	8%	26%	46%
Generally accepted accounting principles				
Others	6%	16%	20%	42%
Total	20%	26%	54%	100%

Table 8.5: Standards used for cost accounting system (N =50)

Source: Analysis of questionnaire.

These answers highlighted the effect of legislation on cost practice (see section 9.2.2.3). They also raised the question of why the respondents answered differently. In order to explore this question, the percentage  $(42\%)^{187}$  of others was analysed, due to it being the second highest percentage, following the 46% which stated government cost accounting standards.

The question is this cost system adequate for your needs? is illustrated in table 8.6

187	<sup>87</sup> Effect of legislation on cost practice in the GOGCWS (N =21)									
Average	of	previous	Technical	Committees	within	There	is	no		
vears.			GOGCWS.			standard				

Source: Analysis of questionnaire.

26%

8%

4%

I do not know

4%

These answers - (26%) stated that the standards used for the GOGCWS were the average of previous years - and did not conflict with the answers of - 46%, which stated government cost accounting standards. It could have been that the standards used for cost accounting in the GOGCWS related to government cost accounting standards based on the average for the previous year.

Adequacy of the cost	N			
accounting system	High	Medium	Low	Total
No needs			4%	4%
Few needs	12%	4%	14%	30%
Most needs	8%	18%	22%	48%
All needs		4%	12%	16%
I do not know			2%	2%
Total	20%	26%	54%	100%

Table 8.6: Adequacy of the cost accounting system (N = 50)

Source: Analysis of questionnaire.

The analysis of adequacy of the cost accounting system above shows that the majority of respondents (48%) viewed the cost accounting system within the GOGCWS as adequate for most of their needs. It was noted that at senior managerial level 60% (12% / 20%) perceive the cost accounting system as adequate only for few needs. The senior managerial levels have more an overview of the GOGCWS and this could account for this result.

#### 8.6.2 Changes in cost management

On the topic of the cost method being revised in the GOGCWS "Has there been a major revision to the method used in the GOGCWS for estimating and measuring product cost?", see table 8.7

Table 8.7: Any revision of methods used for estimating and measuring product cost (N =50)

Yes	10%
No	86%
I do not know	4%

Source: Analysis of questionnaire.

This result was confirmed by the analysis of the documentation of the GOGCWS which showed that although the economic forms and legal structure organisation changed during the organisation's life (see section 8.2.1), there was no change in the cost method<sup>188</sup>. This respondent's answer raised the question "why was there no change?" (For more detail, see section 11.2.4). Unintentional changes to cost management in the GOGCWS "may occur in the absence of systems to monitor the execution of the routines where the rules and routines are not sufficiently understood or accepted by the actors" (Burns and Scapens, 2000: 10).

Regarding the question whether changes should be made in the cost department - observation from the interviews, 45% of interviewees could not comment or give any suggestions or advise about the cost department. Of the remaining interviewees, 55% highlighted some suggestions for the cost department such as:

(a) "Introduce computers"<sup>189</sup> ("technology"<sup>190</sup>)<sup>191</sup> and "excellent use of it"<sup>192</sup>

(b) "Design accurate planning before taking decisions"<sup>193</sup> and "make the connection and integration between administrations in order to decrease the cost"<sup>194</sup>

(c) "Evaluate the booking value"<sup>195</sup>, "choose skilled workers"<sup>196</sup> and "train them"<sup>197</sup>

(d) "Establish cost centres"<sup>198</sup> and "control of the cost element"<sup>199</sup>, "look for the fairest way in cost distribution"<sup>200</sup> and "give higher authority for the cost department"<sup>201</sup>, "it should be seen at the same level as the central administration"<sup>202</sup>

<sup>192</sup> Accountant in the Participating Accounts Department, Chief of the Bills Auditing and Stamp Department, Accountant in the Cost Department, Auditing and Accountant in the Cost Department.

<sup>193</sup> The chief of the Balance Sheet Administration.

<sup>&</sup>lt;sup>188</sup> The GOGCWS had followed the Unified Accounting System since the 1960s.

<sup>&</sup>lt;sup>189</sup> Accountant in the Cost Department, Accountant in the Participating Accounts Administration, and Accountant of the Statistics Department

<sup>&</sup>lt;sup>190</sup> Chief of Central Accounts Department and Chief of Balance Sheet Administration.

<sup>&</sup>lt;sup>191</sup> Observation from questionnaire, regarding the computer based configuration of the cost accounting system in the GOGCWS, 66% of respondents chose a non automated system. On the topic of the principal method of data entry, the majority of respondents 92% chose manual.

<sup>&</sup>lt;sup>194</sup> The chief of the Documentary Credit Department

<sup>&</sup>lt;sup>195</sup> Accountant in the deduction and Modifications Department

<sup>&</sup>lt;sup>196</sup> Accountant in the Participating Accounts Administration.

<sup>&</sup>lt;sup>197</sup> The chief of the Balance Sheet Administration.

<sup>&</sup>lt;sup>198</sup> Accountant in the Central Accounts Department. This was a very surprising result because it contradicted the analysis of the description of the control structure at the GOGCWS as mentioned previously. But the reason could be that, although the cost centre included in the rules (procedures) of the GOGCWS it disappeared from the routines (activities) of the organisation, and this goes back to the argument regarding the differences between rules and routines highlighted by Burns and Scapens (2000).

There were two observations from this result. Firstly, half of the interviewees (50%) realised that the cost method was not appropriate for the GOGCWS and this was the first step to change. So the question raised again was "*why was there no change?*" (See the answer in section 11.2.4). Secondly, the interviewees were very fearful<sup>203</sup> of talking against the policy of the government (the policy of the GOGCWS).

#### 8.6.3 Measuring and estimating product cost

For the present method of measuring product cost *How do you consider the present method of measuring product cost and practice?*, see table 8.8

	N.			
Measuring product cost	High	Medium	Low	Total
Straightforward and simple	6%	12%	28%	46%
seems a reasonable compromise	2%		6%	8%
Needs significant change	. 6%	4%	2%	12%
Need simplifying	6%	10%	16%	32%
Others			2%	2%
Total	20%	26%	54%	100%

Table 8.8: Perception of measuring product cost (N =50)

Source: Analysis of questionnaire.

Just under half of respondents 46%, chose straightforward and simple. Maybe the reason for this answer was that the cost accounting system depended on historical costs. Furthermore, the documentation of the GOGCWS (organisational rules of the GOGCWS) confirmed that the measuring of product costs depended on the actual (historical) costs. Moreover, the result showed that 44% of respondents stated that the

<sup>200</sup> The vice chairman for Technical Affairs.

<sup>201</sup> Accountant in the Cost Department.

<sup>202</sup> Accountant in the Cost Department.

<sup>203</sup> Most of the interviewees before they started the interview asked the researcher to keep the interview very confidential otherwise they were concerned they might be dismissed from their jobs.

In addition to that the interviewees or the respondents sometimes gave their own opinion and other times gave the view of the organisation and this could have created confusion.

<sup>&</sup>lt;sup>199</sup> Accountant in the Central Accounts Department

present method of measuring the product costs needed to change (whether simple or significant) and this answer achieved reliability when the researcher asked the respondents whether a cost system was adequate for their needs (see section 8.6.1). Regarding depreciation methods used in cost accounting system of the GOGCWS, *what depreciation methods are used in your cost accounting system*? The majority of respondents (98%) chose straight line. The perception of depreciation methods used in cost accounting systems of the GOGCWS were consistent with the actual method used.

On the topic of "measuring the cost of resources consumed", *how do you measure the cost of resources consumed*?, the majority of respondents (92%) stated according to actual cost. This answer was similar to the fifth aim of the cost accounting system of the rules (documentation) of the GOGCWS. So, this meant there was no difference between the rules (documentation) of the GOGCWS and the (routines) practices related to measuring the cost of resources based on historical cost.

### 8.6.4 Water Funding

The perception of competitive intensity of capital and financing for water in the GOGCWS is illustration in table 8.9.

Table 8.9: Perception	of competitive	intensity	of capital	and fina	ncing in	the
	GOGC	WS (N =50	0)			

			<b>Competitive Intensity</b>							
M	arket Conditions		1	2	3	4	5			
1	Availability of	Easy to obtain	36%	34%	10%	10%	10%	Hard to obtain		
	funds	funds						funds		
2	Source of funds	Many sources	14%	56%	10%	8%	12%	One major		
				204	н. На 1			source		

Source: Analysis of questionnaire.

<sup>&</sup>lt;sup>204</sup> Although there was only one major source (the government) for funding (subsidisation) of water in the GOGCWS, it could have been answered this way because there were many institutions and organisations - e.g. the National Investment Bank, The Finance Ministry, Ministry of Planning, The Cairo region, etc, - represented the government. Also, there were a variety of customers such as: domestic, business, public, etc.

This result showed the respondents' perception was that there was no difficulty in funding for water in the GOGCWS (observation from the questionnaire). Comparing this observation with the documentation of the GOGCWS, if the price of filtered water did not increase, the governmental subsidisation would be 737<sup>205</sup> million E.P. annually (during 1999- 2003) (The GOGCWS, no date source given, a: 17). The GOGCWS would borrow this amount of money from The National Investment Bank (at 14% interest) and the result being that subsidisation will be 840 million E.P. The question here is "can the government provide subsidisation for the new investments and replacement, and renew the present system at the same time?" If the answer to the question was no, this meant the government should look at different methods for managing the water industry in the GOGCWS (opportunity cost).

The residents of Cairo had the lowest priced water (The GOGCWS, no date source given, a: 17). This may seem to have been a wise policy but it had some effects. The low price did not provide the GOGCWS with the financial resources to run, maintain and renew the system in order to provide the quantities of good quality water demanded by customers.

In conclusion, it was noted that there was a big problem in the finance and financial resources in the GOGCWS. Furthermore, there was a big difference between what the managers thought about the funding of the water in the GOGCWS and in reality.

## 8.6.5 Cost information, reporting and decision-making

#### **8.6.5.1** Use of cost information

The question what statements describe how data from this cost accounting system are primarily used?, is illustrated in table 8.10

<sup>&</sup>lt;sup>205</sup> This means the government subsidised the GOGCWS by more than four and a quarter times their total revenue for 1999/2000 = 172,593,108.01 Egyptian pounds).

			I	Man	agei	ial I	Level	—,»	Total	
Using data from cost accounting		Yes %	6		No %	,	I do not know %			%
· · ·	H	M	L	Н	Μ	L	Н	Μ	L	
To manage an activity or a programme	14	12	20	6	14	32			2	100
To measure programme performance	10	12	22	10	14	26			6	100
To determine selling price			6	20	26	48				100
To value inventory	20	10	42		14	8		2	4	100
To value cost sales	20	10	42		14	8		2	4	100
To satisfy legislative requirements	20	24	52		2	2				100

Table 8.10: Using data from cost accounting (N =50)

Source: Analysis of questionnaire.

This was not unexpected result for the researcher - 72% of respondents stated the data was used to value inventory and the same percentage stated the data was used to value cost of sales. This was because it was similar to the second aim of the cost accounting system which was providing the right basis to evaluate the finished production, uncompleted production and the work under implementation at the end of financial period. The majority of respondents 96% stated the data was used to satisfy legislative requirements (see section 9.2.2.3).

The researcher took the result of the analysis of the questionnaire, which highlighted that the data created from cost accounting was not used to determine the selling price. Further, by asking the interviewees "who determines the price of the water", they stated the selling price was determined outside of the organisation. Furthermore, it was noted that Cairo's governor determined the price of water in Cairo City and every governor determined the selling price?" The question raised here was "why does the governor determine the selling price?" The reason might be the effect of legislation which stated that water utilities in Egypt were subject to the control of both the governor and the Ministry of Housing and Public Utilities (see section 8.1).

On the topic of *to what extent is cost accounting data used in budget formulation?*, see table 8.11

Using data from cost	N	Total		
accounting in budget	High	Medium	Low	
Usually the basis of estimation	20%	24%	46%	90%
Rarely used		2%	8%	10%
Total	20%	26%	54%	100%

Table 8.11: Using data from cost accounting in budget formulation (N =50)

Source: Analysis of questionnaire.

This answer was no different to the information in the documentation of the company, which said that the aim of cost accounting system was to provide accounting information according to the planning budget.

#### **8.6.5.2 Reporting of cost accounting**

The question for what purposes are reports generated from this cost accounting system?, is illustrated in table 8.12

	Managerial Level									Total
	Yes %				No %			I do not know %		
	Н	M	L	Н	Μ	L	Η	M	L	
Financial reporting	20	22	52		2	2		2		100
Developing prices	4	2	16	16	24	38				100
Inventory control	10	14	30	10	10	24		2		100
Managerial control	2	18	26	18	8	28				100
Budget preparation	18	26	50	2		4				100
Budget execution	6	10	16	14	16	38				100

 Table 8.12: Reporting purposes of cost accounting data (N = 50)

Source: Analysis of questionnaire, H =High level, M =Medium level, L =Low level.

This result (94% respondents stated yes for financial reporting) raised the question of whether the accounting system in the GOGCWS was designed to help financial accounting, management accounting or both. In addition, the majority of respondents (78%) stated no in developing prices. By asking the respondents what they meant by developing prices, they explained that costs should be considered when pricing and not necessarily for price to be based on cost.

On the topic of who uses the reports generated from the cost accounting system?, see table 8.13

Users of cost accounting		Managerial level						
reporting	High	Medium	Low	Total				
Financial managers	8%	22%	42%	72%				
Budget analysts	6%		4%	10%				
Others	6%	4%	8%	18%				
Total	20%	26%	54%	100%				

Table 8.13: Users of cost accounting reports (N =50)

Source: Analysis of questionnaire.

This result indicated that the accounting system served financial accounting.

Regarding the level at which cost information flowed within the GOGCWS (observation from the interviews), the majority of interviewees (94%) said that the cost information goes to a high level of command. This result was the outcome of the professional hierarchical bureaucracy with which the GOGCWS was consistent.

## 8.6.5.3 Decision-making and cost management

On the topic of using cost information in decision-making in the GOGCWS (observation from the company's documentation) cost accounting helps to set the policy and decision making at all levels (see table 8.14).

Cost information	%
Plays a part in decision making	55
Does not play any part in decision making	32
I do not know	13

Table 8.14: Use of cost accounting reports in decision-making (N = 50)

Source: Analysis of interviews

By analysing the percentage who answered that cost information plays a part in decision-making,  $62\%^{206}$  of them answered "I think". This means they were not actually

<sup>&</sup>lt;sup>206</sup> 62% out of 55% which has shown in table 8.14

sure whether cost information played a part in decision-making, and they stated their answers on the opinion that senior management takes no decisions without looking at cost information.

#### 8.6.6 Financial management system and cost accounting system

The question what financial management systems/functions are integrated with this cost accounting system?, is illustration in table 8.15

Table 8.15: Integration between	financial	management	system a	nd cost a	ccounting
	system	(N =50)			

			]	Mana	gerial	Leve	1			Tot
		Yes %			No %		I do i	al		
	H	Μ	L	H	Μ	L	Н	M	L	%
General ledger	20	22	44		4	8			2	100
Programme accounting	18	26	46	2	·	8				100
Payroll	18	20	46	2	6	8				100
Inventory control	6	14	20	14	12	34				100
Property, plant and equipment	16	16	38	4	10	16				100

Source: Analysis of questionnaire, H =High level, M =Medium level, L =Low level.

Table 8.15 shows that the majority of respondents stated that the financial management systems were integrated with the cost accounting system of the GOGCWS. The reason for this answer may be that the GOGCWS followed the Unified Accounting System, which did not distinguish between financial, and cost accounting systems.

From the analysis of the interviews, in answer to the question *are the elements of cost in your cost accounting system reconciled with your general ledger control accounts (e.g. labour to accrued payroll account)?* Just under 44% of respondents answered monthly. Of the remaining respondents 20% answered others, 18% answered quarterly, 16% answered annually and 2% answered no.

### **8.6.7 Distinguishing between direct and indirect costs**

From the analysis of the interviews, the majority of interviewees (84%) could not answer the question regarding how overhead costs were managed and allocated within the GOGCWS. Even the interviewees (16%) who answered the question, said they calculated the total cost for each centre each year and added these costs to the production centres. So, the researcher concluded that no one in the GOGCWS, including staff in the cost department, understood the procedures used in cost management.

In conclusion, the accounting system in the GOGCWS was linked to the financial accounting function and not the management accounting function. There were no substantial changes in the cost methods of the GOGCWS; furthermore, there were no differences between the rules (documentations) and the routines (practices) related to measuring the cost of resources. Cairo's governor determined the price of water in Cairo City (and every governor determines the selling price of water in his/her city). Essentially, the cost system reflects the organisational structure of the GOGCWS. The emphasis across the whole company had been limited to conventional budgetary control with full allocation of overheads.

Cost information went to senior management. Furthermore, cost information served several purposes, perhaps the most important was to satisfy legislative requirements. It can be argued that cost information was not the first element to be taken into consideration to determining the water's price as there was an effect of legislation on cost practice.

#### 8.7 Cost management in the GOGCWS

In cost management literature, (evidence of renewed interest could be seen from the publication and reaction to Johnson and Kaplan, 1987). A great deal of recent cost management literature centred on activity-based costing (Banker and Hughes, 1994; Cooper and Kaplan, 1992; Datar and Gupta, 1994). However, other modern cost management tools such as benchmarking (Elnathan and Kim, 1995), and just-in-time inventory systems (Alles, et al., 1995) have also recently been examined. In order to achieve the aim of this section, it is divided into seven sub-sections.

## 8.7.1 Throughput accounting model

The use of throughput accounting in the General Organisation for Greater Cairo Water Supply (the biggest water organisation in Egypt can be seen in table 8.16).

Planning		Managerial level														
and	Use	Use now %			Plan to use			ed in	the	Not used or			I	do no		
control					in the	9	pa	ast, b	ut	pl	planned to		k	now <sup>o</sup>	Total	
process				fu	ture	%	not	t now	%	us	use at all %					%
<b>L</b>	Η	M	L	Η	M	L	Η	Μ	L	H	M	L	Η	M	L	
Throughput	10	6 ·								2			8	20	54	100
accounting																

Table 8.16: Use of throughput accounting in the GOGCWS (N =50)

Source: Analysis of questionnaire, H =High level, M =Medium level, L =Low level.

Table 8.16 shows that the throughput accounting model was not used in the GOGCWS. Throughput accounting costs focus on the short term with overheads attributed from product accounting to usage of existing manufacturing bottlenecks. The result of questionnaires relating to throughput accounting was consistent with Dugdale, et al. (1996) who argued that the idea of building up product costs on the throughput accounting principle is probably impracticable (for the principle of throughput accounting, see section 4.1).

#### 8.7.2 Benchmarking

Regarding the use of benchmarking in the GOGCWS is illustrated in table 8.17.

	Managerial level																	
Planning	Us	Use now %			Use now %			10w % Plan to use			Used in the			Not used or			do n	
and				in the			past, but			planned to			know %					
control				future %			not now %			use at all %						Total		
process	H	M	L	H	M	L	H	M	L	H	M	L	Η	M	L	%		
Benchmark -ing	2	4	10	6		12			2	12	20	20		2	10	100		

Table 8.17: Use of benchmarking in the GOGCWS (N = 50)

Source: Analysis of questionnaire, H =High level, M =Medium level, L =Low level.

The findings of this research (non-uses of benchmarking in the GOGCWS) was consistent with Clark and Merton (1997). They see benchmarking as a tool for management change. There was no change in the management process in the General Organisation for Greater Cairo Water Supply as observed from the interviews.

The core idea of benchmarking was that performance should include three performance measures: (a) cost minimisation (b) cycle time minimisation (c) improvement of quality (see section 4.1). These three concepts were similar to the mission of the GOGCWS which was "To earn the trust and satisfaction of our customers and stakeholders by providing Greater Cairo with sufficient quantity of high quality water at fair price which contributes to the development of society" (Black and Veatch International, 1997: 4).

From the above results of the questionnaires related to benchmarking, its core idea, and the mission of the GOGCWS, a gap was observed between theory and practice. The model of cost management observed at the GOGCWS (or the design of the accounting system) did not give priority to cost management whereas the theory did give priority. The reason for this could have been the effect of the Government on cost accounting. This supported the argument of Geiger and Ittner (1996) which was that mandated government cost accounting systems were positively associated with cost system elaborateness, but were not associated with greater internal use of cost accounting data.

#### 8.7.3 Budgeting

Budgeting played a crucial role in the water industry in Egypt, and particularly in the case of the GOGCWS, concerning the use of operational budgeting procedures in the GOGCWS (see table 8.18).

							Μ	anag	geria	l lev	<b>el</b>									
Planning	Use now %		Use now %		Use now %		Plan to us		Plan to use		Use	ed in	the	Not	t used	l or	I	do no	ot 24	
and control process				future %			not now %			use at all %			KIIOW 70			Total				
•	H	M	L	Η	M	L	H	M	L	Η	M	L	Η	Μ	L	%				
Operational budgeting procedures	20	24	46			2						2		2	4	100				
Capital budgeting procedures	20	22	42			2					4				10	100				

Table 8.18: Use of Operational and Capital budgeting in the GOGCWS (N = 50)

Source: Analysis of questionnaire, H =High level, M =Medium level, L =Low level.

This result showed that the majority of respondents (90%) stated that they used operational budgeting procedures<sup>207</sup>. 84% of respondents stated they utilised capital budgeting procedures<sup>208</sup>. These answers indicated that GOGCWS was highly controlled internally and externally because views were that the budget was an internal and external regulative model and a critical reflection about what the staff were doing (see figures 8.5 and 8.6).

It was valuable to look at whether there was any difference or similarity between rules (documentations), routines (practice) and theory related to the meaning of the budget, preparation and principles preparation.

The budget was a sub-system of the accounting information system and contained the estimation of the financial activities in a comprehensive financial plan. This considered the co-ordination between the objectives, facilities, and probability circumstances (The GOGCWS, no date source given, c: 2). So, in theory it is an effective tool for planning, control and performance evaluation. Furthermore, the budget process helped in making crucial decisions regarding resources<sup>209</sup>, that created goods and services, and supplied data to enable the product managers to manage the planning and control loops around their particular responsibility. So, the researcher observed no difference in the role of the budget between the rules (documentations) of the GOGCWS and the normative theory.

With regard to the description of the annual budgeting process at the GOGCWS, there were stages of budget preparation which are illustrated in figure 8.5.

<sup>&</sup>lt;sup>207</sup> 2% of respondents (one respondent) answered that there were plans to use this in the future. The reason maybe that she had spent all 25 years of her employment recording in one book (the service requirement) and yet got no a qualification to understand where the information would be used. Another 2% of respondents (one respondent) answered that it has not been used in the past and there are no plans to use it at all. The reason could be that he worked in the Bills Auditing and Stamp department and had no idea about the budget. Of the remaining respondents 6% answered that we do not know.

<sup>&</sup>lt;sup>208</sup> One respondent answered that there are plans to use this in the future and maybe the reason why is that his experience in work is very limited (4 years only), the nature of his work was only to record some expenses and he had not enough qualifications. Two respondents answered that there are no plans for use at all. One of them worked as the chief of the Bills Auditing Department and the other was the chief of the Bills Auditing and Stamp Department and the nature of work for both of them had no connection with formal planning and control procedures. Five respondents answered that we do not know.

<sup>&</sup>lt;sup>209</sup> The argument of normative theory.

## Figure 8.5: Organisational model of stages of budget preparation



The objectives of the GOGCWS should be determined before the detailed programmes are set<sup>210</sup>.

Each administration determined their requirement and sent it to the budget administration. The budget administration categorised these requirements to the financial sectors according to The Unified Accounting System (see section A1.2.4.2).

To achieve integration and quasi-resolve the conflict<sup>211</sup>.

To be a guide for implementation and performance standard.

Source: Based on the GOGCWS, (no date source given, b), pp. 2-4

The perception of the interviewees regarding budget preparation was considered. For example, the Chief of Balance Sheet Administration in the GOGCWS said there were some steps in preparing the operational budget, which were explained as follows:

<sup>&</sup>lt;sup>210</sup> (Source: The GOGCWS, no date source given, b: 2).

<sup>&</sup>lt;sup>211</sup> For example, the purchasing plan of chlorinate was affected by the program of water production, credit facilities, etc. From another aspect, the purchasing plan affected the program of water production, amount owing to others, the financial position of the organisation and the size of the inventory of chlorinate. So if there was no co-ordination between detailed plans the result might be incorrect. From another aspect, co-ordination between detailed plans might need simplification, because all aspects needed to be carefully considered as each affected the other e.g. the purchasing plan was not compatible with the space available within the organisation.

# Figure 8.6: Stages of operational budget preparation

Sending of a letter to all stations, dragnets, different	Receiving at the expenses of the
administrations to determine their requirements such	last year
as (raw materials, new employees, etc)	
From the two elements above we could prepare	
The operational budget for the next year	
and then	
Send the operational budget to the Financial Ministry	
The Financial Ministry gave subsidisation which was	usually less than requested in the

budget

Source: Chief of Fixed Assets Department interview.

Regarding the question how do you describe the annual budgeting process at your company?, is illustrated in table 8.19

	Ma	nagerial lev	vel	
Budgeting process	High	Medium	Low	Total
Top-down procedures.	8%	10%	10%	28%
Top management set parameters: organisation units	10%	6%	16%	32%
prepare budgets: top management adjust/decide.				
Bottom-up procedures.	2%	8%	22%	32%
Discussion and participation procedures.		2%	6%	8%
Total	20%	26%	54%	100%

## Table 8.19: Description of the budgeting process in the GOGCWS (N =50)

Source: Analysis of questionnaire.

This means the budget process at the GOGCWS had extremely hierarchical in its procedure.

Based on the analysis of the questionnaires and interviews, there was no substantial difference between rules (documentations) of the GOGCWS and the routines (practices)

related to budget preparation. All of them agreed that each administration determined their requirement and sent it to the budget administration.

The final point to highlight were the general principles for preparing the budget. From the documentation of the GOGCWS, questionnaires and the personal interviews, there were five points in the principles for preparing the budget as follows:

(a) The budget relied on prediction of water based on the historical data of the GOGCWS such as the amount of production and sales of water

(b) Comprehensive budget

(c) Division of budget time-scale

(d) Expression of the budget in financial terms

(e) The budget was considered  $^{212}$  as a tool of control

A brief discussion for each principle is given below:

(a) The budget relied on prediction<sup>213</sup> of water. The prediction was not guessed. It was dependent on rules and fundamentals such as scientific development, economic elements, the political actions, governmental actions, intentional actions, different legislators and the historical data of the organisation such as the amount of production and sales of water.

The study of prediction could be divided into three stages: the prediction for the immediate term such as the cash budget, the prediction for the short term such as the operation budget and the prediction for the long term such as the capital budget. For instance, the Chief of Commitment (Engagement) Department (Balance Sheet) stated "there are three factors which determine the capital budget (replacement and renewal<sup>214</sup>, completion<sup>215</sup>, new expansions<sup>216</sup>)".

<sup>&</sup>lt;sup>212</sup> By the Accounting Central Institution.

<sup>&</sup>lt;sup>213</sup> Such as the prediction of sales, production, supply, etc.

<sup>&</sup>lt;sup>214</sup> The replacement and renewal projects related to the existent/current stations.

<sup>&</sup>lt;sup>215</sup> The completion projects related to the projects needing completion.

<sup>&</sup>lt;sup>216</sup> The new expansions were those that should consider future financial needs.

(b) Comprehensive budget. The budget should contain several aspects such as: sales plan, production plan, investment plan and the finance plan. The Unified Accounting System ensured that the budget should have three aspects (budget in kind<sup>217</sup>, financial budget<sup>218</sup>, and cash budget<sup>219</sup>).

(c) Division of budget's time. The budget would be divided into periods (quarterly or monthly) in order to overcome the obstacles which can affect the organisation, such as changes in the economic environment that were not considered in the budget preparation. This required flexibility<sup>220</sup>, which allowed the evaluation of actual performance, enabled the discovery of variance before it increased and action to be taken to prevent future repetition of these obstacles.

In practice, the GOGCWS did not divide the budget into periods. This meant the GOGCWS did not take into consideration any changes that had occurred during the year. This meant there was a gap between the GOGCWS's rules and general principles for preparing the budget and its practical application.

(d) Express the budget in financial terms. The aim of this role was to co-ordinate between the sub-budgets as they had difficult measurement units, for instance: the raw material budget was expressed by quantities or weights and the staff budget could be expressed by hours of work. In order to arrive at the final budget form, all these budgets had to be calculated and presented financially.

(e) The budget was considered a tool of control. This principle relied on two rules: (1) linking budget estimations by the responsibilities centres (this meant distributing the budget's estimation to the responsible departments and administrations, to find their

<sup>&</sup>lt;sup>217</sup> The budget in kind contained the goods requirements, the service requirements and staff. It was called the budget in kind because it expressed by quantities before translating it to cash values.

<sup>&</sup>lt;sup>218</sup> The financial budget translated the budget in kind to cash frame and it explained the financial plan of the organisation.

<sup>&</sup>lt;sup>219</sup> The cash budget contained the expected cash payments and payments received during the period of the budget. It helped to study the financial structure of the organisation.

<sup>&</sup>lt;sup>220</sup> The division of the budget's time helped, due to its flexibility, to review the estimation of the budget each period by less effort and less time in order to meet any changes that were not considered before.

weak and strong points, and then identifying the necessary corrective measures (2) Participation by the responsibility centres in preparing the budget. These two rules worked in practice and that was clear when one of the accountants in Participating Accounts Department stated "each administration determines their requirements and sends it to the responsible administration for preparing the budget and then the estimations of budget are prepared according to this basis".

The budget plan served two main aims (planning and control). There was no benefit from planning if it was not followed by control (The GOGCWS, no date source given, b: 22). Control should pass through three steps

(a) Comparison between actual and planning performance and determining the variances: This was achieved through the budget estimate record. It contained a column for the actual expenditure and a column for the estimated budget so that The Financial Ministry could determine the variance.

(b) Variance analysis: The variances were usually divided into two elements. The quantity element (the difference between the actual sales and the sales plan according to the planned price) and the valorem element (the difference between the planned price according to the actual quantities).

(c) Taking actions<sup>221</sup> which were necessary to correct the variances.

The reasons for the variances were:

(a) The amount of activity did not suit the staff (quantity and skills)

(b) Random decrease of the required funds from The Financial Ministry without knowing the needs of the organisation

(c) The lack of standard to measure the efficiency of the use of resources (post estimation)

In conclusion, The GOGCWS was ruled by bureaucracy from both internal and external influences and there was no substantial difference between the rules (documentation) of the GOGCWS and the normative theory related to the role and preparation of the

<sup>&</sup>lt;sup>221</sup> After the variances were determined, analysed and responsibility confirmed, then managerial reports were prepared (including suggestions on how to correct the variances) and sent to the responsible levels in order to correct the variances.

budget. On the other hand, there was a gap between the normative theory and the practice of the GOGCWS related to the role of budgets.

The accounting system in the GOGCWS tended to focus wholly on financial accounting and its institutional setting and chose to neglect management accounting. Several studies informed by institutional theory show that the financial controls devised to discipline resource utilisation in public-sector organisations (for example, external financing limits) are little used further down the organisational hierarchy (e.g. Ansari and Euske, 1987; Pettersen, 1995) or assume the relatively symbolic role of legitimising the organisation to funding bodies (e.g. Brunsson, 1989; Czarniawska-Joerges and Jacobsson, 1989).

#### 8.7.4 Activity based costing

ABC was not used in the water industry in Egypt. Regarding the question asked of 37 interviewees in the General Organisation for Greater Cairo Water Supply *could you please tell me about the methods used in your company for measuring product cost?*, see table 8.20

73%	stated Full Absorption Costing
22%	stated Activity Based Costing
5%	stated that they did not know

Table 8.20: Methods used for measuring product cost (N =37)

Source: Analysis of interviews

The researcher also checked the validity of the answer of the interviewees by asking in the questionnaire *what is the product/service cost measurement practice in use at your company?*, see table 8.21

Table 8.21: Methods used for measuring product cost (N =50)

Methods of cost	N			
measurement	High	Medium	Low	Total
Full absorption costing model.	12%	18%	40%	70%
Activity-based costing	8%	8%	12%	28%
Others			2%	2%
Total	20%	26%	54%	100%

Source: Analysis of questionnaire.

This result raised the question of why 22% of interviewees and 28% of respondents answered activity based costing, when the method used in the GOGCWS for measuring product cost was the full absorption costing model (based on the analysis of the documentation of the GOGCWS). In order to answer this question, the researcher discovered that the 22% of interviewees who answered Activity Based Costing misunderstood what this was. The reason for this misunderstanding could have been the Unified Accounting System which the GOGCWS followed (see section 8.5). This divided production into production Centres (5), Productive Services Centres (6), Marketing Services Centres (7), and Administrative and Financial Services Centres (8). So, it could have been that the reason for these results was that respondents or interviewees thought that when they calculated the cost for each centre alone and then calculated all together, this was Activity Based Costing.

There was no empirical evidence of ABC use in the water company in Egypt. Research was concentrated on one case study however, all water organisations in Egypt are governed by the same law (see section 8.1).

#### 8.7.5 Activity based management

ABM is not used in the GOGCWS in Egypt. The researcher discovered this from the documentation of the GOGCWS and observations from the questionnaire, see table 8.22

		Managerial level														
Planning and control	Use now %			Pla	Plan to use in the			Used in the past, but			Not used or planned to		I do not know %			
process				future %		not now %			use at all %					Total		
<b>F</b>	Η	M	L	H	Μ	L	Η	M	L	Η	M	L	Η	M	L	%
Activity Based Management	2		2	6	2					10	2	2	2	22	50	100

Table 8.22: Use of ABM in the GOGCWS (N =50)

Source: Analysis of questionnaire, H =High level, M =Medium level, L =Low level.

This answer was expected because the GOGCWS followed the full absorption cost model. Furthermore, activity based management is extension of activity based costing (see section 4.4) and the result of analysis of the above point (activity based costing)

concluded that there was no empirical evidence of ABC in the GOGCWS. So, the researcher suspected that ABM was not used in the GOGCWS.

#### 8.7.6 Strategic Management Accounting (SMA)

Planning and decision-making are a reasonable basis in a stable or semi-stable environment but are more difficult in a dynamic environment. The present environment in Egypt was characterised by dynamic political, economic, social and legal elements. Egypt was transferring from the central national system to the economic market and it was necessary that the organisation should be proficient both managerially and economically. The lack of opposition to these changes and new indicators leads to high risks such as: losing control of the direction of the organisation; the loss of GOGCWS's leadership position because it was considered the biggest water organisation in the developing countries<sup>222</sup>; re-organisation of the organisation by other external influences or making management alternations to achieve change (The GOGCWS, 1996: 7).

Regarding the question *how do you describe the strategy of your company*?, see table 8.23

Description of the GOGCWS's	M	el		
strategy	High	Medium	Low	Total
As a unitary organisation.	8%	14%	18%	40%
As a multidivisional organisation.	12%	8%	34%	54%
Others.		4%	2%	6%
Total	20%	26%	54%	100%

Table 8.23: Description of the GOGCWS's strategy

Source: Analysis of questionnaire

Perhaps the reason for these different answers is that the chief of the fixed asset department and the chief of the cost department said "the GOGCWS has one strategy and this strategy is divided into sub-strategies for each department". Furthermore, Loft (1991) argued that large companies' strategies were trying to dominate the market by driving competitors out (this was not the case for the GOGCWS) or lead to monopoly

<sup>&</sup>lt;sup>222</sup> Egypt was one of the developing countries. The term "developing countries" has been used not to suggest that such countries can or should be seen as a homogeneous whole, but simply to broadly describe countries which were characterised by relatively low levels of per capita income, possess limited industrialisation and exhibit a constrained infrastructure (Vullimary et al., 1990).

and in turn to bureaucracy and wastefulness. The GOGCWS had a bureaucratic system (see section 8.3)

The strategic planning in the GOGCWS had a process which could be illustrated as follows:

(a) Planning: the approval<sup>223</sup> of the strategic planning process of the GOGCWS

(b) Clarifying the formal and informal responsibilities issued for GOGCWS

(c) Clarifying the mission of GOGCWS and the value's owner benefit

(d) Evaluating the external environment (power and directions, owners' benefit, competition and co-operating)

(e) Evaluating the internal environment (resources, performance, and present strategy)

(f) Determining the strategic subjects which face the organisation such as service level, costs, finance, and management

(g) Formulating the required strategy to manage all main subjects such as: determining the scientific alternative, obstacles in achieving these alternatives, the main suggestion to overcome these obstacles, planning work for the suggestions and working out a program for the plan

(h) Setting effective organising strategy for the future

There were different levels for strategic planning such as; high command, medium and executive command.

So far, this analysis has shown that there was no change in cost management model of the GOGCWS. The following sub-section provides evidence of resistance to change of cost management model within the GOGCWS. This resistance to change has been created by the external institutional pressures (the Egyptian government policy).

#### 8.7.7 The Consultants

Black and Veatch International and its associated sub-contractors carried out a Management, and a Training and Systems Strengthening (MTSS) project under the Institutional Development Component, Cairo Water II. The project covered the period

<sup>&</sup>lt;sup>223</sup> There are several institutions which were involved in the approval of the strategic planning process of the GOGCWS such as; Cairo governorate, Finance Ministry (treasury department), Ministry of planning, The Accounting Central Institution, etc.

25 January 1993 through to 31 December 1997. The project was funded by USAID<sup>227</sup>. The aim of the Cairo Water Supply II (the Institutional Development Component), was to provide an organisational and technical support system to the authority. After a project start-up workshop in May 1993, the project was re-focused to prioritise managerial autonomy and financial viability.

The project concluded with several recommendations. Perhaps the most appropriate to this research are as follows: (a) cost would be recovered by a fair, economic price for water, (taking into accounts social concerns and realities) (b) The organisational structure for delivering these economic, high quality services would be decentralised, performance-based and committed to excellence. This value-oriented vision, strategically aligned the GOGCWS with the Egyptian Government's private-sector-led economic development framework. This would enable the utility to direct the water sector away from one funded exclusively by private sector participation.

The consultants (Black and Veatch International) therefore suggested a public ownership of the source of water and private ownership for its distribution. They challenged the cost system of the GOGCWS and offered an alternative mode of cost management for the water industry. This alternative of cost management was based on re-organisation structure of the GOGCWS (see figures A 1.5, A 1.5.1 and A1.5.2).

What happened to all these recommendations? It was found that they were not put into practice. Why? It may be that it was against government policy for the management of the GOGCWS (For more explanation of the above, see section 11.2.4, which focus on the lack of change at the GOGCWS). Recommendations of the consultants (Black and Veatch International) were withheld by the chief-of-legal-affairs of the organisation.

<sup>&</sup>lt;sup>224</sup> The U.S. Agency for International Development (USAID) is an independent government agency that controls foreign assistance and humanitarian aid in order to advance the political and economic interests of the United States. Since 1975, the United States has given over \$2.8 billion to urban water and wastewater infrastructure projects. The United States has co-operated with the Egyptian government in programmes to expand water and wastewater infrastructures, to improve the delivery of services, and to ensure that adequate policies are in place to support further development in this area. USAID is currently implementing a water and sanitation programme which is valued at approximately \$1 billion in eleven cities. This funding is for the construction of major new water and wastewater treatment, and conveyance facilities. Water and wastewater activities are implemented in close co-operation with several Egyptian agencies, (including nine economic utility organisations, the Ministry of Housing, Public Utilities, and New Communities).

### **8.8** Conclusion

From the analysis presented so far, it could be argued that GOGCWS contains features of an institutional political perspective discussed earlier in chapter 2. These were as follows: Firstly, it was preoccupied with formal structures and legal systems (see sections 8.1 and 8.2). Secondly, the approach emphasised detailed accounts of particular political systems, resulting in "configurative description", or, in other words, interactive, descriptive accounts of inter linked rules, rights, and procedures (see section 8.3). Thirdly, the approach was conservative in the sense that it emphasised the "permanent and unchanging" (see section 8.6.2). Finally, the work was largely non-theoretical, (i.e. more attention was given to historical re-constructions of specific institutional forms, see section 8.7.3).

This chapter has revealed the complexity of the environment of the Egyptian water industry and the GOGCWS in particular (see section 8.2). In addition, Diab (1999) argued that there are many deficiencies in Egyptian water utilities that can be traced to institutional issues and constraints, largely due to a cumbersome, hierarchical organisation without regulatory oversight. This lack of local autonomy prevents effective and efficient operation of the water or wastewater system (The GOGCWS, no date source given, a: 44). This chapter has highlighted the changes of the economic forms and legal structure of the GOGCWS (see section 8.2.1). From the neo-institutional economic theory perspective, the chapter points out that the GOGCWS has hierarchical structure (see section 8.3). Furthermore, it has illustrated that the GOGCWS is highly controlled (see section 8.4).

This chapter points up the limitation of the aims of the cost accounting system in the GOGCWS (see section 8.6). Moreover, it has highlighted the use of cost information in general and the use of cost information for decision-making in particular, which concluded that cost information has a very limited role in decision-making. The chapter has also illustrated the cost method used in the GOGCWS (full absorption costing) and stressed that the method of cost accounting in the GOGCWS is similar to traditional management accounting not contemporary management accounting (see section 8.7).

The chapter showed that the GOGCWS followed the traditional view which was that costs were best controlled by the department responsible for minimising the variance between budgeted and actual cost by cost element. The chapter has also suggested alternative modes to cost management to those suggested by consultants, Black and Veatch International, who suggested a public ownership of the source of water, and private ownership for its distribution, but this was not put into practice at the GOGCWS. From the neo-institutional economic theory perspective, the Black and Veatch International recommended that the GOGCWS should move from a hierarchical structure to adopt a hybrid approach.

In conclusion, the GOGCWS lead to monopoly and in turn to bureaucracy and waste. The researcher concludes that the GOGCWS did not have strategic cost management or an economic optimisation cost management policy but might have accounting and control cost management policy. Institutional theorists have tended to view the emergence of organisational control practices as a process of relatively passive adaptation (DiMaggio, 1988; Powell, 1991; Scott, 1995; Czarniawska and Sevon, 1996). More recent research<sup>225</sup> shows that organisations may, under certain circumstances, pro-actively influence their relations to different constituencies.

<sup>&</sup>lt;sup>225</sup> Abernethy and Chua (1996) found that the scope for managerial choices and which controls to implement can be significant even where certain constituencies exercise considerable power.

# Chapter 9

# Cost management and external environment

## 9.0 Introduction

Figure 5.1 had shown the importance of external environment for cost management. This chapter is going to explore the question "Was there a relationship between cost management<sup>226</sup> and P (CI)<sup>227</sup> and P (Un)<sup>228</sup> which relates to the three areas of the external environment (product market, factor market and legislation)? As illustrated in figure 9.1, the first three elements (efficiency, optimisation and strategy) were seen to derive from the interplay of forces in the external environment.

Identifying the factors that drive costs is central to effective cost management (Porter, 1985; Shank, 1989; Shank and Govindarajan, 1992; 1993). "Understanding cost behaviour means understanding the complex interplay of "Cost drivers" at work in any given situation" (Shank and Govindarajan, 1992: 29). 'Cost drivers' research to model cost variability has sparked off an emergence of research literature into the relationships between overhead costs and their hypothesised underlying causes. Several factors are identified, (e.g. complexity and efficiency, Foster and Gupta, 1990); capacity, product and process variables, (Banker and Johnson, 1993; Data et al., 1993); and product mix heterogeneity (Anderson, 1995). Rouse and Putterill (2000) argue that while the above factors are relevant, this list of cost drivers is incomplete. In particular, environmental factors<sup>229</sup> are component explanations of cost variability. They argue that not only has cost management research paid little attention to the impact on costs by environmental factors, it has also been slow to recognise and incorporate theories which underpin other disciplines.

<sup>&</sup>lt;sup>226</sup> The objectives of cost management are efficiency (see section 4.1), optimisation (see section 4.4) and strategy (see section 5.2).

<sup>&</sup>lt;sup>227</sup> P (CI) refers to the perception of competitive intensity, (i.e. the general level of market conditions which make profitability difficult).

<sup>&</sup>lt;sup>228</sup> P (Un) refers to the perception of unpredictability, (The overall degree of variability from period to period, which make forecasting and planning difficult).

<sup>&</sup>lt;sup>229</sup> Environmental factors have been incorporated into analyses in research literature (Jesson, et al., 1987; Cook, et al., 1994; Lovell, et al., 1994).

#### Figure 9.1: The relationship between cost management and the external



Figure 9.1 builds on Institutional theory (a theory which views organisations as operating within a social framework of norms, values, and assumptions about what constitutes 'appropriate' economic behaviour, Oliver, 1997). According to Pfeffer and Salancik (1978: 3) "a good deal of organisational behaviour can be understood by knowing something about the organisation's environment<sup>230</sup> and the problems it faces for obtaining resources. What happens in an organisation, is not only a function of its internal structure, leadership, procedures, or goals. It is also a consequence of the environment, and the particular contingencies and constraints from that environment".

Powell and DiMaggio (1991: 13) argue that it is important not to see the environment as separate from the company and its organisational processes. The environment can penetrate the organisation, creating ways in which actors view the world and make sense of their organisational activity. The environment creates categories through which organisational activity is constructed. Institutional theory according to Scott (1995) is a

<sup>&</sup>lt;sup>230</sup> This research argues that the environment of the GOGCWS could explain three main factors: product market, factor market and legislation.

continuation and extension of the intellectual revolution that introduced open systems<sup>231</sup> into the study of organisations.

Scott (1987a) identifies the diversity of institutional sources, (i.e. regulatory structures, governmental requirements, laws, ideologies, educational systems and professions). Hence, the institutional environment is of "a complex and multiple character" (DiMaggio and Powell, 1983; Meyer and Scott, 1983a). Oliver (1991) argues that in dealing with institutions, organisations may use a variety of strategies. These may range from passive conformity, compromise and avoidance, to defiance and proactive manipulation. In other words, organisations invariably conform to their institutional environment. Organisations respond in a variety of ways to the normative, regulatory structure, depending on the situation and its possible consequences (Bergevarn, et al., 1995)

Scott and Meyer (1991: 117) propose the idea of "societal sectors", to describe the way in which organisational environments are specific and variable, rather than general and uniform. Scott and Meyer (1991) claim that the technical and institutional perspectives are independent dimensions, rather than polarised extremes. Some organisations are subject to both kinds of pressures. Scott and Meyer cite public utilities as an example.

Carruthers (1995) notes that the distinction between the technical and the institutional perspective is frequently overdrawn. He claims that non-profit organisations (such as the GOGCWS and other less competitive environments), have sufficient 'slack' to allow institutional effects to come into play. Similarly, Powell (1991: 185, 188) contends that institutional effects are visible even in technical environments, and it can be difficult to separate the two.

Neo-institutionalists would prefer to keep the technical and institutional perspectives as analytically separate dimensions, but the two are often confused. For instance, institutional processes are routinely disguised as technical ones. De-coupling plays an important role in this process. Its institutionally prescribed appearance (via a formal structure), does not compromise actual operations. Researchers such as Brignall and

<sup>&</sup>lt;sup>231</sup> Open systems theory modified existing approaches by focusing on the wider environment as it constrains, shapes and penetrates the organisation (Scott, 1992).
Modell, 2000; Abernethy and Chua, 1996; Pettersen, 1995; Ansari and Euske, 1987; Covaleski and Dirsmith, 1983, all argue that a way of dealing with conflicting institutional pressures in public sector organisations, is to de-couple<sup>232</sup> the control systems used at different levels of the organisation.

Cost drivers have been categorised by Shank and Govindarajan (1993) into: (a) structural (scale, scope, technology and complexity) and (b) executional (workforce involvement, plant layout, product configuration and capacity utilisation). Rouse and Putterill (2000) classified cost drivers into: (a) policy environment<sup>233</sup> (b) physical environment<sup>234</sup> and (c) market environment. This classification of cost drivers includes three areas of the external environment - (the product market, the factor market and legislation). The factor market includes raw materials and supply, labour, management, capital and financing, and technology).

In order to understand the complex processes of the Egyptian Water Industry, it was necessary to focus on micro-processes. However, it was impossible to ignore the broader (macro-institutional) dimension. To understand micro-processes it was necessary to recognise the institutional context both within the organisation itself, (i.e. the organisation's rules, routines and institutions) and outside, (i.e. the broader social, economic and political institutions of the organisational field and the society in which the organisation operates) (Burns and Scapens, 2000). This chapter has focused on the micro level (use of cost information at the GOGCW) and the macro level P (CI) and P (Un). These are related to three areas of the external environment: (product market, factor market and legislation).

<sup>&</sup>lt;sup>232</sup> De-coupling can be identified as a process of breaking down the structural elements of different parts of the organisation in response to institutional pressures, (in order to comply with prescribed norms, Meyer and Rowan, 1977; Brignall and Modell, 2000).

<sup>&</sup>lt;sup>233</sup> This includes factors, highlighted by Shank and Govindarajan (1993). Rouse and Putterill (2000) argue that these cost drivers relate closely to the fixed factors or constraints faced by an organisation. Changes to these drivers were not usually a short-run operation-arising as they do from internal choices affecting organisational structure and process (p. 369).

<sup>&</sup>lt;sup>234</sup> The Physical environment, in which the organisation operates, has an "external flavour" and provides difficult challenges for management to improve performance.

This chapter is divided into seven sections. Section one describes how the data was presented and highlights the variables which measure cost management (efficiency, optimisation and strategy). Cost management is seen in relation to CI and Un that is related to the product market, factor market and legislation. Section two gives a descriptive analysis of the data. Section three examines the reliability and validity of the data, by using alpha coefficient and factor analysis. Section four examines the restructured data. Section five addresses the question "Was there a relationship between cost management and P (CI) and P (Un) which relates to the three areas of the external environment (product market, factor market and legislation).

Section six offers a brief discussion of the results. The final section outlines the main conclusions of the chapter.

#### **9.1 Data**

On the use of cost information, Al-Hazami (1995) identified the three objectives of cost management: efficiency, optimisation and strategy. Cost management practices were measured by using forty-seven items (questions<sup>235</sup>). Fifteen items measured the use of cost information for system maintenance and efficiency. Eighteen items measured the use of cost information for system optimisation. Fourteen items measured the use of cost information for system strategy. A Likert scale<sup>236</sup> was utilised, (ranging from 1 = extremely important to 5 = not important).

The perception of 'intensive competition' and 'unpredictability' relate to the three areas of the external environment<sup>237</sup> (product market, factor market and legislation). The GOGCWS perception of the external environment for (CI) and (Un) were measured using eighty-six items, (forty-three items for P (CI) and forty-three items for P (Un)). Fourteen items measured the product market (seven for P (CI) and seven for P (Un)). Sixty-four items measured the factor market (thirty-two for P (CI) and thirty-two for P (Un)). The factor market is divided into five sub-sectors: raw materials and supply (seven items for P (CI) and seven items for P (CI) and

 $<sup>^{235}</sup>$  See section A2.1 (Question 8 in section 3.4).

<sup>&</sup>lt;sup>236</sup> The range of Likert scale was 1 = extremely important 2 = very important 3 = important 4 = not so important 5 = Not important and 6 = I do not know.

<sup>&</sup>lt;sup>237</sup> See section A2.1 (Question 1 - the company environment).

seven items for P (Un)). Management (four items for P (CI) and four items for P (Un)). Capital and financing (four items for p (CI) and four items for P (Un)). Technology (ten items for P (CI) and ten items for P (Un)). Eight items measured legislation (four items for P (CI) and four for P (Un)). Respondents were asked to assess various sectors of the company's environment, on the dimensions of P (CI) and P (Un) by using a Likert scale<sup>238</sup>.

There were two questions<sup>239</sup> relating to the GOGCWS environment. In one question respondents were asked to prioritise the significance of environmental sectors (product market, factor market or legislation) by percentage. The second question asked respondents to rank the sub-sectors of the factor market, (including raw materials and supply, labour, management, capital and financing, and technology).

This chapter has focused on the perceptions of the GOGCWS managers. The sample study included all managerial levels. Functional levels were also included in the sample, (i.e. financial, technical and distribution).

<sup>&</sup>lt;sup>238</sup> For competitive intensity, the Likert scale ranged from 1 = low to 5 = high. For Unpredictability, the Likert scale was 1 = very predictable 2 = predictable 3 = average 4 = unpredictable and 5 = very unpredictability 6 = I do not know.

<sup>&</sup>lt;sup>239</sup> See section A2.1 (Questions 2 and 3 - the company environment).

## 9.2 Data analysis one

### 9.2.1 Cost information

### 9.2.1.1 Use of cost information for efficiency

On the use of cost information for system efficiency, see table 9.1

	Sig	gnifican	ce of cost	t informa	tion use	at		
Cost information role	company							
	Extremely important	Very important	Important	Not so important	Not important	I do not know		
To comply with procedures of reporting	16%	40%	38%	4%		2%		
Basis for audit	12%	38%	34%	2%	10%	4%		
Planning of period costs (budgeting)	28%	48%	16%	2%	4%	2%		
Management and control of period costs	26%	34%	20%	14%	6%			
Role of standard costing in assessing performance	10%	40%	18%	14%	18%			
Identification of corrective measures	14%	20%	26%	20%	20%			

# Table 9.1: Descriptive analysis of the use of cost information for system efficiency(N = 50)

Source: Analysis of questionnaire

The researcher observed that this result emerged as 'normative' perspective (Bjornenak and Olson, 1999: 332). The normative view is related to traditional (technical) management accounting, not contemporary management accounting. For example, in standard costing or budgeting, the normative perspective is invariably based on ex post benchmarks, *not* on forward-looking market prices. The normative pressure is therefore based on historical observations and data, *not* expectations for future forward planning.

Brignall and Modell (2000) argue that an institutional and political process is associated with budgeting (Wildavsky, 1975; Jonsson, 1982; Covaleski and Dirsmith, 1983; 1986). These researchers focus on the bargaining process inherent in public sector budgeting. They differentiate between 'advocates' (e.g. lower-level managers, or representatives of certain professional groups), and 'guardians' (e.g. senior level staff, and politically-elected officials representing funding-bodies).

Regarding the use of cost information for system efficiency, see table 9.1a

Table 9.1a: Descriptive analysis	of the use of	cost information	for system	efficiency
	(N =50)	) N		

	Significance of cost information use at						
Cost information role	company						
	Extremely important	Very important	Important	Not so important	Not important	I do not know	
To achieve targets of production	4%	8%	6%	20%	60%	2%	
To achieve large profits	4%	10%	2%	14%	70%		
To achieve certain financial measurements	10%	6%	6%	22%	52%	4%	

Source: Analysis of questionnaire

This result found that the economic perspective does *not* play an important role in shaping the policy of the GOGCWS. However, the political and social element *does* play an important role in shaping the GOGCWS policy.

For a summary of the significance of cost information for system efficiency (at the GOGCWS), see table 9.1b

Cost information role	Overall mean <sup>240</sup>
	score <sup>241</sup>
Planning of period costs (budgeting)	2.1
Compliance with procedures of reporting	2.4
Management and control of period costs	2.4
Basis for audit	2.7
Role of standard cost in assessing performance	2.9
Identification of necessary corrective measures	3.1
Perceived importance for meeting budgets	3.3
Inventory Evaluation	3.4
To provide a means of communication	3.5
Measuring efficiency of production performance	3.6
To enhance cost reduction programs	3.6
Performance evaluation of management	3.8
To achieve certain financial measurement	4.1
To achieve target of production	4.3
To achieve a large profit	4.4

Table 9.1b: Reported significance of cost information for system	efficiency used at
the GOGCWS $(N = 50)$	

 $<sup>^{240}</sup>$  One of the most important ways of summarising a distribution of values for a variable, is to establish its central tendency - (the typical value in a distribution). Where do values in a distribution tend to concentrate? Statisticians are referring to a number of different measures when they talk about averages (such as mean, median, mode, Bryman and Cramer, 1990: 82).

Gould (1991), (a palaeontologist who is well known for his popular writings on science), illustrates the first two of these measures of average when he writes: "A politician in power might say with pride, "The mean income of our citizens is \$15000 per year". The leader of the opposition might retort, "But half our citizens make less than \$10000 per year". Both are right, but neither cites a statistic impassive objectivity. The first invokes a mean, the second a median" (1991: 473).

The Mean is the value closest to all the other values. In order to calculate the mean: Total all the cases divided by the number of cases. Residual from the mean: the difference between each case in a distribution and the mean: if this value is positive, the value is above the mean; if it is negative, it is below the mean (Bryman and Cramer, 1999).

<sup>&</sup>lt;sup>241</sup> The range of the Likert scale was 1 = extremely important 2 = very important 3 = important 4 = not so important 5 = not important.

#### 9.2.1.2 Use of cost information for system optimisation

On the use of cost information for system optimisation, see table 9.2.

	Sig	gnifican	ce of cost	t inform:	ation use	at	
Cost information role	company						
	Extremely important	Very important	Important	Not so important	Not important	I do not know	
Planned and actual product costs	46%	34%	10%	6%	2%	2%	
Importance of a flexible budget for manufacturing cost control	6%	30%	40%	8%	12%	4%	
Variance analysis and exception reporting	34%	24%	18%	14%	6%	4%	
To stimulate control of conversion cost	26%	38%	14%	4%	14%	4%	
Importance of marketing cost analysis	14%	16%	32%	8%	28%	2%	

# Table 9.2: Descriptive analysis of the use of cost information for system optimisation (N =50)

Source: Analysis of questionnaire

From the perspective theory, respondents saw management accounting from a hierarchical view (institutional hierarchy) (see section 5.2.2). Management accounting was seen as an instrument for calculation, implementation, and control that belonged to the tactical and administrative level. Management accounting was only one out of several functional disciplines within the organisation (Ansoff, 1965 and Anthony, 1965).

	Sig	gnifican	ce of cost	informa	tion use	at		
Cost information role	company							
	Extremely important	Very important	Important	Not so important	Not important	I do not know		
To maintain certain profit margins	6%	2%	4%	10%	74%	4%		
Product mix decision	4%	8%	8%	22%	58%			
Development of better manufacturing methods	6%	8%	6%	20%	60%			
Importance of product cost as an input to pricing decision	12%	4%		12%	70%	2%		
Transfer prices for internal services	2%	2%	8%	14%	68%	6%		
To enhance equipment productivity	6%	8%	24%	24%	36%	2%		
To motivate people to do better	2%	10%	8%	16%	56%	8%		
To emphasis throughput rate	10%	10%	2%	28%	46%	4%		
Measuring of efficiency and capacity utilisation	4%	16%	10%	26%	40%	4%		
To determine economic production	6%	8%	10%	26%	42%	8%		
Make or buy decisions	6%	10%	22%	26%	36%			
Personal development and workforce improvement	2%	16%	12%	10%	60%			

# Table 9.2a: Descriptive analysis of the use of cost information for systemoptimisation (N =50)

Source: Analysis of questionnaire

The initial analysis shows that management accounting in the GOGCWS had no new techniques<sup>242</sup>. The GOGCWS followed a 'traditional' management accounting system, not contemporary management accounting. It was clear that the social or political element was more important than the economic element, (if cost information was unimportant to determine economic production or enhance productivity). Analysis from the interviews showed that most interviewees recommended that the price of water be determined 'outside' of the GOGCWS itself.

<sup>&</sup>lt;sup>242</sup> The new techniques in this thesis, were based on their importance in current management accounting literature (e.g. activity-based costing, activity-based management, target costing, strategic management accounting, transfer prices).

For a summary of the significance of cost information for system optimisation used at the GOGCWS, see table 9.2b below:

Cost information role	Overall mean score <sup>243</sup>
Planned and actual product cost	1.9
Variance analysis and exception reporting	2.5
To stimulate control of conversion cost	2.5
Importance of flexible budget for manufacturing cost control	3
Importance of marketing cost analysis	3.3
Motivating efficiency improvement	3.7
To enhance equipment productivity	3.8
Make or buy decisions	3.8
Measuring of efficiency and capacity utilisation	3.9
To emphasise throughput rate	4
To determine the economic production amount	4.1
Personal development and workforce improvement	4.1
Product mix decisions	4.2
Development of better manufacturing methods	4.2
Importance of product cost as an input to pricing decision	4.3
To motivate people to do better	4.4
To maintain certain profit margins	4.6
Transfer prices for internal services	4.6

Table 9.2b: Reported significance of cost information for system optimisation usedat the GOGCWS (N =50)

Source: Analysis of questionnaire

.

<sup>&</sup>lt;sup>243</sup> The range of Likert scale was 1 = extremely important 2 = very important 3 = important 4 = not so important 5 = not important.

#### 9.2.1.3 Use of cost information for system strategy

Regarding the use of cost information for system strategy, see table 9.3

	Significance of cost information use at						
Cost information role			com	pany			
	Extremely important	Very important	Important	Not so important	Not important	l do not know	
To forecast the future	14%	34%	28%	6%	14%	4%	
Evaluation of investment projects	10%	30%	20%	18%	20%	2%	
Improvement of product characteristics	6%	10%	4%	22%	58%		
Planned and actual customer	2%	10%	4%	8%	76%		
Importance of competitor cost analysis		4%	2%	2%	92%		
Market share analysis	4%	6%		6%	84%		
New product development	2%			6%	92%		
Market development	6%	2%	2%	4%	86%		
Importance of product configuration		6%		4%	90%		
New product development	2%			6%	92%		
Decisions of company strategy	6%	16%	14%	26%	38%		
Product quality analysis	8%	8%	6%	30%	48%		
Importance of product configuration	6%			4%	90%		

Table 9.3: Descriptive analysis of the use of cost information for system strategy (N =50)

Source: Analysis of questionnaire

The result showed that cost information had no role or a very limited role in shaping the strategy of the GOGCWS. This meant cost management in the GOGCWS was clearly absent. Comparing this result with the rules (documentation) of the GOGCWS (mentioned in section 8.7.6), it showed that the role of accounting was extremely important in supporting the development of strategy. The researcher concluded that there was a 'gap' between the theory and practice of cost management in the GOGCWS.

If cost information was not important for deciding the strategy and competitor cost analysis<sup>244</sup> of the GOGCWS, what factor *did* affect the shaping of strategy policy of the GOGCWS? The researcher observed that political and social elements had more effect than the economic element (see section 10.1.1).

For a summary of the significance of cost information use for system strategy at the GOGCWS, see table 9.3a below:

	245
Cost information role	Overall mean score
To forecast the future	2.8
Evaluation of investment projects	3.1
Decisions of company strategy	3.7
Product quality analysis	4
Importance of technology configuration	4
Improvement of product characteristics	4.2
Planned and actual customer profitability	4.5
Market growth analysis	4.5
Market development	4.6
Importance of product improvement	4.6
Market share analysis	4.7
Importance of competitor cost analysis	4.8
Importance of product configuration	4.8
New product development	4.9

Table 9.3a: Reported significance of cost information for system strategy used	l at
the GOGCWS (N =50)	

Source: Analysis of questionnaire

The result showed that cost management had a very limited role in strategy at the GOGCWS. This may have resulted from the policy of managing the water in the GOGCWS (for the public sector, see section 7.2.4).

<sup>&</sup>lt;sup>244</sup> Because Control of the Egyptian water industry was under a monopoly.

<sup>&</sup>lt;sup>245</sup> The range of the Likert scale was 1 = extremely important 2 = very important 3 = important 4 = not so important 5 = not important.

#### 9.2.2 Company environment

As mentioned in section 9.1, this research concentrated on managers perceptions of competitive intensity and unpredictability (related to three areas of the external environment: product market, factor market and legislation, Al-Hazami, 1995). This research adopted institutional theory (as opposed to contingency theory, see section 2.1). However, it is necessary to examine the core idea of each of them. Institutional theory suggests that organisations gain legitimacy by conforming to external expectations of cost management practice, whilst separating their internal activities from externally-focused symbolic systems. Conversely, contingency theory suggests that organisational practices are driven by both the task environment, and the technical nature of the work. The closer the "fit" between technical features of the environment and cost management practice, leads to an increase in efficiency and effectiveness.

#### 9.2.2.1 Product market

Managers' perception of competitive intensity (CI) and unpredictability (Un) of product market was measured by using 14 items (7 for P (CI) and 7 for P (Un)). Regarding the P (CI) of product market in the GOGCWS, see table 9.4

		Competitive Intensity							
M	arket Conditions		1	2	3	4	5		I do not know
1	Market Change (sales)	Declining	8%	2%	32%	48%	8%	Expanding market	2%
2	Competition intensity	Low	100%					High	
3	Customer variety	One major customer	2%	4%	6%	26%	62%	Large variety	
4	Profit margins	Narrow	82%	6%	6%	4%	2%	High	
5	Product innovation	Well established	38%	18%	10%	18%	16%	Always new products	
6	Nature of competition	Well established competitors	98%	2%				Always new competitors	
7	Price of products	Stable	28%	30%	16%	14%	12%	Fluctuate prices.	

Table 9.4: P (CI) of the product market in the GOGCWS (N =50)

Source: Analysis of questionnaire

				Unpre	dictability		
	Market Conditions	Very predictable	Predictable	Average	Unpredictable	Very Unpredictable	I do not know
1	Market Change (sales)	48%	34%	12%	4%		2%
2	Competition intensity	96%	4%				
3	Customer variety	80%	20%				
4	Profit margins	92%	6%	2%			
5	Product innovation	52%	34%	10%	4%		
6	Nature of competition	96%	4%				
7	Prices of product	78%	18%	4%			

Table 9.4a: P (Un) of the product market in the GOGCWS (N =50)

## 9.2.2.2 Factor market

The perception of (CI) and (Un) of Factor market was measured by using 64 items (32 for P (CI) and 32 for P (Un)) (as illustrated in table 9.5). The factor market was divided into five sub-sectors.

 Table 9.5: Measure of factor market

Sub-sectors of factor market	P (CI) measured by	P (Un) measured by
Raw materials and supply	7 items	7 items
Labour	7 items	7 items
Management	4 items	4 items
Capital and financing	4 items	4 items
Technology	10 items	10 items

A brief descriptive analysis of each sub-sector is given in turn.

## 9.2.2.2.1 Raw material and supply

Regarding the P (CI) of raw materials and supply, see table 9.6

				Com	petitive	Intens	ity		
Ma	rket Conditions		1	2	3	4	5		I do not know
1	Supply	Shortage of supply		2%	8%	22%	68%	Over supply	
2	Source of supply	One major source	10%	8%	8%	40%	30%	Plenty of sources	4%
3	Competition between suppliers	No competition	18%	6%	32%	20%	22%	Intensive competition	2%
4	Cost of raw material	Low	2%	2%	16%	52%	20%	High	8%
5	Delivery time	Short	40%	18%	14%	4%	2%	Long	22%
6	Nature of suppliers	Well established	30%	38%	6%	12%	10%	Always new suppliers	4%
7	Prices of raw material	Stable prices	4%	6%	14%	50%	24%	Fluctuate prices	2%

Table 9.6: P (CI) of raw materials and supply in the GOGCWS (N =50)

Source: Analysis of questionnaire

Regarding the P (Un) of raw materials and supply, see table 9.6a

Table 9.6a: P (Un) of raw materials and supply in the GOGCWS (N =50)

				Unpre	edictability		
	Market Conditions	Very predictable	Predictable	Average	Unpredictable	Very Unpredictable	I do not know
1	Supply	74%	22%	4%			
2	Source of supply	56%	32%	8%			4%
3	Competition between suppliers	52%	30%	10%	6%		2%
4	Cost of raw material	38%	34%	16%	4%		8%
5	Delivery time	44%	24%	8%	2%		22%
6	Nature of supplier	56%	28%	8%	2%		6%
7	Prices or raw material	62%	26%	8%	2%		2%

Source: Analysis of questionnaire

The reason for the above result could have been because water in Egypt was under monopoly control, which prevented competition and achieved predictability for raw materials and supply.

## 9.2.2.2.2 Labour

For the P (CI) related to labour, the result was as shown in table 9.7

				Cor	npetitiv	ve Inter	nsity		
	Market Conditions		1	2	3	4	5		I do not know
1	Availability of skilled labour	Shortage	24%	14%	10%	24%	28%	Stable	
2	Competing employers	Few	72%	16%	8%		4%	Many	
3	Wage rates	Low	6%	8%	74%	12%		High	
4	Trade union	Not militant	28%	10%	24%	28%	10%	Militant	
5	Labour turnover	Low	74%	16%	8%	2%		High	
6	Labour dependency	Low	4%	8%	2%	12%	74%	High	
7	Industrial relation	Poor	4%	4%	36%	22%	28%	Good	6%

 Table 9.7: P (CI) of the labour in the GOGCWS (N =50)

Source: Analysis of questionnaire

For the result of P (Un) related to labour, see table 9.7a

Table 9.7a: P (Un) of the labour in the GOGCWS (N =50)

	· · · · · · · · · · · · · · · · · · ·	Unpredictability										
	Market Conditions	Very predictable	Predictable	Average	Unpredictable	Very Unpredictable	I do not know					
1	Availability of skilled	64% .	28%	8%								
	labour											
2	Competing employers	72%	22%	4%	2%							
3	Wage rates	72%	26%	2%								
4	Trade unions	46%	50%	4%	· • • •							
5	Labour turnover	68%	26%	6%								
6	Labour dependency	80%	16%	2%	2%							
7	Industrial relations	62%	24%	8%			6%					

Source: Analysis of questionnaire

It was observed that the way of managing labour in the GOGCWS created obstacles for efficiency, optimisation or strategy because every worker knew that they would get their salary at the end of each month, and therefore would not change their job.

#### 9.2.2.3 Management

Regarding the P (CI) related to management, see table 9.8

			Competitive Intensity								
M٤	arket Conditions		1	2	3	4	5		I do not know		
1	Availability of qualified managers	Shortage	48%	14%	16%	10%	12%	Stable			
2	Salary of management	Low	48%	26%	26%			High			
3	Competition for good managers	Low	48%	6%	12%	8%	26%	High			
4	Staff turnover	Low	4%		14%	10%	72%	High			

Table 9.8: P (CI) of the management in the GOGCWS (N =50)

Source: Analysis of questionnaire

When asked about competition for "good" managers, the majority of interviewees objected to the word "good", but 52% of interviewees stated that competition was low<sup>246</sup>. This result may have two explanations. One explanation was that while managers did their best, there were regulations which prevented them from achieving what they wanted (institutional theory-regulative structure). The second explanation was that the GOGCWS managers were working in their own interests and not those of the company. The organisation itself gave them no motivation or incentive to work hard. Salaries were high and both managers and workers knew they would get paid regardless of whether they worked or not. This was true of most public sector organisations in Egypt.

<sup>&</sup>lt;sup>246</sup> 34% of respondents claimed that competition among managers was high. Respondents stated that managers compete with each other in order to gain a high position (but not through their actual work).

Regarding the P (Un) of management, see table 9.8a

				Unpro	edictability		
	Market Conditions	Very predictable	Predictable	Average	Unpredictable	Very Unpredictable	I do not know
-1	Availability of qualified managers	66%	32%	2%			
2	Salary of management	66%	26%	8%			
3	Competition for good managers	66%	26%	8%			
4	Staff turnover	72%	24%	4%			

Table 9.8a: P (Un) of the management in the GOGCWS (N =50)

Source: Analysis of questionnaire

A possible reason for these answers was that the method of choosing a manager was dependent on how many years they had worked in the organisation (not on their qualifications or work). This was made clear when interviewees<sup>247</sup>, suggested that an alternative to privatisation would be "choosing effective workers for leadership".

## 9.2.2.4 Capital and financing

For the P (CI) of capital and financing, see table 9.9

			Competitive Intensity									
Market Conditions			1	2	3	4	5		I do not know			
1	Availability of funds	Easy to obtain funds	36%	34%	10%	10%	10%	Hard to obtain fund				
2	Sources of funds	One major source	12%	8%	10%	56%	14%	Many sources				
3	Cost of finance	Low	8%	6%	16%	28%	36%	High	6%			
4	Staff turnover	Low	70%	16%	8%	6%		High				

Table 9.9: P (CI) of Capital and financing in the GOGCWS (N = 50)

Source: Analysis of questionnaire.

<sup>&</sup>lt;sup>247</sup> Accountants in the Participating Accounts Department.

There were influences on the source of funding of Government cost accounting practices. Geiger and Ittner (1996) argue that units relying on 'revolving' funds were more likely to use cost system data (managing activities or programs) than units receiving all of their funding through reimbursement. This could have been true of the GOGCWS. From the initial analysis, it appeared that there was very little focus on using cost information for strategy. There was also a validity between the answers of respondents (who stated that the cost of finance was high) and the actual documentation of the GOGCWS (see section 8.6.4) which showed that governmental subsidisation was 737 million E. P. annually for the GOGCWS. This amount of money would invariably be borrowed from The National Investment Bank at 14% interest, and the result of this would be a subsidisation of 840 million of E.P.

For the P (Un) of capital and financing, see table 9.9a

		Unpredictability										
	Market Conditions	Very	Predictable	Average	Unpredictable	Very	I do not					
		predictable				Unpredictable	know					
1	Availability of funds	60%	32%	8%								
2	Sources of funding	58%	38%	4%								
3	Cost of finance	50%	34%	10%			6%					
4	Staff turnover	66%	32%	2%								

Table 9.9a: P (Un) of capital and finance in the GOGCWS (N =50)

Source: Analysis of questionnaire.

# 9.2.2.5 Technology

For the P (CI) of technology, see table 9.10

				Cor	npetitiv	ve Inter	nsity		
M٤	arket Conditions	Low	1	2	3	4	5	High	I do not know
1	Availability of equipment	Shortage		2%	2%	26%	66%	Stable	4%
2	Equipment suppliers	One major suppliers	2%	4%	18%	28%	38%	Many	10%
3	Competition for equipment.	No competition	72%	10%	8%	4%	2%	Keen competition	4%
4	Delivery of equipment	Short	16%	14%	26%	10%	10%	Long	24%
5	Equipment failure	Usual failure	20%	26%	18%	4%	4%	Frequent failure	28%
6	Nature of manufacturing technology	Simple	28%	28%	24%	6%		Complex	14%
7	Nature of system technology	Simple	54%	14%	20%	2%	4%	Complex	6%
8	Technology development	Negligible change	32%	32%	8%	18%	8%	Common change	2%
9	Equipment obsolescence	Short life	2%	4%	12%	16%	60%	Long life	6%
10	Capital labour substitution	Slow	48%	24%	12%	10%	4%	Fast	2%

Table 9.10: P (CI) of Technology in the GOGCWS (N =50)

Source: Analysis of questionnaire.

It was clear from the above result that there was face validity of the data. For example, on the topic of obsolete equipment, 76% of respondents stated that the organisation's 'equipment' was unchanged, and in their view was likely to remain that way. It was therefore unsurprising when 72% of respondents stated that capital/ labour substitution was slow.

				Unpre	edictability	1	<u></u>
	Market Conditions	Very predictable	Predictable	Average	Unpredictable	Very Unpredictable	I do not know
1	Availability of equipment	58%	32%	6%			.4%
2	Equipment suppliers	52%	28%	10%	2%		8%
3	Competition for equipment.	50%	34%	8%	4%		4%
4	Delivery of equipment	40%	20%	14%	2%		24%
5	Equipment failure	38%	22%	8%	4%		28%
6	Nature of manufacturing technology	42%	32%	12%	2%		12%
7	Nature of system technology	48%	32%	14%	2%		4%
8	Technology development	54%	32%	12%	2%		
9	Equipment obsolescence	60%	28%	4%	4%		4%
10	Capital labour substitution	56%	32%	8%			4%

Table 9.10a: P (Un) of technology in the GOGCWS (N =50)

Although 56% of respondents stated that the technology element was the least significant<sup>248</sup> aspect of the factor market, the researcher observed that the technical aspect was much more important than the financial or management aspect (especially in the water industry). This is because the water industry serves fundamental human requirements (see section 8.4).

#### 9.2.2.3 Legislation

Legislation represented a third of the external environment. The perception of (CI) and (Un) of legislation was measured by using 8 items (4 items for P (CI) and 4 for P (Un)).

<sup>&</sup>lt;sup>248</sup> For the rank priority significance of the factor market, 64% respondents stated that capital and financing was the first element. 42% respondents stated that raw materials and supply was the second element. 44% respondents chose labour as the third element. 36% respondents saw management as the fourth element, while 56% respondents stated that technology was the least important element.

On the topic of the P (CI), see table 9.11

Market Conditions				Inte	nsive C	Compet	ition		
		Low	1	2	3	4	5	High	I do not know
1	Regulators in the industry	One major regulator	20%	12%		16%	52%	Many regulator	
2	A mount of legislation	Small	22%	18%	2%	18%	36%	Large	4%
3	Nature of legislation	Well established	32%	30%	18%	14%	6%	Frequent change	
4	Effect of legislation	Negligible				12%	88%	Heavy	

Table 9.11: P (CI) of the legislation in the GOGCWS

Source: Analysis of questionnaire.

These results showed the validity of the data. For example, 68% of respondents stated that there were too many regulators in the water industry, and this answer was consistent with the results from the analysis of documentation (see section 8.4). There were many regulators (both technical and financial). Furthermore, 54% of respondents stated that there was too much legislation and this was clear from section A1.2.1.1 (which included general laws and decrees, laws related to management development, etc). In addition all respondents (100%) stated that legislation exerted a strong influence.

For the P (Un) of legislation, see table 9.11a

Table 9.11a: P (Un) of the legislation in the GOGCWS (N = 50)

Market Conditions		Unpredictability						
		Very predictable	Predictable	Average	Unpredictable	Very Unpredictable	I do not know	
1	Regulators in the industry	68%	20%	8%	2%		2%	
2	Amount of legislation	62%	22%	6%	4%		6%	
3	Nature of legislation	44%	42%	14%				
4	Effect of legislation	76%	24%					

Source: Analysis of questionnaire.

There has been much institutional accounting research Covaleski, et al., 1993; 1996; Mezias, 1990; Geiger and Ittner, 1996 all focused on external legitimisation in the shaping of accounting practice. Covaleski, et al., (1993) claimed that pressures exerted by the US Federal Government were crucial in the development of a case-mix accounting system in the health industry. Mezias (1990) described how various institutional factors (laws, expectations and professionalism) influenced the formulation of US accounting policies. Geiger and Ittner (1996) studied the influence of sources of funding and legislative requirements on government cost accounting practices. Their hypothesis was that "mandated government cost accounting systems are not associated with greater internal use of cost accounting data". This is particularly relevant to the GOGCWS research study.

The researcher previously held the view that legislation in Egypt played an important role in cost management. This view was confirmed by the respondents ranking of the significance of various sectors of the external environment, (product market, factor market and legislation). The results were as follows:

Seventy two percent of respondents stated that *legislation* was the most important element. 28% stated that the *factor market* was the most important element, whilst no respondents considered the product market as the most important element in the environment sector.

"Was there a relationship between cost management and P (CI) and P (Un) which relates to the three areas of the external environment (product market, factor market and legislation)? A difficulty facing the researcher here was how do we know each scale measures a single idea? For example, how does the researcher know that the set 15 items do actually measure efficiency, etc. In other words, how can we assess the reliability and validity of a construct? In order to answer these questions, the reliability and factor analysis (using SPSS) was conducted as follows

## 9.3 Reliability and validity of construct

#### 9.3.1 Reliability

The reliability of a measure refers to its consistency<sup>249</sup> (Bryman and Cramer, 1999: P.64). Internal reliability was important in relation to multiple-item scales. It raised the question of whether each scale was measuring a single idea and whether the items that make up the scale were internally consistent (p.65). Cronbach's alpha<sup>250</sup> is computed in terms of the average inter-correlation among items measuring the concept. Alpha<sup>251</sup> coefficient was used to examine reliability and validity of the constructs. Alpha coefficient provided general support for the constructs and instruments used in this research and also indicated that some refinement may be necessary. Measures of reliability are shown in table 9.12

Table 9.12: Alpha coefficients	of Reliability	for the items	of cost	management (N
	=50)			

Indicator	Alpha
	coefficients
Cost practice (reporting)- Different uses of cost information	II _II II II II II II II II II _II II
Uses of cost information for system maintenance and efficiency	0.82
Uses of cost information for system optimisation	0.89
Uses of cost information for system adaptation	0.91

Source: Analysis of questionnaire.

<sup>&</sup>lt;sup>249</sup> This entailed two separate aspects - external reliability and internal reliability. External reliability referred to the degree of consistency of a measure over several times (Bryman and Cramer, 1999: 64). This was not applicable to the GOGCWS research study.

<sup>&</sup>lt;sup>250</sup> There were two procedures. Split-half and cronbach's alpha could be readily computed in SPSS for estimating internal reliability.

 $<sup>^{251}</sup>$  The closer Cronbach's alpha is to 1, the higher the internal consistency reliability. Therefore, reliabilities less than 0.6 are generally considered to be poor, those in the 0.7 range to be acceptable, and those over 0.8 to be good.

	Competitive	Unpredictability
External environment	Intensity	
Product market	6405	.3741
Factor market:		
Raw material	.2374	.7753
Labour	0612	.6580
Management	.2360	.7074
Capital and financing	.0607	.6151
Technology	.6130	.7732
Legislation	.5221	.7876

Table 9.12a: Alpha coefficients of Reliability for factor analysis of P (CI) and P (Un) of the GOGCWS environment (N =50)

Source: Analysis of questionnaire.

Values which were nearer to one indicated greater reliability than values nearer to zero. The measures suggest that respondents answered in a similar way to each indicator of cost management, but in different way to company environment. The next step was to use factor analysis, this was used to examine the reliability and validity of the construct.

#### 9.3.2 Factor analysis

#### 9.3.2.1 Use of accounting for Cost management

#### **9.3.2.1.1** Use of cost information for system efficiency

By carrying out a factor analysis<sup>252</sup> of use of cost information for efficiency, the initial step was to compute a correlation matrix for 15 items (see table A2.12). The aim of this

<sup>&</sup>lt;sup>252</sup> Factor analysis enables us to assess the factorial validity of the questions which made up the scales (by telling us the extent to which they seemed to be measuring the same concepts or variables). Factor analysis was used for three main purposes: (a) to assess the degree to which items were tapping the same concept (b) To determine the degree to which they could reduce to a smaller set (c) To try to make sense of the bewildering complexity of social behaviour by reducing it to a more limited number of factors. Bryman and Cramer (1999) argue that two uses of factor analysis can be distinguished as (a) Exploratory - in which the relationships between various variables were examined without determining the extent to which the results fit a particular model (b) Confirmatory factor analysis - comparing the solution found, against a hypothetical one.

step was to decide whether to carry out factor analysis<sup>233</sup>. According to the result of a correlation matrix<sup>254</sup>, the researcher thought it worthwhile to conduct a factor analysis.

There were several kinds of methods available on SPSS, such as alpha, image, maximum likelihood factoring, principal components and factor analysis (called principal-axis factoring in SPSS). Bryman and Cramer (1999) claim that the two widely used forms for factor analysis are 'principal component' and 'principal-axis factoring'. The researcher chose the principal component method<sup>255</sup>. The rationale for choosing principal-components analysis, was that it analysed all the variance of a score or variable, including its unique<sup>256</sup> variance.

Table 9.13 illustrates the SPSS output showing the initial factors produced by a principal component analysis (for using cost information for efficiency) and the amount of variance they account for (their 'eigenvalue').

<sup>&</sup>lt;sup>253</sup> If there is no significant correlation between these items, they are unrelated and we would not expect them to form one or more factor analysis (Bryman and Cramer, 1999: 273).

<sup>&</sup>lt;sup>254</sup> Most of the items were significantly correlated at over the 0.05 level (either positively or negatively, with one another, see table A2.12a). They suggested that they might constitute one or more factors.

<sup>&</sup>lt;sup>255</sup>The difference between 'principle-components' analysis and 'principal-axis' factoring is how they handle unique variance.

<sup>&</sup>lt;sup>256</sup> Unique variance is a combination of both specific variance and Error variance (for more details on 'variance' see Bryman and Cramer, 1999: 274).

	Initial Eigenvalues			Extract	Rotation		
Compone					Loadings		
nt	Total	% of	Cumulati	Total	% of	Cumulati	Total
		Variance	ve		Variance	ve	
1	4.608	30.718	30.718	4.608	30.718	30.718	3.238
2	1.777	11.845	42.563	1.777	11.845	42.563	2.238
3	1.597	10.649	53.212	1.597	10.649	53.212	2.474
4	1.368	9.120	62.332	1.368	9.120	62.332	2.671
5	1.169	7.796	70.129	1.169	7.796	70.129	1.823
6	.770	5.136	75.265				
7	.733	4.886	80.151				
8	.692	4.615	84.766				
9	.489	3.318	88.084				
10	.468	3.118	91.202				
11	.406	2.710	93.912				
12	.295	1.969	95.881				
13	.278	1.856	97.736				
14	.191	1.275	99.011				
15	.148	.989	100.000			· · · · · · · · · · · · · · · · · · ·	

Table 9.13: Total variance explained for Using Cost Information for Efficien	Table 9.13	3: Tota	al variance	explained f	or Using	Cost ]	Information	for Efficienc
--	------------	---------	-------------	-------------	----------	--------	-------------	---------------

The next step was to reduce the number of variables. There were two main criteria used for deciding which factors to exclude, - 'Kaiser's Criterion' and the 'Scree'<sup>257</sup> test. This research study followed Kaiser's criterion<sup>258</sup>, which was to select those factors which had an eigenvalue of greater than one (this meant the first five factors were selected for

<sup>&</sup>lt;sup>257</sup> The Scree test was introduced by Cattell in 1966. In the Scree test, a graph was drawn of the descending variance accounted for by the factors initially extracted (Bryman and Cramer, 1999: 282). The term "Scree", in fact, was a geological one for describing the debris found at the bottom of a rocky slope and implied that these factors were not very important.

<sup>&</sup>lt;sup>258</sup> The rationale for choosing a Scree test was that the choice of criterion might depend on the size of the average communalities and the number of variables (Bryman and Cramer, 1999: 277). Furthermore, the Kaiser criterion had been recommended for situations where the number of variables was less than 30. The number of using cost information for efficiency was 15 variables, and the average communality was greater than 0.07 (the average communality for using cost information for efficiency was just over 0.07).

using cost information for efficiency. The principal components produced a 'Component matrix' (see table A2.2), which showed the relationship between each item and a factor was expressed as a correlation or loading.

The first factors extracted from the analysis were those which accounted for the maximum amount of variance. As a consequence, what they present might not be easy to interpret, (since items would not correlate as highly with them as they might). In fact, most of the items would fall on the first factor, although their correlation within it may not be that high.

In order to increase the interpretability of factors, they were rotated to maximise the loading of some of the items. A number of ways were developed to rotate factors, Bryman and Cramer (1999) argue that the two most commonly used methods are orthogonal rotation and oblique rotation. Oblique rotation<sup>259</sup> was used. Oblique rotation produced by the oblimin method in SPSS gave three matrixes; - a pattern matrix<sup>260</sup>, the structure matrix<sup>261</sup>, and the component matrix<sup>262</sup> (see tables A2.2, A2.2a and A2.2b). The structure matrix was made up of weights which reflected the unique variance each factor contributed to a variable. This was a matrix generally used to interpret the factors. According to the structure matrix of using cost information for efficiency, the researcher ended up with five factors. For the illustration of these five factors and the items included in each factor, see table 9.13a

<sup>&</sup>lt;sup>259</sup> The advantage of orthogonal rotation was that the information the factor provided was not redundant, (since a person's score on one factor is unrelated to their score on another). The disadvantage of orthogonal rotation was that the factors may have been forced to be unrelated, whereas in real life they might be related. This might be less likely with oblique rotation (Bryman and Cramer, 1999: 279).

<sup>&</sup>lt;sup>260</sup> A pattern matrix consisted of calculations between the variables and the factors (see table A2.2a).

<sup>&</sup>lt;sup>261</sup> See table A2.2b.

 $<sup>^{262}</sup>$  The component matrix showed the correlations between the factors. It was difficult to estimate the amount of variance accounted for oblique factors since the variance was shared between the correlated factors (see table A2.2).

Purposes of cost information for efficiency	Mean score	Factor
	of efficiency	
1- To achieve certain financial measurements	3.3	Measurement.
2- To enhance cost reduction programs		
3- Identification of necessary correctives		
4- Compliance with procedures of reporting		
5- Perceived importance for meeting budgets		
1- Inventory valuation	3.1	Evaluation.
2- Basis for audit		
1- To provide a means of communication	3.8	Accidental variable.
2- Measuring efficiency of production		
performance		
3- To achieve large profits		
1- Management and control for period costs	2.5	Cost efficiency.
2- Role of standard cost in assessing		
performance		
3- Planning of period costs (budgeting)		
1- Performance evaluation for management	4.1	Performance.
2- To achieve production target		

Table 9.13a: Factors of Use of Cost Information for Efficiency (N = 50)

#### 9.3.2.1.2 Use of cost information for system optimisation

By carrying out a factor analysis using the same steps as followed in the use of cost information for system maintenance and efficiency, the system failed to create a pattern matrix and a structure matrix. The researcher looked for the reason for failure and concluded that it might be that the average of communality was less than 0.70. Hence, Scree test and Varimax<sup>263</sup> rotation with Kaiser normalisation were used (see table A2.3).

<sup>&</sup>lt;sup>263</sup> According to Varimax rotation, it may not be appropriate to use the significance level of the factor loading since it depends on the size of the sample. There are two ways: Firstly, items or variables which correlate less than 0.3 with a factor, are omitted from consideration. An alternative criterion to use is the correlation above which no item correlates highly, with more than one factor. The advantage of this rule is that factors are interpreted in terms of items unique to them.

The researcher ended up with five factors for use of cost information for system optimisation. An illustration of these five factors and items included in each factor are detailed in table 9.14 below.

Purposes of cost information for	Mean score of	Factor
optimisation	optimisation	х. Х.
1- Development of better manufacturing	4.1	Human resources
2- Personal development and workforce		and productivity
improvement		development.
3- To motivate people to do better		
4- To enhance equipment productivity		
5- Importance of marketing cost analysis		
6- Transfer prices for internal services		
1- Determine the economic production	4.2	Economic evaluation
amount		
2- To maintain certain profit margins		
3- Measuring of efficiency and capability		
utilisation		
4- Importance of product cost as input to		
pricing decision		
1- Variance analysis and exception reporting	2.5	Cost optimisation
2- Planned and actual product cost		
3- Importance of flexible budget		
4- To stimulate control of conversion cost		
1- Motivating efficiency improvement	3.5	Motivation
2- To emphasise throughput rate		
1- product mix decision	4	Decision processes
2- Make or buy decision		

Table 9.14: Factors of Use of Cost Information for Optimisation (N =50)

Source: Analysis of questionnaire

## 9.3.2.1.3 Use of cost information for strategic objectives

By carrying out factor analysis using the same steps as followed in uses of cost information for system maintenance and efficiency, the researcher ended up with three factors for use of cost information for system strategy (see table A2.4). An illustration

of these three factors (and items included in each factor), are detailed in table 9.15 below.

Purposes of cost information for Strategic	Mean score of	Factor
objective	strategy	
1- Importance of technology configuration	4	Market evaluation.
2- Improvement of product characteristics		
3- Product quality analysis		
4- Decisions of company strategy		
5- Market growth analysis		
6- To forecast the future	м. Н	
7- Market development		
1- Importance of competitor cost analysis	4	Investment/competit
2- Evaluation of investment projects		or cost.
1- Importance of product configuration	4.6	Product
2- Importance of product improvement		development.
3- New product development		
4- Planned and actual customer profitability		
5- Market share analysis		

Table 9.15: <b>J</b>	Factors of Us	e of Cost I	nformation	for Strategy	(N = 50)
----------------------	---------------	-------------	------------	--------------	----------

Source: Analysis of questionnaire.

## 9.3.2.2 Company (The GOGCWS Environment)

#### 9.3.2.2.1 Product market

#### P (CI)

.

By carrying out factor analysis using the same steps as followed in uses of cost information for system maintenance and efficiency, the SPSS gave this computer message. "There are fewer than two cases. At least one of the variables has zero variance, or there is only one variable in the analysis. No further statistics will be computed".

Hence the researcher considered all items, which included one factor in the product market (see table 9.16).

	Factor	Raw material and supply (Competition).
1	Product market	1- Market change (sales)
		2- Competition intensity
		3- Customer variety
		4- Profit margins
		5- Product innovation
	·	6- Nature of competition
		7- Price of products

Table 9.16: Factor of P (CI) in the product market (N =50)

## P (Un)

Factor analysis produced three factors for the product market - unpredictability (see table A2.5). For an illustration of these three factors and items included in each factor, see table 9.16a below.

	Factor	Mean	Unpredictability	Mean score
1	Price change	1.5	1- Product innovation	1.7
			2- Price of product	1.3
2	<b>Customer/competition</b>	1.1	1- Customer variety	1.2
			2- Nature of competition	1
3	Growth	1.5	1- Market change (sales)	1.8
			2- Profit margin	1.1

Table 9.16a: Factors of P (Un) of the product market (N =50)

Source: Analysis of questionnaire.

#### 9.3.2.2.2 Factor market

#### 9.3.2.2.1 Raw materials and supply

#### **P (CI)**

Factor analysis produced three factors for raw materials and supply - competition (see table A2.6). For an illustration of these three factors and items included in each factor, see table 9.17.

	Factor	Raw material and supply (Competition)
1	Suppliers	1- Sources of supply
		2- Competition between suppliers
		3- Nature of suppliers
2	Raw materials	1- Cost of raw material
1		2- Price of raw material
3	Time	1- Supply
		2- Delivery time

Table 9.17: Factors of P (CI) of raw materials and supply (N =50)

## P (Un)

Factor analysis produced two factors for raw materials and supply - P (Un) (see table A2.6a). For an illustration of these two factors and items included in each of them, see table 9.17a. below.

Factor		Mean	Raw material and supply	Mean score
			(unpredictability)	
1		1.8	1- Sources of supply	1.7
			2- Competition between suppliers	1.8
			3- Cost of raw material	2.2
			4- Nature of suppliers	1.8
			5- Price of raw material	1.6
2		1.9	1- Supply	1.3
			2- Delivery time	2.6

Table 9.17a: Factors of P (Un) of raw materials and supply (N =50)

Source: Analysis of questionnaire.

#### 9.3.2.2.2.2 Labour

## **P (CI)**

Factor analysis produced two factors for labour - P (CI) (see table A2.7). For an illustration of these two factors and items included in each of them, see table 9.18.

	Factor	Labour (Competition)
1		1- Availability of skilled labour
		2- Wage rate
		3- Labour turnover
		4- Relations between water organisations.
2		1- Competing employers
		2- Trade union

Table 9.18: Factors of P (CI) of labour (N =50)

#### P (Un)

Factor analysis produced two factors for labour - P (Un) (see table A2.7a). For an illustration of these two factors and items included in each of them, see table 9.18a below.

Factor Mean Labour (Unpredict		Labour (Unpredictability).	Mean score	
1		1.5	1- Availability of skilled labour	1.4
			2- Wage rate	1.3
			3- Trade union	1.6
			4- Labour turnover	1.4
			5- Relations between water	1.7
			organisations	
2		1.3	1- Competing employers	1.4
			2- Labour dependency	1.3

Table 9.18a: Factors market of P (Un) of labour (N =50)

Source: analysis of questionnaire.

#### 9.3.2.2.3 Management

#### P (CI)

Factor analysis produced two factors for management - P (CI) (see table A2.8). For an illustration of these two factors and items included in each of them, see table 9.19.

	Factor	Management (Competition).
1		1- Availability of qualified managers
		2- Staff turnover
2		1- Salary of management
		2- Competition for good managers

Table 9.19: Factors of P (CI) of management (N =50)

## P (Un)

A factor analysis produced all items related to only one factor (see table A2.8a). For an illustration of the items (see table 9.19a).

Table 9.19a: Factor of P (Un) of management (N =50)

Factor		Mean	Mean Management (Unpredictability).	
1		1.38	1- Availability of qualified managers	1.4
			2- Salary of management	1.4
			3- Competition for good managers	1.4
			4- Staff turnover	1.3

Source: Analysis of questionnaire.

# 9.3.2.2.4 Capital and financing

## **P (CI)**

Factor analysis produced two factors for capital and financing - P (CI) (see table A2.9). For an illustration of these two factors and items included in each of them, see table 9.20.

	1	able 9.20: Fact	ors of P	(CI) 01	capital	and finan	cing (N = 5)	U)
_								

	Factor	Capital and financing (Competition)
1		1- Cost of finance
		2- Staff turnover
2		1- Availability of funds
		2- Sources of funding

Source: Analysis of questionnaire.

## P (Un)

Factor analysis showed that all items were related to only one factor (see table A2.9a). For illustration of the items see table 9.20a.

	Factor	Mean	Capital and financing (Unpredictability)	Mean score
1		1.5	1- Availability of funding	1.5
			2- Source of funds	1.5
			3- Cost of finance	1.8
			4- Staff turnover	1.4

# Table 9.20a: Factor of P (Un) of capital and financing (N =50)

Source: Analysis of questionnaire.

## 9.3.2.2.5 Technology

## P (CI)

Factor analysis produced three factors for technology - P (CI) (see table A2.10). For an illustration of these three factors and items included in each factor, see table 9.21.

	Factor	Technology (Competition)
1		1- Availability of equipment
		2- Equipment suppliers
		3- Delivery of equipment
		4- Equipment failure
		5- Nature of manufacturing technology
2		1- Obsolete equipment
		2- Capital labour substitution
3		1- Competition for equipment
		2- Nature of system technology
		3- Technology development

Table 9.21: Factors of P (CI) of technology (N =50)

Source: Analysis of questionnaire.

## P (Un)

Factor analysis produced three factors for technology - P (Un) (see table A2.10a). For an illustration of these three factors and items included in each factor, see table 9.21a.

	FactorMeanTechnology (Unpredictability)		Mean score	
1		1.7	1- Competition for equipment	1.8
			2- Nature of system technology	1.9
			3- Technology development	1.6
			4- Capital labour substitution	1.7
2		2.45	1- Equipment suppliers	1.9
			2- Delivery equipment	2.7
			3- Equipment failure	2.9
			4- Nature of manufacturing	2.2
			technology	
3		1.66	1- Availability of equipment	1.6
			2- Obsolete equipment	1.7

Table 9.21a: Factors of P (Un) of technology (N =50)

# 9.3.2.2.3 Legislation

## P (CI)

Factor analysis produced two factors for legislation - P (CI) (see table A2.11). For an illustration of these items, see table 9.22.

	Factor	Legislation (Competition)
1		1- Regulators in the industry
-		2- Number of legislation
		3- Effect of legislation
2		1- Nature of legislation

Source: Analysis of questionnaire.

# P (Un)

Factor analysis showed that all items related to only one factor (see table A2.11a). For an illustration of the items, see table 9.22a

.
Table 9.22a: Factor of P (Un) of legislation (N =50)

	Factor	Mean	Legislation (Unpredictability).	Mean score
1		1.6	1- Regulators in the industry	1.5
			2- A mount of legislation	1.8
			3- Nature of legislation	1.7
			4- Effect of legislation	1.2

Source: Analysis of questionnaire.

According to the reliabilities and factor analysis, the researcher ended up with the following factors (see table 9.23).

Table 9.23: Factors of cost management and P (CI) and P (Un) of the	e product
market, factor market and legislation	

	Subject matte	r	Number of
Main	Branch	Sub-branch	factors
Cost management	Efficiency		Five
	Optimisation		Five
	Strategy		Three
Competitive	Product market		One
Intensity	Factor market	Raw material and supply	Three
		Labour	Two
		Management	Two
		Capital and financing	Two
		Technology	Three
	Legislation		Two
Unpredictability	Product market		Three
	Factor market	Raw material and supply	Three
		Labour	Two
		Management	One
		Capital and financing	One
		Technology	Three
	Legislation		One

Source: Analysis of questionnaire.

These numerous factors hindered the researcher in his attempt to address the question "Was there a relationship between cost management and perception of competitive intensity and unpredictability. It was therefore important to restructure the data as outlined in the next section.

## 9.4 Restructured data

## 9.4.1 Cost management

Three factors represented cost management. The first factor represented efficiency, which resulted from the aggregate of 5 factors of efficiency (see table 9.13a). The second factor represented optimisation, which resulted from the aggregate of 5 factors of optimisation (see table 9.14). The third factor represented strategy which resulted from the aggregate of 3 factors of strategy (see table 9.15). The rationale for choosing the aggregate factor for efficiency into 5 factors, optimisation into 5 factors and *strategy* into 3 factors, - the alpha coefficients of reliability and validity construct (see table 9.12) were very good. Furthermore, the correlation and the significance level were very good between the original items of efficiency (see table A2.12a), and the same was noted for optimisation (see table A2.13a) and strategy (see table A2.14a).

### 9.4.2 The GOGCWS environment

#### 9.4.2.1 Product market

One factor for P (CI) of the product market was acceptable (-0.64). This resulted from the aggregate of 7 items (see table 9.16) according to the computer message of the SPSS and the alpha coefficient of these items. Three factors emerged for the P (Un) of the product market as a result from factor analysis (see table 9.16a), because the alpha coefficients were poor and there was no strong correlation between them (see tables A2.15 and A2.15a).

#### 9.4.2.2 Factor market

### 9.4.2.2.1 Raw material and supply

Three factors for the P (CI) of raw material and supply resulted from factor analysis (see table 9.17), because the alpha coefficient was poor (0.24) and there was no strong correlation between them (see table A2.16). One factor for P (Un) of raw material which resulted from the aggregate of two factors came from SPSS (see table 9.17a), because

the alpha coefficients and the level of significance between the items formulating the two factors was acceptable (see tables 9.12a, A2.17 and A2.17a).

### 9.4.2.2.2 Labour

Two factors for the P (CI) of labour resulted from factor analysis (see table 9.18) because the alpha coefficient was very poor between the two factors (-0.16), and between the original items (-0.0612). There was also no strong correlation found between them (see table A2.18). One factor for P (Un) of labour resulted from the aggregate between the two factors which resulted from factor analysis (see table 9.18a). The reason for the combination between the two factors was because (although the alpha coefficient was poor between the two factors "0.40"), the alpha coefficient between the original items was acceptable (0.658). There was also a strong correlation and significance between the items formulating these two factors (see tables A2.19 and A2.19a).

### 9.4.2.2.3 Management

Two factors for P (CI) of management resulted from factor analysis (see table 9.19) because the alpha coefficient was very poor between the two factors (0.0004) and between the original items (.2360). In addition, there was no strong correlation found between them (see table A2.20). One factor for P (Un) of management resulted from factor analysis (see table A9.19a), because the items had strong correlation and the significance between them (tables A2.21 and A2.21a) and the alpha coefficients between items, were acceptable (0.7074).

## 9.4.2.2.4 Capital and financing

Two factors for P (CI) of capital and financing resulted from factor analysis (see table 9.20) because the alpha coefficient was very poor between the two factors (0.0605) and also between the original items (.0607). In addition, there was no strong correlation between them (see table A2.22). One factor for P (Un) of capital and financing resulted from factor analysis (see table 9.20a), because the items had strong correlation and the significance between each of them (tables A2.23 and A2.23a) and the alpha coefficients between the items, were acceptable (0.6151).

#### 9.4.2.2.5 Technology

Three factors for P (CI) of technology resulted from factor analysis (see table 9.21) because the alpha coefficient was very poor between the three factors (0.2461), and there was no strong correlation or significance between these original items (see tables A2.24 and A2.24a). Three factors for the P (Un) of technology resulted from factor analysis (see table 9.21a), because the alpha coefficient was not acceptable between these three factors (0.5505), and there was no strong correlation or significance between these three factors (see tables A2.25 and A2.25a).

#### 9.4.2.3 Legislation

Two factors for P (CI) of legislation resulted from factor analysis (see table 9.22), because the alpha coefficient was very poor between the two factors (0.0586) and there was no strong correlation or significance found between the original items (see tables A2.26 and A2.26a). One factor for P (Un) of legislation resulted from SPSS (see table 9.22a) because the alpha coefficient was acceptable between the items (0.7876), and there was a strong correlation and significance between the items which formulate the factor (see tables A2.27 and A2.27a).

#### 9.4.3 New factors of cost management and external environment

In conclusion, the researcher ended up with several factors. Some of them represented cost management while others represented the P (CI) and P (Un) in the external environment. These related to three areas: the product market, the factor market (raw material and supply, labour, management, capital and financing, and technology) and legislation. The number of these factors was illustrated in table 9.24.

	Subject matte	r	Number of
Main	Branch	Sub-branch	factors/ Mean
			score
Cost	Efficiency		One (3.3)
management	Optimisation		One (3.7)
U	Strategy		One (3.96)
P(CI)	Product market		One
	Factor market	Raw material and supply	Three
		Labour	Тwo
		Management	Two
		Capital and financing	Тwo
		Technology	Three
	Legislation		Two
P (Un)	Product market		Three (1.3)
	Factor market	Raw material and supply	One (1.8)
		Labour	One (1.4)
		Management	One (1.4)
		Capital and financing	One (1.5)
		Technology	Three (1.95)
	Legislation		One (1.5)

 Table 9.24: New factors of cost management and P (CI) and P (Un) of product market, factor market and legislation

Source: Analysis of questionnaire.

## 9.5 Data analysis two

Was there any relationship between cost management and the P (CI) that related to the product market, factor market and legislation? The results were as shown in tables 9.25 and 9.25a.

## Table 9.25: Pearson correlations between Cost practice and perception of theGOGCWS environment related to P (CI) (N= 50).

						I	factor m	arket							
Cost	Product	Raw	materia	nl &	L	abour	Mana	geme	Car	oital &	Tec	hnolog	у	Legisl	ation
management	market		supply				n	t	fina	ancing					
		F1	F2	<b>F</b> 3	<b>F</b> 1	F2	F1	F2	F1	F2	F1	<b>F2</b>	F3	<b>F1</b>	<b>F2</b>
Efficiency	276*	.265*	.246*	1	.290*						.266*			.310*	<u> </u>
Optimisation	328*	.299*		1	.247*					.314*	.357**	1			
Strategy	356**	.269*				256*	.242*								

Table 9.25a: The significance levels of the Pearson correlations between Cost practice and perception of the GOGCWS environment related to P (CI) (N=50).

						1	Factor ma	arket							
Cost management	Product market	Raw	<sup>7</sup> materia supply	ul &	L	abour	Mana n	geme t	Car fin:	oital & ancing	Teo	chnolog	у.	Legisl	ation
		F1	F2	<b>F</b> 3	F1	F2	<b>F</b> 1	F2	F1	F2	F1	F2	<b>F</b> 3	<b>F1</b>	<b>F2</b>
Efficiency	.026	.032	.042		.021						.031			.014	<u> </u>
Optimisation	.010	.018			.042	<u> </u>				.013	.006				<del>                                      </del>
Strategy	.006	.029				.036	.045								

\*\*. Correlation is significant at the 0.01 level (1-tailed).

\*. Correlation is significant at the 0.05 level (1-tailed).

The result showed that there was a negative correlation between the product market and cost management (efficiency, optimisation and strategy). There was a positive correlation between legislation and efficiency, but there was no correlation between legislation and optimisation or strategy. Regarding the factor market, raw material and supply had a positive correlation with cost management (efficiency, optimisation and strategy). Labour had a positive correlation with efficiency and optimisation, but had a negative correlation with strategy. Management had a negative correlation with efficiency and no correlation with optimisation. Capital and financing had a negative correlation with strategy. Finally, technology had a positive correlation with efficiency and had no correlation with strategy.

Was there any relationship between cost management and P (Un) related to the product market, factor market and legislation? The results were as shown in tables 9.26 and 9.26a.

Table 9.26: Pearson correlations between Cost practice and perception of th	e
GOGCWS environment related to P (Un) (N= 50).	

Cost	Pro	duct r	narket			Facto	or market				Legisla
management				Raw	Labour	Manage	Capital and	1	Fechnology		tion
	F1	F2	F3	material and supply		ment	financing	FI	F2	F3	
Efficiency										1	
Optimisation				.272*					.318*		
Strategy									.242*		

# Table 9.26a: The significance levels of the Pearson correlations between Cost practice and perception of the GOGCWS environment related to P (CI) (N= 50).

Cost	Pro	duct n	narket			Facto	or market				Legisla
management				Raw .	Labour	Manage	Capital and	Te	chnology		tion
	F1	F2	F3	material and supply		ment	financing	F1	F2	F3	
Efficiency											
Optimisation				.028					.012		
Strategy									.045		

\*\*. Correlation is significant at the 0.01 level (1-tailed).

\*. Correlation is significant at the 0.05 level (1-tailed).

The result showed that there was no correlation between the product market and cost management (efficiency, optimisation and strategy), and none between legislation and cost management. Regarding the factor market, labour and management, and capital and financing had no correlation with cost management (efficiency, optimisation and strategy). Raw material and supply had a positive correlation only with optimisation and no correlation with efficiency or strategy. Technology had a positive correlation with optimisation with optimisation and strategy, but had no correlation with efficiency.

## 9.6 Discussion

In conclusion, the researcher found that there was a correlation between the cost practice and perception of the GOGCWS environment (illustrated in tables 9.27 and 9.27a).

Table 9.27: Pearson correlations between Cost practice and perception of GOGCWS environment (product market, factor market and P (Un) (N= 50).

		H	actor mark	et		Cost Practice		Legis	lation		Product ma
	Raw	Technol	Capital	Manage	Labour			P (CI)	P (Un)		P (CI) P
	material	ogy	investment	ment							
P (CI)	.265*	.266*	•	•	.290*	Efficiency		.310*	•		276*
P (Un)	•	٠	•	•	•						
P (CI)	.299*	.357*	.314*	•	.247*	Optimisation		•	•		328
P (Un)	.272*	.318*	•	٠	٠			-			
P (CI)	.269	•	•	.242*	256*	Strategy		•	•		356**
P (Un)	.•	.242*	•	•	•						
• Corr	elation do	es not exis	st + Correls	ation is sign	nificant, po	ositive < 0.5 + or - Correlati	on is significan	it, negativ	e < 0.5	_	

Table 9.27a: The significance levels of the Pearson correlations between Cost practice and perception of GOGCWS environment (product market, factor market and legislation) related to P (CI) and P (Un) (N= 50).

		Ţ	actor mark	et		Cost Practice
	Raw	Technol	Capital	Manage	Labour	
	material	ogy	investment	ment		
P (CI)	.032	.031	•	•	.021	Efficiency
P (Un)	•	•	•	•	•	
P (CI)	.018	.006	.013	•	.042	Optimisation
P (Un)	.028	.012	•	•	•	
P (CI)	.029	•	•	.045	.036	Strategy
P (Un)	•	.045	•	•	•	

	,		1	
lation	P (Un)	•	•	•
Legis	P (CI)	.014	•	•

ma	Р	•	•	•
Product )	P (CI)	.026	.010	.006

There was a negative relationship between efficiency and the P (CI) related to the product market. There was a positive relationship with the P (CI) related to legislation, raw material and supply, labour, and technology and efficiency. The researcher referred to this as 'technical' rather than 'economic' efficiency (see section 4.1). Furthermore, the P (CI) in relation to legislation, appeared to play a more important role than both the product market and factor market.

There was a negative relationship between optimisation and the P (CI) related to product market. There was a positive relationship with the P (CI) related to raw material and supply, labour, technology, capital investment and optimisation. In addition, there was a positive relationship between optimisation and the P (Un) related to raw material and supply and technology. The researcher observed that the managers tried to achieve results in the organisation, but there were certain rules and controls which hindered them.

There was a negative relationship between strategy and the P (CI) related to product market and labour. There was a positive relationship between raw material and supply, management and strategy. There was a positive relationship between strategy and the P (Un) related to technology. It was clear that technology was the most important aspect because the water service deals with fundamental human requirements.

In short, legislation and the factor market had *positive* pressure of cost management, while the product market had *negative* pressure of cost management. It was important to examine the relationship between cost management and the P (CI) and P (Un) that are related to the external environment (from all three managerial levels - senior 'high' level, medium level and junior 'low' level), see tables 9.28, 9.28a, 9.29, 9.29a, 9.30 and 9.30a).

Table 9.28: Pearson correlations between Cost practice and perception of GOGCWS environment (product market, factor market and legislation) related to P (CI) and P (Un) (N= 9)/ managerial level (1= high level).

		H	actor mark	et		Cost Practice		Legisl	ation		Product	ma
	Raw	Technol	Capital	Manage	Labour			P (CI)	P (Un)	.1	P (CI)	
	material	ogy	investment	ment								
P (CI)	•	895**	•	.918**	769*	Efficiency	<u> </u>	**667.	•	Į	*669	
P (Un)	•	•	•	•	•							
P (CI)	.792*	792*	•	.832**	673*	Optimisation	1	.944**	•		807**	
P (Un)	•	•	•	•	•							
P (CI)	.845**	778*	678*	.823**	725*	Strategy		.923**	•		814**	
P (Un)	•	•	•	•	•							
¢	-	•	(	•		• • •	,	.	ļ			

 Correlation is significant, negative < 0.5</li> Correlation does not exist + Correlation is significant, positive < 0.5 + or •

Table 9.28a: The significance levels of the Pearson correlations between Cost practice and perception of GOGCWS environment (product market, factor market and legislation) related to P (CI) and P (Un) (N= 9)/ managerial level (1= high level).

		F	actor mark	iet	
	Raw	Technol	Capital	Manage	Labour
	material	ogy	investment	ment	
P (CI)	•	.001	•	000.	.016
P (Un)	•	•	•	•	•
P (CI)	.011	•	•	.005	.047
P (Un)	•	•	•	•	•.
P (CI)	.004	.014	.045	.006	.027
P (Un)	•	•	•	•	•

Cost Practice	Legislati	on
	P (CI)	P (
Efficiency	.013	•
Optimisation	000.	•
Strategy	000.	•

ma	P (	•	•	•
Product	P (CI)	.049	600'	.008

P (Un)

The perception of the senior level GOGCWS managers was that "legislation, raw material, management and product market, technology, capital investment and labour" all had negative pressure of cost management. The perception of senior managers in the GOGCWS was similar to that of the GOGCWS<sup>264</sup> itself (except that technology, capital investment and labour had positive pressure of cost management).

<sup>&</sup>lt;sup>264</sup> Legislation and factor market have positive pressure of cost management. Product market has negative pressure of cost management.

Table 9.29: Pearson correlations between Cost practice and perception of GOGCWS environment (product market, factor market and P (Un) (N= 14)/ managerial level (2= medium level).

		F:	actor mark	et		Cost Practice		Legis	ation		Product	ma
	Raw	Technol	Capital	Manage	Labour		<u> </u>	P (CI)	P (Un)		P (CI)	ď
	material	ogy	investment	ment								
P (CI)	•	634*	•	•	•	Efficiency		•	•		•	•
P (Un)	•	•	•	•	•							
P (CI)	•	.730**	•	•	•	Optimisation		•	•		•	•
P (Un)	•	•	•	•	•							_
P (CI)	•	.560*	•	•	•	Strategy	1	•	•		•	•
P (Un)	•	•	•	•	•							
<ul> <li>Corr</li> </ul>	elation do	es not exist	t + Correls	ation is sign	ifficant, po	sitive $< 0.5 + \text{or} - \text{Correlation}$ is	significant	t, negativ	e < 0.5	•		

Table 9.29a: The significance levels of the Pearson correlations between Cost practice and perception of GOGCWS environment (product market, factor market and legislation) related to P (CI) and P (Un) (N= 14)/ managerial level (2= medium level).

		Γ <b>ι</b>	actor mark	et	
	Raw	Technol	Capital	Manage	Labour
	material	ogy	investment	ment	-
P (CI)	•	.015	•	•	•
P (Un)	•	•	•	•	•
P (CI)	•	.003	•	•	•
P (Un)	•	•	•	•	•
P (CI)	•	.037	•	•	•
P (Un)	•	•	•	•	•





ma	Р	•	•	•
Product	P (CI)	•	•	•

The perception of the GOGCWS managers (medium level) was that technology had negative pressure of efficiency, and positive pressure of optimisation and strategy (see table 9.29). However, they did not comment on the product market or legislation, or the remaining elements of the factor market. Comparing this result to the result from the perception of the GOGCWS itself, it did not fully represent the perception of the GOGCWS.

The only response from the junior managers level to perceptions of the impact of the external environment on cost practices, was capital investment had positive pressure of optimisation, and legislation had negative pressure of optimisation, (as illustrated in table 9.30).



		H	actor mark	tet		Cost Practice		Legisl	ation		Product	ma
	Raw	Technol	Capital	Manage	Labour		<u> </u>	P (CI)	P (Un)	3	P (CI)	Р
	material	ogy	investment	ment								
P (CI)	•	•	•	•	•	Efficiency		•	•		•	
P (Un)	•	•	•	•	•.							
P (CI)	•	•	.386*	•	•	Optimisation	1	400*	•		•	
P (Un)	•	•	•	•	٠							
P (CI)	•	•	•	•	•	Strategy		•	•		•	
P (Un)	•	•	•	•	•							
<ul> <li>Cori</li> </ul>	elation do	es not exis	st + Correl	ation is sigi	nificant, po	sitive $< 0.5 + $ or - Correlation is s	significant,	, negative	e < 0.5	-		

Table 9.30a: The significance levels of the Pearson correlations between Cost practice and perception of GOGCWS environment (product market, factor market and legislation) related to P (CI) and P (Un) (N= 27)/ managerial level (3= low level).

		Ξ.	actor mark	et	
	'Raw	Technol	Capital	Manage	Labour
	material	ogy	investment	ment	
P (CI)	•	•	•	•	•
P (Un)	•	•	•	•	•
P (CI)	•	•	.047	•	•
P (Un)	•	•	•	•	•
P (CI)	•	•	•	•	•
P (Un)	•	•	•	•	•







It was necessary to examine the relationship between cost management and the P (CI) and P (Un) (which are related to the external environment), from a functional viewpoint (accounting, technical and administrative (see tables 9.31. 9.32 and 9.33).

Table 9.31: Pearson correlations between Cost practice and perception of the Chief of Cost Department of GOGCWS environment (product market, factor market and legislation) related to P (CI) and P (Un)/ Mean score - Functional level (accounting)

		Ĭ.	actor mark	tet		Cost Pr	actice		Legisla	tion	<u> </u>	Produc
												marke
	Raw	Techn	Capital	Manage	Labou			P	(CI)	P	- <u>L</u>	P (CI)
	materi	ology	investme	ment	ч					( <b>Un</b> )		
	al		nt									<u></u>
P (CI)	2.86	2.7	2.5	3.75	2.14	Efficiency	3.7		4.5	H		2.14
Ь	1.14	2.5	1	1	1.14							
(Un)								;				
P (CI)	7	2.7	2.5	3.75	2.14	Optimisatio	n 4.55		4.5		'	2.14
Р	1.14	2.5	1	1	1.14							
(Un)												<u></u>
P (CI)	2.86	2.7	2	3.75	2.14	Strategy	5.1		4.5	1	1	2.14
Ь	1.14	2.5	1	1	1.14					<u> </u>		
(Un)												

- Correlation is significant, negative < 0.5 Correlation does not exist + Correlation is significant, positive < 0.5 + or</li>

Table 9.32: Pearson correlations between Cost practice and perception of the Vice Chairman for Technical Affairs of GOGCWS environment (product market, factor market and legislation) related to P (CI) and P (Un)/ (Mean score) - Functional level (technical).

		F			ſ		annour).	ſ	-	ļ			
		<b>-</b>	actor mark	set .			Cost Pract	ice		Legisl	lation		Produ
_													marke
_	Raw	Techn	Capital	Manage	Labou				<u> </u>	P (CI)	P		P (CI)
	materi	ology	investme	ment	ų						(Un)		
	al		nt										
P (CI)	7	2.4	1.5	2.5	2.41		Ifficiency	1.53		2	1	- I	2.86
4	1	1	1		1							***	
( <b>Un</b> )													
P (CI)	7	2.4	5	2.5	2.14	0	<b>Dptimisation</b>	1.5		2	1		2.86
d		1	7	-	1								
(Un)												14. 18	
P (CI)	2	1	2	2.5	2.14	S	trategy	1.43		2	1		2.86
	1	2.4	1	Ţ	1								
( <b>Un</b> )			v					-		<u></u>			
Corr	elation do	es not exis	st + Correl	ation is sign	ificant, po	ositive $< 0$ .	5 + or - Corr	elation is sig	gnificant	, negative	e < 0.5	_	

Table 9.33: Pearson correlations between Cost practice and perception of the Chief of General Chief of Meters Administration ofGOGCWS environment (product market, factor market and legislation) related to P (CI) and P (Un)/ (Mean scored) - Functional level (administrative)

		Ť	actor mark	ket		L	Cost Practi	ice	L	Legisl	ation	<b></b>	Pro	duc
													mai	rke
	Raw	Techn	Capital	Manage	Labou					P (CI)	Р	<u> </u>	P (CI)	
	materi	ology	investme	ment	r						(Un)			0
	al		nt	~										
P (CI)	2.6	1.4	1	4	1.9		fficiency	3.33	<u>.</u>	4.5	1.25		2.29	
Ь	1.4	2.1	7	1.25				<u>.</u>						
(Un)						-								
P (CI)	5	2.1	7	4	1.86	0	)ptimisation	3.78		4.5	1.25		2.29	
d	1.43	1.4	Ħ	1.25	1									
(Un)									· •					
P (CI)	2.57	2.1	5	4	1.86		Strategy	4.5	<u></u>	4.5	1.25	I	2.29	
ď	1.43	1.4	<b>H</b>	1.25	1									
(Un)														
• Cor	relation do	es not exis	st + Correl	ation is sigr	ifficant, p	ositive $< 0$	.5 + or - Corr	relation is sig	nificant	, negative	3 < 0.5	_		

The results showed (the analysis of managerial and functional levels), that legislation is the most important factor affecting cost management. This result could be explained by institutional theory. Institutional theorists argue that pressures to conform to accepted practices, can be particularly powerful in government organisations (Meyer and Rowan, 1977; DiMaggio and Powell, 1983; Gupta et al., 1994; Geiger and Ittner, 1996). Since the survival of government units depend primarily on the support of external constituents, (and only secondarily on actual performance, Gupta et al., 1994), conforming to accepted social norms and external requirements is necessary to maintain organisational legitimacy (thereby strengthening support and ensuring continued funding). Scott (1987) argued that in institutional environments (such as government organisations), environmental agents have the authority to impose organisational practices on subordinate units, or specify conditions for remaining eligible for the continuation of funding. Consequently, subordinate organisations are likely to show little resistance to the implementation of mandated practices, but the changes will tend to be superficial and loosely coupled to participant action.

This result could be explained by neo-institutional theory which is a theory "based on the premise that organisations respond to pressures from their institutional environment<sup>265</sup> and adopt structures and/or procedures that are socially accepted as being the appropriate organisational choice" (Carpenter and Feroz, 2001). This shows that the GOGCWS relates to political organisation. This means the GOGCWS's legitimacy is threatened by monopoly. Procedures and management play no direct part in generating resources for the GOGCWS, it is assumed that the customer is not interested in such things but only in the GOGCWS product.

## 9.7 Conclusion

This chapter identified the perceptions of managers in the GOGCWS regarding legislation, product market and factor market which affect accounting work in the GOGCWS generally, and cost management especially. From a strategic cost management perspective, this chapter elevated environmental factors in importance as

<sup>&</sup>lt;sup>265</sup> For a definition of the term "institutional environment" (see section 7.2.5).

major drivers of cost and in particular, highlighted their related interaction with management strategy and policy.

This chapter observed that management accounting in the GOGCWS was based on ex post benchmarks (historical observations), not on forward-looking market prices (expectations for the future). The result was, that management accounting had no new techniques operating in the GOGCWS. Furthermore, management accounting in the GOGCWS was based on a hierarchical perspective (institutional hierarchy). Cost information had no role (or a very limited) role in shaping the strategy of the GOGCWS. In addition, government funds for the GOGCWS had an effect on the use of cost information.

The economic element did *not* play an important role in shaping the policy of the GOGCWS. The political or social element did play an important role in shaping the policy of the GOGCWS. In other words, the GOGCWS represents a specific embodiment of the role "political and legal structures play, in shaping the institutional frameworks within which various organisations operate" (Scott, 1987: 409; March and Olsen, 1989; Bealing, et al., 1996). The analysis also suggests that the GOGCWS adopted institutionalised structures in order to gain legitimacy and social support. This suggestion is not new to institutional theory. It has been suggested by several authors (e.g. Meyer and Rowan, 1977; DiMaggio and Powell, 1983; 1991). Furthermore, the political, social and cultural elements had strong influence on the decision to privatise and regulate the water industry in Egypt.

Social context shaped economic theory, and accounting was both independent of, and connected to economic theory. The way of managing labour in the GOGCWS created obstacles for efficiency, optimisation and strategy. Furthermore, managers of the GOGCWS tried to achieve results, but there were enforced regulations which prevented them from achieving what they wanted (institutional theory- regulative structure). The researcher noted that the most important element affecting the strategy of the GOGCWS was legislation, the factor market and the product market.

## Chapter 10

## **External and Internal Environments of the GOGCWS**

## **10.0 Introduction**

The objective of this research was to understand, describe and explain the role of cost management within the water industry. Institutional theory was adopted as a framework for explaining the role of cost management in the Egyptian Water Industry via a case study of the GOGCWS. Institutional theory focuses on both the external (Covaleski, et al., 1993; 1996; Mezias, 1990) and internal environment (Burns and Scapens, 2000). The aim of this chapter is to discuss the external and internal environment of the General Organisation for the Greater Cairo Water Supply. The chapter is divided into three sections.

Section one examines research evidence from the external environment. It discusses Egyptian government policy in its management of the water industry. It also examines the perceptions of the GOGCWS managers on the issue of privatisation in the water industry. It looks at the supply and demand of water in the Egyptian water services industry.

Section two examines evidence from the internal environment. It outlines the current cost practice used in the GOGCWS. The final section sums up the main conclusions and implications of the previous sections.

## **10.1 The External environment**

#### **10.1.1 Government policy**

The Egyptian Prime Minister has said Egypt is committed to economic reform despite the economic slowdown it has suffered over the past two years (XINHUA, 05/02/2002). He confirmed that privatisation would continue. He argued that it helps improve competitiveness. The new Egyptian government signalled that the privatisation process will broaden to include not only state enterprises but also the public utilities and publicsector authorities (Business Middle East, 16/03/2000). However, Egypt's privatisation programme is in no way straightforward and an array of obstacles must be negotiated, many of which are not immediately clear.

The partial privatisation of Egypt's state electricity companies began with a law, passed by parliament and ratified by the president. It allowed the country's eight regional distribution companies to be transferred to a joint stock company. The Egyptian cabinet determined how much capital in the joint stock company would be offered to stock market investors (Modern Power Systems, 98). However, investors were not allowed to exceed 49 percent of the shares in each of the eight companies. The new company was responsible for power generation and buying-power from private projects. It also controlled interconnection between Egypt and its neighbouring countries.

This research does not aim to compare the water industry in Egypt with that of France and England. Rather it aims to offer alternative governance structures for the water industry in different countries (such as the UK where water is provided by a highly regulated market and France where water services is hybrid structure). By holding up other structures as alternatives, the Egyptian water industry may now consider other control mechanisms (such as market or hybrid).

The focus of this research was the water industry in Egypt. As mentioned in section 7.2.5, by Williamson, 1985; 1996; Spicer and Ballew, 1983; Colbert and Spicer, 1995; Dietrich, 2001), the management of the water industry in Egypt was hierarchical. The aim of this section is to examine the perceptions of the GOGCWS managers on the issue of privatisation (a governance structure in the water industry). The results were as shown in table 10.1

000C WS: (II - 57)					
	Managerial level			Total	
Privatisation or regulation	High	Medium	Low		
Object to privatisation		10%	32%	42%	
Supportive of privatisation		10.5%	10.5%	21%	
Did not object or agree about privatisation.	3%	3%	5%	11%	
Refused to talk about privatisation	3%	5%	18%	26%	
Total	6%	28.5%	65.5%	100%	

# Table 10.1: Perception of privatisation or regulation of the water industry in the GOGCWS. (N =37)

Source: Analysis of interviews.

The interviewees who were opposed to privatisation gave reasons for their objections. Interviewee response has been divided into four categories: Cost, price, control and water requirements. Typical responses included the following:

- Privatisation will lead to a decrease in both salaries and workers<sup>266</sup>.
- Privatisation will lead to price<sup>267</sup> increases and go against public interest<sup>268</sup>.
- Price increases will affect the indigent <sup>269</sup>.
- A price increase would affect the social element<sup>270</sup>.
- It would affect the political position<sup>271</sup>.
- The most important element is the political element $^{272}$ .
- Water is an essential life element<sup>273</sup> and life's artery<sup>274</sup>.

<sup>268</sup> Auditing in the maturities Department.

<sup>269</sup> Chief of Loans Accounts and Documentary Credit Department.

<sup>270</sup> Chief of the Bills Auditing and Stamp Department and Accountant in the Deduction and Modifications Department.

<sup>&</sup>lt;sup>266</sup> Accountant in the Participating Accounts Department.

<sup>&</sup>lt;sup>267</sup> Accountant in the Cost Department, Accountant in the Cost Department, Accountant in the Cost Department, Accountant in the Central Department and Accountant in the Participating Accounts Administration.

<sup>&</sup>lt;sup>271</sup> Chief of the Commitment Department.

<sup>&</sup>lt;sup>272</sup> Accountant in the Cost Department.

<sup>&</sup>lt;sup>273</sup> Accountant in the Bills Auditing and Stamp Department.

<sup>&</sup>lt;sup>274</sup> Accountant in the Cost Department.

- The General Organisation for the Greater Cairo Water Supply provides water to 25% of the population in Egypt, so how can we let 25% of the population be under the control of the private sector<sup>275</sup>?
- We cannot have water under anyone's  $control^{276}$ .

Results from the interviews show that 42% of respondents were opposed to privatisation. This is consistent with the basic premise of institutional theory which posits that "an organisation's tendency is to conform to established norms, traditions and social influences in both their internal and external environments. This conformity extends to their structures and practices (as in the case of the GOGCWS). Institutional theory claims that successful governments are those that gain support and legitimacy by conforming to social pressures" (Carpenter and Feroz, 2001).

It is further argued that these results could be explained in terms of the religious beliefs of the Egyptian people who follow the faith of Islam. In Islam, water laws prohibit the monopoly of water by those nearest the source (International El-Aharam, 2000). Islamic laws forbid those with a plentiful supply of water to keep it as surplus. These laws also prohibit the sale of water (even if it is surplus to requirements). Islamic law forbids the wasting and excessive use of water, and advocates water-cleaning or anti-pollution programmes.

It was noted that at senior managerial level, no one objected to privatisation. Half of the junior managerial levels objected (32%). The senior managerial levels have more of an overview of the management of the organisation and this could account for this result. Senior managers realised the GOGCWS needs to change from a hierarchically-based organisation to one market-based (or a mix between the two). Consultants Black and Veatch International had made similar suggestions (see section 8.7.7).

None of the senior managerial staff were in favour of privatisation. This may be a fear of the consequences of going against government policy. Douglas (1986) argues that individuals see answers as being "right" only if they fit in with institutional thinking.

<sup>&</sup>lt;sup>275</sup> Chief of commitment.

<sup>&</sup>lt;sup>276</sup> Accountant in the Cost Department.

This pressure to conform, prevents employees from giving their own opinions, out of fear that they may conflict with the institution's ethos.

A possible reason for the high percentage of interviewees (26%) who refused to talk about privatisation at all, may have been born out of fear of talking against the policy of the government. It is clear from sections 8.1 and 8.4 that the GOGCWS had a regulative pillar (the third pillar in institutional theory, Scott, 1995; see section 2.3). *Fear* is central to the regulative pillar (North, 1990 and Scott, 1995). Hence institutional pressures may work in constraining the activities of the GOGCWS managers.

21% of respondents who agreed with privatisation in the water industry did not approve of selling the water companies to the private sector. Interviewees wanted to see the Egyptian water industry run on a basis of both privatisation *and* regulation. In other words, they wanted to adopt a hybrid approach similar to that of France, see section 7.2.3). This was observed when The Chief of Statistics Department stated that "Privatisation will increase quality but will also increase price. In order to solve this problem, the ownership of the organisation should be between the private sector and the state, but the greater percentage of ownership should be with the state (over 50%)".

The interviewees, who did not oppose or agree with privatisation (11%), stressed the importance of the social element. This was clear when an accountant in the Costs department stated "It is impossible to privatise the water industry because investors want a profit margin and this means an increase in price". This statement showed there was a very negative view of privatisation in the GOGCWS.

Some interviewees suggested alternatives to privatisation of the water industry. These suggestions were classified into five categories: leadership, regulation and control, cost and price, loss and consumption and privatisation. Typical suggestions were "Leaders<sup>277</sup> chosen should be both judicious and competent"<sup>278</sup>. With regard to regulation and control, respondents suggested restrictions and the calling to account of management at every level, (even the chairman of the organisation<sup>279</sup>). They advocate exercising stricter

<sup>&</sup>lt;sup>277</sup> Accountant in the Participating Accounts Department.

<sup>&</sup>lt;sup>278</sup> Accountant in the participating Accounts Administration.

<sup>&</sup>lt;sup>279</sup> Chief of the Bills Auditing Department (Collection).

controls on senior management. These should be put into effect rather than merely exist on paper<sup>280</sup>.

On the subject of cost and price, the interviewees suggested a price increase. They recommended obtaining a subsidy from the state to make up the difference between price and cost<sup>281</sup>, (thereby reducing costs through continuous maintenance of equipment and a reduction in water consumption)<sup>282</sup>. The focus of this category was clearly important. This was apparent when the Chief of the Balance Sheet Administration stated, "The only solution is to look at methods which can reduce costs and raise prices".

Regarding loss and consumption of water, respondents suggested reducing the loss of water by following new technology<sup>283</sup>, and putting a meter in each flat would lead to less consumer consumption of water<sup>284</sup>. On the issue of privatisation, respondents suggested privatising distribution, management and collection, but without transferring ownership of the organisation<sup>285</sup>. Conversely, they were against the privatisation of production<sup>286</sup>,

Although this research is based on a single company, the company itself provides water for 25% of the population in Egypt. This research argued that "the social context shaped economic theory, and accounting was both independent of and related to economic theory". Carpenter and Feroz (2001) claim that isomorphic pressures will cause organisations to become increasingly homogenous (within given domains), and lead them to conform to expectations of the wider institutional environment.

<sup>&</sup>lt;sup>280</sup> Accountant in the Deduction and Modifications Department.

<sup>&</sup>lt;sup>281</sup> Accountant in the Planned Budget Department.

<sup>&</sup>lt;sup>282</sup> Accountant in the Cost Department and another accountant in the Cost Department.

<sup>&</sup>lt;sup>283</sup> Chief of Loans and Grants Department.

<sup>&</sup>lt;sup>284</sup> Accountant in the Bills Auditing and Stamp Department

<sup>&</sup>lt;sup>285</sup> Accountant in the Planned Budget Department.

<sup>&</sup>lt;sup>286</sup> Chief of the Central Accounts Department.

The Chief of the Documentary Credit Department supported this argument when he stated that "privatisation in the water industry leads to a higher quality and decreases cost by about 50%, but he adds, we should consider the social element". He believes that if management of the water industry in Egypt was privatised (market based), it would be more beneficial. He emphasised the social element which opposed privatisation in the Egyptian water industry.

The importance of the social element is made clear by the argument put forward by Young (1996) and Douglas (1986). Young (1996) contended that individuals draw upon their institutional commitments when recognising problems which require decision-making, (and in suggesting specific proposals as feasible and correct "solutions" to such problems). A social element underlies all thinking and problem-solving. In recognising the social element, Douglas (1986) argues that individuals see answers as being "right" only if they fit in with institutional thinking. This pressure to conform therefore prevents employees from giving their own opinion out of fear that they may conflict with the institution's ethos.

### **10.1.2 Supply and Demand of water in Egypt**

Demand for water is growing in the 21st century. The most important factor affecting demand of water is the rise in population. For example, the population of Egypt rose from 55 million in 1994 to 63 million in the year 2000. It is expected to increase to 86 million by the year 2025. The demand for water will inevitably increase. The future looks bleak if Egypt does not succeed in formulating and implementing a water policy which can match the limited fresh water supply with increasing demand. The per capita water resources are expected to drop from  $922 m^3$  per year (1990) to  $337 m^3$  in the year 2025. This will potentially change the focus from cost management of supply, to cost management of demand. The Ministry of Public Works and Water Resources (MPWWR) in Egypt had highlighted several cases - (especially in the distant regions of the Nile) - where the cost of the desalination of seawater was less than the cost of transferred Nile water. They concluded that the desalination of seawater should be increased. De-salinated water in Egypt was approximately 30 million  $m^3$ /year (International El-Aharam, 4/3/2000). If the present management practice and cropping patterns continue, it could mean that up to 60 % of agricultural land would not be irrigated.

The main problems facing the water industry of Egypt (see section 7.2.4) and the GOGCWS, is management of water demand. This problem has arisen for several reasons. The increase in population is one such reason. For example, the population of Cairo is 13 million. By 2025, three millions tourists are expected to visit the city per day. This poses a potential problem for the demand on water supplies.

The following example illustrates the potential problems an increased demand for water is likely to produce. In 1968, the population of Giza was 610,000. Today it is one of the largest cities in the world (the GOGCWS, no date source given, a: 17). The population of Shoubra El-Khama was 185,000 in 1968. In 1986 it was 700,000. Today it is the fourth largest city in Egypt (Egyptian Geographic Association, no date source given: 2).

Regarding price, the GOGCWS charges different prices (i.e. business sector rates differ from those of domestic houses). Table 10.2 presents an outline of (1) the purpose of water (2) the rate of each purpose (3) quantities of water and (4) the value of water sold in 1999/2000.

Purpose	Rate	Ouantity $m^3$	Value E. P. <sup>287</sup>
		2	
Houses	.12, .16	900,091,063	115,006,510.72
Fixed rate	.12	5,063,982	607,676.27
Places of Worship	.08	2,639,883	211,190.64
Society clubs	.10	4,654,748	465,474.8
Charitable associations	.15	3,684,640	552,696.00
Health insurance hospitals	.23	1,379,018	317,174.14
Unions	.25	323,970	80,992.50
Bakeries and garages	.28	3,912,553	1,095,514.84
Government factories	.30	1,097,853	329,355.90
Building order	.34	5,055,170	1,718,757.80
Commercial shops	.36	44,158,915	15,897,209.40
Clubs	.50	511,982	255,991.00
Large factories	.60	31,665,959	18,999,575.40
Embassies	.75	977,885	733,413.75
Tourist investment	1.25	12,835,303	16,044,128.75
Water treatment	.10	1,037,487	103,748.70
Investment treatment	.15	1,037,202	155,580.30
Annual fixed rate		181,177	18,117.70
Total		1,020,308,790	172,593,108.61

 Table 10.2: Purposes of water and the quantities for the year 1999/2000

Source: This information was obtained during fieldwork from the accounts department of the GOGCWS.

The cost of water supply outweighs the revenue of water supply. Taking the cost and average revenue of water houses as an example. The cost per  $m^3 = 0.46$  pence (see section 1.5). By contrast, average revenue per  $m^3$  for houses = 0.16 pence. This meant the uncover cost of supply of water houses = 0.30 \* 900,091,063 = 270,027,318 E. P.

<sup>&</sup>lt;sup>287</sup> E. P. means Egyptian pound.

#### **10.2 Internal environment**

### **10.2.1 Current cost practice in the GOGCWS**

The management accounting procedure used in the GOGCWS (see tables 9.1 - 9.3a) was consistent with the development of literature in management accounting up until 1970 (see table 3.2). The GOGCWS used budgets (see section 8.7.3) and full absorption costing (see section 8.7.4). Its budget was governed by functional bureaucracy (and contributed to the pursuit of the rationale in the allocation of resources). Furthermore, the emphasis on budgeting and standard costing (see section 9.2.1.1) resulted in increased control within the GOGCWS. Brignall and Modell, 2000; Wildavsky, 1975; Jonsson, 1982; Covaleski and Dirsmith, 1983; 1986 all found that budgeting was associated with institutional and political processes (see section 9.2.1.1)

The budget of the GOGCWS can be described in terms of DiMaggio and Powell's (1983: 563) definition: "Budgeting is one specific form of coercive isomorphism that governments often require for funding". Covaleski and Dirsmith (1989: 585) adopted an institutional perspective to examine the role of power in the budgetary process. They found that the state's budget, (which is the end product of many processes of institutionalisation), is controlled by issues of power and by individuals working in their own self-interests.

Was there dissatisfaction with the conventional costing system at the GOGCWS? Conventional costing systems allocated overhead costs on simplistic measures (such as direct labour, etc). These systems within the GOGCWS ignored overhead costs for calculating the costs of products (Innes and Mitchell, 1995 and Kaplan 1994, see section 4.3). Why was activity-based costing not practised in spite of its many advantages? - (as outlined in section 4.3). It was observed that the main reason that the GOGCWS had no authority or choice in its accounting system was because it was a public sector economic organisation. The Unified Accounting System was applied on an economic organisation within the public sector by law, (incremental model). This goes back to the argument of Carpenter and Feroz (2001), who found that previous governmental accounting research (based on economic theory) ignored how institutional and organisational pressures constrain accounting choice in the public sector.

On the topic of activity-based management (mentioned in section 8.7.5) it was noted that ABM was not used in the GOGCWS because the organisation took the functional view rather than the process view. The functional view sees costs as best controlled by department managers (see section 8.3). Managers were responsible for minimising the variance between budget and actual costs by cost element (see sections 8.7.3 and 8.8). Cost variance analysis in the GOGCWS seemed to be designed in a way which meant managers were responsible for the variances under their control. This meant that the emphasis of the cost model used in the GOGCWS was on efficient use of resources. This is what Harmmer and Champy, 1993; Cooper et al., 1992 and Trussel and Bitner, 1998 call a 'traditional' view of cost management (see section 4.4). In contrast the ABM view sees costs as best controlled by managing the workload, eliminating non-value-added activities and continuously improving value-added activities. This would then be an effective use of resources (Brimson, 1996).

Which methods of cost management were best suited to the GOGCWS? Berry, et al., (1995: 55) found that "the structures of traditional cost accounting (e.g. full absorption costing) fit beautifully with the notion of functional bureaucracy. These structures are a classic case of dis-aggregation of hierarchic elements to lower-order activities".

As illustrated in table 8.30, the GOGCWS focuses on operational and capital budgets. A budget was more than a financial statement. It was a record of how real sources (expressed in monetary terms) allocated different activities. It was observed that the budget process at the GOGCWS followed the incremental model of budgeting. (This observation was based on analysis of documentation<sup>288</sup> and staff perception at the GOGCWS<sup>289</sup>). 'Incrementalism' refers to a particular view of the budget decision-making process (Jonsson, 1982). What occurred during a given period determined what would happen in the future at the GOGCWS. It was expressed as "This year's budget is estimated <sup>290</sup> by the amount spent last year".

<sup>&</sup>lt;sup>288</sup> See figure 8.5

<sup>&</sup>lt;sup>289</sup> See figure 8.6 (analysis of interviews) and table 8.30 (analysis of questionnaires).

<sup>&</sup>lt;sup>290</sup> This was the case of the GOGCWS (see figure 8.6).

#### 10.2.2 Managers' perceptions of cost management

Cost management was defined in this thesis as has three elements, - efficiency, optimisation and strategy. Table 10.3 shows when managers of the GOGCWS were asked, "Do you have cost management for efficiency?" The answer across the board was yes (92%). However, when actual use of cost information for efficiency was analysed, the mean score shows that there *is* use of cost information for efficiency. However, it was very low for optimisation and almost disappeared for strategy.

Cost management	Do you have cost management? (Yes)	The use of cost information for (Mean score <sup>291</sup> )
Efficiency	92%	3.3
Optimisation	76%	3.7
Strategy	70%	3.96

**Table 10.3: Managers Perception of Cost Information in the GOGCWS** 

Source: Analysis of questionnaire

The mean scores in the three Likert categories (in table 10.3) appear similar, however, a big difference can be detected between them. It should be noted that this is the average mean score and respondents of the questionnaire numbered only fifty. The average mean score appears to show a difference (i.e. less than 3.5 is 'important' whereas higher than 3.5 is 'not important').

As mentioned in section 9.2.1.1, the use of cost information for efficiency was measured by fifteen items (eight of which had an overall mean score of less than 3.5. six items showed an overall mean score higher than 3.5, see table 9.1b). The use of cost information optimisation was measured by eighteen items. Five items had an overall mean score of less than 3.5, while thirteen items had an overall mean score of higher than 3.5 (see table 9.2b). The use of cost information for strategy was measured by fourteen items. Two items had an overall mean score of less than 3.5 (see table 9.2b). The use of cost information for strategy was measured by fourteen items. Two items had an overall mean score of less than 3.5 (see table 9.2b).

<sup>&</sup>lt;sup>291</sup> The range of the Likert scale was 1 =extremely important, 2 =very important, 3 =important, 4 =not so important, 5 =Not important and 6 =I do not know.

Managers' perception of legislation is the most important factor in cost management for the GOGCWS followed by the factor market. Perception of the product market had no impact on cost management at the GOGCWS (see chapter 9). This was unsurprising given that the hypothesis working in this research was: "If managers perceive a competitive, uncertain environment, then cost management will inevitably increase". As it is, the GOGCWS runs efficiently (see section 9.2.1.1). However the company suffered from a lack of change and focus in the organisation itself. The managers had no strategic orientation.

The results (see tables 10.3) show that managers of the GOGCWS realised the accounting system used in the organisation was inadequate. It was observed that many of the suggestions put forward by academics<sup>292</sup> (regarding improvement in management accounting) were similar to those advocated by interviewees within the GOGCWS. For example, Coates and Longden (1989) were among the first to note that computer technology would obviate the need for standard costing systems and make it possible to work entirely on-line with actual costs. Similarly interviewees<sup>293</sup> of the GOGCWS felt "it would be beneficial to introduce computers into the company and make use of them".

### **10.3 Conclusion**

This chapter shows how institutional pressure was worked in shaping management policy of the Egyptian water companies. It was noted the GOGCWS was 'boxed-in' and was not receptive to new ideas.

This chapter highlighted the method of cost management was used in the GOGCWS which was based on budget, full absorption costing and standard costing. It also illustrated there was no change of cost management in the GOGCWS that had been further explained in the following chapter.

The chapter showed that the GOGCWS's managers were not stupid, or ignorant. It also highlighted the managers of the GOGCWS realised the method of cost management

<sup>&</sup>lt;sup>292</sup> (Coates and Longden, 1989; Littler and Sweeting, 1989; Hope and Hope, 1995; Cooper, 1996).

<sup>&</sup>lt;sup>293</sup> Accountant in the Cost Department, Accountant in the Cost Department, Accountant in the Participating Accounts Administration, Accountant of the Statistics Department.

was not relevant and there was a need for change (see section 10.2.2). This chapter illustrated that the GOGCWS focused on the first element of cost management (efficiency) while the two elements of optimisation and strategy were given less attention. This argument will be further developed in chapter eleven.

## Chapter 11 Discussion

## **11.0 Introduction**

Chapter ten showed how the GOGCWS focused on the element of cost management (efficiency). The two elements of optimisation and strategy had not previously been addressed by the GOGCWS. It was also highlighted in chapter ten that there was no change of cost management within the GOGCWS. This chapter aims to explain why there was no change in cost management by adopting two propositions. The first suggested that government policy (coercive isomorphism) shaped the limited use of cost management for efficiency, optimisation and strategy. The second proposition assumed that organisations are the 'theatre' in which institutions are visible.

The objective of this research was to understand, describe and explain the role of cost management within the water industry. Institutional theory adds the social and political elements that are typically absent or de-emphasised in the rational instrumental approach, (which guides most of the recent research on cost management, see section 2.1). The aim of this chapter is to discuss the GOGCWS research findings in the light of the theoretical framework that has been adopted. It also relates the findings back to the literature review. The chapter is divided into three sections.

Section one evaluates the theories used to explain cost management. It relates the different theories to the three elements of cost management (efficiency, optimisation and strategy). It finds the existing theories of cost management are inadequate and incomplete (Scott, 1987).

Section two explains the findings of this research. It focuses on the role of accounting within the GOGCWS and the relationship between accounting and GOGCWS decision-making. It examines the accounting system of the GOGCWS<sup>294</sup> from an institutional perspective. The section offers an explanation for the lack of change in the GOGCWS (institutional imprinting) through neo-institutional theory. It examines the effect of the macro environment on the GOGCWS from an institutional perspective. It also links the

<sup>&</sup>lt;sup>294</sup> The Unified Accounting System.

GOGCWS to both institutional isomorphism (Carpenter and Feroz, 2001), and the political organisation (Brunsson, 1994).

The final section sums up the main conclusions and implications of the previous sections. It also looks at how this study relates to other studies and the similarities and the differences found therein.

## **11.1 Cost Management**

Cost management is a major theme in management accounting. It acts both as a guideline for the allocation of costs to different cost objects, and as a base for the design of accounting systems (Shillinglaw, 1989). Shank and Govindarajan (1989) defined strategic cost management (SCM) as "the managerial use of accounting information explicitly directed at one of the four<sup>295</sup> stages of the strategic management cycle (Shank, 1989: 50). Cooper and Slagmulder (1998b; 1998c) claimed that enterprise-wide strategic cost management systems require two distinct models. The first model needs to address operational improvement and efficiency, while the second needs to identify the firm's source of profitability. It was observed the two models highlighted by Cooper and Slagmulder had three elements of cost management, namely: efficiency, optimisation and strategy.

Cost management for efficiency has its roots in transaction theory and absorption costing (see section 4.1). As mentioned in section 4.1, there are three ways of utilising cost management to increase efficiency, (i.e. throughput costing, benchmarking and target costing). Some researchers claim that ABC focused on efficiency (Bjornenak and Olson 1999).

It was observed that the control of efficiency in the GOGCWS (via budgeting<sup>296</sup>, the reporting process<sup>297</sup>, standard costing<sup>298</sup> (as part of the design and planning process) led to an emphasis placed on calculation and analysis of variance. These findings support

<sup>&</sup>lt;sup>295</sup> These are formulation, communication, implementation and action.

<sup>&</sup>lt;sup>296</sup> See section 8.7.3.

<sup>&</sup>lt;sup>297</sup> Reports generated from the cost accounting system, found that the majority of respondents (94%) agreed for introducing financial reporting (for further analysis see section 8.6.5.2).

<sup>&</sup>lt;sup>298</sup> See section 9.2.1.1.
the argument of Ezzamel, et al., (1994). They found that companies were focusing on improving the technical capabilities of their management accounting system, in order to improve their costs calculation and enhance their efficiency. However, control of efficiency and standard-costing may have become obscured by the attention given to observation and analysis of variance.

Cost management for optimisation has been written from a managerial economic-theory perspective. There are certain models of cost management which can be utilised to achieve optimisation (such as the ABM, see section 4.4). Cost management optimisation research is underpinned by contingency theory (Simons, 1987; 1990; Govindarajan and Gupta, 1985; Shank, 1989, see section 5.2.2).

Optimisation was not a central concern to the GOGCWS (see tables 9.2, 9.2a, 9.2b and 9.27). It was observed that there was a similarity between the cost management method at the GOGCWS (budget and absorption costs), and Anglo-American literature (see section 5.1.2). In Anglo-American literature, costs include quantity components and a value component. Value components can be measured in two ways. The first involves the assignment of payments (in the form of actual or past market prices) to the consumed input factor. (This was how<sup>299</sup> cost was calculated in the GOGCWS). The second approach adds opportunity costs to pagatory costs<sup>300</sup>.

Is there any room for Opportunity costs in the Egyptian Water Industry? Several respondents suggested using water technology<sup>301</sup> (i.e. using technology to provide water resources). This could be achieved by re-cycling drainage water (i.e. making multiple

<sup>&</sup>lt;sup>299</sup> For the measurement of cost of resources in the GOGCWS, the majority of respondents (92%) chose actual cost as apposed to other forms of costing (see section 8.6.3).

<sup>&</sup>lt;sup>300</sup> "Pagatory" costs refer to 'cash payments'.

<sup>&</sup>lt;sup>301</sup> Egypt was one of the first countries to recycle water. River water has multiple use. It was used in generating power from the High dam, and subsequently for water navigation, tourism, agriculture, drinking and re-use of drainage water. This meant that water was being used several times to achieve utmost utilisation (International El-Aharam, 04/03/2000).

There were two ways to overcome decrease of water resources in future. Firstly, using water technology in agriculture would consume around 85% of water resources (such as irrigation pumps). This would lead to a decreased water consumption of over 50%. This would mean increasing water resources by over 40%. The second way involves the removal of salt from seawater and hard water. The technology

developed in this field could bring the cost of de-salination to  $2 \text{ E.B.}/m^3$ . This poses the question "Is the conveyance of distant Nile water for drinking, economical or not?"

use of it) and using technology for irrigation). The Ministry of Public Works and Water Resources (MPWWR) in Egypt explained how opportunity costs could play a crucial role in the water industry (see section 10.1.2).

Much of cost management strategy research has been written from an industrialeconomic perspective. Bellis, et al., (1999) claimed that strategic cost management and strategic management accounting have previously been the focus of researcher interest. Wong (1996) used a case study to illustrate how an activity-based approach to strategic cost management can give management a 'richer' and 'robust' understanding of cost dynamics. Cooper and Slagmulder (1998a) asked, "What is strategic cost management?" They found it to be the application of cost management techniques, which improves the strategic position of the firm thereby reducing costs. Simmonds (1981; 1982; 1985; 1986) suggested adding more relevant market data to management records and a regular reporting system. He emphasised the need for the relative competitive data on levels and tends on prices, costs, profit and volume changes for the firm and its competitors (see section 5.2.1).

Porter (1985) defined ten structural cost drives which may have originated the cost driver concept (see sections 5.1.3 and 5.5). This set of cost drivers was further developed in strategic management accounting literature (see sections 5.2.2 and 5.2.3) (Riley, 1987; Shank and Govindarajan, 1993a). Here cost drivers were divided into two groups: structure cost drivers and executional cost drivers. Most of strategic management accounting research was based on the assumption that behaviour was governed by 'rational economic sense'. However this did not explain the strategy of the GOGCWS (which itself was based on economic, political and social elements).

As mentioned in chapter ten, the focus of the GOGCWS was on efficiency and there was no change in cost management within the organisation. In order to explain why there was no change in cost management in the GOGCWS, two propositions were required. Firstly, government policy (coercive isomorphism) shaped the limited use of cost management for efficiency, optimisation and strategy. Second proposition assumed organisations are the 'theatre' in which institutions are visible.

## **11.2 Explanation of the findings**

Proposition One: Government policy (coercive isomorphism) shaped the limited use of cost management

## **11.2.1** Role of accounting in the GOGCWS

Dietrich (2001) outlined four different roles of accounting (see section 3.2.1). These were as (a) passive optimiser (b) passive co-ordinator (c) active co-ordinator and (d) active regulator (Dietrich, 2001). It was argued that the role of accounting within the GOGCWS worked as a passive optimiser<sup>302</sup>, because authority in the GOGCWS seemed to be created for constraint<sup>303</sup> purposes, not co-ordination purposes (see section 3.2.1).

Burchell, et al., (1980) outlined their four roles of accounting. These were (a) answer machines (b) assistance through decision-support systems (c) ammunition machines and (d) organisational rationalisation machines. While the four roles outlined above were not necessarily all applicable to the role of accounting within the GOGCWS, nevertheless point (a) answering machines<sup>304</sup> did relate to the role of accounting at the GOGCWS. For example, the accounting system of the company provides a stock control system and credit control routines.

The role of accounting within the GOGCWS could be seen as "a passive optimiser" and "answering machines". There was a similarity between the two. The contract in the GOGCWS was based on a universal "knowledge that", rather than a "learning based" "knowledge how" (passive optimiser). Therefore uncertainty over the objectives at the GOGCWS (and their cause and effect) was very low (answering machine).

## 11.2.2 The Unified Accounting System and Institutional theory

It was observed that the Unified Accounting System contained cognitive, normative and regulative structures (Scott, 1995). The system involved activities that provided stability

 $<sup>^{302}</sup>$  In this instance, information-use at the GOGCWS was described in a "knowledge that" manner (Dietrich, 2001, see section 9.2.1).

<sup>&</sup>lt;sup>303</sup> An accountant in the Cost Department of the GOGCWS stated that "there has not been any change in the organisational structure of the GOGCWS for 25 years. This does not give any opportunity for the skilled worker to introduce anything new". Similarly, another accountant in the Central Accounts Department claimed that "The organisational structure does not fit with the organisation's mission".

<sup>&</sup>lt;sup>304</sup> Simon (1960) called this 'structured decision-making'.

and meaning to social behaviour (see section 8.5). On the whole, the Unified Accounting System was essentially a normative structure. It was also observed that the different elements of institutional theory (culture, social structure and routines, see section 2.3) were apparent within the Unified Accounting System. As mentioned in section 2.3, there is no universal definition of "institutional". However, according to Scott (1995) an 'institution' has three pillars: cognitive, normative and regulative,

# The Cognitive pillar

The Unified Accounting System had an accounting index<sup>305</sup> based on three elements: (a) classification of accounts (b) code of accounting and (c) explaining the core idea of each. This shows that the Unified Accounting System contained a cognitive element (which included constitutive rules). These rules involved the creation of categories and the construction of typifications (see section 2.3).

The Unified Accounting System required records at the GOGCWS to be divided into four groups (see section 8.6). The Unified Accounting System took the number code and divided the production into: Production Centre, Productive Services Centre, Marketing Services Centre and Administrative, and Financial Services Centres. This ensured accuracy and ease in the collection of accounting information and its classification (see section A1.2.4.2).

#### The Normative pillar

It was observed that the accounting system in the GOGCWS had a core of functional claims (see section 8.5). This was clear from the first objective of the Unified Accounting System (outlined in section 8.5.1). This objective provides the information and analytical tools which are a requirement of planning, execution and control at different levels. The role of accounting in the GOGCWS is a normative structure, because the state enforces the Unified Accounting System onto economic organisations within the public sector (see section 8.5). Therefore, state agencies (i.e. the Accounting Central Institution and the Finance Ministry) exert influence over choice of accounting practice (see section 8.4).

<sup>&</sup>lt;sup>305</sup> See section A1.2.4.2.

It was argued that the Unified Accounting System was a normative system (see section 2.3). It defined the objectives (see section 8.5.1) and also designated a way to pursue them (see section A1.2.4.2). The Unified Accounting System could be seen as the source of accounting rules and procedures (routines) operating at the GOGCWS<sup>306</sup>. Although there were changes implemented in the economic forms and legal structure of the GOGCWS (see section 8.2.1), the researcher observed that these changes are what Nelson and Winter (1982) call "evolutionary<sup>307</sup> change".

#### The Regulative pillar

It was noted that the Unified Accounting System was a regulative process. This was apparent from analysing documentation of the GOGCWS (see section 8.1). Analysis of the interviews revealed that most of the interviewees felt they had to satisfy requirements of both the Accounting Central Institution, and the Finance Ministry. As mentioned in section 2.3, fear, force and experience were central factors of the regulative process. Fear was a major factor (see section 8.6.2).

It is argued that the Unified Accounting System had a large number of routines. This supports the argument of Nelson and Winter, 1982 (see section 2.3). It was argued that the only way to change the management accounting system at the GOGCWS was by implementing *formal* rules. In other words, in order to change management accounting system at the GOGCWS, the Accounting Central Institution has to implement formal rules for change.

As mentioned in (section 8.5), the GOGCWS was required to and did follow the Unified Accounting System. Institutional theory was adopted to explain accounting choice in organisations, where actors working in their own interests cannot exert influence over choice of accounting practice (Carpenter and Feroz, 2001). At the GOGCWS, this was because the Accounting Central Institution forced the GOGCWS to follow the Unified Accounting System.

<sup>&</sup>lt;sup>306</sup> See the relationship between rules and routines in section 2.3.

<sup>&</sup>lt;sup>307</sup> A similar point was made by Burns and Scapens (2000). They differentiated between 'revolutionary' and 'evolutionary' change. Revolutionary change is a fundamental distribution to existing routines and institutions. Evolutionary change is incremental, with only minor distribution to existing routines and institutions.

#### **11.2.3 Institutional imprinting**

The focus of management accounting in the GOGCWS was on the scope dimension. This supported the argument of Bjornenak and Olson (1999) (see section 3.2.2). This was evident from analysis of documentation of the budget system (see section 8.7.3). The Unified Accounting System in the GOGCWS had longevity (see section 8.6.2); what Bjornenak and Olson (1999) called 'a continuous view of the system itself'.

In institutional theory literature, this was often referred to as *organisational imprinting* (Kimberly, 1975; Mezias, 1990; Scott, 1987; Stinchcombe, 1965). Organisational imprinting referred to the process whereby organisations maintain certain practices adopted at the time that the organisation was founded. Hence, these organisational practices became accepted as an effective means to account for public monies (see section 8.5). This also applied to the GOGCWS. When state governments dictated accounting practices in their constitutions and statutes in this way, they were offering only one choice of accounting method, (thereby excluding alternative methods). 'Organisational imprinting' for state governments established a social reality of correct procedures to account for public monies. This was apparent in the regulations and statues of the UAS.

#### Proposition Two: Organisations are the 'theatre' in which institutions are visible.

Government statues dictated choice of accounting method in Egyptian water companies. This research argued that accounting bureaucrats working within the Egyptian Water Industry carry political influence/power (see sections 8.1 and 8.5). It also argued that Government statues (decision-makers) had been influenced by their institutionalisation. They exert this power in order to retain outmoded accounting practices.

It was noted that the accounting system of the GOGCWS was consistent with historical<sup>308</sup> institutional theory (see section 2.2). This was confirmed through interviewee feedback<sup>309</sup> where the following responses were common:

<sup>&</sup>lt;sup>308</sup> Historical-based analysis was useful for relating what they were trying to maximise and why they emphasised certain goals over others (Thelen and Steinmo, 1992).

<sup>&</sup>lt;sup>309</sup> See section 8.7.3.

- "The most important indicators for the operational budget are the expenses and revenues of last year"<sup>310</sup>
- "The budget of the last three years is taken as a guide in order to prepare the operational budget"<sup>311</sup>
- "There are certain steps taken in preparing the operational budget, firstly, we look at the expenses of last year"<sup>312</sup>
- "In order to prepare the operational budget and capital budgeting, we look at the balance sheet of the last three years"<sup>313</sup>
- "In order to prepare the budget we look at the expenses and the revenues of last year and add 10%".

## 11.2.4 The lack of change at the GOGCWS

Why was there no change at the GOGCWS? Government policy played an important role here. This was apparent from an analysis of the interviews and questionnaires. Analysis of questionnaire data showed that legislation had a greater effect than the product market and the factor market on cost practice (see table 9.27). The Chief of Commitment<sup>314</sup> stated "I think if there are any plans for changes in cost measurement practice, it would be necessary to get support from the state". Another interviewee<sup>315</sup> added, "If we want to change the cost method in the GOGCWS, we have to change the regulations".

Managers of the GOGCWS were aware of the limitations of the Unified Accounting System. Why did they not change it? It was found that managers had no choice of accounting system, and fear made them reluctant to change it. This altered the focus from a micro-process view of the GOGCWS to a macro-process one of the government policy. In other words, it highlighted the effect of the macro on the GOGCWS.

<sup>&</sup>lt;sup>310</sup> Chief of Bills Auditing Department (collection).

<sup>&</sup>lt;sup>311</sup> Chief of Commitment (Engagement) Department (Balance Sheet).

<sup>&</sup>lt;sup>312</sup> Chief of Fixed Assets Department.

<sup>&</sup>lt;sup>313</sup> Chief of Balance Sheet Administration.

<sup>&</sup>lt;sup>314</sup> (Engagement) Department (Balance Sheet).

<sup>&</sup>lt;sup>315</sup> Auditing (Maturities Department).

Institutional theorists claim that because an organisation (like the GOGCWS) depends on its survival for support of external constituents, it must conform to accept social norms. This reasoning applies to government organisations (Meyer and Rowan, 1977; Meyer et al., 1987; DiMaggio and Powell, 1991, Scott, 1987). These institutionalised expectations were expressed in a broad class of elements. They include rules, blueprints for action, standard operating procedures and impersonal prescription (Meyer and Rowan, 1977, Scott, 1987). Meyer and Rowan (1977) claimed that the formalised coordination and control systems of organisational structure, served as "rationalised and impersonal prescriptions which identify various social purposed. This demonstrates institutional conformity as technical, (affecting control) and specifying appropriate means to pursue technical purposes of rationality".

Scott and Meyer (1994) stated that there were two schools of thought on the relationship of environments to organisations. The first saw the environment as a structure of power, (posing problems which the organisations must rectify (Pfeffer and Salancik, 1978, Zald, 1970; 1984). Accounts were thus likely to arise in response to demands made by powerful elements in the environment. The second theory concentrated less on the power of environmental demands for particular products or exchanges, than on the power of the environment as a supplier of cultural material for the organisation. Both views should be taken into consideration.

It has been shown (see section 8.5) that the environment of the GOGCWS constrained managers in their choice of accounting system. Similarly the organisation failed to put into practice any recommendations (i.e. alternatives to the method of cost management, as outlined by consultants). Moreover, there was a tendency to not even disclose such recommendations 'coercive isomorphism'. The second view highlights government policy, (arguing that the water industry in Egypt can be explained in terms of the religious beliefs of the Egyptian people-culture).

This argument offered only a partial insight into water organisations. This argument failed to take into account the inadequate accounting practices of water organisations in Egypt. Previous research had led theorists to shift from a technology-focus (as the primary determinant of structure) to one emphasising the role played by the environment (Meyer and Scott, 1983a). Organisational environments vary in complexity

of resource and power arrangements (Pfeffer and Salancik, 1978). They also differed in the configuration of their wider structures and legitimating rules (DiMaggio and Powell, 1983; Meyer and Rowan, 1977).

Institutional theorists argued that pressures to conform to accepted practices can be particularly powerful in government organisations (Meyer and Rowan, 1977; DiMaggio and Powell, 1983; Gupta et al., 1994; Geiger and Ittner, 1996). The survival of government units depended primarily on the support of external constituents, and only secondarily on actual performance (Gupta et al., 1994). Conforming to accepted social norms and external requirements was required to maintain organisational legitimacy. This strengthens support and ensures continued funding.

Scott (1987) claimed that (institutional environments such as) government organisations and environmental agents<sup>316</sup> had the authority to impose their own organisational practices. The GOGCWS for example was required to follow the Unified Accounting System. The government can therefore dictate subordinate units to comply, or even specify conditions for their remaining eligibility for continuation of funding. As a result, subordinate organisations were likely to show little resistance to the implementation of mandated practices. Changes will tend to be superficial and only arise a result of participant action.

The fact that (there was no change in cost management at the GOGCWS) may be explained by through neo-institutional theory. "Institutional theory was based on the premise that organisations respond to pressures from their institutional environments<sup>317</sup>. They adopted structures or procedures which were socially-accepted as being the appropriate organisational choice" (Carpenter and Feroz, 2001). It was assumed that the customer was only interested in the GOGCWS product. Institutional theory proposed that many elements of formal organisational structure, policies and procedures, arose as a result of societal expectation of what constitutes acceptable organisational practice (Meyer and Rowan, 1977; DiMaggio and Powell, 1983; 1991; Scott, 1987; Bealing et al., 1996).

<sup>&</sup>lt;sup>316</sup> (E.g. the Accounting Central Institution in Egypt).

<sup>&</sup>lt;sup>317</sup> For the term institutional environment (see section 7.2.5).

It was observed that the GOGCWS was what Brunsson (1994) calls a "political organisation"<sup>318</sup>. This observation was built from both an understanding of the term "political organisation" (Brunsson, 1994), and from an analysis of the GOGCWS (outlined in chapters 8, 9 and 10). The political organisation contained a hierarchy in which citizens were the clients or 'principals' (see section 8.3). The political organisation was 'a public utility' (see sections 7.1 and 8.3). An organisation did the task assigned to it but had no intrinsic value if this role disappears (see section 8.2.2). A political organisation claimed its services were useful for everyone in its consistency and was under democratic control. Typically "control" in political organisations was concerned with rules. In political organisations, the (publically-appointed) leadership prescribes the procedures and methods used (see section 8.4).

The GOGCWS was what Brunsson (1994) calls "political-isation". 'Political-isation' occurs when an individual organisation takes on some the characteristics of a political organisation. This included attempts to anchor the organisation in a hierarchy of super-ordinate and comprehensive ideas and values, whilst also addressing the wider public (not merely its own customers). The GOGCWS was essentially a highly political organisation<sup>319</sup>

The "water environment"<sup>320</sup> in Egypt was very complex (see section 8.1). There was pressure for the water companies in Egypt to satisfy legislative requirements (see sections 8.4 and 9.6). The water environment in Egypt was a structure of power, (posing problems which organisations need to rectify, see section 7.2.4). Accounting arose in response to demands made by powerful elements in the environment.

## 11.2.5 Rationale for choosing the Budget incremental model

As mentioned in section 10.2.1.1, the GOGCWS had adopted an incremental budget model. Why did decision-makers of the GOGCWS follow an incremental model? Why

<sup>&</sup>lt;sup>318</sup> A political organisation is "an institution on which most state and municipal organisations are based" (Brunsson, 1994: 324).

<sup>&</sup>lt;sup>319</sup> A highly political organisation may be attracted by management techniques which focus on results and action (Brunsson, 1994).

<sup>&</sup>lt;sup>320</sup> See sections A1.2.1.1 and 8.4.

do decision-makers in the GOGCWS follow the budget process as a method of cost management? Institutional theory offers an explanation. The institutional theoretical explanation can be divided into two levels: (a) the macro level (b) the individual level.

At the macro level, the Accounting Central Institution in Egypt exerted a constant institutional pressure for the GOGCWS to use the Unified Accounting System. An incremental budget was one of its tools (see section 8.5). Scott (1987) argued that in institutional environment such as government organisations (like the GOGCWS), environmental 'agents' have the authority to impose organisational practices on subordinates. They can also specify conditions for eligibility of funding. Consequently, subordinate organisations were likely to show little resistance to the implementation of mandated practices (For more discussion of the macro level, see section 8.6).

Regarding the individual level, Douglas (1986) argued that institutions provide conceptual schemas (or analogies) that can be used in the decision-making. In the decision-making process, individuals were faced with the choice of expanding time and energy on difficult problems, or employing existing methods. Institutions often opt to save time and effort in resolving problems. Institutions can thus be seen as 'machines' of decision-making and thought, which are "in danger of turning individual thought over to automatic pilot" (Douglas, 1986: 63). As mentioned in section 11.2.1, the role of accounting within the GOGCWS can therefore be seen as answering machine and passive optimiser.

Attempts to save time and effort *should* be made by institutions, however our reliance upon them to provide analogies and conceptual schemas is not without its costs (Young, 1996). Institutions can systematically direct individual memory, and channel our perceptions into forms compatible with the relations they authorise. They fix processes that are essentially dynamic, they hide their influence, and they arouse our emotions an important issue. They essentially endow themselves with 'rightness'. Any problems are automatically transformed into organisational problems. The solutions they suggest come only from a limited range of experience. If the institution is one that depends on participation, (e.g. the GOGCWS), it should answer the call for "more participation". Institutions are shown to have the megalomania of a computer, "where the whole vision of the world is its own program" (Douglas, 1986: 92).

Institutional thinking may standardise and generalise our framing of particular problems, and limit our ways of conceptualising particular issues. This was apparent with the junior managers at the GOGCWS who objected to the issue of privatisation (see section 10.1.1). In this way, institutions create "shadowed places" (in which nothing can be seen nor any questions asked, Douglas, 1986: 69). While institutions save us time and effort in the decision-making process, they also limit the types of solutions, which are "appropriate". Government policy did not allow the method of cost management suggested by consultants (Black and Veatch International) to be adopted within the GOGCWS. They may stifle our creativity in developing novel approaches to issues. They may also inhibit our ability to recognise particular issues as 'problems' to be resolved.

The "shadowed places" created by institutions emerge in part because individuals draw upon institutions to classify "things" into categories. In categorising and classifying, we draw boundaries around those things that will be influenced, and those that will be excluded from consideration within a particular context (i.e. the UAS). For example, we draw upon our institutions to recognise and classify certain things of an "economic" nature. If institutional commitments suggest that economic things are to be valued more than non-economic things (or vice versa), then the assignment of things as either 'economic' or 'non-economic' imbues them with a particular moral and political content (Douglas, 1986: 63).

# **11.3 Conclusion**

The explanations of the findings of this research centred around two proposition: (a) Government policy (coercive isomorphism) shaped the limited use of cost management for efficiency, optimisation and strategy (b) Organisations are the 'theatre' in which institutions are visible. In other words, the explanation of the findings had moved from an institutional functionalist view (first proposition) to an institutional interpretative view (second proposition)

Cost management (optimisation and strategy) was not a central concern at the GOGCWS (see tables 9.2 - 9.3a and table 10.3). However, technical efficiency was a major concern (see tables 9.1, 9.1a and 9.1b). Although managers of the GOGCWS were aware the accounting system used in the organisation was not ideal, there had been

no change in its system since the 1960s. Managers had no choice of accounting system and fear made them reluctant to change it. Professionalisation of the government accounting community in Egypt (the Accounting Central Institution), creates a constant institutional pressure for the GOGCWS (and other water companies in Egypt, see section 8.5), to use the Unified Accounting System.

The role of accounting within the GOGCWS can be seen as a "passive optimiser" and "answering machines". The focus of management accounting in the organisation is on the scope dimension, and the accounting system employed has longevity (institutional imprinting). This chapter has shown that the accounting system in the GOGCWS adopted an incremental budgeting model, standard costing and absorption costs.

The Unified Accounting System can be seen as an outcome of coercive isomorphism<sup>321</sup>. "Coercive isomorphism" is the way in which organisations (such as the GOGCWS), may be subject to external pressures from general cultural expectations (DiMaggio and Powell, 1983), (e.g. the GOGCWS, or organisations upon which they are dependent, Carruthers, 1995). Similarly, Egyptian government regulations coerce the GOGCWS into adopting the Unified Accounting System.

As mentioned in section 11.2.2, the Unified Accounting System contained cognitive, normative and regulative pillars (Scott, 1995). It was clear from chapters 8, 9 and 10, that the external environment inhibited cognitive learning within the GOGCWS. The effect of the external environment at the GOGCWS, shaped its accounting system. In other words, the institutional pressures dominated the use of accounting in the GOGCWS.

The findings of the research were consistent with the argument advanced by DiMaggio and Powell (1991, 1983) who argue that 'coercive isomorphism' is the way organisations (such as the GOGCWS) may be subject to external pressures as a result of cultural expectations. This also applied to organisations upon which other organisations are dependent (Carruthers, 1995). Similarly, Egyptian government regulations coerced

<sup>&</sup>lt;sup>321</sup> DiMaggio and Powell (1991) distinguish between "competitive" and "institutional" isomorphism. Within 'institutional isomorphism', they distinguish between coercive, mimetic and normative isomorphism.

the GOGCWS into adopting the Unified Accounting System. These conclusions (drawn from analysis of the documentation), showed that there were several institutions responsible for financial control in the GOGCWS (such as the Cairo region, the Finance Ministry, the Ministry of Planning and the Accounting Central Institution, see section 8.4.2).

The findings of the research were consistent with (see section 7.2.2.1) the argument put forward by Ogden (1995) who claimed that before privatisation of the UK water industry, accounting emphasised cost-reduction in order to meet Government-determined performance aims (and thereby improve organisational efficiency). Management control in the GOGCWS relates to programmed controls (see section 5.2.2). Data was communicated through formal reporting procedures.

Results of this research concur with the argument of Scapens (1991). Management accounting texts suggested that in many situations, absorption costing cannot provide relevant costs decision-making and control (Scapens, 1991; see section 5.1.1). In practice, absorption costing seems to be the dominating technique in most Western countries (e.g. the U.S.A. Howell, et al., 1987; the U.K. Drury and Tayels, 1994; 2000; Sweden: Ask and Ax, 1992) and socialist countries (e.g. Egypt, see section 8.7.4). It was argued that the methods of management accounting used in the GOGCWS were compatible with the techniques of management accounting adopted in Germany (see table 5.1). For example, the cost method used in the GOGCWS is based on absorption costing, which represented costing practice (a major characteristic of the German tradition).

The findings of this research also support the argument of Hope and Hope (1995: 213) who claim that most organisations remain "tied" to archaic accounting systems and management structures. They stress the need for 'horizontal information systems'<sup>322</sup> and even interviewees<sup>323</sup> in this study called for "a connection and integration between administrations in order to decrease costs" (p. 213).

<sup>&</sup>lt;sup>322</sup> Horizontal systems provide managers with a new framework for measuring the 'real' performance of their business. As such, the horizontal information system can bring new methods of cost management into existence (i.e. activity based management).

<sup>&</sup>lt;sup>323</sup> The chief of Balance Sheet Administration and Chief of the Documentary Credit Department.

Cost accounting in the GOGCWS was similar to what Shark (1943) and Coward (1944) called "konstnadssted"<sup>324</sup>. Coward (1944: 49) argued that "if a manufacturing company follows the conversion of materials in different steps throughout the factory, then it is natural to divide the production process into sub-processes". This was apparent in the GOGCWS (see section 8.4.1.1). It was observed that the cost accounting system in the GOGCWS was not made clear to staff (see section 8.6.7). This relates back to the argument of Brignall, et al., (1991) (see section 5.1.3) who found that cost traceability varies systematically, (being greatest in professional services and least in mass services, such as the water industry, see section 5.1.3).

It was observed that there was a similarity between the method of cost management in the GOGCWS and methods found in Anglo-American literature (see section 5.1.2). In Anglo-American literature, costs always include quantity components and value components. Value components can be measured in two ways. The first involves the assignment of payments in the form of actual or past market prices, to the consumed input factor. (This was similar<sup>325</sup> to how cost was calculated in the GOGCWS). The second approach added opportunity costs to 'pagatory' costs. "Pagatory" costs refer to "cash" payments.

This chapter highlighted the similarities and differences between this study and other studies. However, it did not focus on the contribution to knowledge which is addressed in the following chapter.

<sup>&</sup>lt;sup>324</sup> Konstnaddsted is an alternative classification of cost pools. 'Department' is the natural choice for Konstnaddsted.

<sup>&</sup>lt;sup>325</sup> On the topic of measuring cost of resources in the GOGCWS, the majority of respondents (92%) chose 'actual cost' as opposed to other forms of costing (see section 8.6.3).

# Chapter 12 Conclusion

# **12.0 Introduction**

The aim of this chapter is to summarise the results of the research and identify its contribution to knowledge. This research has reported on the findings of one case study - conducted in an Egyptian public sector organisation (the GOGCWS). The results and contributions of the research have been presented, analysed and developed in chapters eight, nine, ten and eleven.

This chapter summarises the research and the research findings. It has four sections. Section one presents an overview of the thesis and gives a brief background to the 'knowledge problem' of cost management.

Section two identifies the contributions to knowledge this thesis makes to cost management and management accounting. Some contributions relate to institutional theory and the case study approach.

Section three discusses limitations of this research and offers suggestions for future research.

The final section concludes with comments on the research process and learning experience.

# 12.1 An overview

#### **Objectives and research questions**

This research endeavoured to understand and explain the role of cost management in the service water industry in Egypt via a case study of the GOGCWS. Cost management was examined from different viewpoints, including (a) accounting (the managerial and economic perspective) and (b) Government policy (social and political perspective).

With this objective in mind, empirical questions to be addressed in this research were:

- What was the cost structure for the provision of an Egyptian water service? (see section 8.6).
- What was the rationale of the water companies for their cost management? (see sections 8.5 and 11.2).
- What methods of cost management were used in the Egyptian water industry? (see sections 8.6 and 8.7).
- What was the role of accounting (its nature and function) in controlling the Egyptian water companies? (see sections 3.1, 8.5 and 11.2.1).

This thesis developed an explanation of an Egyptian water company. The provision of water is an important public sector service, world-wide, with significant economic and social impact (see section 7.1). The motivation to conduct this study arose from both theoretical and practical reasons (see section 1.3).

#### Cost Management

Largely rational economic theory and neo-classical theory (the normative view) has been the basis of most cost management research. A major criticism of rational economic theory is that it failed to consider the behaviour of the firm, neglected the social and political context, assumed the imperfect market does not exist and neglected the monopoly market, uncertainty and unpredictability.

There are different approaches in cost management (e.g. the economic approach and the strategic approach, see section 11.1). Much of the empirical research on cost management, (especially in activity-based costing and strategic management accounting), has been based on studies of private sector implementation. This study focused on a public utility (see section 7.1) and *not* the private sector, and hence this made the research difficult. As mentioned in section 6.2.1.3, certain projects seem to have universal sensitivity in developing countries: political power structures and government decision-making processes, social problems that reflect national image. Water is an emotive subject in all of them. Therefore, access to data was difficult given the constraints exerted by Egyptian Government policy.

#### Theoretical framework

Institutional theory was adopted as a framework for explaining the observed practice of cost management in the GOGCWS, because it adds the social and political elements that were typically absent or de-emphasised in the rational instrumental approach (which guides most of the recent research on cost management, see section 2.1). Institutional theory focuses on both the *external* (Covaleski, et al., 1993; 1996; Mezias, 1990) and *internal* environment (Burns and Scapens, 2000).

As mentioned in sections 2.1 and 3.3.2, three institutional theoretical frameworks had been used in the accounting literature, - namely old institutional economic theory (Scapens, 1994; Burns and Scapens, 1998; 2000), neo-institutional economic theory (transaction cost) (Williamson, 1975; 1985; 1991; 1996) and neo-institutional sociological theory (Miller, 1994). This research adopted neo-institutional economic theory and utilises new empirical data. This research therefore focused on the regulative pillar and adopted a micro focus (see figure 2.2). The study took a process approach (see section 2.3).

#### Methodology and method

This research adopted a hybrid approach (i.e. between inductive and deductive), but mainly an inductive approach. The rationale for choosing a hybrid approach was because the aims of this research were to describe what is going on (i.e. to describe the current cost structure for the provision of water in Egypt); and to understand and explain why it has been put into effect (i.e. the companies' rationale for their choice of cost structure). The research also took a nomalist view (ontology - the rationale for which can be found in section 6.1.2). It also takes a 'voluntaristic'<sup>326</sup> view of human behaviour than which guides contingency research (Child, 1972; Schreyogg, 1980). The voluntaristic view was noticeably absent in early neo-institutional research (DiMaggio, 1988 and Oliver, 1991). This approach followed from more recent advances in institutional theory which call for more-in-depth, interpretative analyses of the dialectic between stability and change (Czarniawska and Servon, 1996).

<sup>&</sup>lt;sup>326</sup> 'Voluntaristic' view assumes that human beings are free-willed, act independently of external stimuli and exercise freedom of choice (Burrel and Morgan, 1979).

A case study approach was adopted in this research (see section 6.2.1). It entailed collecting data for an examination of the GOGCWS<sup>327</sup> (see chapters 8, 9, 10 and appendix). A case study is both a process of induction and deduction (see section 6.1.1). The case study is mainly explanatory (theoretical framework), and exploratory in practice. A case study approach is a useful research strategy for both generating and developing the theoretical arguments put forward in this thesis (see section 6.2.1.1).

This methodological position led the researcher to use multiple sources for collecting data (including interviews, questionnaires and company documents of the GOGCWS). Multiple sources of evidence (triangulation, see section 6.2.1.3), were used in order to examine the context of the GOGCWS (Yin, 1984). Semi-structured interviews were carried out. Archival documents were reviewed and press articles, state statutes, and official documents were assessed. A questionnaire was used as the basis for the questioning in interviews, (which were held with all managerial levels). The rationale behind this integration (a method known as 'triangulation'), is that limitations of one method can be counter-balanced by the strength of another (Hoque and Hopper, 1994).

#### **12.2** Contribution to knowledge

This thesis gave an insight into the role of cost management and management accounting practice in the Egyptian water industry.

#### 12.2.1 The role of Cost management and Management accounting

This research was based upon a model of cost management which had three elements of cost management: efficiency, optimisation and strategy. These three elements are found in different theories (Incremental theory, Accounting theory, Economic theory (efficiency), Marginal and Opportunity cost (optimisation) and Strategic Management Accounting (strategy). Few studies have discussed these three objectives. Even these few studies (which focus on these three objectives), use a different theoretical framework, (for example Al-Hazami, 1995 used contingency theory to explain cost policy and practice in UK manufacturing industries).

<sup>&</sup>lt;sup>327</sup> Although, Yorkshire Water Company had been stood as an example of UK water system and also the French water but the focus of this research on the Egyptian water industry (the GOGCWS).

This thesis has contributed to our understanding of the role of cost management in the Egyptian water industry, by providing empirical evidence of cost management practice in one organisation. There were no other empirical studies cited in academic literature of Egyptian companies. This research was conducted in a Third world "socialist" country whose environment is different from those of Western developed economies, (on which previous reported research is based). Chapters eight and ten provided an overview of the environment where this research was conducted.

Much of the empirical literature on cost management (especially activity-based costing and strategic management accounting) is based on studies of private sector organisation implementation. This study focused on public utilities rather than the private sector (see section 7.1). It is hoped that it will contribute to knowledge development in this field.

The explanations of the findings of this research centred around two proposition: (a) Government policy (coercive isomorphism) shaped the limited use of cost management for efficiency, optimisation and strategy (b) Organisations are the 'theatre' in which institutions are visible. In other words, the explanation of the findings had moved from an institutional functionalist view (first proposition) to an institutional interpretative view (second proposition)

The research findings of this study support the work of Berry, et al., 1985; Covaleski et al., 1985; Ansari and Euske, 1987; Covaleski and Dirsmith, 1991; Lapsley, 1994; Geiger and Ittner, 1996. They all concluded that a public sector cost accounting system is generally implemented to satisfy government mandates, and is rarely used for internal decision-making and control purposes. The findings of the GOGCWS research are consistent with the argument put forward by Jones and Dugdale (2001) who claimed that management accounting appeared more significant in the UK than in some other countries, (e.g. Japan or Germany, Jones et al., 1993; Jones, 1995). Nevertheless in Egypt, the information constructed through the application of accounting measurements, techniques and criteria does not seem to dominate in decision-making and control. The reason for that is government policy or coercive isomorphism shaped the limited use of cost management for efficiency, optimisation and strategy.

This research argued that the role of management accounting in the GOGCWS relates to technical management accounting (which does not provide useful information for optimisation and strategy 'see table 9.27'). The prescribed roles of management accounting in the GOGCWS emphasised the reporting and control functions within a bureaucratic hierarchy. In order to change the role of management accounting within the GOGCWS, the focus should be on organisational processes (the accounting process), not on organisational units (the accounting department). In other words, the GOGCWS needs to improve performance measurement (see section 4.1 and Beretta, et al., 1998).

#### 12.2.2 Institutional theory and case study

Institutional theory was particularly useful for understanding public sector government (Bealing, et al. 1997). This research found that institutional theory was a theoretical lens with which to view the role of cost management in the GOGCWS. This thesis used institutional theory to explain the rationale of the cost management in the GOGCWS (see section 11.2). Previously, there had been very little empirical evidence of cost management in the Egyptian water industry (e.g. the study of "Black and Veatch International", 1997).

The study examined whether government policy (a coercive institutional) pressure had a greater influence for the choice of accounting system in the GOGCWS. Previous research has utilised institutional theory in order to explain *change* in an organisation. This research adopted institutional theory in order to explain the unchanging and relatively static environment of the GOGCWS. 'Organisational imprinting<sup>328</sup>, which has been used by researchers (Stinchcombe, 1965; Kimberly, 1975; Scott, 1987 and Mezias, 1990). 'Organisational imprinting' for state governments (i.e. the UAS' regulations and statues), establish a 'social reality' of correct procedures to account for public monies (see section 11.2.3).

This thesis found evidence that the Egyptian government directly influenced methods and systems of cost management at the GOGCWS. Consultants (Black and Veatch International) were only indirectly influential in that they could inform government

<sup>&</sup>lt;sup>328</sup> 'Organisational imprinting' refers to the process whereby organisations (such as GOGCWS) tend to maintain certain practices. For example the Unified Accounting System had been in effect since the GOGCWS was founded.

policy, and advise which accounting methods and systems should be employed in the organisation.

It was observed at the GOGCWS, that culture, government policy, structure, institutional pressures, fear and force influenced participant perception and situational and organisational objectives. (These factors are all part of the third pillar which is Regulative in institutional theory, see section 11.2.2). These observations led to the conclusion that cost management practices at the case organisation (the GOGCWS) was influenced by institutional pressures.

Institutional thinking of the Egyptian government policy "standardised" managers perceptions of the GOGCWS' framing of problems. It also limited their ways of conceptualising particular issues. In this way institutions create "shadowed places" (Douglas, 1986; see section 11.2.5). The 'shadowed places' created by institutions emerge in part because managers draw upon institutions to classify things into categories. In categorising, they draw boundaries around those things which will be influenced, and those that will be excluded from consideration. The institutional commitments of the GOGCWS managers, mean that water was valued more (as a commodity) than non-economic things.

Analysis of this research differed from contingency theory (which guides most recent research in cost management). While technical (or contingent) explanations of cost management were not rejected, they were clearly viewed as incomplete (Scott, 1987). Institutional theory adds the social and political elements, which were typically absent or de-emphasised in the rationally instrumental approach. One difference between contingency and institutional theory is that institutional theorists believe organisations respond to external pressures by adopting externally-driven accounting system practices. They do this without evidence that efficiency will increase as a result (DiMaggio and Powell, 1983; Gupta et al., 1994; Abernathy and Chua, 1995). Furthermore, institutional theory had been seen in this research as interpretative theory (not functional theory, i.e. contingency theory).

Institutional theory adds a broader (extra-organisational) institutional dimension to the analysis and discussion of this thesis. The framework also led to a clear understanding of intra-organisational processes of cost management. Through such research, it is hoped that researchers and managers will better able to anticipate the issues and difficulties involved in cost management.

Burns and Scapens (2000) adopted Old institutional economic theory and used old data (1990, 1993). They focused their attention on the cognitive pillar (see figure 2.2). This research utilises neo-institutional economic theory and uses new empirical data (unlike the studies of Geiger and Ittner, 1996; Burns and Scapens, 2000). This research therefore focuses on the regulative pillar and adopts a micro focus (see figure 2.2). Previous management accounting research (utilising institutional theory) neglected empirical data somewhat (e.g. Scapens, 1994; Brunsson, 1994; Carruthers, 1995; Walker, 1998; Spekle, 2001).

The case study method was used for studying complex social processes within a single organisation (Scapens, 1990). The case itself was both exploratory and explanatory (Spicer, 1992). This thesis has demonstrated how a case study approach has helped improve knowledge of cost management and management accounting.

# 12.3 Limitations and future research

### 12.3.1 Limitations

This study had several limitations, (some related to theory, - others to empirical work. Much of this research was written from an economic perspective. Other research exists which examines the different models of cost management (e.g. the ABC, ABM and SMA), but limited research exists regarding the development or constraints of cost management. Values and beliefs are two factors which illustrate the emotional aspect of accounting. This is a neglected feature of the social analysis of accounting. Jones and Dugdale (2001) suggest that emotions are important in understanding accounting.

There were several limitations to the empirical work. For example the research was based on the GOGCWS, - a single complex case study. Had a different case study been selected (e.g. the General Organisation for the Greater Alexandria Water Supply), this may have resulted in a different outcome. However, the legal framework and accounting system of all Egyptian water companies are similar to each other. There are also limitations to using case studies as a research method (see section 6.2.1.3)

There was limited access to the organisation itself, which meant that the investigation focused solely on staff working in the GOGCWS. This limited the research to the perception of only the GOGCWS managers. The policy decision-makers and staff who actually controlled the GOGCWS (such as the Central Accounting Institution) were therefore excluded.

Other problems arose from limited access to data. The problem arose from sensitivity of certain data (as mentioned in section 6.2.1.3). These include the political power structure, government decision-making processes, religious and social problems (which reflect the national image) and the water industry itself. This is not a unique problem when gathering research data. Hvidt (1998) in his research (an analysis of water resource planning in Egypt), found a distinct lack of data available. More data could have been obtained for this research had more information from the GOGCWS had been available. A different case study could have been examined had more time been available for fieldwork, (e.g. the General Organisation for the Greater Alexandria Water Supply<sup>329</sup> 'GOGAWS' could have been cited as an example).

The case material presented in chapters seven, eight, nine and ten represented the researcher's own observations and understanding. Much effort was made to ensure the reliability and validity of the study (see sections 6.2.1.3 and 9.3). There was no attempt to portray the accounts as absolute reflections of events. There is always the possibility that a different researcher, using different methodology and method could arrive at different conclusions.

#### 12.3.2 Future research

The main objective of this research was to develop a theoretical explanation of observations of an organisational setting. The primary focus of this research was mainly intra-organisational, and to a lesser degree, extra-organisational. This requires further research.

Government policy exerted much power and influence on cost management systems in the GOGCWS, and this was clear from the analysis (see chapters seven, eight, nine and

<sup>&</sup>lt;sup>329</sup> The researcher had interviewed very few of the staff in the GOGAWS by informal way. He failed to get the access to the GOGAWS.

ten). For example, the recommendations of consultants (Black and Veatch International) which aim to set the market-base rationality for the GOGCWS were rejected by the organisation. This can be explored further in order to understand the reasoning of the policy decision-makers who rejected market-based rationality.

It was argued that there was a lack of change in management and cost management at the GOGCWS (see section 11.2.4). It was found that the environment of the GOGCWS constrained managers in their choice of accounting system. The organisation failed to put into practice recommendations as outlined by the consultants. Moreover, there was even a tendency not to disclose such recommendations (coercive isomorphism). It was noted that government policy of the water industry can be explained in terms of the religious beliefs of the Egyptian people (culture). This latter area needs to be developed through future research.

The study raised essential points about problems in the Egyptian water industry (see chapter 7). Further research can focus on alternative ways of paying for water (such as a rateable value for domestic use and metered value for business use). Future research should address the issue of how price systems can be based on cost in order to focus on cost management (efficiency<sup>330</sup>, optimisation and strategy in the GOGCWS). The research showed clearly how the water system operates in Egypt and the GOGCWS (see section 7.2.4 and chapters 8, 9 and 10). However, the research did not devote enough time to the issue of *which* changes could have been implemented in the Egyptian water system, (see section 7.2). One external factor which could motivate the GOGCWS managers in the acquisition and use of cost management systems, is competition and uncertainty of funding. This area would need further investigation.

# 12.4 Reflections on the learning process

The research process was a learning exercise for the researcher. Valuable practical research experience was gained in the process. There were problems however with the environment in which the research was conducted. Unlike Western environments (where most research methods literature is based), case-based organisational research by doctoral students and other researchers is noticeably absent in Egypt. 'Secrecy' and lack

<sup>&</sup>lt;sup>330</sup> This refers to economic efficiency, *not* technical efficiency.

of confidentiality was also a problem. The biggest problem however was with participant understanding of what the term 'research' actually meant. Hence they were unsure of the type of information that the researcher required. Many participants were initially reluctant to express their perceptions and experiences, and limited themselves to such information as number of employees, etc. This was perhaps attributable to the researcher's inexperience. Could this been anticipated, interviews and discussions could have begun with clear explanations of the type of information required.

There were also problems with participant understanding of some the expressions and terms used. This was particularly a problem with lower level staff. Measures had been taken to ensure that the data collection instrument<sup>331</sup> did not contain commercial, economic or accounting jargon. However in follow-up questions, comments and discussions the researcher did tend to use jargon. The researcher had to explain the meaning of many terms (mainly through examples). This naturally took a considerable amount of time. Some participants also responded in unfamiliar slang, some of which was industry-specific. This problem could have been avoided had the researcher familiarised himself with such terms, before embarking on the fieldwork.

Participation and access were secured through the Egyptian government and the GOGCWS itself. Initially the researcher intended to study two organisations (the GOGCWS and the General Organisation for the Greater Alexandria Water Supply), but this did not prove possible. Many of the organisations, approached could not afford the time or were afraid to talk against government policy. During data and interpretative analysis it was difficult to concentrate on one case at a time. There was interplay between the GOGCWS and Yorkshire Water Company. This interplay, though sometimes cumbersome, was helpful. This illustrates the value of looking at more than one case study, (case studies, which are not necessarily case-comparative).

The research makes a significant contribution to cost management and management accounting literature in Egypt. Previously, there has been little research of this kind. With more studies like this, organisations in Egypt may now become more 'research-friendly'.

<sup>&</sup>lt;sup>331</sup> (As reported in Appendix two).

## • References

- Abernathy, M. and Chua, W. (1995)- "A Field Study of Control System Change: The Impact of Institutional Inducement on Managerial Choice", Paper Presented at The Third European Management Control Symposium, Imperial College, London, July.
- Abernethy, M. and Chua, W. (1996)- "A Field Study of Control Systems Redesign: The Impact of Institutional Processes on Strategic Choice", Contemporary Accounting Research, 13, pp. 569-606.
- Accounting Central Institution (1994)- "The Control of the General Organisation for Greater Cairo Water Supply", The General Control.
- Accounting Central Institution (no date source given and forbidden to publish)-"The Accounting System of the GOGCWS and Analysis of its Resources, Liabilities and Financial Statements".
- Adamany, H. and Gonsalves, F. (1994)- "Life Cycle Management: An Integrated Approach to Managing Investments", Journal of Cost Management, Summer, pp. 35-48.
- Adcroft, A. Cutler, T. Haslam, C. Williams, J. and Williams, K. (1991)- "Hanson and ICI: The Consequences of Financial Engineering", Occasional Paper On Business, Economy and Society, No. 2, University of East London, London.
- Adler, P. (1985)- "Wheeling and Dealing: An Ethnography of an Upper Level Drug Dealing and Smuggling Community", New York: Columbia University Press.
- Ahmed, M. (1992)- "A Critical Evaluation of The Methodological Underpinnings of Management Accounting Research: An Alternative Institutional Economic Framework", Unpublished PhD Thesis, University of Manchester.
- Ahmed, M. and Scapens, R. (1991)- "Cost Allocation Theory and Practice: The Continuing Debate", In Ashton, D. Hopper, T. and Scapens, R. "Issues in Management Accounting", Prentice Hall, pp. 30-60.
- Alchian, A. and Demsetz, H. (1972)- "Production, Information Costs and Economic Organisation", American Economic Review, Vol. 62, pp. 777-795.
- Alexander, J. (1983)- "Theoretical Logic in Sociology: The Classical Attempt at Theoretical Synthesis: Max Weber", Vol. 3, Berkeley: University of California Press.

- Alford, L (1993)- "Towards A New Public Management Model: Beyond "Managerialism" and its Critics", Australian Journal of Public Administration, 52, pp. 135-148.
- Al-Hazami, M. (Ph.D. Thesis- 1995)- "Cost Policies And Cost Practices in UK Manufacturing Industry", Manchester Business School.
- Allen, D. (1985)- "Strategic Management Accounting", Management Accounting (CIMA), March, pp. 25-27.
- Alles, M. Datar, S. and Lambert, A. (1995)- "Moral Hazard and Management Control in Just-in-time Settings", Journal of Accounting Research, 33 Suppl., pp. 177-204.
- American Accounting Association (1977)- "Statement on Accounting Theory and Theory Acceptance", American Association Committee on Concepts and Standards for External Financial Reports (Sarasota).
- Anderson, M. Edwards, J. and Mathews, D. (1997)- "The Priesthood of Industry: The Rise of the Professional Accountant in British Management", Oxford: Oxford University Press.
- Anderson, S. (1995)- "A framework for Assessing Cost Management System Changes: The Case of Activity Based Costing Implementation at General Motors, 1986-1993", Journal of Management Accounting Research, 7, Fall, pp. 1-51.
- Anderson, S. (1995)- Measuring the Impact of Product Mix Heterogeneity on Manufacturing Overhead Cost", The Accounting Review, July, pp. 363-387.
- Andrew, R. (1971)- "The Concept of Corporate Strategy", Dow Jones-Irwin Inc., Homewood, Illinois.
- Anonymous, (1994)- "The Organisation of Water Supply and Drainage Services in Fr", Water Engineering and Management; Des Plaines.
- Anotos, J. (1992)- "Activity-Based Management for Service, Not-for-profit, and Governmental Organisations", Journal of Cost Management, Summer, pp. 13-23.
- Ansari, S. and Euske, K. (1987)- "Rational, Rationalising and Reifying Uses of Accounting Data in Organisations", Accounting, Organisations and Society, Vol. 12, No. 6, pp. 549-570.
- Ansoff, H. (1965)- "Corporate Strategy, An Analytic Approach to Business Policy For Growth and Expansion", Penguin Books, McGraw-Hill, England.
- Anthony, N. (1965)- "Planning and Control Systems, A Framework For Analysis", Harvard University, Boston, USA.

- Anthony, R. Dearden, J. and Bedford, N. (1989)- "Management Control System", Homewood, Business One Irwin.
- Archer, M. (1995), "Realist Social Theory: The Morphogenetic Approach", Cambridge, Cambridge University Press.
- Argyris, C. (1985)- "Strategic Change, and Defensive Routines", New York, Harper and Row.
- Argyris, C. (1990)- "Overcoming Organisational Defenses: Facilitating Organisational Learning", Needham, MA, Allyn and Bacon.
- Argyris, C. and Kaplan, R. (1994)- "Implementing New Knowledge: The Case of Activity Based Costing", Accounting Horizons, 8 (3), September, pp. 83-105.
- Argyris, C. and Schon, D. (1974)- "Organisational Learning: A Theory of Action Perspective", Addison-Wesley.
- Armitage, H. and Nicholson, R. (1993)- "Activity-Based Costing: A Survey of Canadian Practice", Issues Paper No. 3, Hamilton, Ontario: The Social of Management Accountants of Canada.
- Armstrong, P. (1985)- "Changing Management Control Strategies: The Role of Competition Between Accounting and Other Organisation Professions", Accounting, Organisations and Society, 10, pp. 129-148.
- Armstrong, P. (1987)- "The Rise of Accounting Controls in British capitalist Enterprises", Accounting Organisation and Society, Vol. 12, pp. 415-436.
- Armstrong, P. (1987a)- "Engineers, Management and Trust", Work, Employment and Society, Vol. 4, pp. 1-28.
- Arrow, K. (1985)- "The Economics of Agency" In Pratt, J. and Zeckhauser, R. (eds) "Principals and Agents: The Structure of Business", (Boston, MA: Harvard Business School Press, pp. 37-51.
- Ashworth, P. (1997)- "The Varity of Qualitative Research. Part one: Introduction to The Problem", Nurse Education Today, 17, pp. 215-218.
- Ask, U. and Ax, C. (1992)- "Trends to the Development of Product Costing Practices and Techniques- A survey of Swedish Manufacturing Industry", A Paper Presented at the Annual Congress of the European Accounting Association, Madrid, Spain.
- Atkinson, A. (1990)- "Book Review of Burns, W. and Kaplan, R. (eds) "Accounting and Management: Field Study Perspectives", (Harvard Business School Press, 1987), Contemporary Accounting Research, 6, pp. 931-961.

- Ax, C. and Ask, U. (1995)- "Cost Management", Produktkakylering och ekonomistyring itveckling. Lund, Studentlitterature.
- Aylen, J. (1988)- "The Privatisation of The British Steel Corporation", Fiscal Studies, August, pp. 1-25.
- Baecker, D. (1992)- "The Writing of Accounting", Stanford Literature Review, pp. 157-178.
- Bailey, D. (1978)- "Marx on Accounting", The Accountant, January 5th.
- Banker, R. and Hughes, J. (1994)- "Product Costing and Pricing", The Accounting Review, 69 (3), pp. 479-494.
- Banker, R. and Johnson, H. (1993)- "An Empirical Study of Cost Drivers in the U.S. Airline Industry", The Accounting Review, July, pp. 576-601
- Bariff, M. and Galbraith, J. (1978)- "Intra-Organisational Power Considerations For Designing Information Systems", Accounting, Organisations and Society, 3, pp. 15-27.
- Baritz, L. (1960)- "The Servants of Power", Wesleyn University Press.
- Barry, D. and Elmes, M. (1997)- "Strategy Retold: Toward a Narrative View of Strategic Discourse", Academy of Management Review, 22, pp. 429-452.
- Baumol, W. (1982)- "Contestable Markets: An Uprising in The Theory of Industry Structures", American Economic Review, March, pp. 1-15.
- Baumol, W. (1983)- "Minimum and Maximum Pricing Principles For Residual Regulation", In Danielsen, A. and Kamerschen, D. (eds). "Current Issues in Public Utility Economics", Lexington Books.
- Baumol, W. Panzar, J. and Willig, R. (1988)- "Contestable Markets and The Theory of Industry Structure", San Diego.
- Bealing, Jr. Dirsmith, M and Fogarty, T. (1996)- "Early Regulatory Actions By The SEC: An Institutional Theory Perspective on the Dramaturgy of Political Exchanges", Accounting, Organisations and Society, Vol. 21, No. 4, pp. 317-338.
- Becker, S. and Neutheuser, D. (1975)- "The Efficient Organisation", North-Holland.
- Bellis-Jones, R. Oldman, A. and Mills, R. (1999)- "Trends in Cost Management", Management Accounting, London, Jul/Aug.
- Beretta, S. Dossi, A. and Grove, H. (1998)- "Methodological Strategies for Benchmarking Accounting Processes", Benchmarking For Quality Management and Technology, Vol. 5, Issue 3, ISSN 351-3036

- Berger, P. and Luckman, T. (1966)- "The Social Construction of Reality", New York: Doubleday.
- Berger, P. and Luckmann, T. (1967)- "The Social Construction of Reality", New York: Doubleday Anchor.
- Bergevarn, L. Mellemvik, F. and Olson, O. (1995)- "Institutionalization of Municipal Accounting- A Comparative Study Between Sweden and Norway", Vol. 11, No. 1, pp. 25: 41.
- Berliner, C. and Brimson, J. (1988)- "Cost Management Today's Advanced Manufacturing: The CAM-I Conceptual Design", Boston, MA, Haravard Business School Press.
- Berry, A. (1983)- "Management Control and Methodology", In Lowe, T. and Machin, J. (eds), "New Perspective in Management Control", London: Macmillan.
- Berry, A. Ahamed, A. Cullen, J. Dunlpo A. and Seal, W. (1997)- "The Consequences of Inter-firm Supply Chains for Management Accounting", Management Accounting, November.
- Berry, A. Broadbent, J. and Otley, D. (1995), Management Control Theories, Issues and Practices", Macmillan Press Ltd.
- Berry, A. Capps, T. Copper, D. Ferguson, P. Hopper, T. and Lowe, E. (1985)-"Management Control in An Area of the NCB: Rationales of Accounting Practices in A Public Enterprise", Accounting, Organisations and Society, Vol. 10, No. 1, pp. 3-28.
- Berry, A. Capps, T. Copper, D. Hopper, T. and Lowe, E. (1985)- "NCB Accounts-A mine of Disinformation?", Accountancy, January, pp. 10-12.
- Berry, A. Laughton, E. and Otley, D. (1991)- "Control in Financial Service Company (RIF): A Case Study", Management Accounting research, Vol. 2, No. 2, pp. 1-39.
- Bescos, P. and Mendiza, C. (1995)- "ABC in France", Management Accounting, Vol. 76, No. 10, pp. 33-41.
- Bettleheim, C. (1976)- "Economic Calculation and Forms of Property", Routledge and Kegan Paul.
- Bhimani, A. and Pigott, D. (1992)- " Implementing ABC: A case Study of Organisational and Behavioural Consequences", Management Accounting Research.
   3, pp. 119-132.
- Birnberg, J. (1980)- "The Role of Accounting in Financial Disclosure", Accounting, Organisations and Society, pp. 71-86

- Birnberg, J. Turopolec, L. and Young, S. (1983)- "The Organisational Context of Accounting", Accounting, Organisations and Society, pp. 111-129
- Bishop, M. and Kay, J. (1988)- "Does Privatisation Work? Lessons From the U.K.", Centre for Business Strategy, London Business School.
- Bjornenak, T. (1997)- "Conventional Wisdom and Costing Practices", Management Accounting Research, 8, pp. 367-382.
- Bjornenak, T. (1997)- "Diffusion and Accounting: The Case of ABC in Norway", Management Accounting Research, 8, pp. 3-17.
- Bjornenak, T. and Mitchell, F. (1999)- "A Study of the Development of the Activity-Based Costing Journal Literature 1987-1998", Working Paper, University of Edinburgh.
- Bjornenak, T. and Mitchell, F. (2000)- "A Study of the Development of the Activity Based Costing Journal Literature 1987 –1998", Paper Presented at the MARG Workshop on Reflections on Management Accounting Change, London School of Economics.
- Bjornenak, T. and Olson, O. (1999)- "Unbundling Management Accounting Innovation", Management Accounting Research, 10, pp. 325-338.
- Black and Veatch International (1997)- "Management, Training and System Strengthening for the General Organisation for Greater Cairo Water System", MTSS Phase 1: Final Project Report, Volume 1, Cairo Water II- Institutional Development Component, USAID Project No. 263-0193, 31 December.
- Blaikie, N. (1995)- "Approaches to Social Enquiry", Polity Press.
- Blumer, H. (1956)- "Sociological Analysis and The Variable", American Sociological Review, Vol. 21, No. 6, pp. 683-690.
- Blumer, H. (1969)- "Symbolic Interactionism", Englewood Cliffs, NJ: Prentice-Hall.
- Boland, R. (1993)- "Accounting and the Interpretive Act", Accounting, Organisations and Society, Vol. 18, No. 2/3, pp. 125-146.
- Boland, R. and Pondy, L. (1983)- "Accounting in Organisations: A Union of Natural and Rational Perspectives", Accounting, Organisations and Society, pp. 223-234
- Bonbright, J. (1961)- "Principles of Public Utility Rates", New York: Columbia University Press.

- Booth, P. and Giacobbe, F. (1995)- "Activity-Based Costing in Australian Manufacturing Firms: The State of Play", Contemporary Perspectives on Management Accounting, in Press.
- Bougen, P. (1989)- "The Emergence, Roles and Consequences of An Accounting -Industrial Relations Interaction", Accounting Organisations and Society, pp. 203-234.
- Bourn, M. and Ezzamel, M. (1986)- "Organisational Culture In Hospitals in The N.H.S", Financial Accountability and Management, Autumn, pp. 203-225.
- Bower, J. (1970)- "Managing the Resource Allocation Process", Graduate School of Business Administration, Harvard University.
- Boyns, T. and Edwards, R. (1997)- "Cost and Management Accounting in Early Victorian Britain: A Chandleresque Analysis?", Management Accounting Research, 8, pp.19-46.
- Brignal, S. (1997)- "A Contingent Rationale For Cost System Design in Services", Management Accounting Research, 8, pp. 325-346.
- Brignall, S. and Modell, S. (2000)- "An Institutional Perspective on Performance Measurement and Management in the 'New Public Sector'", Management Accounting Research, 11, pp. 281-306.
- Brignall, T. Fitzgerald, L. Johnson R. and Silvestro, R. (1991)- "Product Costing in Service Organisations", Management Accounting Research, Vol. 2, pp. 227-248.
- Brimson, J. (1991)- "Activity Accounting", New York.
- Brimson, J. (1996)- "The Basics of Activity- Based Management", In Drury, C.
  "Management Accounting Handbook", CIMA.
- Broadbent, J. and Guthrie, J. (1992)- "Changes in The Public Sector: A Review of Recent "Alternative" Accounting Research", Accounting, Auditing and Accountability Journal, No. 2, pp. 3-31.
- Broadbent, J. Laughlin., R. and Read, S. (1991)- "Recent Financial and Administrative Changes in The N.H.S.: A Critical Theory Analysis", Critical Perspectives on Accounting, pp. 1-24.
- Bromwich M. and Hong, C. (1999)- "Activity-Based Costing Systems and Incremental Costs", Management Accounting Research, 10, pp. 39-60.
- Bromwich, M. (1988)- "Managerial Accounting Definition and Scope-from A Managerial View" Management Accounting (CIMA), 66 (8), pp. 26-27.

- Bromwich, M. (1990)- "The Case for Strategic Management Accounting: The Role of Accounting Information for Strategy in Competitive Markets", Accounting, Organisations and Society, 15 (1/2), pp. 27-46.
- Bromwich, M. and Bhimani, A. (1989)- "Management Accounting: Evolution Not Revolution", CIMA, London.
- Brown, R. and Sproghe, H. (1987)- "Government Managerial Accounting: What and Where is it?", Public Budgeting and Finance, (August), pp. 35-46.
- Bruno, A. and Leidecker, J. (1984)- "Identifying and Using Critical Success Factors", Long Range Planning, Vol. 17. No. 1, pp 23 - 32.
- Bruns, W. and Kaplan, R. (1987)- "Introduction: Field Studies in Management Accounting", In Bruns, W. and Kaplan, R. (eds), "Accounting and Management: Field Study Perspectives", Boston, Harvard Business School Press.
- Bruns, W. and Waterhouse, J. (1975)- "Budgetary Control and Organisation Structure", Journal of Accounting Research, Autumn, pp. 177-203.
- Brunsson, N. (1989)- "The Organisation of Hypocrisy", New York, John Wiley.
- Brunsson, N. (1994)- "Politicization and 'Company-ization'- on Institutional Affiliation and Confusion in the Organisational World", Management Accounting Research, (5), pp. 323-335.
- Bryan, J. (1998)- "Inset Appointments: Meeting Supply Through Trading Water Rights", Economic Affairs, Vol. 18, No. 2, June.
- Bryman, A. (1988)- "Quantity and Quality in Social Research", London and New York.
- Bryman, A. and Cramer, D. (1990)- "Quantitative Data Analysis for Social Scientists", London: Routledge.
- Bryman, A. and Cramer, D. (1999)- "Quantitative Data Analysis With SPSS Release
  8 for Windows- A Guide for Social Scientists", London: Routledge.
- Buchanan, D. and Badham, R. (1999)- "Power, Politics and Organisational Change: Winning the Turf Game", Sage Publishing, London.
- Buchanan, D. Boddy, D. and McCalmam, J. (1988)- "Getting in, Getting on, Getting out and Getting Back" In Bryman, A. (ed.) "Doing Research In Organisations", London: Routledge.
- Burawoy, M. (1979)- "Manufacturing Consent", Chicago: University of Chicago Press.

- Burchell, S. Clubb, C. Hopwood, A. and Hughes, J. (1980)- "The Roles of Accounting in Organisations and Society", Accounting, Organisations and Society, Vol. 5, No. 1, pp. 5-27.
- Burns, J. (1997)- "The Institutionalisation of Accounting Routines: Keano Ltd.", Proceedings of the 1995, Management Control Association Symposium, pp. 217:233.
- Burns, J. (1998)- "A Critique of New Institutional Economics: Towards A Processual, Institutionalist Perspective of Management Accounting Change", Mimeo, School of Accounting and Finance, University of Manchester.
- Burns, J. (2000)- "The Dynamic of Accounting Change: Inter-Play Between New Practices, Routines, Institutions, Power and Politics", Accounting, Auditing and Accountability Journal, Vol. 13, Issue5, ISSN 0951-3574
- Burns, J. (2000a)- "Institutional Theory in Management Accounting Research: The Past, The Present, and The Future", Paper Presented at The Second Workshop on Management Accounting Change, Oslo, Norway, May 11-14.
- Burns, J. (Ph.D. Thesis- 1996)- "The Routinization and Institutionalisation of Accounting", University of Manchester.
- Burns, J. and Scapens, R. (1998)- "Conceptualising Management Accounting Change: An Institutional Framework".
- Burns, J. and Scapens, R. (2000)- "Conceptualising Management Accounting Change: An Institutional Framework", Management Accounting Research, 11, pp. 3-25.
- Burns, L. and Wholey, D. (1993)- "Adoption and Abandonment of Matrix Management Programs: Effects of Organizational Characteristics and Interorganizational Networks", Academy of Management Journal, 36, pp. 106-138.
- Burrel, G. and Morgan, G. (1979)- "Sociological Paradigams and Organisational Analysis", Heineman.
- Business Middle East (2000)- "Egypt: Practical Guide to Privatisation", 16 March.
- Byatt, Icr. (1997)- "The Regulation of Public Utilities: The Case of The Water Companies", Liverpool Economic and Statistical Society, 3 March.
- Campbell, D. (1975)- "Degrees of Freedom and The Case Study", Comparative Political Studies, 8, pp. 178-193.

- Carpenter, V. and Feroz, E. (1992)- "GAAP As A symbol of Legitimacy: New York State's Decision to Adopt Generally Accepted Accounting Principles for External Financial Reporting", Accounting, Organisations and Society, 17 (7), pp. 613-643.
- Carpenter, V. and Feroz, E. (2001)- "Institutional Theory and Accounting Rule Choice: An Analysis of Four US State Governments' Decisions to Adopt General Accepted Accounting Principles", Accounting, Organisations and Society, 26, pp. 565-596.
- Carruthers, B. (1995)- "Accounting Ambiguity, and The New Institutionalism", Accounting, Organisations and Society, Vol. 20, 4, pp. 313-328
- Cave, M. and Mills, R. (1992)- "Cost Allocation in Regulated Industries", Charted Institute of Public Finance and Accountancy.
- Chambers, D. (1988)- "Managing Operations and The Relevance of Privatisation", In Ramanadham, V. (ed.), "Privatisation in the U.K.", London: Routledge.
- Chandler, A. (1962)- "Strategy and Structure", MIT Press.
- Chandler, A. (1977)- "The Visible Hand: The Marginal Revolution In American Business", Cambridge, Massachusetts, The Belknap Pree of Harvard University Press.
- Child, J. (1972)- "Organisational Structure, Environment and Performance: the Role of Strategic Choice", Sociology, 6, pp. 1-22
- Christensen, J. and Demski, J. (1995)- "The Classical Foundations of 'Modern' Costing", Management Accounting Research, 6, pp. 13-32.
- Chua, W. Lowe, T. and Puxty, T. (1988)- "Critical Perspective in Management Control", London: MacMillan.
- Churchman, C. (1971)- "The Design of Inquiring Systems", Basic Mooks.
- Cicourel, A. (1982)- "Interviews, Surveys, and The Problem of Ecological Validity", American Sociologist, Vol. 17, No. 1, pp. 11-20.
- CIMA (The Chartered Institute of Management Accountants) (2000)- "Management Accounting Official Terminology", CIMA.
- Clark, A. and Marton, S. (1997)- "A Benchmarking Tool For Change Management", Business Process Management Journal, Vol. 3, No. 3.
- Clark, J. (1923)- "Studies in The Economics of Overhead Costs", Chicago, University of Chicago Press.
- Clark, T. and Salaman, G. (1995)- "The Management Guru As Organisational Witchdoctor", Organisation, Vol. 3, No. 1, pp. 85-107.
- Clarke, A. and Cochrance, A. (1989)- "Inside The Machine: The Left and Finance Professionals in Local Government", Capital and Class, pp. 36-61.
- Clarke, A. and Marton, S. (1997)- "A Benchmarking Tool For Change Management", Business Process Management Journal, Vol. 3, No. 3.
- Cmnd 3437 (1967)- "National Industries: A Review of Economic and Financial Objectives", London: HMSO.
- Cmnd 9734 (1986)- "White Paper on Privatisation of The Water Authorities In England and Wales", London: HMSO.
- Coase, R. (1937)- "The Nature of the Firm", Economic 4, pp. 385-405.
- Coase, R. (1983)- "The New Institutional Economics", Journal of Institutional and Theoretical Economics, 140, pp. 229-231.
- Coates, J. and Longden, S. (1989)- "Management Accounting: The Challenge of Technical Innovation", CIMA, London.
- Cobb, I. Innes, J. and Mitchell, F. (1992)- "Activity-Based Costing: Problems in Practice", London, CIMA.
- Cobb, I. Innes, J. and Mitchell, F. (1993)- "Activity-Based Costing Problems: The British Experience", Advances in Management Accounting, Vol. 2, pp. 63-83.
- Coch, L. and French, J. (1948)- "Overcoming Resistance to Change. Human Relations", 1, pp. 512-532.
- Cohen, A. (1978)- "Ethnographic Method in The Real Community", Sociology Ruralis, Vol. 18, No. 1, pp. 1-22.
- Colbert, G. and Spicer, B. (1995)- "A multi-Case Investigation of A Theory of the Transfer Pricing Process", Accounting, Organisational and Society, 20 (6), pp. 423-456.
- Commons, J. (1924)- "The Legal Foundations of Capitalism", New York: Macmillan.
- Cook, B. (1998)- "Doing Educational Research in A developing Country: Reflections on Egypt", Compare, Vol. 28/1, pp. 93-101
- Cook, W. Kazakov, A. and Roll, Y. (1994)- "On the Measurement and Monitoring of Relative Efficiency of Highway Maintenance Patrols", Chapter 10, in Charnes, A. Cooper, W. and Lewin, L. Seiford (eds), "Data Envelopment Analysis: Theory, Methodology and Applications", New York, Kluwer Academic Publishers.
- Cooley, C. (1965)- "Social Organisation", Glencoe, IL: Free Press.

- Cooper, D. and Hopper, T. (1988)- "Debating Coal Closures", Cambridge University Press, Cambridge.
- Cooper, F. (1992) -"GOGCWS Legal Background", Report 15 October.
- Cooper, R. (1988)- "The Rise of Activity-based Costing Part Two: When Do I Need An Activity Based Costing System?" Journal of Cost Management, pp. 41-48.
- Cooper, R. (1995)- "Activity- Based Costing: Theory and Practice", In Brinker, B.
  "Handbook of Cost Management", Boston: Warrn, Gorham and Lamont, pp. B1-1
  B1-33.
- Cooper, R. (1996)- "Costing Techniques to Support Corporate Strategy: Evidence From Japan", Management Accounting Research, Vol. 7, pp. 219-246.
- Cooper, R. and Kaplan, R. (1987)- How Cost Accounting Systematically Distorts Product Costs", In Burns, W. and Kaplan, R. eds. (1987)- "Accounting and Management: Field Study Perspectives", Boston, MA, Harvard Business School Press, pp. 169-203.
- Cooper, R. and Kaplan, R. (1988)- "Measure Costs Right: Make The Right Decisions", Harvard Business Review, September/October, pp. 96-103.
- Cooper, R. and Kaplan, R. (1991)- "Profit Priorities From Activity-Based Costing", Harvard Business Review, May-June, pp. 130-135.
- Cooper, R. and Kaplan, R. (1991)- "The Design Of Cost Management Systems", Prentice-Hall.
- Cooper, R. and Kaplan, R. (1992)- "Activity-Based Systems: Measuring the Costs of Resource Usage", Accounting Horizons 6, 3, pp. 1-13.
- Cooper, R. and Kaplan, R. (1998)- "Cost and Effect", Boston, Harvard Business School Press.
- Cooper, R. and Slagmulder, R. (1998a)- "Strategic Cost Management: Introduction to Enterprise-Wide Cost Management", Management Accounting, New York, August 1998.
- Cooper, R. and Slagmulder, R. (1998b)- "Operational Improvement and Strategic Costing", Management Accounting, New York, September.
- Cooper, R. and Slagmulder, R. (1998c)- "Cost Management for Internal Marke Y", Management Accounting, New York, April.
- Cooper, R. and Zmud, R. (1990)- "Information Technology Implementation Research: A Technological Diffusion Approach", Management Science, 36 (2), pp. 123-139.

- Cooper, R. Kaplan, R. Maisel, L. Morrissey, E. and Oehm, R. (1992)-"Implementing Activity-Based Cost Management: Moving From Analysis to Action", Institute of Management Accountants, Montvale.
- Corrigan, J. (1996)- "ABC Not Easy in Australia: Survey", Australian Accountant, pp. 51-53.
- Cost Management Group of the Institute of Management Accountants (1993). Issue No. 28.
- Cost Management Group of the Institute of Management Accountants (1996), Cost Management Update, Issue No. 62.
- Covaleski, M. and Dirsmith, M. (1983)- "Budgets As A Means of Control and Loose Coupling", Accounting, Organisations and Society, 8, pp. 323-340.
- Covaleski, M. and Dirsmith, M. (1986)- "The Budgetary Process of Power and Politics", Accounting, Organisations and Society, 11, pp. 193-214.
- Covaleski, M. and Dirsmith, M. (1988)- "The Budgetary Process of Power and Politics", Accounting, Organisations and Society 11, pp. 193-214.
- Covaleski, M. and Dirsmith, M. (1991)- "The Management of Legitimacy and Politics in Public Sector Administration", Journal of Accounting and Public Policy, pp. 135-156.
- Covaleski, M. Dirsmith, M. and Jablonsky, S. (1985)- "Traditional and Emergent Theories of Budgeting: An Empirical Analysis", Journal of Accounting and Public Policy, pp. 277-300.
- Covaleski, M. Dirsmith, M. and Michelman, J. (1993)- "An Institutional Theory Perspective On the DRG Framework, Case- Mix Accounting System and Health Care Organisations", Accounting, Organisations and Society, 18, 1, pp. 65-80.
- Covaleski, M. Dirsmith, M. and Samuel, S. (1996)- "Managerial Accounting Research: The Contributions of Organisational and Sociological Theories", Journal of Management Accounting Research, 8, pp. 1-35.
- Coward, D. (1944)- "Kostnadsregning Industtribedrifter", Aschehaug.
- Craib, I. (1992)- "Anthony Giddens", London: Routledge.
- Cullen, M. (1975)- "The Statistical Movement in Early Victorian Britain", Harvester.
- Cyert, R. and March, J. (1963)- "A Behavioral Theory of the Firm", Englewood Cliffs, NJ: Prentice-Hall.

- Czarniawska-Joerges, B. and Jacobsson, B. (1989)- "Budget in A Cold Climate", Accounting, Organisations and Society, 14, pp. 29-39.
- Czarniawska-Joerges, B. and Sevon, G. (1996)- "Introduction" In Czarniawska, G. and Sevon (eds), "Translating Organisational Change", Berlin, DeGruyter.
- Daft, R. and Macintosh, N. (1978)- "A New Approach to Design and Use of Management Information", California Management Review, pp. 82-92.
- Damanpour, F. and Evan, W. (1984)- "Organisational Innovation and Performance: The Problem of "Organisational Lag"", Administrative Science Quarterly, Vol. 29, No. 3.
- D'Andrade, R. (1984)- "Cultural Meaning Systems", pp. 88-119 In Shweder, R. and LeVine, R. "Culture Theory: Essays On Mind, Self, and Emotion", Cambridge, UK: Cambridge University Press.
- Daniels, A. (1999)- "The Implementation of an ABC System in ABB", Paper Presented at the ENROC Conference on Management Accounting Change, University of Manchester.
- Darlington, J. Innes, J. Mitchell, F. and Woodward, J. (1992)- "Throughput Accounting: The Garret Automotative Experience", Management Accounting, pp. 32-38.
- Data, S. Kekre, S. Mukhopadhay, T. and Srinivasan, K. (1993)- "Simultaneous Estimation of Cost Drivers", The Accounting Review, July, pp. 602-614.
- Datar, S. and Gupta, M. (1994)- "Aggregation, Specification and Measurement Errors in Product Costing", The Accounting Review 69 (4), pp. 576-591.
- Davis, G. and Powell, W. (1992)- "Organisation-Environment Relations", In Dunnette, M. and Hough, L. (eds), "Handbook of Industrial and Organisational Psychology", Vol. 3, Palo Alto, CA: Consulting Psychologists Press, pp. 317-375.
- Dawson, P. (1994)- "Organisational Change: A Processual Approach", Paul Chapman Publishing, London.
- Dawson, P. (1996)- "Technology and Quality: Change in the Workplace", Institutional Thompson Business Press, London.
- De Vaus, D. (1996)- "Survey in Social Research", UCL Press.
- Dean, G. Joye, M. and Blayney, P. (1991)- "Strategic Management Accounting Survey: Overhead Cost Allocation and Performance Evaluation Practices of Australian Manufacturers Monograph No. 8, the Accounting and Finance

Foundation Within The University of Sydney and National Industries Extension Services".

- Dent, J. (1990)- "Strategy, Organisation and Control: Some Possibilities For Accounting Research", Accounting, Organisations and Society, 15 (1/2), pp. 3-25.
- Dent, J. (1991)- "Accounting and Organisational Cultures: A Field Study of The Emergence of A New Organisational Reality", Accounting Organisations and Society, 16, pp. 705-732.
- Dermer, J. (1990)- "The Strategic Agenda: Accounting For Issues And Support", Accounting, Organisations and Society, 15 (1/2), pp. 67-76.
- Dermer, J. and Lucas, R. (1986)- "The Illusion of Managerial Control", Accounting, Organisations and Society, 11 (6), pp. 471-482.
- Diab, A. (1999)- "Egypt and Future Challenges Drinking Water and Sewerage Sector", Egyptian Review of Development and Planning, Vol. 7, No. 2, December, pp. 178: 241.
- Diamond, P. and Rothschild, M. (1985)- "Uncertainty in Economics: Readings and Exercises", SanSiego: Academic Press.
- Dietrich, M. (1993)- "Total Quality Control, Just-in-Time Management, and The Economics of The Firm", London, Routledge.
- Dietrich, M. (1994)- "Transaction Cost Economics and Beyond: Towards A New Economics of The Firm", London, Routledge.
- Dietrich, M. (1997)- "Strategic Lock-in As A Human Issue: The Role of Professional Orientation" In Magnusson, L. Ottoson, J. (eds) "Evolutionary Economics and Path Defendence", Cheltenham, Edward Elgar
- Dietrich, M. (2001)- "Accounting for The Economic of The Firm", Management Accounting Research, 12, pp. 3-20.
- Dietrich, M. and Roberts, J. (1997)- "Beyond the Economics of Professionalism", In Broadbent, J. and Dietrich, M. and Roberts (eds) "The End of the Professions? The Restructuring of Professional Work", London, Routledge.
- Dietrich, M. and Roberts, J. (1999)- "Conceptualising Professionalism: Why Economics Needs Sociology" American Journal of Economics and Sociology, 58 (4), pp. 807-828.
- DiMaggio, P. (1988)- "Interest and Agency in Institutional Theory", In Zucker, L. (eds), "Institutional Patterns and Organisations: Culture and Environment", Cambridge, MA: Bellinger Publishing CO.

- DiMaggio, P. and Powell, W. (1983)- "The Iron Cage Revisited: Institutional Isomorphism and Collective Rationality in Organizational Fields", American Sociological Review, 48, pp. 147-160.
- DiMaggio, P. and Powell, W. (1991)- "Introduction", In Powell, W. and DiMaggio, P. (eds) "The New Institutionalsim in Organisational Analysis", Chicago, University of Chicago Press, pp. 1-38.
- Donaldson, L. (1995)- "American Anti-Management Theories of Organisation: A Critique of Paradigm Proliferation", Cambridge: Cambridge University Press.
- Douglas, M. (1986)- "How Institutions Think", Syracuse, NY: Syracuse University Press.
- Drury, C. and Dugdale, D. (1992)- "Survey of Management Accounting Practice", In Drury, C. (eds.) "Management Accounting Handbook", Oxford, Butterworth-Heinemann in Conjunction With CIMA.
- Drury, C. and Tayles, M. (1994)- "Product Costing in UK Manufacturing Organisations", The European Accounting Review, Vol. 3, No. 3, pp. 443-469.
- Drury, C. and Tayles, M. (1995)- "Issues Arising From Surveys of Management Accounting Practice", Management Accounting Research, 6, pp. 267-280.
- Drury, C. and Tayles, M. (2000)- "Cost System Design and Profitability Analysis in UK Companies", CIMA.
- DTI (Department of Trade and Industry) (1997) -"Supply Chain Networks Group: Supply Chain Management Attitude Survey", Report of The Supply Chain Working Group, April.
- Dugdale, D. and Jones, T. (1994)- "Throughput and Accounting: A History of Ideas", Paper Presented for Management Accounting Research Group Conference. University of Aston.
- Dugdale, D. Kennedy, A. Sugden, K and Scarlett, R. (1996)- "Management Accounting Applications: Practical Element", The Chartered Institute of Management Accountants.
- Dunk, A. (1989)- "Budget Emphasis, Budgetary Participation and Managerial Performance: A Note", Accounting, Organisations and Society, Vol. 14, No. 4, pp. 321-324.
- Dunk, A. (1989)- "Management Accounting Lag", ABACUS, Vol. 25, No. 2, pp. 149-155.

- Durden, C. and Mak, Y. (1999)- "Reporting of Overhead Variances: A Cost Management Perspective", Journal of Accounting Education, Volume 17, Issues 2/3, pp. 321-331
- Dutton, J. and Dukerich, J. (1991)- "Keeping An Eye On The Mirror: Image and Identity in Organizational Adoption", Academy of Management Journal, 34, pp. 517-554.
- Easton, D. (1965)- "A Framework For Political Analysis", Englewood Cliffs, NJ: Prentice Hall.
- Eckstein, H. (1963)- "A Perspective on Comparative Politics, Past and Present", pp. 3-32 In Eckstein, H. and Apter D. "Comparative Politics", New York: Free Press.
- Edwards, K. and Emmanuel, C. (1990)- "Diverging Views on The Boundaries of Management Accounting", Management Accounting Research, Vol. 1, pp. 51-63.
- Edwards, P. and Shaoul, J. (1996)- "Reporting Accounting", Working Paper, Manchester University, Manchester.
- Egyptian Geographic Association (no date source given)- "The Executive Summary of Growth Population and Constructional Development in Great Cairo Water From 1996 to 2010", Research Presented to General Organisation for Greater Cairo Water Supply.
- Eisenhardt, K. (1989)- "Building Theories From Case Study Research", Academy of Management Review, 14, pp. 532-550.
- Ellram, L. (1991)- "Supply Chain Management: The Industrial Organisation Perspective", International Journal of Physical Distribution and Logistics Management, 21 (1), pp. 13-22.
- Elnathan, D. and Kim, O. (1995)- "Partner Selection and Group Formation in Cooperative Benchmarking", Journal of Accounting and Economics 19 2/3, pp. 345-364.
- Emmanuel, C. Otley, D. and Merchant, K. (1992)- "Reading in Accounting for Management Control", Chapman and Hall.
- Emore, J. and Ness, L. (1991)- "The Slow Pace of Meaningful Change in Cost Systems", Journal of Cost management, Vol. 4, No. 4, pp. 36-45.
- Espeland, M. and Bourn, M. (1990)- "The Roles of Accounting Information Systems in An Organisation Experiencing A Financial Crisis", Accounting, Organisations and Society, Vol. 15, No. 5, pp. 77-96.

- Espeland, W. and Hirsch, P. (1990)- "Ownership Changes, Accounting Practice and The Re-Definition of The Corporation", Accounting, Organisations and Society, pp. 77-96
- Evan, W. (1996)- "Organisational Lag", Human Organisation, Vol. 25, No. 1, pp. 51-53.
- Ezzamel, M. (1992)- "Corporate Governance and Financial Control" In Ezzamel, M. Heathfield, D. (eds), "Perspectives on Financial Control", London, Chapman and Hall, pp. 3-26.
- Ezzamel, M. Lilley, S. and Willmott, H. (1994)- "Accounting for Management and Management Accounting Reflections on Recent Changes in UK Companies", Paper Presented at the European Accounting Association Congress, Venice.
- Fetterman, D. (1989)- "Ethnography: Step by Step", Applied Social Research Methods Series", Vol. 17. Newbury Park, CA: Sage.
- Filstead, W. (1979)- "Qualitative Methods: A needed Perspective in Evaluation Research", In Cook, T. and Reichardt, C. (eds), "Qualitative and Quantitative Methods in Evaluation Research", Beverly Hills, Calif.: Sage, pp. 33-48.
- Fitzgerald, L. Johnson, R. Brignall, T. Silvestro, R. and Voss, C. (1991)-"Performance Measurement in Service Business", London, CIMA.
- Fligstein, N. (1987)- "Intra-organisational Power Struggles: The Rise of Finance Personnel to Top Leadership in Large Corporations, 1919-1979", American Sociology Review, 52, pp. 44-58.
- Fligstein, N. (1991)- "The Structural Transformation of American Industry: An Institutional Account of the Causes of Diversification in the Largest Firms, 1919-1979", In Powell, W. and DiMaggio, P. (eds.), "The New Institutionalism in Organizational Analysis", pp. 311-336, Chicago: University of Chicago Press.
- Fligstein, N. (1998)- "The Politics of Quantification", Accounting Organisations and Society, 23 (3), pp. 325-331.
- Forster, N. (1995)- "The Analysis of Company Documentation", In Cassell, C. and Symon, G. "Qualitative Methods in Organisational Research", Sage Publications.
- Foss, N. (1993)- "The Theory of the Firm: Contractual and Competence Perspectives", Journal of Evolutionary Economics, 3, pp. 127-144.
- Foster, G. and Gupta, M. (1990)- "Manufacturing Overhead Cost Driver Analysis", Journal of Accounting and Economics, January, pp. 309- 337.

- Frank, W. (1990)- "Back to The Future: A Perspective View of J. Maurice Clark's Studies in The Economics of Overhead Costs", Journal of Management Accounting Research, Fall, pp. 155-166.
- Froud, J. Haslam, C. Johal, S. Shaoul, J. and Williams, K. (1996)- "Stakeholder Economy?" Capital and Cost Class, Autumn, Vol. 60, pp. 119- 34.
- Fry, G. (1988)- "The Thatcher Government, The Financial Management Initive and The "New" Civil Service", Public Administration, Spring, pp. 1-20.
- Galbraith, J. (1973)- "Designing Complex Organisations", Addison-Wesley.
- Galloway, D. and Waldron, D. (1988)- "Throughput Accounting: The Need for A New Language for Manufacturing", Management Accounting (UK), November, pp. 32-33.
- Geiger, D. and Ittner, C. (1996)- "The Influence of Funding Source and Legislative Requirements on Government Cost Accounting Practices", Accounting, Organisations and Society, Vol. 21, No. 6, pp. 549-567.
- Gerry, J. (2000)- "Micro Processes of Institutional Change in The Context of Privatisation", Academy of Management Review, Jul. 2000, Vol. 25, Issue 3, pp. 572-
- Giddens, A. (1984)- "The Constitution of Society: Outline of The Theory of Structuration", Cambridge: Polity Press.
- Giddens, A. (1992)- "The Transformation of Intimacy: Love, Sexuality and Eroticism in Modern Societies", Cambridge: Polity.
- Gietzmann, M. (1991)- "Optimisation and Satisficing Sensitivity Analysis", In Ashton, D. Hooper, T. and Scapens, R. "Issues in Management Accounting", Prentice Fall, pp. 2333-255.
- Gill, J. and Johnson, P. (1991)- "Research Methods Managers", Paul Chapman Publishing Ltd.
- Gilling, D. (1976)- "Accounting and Social Change", The International Journal of Accounting Education and Research, Spring, pp. 59-72.
- Gioia, D. Thomas, J. Clark, S. and Chittipeddi, K. (1994)- "Symbolism and Strategic change in Academia: The Dynamics of Sense Making and Influence". Organization Science, 5, pp. 363-383.
- Glaser, B. and Strauss, A. (1967)- "The Discovery of Grounded Theory: Strategies For Qualitative Research", Chicago: Aldine.
- Goldartt, E. and Fox, J. (1989)- "The Goal", Aldershot, Gower.

- Goldratt, E. (1994)- "It's Not Luck", London: Gower.
- Goode, W. and Hatt, P. (1952)- "A Case For the Case Study", University of North.
- Goold, M. and Quinn, J. (1990)- "The Paradox of Strategic Controls", Strategic Management Journal, Vol. 11, pp 43-57.
- Gordon, L. and Miller, D. (1976)- "A Contingency Framework For The Design of Accounting Information Systems", Accounting, Organisations and Society, pp. 59-70.
- Gordon, L. and Narayanan, V. (1984)- "Management Accounting System, Perceived Environmental Uncertainty and Organisation Structure: An Empirical Investigation", Accounting, Organisations and Society, Vol. 9, pp. 33-47.
- Gore, A. (1993)- "From Red Tape to Results: Creating a Government That Works Better and Costs Less", New York: Random House.
- Gorry, G. and Scott-Morton, M. (1971)- "A Framework For Management Information Systems", Sloan Management Review, Fall.
- Gosselin, M. (1997)- "The Effect of Strategy and Organisational Structure on The Adoption and Implementation of Activity-Based Costing", Accounting, Organisations and Society, 22, (2), pp. 105-122.
- Gottschalk, L. (1968)- "Understanding History: A Primer of Historical Method", New York: Knopf.
- Gould, S. (1991)- "The Median Isn't The Message, Bully For Brontosaurus: Reflections in Natural History", London: Hutchison Radius.
- Govindarajan, V. (1984)- "Appropriateness of Accounting Data in Performance Evaluation: An Empirical Examination of Environmental; Uncertainty As Interviewing Variables", Accounting, Organisations and Society, pp. 125-135
- Govindarajan, V. and Gupta, A. (1985)- "Linking Control Systems to Business Unit Strategy: Impact on Performance"; Accounting, Organisations and Society; Vol. 10, No. 1, pp 51-66.
- Govindarajan, V. and Shank, J. (1992)- "Strategic Cost Management: Tailoring Controls to strategies", Journal of Cost Management, Fall, pp. 14-24.
- Granovetter, M. (1985)- "Economic Action and Social Structure; The Problem of Embeddedness", American Journal of Sociology, 91, pp. 481-510.

- Gray, A. and Jenkins, W. (1986)- "Accountable Management in British Central Government: Some Reflections on The Financial Management Initiative", Financial Accountability and Management, Autumn, pp. 171-186.
- Gray, B. (1990)- "The Enactment of Management Control Systems: A Critique of Simons", Accounting, Organisations and Society, Vol. 15, No. 1/2, pp. 145-148.
- Green, F. and Amenkhienan, F. (1992)- "Accounting Innovations: Across-Sectional Survey of Manufacturing Firms", Journal of Cost Management, Vol. 5, No. 1, pp. 58-64.
- Greenwood, R. and Hinings, C. (1993)- "Understanding Strategic Change: The Contribution of Archetypes", Academy of Management Journal, 36, pp. 1052-1081.
- Greenwood, R., and Hinings, C. (1996)- "Understanding Radical Organizational Change: Bringing Together The Old and New Institutionalism", Academy of Management Review, 21, pp. 1022-1054.
- Gummesson, E. (1991)- "Qualitative Methods in Management Research", London: Sage.
- Gupta, P. Dirsmith, M. and Fogarty, T. (1994)- "Co-ordination and Control in A Government Agency: Contingency and Institutional Perspectives on GAO Audits", Administrative Since Quarterly, pp. 264-284.
- Halfpenny, P. (1979)- "The Analysis of Qualitative Data", Sociological Review, Vol. 27, No. 4, pp. 799-825.
- Hall, P. (1986)- "Governing the Economy: The Politics of State Intervention in Britain and France", Cambridge, UK: Polity Press.
- Hamilton, W. (1932)- "Institution", In Seligman, E.R.A. and Johnson, A. (eds), Encyclopaedia of Social Science, 73, 4, pp. 560-595.
- Hannan, M. and Freeman, J. (1989)- "The Population Ecology of Organisations", American of Sociology, Vol. 82, pp. 929-964.
- Hargert, M. and Morris, D. (1989)-"Accounting Data For Value Chain Analysis", Strategic Management Journal, 10(2), March-April, pp. 175-188.
- Harland, C. (1996)- "Supply Chain Management: Relationships, Chains and Networks", British Journal of Management, 7, Special Issue, S 63- S80.
- Harmmer, M. and Champy, J. (1993)- "Reengineering the Corporation: A Manifesto For Business Revolution", Harper Business, New York.
- Harper, W. (1988)- "Privatisation in The Water Sector", In Ramanadham, V. (ed.), "Privatisation in The U.K.", London: Routledge.

- Harrow, J. and Willcocks, L. (1990)- "Public Services Management: Activities, Initiatives and Limits to Learning", Journal of Management Studies, 27, pp. 281-305.
- Hart, O. (1995)- "Firms, Contracts and Financial Structure", Oxford, Clarendon Press.
- Hartley, J. (1995)- "Case Study In Organisational Research", pp. 208-229. In Cassell, C. and Symon, G. "Qualitative Methods In Organisational Research", Sage Publication.
- Hartman, S. (1993)- "Strategic Accounting or Accounting Strategy", Paper Presented to 16<sup>th</sup> E.A.A. Congress.
- Hartmann, S. (1999)- "Strategic Accounting Relationships: The Role of Management Accounting Seen From a Strategic Perspective",
- Hartnett, N. Lowry, J. and Luther, R. (1994)- "ABC and Australian Accounting Standards", Australian Accountant, pp. 19-21.
- Hax, H. (1965)- "Die Koordination Von Entscheidungen", Koln, Carl Heymanns.
- Hayes, D. (1977)- "The Contingency Theory of Managerial Accounting", The Accounting Review, pp. 22-39.
- Heald, D. (1989)- "The U.K.: Privatisation and Its Political Context", In Vickers, J. and Wright, V. (eds), "The Politics of Industrial Privatisation in Western Europe", London: Weidenfeld and Nicolson.
- Heald, D. and Steal, D. (1986)- "Privatising Public Enterprises: An Analysis of The Government's Case", In Kay, J. Mayer, C. and Thompson, D. "Privatisation and Regulations - The U.K. Experience", Oxford: Clarendon Press.
- Henzel, F. (1931)- "Erfassung and Verrechnung De Gemeinkosten in Der Unternehmung", Berlinwien.
- Hills, J. (1986)- "De-regulating Telecoms: Competition and Control in The U.S., Japan and Britain", London: Francis Pinter.
- Hines, R. (1988)- "Financial Accounting: In Communicating Reality We Construct Reality", Accounting, Organisations and Society, pp. 251-261.
- Hiromoto, T. (1988)- "Another Hidden Edge: Japanese Management Accounting", Harvard Business Review, July-August, pp. 22-26.
- Hirst, M. (1983)- "Reliance on Accounting Performance Measures, Task Uncertainty, and Dysfunctional Behaviour: Some Extension", Journal of Accounting Research, pp. 596-605.

- Hirst, P. (1976)- "Social Evolution and Sociological Categories", George Allen and Unwin.
- HMSO, (1986)- "Privatisation of The Water Authorities in England and Wales", White Paper, Cmnd 9734, HMSO, London.
- Hoaglin, D. Light, R. McPeek, B. Mosteller, F. and Stoto, M. (1982)- "Data For Decisions: Information Strategies For Policy-Makers", Cambridge, MA. Abt Books.
- Hodgson, G. (1988)- "Economics and Institutions: A Manifesto for A Modern Institutional Economics", Cambridge: Polity Press.
- Hodgson, G. (1993)- "Economics and Evolution", Cambridge, Polity Press.
- Hood, C. (1991)- "Public Management For All Seasons?", Public Administration, Spring, pp. 3-19.
- Hope, T. and Hope, J. (1995)- "Transforming The Bottom Line: Managing Performance With The Real Numbers", Nichplas Brealey Publishing, London.
- Hopper, T. and Armstrong, P. (1991)- "Cost Accounting, Controlling Labour and The Rise of Conglomerates", Accounting, Organisations and Society, 16, pp. 405-438.
- Hopper, T. Storey, J. and Willmott, H. (1987)- "Accounting For Accounting: Towards the Development of A Dialectic View", Accounting, Organisations and Society, pp. 437-456.
- Hopwood, A. (1974)- "Accounting and Human Behaviour", Haymarket Publishing.
- Hopwood, A. (1983)- "On Trying to Study Accounting in The Context in Which It Operates", Accounting, Organisations and Society, pp. 287-305.
- Hopwood, A. (1984)- "Accounting and The Pursuit of Efficiency", In Hopwood, A. and Tomkins, C. eds. "Issues in Public Sector Management", Deddington, Philip Allan.
- Hopwood, A. (1987)- "The Archaeology of Accounting Systems", Accounting, Organisations and Society, pp. 207-234.
- Hoque, Z. (1993)- "Management Control in Public Sector Enterprises: A Case Study of A Budgeting in The Jute Industry of Bangladesh", Unpublished PhD Thesis, University of Manchester.
- Hoque, Z. and Hopper, T. (1994)- "Rationality, Accounting and Politics: A case Study of Management Control in A Bangladeshi Jute Mill", Management Accounting Research, 5 (1), pp. 5-30.

- Horngren, C. (1995)- "Management Accounting: This Century and Beyond", Management Accounting Research, 6, pp. 281-286.
- Horngren, C. Bhaimani, A. Foster, G. and Datar, S. (1999)- "Management and Cost Accounting", Prentice Hall Europe.
- Hoskin, K. and Macve, R. (1992)- "The Genesis of Modern Cost Management: A Reappraisal of the Significance of European Initiatives", A Paper Presented at The EAA, 1992.
- Howe, K. and Rasmussen E. (1986)- "Public Utility Economics and Finance", Printice-Hall, INC., Englewood Cliffs, New Jersey.
- Howell, R. Brown, J. Soucy, S. and Seed, A. (1987)- "Management Accounting in The New Manufacturing Environment", Montvale, National Association of Accountants.
- http://www.Oxfordshire.gov.uk- "Water- Where are we going? Agenda 21".
- Hughes, E. (1936)- "The Ecological Aspect of Institutions", American Sociological Review 1, pp. 180-189.
- Hughes, E. (1939)- "Institutions", In Park, R. "An Outline of The Principles of Sociology", New York: Barnes and Noble.
- Humphrey, C. (1991)- "Accountable Management in The Public Sector", In Ashton,
  D. Hopper, T. and Scapens, R. (Eds), "Issues in Management Accounting", Prentice-Hall, London.
- Humphrey, C. and Scapens, R. (1990)- "Whatever Happened to The Liontamers? An Examination of Accounting Change in The Public Sector", Paper Presented at The Workshop on Changing Notions of Acceptability, University of Manchester, June.
- Humphrey, C. and Scapens, R. (1992)- "Whatever Happened to The Liontamers? An Examination of Accounting Change in The Public Sector", Local Government Studies, Vol. 18, No. 3, pp. 141-147.
- Hvidt, M. (1998)- "Water Resources Planning in Egypt", The Middle Eastern Environment Published by StMalo Press. April 24.
- Innes, J. and Mitchell, F. (1990)- "The Process of Change In Management Accounting", Management Accounting Research, pp. 3-19.
- Innes, J. and Mitchell, F. (1991)- "ABC: A Survey of CIMA Members", Management Accounting (UK), pp. 28-30.

- Innes, J. and Mitchell, F. (1995)- "A Survey of Activity-Based Costing in The UK's Largest Companies", Management Accounting Research, Vol. 6, pp. 137-153.
- Innes, J. Mitchell, F. and Sinclair, D. (2000)- "Activity-Based Costing in The U.K.'s Largest Companies: A Comparison of 1994 and 1999 Survey Results", Management Accounting Research, 11, pp. 349-362.
- International El-Aharam (2000)- "Recycling Water", 04/03/2000
- International El-Aharam, (2000)- "The Rights of Source and Riverbed Countries in The International Agreements", 6th February (Written by Elkatee, M.).
- Jaccoby, S. (1990)- "The New Instituitonalism: What Can It Learn From the Old?", Industrial Relations 29, pp. 316-359.
- Jarillo, J. (1988)- "On Strategic Networks", Strategic Management Journal, 9, pp. 31-41.
- Jeans, M. and Morrow, M. (1989)- "The Practicalities of Using Activity-Based Costing", Management Accounting, pp. 24-44.
- Jensen, M. and Meckling, W. (1976)- "Theory of The Firm: Managerial Behaviour, Agency Costs and Ownership Structure", Journal of Financial Economics, Vol. 3, October, pp. 305-360.
- Jepperson, R. (1991)- "Institutions, Institutional Effects and Institutionalism", In Powell, W. and DiMaggio, P. (eds.), "The New Institutionalism in Organizational Analysis", pp. 143-163, Chicago: University of Chicago Press.
- Jesson, D. Maystin, D. and Smith, P. (1987)- "Performance Assessment in The Education Sector: Educational and Economic Perspectives", Oxford Review of Education, 13 (3), pp. 249- 266.
- Johnson, G. and Scholes, K. (1993)- "Exploring Corporate Strategy", Hemel Hempstead: Prentice-Hall.
- Johnson, H. (1983)- "The Search For Gain in Markets and Firms: A Review of the Emergence of Management Accounting Systems", Accounting Organisations and Society, 8, (2/3), pp. 139-146.
- Johnson, H. (1992)- "Relevance Regained: From Top-down Control to Bottom-Up Improvement", New York, Free Press.
- Johnson, H. and Kaplan, R. (1987)- "Relevance Lost: The Rise and Fall of Management Accounting", Harvard Business School Press, Boston, MA.
- Johnson, T. (1988)- "Activity-Based Information: Blueprint For World-Class Management Accounting", Management Accounting (USA), June, pp. 23-30.

- Johnson, T. (1990)- "Beyond Product Costing: Challenges to Cost Management's Conventional Wisdom", Journal of Cost management, pp. 15-21.
- Jones, C. (1992)- "The Attitude of Owner-Managers Towards Accounting Control Systems Following Management Buy-out", Accounting, Organisations and Society, pp. 151-168.
- Jones, T. (1995)- "Accounting and the Enterprise: A social Analysis", London: Routledge.
- Jones, T. and Dugdale, D. (1998)- "Theory of Constraints: Transforming Ideas?", British Accounting Review, Vol. 30, No. 1, pp. 71-93.
- Jones, T. and Dugdale, D. (2001)- "The Concept of Accounting Regime", Critical Perspectives on Accounting, 12, pp. 35-63.
- Jones, T. Currie, W. and Dugdale, D. (1993)- "Accounting and Technology in Britain and Japan: Learning From Field Research", Management Accounting Research, Vol. 4, No.2, pp. 109-137.
- Jonsson, S. (1982)- "Budgetary Behaviour in Local Government- A Case Study Over Three Years", Accounting, Organisations and Society, 7, pp. 287-304.
- Joye, M. and Blaney, P. (1990)- "Cost management Accounting Practices in Australian Manufacturing Companies: Survey Results", Monograph No. 7, The Accounting and Finance Foundation, The University of Sydney.
- Kahn, A. and Shew, W. (1987), "Current Issues in Telecommunications Regulation: Pricing", Yale Journal On Regulation, Vol. 4, No. 2, pp. 191-256.
- Kamin, L. (1974)- "The Science and Politics of IQ", Penguin.
- Kaplan, R. (1983)- "Measuring Manufacturing Performance: A New Challenge for Managerial Accounting Research", The accounting Review, October, pp. 686-705.
- Kaplan, R. (1984)- "The Evolution of Management Accounting", The Accounting Review, 59, pp. 390-418.
- Kaplan, R. (1988)- "One Cost system Isn't Enough", Harvard Business Review, pp. 61-66.
- Kaplan, R. (1992)- "In Defence of Activity-Based Cost Management", Management Accounting (US), Nov. pp. 58-63.
- Kaplan, R. (1994)- "Management Accounting (1914- 1994): Development of New Practice", Management Accounting Research, 5, pp. 247-260.
- Kaplan, R. (1995)- "New Roles of Management Accountants", Journal of Cost Management, Vol. 9, No. 3, pp. 6-13.

- Kaplan, R. and Norton, D. (1992)- "The Balanced Scorecard Measures That Drive Performance", Harvard Business Review, January-February.
- Kaplan, R. and Norton, D. (1996)- "Using the Balance Scorecard As A Strategic Management System", Harvard Business Review, Jan-Feb.
- Kato, Y. (1993)- "Target Costing Support Systems: Lessons From Leading Japanese Companies", Management Accounting Research, 4, pp. 33-47.
- Kay, J. and Thompson, D. (1986)- "Privatisation: A Policy in Search of A Rationale", Economic Journal, Vol. 96, pp. 18-32.
- Kay, J. and Thompson, D. (1988)- "Privatisation: A policy in Search of Rationale", Economic Journal, pp. 18- 32.
- Kay, J. Mayer, C. and Thompson, D. (1986)- "Privatisation and Regulations the U.K. Experience", Oxford: Clarendon Press.
- Kazdin, A. (1982)- "Single-Case Experimental Design in Clinical Research and Practice", In Kazdin, A. and Tuma, A. (eds) "New Directions for Methodology of Social and Behavioural Science: Single Case Research Designs", San Francisco, Jossey-Bass Inc. pp. 33-48.
- Kendrick, J. (1970)- "The Historical Development of National Income Accounts", History of Political Economy, Vol. 2, No. 3.
- Kenway, P. (1999)- Privatised Water: Who's Paying the Price? New Stateman, 26 July.
- Khandwalla, P. (1972)- "The Effect of Different Types of Competition on The Use of Management Controls", Journal of Accounting Research, Autumn, pp. 275-285.
- Kimberly, J. (1975)- "Environmental Constraints and Organisational Structure: A Comparative Analysis of Rehabilitation Organisations", Administrative Science Quarterly, 20 (1), pp. 1-9.
- King, N. (1995)- "The Qualitative Research Interview", In Cassell, C. and Symon,
  G. "Qualitative Methods In Organisational Research", Sage Publication.
- Klein, B. Crawford, R. and Alchian, A. (1978)- "Vertical Integration, Appropriate Rents, and The Competitive Contracting Process", Journal of Law and Economics, October, pp. 297:326.
- Klein, P. and Shelanski, H. (1994)- "Empirical Research in Transaction Cost Economics: A Survey and Assessment", Unpublished Manuscript, Haas School of Business, University of California, Berkeley.

- Kloock, J. and Schiller, U. (1997)- "Marginal Costing: Cost Budgeting and Cost Variance Analysis", Management Accounting Research, 8, pp. 299-323.
- Knights, D. and Collinson, D. (1987)- "Disciplining The Shop Floor: A Comparison of The Disciplinary Effects of Managerial Psychology and Financial Accounting", Accounting, Organisations and Society, Vol. 12, No. 5, pp. 457-477.
- Kondra, A. and Hinings, C. (1998)- "Organizational Diversity and Change In Institutional Theory", Organization Studies, 19, pp. 743-767.
- Koutsoyiannis, A. (1982)- "Non-Price Decisions: The Firm in A modern Context", London, The Macmillan Press.
- Krasner, S. (1988)- "Sovereignty: An Institutional Perspective", Comparative Political Studies, 21, pp. 66-94.
- Kvale, S. (1983)- "The Qualitative Research Interview: A Phenomenological and A Hermeneutical Mode of Understanding", Journal of Phenomenological Psychology, 14, pp. 171-196.
- Kwon, T. and Zmud, R. (1987)- "Unifying and Fragmented Models of Information System Implementation" In Boland, R. and Hirschheim, R. (eds), "Critical Issues in Information System research", New York, John Wiley.
- Laffont, J. (1989)- "The Economics of Uncertainty and Information", Cambridge, MA: MIT Press.
- Lamming, R. (1993)- "Beyond Partnership: Strategies For Lean Supply", Prentice Hall, New York.
- Lancaster, K. (1966)- "A New Approach to Consumer Theory", Journal of Political Economy, pp. 132-157.
- Lancaster, K. (1979)- "Varity, Equity and Efficiency: Product Variety in An Industrial Society", New York: Columbia University Press.
- Lapsley, L. (1994)- "Responsibility Accounting Revived? Market and Budgetary Reforms in Health Care", Management Accounting Research, pp. 337-352.
- Laughlin, R. (1991)- "Environmental Disturbances and Organisational Transitions and Transformations: Some Alternative Models", Organisational Studies, Vol. 12, No. 2, pp. 209-232.
- Lehman, C. and Tinker, T. (1987)- "The 'Real' Cultural Significance of Accounts", Accounting, Organisations and Society, Vol. 12, No. 5, pp. 503-522.
- Lemond, S. (1992)- "Activity-Based Management: An Australian Perspective", Journal of Cost Management, Vol. 6, No. 2, pp. 42-46.

- Lere, J. and Saraph, J. (1995)- "Activity Based Costing For Purchasing Managers' Cost and Pricing Determinations", International Journal of Purchasing and Materials Management, Fall, pp. 25-31.
- Liebow, E. (1967)- "Tally's Corner", Boston, Mass.: Little, Brown.
- Littlechild, S. (1986)- "Economic Regulation of Privatised Water Authorities", Department of the Environment: HMSO.
- Littler, D. and Sweeting, R. (1989)- "Management Accounting: The Challenge of Technological Innovation- High Technology Industries", CIMA, London.
- Littleton, A. (1953)- "Structure of Accounting Theory", American Accounting Association.
- Loasby, B. (1998)- "The Concept of Capabilities", In Foss, N. Loasby, B. (eds), "Economic Organisation, Capabilities and Co-ordination: Essays in Honour of Richardson, G., London, Routledge.
- Lockwood, D. (1964)- "Social Integration and System integration" In Zollschan, G. and Hirsch (eds), "Explorations in Social Change", London:Routledge and Kegan Paul.
- Loft, A. (1986)- "Towards A Critical Understanding of Accounting: The Case of Cost Accounting in The U.K. 1914-1925", Accounting. Organisations and Society, Vol. 11, No. 2, pp. 137-169.
- Longlois, R. and Robertson, P. (1995)- "Firms, Markets and Economic Change: A Dynamic Theory of Business Institutions", London, Routledge.
- Longman, D. and Schiff, M. (1955)- "Distribution Cost Analysis", Homewood, Illinois, Irwin.
- Lord, B. (1996)- "Strategic Management Accounting: The Emperor's New Clothes?" Management Accounting Research, 7, pp. 347-366.
- Lord, R. and Smith, J. (1983)- "Theoretical, Information Processing and Situational Factors Affecting Attribution Theory Models of Organizational Behavior", Academy of Management Review, 8, pp. 50-60.
- Lovell, C. Walters, L. and Wood, L. (1994)- "Steatified Models of Education Production Using Modified DEA and Regression Analysis", Chapter 17, In Charnes, A. Cooper, W. and Lewin, L. Seiford (eds), "Data Envelopment Analysis: Theory, Methodology and Applications", New York, Kluwer Academic Publishers.

- Lowe, E. and Puxty, T. (1989)- "The Problems of A Paradogm: A Critique of The Prevailing Orthodoxy in Management Control", In Chue, W. Lowe, T. and Puxty, T. (eds), "Critical Perspectives in Management Control", London: MacMillan.
- Lowe, T. and Machin, J. (1983)- "New Perspectives in Management Control", Macmillan, Hundmills.
- Lucas, H. (1990)- "Performance and The Use of Information System", Management Science, 21 (8), pp. 908-919.
- Lukka, K. and Granlund, M. (1994)- "Cost management Practice in Finland", A Paper at The Annual Congress of The European Accounting Association, Venice, Italy.
- Machlup, F. (1967)- "Theories of The Firm: Marginalist, Behavioural, Managerial" American Economic Review, 57, March, pp. 1-33.
- Macintosh, N. (1994)- "Management Accounting and Control Systems: An Organisational and Behavioural Approach", Chuchester. Wiley.
- Macintosh, N. and Scapens, R. (1990)- "Structuration Theory in Management Accounting", Accounting, Organisations and Society, 15 (5), pp. 455-477.
- Madsen, V. (1963)- "Regnkabsvesenets Opgaver Og Problem I ny belysing", 2nd Edition. Copenhagen, Gyldendal
- Malmi, T. (1997)- "Towards Explaining Activity-Based Costing Failure: Accounting and Control In A Decentralized Organisation" Management Accounting Research, 8, pp. 459-480.
- Malmi, T. (1999)- "Activity-Based Costing Diffusion Across Organisations: An Exploratory Empirical Analysis of Finnish", Accounting Organisation and Society, 24, November, pp. 649-672.
- March, J. (1981)- "Decisions in Organizations and Theories of Choice", In Van de Ven, A. and Joyce, W. (eds.), "Perspectives On Organizational Design and Behavior", pp. 205-244, New York: Wiley.
- March, J. (1988)- "Decisions in Organisations", Oxford: Basil Blackwell.
- March, J. (1994)- "A Primer On Decision Making: How Decisions Happen", New York: Free Press.
- March, J. and Olsen, J. (1984)- "The New Institutionalism: Organisational Factors in Political Life", American Political Science Review, 78, pp. 734-749.
- March, J. and Olsen, J. (1989)- "Rediscovering Institutions: The Organisational Basis of Politics", New York: Free Press.

- March, J. and Olsen, J. (1995)- "Democratic Governance", New York: Free Press.
- March, J. and Simon, H. (1958)- "Organisations", New York: John Wiley.
- Markus, H. and Zajonc, R. (1985)- "The Cognitive Perspective in Social Psychology", pp. 137-230 In Lindzey, G. and Aronson E. "Handbook of Social Psychology", Vol. 1, 3rd ed. New York: Random House.
- Markus, M. (1983)- "Power, Politics and MIS Implementation", Communication of A.C.M., June, pp. 430-444.
- Markus, M. and Pfeffer, J. (1983)- "Power and The Design and Implementation of Accounting and Control Systems", Accounting, Organisations and Society, 8, pp. 205-318.
- McClintock, C. Brannen, D. and Maynard- Moody, S. (1979)- "Applying The Logic of Sample Surveys to Qualitative Case Studies: The Case Cluster Method", Administrative Science Quarterly, 24, pp. 612-629.
- McGowan, A. and Klammer, T. (1997)- "Satisfaction With Activity-Based Costing Cost Management Implementation", Management Accounting Research, 9, pp. 217-237
- Mellemvik, F. Monsen, N. and Olson, O. (1988)- "Functions of Accounting A Discussion", Scandinavian Journal of Management, pp. 101-119.
- Merchant; K. (1981)- "The Design of The Corporate Budgeting System: Influences on Managerial Behaviour and Performance", The Accounting Review, Vol. 56, No. 4, pp. 813-829.
- Merchant, K. (1984)- "Influence on Departmental Budgeting: An Empirical Examination of A Contingency Model", Accounting, Organisations and Society, pp. 291-307.
- Meredith, E. (1985)- "Accounting in Water Industry", London, ICMA.
- Merton, R. (1936)- "The Unanticipated Consequences of Purposive Social Action", American and Sociological Review, 1, pp. 894: 904.
- Merton, R. (1957)- "Bureaucratic Structure and Personality", pp. 195-206 In Merton, R. "Social Theory and Social Structure", 2nd ed. Glencoe, IL: Free Press.
- Meyer, J. and Rowan, B. (1977)- "Institutional Organizations: Formal Structure As Myth and Ceremony", American Journal of Sociology, 83, pp. 340-363.
- Meyer, J. and Rowan, B. (1978)- "The Structure of Educational Organisations", In Meyer, M. et al. (eds.), "Environments and Organisations", Jossey-Bass.

- Meyer, J. and Scott, W. (1983a)- "Centralization and Legitimacy Problems of Local Government" In Meyer, J. and Scott, W. (1983b)- "Organizational Environments: Ritual and Rationality", pp. 199-215, Beverly Hills, CA: Sage.
- Meyer, J. Boli, J. and Thomas, G. (1987)- "Ontology and Rationalization in The Western Cultural Account", pp. 12-37 In Thomas, G. Meyer, M. John, W. Ramirez, F. and Boli, J. "Institutional Structure: Constituting State, Society, and the Individual", Newbury Park, CA: Sage.
- Meyer, W. and Rowan, B. (1991)- "Institutionalised Organisations: Formal Structure as Myth and Ceremony", In Powell and DiMaggio: "The New Institutionalism in Organisational Analysis", The University of Chicago Press.
- Mezias, S. (1990)- "An Institutional Model of Organisational Practice: Financial Reporting at The Fortune 200", Administrative Science Quarterly, 35, pp. 431-457.
- Miles, M. (1979)- "Qualitative Data As An attractive Nuisance: The Problem of Analysis", Administrative Science Quarterly, 24, pp. 590-601.
- Miles, R. and Snow, C. (1978)- "Organisational Strategy, Structure, And Process", New York, McGraw-Hill.
- Miller, J. and Volluman, T. (1985)- "The Hidden Factory", Harvard Business Review, September/ October.
- Miller, P. (1989)- "Calculating Selves and Calculable Spaces", Paper Presented at Workshop on Controlling Social Life, European University Institute, Florence.
- Miller, P. (1994)- "Accounting As Social and Institutional Practice: An Introduction", In Hopwood, A. and Miller, P. (eds.) "Accounting As Social and Institutional Practice", Cambridge: University of Cambridge Press.
- Miller, P. and O'Leary, T. (1987)- "Accounting and The Construction of The Governable Person", Accounting, Organisations and Society, pp. 235-266.
- Miller, P. and O'Leary, T. (1990)- "Making Accountancy Practical", Accounting, Organisations and Society, pp. 479-498.
- Miller, P. and O'Leary, T. (1993)- "Accounting Expertise and the Politics of the Product: Economic Citizenship and Modes of Corporate Governance", Accounting, Organisations and Society, 19 (2/3), pp. 187-206.
- Miller, P. Hopper, T. and Laughlin, R. (1991)- "The New Accounting History: An Introduction", Accounting, Organisations and Society, pp. 395-404.
- Milne, R. (Ph.D. Thesis 1980), "Budgetary Slack", University of Illinois at Urbana-Champaign.

- Mintzberg, H. (1973)- "The Nature of Managerial Work", Harper and Row, New York.
- Mintzberg, H. (1978)- "Patterns In Strategy Formulation", Management Science, 24(9), May, pp. 934-948.
- Mintzberg, H. (1983)- "Power In And Around Organisations", Prentice Hall, Englewood Cliffs, N.J.
- Mitchell, A. and Sikka, P. (1993)- "Accounting For Change: The Institutions of Accountancy", Critical Perspectives On Accounting", Vol. 4, No. 1, March, pp. 29-52.
- Mitchell, A. Puxty, A. Sikka, P. and Willmott, H. (1992)- "Accounting For Change: Proposals For Reform Of Audit and Accounting", Discussion Paper No. 7, Fabian Society, London.
- Mitchell, J. (1990)- "Britain: Privatisation As Myth". In Richardson, J. (ed.) "Privatisation and De-regulation in Canada and Britain", Aldershot: Dartmouth.
- Moe, T. (1984)- "The New Economic of Organisation", American Journal of Political Science, 28, pp. 739-777.
- Mohr, L. (1982)- "Explaining Organisational Behaviour", San Francisco: Jossey-Bass.
- Monden, Y. and Hamada, K. (1991)- "Target Costing and Kaizen Costing In Japanese Automobile Companies", Journal of Management Accounting Research, Vol. 3, pp. 16-43.
- Moore, J. (1985)- "The Success of Privatisation", Text of Speech Made When Opening Hoare Govett Ltd's New City Dealing Rooms on 17 July, HM Treasury Press Release 107/85, Reprinted In Part In Kay, J. Mayer, C. and Thompson, D. (Eds) (1986), Privatisation and Regulation: The UK Experience", Clarendon Press, London.
- Most, K. (1963)- "Marx and Management Accounting", The Accountant, August 17.
- Mouritsen, J. (1994)- "Rationality, Institutions, and Decision Making: Reflections on March and Olsen's Rediscovering Institutions", Accounting, Organisations and Society, 19 (2), pp. 193-211.
- Nahapiet, J. (1988)- "The Rhetoric and Reality of An Accounting Change: A Study of Resource Allocation", Accounting, Organisations and Society, pp. 333-358.

- Nanni, A. Dixon, J. and Vollmann, T. (1992)- "Integrated Performance Measurement: Management Accounting to Support The New Manufacturing Realities", Journal of Management Accounting Research, 4, pp. 1-19.
- Nee, N. (1992)- "Organisational Dynamic of Market Transition: Hybrid Forms, Property Rights, and Mixed Economy in China", Administrative Science Quarterly, 37, pp. 1-27.
- Neimark, M. and Tinker, T. (1986)- "The Social Construction of Management Control Systems", Accounting, Organisations and Society, pp. 369-396.
- Nelson, R. and Winter, S. (1982)- "An Evolutionary Theory of Economic Change", Boston: Harvard University Press.
- Nisbett, R. and Ross, L. (1980)- "Human Inference: Strategies and Shortcomings of Social Judgement", Englewood Cliffs, NJ: Prentice Hall.
- Noreen, E. (1990)- "In the Long Run are All Costs Variable: Preliminary Evidence", Paper Presented to Management Control Association Held at The Department of Accounting and Finance, Manchester University.
- Noreen, E. (1991)- "Conditions Under which Activity-Based Cost System Provides Relevant Costs", Journal of Management Accounting Research, 3, pp. 159-168.
- Norkiewicz, A. (1994)- "Nine Steps to Implementing ABC", Management Accounting, Vol. 75, No. 10, pp. 28-33.
- North, D. (1990)- "Institutional Change and Economic Performance", Cambridge, UK: Cambridge University Press.
- Nutt, P. (1998)- "Leverage Resistance and The Success of Implementation Approaches", Journal of Management Studies, 35, pp. 213-240.
- OFWAT, (1990)- "Paying for Water: A Time for Decisions".
- OFWAT, (1991)- "The K Factor What It Is and How It Can Be Changed", 8 October (Revised September 1994).
- OFWAT, (1992)- "Water Charged and Company Profits". 16 July (Revised March 1996).
- OFWAT, (1992a)- "Issues Involved In Regulation of Privatised Water Utilities" Speech by Alan B., Deputy Director General, OFWAT, to The Financing Water Utilities Symposium, Institute for International Research, Malasia Journal.
- OFWAT, (1992b)- "Report on Company Performance In 1991- 92", OFWAT, Birmingham.

- OFWAT, (1993b)- "1992-93 Report On The Capital Investment and Financial Performance of the Water Companies in England and Wales", OFWAT, Birmingham.
- OFWAT, (1994)- "The Role of The Regulator", March (Revised 1997).
- Ogden S. (1995)- "Transforming Frameworks Of Accountability: The Case Of Water Privatisation", Accounting, Organisations and Society. Vol. 20, No. 2/3, pp. 193-218.
- Ogden, S. (1991)- "The Trade Union Campaign Against Water Privatisation: Industrial Impotence or Political Maturity?" Industrial Relations Journal, Spring, pp. 20-35.
- Ogden, S. (1994)- "Accounting For Organisational Performance: The Construction of The Customer in The Privatised Water Industry", Paper Presented to European Institute For Advanced Studies in Management Conference On Accounting and Accountability in The New Public Sector, University of Edinburgh.
- Ogden, S. (1995a)- Transforming Frameworks Of Accountability: The Case Of Water Privatisation, Accounting Organisations and Society, Vol. 20, No 2/3, pp. 193-218.
- Ogden, S. and Anderson, F. (1999)- "The Role Of Accounting In Organisational Change: Promoting Performance Improvements In The Privatised UK Water Industry", Critical Perspectives on Accounting, 10, pp. 91- 124.
- Oliver, C. (1991)- "Strategic Responses to Institutional Processes", Academy of Management Review, 16, pp. 145-179.
- Oliver, C. (1992)- "The Antecedents of Deinstitutionalization", Organization Studies, 13, pp. 563-588.
- Oliver, O. (1997)- "Sustainable Competitive Advantage: Combining Institutional and Resource-Based Views" Strategic Management Journal, 18 (9), pp. 697-713.
- Osborne, D. and Gaebler, T. (1992)- "Reinventing Government: How the Environmental Spirit is Transforming the Public Sector", Reading, MA. Addison-Wesley.
- Otley, D. (1980)- "The Contingency Theory of Management Accounting: Achievement and Prognosis", Accounting, Organisations and Society, Vol. 5, No. 4, pp. 413-428.

- Otley, D. (1989)- "A Strategy For The Development of Theories in Management Control", In Chua, W. Lowe, T. and Puxty, T (eds), "Critical Perspective in Management Control", London: MacMillan.
- Otley, D. and Berry, A. (1980)- "Control Organisation and Accounting", Accounting, Organisations and Society, Vol. 5, No. 2, pp. 231-246.
- Otley, D. and Berry, A. (1994)- "Case Study Research in Management Accounting and Control", Management Accounting Research, 5, pp. 45-65.
- Otley, D. and Berry, A. (1998)- "Case Study Research in Management Accounting and Control", Accounting Education, 7, pp. S105-S127.
- Ouchi, W. (1979)- "A Conceptual Framework For The Design of Organisational Control Mechanisms", Management Science, Vol. 25, No. 9, pp. 833-849.
- Ouchi, W. (1980)- "Markets, Bureaucracies, and Clans", Administrative Science Quarterly, 25, pp. 129-141.
- Parsons, T. (1940)- "The Motivation of Economic Activities", Canadian Journal of Economics and Political Science, 6, pp. 187-203.
- Parsons, T. (1960a)- "A Sociological Approach to The Theory of Organisations", pp. 16-58 In Parso, T. "Structure and Process in Modern Socities", Glencoe, IL: Free Press.
- Parsons, T. (1960b)- "Some Ingredients of A General Theory of Formal Organisation", pp. 59-96 In Parsons, T. "Structure and Process in Modern Societies", Glencoe, IL: Free Press.
- Penning, J. (1985)- "Towards Convergence in Strategic Theory and Practice", In Penning, J. and Associates, "Organisational Strategy and Change", LA: Jossey-Bass, pp. 468-494.
- Perkin, H. (1989)- "The Rise of Professional Society", London, Routedge.
- Peter, R. and Royston, G. (1997)- "Integrating Transaction Cost and Institutional Theories: Toward a Constrained-efficiency", Academy of Management Review, April, Vol. 22, 2, pp. 346
- Pettersen, I. (1995)- "Budgetary Control of Hospital Ritual, Rhetorics and Rationalised Myths?", Financial Accountability and Management, 11, pp. 207-221.
- Pettigrew, A. (1973)- "The Politics of Organisational Decision Making", Tavistock.
- Pfeffer, J. (1978) "The Micro Politics of Organisations", In Meyer, M. "Environment and Organisations", Jossey-Bass.
- Pfeffer, J. (1981)- "Power in Organisations", Ballinger, Cambridge, MA.

- Pfeffer, J. and Salancik, G. (1978)- "The External Control of Organizations", New York: Harper and Row.
- Piper, J. and Walley, P. (1990)- "Testing ABC Logic", Management Accounting (UK), September, pp. 37-42.
- Porter, M. (1980)- "Competitive Strategy: Techniques For Analysing Industries and Competitors", New York, The Free Press.
- Porter, M. (1985)- "Competitive Advantage: Creating and Sustaining Superior Performance", New York: The Free Press.
- Powell, W. (1991)- "Expanding The Scope of Institutional Analysis", In Powell, W. and DiMaggio, P. (eds.), "The New Institutionalism in Organizational Analysis", pp. 183-203, Chicago: University of Chicago Press.
- Powell, W. and DiMaggio, P. (1991)- "The New Institutionalism in Organisational Analysis", Chicago, University of Chicago Press.
- Preston, A. (1995)- "Budgeting, Creativity and Culture", In Ashton, D. et al. (eds.)-"Issues in Management Accounting", London: Prentice Hall, pp. 273-297.
- Primeaux, W. (1979)- "Some Problems With Natural Monopoly", The Antitrust Bulletin.
- Puxty, A. Willmott, H. Cooper, D. and Lowe, T. (1987)- "Models of Regulation in Advanced Capitalism: Locating Accountancy in Four Countries", Accounting, Organisations and Society, pp. 273-291.
- Rajagopal, S. and Bernard, K. (1993)- "Cost Containment Strategies: Challenges For Strategic Purchasing in The 1990s", International Journal of Purchasing and Materials Management, Winter, pp. 17-24.
- Rees, R. (1985)- "The Theory of Principal and Agent", Bulletin of Economic Research, Part 1: pp. 3-26, Part 2", pp. 77-95.
- Richardson, S. and Cullen, J. (2000)- "Autopsy of Change: Contextualising Enterpreneurial and Accounting Potential in the NHS", Financial Accountability and Management in Governments, Public Services and Charities, Volume 16, No. 4, November, pp. 353: 372.
- Rickard, J. (1988)- "Privatisation in The Transport Sector", In Ramanadham, V. (ed.) Privatisation in The U.K.", London: Routledge.
- Ricketts, D. and Gray, J. (1988)- "Managerial Accounting", Boston: Houghton Mifflin.
- Rider, G. (1999)- "Will New Labour Walk on Water", Anderson Consulting.

- Riebel, P. (1974)- "Deckungsbeitrag und Deckungsbeitragsrechnung" In Grochla, E. and Wittmann, W. (eds) "Handworterbuch Der Betriebswirtschaft", 4th Edition, Stuttgart, Poeschel, pp. 1137-1155.
- Riley, D. (1987)- "Competitive Cost Based Investment Strategies For Industrial Companies" In Booz-Allen, H. (eds), "Management Issues", New York, Booz-Allen and Hamilton Inc.
- Ring, P. and Perry, J. (1985)- "Strategic Management in Public and Private Organisations: Implications of Distinctive Contexts and Constraints", Academy of Management Review, 10, pp. 276-286.
- Roberts, J. and Scapens, R. (1985)- "Accounting Systems and Systems of Accountability - Understanding Accounting Practices in Their Organisational Contexts", Accounting, Organisations and Society 10, 4, pp. 443-456.
- Roberts, J. and Scapens, R. (1990)- "Accounting As Discipline", In Cooper, T. Hopper (eds), "Critical Accounts", London, Macillan, pp. 107-125.
- Robins, J. (1987)- "Organisational Economics: Notes on The Use of Transaction Costs Theory in The Study of Organisations", Administrative Science Quarterly, 32, pp. 68-86.
- Robinson, C. (2000), "Competition in The Water Industry", Institute of Economic Affairs.
- Rose, H. (1991)- "Case Studies", In Allan G. and Skinner, C. (eds), "Handbook for Research Students in The Social Sciences", London: Falmer.
- Rosenthal, R. (1966)- "Experimenter Effects In Behavioural Research", New York: Appleton Century-Crofts.
- Roth, H. and Borthick, A. (1989)- "Getting Closer To Real Product Costs", Management Accounting, May, pp. 28-33.
- Rouse, P. and Putterill, M. (2000)- "Incorporating Environmental Factors Into A Highway Maintenance Cost Model", Management Accounting Research, 2000, 11, pp. 363-384.
- Russell, P. (1991)- "Department on Trade and Industry Investigations", In Sherer, M. and Turley, S. (eds), "Current Issues in Auditing", Chapman Publishing Ltd, London.
- Rutherford, M. (1994)- "Institutions in Economics: The Old and New Institutionalism", Cambridge, Cambridge University Press.

- Ryan, B. Scapens, R. and Theobald, M. (1993)- "Research Method and Methodology in Finance and Accounting", TJ Press Ltd, Padstow, Cornwall.
- Sales, O. Blake, J. and Wraith, P. (1995)- "Learning ABC in Spain... With No Sour Grapes", Management Accounting (UK), Vol. 73, No. 9, pp. 36-37.
  - Sathe, V. (1975)- "Contemporary Theory of Organisation Structure", In Livingstone, J. (ed.), "Managerial Accounting: The Behavioural Foundations", Grid Publishing.
  - Scapens, R. (1988)- "Management Accounting Researching Practice: A Review of the Practice of Research", Paper Prepared For The Annual Conference of The Accounting Association of Australia and New Zealand.
- Scapens, R. (1990)- "Researching Management Accounting Practice: The Role of Case Study Methods", British Accounting Review, 22, pp. 259-281.
- Scapens, R. (1991)- "Management Accounting: A Review of Recent Developments", 2nd edition, London, Macmillan.
- Scapens, R. (1994)- "Never Mind the Gap: Towards An Institutional Perspective of Management Accounting Practices", Management Accounting Research, Vol. 5 (3/4), pp. 301- 321.
- Scapens, R. and Arnold, J. (1986)- "Economics and Management Accounting Research", In Bromwich, M. and Hopwood, A. (eds.) "Research and Current Issues in Management Accounting", Pitman, pp. 78-102.
- Scapens, R. and Macintosh, N. (1996)- "Structure and Agency in Management Accounting Research: A Response to Boland's Interpretive Act", Accounting, Organisations and Society, Vol. 21, No. 7/8, pp. 675-690.
- Scapens, R. and Roberts, J. (1993)- "Accounting and Control: A Case Study of Resistance to Accounting Change", Management Accounting Research, 4, pp. 1-32.
- Schonberger, R. (1986)- "World-Class Manufacturing", New York, Free Press.
- Schonberger, R. (1991)- "Performance Measurement: The Integration of Time and Customer", CAM-I Programme Meeting, Cincinatti, Ohio, September 11-12.
- Schreyogg, G. (1980)- "Contingency and Choice in Organisation Theory", Organisation Studies, 1, pp. 305-326.
- Scott, W. (1987)- "Organisations: Rational, Natural and Open Systems", (2nd ed.).
  Englewood Cliffs, NJ: Prentice Hall.
- Scott, W. (1987a)- "The Adolescence of Institutional Theory", Administrative Science Quarterly, pp. 493-511.

- Scott, W. (1992)- "Organisational Environments: Ritual and Rationality", pp. 129-153, Beverly Hills, CA: Sage
- Scott, W. (1994)- "Institutions and Organizations: Toward a Theoretical Synthesis", In Meyer, J. and Scott, R. (eds.), "Institutional Environments and Organizations", pp. 55-80, Thousand Oaks, CA: Sage.
- Scott, W. (1995)- "Institutions and Organisations", Sage Publications.
- Scott, W. and Meyer, J. (1983)- "The Organisation of Social Sectors", pp. 129-153 In Meyer, J. and Scott, R. "Organisational Environments: Ritual and Rationality", Chicago: University of Chicago Press, 1991.
- Scott, W. and Meyer, J. (1991)- "The Organisation of Societal Sectors: Propositions and Early Evidence", In Powell, W. and DiMaggio, P. (eds), "The New Institutionalism in Organisational Analysis", Chicago: University of Chicago Press.
- Scott, W. and Meyer, J. (1994)- "Institutional Environments and Organisations -Structural Complexity and Individualism", Sage Publications.
- Seal, W. Berry, A. and Cullen, J. (1999a)- "Inter-firm Accounting: A Structuration Perspective", Paper presented in Workshop on Accountability in Networks, Uppsala University, September.
- Seal, W. Cullen, J. Dunlop, A. Berry, A. and Ahamed, M. (1999)- "Enacting A European Supply Chain: A Case Study on The Role of Management Accounting", Management Accounting Research. 10, pp. 303-322.
- Searle, G. (1971)- "The Quest For National Efficiency", Blackwell.
- Searle, J. (1969)- "Speech Acts: An Essay in The Philosophy of Language", Cambridge, UK: Cambridge University Press.
- Selznick, P. (1948)- "Foundations of The Theory of Organisation", American Sociological Review, 13, pp. 25-35.
- Selznick, P. (1957)- "Leadership in Administration", New York.
- Shank, J. (1989)- "Strategic Cost Management: New Wine or Just Bottles?", Journal of Management Accounting Research, 1, pp. 47-65
- Shank, J. and Govindarajan, V. (1988)- "Marketing Strategy Explicit in Cost Analysis: A Case Study", Sloan Management Review, pp. 19-29.
- Shank, J. and Govindarajan, V. (1989)- "Strategic Cost Analysis", New York, Irwin.
- Shank, J. and Govindarajan, V. (1989)- "Strategic Cost Analysis: The Evolution From Managerial to Strategic Accounting", Homewood, IL, Irwin.

- Shank, J. and Govindarajan, V. (1992)- "Strategic Cost Management and The Value Chain", Journal of Cost Management, Winter, pp. 5-21.
- Shank, J. and Govindarajan, V. (1992a)- "Strategic Cost Analysis of Technological Investment", Sloan Management Review, Fall, pp. 39-51.
- Shank, J. and Govindarajan, V. (1992b)- "Strategic Cost Management: The Value Chain Perspective", Journal of Management Accounting Research, 4, pp. 179-197.
- Shank, J. and Govindarajan, V. (1992c)- "Strategic Cost Management and The Value Chain", Journal of Cost Management, Winter, pp. 5-21.
- Shank, J. and Govindarajan, V. (1993)- "What Drives Cost", Advances in Management Accounting, 2, pp. 27-47
- Shank, J. K. and Covindarajan, V. (1993a)- "Strategic Cost Management The New Tool For Competitive Advantage", The Free Press.
- Shaoul, J. (1997)- "The Power of Accounting Reflecting On Water Privatisation", Accounting, Auditing and Accountability Journal, Vol. 10 No. 3, pp.382-405
- Shark, L. (1943)- "Selvkostberegninger of Bokforing", Forretningsokonomisk Forlag.
- Shields, M. (1995)- "An Empirical Analysis of Firms Implementation Experiences With Activity- Based Costing", Journal of Management Accounting Research, 7, pp. 148-166.
- Shields, M. and Young, M. (1989)- "A Behavioural Model For Implementing Cost Management Systems", Journal of Cost Management, Winter, pp. 17-27.
- Shillinglaw, G. (1989)- "Management Cost Accounting: Present and Future", Journal of Management Accounting Research, 1, pp. 33-46.
- Shim, E. and Sudit, E. (1995)- "How Manufacturers Price Products", Management Accounting, Vol. 76, No. 8, pp. 37-39.
- Silvestro R. Fitzgerald, L. and Johnson R. (1992)- Towards A Classification of Service Processes", International Journal of Services Industry Management, 3, pp. 62-75.
- Simmonds, K. (1981)- "Strategic Management Accounting", Management Accounting (CIMA), April, pp. 26-29.
- Simmonds, K. (1982)- "Strategic Management Accounting For Pricing: A Case Example", Accounting and Business Research, Summer, pp. 206-214.
- Simmonds, K. (1985)- "How to Compete", Management Today, pp. 39-43.

- Simmonds, K. (1986)- "The Accounting Assessment of Competitive Position", European Journal of Marketing, Vol. 20, No. 1, pp. 16-31.
- Simon, H. (1957)- "Administrative Behaviour", 2nd ed. New York: Macmillan. Simon, H. (1960)- "The New Science of Management Decision", Harper and Row.
- Simons, R. (1987)- "Accounting Control Systems and Business Strategy: An Empirical Analysis"; Accounting, Organisations and Society; Vol. 12. No. 4., pp 357-374.
- Simons, R. (1990)- "The Role of Management Control System in Creating Competitive Advantage: New Perspective", Accounting, Organisations and Society, 15 (1/2), pp. 127-143.
- Simons, R. (1991)- "Strategic Orientation and Top Management Attention to Control Systems", Strategic Management Journal, 12, pp. 49-62.
- Simons, R. (1992)- The Strategy of Control", CA Magazine, March, pp. 44-50.
- Simons, R. (1995)- "Levers of Control", Harvard Business School Press.
- Sink, D. (1985)- "Productivity Management: Planning, Measurement Evaluation, Control and Improvement", John Wiley.
- Skocpol, T. (1985)- "Bringing the State Back In: Strategies of Analysis in Current Research", pp. 3-37 In Evans, P. Ruesschemeyer, D. and Skocpol, T. "Bringing the State Back In", UK: Cambridge University Press.
- Skocpol, T. (1992)- "Protecting Soldiers and Mothers: The Political Origins of Social Policy in The United State", Cambridge, MA: Harvard University Press.
- Slack, N. Chambers, S. Harland, C. Harrison, A. and Johnson, R. (1998)-"Operation Management", London, Pitman.
- Smith, M. (1995)- "New Tools For Management Accounting: Putting Activity Based Costing and Non-financial Indicators to Work".
- Smith, M. (1997)- "Strategic Management Accounting Text and Cases", Great Britain.
- Soin, K. Seal, W. and Cullen, J. (2001)- "ABC and Organisational Change: An Institutional Perspective", Management Accounting Research, 13, pp. 249-271.
- Solomons, D. (1952)- "The Historical Development of Costing", In Solomons (ed.), "Studies in Costing", Sweet and Maxwell.
- Sorensen, P. and Israelsen, P. (1996)- "The Diffusion of "Cost Management' Techniques in Danish Manufacturing Companies", A Paper Presented at The Annual Congress of The European Accounting Association, Bergen, Norway, Stuttgart, Poeschel, pp. 1137-1155.

- SouthWest Water (2000)- "Annual Report and Accounts 2000".
- SouthWest Water (2000)- "Summary Report and Accounts 2000".
- Spekle', R. (2001)- "Explaining Management Control Structure Variety: A Transaction Cost Economics Perspective", Accounting, Organisations and Society, 26, pp. 419-441.
- Spendolini, M. (1992)- "The Benchmarking Book", Amacom, New York
- Spicer, B. (1988)- "Towards An Organisational Theory of the Transfer Pricing Process", Accounting Organisations and Society, 13 (3), pp. 303-322.
- Spicer, B. (1992)- "The Resurgence of Cost and Management Accounting: A Review of Some Recent Developments in Practice, Theories and Case Research Methods", Management Accounting Research, 3, pp. 1-37.
- Spicer, B. and Ballew, V. (1983)- "Management Accounting Systems and the Economics of Internal Organisation", Accounting Organisations and Society, 8 (1), pp. 73-96
- Stamp, E. and Marley, C. (1970)- "Accounting Principles and The City Code: A Case For Reform", Butterworths, London.
- Stamp, E. Dean, G. and Wolnitzer, P. (1980)- "Notable Financial Causes Celebres", Arno Press, New York.
- Standifird, S. and Marshall, R. (2000)- "The Transaction Cost Advantage of Guanxi-Based Business Practices", Journal of World Business, Spring, Vol. 35, Issue 1, pp. 21-45.
- Staubus, G. (1971)- "Activity Costing and Input-Output Accounting", Homewood, IL: Irwin, Richard.
- Stewartt 111, G. (1994)- "EVA tm: Fact and Fantasy", Journal of Applied Corporate Finance, Vol. 7, No. 2, pp. 71-84.
- Stinchcombe, A. (1965)- "Social Structure and Organisations", In March, J. "Handbook of Organisations", pp. 142-193, Chicago, IL: Rand McNally.
- Stinchcombe, A. (1968)- "Constructing Social Theories", Chicago: University of Chicago Press.
- Strauss, A. and Cobrin, J. (1990)-" Basics of Qualitative Research: Grounded Theory Procedures and Techniques", Sage, Newbury Park, CA.
- Sudman, S. and Bradburn, N. (1982)- "Asking Questions: A Practical Guide to Questionnaire Design", San Francisco: Jossey-Bass.

- Summerton, N. (1999)- Privatised Water: Who's Paying The Price? "New Stateman", 26 July.
- Sutherland, G. (1977)- "The Magic of Measurement, Transactions of The Royal Historical Society", 5th Series, Vol. 27.
- Swieringa, R. and Wieck, K. (1987)- "Management Accounting And Action", Accounting, Organisations and Society, pp. 293-308
- Tani, T. Okano, H. Shimizn, N. Iwabuchi, Y. Fukuda, J. and Cooray (1994)- "Target Cost management in Japanese Companies: Current State of The Art", Management Accounting Research, Vol. 5, 1, pp. 67-81.
- Tayles, M. and Walley, P. (1994)- "Integrating Manufacturing and Management Accounting Strategy", Paper Presented at The British Accounting Association Conference at Winchester.
- Taylor, B. and Graham, C. (1992)- "Information For Strategic Management", Management Accounting (CIMA), January, pp. 52-54.
- Thames Water (2000)- "Annual Report and accounts 2000".
- Thames Water (2000)- "Summary Report and Accounts 2000".
- The Concise Oxford Dictionary (1983)- Oxford, Oxford University Press.
- The Confederation of British Industry (1991)- "Efficient Water Management Guidelines For Business", Gwynne Printers Ltd Hurstpierpoint, Sussex.
- The Financial Times (1997)- "Prescott Gets Tough on Water Leaks", 20 May.
- The General Organisation for Greater Cairo Water Supply (1996)- "Introduction for Strategic Planning", Workshop in GOGCWS, 14 Feb.
- The General Organisation for Greater Cairo Water Supply (no date source given, a) "The Strategic Plan for the General Organisation for Greater Cairo Water Supply for 1998-2008".
- The General Organisation for Greater Cairo Water Supply (no date source given, c) "Greater Cairo Water", The Administration of Public Relations.
- The General Organisation for Greater Cairo Water Supply (no date source given, b) "The General Organisation for Greater Cairo Water Supply, Program of the Accounting Systems The Original Memorial". The General Administration for Training.
- Thelen, K. and Steinmo, S. (1992)- "Historical Institutionalism in Comparative Politics". In Steinmo, S. Thele, K. and Longstreth, F. "Structuring Politics:

- Historical Institutionalism in Comparative Analysis", Cambridge, UK: Cambridge University Press.
- Thobani, M. (1997)- "Formal Water Market: Why, When, and How to Introduce Tradable Water Rights", The World Bank Research Observer, Aug.
- Thomas, A. (1974)- "The Allocation Problem: Part Two", Studies in Accounting Research, No. 9, American Accounting Association.
- Thomas, D. (1986)- "The Union Response to De-nationalisation", In Kay, J. Mayer, C. and Thompson, D. (eds), "Privatisation and Regulation - The U.K. Experience", Oxford: Clarendon Press.
- Thompson, G. (1994)- "Early Double-Entry Bookkeeping As Social and Institutional Practice", Cambridge: Cambridge University Press.
- Thompson, J. (1967)- "Organisations in Action", New York: McGraw-Hill.
- Thompson, J. and Tuden, A. (1959)- "Strategies, Structures and Processes of Organisational Decision", In Thompson, J. et al. (eds.), "Comparative Studies in Administration", University of Pittsburgh Press.
- Tolbert, P. and Zucker, L. (1984)- "Institutional Sources of Change in The Formal Structure of Organizations: The Diffusion of Civil Service Reform, 1880-1935", Administrative Science Quarterly, 28, pp. 22-39.
- Tolbert, P. and Zucker, L. (1986)- "The Institutionalization of Institutional Theory". In Clegg, S. Hardy, C. and Nord, W. (eds.), "A Handbook of Organization Studies", pp. 175-190. London: Sage.
- Tool, M. (1993)- "The Theory of Instrumental Value: Extensions, Clarifications", In Tool, M. (eds), "Institutional Economics", Boston and London, Kluwer Publishers, pp. 119-159.
- Trice, H. and Beyer, J. (1984)- "Studying Organizational Cultures Through Rites and Ceremonials", Academy of Management Review, 9, pp. 653-669.
- Trussel, J. and Bitner, L. (1998)- "Strategic Cost Management: An Activity-Based Management Approach", Management Decision, 36/7, pp. 441-447.
- Turney, P. (1991a)- "How Activity-Based Costing Helps Reduce Cost", Journal of Cost management for Manufacturing Industry, pp. 29-35.
- Turney, P. (1991b)- "Common Cents: The ABC Performance Breakthrough", Cost Technology, Hillsboro.
- Tversky, A. and Kahneman, D. (1974)- "Judgement Under Uncertainty", Science, 185, pp. 1124-1131.

- Utton, M. (1986)- "The Economics of Regulation Industry", Oxford: Blackwell.
- Veblen, T. (1898)- "Why Is Economics Not An Evolutionary Science", Quarterly Journal of Economics 12, pp 373-397.
- Veblen, T. (1919)- "The Place of Science in Modern Civilisation and Other Essays", New York: Huebsch.
- Vickers J. and Wright, V. (1989)- "An Overview", In Vickers, J. and Wright, V. (eds), "The Politics of Industrial Privatisation in Western Europe", London: Cass.
- Vickers, J. and Yarrow, G. (1988)- "Privatisation: An Economic Analysis", Cambridge, MA: M.I.T. Press.
- Vollmers, G. (1994)- "Relevance Found: A Rereading of the Cost Accounting Literature and A Reassessment of The Decline of Cost Accounting", Paper Submitted to Fourth Interdisciplinary Perspectives on Accounting Conference, University of Manchester.
- Waldron, D. and Galloway, D. (1988)- "Throughput Accounting 2: Ranking Products Profitability", Management Accounting, December, pp. 34-35.
- Waldron, D. and Galloway, D. (1989a)- "Throughput Accounting 3: A Better Way to Control Labour Costs", Management Accounting, January, pp. 32-33.
- Waldron, D. and Galloway, D. (1989b)- "Throughput Accounting 4: Moving to Complex Products", Management Accounting, pp. 40-41.
- Walker, M. (1998)- "Management Accounting and The Economics of Internal Organisation: A Review Essay", Management Accounting Research, 9, pp. 21-30.
- Walsh, J. (1995)- "Managerial and Organizational Cognition: Notes From a Trip Down Memory Lane", Organization Science, 6, pp. 280-321.
- Ward, K. (1992)- "Strategic Management Accounting", Butterworth-Heinemann Ltd.
- Ward, K. (1993)- "Corporate Financial Strategy", Oxford, Butterworth-Heinemann.
- Water Bulletin, (1992)- 30 October.
- Waterhouse, J. and Teissen, P. (1978)- "Contingency Framework for Management Accounting Systems Research", Accounting, Organisations and Society, pp. 65-76.
- Watson, D. (975)- "Contingency Formulations of Organisational Structure: Implications for Managerial Accounting", In Livingstone, J. "Managerial Accounting: The Behavioural Foundations", Gird.
- Watson, G. (1976)- "Resistance to Change", In Bennis, W. Benne, K. Chin, R. Corey, K. (eds.), "The Planning of Change", pp. 488-497, New York: Holt, Rinehart and Winston.
- Watts, R. and Zimmerman, J. (1986)- "Positive Accounting Theory: A Ten Year Perspective", Englewood Cliffs NJ: Prentice-Hall, Inc.
- Weber, M. (1968)- "Economy and Society: An Interpretative Sociology", Vols 3, New York: Bedminister.
- Weber, M. (1969)- "Economy and Society", Vols 3, Bedminster Press.
- West, L. (1999)- "Making Wave Overseas", March.
- Westney, D. (1993)- "Institutionalization Theory and The Multinational Corporation", In Ghoshal, S. and Westney, D. (eds.), "Organization Theory and the Multinational Corporation", pp. 53-76, New York: St. Martin's Press.
- Whitehead, C. (1988)- "Introduction: Theory and Practice", In Whitehead, C. (ed.), "Reshaping the Nationalised Industries", Oxford: Policy Journals.
- Whyte, W. (1984)- "Learning From the Field: A Guide From Experience", Beverly Hills, Claif.: Sage.
- Wildavsky, A. (1975)- "Budgeting. A Comparative Theory of Budgeting", Boston, Little Brown and Co.
- Williams, A. and Giardina, E. (1993)- "Efficiency in The Public Sector: The Theory and Practice of Cost-Benefit Analysis", Published by Edward Elgar Publishing Limited.
- Williams, J. and Ashford J. (1994)- "Management Accounting Control System: Knowledge", Charted Institute of Management Accountants, London.
- Williamson O. (1992)- "Markets, Hierarchies, and The Modern Corporation: An Unfolding Perspective", Journal of Economic Behavior and Organization, 17, pp. 335-352.
- Williamson O. (1993)- "Transaction Cost Economics and Organization theory", Industrial and Corporate Change, 2, pp. 107-156.
- Williamson, O. (1975)- "Markets and Hierarchies: Analysis and Antitrust Implications", New York: Free Press.
- Williamson, O. (1979)- "Transaction Cost Economics: The Governance of Contractual Relations", Journal of Law and Economics, 22, pp. 233-261.
- Williamson, O. (1980)- "The Emergence of The Invisible Hand: Implications For Industrial Organisation" In Chandler, A. and Deams, H. (eds) "Managerial

- Hierarchies: Comparative Perspectives on The Rise of The Modern Business Enterprise", Cambridge, Mass., Harvard University Press.
- Williamson, O. (1981a)- "The Economics of Organization: The Transaction Cost Approach", American Journal of Sociology, 87, pp. 548-577.
- Williamson, O. (1985)- "The Economic Institutions of Capitalism: Firms, Markets and Rational Contracting", New York, Free Press.
- Williamson, O. (1991)- "Comparative Economic Organisation: The Analysis of Discrete Structural alternatives", Administrative Science Quarterly, 36, pp. 269-296.
- Williamson, O. (1996)- "The Mechanisms of Governance", New York: Oxford University Press.
- Wilson, R. (1991)- "Strategic Management Accounting", In Ashton, D. Hooper, T. and Scapens, R. "Issues in Management Accounting", Prentice Fall, pp. 82-105.
- Winter, S. (1991)- "On Coase, Competence and The Corporation". In Williamson,
   O. and Winter, S. (eds.), "The Nature of The Firm: 179-195", New York: Oxford University Press.
- Wong, M. (1996)- "Strategic Management Accounting", Management Accounting, London, April.
- Xin, K. and Pearce, J. (1994)- "Guanxi: Good Connections As Substitutes for Institutional Support", Academy of Management Best Paper Proceeding, pp. 163: 167.
- XINHUA, (2000)- "Egypt Committed to Economic Reforms Despite Challenges Ahead: PM", 02/05/2002.
- Yin, R. (1981)- "The Case Study Crisis: Some Answers", Administrative Science Quarterly, 26, pp. 58-65.
- Yin, R. (1984)- "Case Study Research: Design and Methods", Beverly Hills, CA: Sage Publications.
- Yin, R. (1989)- "Case Study Research: Design and Methods", Beverley Hills, CA: Sage.
- Young, J. (1996)- "Institutional Thinking: The Case of Financial Instruments", Accounting, Organisations and Society, Vol. 21, pp. 487-512.
- Young, M. and Selto, F. (1990)- "Managerial Accounting Information and Modern Manufacturing: A Review of The Literature and Directions for Research", Working Paper, Division of Accounting University of Colorado, Boulder, Colorado.

- Young, S. and Selto, F. (1991)- "New Manufacturing Practices and Cost Management: A Review of The Literature and Directions For Research", Journal of Accounting Literature", 10, pp. 265-297.
- Zald, M. (1970)- "Political Economy: A Framework for Comparative Analysis" In Zald, M. (ed.), "Power in Organisations", pp. 221-261, Nashville, TN: Vanderbilt University Press.
- Zald, M. (1984)- "The Sociology of Enterprise, Accounting and Budget Rules: Implications For Organisation Theory", Accounting, Organisations and Society, 11, pp. 327-340.
- Zeff, S. (1972)- "Foreign Accounting principles in Five Countries: A History and An Analysis of Trends", Stipses Publishing Company.
- Zimmerman, J. (1976)- "Budget Uncertainty and The Allocation Decision In A Non-profit Organisation", Journal of Accounting Research, Autumn, pp. 301-319.
- Zimmerman, J. (1997)- "Accounting For Decision Management and Control", 2nd edition, ChigacagoI11, Richard D. Irwin.
- Zucker, L. (1977)- "The Role of Institutionalisation in Cultural Persistence", American Sociology Review, 42, pp. 726-743.
- Zucker, L. (1983)- "Organizations As Institutions", In Bacharach, S. (ed.), "Research in The Sociology of Organizations", pp. 1-43, Greenwich, CT: JAI Press.

## **Appendix One**

#### A1.1 Letters of the access for the GOGCWS

Sheffield Hallam University Pond St. Sheffield S1 1WB

March 13th 2000 To. Professor El-Sharkawy. Embassy of Arab Republic of Egypt. London

Dear Professor El-Sharkawy,

I write with respect to Mr. Yasser Abd-El-Naby; MM 8048

Mr. Abd-El-Naby began his studies at the university in February 1999. The school has recommended that Mr. Abd-El-Naby transfer from M.Phil. to Ph.D. and this will be going to the next university research degree committee (16/2/2000); this follows a presentation and examination of the proposal (Accounting Information System in the water Industry: The case of Cost Management). He is currently undergoing the second and final semester of his research training programme having successfully completed the first semester.

Mr. Abd-El-Naby has managed to build his theoretical framework and his empirical work will be on "The General Organisation for Greater Cairo Water Supply". I wonder if you can help him to get access to this company by no later than the beginning of May 2000 in order to finish his thesis on time (no later than the beginning of September 2001).

Yours sincerely,

**Professor Anthony J. Berry.** 

Sheffield Hallam University Pond St. Sheffield S1 1WB

April 20th 2000

To. Professor El-Sharkawy. Embassy of Arab Republic of Egypt. London

Dear Professor El-Sharkawy,

I write with respect to Mr. Yasser Abd-El-Naby; MM 8048

I have pleasure in enclosing the fieldwork plan for Mr. Abd-El-Naby's study of "Accounting Information Systems in the Water Industry: The Case of Cost Management".

I estimate that he will need up to four months (at least three months) in Egypt. A considerable amount of preparation has been taken from published sources.

This will be an original and useful contribution to the management of the water Industry in Egypt and the world.

Yours sincerely,

**Professor Anthony J. Berry.** 

#### Arab Republic of Egypt

#### The Central institution for General Statistical

# Decision of the chief of Central institution for General Statistical by empowerment No. 652 (Year 2000)

Relating to the researcher/ Yasser Kamal El-Sayed Abd-El-Naby - who works as assistant researcher in the Institute of National Planning and is a university student in England- to do his field work to obtain a Ph.D. on "Accounting Information System in The Water Industry: The Case of Cost Management"

#### The chief of secretariat-general sector

\* After seeing the presidential of Egypt issued order (No. 2915 / Year 1964) relating to establishing and organising the Institution

\* And the decision of the Chief of the Institution (No. 231/ Year 1968) related to statistics, consultations and explorations.

\* And the acceptance of the chairman of the Institution (07/07/1998) for the empowerment to the chief of secretariat-general sector to approve the decisions relating to security of data.

\* And the letters of general administration of mission (No. 1186, 60/05/2000), (unnumbered, 60/05/2000)

#### Decision

(1) The researcher/ Yasser Kamal El-Sayed Abd-El-Naby -who works as assistant researcher in the Institute of National Planning and the member of foreign mission in England- to do his fieldwork which is mentioned above.

(2) This research involves the Chairman of the Board of directors, financial manager, cost department manager, accountants, auditors, marketing managers, production managers, some production engineers, consultants of the General Organisation for Greater Cairo Water Supply and the organisations which are involved in setting the pricing policy for water in Egypt. It also involves the people responsible for the control of water organisations in the Accounting Central Institution, the general manager of

public relations in the Ministry of public works and water resources and the ministry of Agriculture. It is conditional that this research is approved by their administration security and the data collected will be used for academic purposes only.

(3) The data will be collected according to the statement which is prepared for this purpose and stamped by the Central Institution for General Statistical (being a 23 page questionnaire), taking up to three months only (from the issue of this decision).

(4) The process of this fieldwork is not allowed to start until this decision is issued.

(5) The researcher should submit two copies of the primary results of this fieldwork to the Central Institution for General Statistical and submit two copies of the final results on completion.

(6) This decision comes into action on the date of issue.

Issued in 20/06/2000

#### Sami Hussan Fauat

# Arab Republic of Egypt The Central institution for General Statistical

<b>Record</b> : R/ AM/ A 674	Subject:
Date: 21/60/2000	
Attached:	

# F. A. O the general manager of the general administration of mission within Ministry of Higher Education.

Goodness greeting and following

Referring to the letters (No. 1186, 60/05/2000) and (unnumbered, 13/60/2000) and its attachments concerning the application for the acceptance for the researcher/ Yasser Kamal El-Sayed Abd-El-Naby - who works as assistant researcher in the Institute of National Planning and is a university student in England- to do his fieldwork for his Ph.D. on "Accounting Information System in the Water Industry: The Case of Cost Management". Also referring to the proposal for above, stamped by the security stamp of the Central Institution for General Statistical, and our letter to you (No. 535, 16/05/2000).

We would like to inform that the Central Institution for General Statistical approves that the researcher can do the fieldwork mentioned above. Attached to the decision of the chief of Central institution for General Statistical empowerment No. 652 (Year 2000) which is a requirement for this role. The researcher should submit two copies of the primary results of this fieldwork to the Central Institution for General Statistical and submit two copies of the final results on completion according to No.5 of the decision.

#### Many thanks

The general manager for general administration of security.

Arab Republic of Egypt Ministry of Higher Education Mission Department

Financial supervision and emissary care The Administration of emissary cares Department of Scientific Communication

#### Dear Sir, Chairman of General Organisation for Greater Cairo Water Supply

#### Goodness greeting and following

I would like to inform you that Mr. Yasser Kamal El-Sayed Abd-El-Naby -who works as assistant researcher in the Institute of National Planning and is a university student in England studying for a Ph.D. in "Accounting Information System in the Water Industry: The Case of Cost Management "- came to Egypt to do his fieldwork which will involve:

The chairman of the Board of Directors, Financial Manager, Cost Department Manager, accountants, auditors, marketing managers, production managers, some production engineers, consultants of General Organisation for Greater Cairo Water Supply and the organisations which are involved in setting the pricing policy for the water in Egypt.

I would like to inform you that the Central Institution for General Statistical agreed that -in the letter to us No. 535 (19/05/2000) and the decision issued at 20/60/2000 - Mr. Yasser Abd-El-Naby may do his fieldwork by condition which is that this research is approved by your administration security. The individual data is by law confidential and the use of this data will be for academic purposes only. The questionnaire consists of 23 pages and stamps from the Central Institution for General Statistical.

There is also permission to give him the following information: -

- Profit and loss accounts, balance sheet and the planning budget.

- Cost accounts.

- The standards of internal and external control.

- A general view of each department, its roles and aims and knowledge of the communication between each department and others for 30 years.

We hope to facilitate his role and give him the data needed.

Many thanks for your contribution.

24/60/2000

Ehab

The general manager

Abo-El-Wafa El-Tiab Abo-El-Wafa

# A1.2 The GOGCWS

#### A1.2.1 The GOGCWS' law and decrees

#### A1.2.1.1 General laws and decrees

- 1- Law No. 47 of 1978 and its amendments; concerning the State Civil Servants System.
- 2- The executive regulation of Law No. 47 of 1978; issued by Decree No. 2 of 1978 and its amendments of the Civil Service Affairs Committee of the Central Agency for organisation and administration.
- 3- Law No. 43 of 1979 and its amendments concerning the Local Administration System.
- 4- Prime Minister's Decree No. 707 of 1979; promulgating the executive regulations of the local administration system law.
- 5- Law No. 53 of 1979; regarding imposing upon government units and public sector to provide basic data for planning of manpower and vocational training in accordance with employment information forms.
- 6- Prime Minister's Decree No. 1309 of 1979; regarding economic general organisations and financing funds of economic nature.
- 7- Law No. 49 of 1977 and its amendments; regarding leasing and selling places and organising relationship between the lessor and the lessee.
- 8- The executive regulations for the law of organising the relationship between the lessor and the lessee.
- 9- Law No. 106 of 1976 and its amendments; regarding directing and organising construction works.
- 10-Law No. 143 of 1981; regarding desert lands.
- 11- State Minister's Decree No. 198 of 1982; issuing the executive regulations of desert lands law.
- 12- Law No. 26 of 1983; regarding sewers and sanitary drainage systems.
- 13- Law No. 16 of 1985; amending provisions for workers in sewers and sanitary drainage.
- 14-Law No. 12 of 1984; regarding Irrigation and drainage.
- 15- Law No. 48 of 1882; regarding the protection of the River Nile and water ways from pollution.
- 16-Law No. 3 of 1982; regarding developmental planning.

- 17-Law No. 59 of 1979; regarding establishment of new developmental communities.
- 18-Development and Housing Minister's Decree No. 600 of 1982; promulgating the executive regulations for law No. 3 of 1982 re-developmental planning
- 19-Law No. 25 of 1992; amending some provisions of law of directing and organising construction works, law of establishing fund to finance economic housing projects and law of developmental planning.
- 20- Circular letter No. 30 of 1987; regarding authorities and responsibilities included in law 47 of 1978.
- 21- Prime Minister's Decree No. 615 of 1986; regarding maximum wages and in-kind in government units, local administrative units, general organisations and authorities.
- 22-Law No. 27 of 1994, regarding arbitration in civil and trade articles.
- 23- Presidential Decree No. 399 of 1987; regarding agreement of regional Cairo centre for international trade arbitration.
- 24-Law No. 144 of 1988; amending law of the central agency for accounting.
- 25- Presidential Decree No. 129 of 1964; promulgating the law of the central agency for accounting.
- 26-Presidential Decree No. 796 of 1972; regarding formation and responsibilities of the Training High Council.
- 27-Law No. 118 of 1964 and its amendments; regarding establishing the central agency for organisation and administration.
- 28-Presidential decree No. 1085 of 1964; concerning organising and determining responsibilities of the Central Departments of the Central Agency for Organisation and Administration.
- 29- Presidential Decree No. 2915 of 1964; concerning establishing the Central Agency for Mobilisation and Statistics.
- 30-Law No. 54 of 1964; concerning the administrative control organisation.
- 31-Presidential Decree of law No. 35 of 1960; concerning statistics and census.
- 32- Presidential Decree No. 1115 of 1974; concerning the Organisation of the Ministry of Finance.
- 33-Presidential Decree No. 1100 of 1974; determining responsibilities of the State Minister of Local Administration and Public Organisations.

#### A1.2.1.2 Laws Related to management development program

1- Law No. 47 of 1987 and amendments; concerning the state civil servants system.

- 2- The executive regulations of law No. 47 of 1978; issued by decree No. 2 of 1978 of the civil service affairs committee of the central agency for organisation and administration; and amended by decree No. 75 of 1992 of the State Minister for Administrative Development.
- Decree No. 1 of 1979 of the Minister of Administrative Development; concerning employing experts.
- 4- Decree No. 2 of 1979 of the Minister of Administrative Development; concerning employing foreign experts.
- 5- Decree No. 3 of 1979 of the Minister of Administrative Development; concerning employing workers who carry out temporary works.
- 6- Decree No. 4 of 1979 of the Minister of Administrative Development; concerning employing graduated apprentices.
- 7- Law No. 39 of 1975 amended by Law No. 49 of 1982; concerning qualifying the handicapped.
- 8- Law No. 46 of 1972; concerning the state council.
- 9- Decree for laws No. 19 of 1959; concerning validity of provisions of the administrative prosecution on the general organisations.
- 10-Law No. 145 of 1957; concerning liquidation of Cairo Water Company.
- 11-Presidential Decree No. 577 of 1957; concerning Cairo Water utility Department.
- 12-Law No. 61 of 1963; promulgating the general organisations law.
- 13-Presidential Decree No. 4417 of 1965; concerning transfer of Cairo Water Utility Department to Arab stock Company.
- 14-Presidential Decree No. 1938 of 1968; concerning establishing the general organisation for Greater Cairo Water Supply.
- 15-Presidential Decree No. 33 of 1977; concerning re-organisation of units of organisation and administration in the various agencies.
- 16-Presidential Decree No. 70 of 1981; concerning organising public service departments.
- 17- Presidential Decree No. 625 of 1981; concerning establishing follow-up offices in the state administrative units and the public sector units.
- 18-Law No. 53 of 1979; concerning obliging the governmental and public sectors to provide basic data for planning manpower and vocational training.
- 19- Presidential Decree No. 118 of 1964; concerning establishing the central agency for organisation and administration.

- 20-Presidential Decree for law No. 42 of 1967; concerning delegation of authorities.
- 21-Law No. 115 of 1983; amending some provisions of state civil servants law No. 47 of 1978.
- 22-Decree No. 75 of 1992 of the Minister of Administrative development; amending some provisions of the executive regulations of state civil servants law.
- 23- Law No. 34 of 1992; amending some provisions of state civil servants law No. 47 of 1978.
- 24-Decree No. 337 of 1992 of the head of the central agency for organisation and administrative; concerning re-approval of the comprehensive assessment of the organisation's job with related remarks.
- 25-Law No. 5 of 1991; concerning leadership civil positions in the state administrative units and the public sector units.
- 26-Prime Minister's Decree No. 1596 of 1991; promulgating the executive regulation of law No. 5 of 1991.
- 27-Presidential Decree No. 796 of 1972; concerning formation and authorities of the training high council.
- 28-Law No. 39 of 1975; concerning qualifying the handicapped.
- 29-Presidential Decree No. 2420 of 1971; concerning organising the governmental system.
- 30-Presidential Decree No. 72 of 1975; concerning organising the Ministry of Housing and Construction.

#### A1.2.1.3 Laws Related to financial viability

- Presidential Decree for promulgating the central agency for accounting law No. 129 of 1964.
- 2- Law No. 127 of 1981; concerning governmental accounting.
- 3- Law No. 105 of 1992, amending provisions of law No. 127 of 1981 concerning governmental accounting.
- 4- Law No. 308 of 1958 and its amendments; concerning administrative sequestegation.
- 5- Law No. 66 of 1963 and its amendments; promulgating the customs law.
- 6- Minister of Finance Decree No. 101 of 1985; concerning customs release procedures for goods imported for ministers, governmental authorities, general organisations and Public sector companies.

- 7- Minister of Finance Decree No. 316 of 1983; concerning the temporary release system on private riding cars.
- 8- Prime Minister's decree No. 615 of 1986; regarding maximum wages and in-kind in government units, local administrative units, general organisations and authorities, banks and public sector companies.
- 9- Minister of Finance Decree No. 228 of 1985; concerning arbitration system in disputes between good's owners and the customs authority.
- 10-Minister of Finance Decree No. 274 of 1986; amending the principles of the temporary release system.
- 11-Presidential Decree No. 186 of 1986; promulgating organising customs exemptions law.
- 12- Presidential Decree for Law No. 1987 of 1986; concerning nullification of taxes and fees annexed with the customs taxes.
- 13-GOGCWS Chairman's Decree No. 101 of 1976; issuing the financial regulations for the budget and accounts.
- 14-Law No. 35 of 1972; concerning protecting of public money.
- 15-Law No. 105 of 1985; concerning the maximum of wages.
- 16-Prime Minister's Decree No. 615 of 1986; concerning the maximum of wages.

#### A1.2.1.4 Laws related to TSOM program

- Law No. 140 of 1956; amended by law No. 56 of 1957; amended by law No. 174 of 1960; and Presidential Decree No. 1356 of 1961; and law No. 209 of 1980; and law No. 187 of 1981; and law No. 129 of 1982; concerning occupying public roads.
- 2- Law No. 84 of 1986; concerning public roads.
- 3- Minister of Manpower Decree No. 53 of 1982; identifying enterprises, safety and vocational health agencies and training authorities.
- 4- Minister of Manpower Decree No. 36 of 1982; concerning forms for statistics of injuries serious accidents and diseases.
- 5- Presidential Decree No. 114 of 1984; concerning formation of consulting agencies for safety and vocational health.
- 6- Minister of Manpower Decree No. 2 of 1985; concerning organising work of the high consulting council for safety and vocational health.
- 7- Law No. 57 of 1978; concerning getting rid of pools and swamps; and preventing excavation works.

- 8- Law No. 38 of 1967; amended by law No. 129 of 1982 concerning public cleanliness.
- 9- Law No. 48 of 1982; concerning protection of the River Nile and waterways from pollution.
- 10-Minister of Irrigation Decree No. 8 of 1983; promulgating the executive regulations for law No. 48 of 1982 concerning protection of the River and waterways from pollution.
- 11-Presidential Decree No. 93 of 1962; amended by law No. 48 of 1982 concerning drainage of liquid waste.
- 12-Law No. 10 of 1990; concerning expropriation of real property for public utility.
- 13- Presidential Decree No. 392 of 1979; organising the Egyptian General Organisation for Standardisation and Production Quality.
- 14-Law No. 106 of 1976; concerning directing and organising construction works.
- 15-Law No. 25 of 1992; amending provisions of law of direction and organising construction works, and law of establishing fund for financing economic housing projects, and law of development planning.
- 16-Presidential Decree No. 431 of 1979; concerning organising the Egyptian Committee for Hydraulics and Water Sources and Enterprises for Irrigation and Drainage.
- 17- Presidential Decree No. of 1986; concerning organising the general organisation for geological survey and mining projects.
- 18-Decree No. 48 of 1968; concerning organising necessary precautions to protect the workers during work from health detriments, work and machines hazards.

#### A1.2.1.5 Laws related to materials management and procurement

- 1- Law No. 9 of 1983 and its amendments; promulgating the law on organisation of adjudications and bidding.
- 2- GOGCWS Decree published in the official journal on 7/7/1986 edition 153; promulgating GOGCWS adjudication and bidding regulations.
- 3- Law No. 31 of 1984; concerning some principles for disposal of specific state property.
- 4- Law No. 7 of 1991; concerning some provisions related to the specific state property.
- 5- Administrative Decree No. 84 of 1981; establishing a unit for inventory control.

- 6- Prime Minister's decree No. 137 of 1978; concerning establishing units for Inventory Control.
- 7- GOGCWS stores regulations; issued in accordance with GOGCWS board of directors approval in meeting on 18 October 1975.
- 8- Government stores regulations.
- 9- Legal opinion by the general assembly of the state council in its meeting on 6 May 1985; obliging applying the provisions of the agreement of the American grant (USAID) in matters related to choice of contracting parties and contracting procedures in contrary to provisions of law No. 9 of 1983 and its executive regulations.

#### A1.2.1.6 Laws and Decrees related to project management and administration

- 1- USAID Regulations.
- 2- Law No. 9 or 1988 and its amendments; concerning adjudications and auctions law.
- 3- GOGCWS adjudications and auctions regulations.
- 4- Presidential Decree No. 627 of 1981 concerning establishing Information and Documentation Centres in the State Administrative Agencies and Public Organisations and identifying their responsibilities.
- 5- Presidential Decree No. 4720 of 1979 concerning the system of maintaining state official Documents and method of their publishing.
- 6- Presidential Decree No. 2915 of 1964 and its amendments; Concerning establishing and organising the central Agency for Mobilisation and Statistics.

#### A1.2.2 Stages of production of fresh water

The production of fresh water should pass through five main stages. Figure A1.1 shows the stages of production of fresh water.



#### Figure A1.1: Stages of production of fresh water

Source: Based on the GOGCWS (no date source given, c), p. 10-11.

The following section will give a brief explanation of each stage.

#### A1.2.2.1 Pulling water from the Nile and pumping it

There are some conditions regarding pulled water.

1- The pulling should be from flowing (streaming) water not stagnant water by the extension of pipes for the appropriate distance and deep inside the waterway of the Nile.

2- The pulling of water should be far away from pollution sources (industrial pollution) or any human crowd on the Nile beach.

3- Setting up protective measures on the pulled pipes to prevent any pollutants from the entrance with the pulled water.

4- The pulling should be by suitable pump of irrigation water with the designated water quantity of station to produce it.

#### A1.2.2.2 Precipitation

In this stage the clarification chemical and preliminary purification (chlorinate) are added and then filtration and finally this goes to the precipitation tank.

#### A1.2.2.3 Filtration

This stage can be illustrated in figure A1.2.

Figure A1.2: Filtration stage



Source: Based on the GOGCWS (no date source given, c), p. 11.

#### A1.2.2.4 Refining

Figure A1.3. will illustrate this stage.

#### Figure A1.3: Refining stage



Source: Based on the GOGCWS (without date, c), P. 11.

#### A1.2.2.5 Pumping to the city dragnet

Pumping to the city dragnet is through earth filtration pumps and is pumped to the customer by appropriate pressure which ranges between 60-100 m.

#### A1.2.3 Organisation structure of the GOGCWS

According to the organisation structure (see figure A1.4), there are 6 central administrations. The following section will illustrate the function of board of directors of the GOGCWS and these 6 central administrations.

#### A1.2.3.1 The function of the board of directors of the GOGCWS

The board of directors of the GOGCWS represents the top of the organisation structure and the important functions of it are as follows:

1- To suggest the general policy of the institution which the organisation manages according to the scope of the general policy of the state.

2- To suggest project development plans and programs for carrying them out.

3- To set the internal rules and the decisions related to finance and administration of the GOGCWS and its workers.

4- To suggest the tariff of water sold.

5- To approve the annual budget plan of the GOGCWS and financial statement.

6- To look at the periodic reports which are introduced for following up the work and its financial position.

#### A1.2.3.2 The functions of the chief of the central administration for projects

The chief of the central administration for projects is responsible for the supervision of preparation for designs for new projects and vertical or horizontal expansion for present projects.

#### A1.2.3.3 The functions of the chief of the central administration for stations

The chief of the central administration for station is responsible for the supervision of the work of the main stations, which is responsible for treatment to clean and filter water. He is also responsible for the supervision of all the requirements to run the station, and maintenance.

#### A1.2.3.4 The functions of the chief of the central administration for dragnets

The chief of the central administration for dragnets is responsible for the supervision of the work of the stations, projects for pressure development, operational programs and the maintenance of the dragnets.

# A1.2.3.5 The functions of the chief of the central administration for financial affairs

The chief of the central administration for financial affairs is responsible for the supervision of preparing the financial plan for the GOGCWS and following up the carrying out of it. He is also responsible for preparing the organisation's budget and financial statements.

# A1.2.3.6 The functions of the chief of the central administration of managerial affairs

The chief of the central administration of managerial affairs is responsible for the supervision of workers affairs such as; salaries, wages, general secretaries, all administrative aspects and preparing the workers plan for the GOGCWS.

#### A1.2.3.7 The functions of the chief of the central administration for legal affairs

The chief of the central administration for legal affairs is responsible for liaising with legal authorities and gives formal (legal) opinions in all subjects that the board of directors enquires about





El-phustat drinking water station, El-Madi drinking water station, Shamal Hellwan drinking water station, Kaver El-help drinking water station and El-tabien drinking 333 The GOGCWS has 13 stations as follows; Ameria drinking water station, Roughd El-Farag drinking water station, Mostured drinking water station, Shoubra El-Kama drinking water station, El-Rouda drinking water station, El-Geza drinking water station, Gezera El-Dahab drinking water station, Embaba drinking water station, water station (The GOGCWS "without date a": 120).





453

.



with members of the board of directors, managers, workers, employers of the GOGCWS to find their view and their expectation (b) Interviews with central government, local govern, popular committees institutions in Great Cairo City in order to find their expectation for the future of the GOGCWS, (c) Interviews with 334 Note: The GOGCWS wrote that these suggestions built on knowing the view of the owners of joint benefits with the GOGCWS through (a) Personal interviews most important customers to find their evaluation and expectation for the GOGCWS and (d) Sample of domestic customers of different level to find the previous aims. The researcher when he was doing documentation analyses found most of all these suggestions in Black and Veatch International (1997).

<sup>&</sup>lt;sup>335</sup> Egypt is transferring from the central system to the economic system market.



Figure A1.5.2 shows the general plan of the organisation structure at the re-organisation stage.



Figure A1.5.2: The plan of the organisational structure at the re-organisation stage

Figure A1.6: The environment of the GOGCWS



#### A1.2.4 The Unified Accounting System

# A1.2.4.1: The general rules for recording the economic organisation's transactions according to the Unified Accounting System

1- All records of the economic organisation's transactions are recorded according to the Accounting index to integrate between the accounts of the economic organisation and the national accounts.

2- Separate between the maturity and payment or collection to prepare the cash  $budget^{336}$ . According to this rule the transaction should be recorded as on credit even if cash and then one of the cash accounts is used when recording the payment or collection as shown in figure A1.7.





Source: Based on the GOGCWS, (no date source given, b), p. 19.

#### A1.2.4.2 The Accounting index

#### The aim of the accounting index

The aim of the accounting index is to help the accounting system to achieve its aim.

<sup>&</sup>lt;sup>336</sup> Cash budget is one of the statements, which GOGCWS has to prepare.

#### The basis of the establishment of the accounting index

There are three bases on which the accounting index is based and these three bases are as follows: (a) classification of accounts, (b) code of accounting and (c) explaining the core idea of each of them.

#### A. Classification of accounts

All accounts which have a connection with the activity of the economic organisation are enumerated and then these accounts are classified into homogeneous groups in order to satisfy the traditional requirements of financial accounting and prepare the national accounts.

Balance sheet accounts are classified to enable the preparation of the financial position and the statement of resources and uses, so we can know the contents of the capital and the financial sources of it.

The result accounts are classified into two groups: uses (expenses) and resources (revenues). The uses are classified as wages, general expenditures and transfers, and the revenues are divided into the current activity revenues and financial investments' interests of transfers. According to this classification the result of the economic organisation can be known (profit or loss).

In addition to the classification of the uses, classification of the accounts depends on the nature of them. It is divided into groups such as: production centres, production services centres, distribution services centres, administrative and financial services centres and capital process centres. This division shows the connection between financial accounting and cost accounting.

#### **B.** Accounts code

The Unified Accounting System takes the number code to ensure accuracy, ease of accounting direction, collection of accounting information and classification of it. The number code takes six levels as follows (see table A1.1);

The code	Accounts
1	Assets.
2	Liabilities.
3	Uses.
4	Resources.
5	Production centres control.
6	Production services centres control.
7	Distribution services centres control.
8	Administration and financial services control.
9	Capital process services control.

 Table A1.1: Accounts code of the level of general accounts

Source: the GOGCWS (no date source given, b), P. 15

The first level is the level of general accounting and this level is coded from 1 to 9 The second level - This level represents the general account and is coded by two numbers (starting from 11 to 99).

The third level - This is the subsidiary account and three numbers (111: 999) code it. The fourth level - This is the branch account and is coded by four numbers (1111: 9999).

The fifth level - This is the minor account and is coded by five numbers (11111: 99999). The sixth level - This is the analysis account level and is coded by 6 numbers (111111: 999999).

C. Explanation of the core idea of the (1) assets and their classification and (2) uses and their classification.

#### Assets and their classification

#### Assets 1

**Fixed assets 11.** These are transferred property or not transferred and tangible or acquisition or produced by the organisation for sales or transferred. Fixed assets include lands, buildings, constructions, installation, roads, machines and equipment.

**Projects on execution 12.** These include the fixed asset in preparation stage.

**Inventory 13.** This has all tangible or non-tangible goods, which the organisation buys for the production process to make complete products, or if the financial period is

finished and they are not completed (uncompleted products). Also it has all goods which the organisation buys to re-sell without any production process.

Long period lending 14. These are loans, which the organisation gives to other companies to be repaid over one year.

**Financial investments 15.** These are the investments of the economic organisation in governmental stock and national or international stocks.

**Debtors 16.** This represents the organisation's rights against other organisations resulting from the business activity of the organisation. It includes agent's commissions, receiving payable and varieties debtors.

**Different debts accounts 17.** This represents the organisation's rights against other organisations resulting the non-normal activity of the organisation. It includes different debtors, other debt balances and payable allocation operational revenues.

**Cash in the treasury and banks 18.** It includes two accounts: (a) cash in the treasury. The Unified Accounting System distinguishes between the cash in treasury of the unit and the cash in the subsidiary' treasury of it. And (b) current account bank. The Unified Accounting System distinguishes between money, which is the result of the process, and money, which is the result of capital process.

#### A1.2.4.3 The book group related to cost

This represents the records and books, which are used in the administration of financial statistics and cost. These records and books are illustrated as follows

*The records and books related to the fixed assets.* These include (a) Statistical book which records in it the direct bounds (bills) which comes in daybooks (journal) and the excluded assets from the goods requirements which come from the financial control on inventory administration. (b) Detailed book for the cost of fixed assets and its depreciation. (c) Special book for the distribution of fixed asset on the different stations and production centre for each asset alone.

*The records and books related to wages.* These are represented by the following records (a) Analytical records for wages to show the detailed wages according to production centres and stations. (b) Social insurance book. (c) Book for individual items. (d) Total wages book.

*The records and books related to goods requirement.* These are represented by the following: (a) Records for recording the excluded goods requirements and relating to the fixed assets. (b) Book recording within it all items of goods requirements.

*The records and books related to service requirements.* This includes as follows: (a) Service requirement book, which has, recorded within it, all items of service requirements<sup>337</sup>. (b) Current transferred expenses book. (c) Apportionment current transferred book. (d) Total book which includes the calculation of cost centres for current transferred expenses apportionment current transferred as total.

*The archives book.* This represents the following books: (a) Cash book daily entry, which has, recorded within it, all cash day-enter, which come from the central accounts administration. (b) A daily entry book which has, recorded within it, all the banks' journals which come from the central accounts administration.

*Other records.* Represented as follows: (a) Statistical records which have recorded within all data related to the cost of  $m^3$  of the water. (b) Record which has, recorded within it, all data sent to different external organisation.

#### Comments of the Central Accounting Institution regarding the book groups

The cost system lacks some forms and data such as; operation order and cards, average burdening for wages and expenses.

<sup>&</sup>lt;sup>337</sup> Service requirements represent the given services from others which are a requirement for the production process, such as expenses of maintenance, expenses of publicity and advertising, hiring out equipment and means of transportation, etc.

## **Appendix Two**

### A2.1 The Questionnaire and Interview guide

#### Introduction

This is an investigation for the Ph.D. research which I am currently preparing at Sheffield Hallam University (U.K.). The research examines the role of cost management in the Egyptian Water Industry. The Study aims to discover and describe the relationship between cost management and cost structure; the current cost structure for the provision of Egyptian Water Services and the rationale of it; methods of cost accounting that were and are used; and finally, the role of accounting in controlling the Egyptian Water Industry.

#### The data will be collected through:

1. Reading and analysing relevant documents: annual reports, financial plans and budgets, progress reports, description of and instruction for cost accounting systems, policy documents- the guide informing managers what they should do when certain things happen-, policies and rules procedures, government rules, 5 years financial statements and other suitable available documents.

2. Personal interviews and questionnaires: this will divide into three phases (steps):

A- The first step: conversation interview will be conducted with planner's/policy decision makers within such as; Ministry of Irrigation (Ministry of Public Works and Water Resources); Ministry of Agriculture; Government advisor; the people who set the price of the water Industry; and finally, conversation interview with the accountants responsible for controlling all the companies of the Water Industry. The aim of this step is to build knowledge basis of the Egyptian Water Industry.

B-The second step: Personal interviews and questionnaires- please see the questionnaires and the interview guide which is attached to this introduction- will be conducted on-site during one or two visits with senior managers, financial directors and managers of cost department of all the Water companies in Egypt.

C- The third step: personal interviews and questionnaires will be conducted during an on-site visit in Great Cairo Water Public Enterprise- A major case study of this research- with relevant persons at various levels of management (senior manager, cost directors, financial directors/ controllers, accounting staff, general marketing manager and general production managers); consultants and engineering's Great Cairo Water Public Enterprise.

Interviewing will be supported by tape-recording which will capture much more than the researcher's memory. Tape-recording will be used where possible provided the interviewee agrees. It is anticipated that the use of a tape-recorder will increase rapport between the interviewer and interviewee.

I should emphasise that the data will be used for academic purposes only and will not be released-under any circumstances- for any other use. Confidentiality will be assured. I appreciate that there are many demands on your time, but the more complete response will ensure a high quality of research.

#### Structure of the questionnaire

1- General background.

#### 2- Cost practice

A. Control structure and responsibility accounting.

B- Planning and control procedures.

C- Product costing (System background - system description - cost recognition and measurement - reporting).

D- Reporting.

3- Cost management policies.

4- Company environment.

# Confidential

### **General Background**

# Section 1: About The Respondent.

(1) Name of person filling the questionnaire:
(2) Job Title:
(3) Job Description:
***************************************
(4) Length of time with company:

# Section 2: About The Company.

(1) Name of the company:
(2) Established date of the company:

- (3) Business activities in your company: .....
- (4) For the last financial year, would you please indicate:

Total sales.	
Total assets.	
No. of employees.	

# (5) Who are your customers? Please, describe the company sales to each customer:

	Company Sales					
<b>Category Of Customer</b>	Less	10-	20-	40-	60-	80-
	than	20%	40%	60%	80%	100%
	10%					
Domestic						
Public (schools, hospitals,)						
Business (agriculture, industry,)						
Others, Please specify:						
#### **Cost Practice**

Section	1:	Control	structure an	ıd res	ponsibility	accounting.
---------	----	---------	--------------	--------	-------------	-------------

Q1: How do you describe the strategy of your company? (Please tick the appropriate				
box)				
As a unitary organisation As a multi-divisional organisation Other, please specify organisation				
Q2: How is the company organised? Please tick the appropriate box)				
According to the nature of products       According to the nature of the geographic areas       According to the nature of the nature of the production process       Other, please specify				
Q3: How do you describe the control structure at your company/division? (Please tick				
the appropriate box)				
Investment centre Profit centre Contribution centre Cost centre Other, please specify				
Q4: How does the control structure fit within the organisation structure? (Please tick the appropriate box)				
A good fit Reasonable fit Problematic fit				

## Section 2: Planning and control procedures.

Q1: Which one of the following formal planning and control approaches are used in your company? (Please tick the most appropriate answer for each process, in the table below:

Planning and control process	is in use now	is planned to be used in the future	was used in the past, but not now	is not used or planned to be used at all	I do not know
Operational budgeting procedures					
Capital budgeting procedures					
Zero based budgeting approach					
Linear programming model					
Transfer pricing model					

Planning and control process	is in use now	is planned to be used in the future	was used in the past, but not now	is not used or planned to be used at all	I do not know
Focus on short run marginal cost model.					
Focus on medium run marginal cost model.					
Focus on long run marginal cost model.					
Product costing (Full absorption costing).					
Product costing (Activity based costing).					
Activity-based management (ABM).					
Throughput accounting.					
Life-cycle costing.					
Value-chain approach to strategic cost					
management.					
The accounting assessment of competitive					
position.					
Qualitative performance measures.					
The balanced scorecard.					
Benchmarking of costs against other					
companies.					
Target costing.					
"Kaizen" cost management (formal, regular					
cost reduction programmes).					
If you have any comment, please write these o	comments	below:			
Q2: Could you tell me about any other existin used in your division/business unit not mentio	g formal j oned abov	planning a	and contro	ol process	5

.....

Q3: How do you describe the annual budgeting process at your company? (Please tick
the appropriate box)
Top-down Top management set parameters: organisation Bottom-up Discussion and participation procedures adjust/decide Discussion and participation procedures adjust/decide
Section 3: Product costing.
3.1 System background.
Q1: How was the cost accounting system developed? (Please tick the appropriate box)
In-house Contractor Software Customer Borrowed from Others, please package prepared another agency specify
Q2: Is this cost system adequate for your needs? (Please tick the appropriate box)
No needs     Few needs     About half our needs     Most needs     All needs
Q3: Has there been a major revision to the method used in your company for estimating and measuring product cost? For example a shift from FAC to ABC. (Please tick the appropriate hox)
Yes I do not know
Yes No I do not know
• Is a major revision in process?
Yes No I do not know
• Has the introduction of the new product costing method led to any of the following
changes:
No changes       Changes to production       Changes to product mix       Changes in the structure of responsibility and       Other, please specify         process       accountability

Q4: In your view, do you consider the present method of estimating and measuring the product cost and practice as; (Please tick the most appropriate answer)

	Estimation	Measurement
Straightforward and simple		
Seems a reasonable compromise		
Needs significant change		
Needs simplifying		
Other, please specify		
If you have any comment, please write these comm	nents below:	
· · · · · · · · · · · · · · · · · · ·		
· · · · · · · · · · · · · · · · · · ·		
Q5: What is the estimated number of accounting/ operating the cost system?	finance personnel di	rectly involved in
Q6: What is the computer based configuration of company?	f the cost accounting	g system in your Other, please specify
Q7: What are the principal methods of data entry?	mputer- Personal comp connected to 1	outer- Other, please arge PC. specify
Q8: What are the sources of funding for the seaccounting system relates?	ervices/operations to	• which the cost
3.2 System description.         Q1: Of what type is this cost accounting system?         Job/worker order       Process cost       A         469	Activity cost	Other, please specify

Q2: How do you measure the cost of resources consumed? (Please tick the appropriate box)

Actual cost	Standard cost	Other, please specify
Q3: What standards	are used for your c	ost accounting activity? (Please tick the
appropriate box)		
Agency standards	government cost accounting standards	Generally accepted Other, please specify

Q4: What statements describe how data from this cost accounting system are primarily used?

	Describe		
	Yes	No	I do not know
To manage an activity or a program			
To measure program performance			
To determine selling price/ user fee or inter/intra agency			
service charge			
To value inventory or cost sales			
To satisfy legislative requirements			
Other, please specify			

Q5: To what extent is cost accounting data used in budget formulation? (Please tick the appropriate box)

Not used at all

Rarely used

Us

Usually the basis of estimation

Q6: What financial management systems/functions are integrated with this cost

accounting system? (Please check all that apply)

General ledger	
Program accounting	
Payroll	
Inventory control	
Property, plant and equipment	
Other,	
Not applicable	

No	Monthly Quarterly Semi-annually Annually Other, please specify
Q8: If you cons	sider that changes is needed with the method of product costing utilised
in your compan	y, Please indicate your reasons:
	······
3.3 Cost recogn	nition and measurement.
Q1: What is the	e product/service cost measurement practices in use at your company?
(Please tick the	appropriate box)
Full absorption	Activity-based Direct costing Marginal costing Other, please
costing model (FAC)	costing model (MC) specify (ABC)
	· · ·
Q2: What depre	eciation methods are used in your cost accounting system?
Straight line	Accelerated Usage related None Other, please specify
Q3: How does t	he system treat gains/losses upon disposal of assets?
Add to basis of new asset	Adjustment to Treated as Treated as Other, please Not applicable
	income or administrative
,	expense expense
	ir overall cost accounting system have a formalized method of
distinguishing h	n overall cost accounting system have a formalised method of
II yes, now are	iney distinguished and measured?
•••••	
05. D	water have one Contraction the interval 1.

If yes, how are these cost al	located to overhead cost pools	or final cost of	ojects?
		•••••••••••••••••••••••••••••••••••••••	
Q6: Do you have operationa	al overhead cost pools in your o	cost accounting	g system?
Yes	No	I do no	t know
If yes, how are these costs a	llocated?	L]	
			•••••
Q7: Do you have General as	nd Administrative cost pools in	your cost acco	ounting
system?			
Yes	No	I do no	t know
If yes, how are these costs a	llocated?	LJ	
•••••			
		•••••	••••••
			••••••
Q8: Do you receive allocation	ons of general and administrati	ve expenses fro	om other
organisational units?			
Yes	No	I do no	t know
3.4 Reporting			
Q1: For what purposes are r	eports generated from this cost	accounting sy	stem (Please
check all that apply).			
Financial reporting			
Developing prices			
Inventory control			

Financial reporting	
Developing prices	
Inventory control	
Managerial control	
Budget preparation	
Budget execution	
Other place specify	
Outer, prease specify	

Q2: Who uses the reports that are generated from the cost accounting system? (Please
check all that apply).
Program managers financial managers Budget analysts Operating managers Other, please specify
Q3: How many different reports are generated from this system?
04: How often do the recipients of these reports give you foodback on management
Q4. How one do the recipients of these reports give you recuback on management
decisions made based on data in the reports? (Please tick the appropriate box)
Always often usually rarely Never
Q5: To what extent does the present product costing model help you to manage cost within the company/division? (Please tick the appropriate box)
To a very great To a great extent To a moderate To some extent To no extent Extent
Q6: Could you please indicate the extent to which you rely on the information provided
under the present product costing model for managerial decision making? (Please tick
the appropriate box)
Totally reliable (no Reliable Reliable cosme to Unreliable Totally unreliable Not error) error
Q7: To what extent, if any, do you fell your cost accounting system provides all the
information that users of these reports need? (Please tick the appropriate box)
To a very great To a great extent To a moderate To some extent To not extent To not extent

Q8: The reporting system provides a series of data relating to cost information. The following statements describe different uses of cost information. In your opinion, please indicate the extent to which you agree or disagree with each of the statements as indication of the importance of the role of cost information play in your company. (Please tick the most appropriate answer for each statements, in the table below).

	Significance of cost information use at your											
			com	pany								
Cost information role	extrem	very	Import	Not so	Not	I do						
	ely	import	ant	import	import	not						
	import	ant		ant	ant	know						
	ant											
Uses for system maintenance and effici	ency.		<u> </u>									
To comply with procedures of reporting												
Basis for audit												
Inventory Evaluation												
Planning of period costs (budgeting)												
Perceived importance for meeting												
budgets												
To provide a means of communication												
To achieve target of production												
To achieve large profit												
Management and control of period costs												
Role of standard cost in assessing												
performance												
Measuring efficiency of production												
performance												
Performance evaluation of management												
Identification of necessary corrective												
measures												
To achieve certain financial												
measurement												
To enhance cost reduction programmes												

	Significance of cost information use at your										
			com	pany							
Cost information role	extrem	very	Import	Not so	Not	I do					
	ely	import	ant	import	import	not					
· · ·	import	ant		ant	ant	know					
	ant										
				I	1 <u>.</u>						
Uses for system optimisation.											
Importance of flexible budget for			. <u>_</u>	· · · · · · · · · · · · · · · · · · ·							
manufacturing cost control											
Variance analysis and exception											
reporting											
Planned and actual product cost											
Motivating efficiency improvement											
To enhance equipment productivity											
To enhance labour productivity											
To motivate people to do better											
To stimulate control of conversion cost											
To emphasis throughput rate											
To maintain certain profit margins											
Measuring of efficiency and capacity											
utilisation											
Make or buy decisions											
Product mix decisions											
Personal development and workforce											
improvement											
Development of better manufacturing											
methods											
Importance of product cost as an input											
to pricing decision											
Transfer prices for internal services											
Importance of marketing cost analysis											

	Significance of cost information use at your										
			com	pany							
Cost information role	extrem	very	Import	Not so	Not	I do					
	ely	import	ant	import	import	not					
	import	ant		ant	ant	know					
	ant		İ								
			·		<u> </u>						
Uses for system adaptation											
To forecast the future											
Evaluation of investment projects											
Decisions of company strategy											
Improvement of product characteristics	-										
Product quality analysis											
Planned and actual customer											
profitability.											
Importance of competitor cost analysis											
Market share analysis											
Market growth analysis											
New product development											
Market development											
Market and competition details											
Importance of product configuration											
Importance of technology configuration											

Any comment:

.....

Q9: In your view, How would you evaluate the cost practices' orientation at your company? (Please tick the most appropriate answer for each statement, in the table below:

	Significance of cost information use at your											
	company											
Cost information Orientation	extrem	very	Import	Not so	Not	I do						
	ely	import	ant	import	import	not						
	import	ant		ant	ant	know						
	ant											
Long run concern with cost and revenue												
information												
Long run concern with little												
consideration for cost information												
Long run corporate plan guidelines												
Short run concern with target costing	· · · · · · · · ·											
Short run concern with flexible costing												
Short run concern and efficiency												
information												
Short run concern and conversion cost												

If you have any comment, please write these comments below:

.....

.....

Q10: What types of information generated by the formal reporting system in your company helps in making decisions? (Please tick the most appropriate box in this table)

Degree of	Extremely	Very	Important	Not so	Not	I do not
Importance	important	important		important	important	know
Quantitative Data						
Qualitative Data						

Q11: Do you have any additional comments about the existing cost practices in your company?

······

#### Section 1:

The followings are three perspectives of cost management policies which are based upon time horizons, long, medium and short run. First, please read the definition of each model and then answer the related questions if that model of cost management is used in your company.

#### Section A: The strategic cost management policy (SCMP).

Definition	Q1: Does your company have a SCMP model?									
"The long term is focused through the notion	Please tick the appropriate box									
of strategic cost management policy (SCMP) is	Yes No I do not know									
a systemic view that refers to the way in which										
the enterprise "cost structure" is positioned in	If your company has a SCMP model. Please									
terms of markets and competition".	answer the following questions, otherwise go to									
This is partly explored in terms of cost	section B.									
structure, and partly in terms of estimates of										
long run marginal cost and whether long-run	Q2: Please specify the period horizons which are									
marginal cost can be affected by the logistic	used in the SCMP model.									
system, organisational, technological and	Years         1         2         3         4         5         6         7         8         9         10									
market arrangements; e.g., understanding of										
marginal costs imposed by particular strategic	Q3: In what ways does your SCMP model differ									
moves.	from above definition.									
	Not at all. There are differences.									
If yes. What are they?										

.....

.....

# Section B: The Economic optimisation cost management policy (ECMP).

.

Definition	Q1: Does your company have a ECMP model?										
" The medium term is approached from a view	Please tick the appropriate box										
of an economic optimisation cost management	Yes No I do not know										
policy (ECMP): assuming that technology.	If your company has a ECMP model. Please										
logistic system, raw material, production and	answer the following questions, otherwise go to										
distribution arrangements. organisational and	section C.										
market arrangements are given then".	Q2: Please specify the period horizons which are										
ECMP refers to the way of optimising the	used in the ECMP model										
activity or product mix to ensure an optimal	Years         1         2         3         4         5         6         7         8         9         10										
short term performance of the enterprise. This											
would be based upon attention to marginal	Q3: In what ways does your ECMP model differ										
revenues and marginal costs over the relevant	from above definition.										
time period.	Not at all There are differences										
If yes, what are they?											
Section C: The Accounting and control cost management policy (ACMP).											
Section C: The Accounting and control cos	t management policy (ACMP).										
Section C: The Accounting and control cos Definition	t management policy (ACMP). Q1: Does your company have a ACMP model?										
Section C: The Accounting and control cos Definition "The short term is approached via the control	t management policy (ACMP). Q1: Does your company have a ACMP model? Please tick the appropriate box										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management	t management policy (ACMP). Q1: Does your company have a ACMP model? Please tick the appropriate box Yes No I do not know										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of	t management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of accounting control, process of budgeting and	t management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please         answer the following questions, otherwise go to										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of accounting control, process of budgeting and reporting. The cost here would probably be	t management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please         answer the following questions, otherwise go to         section 2										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of accounting control, process of budgeting and reporting. The cost here would probably be either full absorption costing (FAC), based on	t management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please         answer the following questions, otherwise go to         section 2         Q2: Please specify the period horizons which are										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of accounting control, process of budgeting and reporting. The cost here would probably be either full absorption costing (FAC), based on tradition methods or based on activity-based	t management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please         answer the following questions, otherwise go to         section 2         Q2: Please specify the period horizons which are         used in the ACMP model										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of accounting control, process of budgeting and reporting. The cost here would probably be either full absorption costing (FAC), based on tradition methods or based on activity-based costing (ABC)".	t management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please         answer the following questions, otherwise go to         section 2         Q2: Please specify the period horizons which are         used in the ACMP model         Years       1         2       3         4       5         6       7       8       9       10										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of accounting control, process of budgeting and reporting. The cost here would probably be either full absorption costing (FAC), based on tradition methods or based on activity-based costing (ABC)".	t management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please         answer the following questions, otherwise go to         section 2         Q2: Please specify the period horizons which are         used in the ACMP model         Years       1         2       3       4       5       6       7       8       9       10										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of accounting control, process of budgeting and reporting. The cost here would probably be either full absorption costing (FAC), based on tradition methods or based on activity-based costing (ABC)".	t management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please         answer the following questions, otherwise go to         section 2         Q2: Please specify the period horizons which are         used in the ACMP model         Years       1         2       3       4       5       6       7       8       9       10         Q3: In what ways does your ACMP model differ										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of accounting control, process of budgeting and reporting. The cost here would probably be either full absorption costing (FAC), based on tradition methods or based on activity-based costing (ABC)".	t management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please         answer the following questions, otherwise go to         section 2         Q2: Please specify the period horizons which are         used in the ACMP model         Years       1         2       3       4       5       6       7       8       9       10         Q3: In what ways does your ACMP model differ       from above definition.										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of accounting control, process of budgeting and reporting. The cost here would probably be either full absorption costing (FAC), based on tradition methods or based on activity-based costing (ABC)".	t management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please         answer the following questions, otherwise go to         section 2         Q2: Please specify the period horizons which are         used in the ACMP model         Years       1         2       3       4       5       6       7       8       9       10         Q3: In what ways does your ACMP model differ       from above definition.       There are differences										
Section C: The Accounting and control cos Definition "The short term is approached via the control of efficiency and accounting cost management policy (ACMP). ACMP is the loop of accounting control, process of budgeting and reporting. The cost here would probably be either full absorption costing (FAC), based on tradition methods or based on activity-based costing (ABC)". If yes, what are they?	<b>t</b> management policy (ACMP).         Q1: Does your company have a ACMP model?         Please tick the appropriate box         Yes       No         I do not know         If your company has a ACMP model. Please         answer the following questions, otherwise go to         section 2         Q2: Please specify the period horizons which are         used in the ACMP model         Years       1         Q3: In what ways does your ACMP model differ         from above definition.         Not at all         There are differences										

.....

#### Section 2:

I am interested in the amount of capital investment expenditure (CIE) that was spent by your company over the last 5 years. Firstly, please write down in the table below the figure (amount) of capital investment expenditure that has been during the last 5 years. Secondly, please indicate how important (significant) relative to your company, under the current situation, is each item of the CIE by making the most appropriate answer in the table below.

	The amount	Perce	eived Re	elevant	Import	ance to	Your					
<b>Classification Of Capital</b>	of capital expenditure	Con	npany U	U <b>nder C</b>	Current	Condit	ions					
Investment Expenditure (CIE)	during the	Extrem	Very	Import	Not so	Not	I do not					
	last 5 years	ely	import	ant	import	import	know					
		import	ant		ant	ant						
	<b>.</b>	ant										
System maintenance Capital Expenditure.												
Major maintenance and												
replacement of facilities												
Improvement required to carry												
on the existing business												
Improvement in operational												
efficiency												
Rationalisation and cost												
reduction programmes												
Product improvements												
System Optimisation Capital Ex	penditure.											
Introduction or expansion of												
new market												
Expansion of the existing												
process												
Research and development												
investment												

Classification Of Capital	The amount of capital expenditure	Company Under Current Conditions										
Investment Expenditure (CIE)	during the last 5 years	Extrem ely import ant	Very import ant	Import ant	Not so import ant	Not import ant	I do not know					
Management and development												
training												
Marketing expenditure												
System adaptation capital expen	iditure	L	I.,,,,	I	1	L	I					
Changes in manufacturing												
process and products												
Changes in manufacturing												
process and market												
Changes in manufacturing												
process, market and products												
Changes in market and products												
Changes in organisation												
(reorganisation)												
Concentration of production in a												
single site		•				1						
Changes in distribution												
arrangements												
Changes in raw material												
arrangements												

#### **Company Environment**

The researcher is interested in your company's relationship with various sectors of the external environment. Specially he would like to elicit the degree of "Intensive Competition" and "Unpredictability" facing your company that are related to three areas of external environment: product market, factor market and legislation.

The following is a brief definition of intensive competition and unpredictability: Intensive Competition: The general level of market conditions that makes profitability difficult.

Unpredictability: The overall degree of variability from period to period that makes forecasting and planning difficult.

Q1: Please tick the most appropriate answers, in the provided shaded areas of the table below, to indicate the degree of "Intensive Competition" and "Unpredictability" facing your company over the last 5 years. In the case if you do not know, please tick the appropriate box of the table below.

			Inte	ensi	ve C	Com	peti	tion		τ	Unpredictability						
M	Warket Conditions		1	2	3	4	5	High .	I do not know	Average Averag		Average	Unpredictabil	Very unpredi ctability	I d0 not know		
Pro	oduct Market									-							
1	Market Changes	Expandi						Declinin									
	(sales).	ng market						g									
2	Competition	Low					· . · .	High									
	intensity																
3	Customers variety.	large						One			n e el						
		variety						major customer									
4	Profit margins	High						Narrow									
5	Product innovation	Well					а. 	Always									
		establish	1. <sup>1</sup> .					new									
		ed		•				products					n i Ning				
6	Nature of	Well						Always						enna di Agita dal			
	competition	establish						new									
		ed						competit									
		competit						OFS						1. A			
		ors															

			peti	Unpredictability												
M	Market Conditions		1	2	3	4	5	High	I do not know	Very pred ble	y licta	predictable	Average	Unpredictabil	Very unpre ctabil	di ity I do not know
7	Prices of products	Stable						Fluctuate prices.								
Fa a	ctor Market. Raw Material		<b>I</b>	· · ·		<u> </u>	<u>.</u>	1					I			
	and Supply.															
1	Supply	oversupp						shortage								
2	Sources of supply	plenty of sources						One major								
3	Competition between	No						Intensive			<u> </u>					
5	suppliers	competiti on						competiti on								
4	Cost of raw material	Low						High								
5	Delivery time	Short	 					Long			101 A.					
6	Nature of suppliers	Always new						Well establish								
7	Prices or raw	Stable						ed Fluctuate								
,	material	prices						prices					14 - 1 - 1			
b	Labour					1		L				1	I	<u></u>	L	, , , <b>, , , , , , , , , , , , , , , , </b>
1	Availability of skilled labour	stable						Shortage								
2	Competing employers	Few						Many								
3	Wage rates	Low						High								
4	Trade union	Not militant						Militant								
5	Labour turnover	Low						High								
6	Labour dependency	High						Low			<u>.</u>					
7	Industrial relation	Good	23 1 1 2 4 1 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1					Poor								
с	Management			1		[]						1		I	<u> </u>	L
1	Availability of	Stable		-				Shortage							· • •	
	qualified managers	Law					-	Link								
2	Salary of management	LOW						righ								

			Inte	ensi	ve C	Com	peti	tion		Unpredictability						
Ma	nrket Conditions	Low	. 1	2	3	4	5	High	I do not know	Very pred ble	y licta	predictable	Average	Unpredictabil	Very unpredi ctability	I d0 not know
3	Competition for	Low						High					1 1			
	good managers															
4	Staff turnover	Low						High								
d	Capital and Financing		- <b>L</b>	1	<b>1</b>	_	1					L	1		· · · · · ·	
1	Availability of fund	Easy to						hard to					- 19			
		obtain fund						obtain fund								
2	Sources of fund	many						One								
2	Bourtes of fund	sources						major								
2	Cost of finance	Low						source			<u>;</u>					
3	Cost of Imance	2011						111.5.1								
4	Staff turnover	Low						High								
е 1	<i>Technology</i> Availability of equipment	Stable						Shortage			·					
2	Equipment suppliers	Many						One major suppliers								
3	Competition for	No						Keen								
	equipment.	on						on								
4	Delivery of	Short		· .				Long								
	equipment										14 J					
5	Equipment failure	usual failure						Frequent failure								
6	Nature of	Simple						Complex								
	manufacturing															
	technology		ta. L													
7	Nature of system	Simple				n Status	an e Russie	Complex								
	technology															
8	Technology	Negligibl						Common								
	development	e change						change								
0	Equipmon <sup>4</sup>	Long life						Short life								
7	abaolooseree	Long me						Short me								
10	obsolescence	class						Fact								
10	Capital labour	siow				:		rast								
	substitution															

Market Conditions			Intensive Competition						Unpredictability						
		Low	1	2	3	4	5	High	I do not know	Very predicta ble	predictable	Average	Unpredictabil	Very unpredi ctability	I d0 not know
Le	Legislation														
1	Regulators in the	One			[	den e 1 - 1		Many							
	industry	major						regulator							
		regulator													
2	Number of	Small	1.00					Large							
	legislation	ľ										j			
3	Nature of legislation	Well		-			1	Frequent			1.14	<u>.</u>			
5	Nature of legislation	establish						change							
	1	ed						Ū							
4	Effect of legislation	Negligibl						Heavy							
		e		n i si Nga ta	1.1						· · ·				
		I		L		1	L	l			1	11.0			
	Overall Asses	sment				· · · · · · · · ·									

Q2: How do you rank the significance of these three sectors of external environment relative to your company in order to express their importance? Then please attach percentage weights to each sector.

	Rank of	priority of si		
Environment sectors	1	2	3	Percentage Weighting (%)
1- Product Market				
2- Factor Market				
3- Legislation				
				100%

Q3: Given the market factor sub-sectors, how would you rank the 5 sub-sectors within the factor product in order to express their priority and significance?

	Rank						
Factor product	1	2	3	4	5		
1- Raw Material and Supply							
2- Labour							
3- Management	<u></u>						
4- Capital and Financing							
5- Technology							

Thank you for completing this questionnaire.

.

#### The Interview Guide

#### **Cost practice**

#### Section 1: Control structure and responsibility accounting.

Q1: Could you please describe the organisation structure of the company?

Q2: How did the existing organisation structure develop?

Q3: Could you please describe the control structure at the company?

Q4: How many responsibility centre do your primary products have to sequentially go through, from purchases to customers?

#### Section 2: Planning and control procedures.

Q1: Could you tell me how the operational business budget is constructed in your division/business unit?

Q2: Could you please tell me how the capital budgeting is constructed in your division business unit?

#### Section 3: Cost accounting.

#### 3.1 System background.

Q1: Could you please tell me about the methods used in your company for measuring product cost? For example: Full Absorption Costing Model, Activity based Costing Model, Direct Costing Model, Variable Costing Model, Marginal Costing Model, Target Costing, Throughput Costing,...)

Q2: Were there any significant changes in the cost measurement practice system used in your company over the last five years?

*If yes,* Please tell me about; When did it happen? What was each change? Who was its champion? What is your opinion about each change? And where are others' opinion about it?

If no, Are there any plans to change the cost measurement practice? Please describe them.

#### 3.2 System description.

Q1: How many steps (levels) through which these cost information flow within your company?

Q2: Is there any role for consultants who designed your cost management in: -

- maintaining it by reviewing over a specific period of time?

- the day to day managing it?

- proposing changes on it which may be: simple changes; significant changes; complete changes of it.

#### 3.3 Cost recognition and measurement.

Q1: Could you tell me how the overhead cost is (1) managed within your company and/or (2) allocated within your company?

#### 3.4 Reporting.

Q1: What are the types of cost information generated by the formal reporting system in your company? For example, Daily plant book reports; Regular management reports; Ad hoc reports; Combination; Other?

Q2: The reporting system provides a series of data relating to cost information, How do your company take action on the basis of these reports?

# A2.2 Analysis of the Questionnaire

	Name	Job title
1	Shapan Ramadan Mohammed	Accountant (Participating Accounts Department).
2	Ali Sadek Mohammed Ali	Accountant (Participating Accounts Department).
3	Gamalat Mustafah Naseer	Accountant (Participating Accounts Department).
4	Momena Tah Mohammed Abd-	Accountant (Bills Auditing and Stamp
	El-Salam	Department).
5	Saeed Ahamed Ahamed	Accountant (Financial Affairs) in participating
		Accounts Administration.
6	Mustafah Husseen El-Kasbani	Chief of Bills Auditing Department (collection).
7	Sameer Ragab Ahamed	Chief of Bills Auditing and Stamp Department.
8	Nagwa Ahamed Abd-El-Hadi	Chief of Statistics Department.
9	Mohammed Mohammed Ahamed	Accountant (Deduction and Modifications
	Deyab	Department).
10	Adel Taha El-Hadi	Chief of Commitment (Engagement) Department
		(Balance Sheet).
11	Mohammed Abd-El-Rasheed	Chief of Loans Accounts and Documentary Credit
	Abd-El-Al	Department).
12	Emad Hesen Ebrahem	Accountant (Cost Department).
13	Batah Salah Abd-El-Mawgood	Accountant (Cost Department).
14	Mervat Abd-El-Hameed	Chief of Fixed Assets Department.
15	Zenab El-Sayed Ali Tantawy	Accountant (Cost Department).
16	Fauzia Abdo Marzouk	Accountant (Cost Department).
17	Mohammed Sayed Hassan	Chief of Central Accounts Department.
18	Magda Fareed Sad	Accountant (Cost Department).
19	Maher Ahamed Sayed	Accountant (Cost Department).
20	Mohammed Sayed Hassan	Chief of Cost Department.
21	Bader Abd-El-Shafy Abd-El-	Accountant (Documentary credit Department).
	Shafy	
22	Yasser Mahammoud Ahamed	Chief of Documentary Credit Department.

Table A2.1: The respondents of the question	nnaire
---	--------

	Name	Job title					
23	Adel El-Bagdadi	Vice Chairman for Technical Affairs.					
24	Asem Mohammed Abd-El-Aziz	Auditing (Maturities Department).					
25	Samir Mohammed Hassan	Accountant (Planned Budget Department).					
26	Marsel William Yakob	Chief of Balance Sheet Administration.					
27	Mohammed Mustafah Amin	Chief of loans and Grants Department.					
28	Hamadi Abd-El-Hakeem Sharawi	Accountant (Cost Department).					
29	Azzam Mohammed Ramdan	Chief of Final Statement of Accounts and Follow-					
		up					
30	Ekram Mohammed Hassanen	Accountant (Cost Department).					
31	Sadi Flah Sad	Accountant (Central Accounts Department).					
32	Mosad Abd-El-Wahed Showra	Accountant (Central Accounts Department).					
33	Amal Habeeb Abd-El-Gani	Accountant (Cost Department).					
34	Abd-El-Halim Attia Abd-El-	Accountant (Cost Department).					
	Halim						
35	Reda Abd-El-Menam Ahamed	Accountant (Cost Department).					
36	Khyeria Ahamed Faried	Accountant (Central Department).					
37	Medhat Abd-El-Hamid Basywni	accountant (Participating accounts Administration.					
38	Magdi Abd-Allah Zaki	Accountant of Statistics Department.					
39	Mahasen Ahamed Refaee	Accountant (Participating Accounts Department).					
40	Abd-El-Aziz Zaki Abd-El-Aziz	Accountant of Participating Accounts Department					
		(Deduction Department).					
41	Ahamed Mosad Mohammed	General Chief Administration of Government and					
		Organisations Accounts.					
42	El-Sayed Ramadan Kamal	General Chief of Inventory Administration.					
43	Rada Abd-El-Hameed Shatah	General Chief of Purchases and Contracts					
		Administration.					
44	Hamdy Mustafah Abd-El-Hameed	General Chief of Revenues Auditing					
		Administration.					
45	Hussam Hassan Abd-El-Aal	General Chief of Meters Administration.					
46	Emad Abd-El-Halim Ali	Senior Accountant (Cost Department).					
47	Ahamed Anwer	Researcher of Planning for Investment Projects.					
48	Aesha Ahamed	Chief of Budget and Follow-up Administration.					

-

	Name	Job title
49	Mohammed El-Demardash	Chief of the central administration for managerial
		affairs.
50	Hassan El-Shasfy	Chief of the central administration for stations.

			Componen	t	
	1	2	3	4	5
To comply with procedures of	.705	-8.225E-02	229	8.351E-02	142
reporting.					
Perceived importance for	.668	.242	134	122	6.349E-02
meeting budgets.					
Planning of period costs	.667	-6.567E-02	208	.452	-6.288E-02
(budgeting).					
Identification of necessary	.633	245	283	308	2.977E-02
measures.					
To achieve certain financial	.625	-8.903E-02	448	274	290
measurement.					
To enhance cost reduction	.596	296	2.017E-02	108	423
programmes.					
To achieve target of production.	.587	.360	-1.092E-02	197	.363
Measuring efficiency of	.547	-9.638E-02	.545	-6.974E-02	269
production performance.					
To achieve large profit.	.514	.435	.291	276	298
Basis for audit.	.511	.447	.273	.250	.271
Inventory evaluation.	.242	.772	260	.311	-9.951E-02
Role of standard cost in	.425	541	.349	.321	4.392E-02
assessing performance.					
To provide means of	· .391	2.811E-02	.685	310	.223
communication.					
Management and control of	.583	229	4.059E-02	.618	.198
period costs.					
Performance evaluation of	.414	286	293	316	.641
management.					

# Table A2.2: Component Matrix<sup>a</sup> for Uses of Cost Information for Efficiency

Extraction Method: Principal Component Analysis.

<sup>&</sup>lt;sup>a</sup> 5 components extracted

Table A2.2a: Pattern Matri	ix <sup>a</sup> for Use:	s of Cost In	formation	for Efficien	cy
	•		Component	t	
	1	2	3	4	5
To achieve certain financial	.863	6.278E-02	-9.695E-02	-8.508E-02	.120
measurement.					
To enhance cost reduction	.680	179	.222	.167	168
programmes.					
Identification of necessary	.630	106	3.471E-02	3.135E-02	.388

.910

.574

.379

-9.526E-02

-9.090E-02

.373

.127

-.357

.283

-.173

.433

-2.184E-02

-.168

.359

.172

.856

.684

.603

-6.251E-02

.242

-.161

-6.672E-02

.292

.312

9.880E-03

.325

4.428E-02

-8.369E-04

.204

-.233

.890

.704

.631

7.170E-02

-2.393E-02

7.096E-02

-.167

.135

.261

.217

-.220

-.164

7.991E-02

-2.385E-02

-1.793E-02

.883

.488

.561

2.571E-02

-.239

.370

-.165

.234

.322

-2.677E-02

1.886E-02

.334

.145

7.760E-02

for

of

of

in

costs

of

**Extraction Method: Principal Component Analysis.** 

period

evaluation

Rotation Method: Oblimin with Kaiser Normalization.

To achieve target of production.

measures.

reporting.

Perceived

Measuring

period costs.

Planning

(budgeting).

Performance

management.

To

Inventory evaluation.

Basis for audit.

meeting budgets.

communication.

provide

production performance.

To achieve large profit.

Management and control of

Role of standard cost

of

assessing performance.

To comply with procedures of

importance

means

efficiency

<sup>&</sup>lt;sup>a</sup> Rotation converged in 14 iterations.

			Componen	t	
	1	2	3	4	5
To achieve certain financial	.852	.190	7.077E-02	.138	.250
measurement.					
To enhance cost reduction	.710	-5.386E-02	.347	.354	-2.870E-02
programmes.					
Identification of necessary	.693	3.609E-02	.189	.243	.490
measures.					
To comply with procedures of	.675	.276	.175	.473	.219
reporting.				·	
Perceived importance for	.516	.486	.324	.233	.385
meeting budgets.					
Inventory evaluation.	.108	.878	-6.958E-02	2.325E-02	-8.647E-02
Basis for audit.	2.517E-02	.616	.462	.392	.231
To provide means of	1.983E-02	3.649E-03	.835	.150	.267
communication.					
Measuring efficiency of	367	1.830E-02	.736	.369	-9.451e-02
production performance.					
To achieve large profit.	.406	.462	.645	-2.409E-02	-3.989E-02
Management and control of	.224	.184	.135	.889	.194
period costs.					
Role of standard cost in	.188	280	.342	.731	5.639E-02
assessing performance.					
Planning of period costs	.507	.356	6.197E-02	.702	.130
(budgeting).					
Performance evaluation of	.272	-6.421E-02	4.523E-02	.195	.892
management.					
To achieve target of production.	.276	.528	.404	.146	.572

# Table A2.2b: Structure Matrix<sup>a</sup> for Uses of Cost Information for Efficiency

Extraction Method: Principal Component Analysis.

Rotation Method: Oblimin with Kaiser Normalization.

<sup>&</sup>lt;sup>a</sup> Rotation converged in 14 iterations.

# Table A2.3: Rotated Component Matrix<sup>a</sup> for Uses of Cost Information for Optimisation

.

	Component						
	1	2	3	4	5		
Development of better manufacturing	.770	-1.926E-02	.201	.186	.367		
methods.							
Personal development and workforce	.742	.502	9.451E-02	119	6.727E-02		
improvement							
To motivate people to do better	.618	.225	4.571E-02	.495	.146		
To enhance equipment productivity	.556	.363	-8.440E-02	.203	.309		
Importance of marketing cost analysis	.507	.264	.387	.118	.195		
Transfer prices for internal services	.420	129	.333	.266	188		
Determine the economic production	.372	.762	7.213E-02	8.225E-02	3.263E-02		
amount							
To maintain certain profit margins	9.662E-02	.671	.225	.556	.154		
To measure of efficiency and capacity	.159	.635	-1.066E-02	4.955E-02	.426		
utilisation							
Importance of product cost an input to	-3.151E-02	.596	.438	.244	.232		
pricing decision							
Variance analysis and exception	9.926E-02	-2.174E-02	.844	9.237E-02	7.500E-02		
reporting							
Planned and actual product cost	2.281E-02	.200	.745	7.031E-02	-2.369E-02		
Importance of flexible budget	.154	127	.547	.363	.427		
To stimulate control of conversion cost	.247	.196	.496	.271	-5.716E-03		
Motivating efficiency improvement	9.044E-02	2.886E-02	.162	.826	.103		
To emphasis throughput rate	.221	.274	.241	.771	3.378E-02		
Product mix decision	.1`30	.231	.199	-8.981E-02	.790		
Make or buy decision	.243	.198	127	.270	.767		

Extraction method: Principal Component Analysis.

Rotation Method: Varimax with Kaiser Normalisation.

<sup>&</sup>lt;sup>a</sup> Rotation converged in 11 iterations.

	Component		
	1	2	3
Importance of technology configuration.	.861	110	.529
Improvement of product characteristics.	.874	349	.532
Product quality analysis.	.822	-6.088E-02	.628
Decisions of company strategy.	.808	4.481E-02	.398
Market growth analysis.	.747	-8.667E-02	.566
To forecast analysis.	.718	-2.435E-02	.132
Market development.	.715	.230	.558
Importance of competitors cost analysis.	.540	657	.498
Evaluation of investment projects.	.531	.606	.341
Importance of product configuration.	.434	4.864E-02	.862
Importance of product improvement.	.463	146	.838
New product development.	.325	259	.791
Planned and actual customer profitability.	.597	505	.690
Market share analysis.	.470	585	.623

Table A2.4: Structure Matrix for Uses of Cost Information for Strategy

**Extraction Method: Principal Component Analysis.** 

Rotation Method: Oblimin with Kaiser Normalization.

	Component			
	1	2	3	4
Market changes (sales)	.486	156	516	338
Competition intensity	5.729E-02	-5.122E-02	-6.143E-02	.913
Customer variety	.136	839	8.325E-02	.203
Profit margins	8.032E-02	2.029E-02	.870	105
product innovation	.776	228	.208	116
Nature of competition	9.488E-02	829	153	172
Prices of product	.790	-2.310E-02	135	.132

Table A2.5: Structure Matrix for the P (Un) of Product market

**Extraction Method: Principal Component Analysis.** 

Rotation Method: Oblimin with Kaiser Normalization.

	Component			
	1	2	3	
Supply	141	.133	861	
Sources of supply	805	-4.801E-02	.121	
Competition between suppliers	.648	.430	-2.903E-02	
Cost of raw material	.215	.787	-2.505E-02	
Delivery time	.181	-1.319E-03	822	
Nature of suppliers	705	.247	-9.529E-02	
Prices of raw material	292	.795	133	

# Table A2.6: Structure Matrix for the P (CI) of Raw Material and Supply

Extraction Method: Principal Component Analysis.

Rotation Method: Oblimin with Kaiser Normalization.

## Table A2.6a: Structure Matrix for the P (Un) of Raw Material and Supply

· · ·	Component		
	1	2	
Supply	8.756E-02	.817	
Sources of supply	.720	.176	
Competition between suppliers	.907	.242	
Cost of raw material	.585	.360	
Delivery time	.417	.721	
Nature of suppliers	.832	.199	
Prices of raw material	.886	.140	

**Extraction Method: Principal Component Analysis.** 

Rotation Method: Oblimin with Kaiser Normalization.

	Component		
	1	2	3
Availability of skilled labour	.663	.254	.190
Competing employers	7.429E-02	.176	.785
Wages rate	.477	.379	.223
Trade union	365	422	.566
Labour turnover	750	.108	.246
Labour dependency	7.977E-02	830	-8.372E-02
The relationship between water	.655	275	203
organisations			

Table A2.7: Structure Matrix for P (CI) of Labour

**Extraction Method: Principal Component Analysis.** 

Rotation Method: Oblimin with Kaiser Normalization.

Table A2.7a:	Structure	Matrix fo	or P	(Un)	of Labour
	Suduit	TIMEE CT TV TC		(01)	UI LADUAL

· · · · · · · · · · · · · · · · · · ·	Component		
	1	2	
Availability of skilled labour	.706	.197	
Competing employers	.129	882	
Wages rate	.733	398	
Trade union	.618	514	
Labour turnover	.666	272	
Labour dependency	.417	754	
The relationship between water organisations	.541	144	

Extraction Method: Principal Component Analysis.

Rotation Method: Oblimin with Kaiser Normalization.

	Component		
	1	2	
Availability of qualified managers	.822	-4.500E-02	
Salary of management	592	.624	
Competition for good managers	.225	.852	
Staff turnover	.692	.264	

 Table A2.8: Structure Matrix for P (CI) of Management

Extraction Method: Principal Component Analysis.

Rotation Method: Oblimin with Kaiser Normalization.

## Table A2.8a: Structure Matrix for P (Un) of Management

	Component
	1
Availability of qualified managers	.803
Salary of management	.769
Competition for good managers	.613
Staff turnover	.748

**Extraction Method: Principal Component Analysis.** 

Rotation Method: Oblimin with Kaiser Normalization.

•

	Com	Component		
	1	2		
Availability of fund	275	.814		
Sources of fund	.486	.641		
Cost of finance	.696	5.394E-02		
Staff turnover	625	.116		

## Table A2.9: Structure Matrix for P (CI) of Capital and Financing

**Extraction Method: Principal Component Analysis.** 

Rotation Method: Oblimin with Kaiser Normalization.

#### Table A2.9a: Structure Matrix for P (Un) of Capital and Financing

	Component
	1
Availability of fund	.867
Sources of fund	.876
Cost of finance	.431
Staff turnover	.781

Extraction Method: Principal Component Analysis.

Rotation Method: Oblimin with Kaiser Normalization.

	Common and			
	Component			
	1 .	2	3	
Availability of equipment	.480	.338	.143	
Equipment suppliers	.603	151	.134	
Competition for equipment	6.089E-02	-6.547E-02	.694	
Delivery of equipment	.734	-141	194	
Equipment failure	.675	.303	.163	
Nature of manufacturing technology	.712	118	.201	
Nature of system technology	.474	6.228E-02	.646	
Technology development	8.079E-03	.184	.676	
Equipment obsolescence	-4.624E-02	.884	1.296E-02	
capital labour substitution	-4.290E-02	.882	.118	

 Table A2.10: Structure Matrix for P (CI) of Technology

۱

.

**Extraction Method: Principal Component Analysis.** 

Rotation Method: Oblimin with Kaiser Normalization.

	Component		
	1	2	3
Availability of equipment	-4.082E-03	.467	.724
Equipment suppliers	.172	.691	.268
Competition for equipment	.782	.383	3.438E-02
Delivery of equipment	.291	.832	2.885E-02
Equipment failure	.364	.514	-4.703E-02
Nature of manufacturing technology	.443	.782	2.165E-02
Nature of system technology	.857	.392	8.110E-02
Technology development	.681	.254	.335
Equipment obsolescence	.281	-2.804E-02	.881
capital labour substitution	.828	.225	6.204E-03

#### Table A2.10a: Structure Matrix for P (Un) of Technology

**Extraction Method: Principal Component Analysis.** 

Rotation Method: Oblimin with Kaiser Normalization.
	Comp	oonent
	1	2
Regulators in the industry	.886	195
Number of legislation	.852	220
Nature of legislation	.158	.912
Effect of legislation	.598	.355

## Table A2.11: Structure Matrix for P (CI) of Legislation

**Extraction Method: Principal Component Analysis.** 

Rotation Method: Oblimin with Kaiser Normalization.

## Component1Regulators in the industry.878Number of legislation.899Nature of legislation.630Effect of legislation.854

## Table A2.11a:Structure Matrix for P (Un) of legislation

**Extraction Method: Principal Component Analysis.** 

Rotation Method: Oblimin with Kaiser Normalizatio

Table A2.12: Pearson correlations -Level Matrixes for Uses of Cost Information for Efficiency

_								I														
To	achieve	certain	financial	measure	ment			.528**	.067	.186	.340**	.432**	002	.244*	.263*	.144	.144	.235	.304*	.579**	1.000	.410**
Identific	ation of	necessar	y	correctiv	ð	measure	s	.351**	.093	.025	.296*	.278*	.115	.396**	.245*	.302*	.243*	.230	429**	1.000	.579**	.424**
Perform	ance	evaluati	on of	manage	ment			.225	.137	186	.143	.371**	.146	.304*	031	.258*	611.	028	1.000	.429**	.304*	.167
Measuri	ßu	efficienc	y of	producti	uo	perform	ance	.263*	.238*	054	.249*	.265*	.448**	.261*	.375**	.283*	.357**	1.000	028	.230	.235	.366*8
Role of	standard	cost in	assessin	ະຄ	perform	ance		.231	.132	235	.309*	.088	.247*	.047	.064	.513**	1.000	.357**	6II.	.243*	.144	.250*
Manage	ment	and	control	period	costs	-		.350**	.387**	.154	.568**	.228	.018	.161	.070	1.000	.514**	.283*	.258*	.302*	.144	.284*
To	achieve	large	profit					.205	.359**	.287*	.124	.442**	.333**	.322*	1.000	.070	.064	.375**	031	.245*	.263*	.304*
To	achieve	target of	producti	uo				.413**	.377**	.317*	.290*	.354**	.329**	1.000	.322*	.161	.047	.261*	.304*	.396**	.244*	.076
To	provide	a means	of	commun	ication			.112	.318*	135	190.	.231	1.000	.329**	.333**	· 180.	.247*	.448**	.146	.115	002	.192
Perceive	p	importa	nt for	meeting	budgets		-	.362**	.309*	.262*	.506**	1.000	.231	.354**	.442**	.228	.088	.265*	.371**	.278*	.432**	.261*
Planning	of	period	costs					.544**	.216	.233	1.000	.506**	.061	.290*	.124	.568**	.309*	.249*	.143	.296*	.340**	.340**
Inventor	У	Evaluati	uo					.113	.401**	1.000		.262*	135	317*	.287*	.154	235	054	186	.025	.168	063
basis for	audit							.279*	1.000	.401**	.216	.309*	.318*	.377**	.359**	.387**	.132	.2388	.137	.093	.067	.168
To	comply	with	procedur	es of	reportin	50		1.000	.279*	.113	.544**	.362**	.112	.413**	.205	.350**	.231	.263*	.225	.351**	.528**	.477**
		Correlation						To comply with procedures of reporting	basis for audit	Inventory Evaluation	Planning of period costs	Perceived important for meeting budgets	To provide a means of communication	To achieve target of production	To achieve large profit	Management and control period costs	Role of standard cost in assessing performance	Measuring efficiency of production performance	Performance evaluation of management	Identification of necessary corrective measures	To achieve certain financial measurement	To enhance cost reduction programmes

\*. Correlation is significant at the 0.05 level (1-tailed). \*\*. Correlation is significant at the 0.01 level (1-tailed).

or Efficiency
Information f
CUSES OF COST
Matrixes for
ations -Leve
earson corre
evels of the P
ignificance le
v2.12a: The s
Table A

			_							1					1	-							
To	achiore	acilieve	certain	financial	measure	ment			000.	.323	860.	.008	.001	.496	.044	.033	.160	.160	.050	.016	000.		.002
Identific	ation of	auon oi	necessar	У	correctiv	Ð	measure	s	.006	.260	.432	.018	.025	.214	.002	.043	.017	.045	.054	100.		000.	.001
Perform		ance	evaluati	on of	manage	ment			.058	.172	860.	.161	.004	.156	.016	414	.035	.206	.425	•	.001	.016	.123
Measuri		۲	efficienc	y of	producti	uo	perform	ance	0.33	.048	.354	.041	.031	.001	.034	.004	.023	.006		.425	.054	.050	.004
Role of	ctandard	DIADUAL U	cost in	assessin	50	perform	ance		.053	.180	.050	.015	.273	.042	.372	.329	000		.006	.206	.045	.160	.040
Manage	t cont		and	control	period	costs			900.	.003	.142	000.	.056	.288	.132	.315	•	000.	.023	.035	.017	.160	.023
To	achiana	מרוונגב	large	profit					.076	.005	.022	.196	.001	600.	.011	•	.315	.329	.004	.414	.043	.033	.016
To	achieve		target of	producti	on				.001	.003	.012	.021	900.	.010	•	.011	.132	.372	.034	.016	.002	.044	.300
To	nrouida	biovide	a means	of	commun	ication			.220	.012	.174	.337	.053		.010	600.	.288	.042	.001	.156	.214	.496	.091
Perceive			пропа	nt for	meeting	budgets			.005	.015	.033	000.		.053	.006	.011	.056	.273	.031	.004	.025	.001	.033
Planning	ų	5	period	costs					000.	.066	.015	•	000	.337	.012	.196	000.	.015	.041	.161	.018	.008	.008
Inventor	;	y r	Evaluati	uo					.217	.002		.051	.033	.174	.012	.022	.142	.050	.354	860.	.432	860.	.333
basis for	andit	aunt							.025		.002	990.	.015	.012	.003	.005	.003	.180	.048	.172	.260	.323	.122
To	comply	unint.	MIN	procedur	es of	reportin .	co		•	.025	.217	000	.005	.220	100.	.076	900.	.053	.033	.058	900.	000	000.
					Sig. (1-tailed)				To comply with procedures of reporting	basis for audit	Inventory Evaluation	Planning of period costs	Perceived important for meeting budgets	To provide a means of communication	To achieve target of production	To achieve large profit	Management and control period costs	Role of standard cost in assessing performance	Measuring efficiency of production performance	Performance evaluation of management	Identification of necessary corrective measures	To achieve certain financial measurement	To enhance cost reduction programmes

ptimisation
for O
rmation 1
t Info
f Cost
Jses of
s for l
Matrixes
_evel
correlations -]
Pearson
le A2.13:
Tab

	Importan	Variance	Planned	Motivati	To	To	To	To	To	Mcasuri	Determi	Make or	Product	Personal	Develop	Importan	Transfer
	ce of	analysis	and	ng	enhance	motivate	stimulate	emphasis	maintain	ng of	ne the	buy	mix	develop	ment of	ce of	prices
	flexible	ķ	actual	efficienc	equipme	people to	control	throughp	certain	cfficienc	economi	decision	decision	ઝ	better	product	for
Correlation	budget	exceptio	product	у.	Ħ	do better	J	ut rate	profit	y and				workforc	manufact	costs in	internal
		u	cost	improve	producti		conversi		margins	capacity	producti			U	uring	pricing	services
		reporting		ment	vity	-	on cost				uo			improve	methods	.,	
Importance of flexible budget	1,000	.527**	.308*	.407**	.245*	317*	.410**	.404**	.288*	.194	.151	.332**	.303*	.146	.474**	.378**	.184
Variance analysis and exception reporting	.527**	1.000	.520**	.231	.036	.251*	.281*	.246*	.314*	.156	.073	040	.147	.087	.280*	.333**	-295*
Planned and actual product cost	.308*	.520**	1.000	.220	.159	.177	.367**	.317*	.297*	.078	.200	.041	203	.164	116	-311*	.228
Motivating efficiency improvement	.407**	.231	.220	1.000	.266*	.402**	.356**	.606**	.437*8	.201	.170	.260*	.102	.083	.258*	.252*	.274*
To enhance equipment productivity	.245*	.036	.159	.266*	1.000	.557**	861.	.330**	.380**	.436**	.443**	.456**	:301*	.512**	.466**	.290*	080.
To motivate people to do better	.317*	.251*	.177	.402**	.557**	1.000	.296*	.530**	**792.	.370**	.422**	.456**	161.	.480**	.573**	.272*	.293*
To stimulate control of conversion cost	.410**	.281*	.367**	.356**	861.	.296*	1.000	.375**	.339**	160	.327*	151.	.156	:311*	.330**	371**	.163
To emphasis throughput rate	.404**	.246*	.317*	.606**	.330**	.530**	.375**	1.000	.700**	.179	.353**	.264*	.161	.266*	.482**	.483**	.267*
To maintain certain profit margins	.288*	.314*	.297*	.437**	.380**		.339**	.700**	1.000	483**	.582**	411**	.281*	.322*	.300*	.672**	.127
Measuring of efficiency and capacity utilisation	.194	.156	.978	.201	.436**	.370**	160	.179	.483**	1.000	.485**	.453**	.337**	.431**	.276*	.402**	052
Determine the economic production amount	.151	.073	.200	.170	.443**	422**	.327*	.353**	.582**	.485**	1.000	.315*	.318*	.640**	.274*	.399**	.176
Make or buy decision	.332**	-,040	.041	.260*	.456**	.456**	151.	.264*	.411**	.453**	.315*	1.000	**909.	.257*	.386**	.257*	.114
Product mix decision	.303*	.147	.203	.102	.301*	161.	.156	.161	.281*	.337**	.318*	**909.	1.000	.312*	.417**	.402**	.041
Personal development and workforce improvement	.146	.087	.164	.083	.512**	.480**	.311*	.266*	.322*	.431**	.640*8	.257*	.312*	1.000	.625**	.378**	.188
Development of better manufacturing methods	.474**	.280*	.116	.258*	.466**	.573**	.330**	.428**	.300*	.276*	.274*	.386**	.417**	.625**	1.000	.237*	.242*
Importance of product costs as an input to pricing decision	.378**	.333**	311*	.252*	.290*	.272*	.371**	.483**	.627**	.402**	**065.	.257*	.402**	.378**	.237*	1.000	.211
Transfer prices for internal services	.184	.295*	.228	.274*	080	.293*	.163	.267*	.127	052	.176	.114	.041	.188	.242*	211	1.000
Importance of marketing cost analysis	.276*	.441**	.300*	.249*	.514**	.452**	.371**	.360**	.499**	.368**	.337**	.338**	.286*	.428**	.500**	.284*	.209
*. Correlation is significant at the 0.05 levels	vel (1-f	ailed)															

Correlation is significant at the 0.05 level (1-tailed).

\*\*. Correlation is significant at the 0.01 level (1-tailed).

n
ţ
isa
H
pti
õ
20
Ę
10I
ati
E.
<u>6</u>
In
st]
õ
f
0
ses
Ď
0r
S.
Xe
ET.
[a]
Σ
vel
ē
-
ns
tio
la
Ĕ
5
a
<b>S</b> 0
ar
Pe
Je
ft
ō
els
ē
el
nc
ca
ij
ng
SI
he
Ε
a
13
Ľ.
e A
pl
_ಡ

	Іпропап	Variance	Planned	Motivati	To	To	To	To	To	Measuri	Determi	Make or	Product	Personal	Develop	Importan	Transfer
	ce of	analysis	and	Bu	enhance	motivate	stimulate	emphasis	maintain	ng of	ne the	'nq	mix	develop	ment of	ce of	prices
t (1_tailad)	flexible	ع	actual	efficienc	cquipme	people to	control	throughp	certain	efficienc	economi	decision	dccision	শ্ব	better	product	for
(nalial)	budget	exceptio	product	ж.	۲	do better	ы. Ч	ut rate	profit	y and				workforc	manufact	costs in	internal
		n reporting	COSI	improv.c ment	producti vity		conversi		marguns	capacity	producti on			e improve	uring methods	pricing	services
ble budget		000.	.015	.002	.043	.012	.002	.002	.021	.089	.147	600.	.016	.156	.000	.003	.101
nd exception reporting	000.		000.	.054	.403	.039	.024	.042	.013	.139	.307	.391	.154	.274	.024	600.	.019
product cost	.015	.000		.063	.135	.109	.004	.012	.018	.295	.082	.389	.079	.128	.212	.014	.056
icy improvement	.002	.054	.063		.031	.002	900.	000	.001	.081	.119	.034	.241	.283	.035	.039	.027
nent productivity	.043	.403	.135	.031		000.	.085	.010	.003	100.	.001	000.	.017	000.	000.	.021	.291
e to do better	.012	.039	.109	.002	000.		.018	000.	000.	.004	100.	000.	.092	000.	000.	.028	610.
ol of conversion cost	.002	.024	.004	900.	.085	.018		.004	.008	.265	.010	.147	.140	.014	.010	.004	.128
ghput rate	.002	.042	.012	000.	.010	000.	.004		000.	.107	900.	.032	.131	.031	000.	000.	.031
n profit margins	.021	.013	.018	.001	.003	000.	.008	000.		000.	900.	.032	.024	.011	.017	000.	.190
iency and capacity utilisation	680.	.139	.295	180.	100.	.004	.265	.107	000.		000.	000.	800.	.001	.026	.002	.360
nomic production amount	.147	.307	.082	.119	.001	100.	.010	.006	000.	000.		.013	.012	.000	.027	.002	111.
ion	600.	.391	.389	.034	000.	000.	.147	.032	.002	000.	.013		000.	.036	.003	.036	.216
ion	.016	.154	.079	.241	.017	.092	.140	.131	.024	.008	.012	000.		.014	.00	.002	.389
nent and workforce improvement	.156	.274	.128	.283	000.	000.	.014	.031	.011	.001	000.	.036	.014		000.	.003	.095
etter manufacturing methods	.000	.024	.212	.035	000.	000.	.010	000.	.017	.026	.027	.003	.001	000.		.049	.045
duct costs as an input to pricing	.003	600.	.014	.039	.021	.028	.004	000.	000.	.002	.002	.036	.002	.003	.049		.071
internal services	.101	.019	.056	.027	.291	.019	.128	.031	.190	.360	HI.	.216	.389	.095	.045	.071	
keting cost analysis	.026	.001	.017	.041	.001	000	.004	.005	000.	.004	.008	.008	.022	.001	000.	.023	.073
												I				ļ	1

Strategy
for
Information
Cost
s of (
Use
s for
atrixes
rel M:
-Lev
correlations
Pearson
2.14:
able A
Ξ

-		تد 						· .	<u> </u>		<u> </u>		•	•	<u> </u>	•	•			
Impote	nce of	product	improv	ement			.219	.224	.407**	.516**	.526**	.653**	.287*	.478**	.571**	.579**	.328**	.653**	1.000	.541**
Import	ance of	product	configu	ration		_	.267*	.334**	.368**	.415**	.507**	.484**	.336**	.534**	.269*	.569**	.645**	1.000	.653**	.465**
Market	develo	pment					.319*	.470**	.490**	.566**	**669'	.328**	.335**	.359**	.598**	.232	1.000	.645**	.328**	.642**
New	product	develo	pment				.168	.245*	.292*	.380**	.375**	.500**	.598**	.393**	.500**	1.000	.232	.569**	.579**	.319*
Market	growth	analysi	S				.293*	.346**	.561**	.675**	.704**	.540**	.462**	.287*	1.000	.500**	.598**	.269*	.571**	.637**
Market	share	analysi	S				.226	.021	.375**	.565**	.512**	.659**	.620**	1.000	.287*	.393**	.359**	.534**	.478**	.449**
Import	ance of	compet	itors	cost	analysi	S	.327*	.028	.399**	.579**	.442**	.598**	1.000	.620**	.462**	.598**	.335**	.336**	.287*	.474**
Planne	d and	actual	custom	er	profita	bility	.308*	.174	.478**	.692**	.567**	1.000	.598**	.659**	.540**	.500**	.328**	.484**	.653**	.532**
Produc	t	quality	analysi	S			.364**	.421**	.526**	.756**	1.000	.567**	.442**	.512**	.704**	.375**	**669'	.507**	.526**	<b>**</b> 00**
Improv	ement	of	product	charact	eristics		.537**	.249*	.582**	1.000	.756**	.692**	.579**	.565**	.675**	.380**	.566**	.415**	.516**	.714**
Decisio	ns of	compa	ny	strateg	y		.585**	.540**	1.000	.582**	.526**	.478**	.399**	.375**	.561**	.292*	.490**	.368**	.407**	.589**
Evaluat	ion of	invest	ment	project	s		.369**	1.000	.540**	.249*	.421**	.174	.028	.021	.346**	.245*	.470**	.334**	.224	.326*
To	forecas	t the	future				1.000	.369**	.585**	.537**	.364**	.308*	.327*	.226	.293*	.168	.319*	.267*	.219	.541**
			Correlation				To forecast the future	Evaluation of investment projects	Decisions of company strategy	Improvement of product characteristics	Product quality analysis	Planned and actual customer profitability	Importance of competitors cost analysis	Market share analysis	Market growth analysis	New product development	Market development	Importance of product configuration	Impotence of product improvement	Importance of technology configuration

.

\*. Correlation is significant at the 0.05 level (1-tailed).

•

\*\*. Correlation is significant at the 0.01 level (1-tailed).

507

.

>
50
Ite
Ľ,
$\mathbf{S}$
Ľ
f
n
ij
na
E
fo
l
st
ĕ
$\mathbf{U}$
6
es
JS
1
ē
ŝ
хе
Ľ.
at
Σ
el
Ň
Ľ
<b>1</b>
n
.e
a
Le
E
ຽ
n
SC
al
Pe
e
th
Jf
S
'el
e
el
nc
(a)
fic
n.
- Bo
ss Co
h
L
8
14
2
A
le
<b>ib</b>
Ë,

					ĺ									
	To	Evaluat	Decisio	Improv	Produc	Planne	Import	Market	Market	New	Market	Import	Impote	_
	forecas	ion of	ns of	ement	t	d and	ance of	share	growth	product	develo	ance of	nce of	
	t the	invest	compa	of	quality	actual	compet	analysi	analysi	develo	pment	product	product	+
Sig. (1-tailed)	future	ment	ny	product	analysi	custom	itors	s	s	pment		configu	improv	<u> </u>
)		project	strateg	charact	s	er	cost					ration	ement	<u> </u>
		s	y	eristics		profita	analysi							
	•					bility	s							
To forecast the future		.004	000.	000.	.005	.015	.010	.058	.019	.121	.012	.031	.063	<u> </u>
Evaluation of investment projects	.004	•	.000	.041	.001	.113	.422	.441	.007	.043	000.	600.	.059	<u> </u>
Decisions of company strategy	000.	000.		000.	000.	000.	.002	.004	000.	.020	000.	.004	.002	<u> </u>
Improvement of product characteristics	000.	.041	000.		000.	000.	000.	000.	000.	.003	.000	.001	.000	<u> </u>
Product quality analysis	.005	.001	000.	.000		000.	.001	.000	000.	.004	000.	000.	000	
Planned and actual customer profitability	.015	.113	000.	000.	000.		000.	000.	000.	000.	.010	000.	000.	· ·
Importance of competitors cost analysis	.010	.422	.002	000.	.001	000.		000.	000.	000	600.	.008	.022	· ·
Market share analysis	.058	.441	.004	000.	000.	000.	000.		.022	.002	.005	.000	000.	
Market growth analysis	.019	.007	000.	000.	000.	000.	000.	.022		000.	000.	.030	000.	· -
New product development	.121	.043	.020	.003	.004	000.	000.	.002	000.		.053	000.	000.	<u> </u>
Market development	.012	000.	000.	000.	000.	.010	600.	.005	000.	.053		000.	.010	•
Importance of product configuration	.031	600.	.004	.001	000.	000.	.008	000.	.030	000.	.000		000.	· ·
Importance of product improvement	.063	.059	.002	000.	000.	000.	.022	000.	000	000.	.010	000.		· ·
Importance of technology configuration	000.	.010	000.	000.	000.	000.	000.	.001	000.	.012	000.	000.	000.	<u> </u>

Correlation	Market changes	Competition	Customer	Profit margins.	Product	Nature of	Prices o
	(sales).	intensity.	variety.		innovation	competition.	product:
Market changes (sales).	1.000	060	.049	109	.183	.140	.173
Competition intensity.	060	1.000	.153	057	040	042	.094
Customer variety.	.049	.153	1.000	000.	.209	.408**	.038
Profit margins.	109	057	000.	1.000	.116	057	032
Product innovation	.183	040	.209	.116	1.000	.085	.349*
Nature of competition.	.140	042	.408**	057	.085	1.000	.094
Prices of products.	.173	.094	.038	032	.349**	.094	1.000
** Correlation is significant at the 0.01 lovel (1 toiled)							

Table A2.15: Pearson Correlation of Level Matrixes for P (Un) of Product Market

. Correlation is significant at the 0.01 level (1-tailed).

Table A2.15a: The Significance levels of the Pearson correlations of Level Matrixes for P (Un) of Product Market

	Market	Competition	Customer	Profit margins.	Product	Nature of	Prices o
Sig. (1-tailed)	changes (sales).	intensity.	variety.		innovation	competition.	product
Market changes (sales).	•	.339	.368	.226	.102	.166	.115
Competition intensity.	.339		.144	.348	.391	.387	.258
Customer variety.	.368	.144		.500	.073	.002	.396
Profit margins.	.226	.348	.500		.212	.348	.413
Product innovation	.183	.391	.073	.212		.278	.007
Nature of competition.	.140	042	.002	.348	.278		.258
Prices of products.	.173	.094	.369	.413	.007	.258	.

					<b>. T. T</b>	و	
Correlation	Supply	Sources of supply	Competition	Cost of raw	Delivery time	Nature of	Prices of ra
			between suppliers	material		suppliers	material
Supply	1.000	054	043	006	.427**	.139	.221
Sources of supply	054	1.000	422**	086	101	.391**	.108
Competition between suppliers	043	422**	1.000	.275*	.105	136	.061
Cost of raw material	006	086	.275*	1.000	860.	044	.374**
Delivery time	.427**	101	.105	860.	1.000	057	052
Nature of suppliers	.139	.391**	136	044	057	1.000	.257*
Prices of raw material	.221	.108	.061	.374**	052	.257*	1.000
*. Correlation is significant at the 0.05 level (1-taile	.(b	**. Correlation i	s significant at the	: 0.01 level (1-tail	led).		

Table A2.16: Pearson Correlation of Level Matrixes for P (CI) of Raw Material and Supply

Table A2.16a: The Significance levels of the Pearson correlations of Level Matrixes for P (CI) of Raw Material and Supply

Sig. (1-tailed)	Supply	Sources of supply	Competition	Cost of raw	Delivery time	Nature of	Prices of rav
			between suppliers	material		suppliers	material
Supply		.356	.385	.484	.001	.169	.062
Sources of supply	.356		.001	.276	.243	.002	.227
Competition between suppliers	.385	.001	•	.026	.234	.174	.336
Cost of raw material	.484	.276	.026		.250	.380	.004
Delivery time	.001	.243	.234	.250		.346	.360
Nature of suppliers	.169	.002	.174	.380	.346		.036
Prices of raw material	.062	.227	.336	.004	.360	.036	

Table A2.17: Pearso	n Correlation	of Level Matri	ixes for P (Un)	) of Raw Mate	rial and Supp	ly	
Correlation	Supply	Sources of supply	Competition	Cost of raw	Delivery time	Nature of	Prices of ra
			between suppliers	material		suppliers	material
Supply	1.000	.096	.150	.061	.221	.087	.089
Sources of supply	.096	1.000	.632**	.263*	.274*	.400**	.562**
Competition between suppliers	.150	.632**	1.000	.404**	.320*	.746**	.751**
Cost of raw material	.061	.263*	.404**	1.000	.374**	.367**	.474**
Delivery time	.221	.274*	.320*	.374**	1.000	.325*	.221
Nature of suppliers	.087	.400**	.746**	.367**	.325*	1.000	**969.
Prices of raw material	.089	.562**	.751**	.474**	.221	**969.	1.000
*. Correlation is significant at the 0.05 level (1-t:	ailed).	. **. Correls	ation is significan	it at the 0.01 leve	il (1-tailed).		

Table A2.17a: The Significance levels of the Pearson correlations of Level Matrixes for P (Un) of Raw Material and Supply

Sig. (1-tailed)	Supply	Sources of supply	Competition	Cost of raw	Delivery time	Nature of	Prices of ra
			between suppliers	material		suppliers	material
Supply		.254	.149	.336	.062	.087	680.
Sources of supply	.254		000.	.033	.027	.002	000.
Competition between suppliers	.149	000.	•	.002	.012	000.	000
Cost of raw material	.336	.033	.002	•	.004	.004	000.
Delivery time	.062	.027	.012	.004	•	.011	.061
Nature of suppliers	.274	.002	000.	.004	.011	•	000.
Prices of raw material	.270	.000	.000	.000	.061	.000	•

				~			
Correlation	Availability of	Competing	Wage rate	Trade union	Labour turnover	Labour	Relation be
	skilled labour	employers				dependency	water comp:
Availability of skilled labour	1.000	.045	.239*	109	350**	046	.139
Competing employers	.045	1.000	.030	.083	.026	084	054
Wage rate	.239*	.030	1.000	105	063	075	.180
Trade union	109	.083	105	1.000	.253*	.105	133
Labour turnover	350**	.026	063	.253*	1.000	091	407**
Labour dependency	046	084	075	.105	091	1.000	.138
The relation between water companies	.139	054	.180	133	407**	.138	1.000
*. Correlation is significant at the 0.05 level (1-ta	ailed).	).**	Correlation is sig	guificant at the 0.0	)1 level (1-tailed).		

Table A2.18: Pearson Correlation of Level Matrixes for P (CI) of Labour

.

Table A2.18a: The Significance levels of the Pearson correlations of Level Matrixes for P (CI) of Labour

Sig. (1-tailed)	Availability of	Competing	Wage rate	Trade union	Labour turnover	Labour	Relation bet
)	skilled labour	employers				dependency	water comp:
Availability of skilled labour		.378	.047	.225	.006	.377	.169
Competing employers	.378	•	.419	.284	.429	.280	.354
Wage rate	.047	.419	•	.234	.333	.302	.105
Trade union	.225	.284	.234		.038	.233	.179
Labour turnover	.006	.429	.333	.038		.265	.002
Labour dependency	.377	.280	.302	.233	.265		.170
The relation between water companies	.169	.354	.105	.179	.002	.170	

Table A2.1	9: Pearson Co	rrelation of L	evel Matrixes	for P (Un) of I	abour		
Correlation	Availability of	Competing	Wage rate	Trade union	Labour turnover	Labour	Relation be
	skilled labour	employers				dependency	water comp
Availability of skilled labour	1.000	092	.339**	.289*	.244*	.120	.165
Competing employers	092	1.000	.219	.405**	.162	.479**	.107
Wage rate	.339**	.219	1.000	.373**	.423**	.478**	.271*
Trade union	.289*	.405**	.373**	1.000	.353**	.323*	.300*
Labour turnover	.244*	.162	.423**	.353**	1.000	.286*	.233
Labour dependency	.120	.479**	.478**	.323*	.286*	1.000	.159
The relation between water companies	.165	.107	.271*	.300*	.233	.159	1.000
*. Correlation is significant at the 0.05 level (1-t:	ailed).	**. Correl	ation is significa	nt at the 0.01 leve	el (1-tailed).		_

Table A2.19a: The Significance levels of the Pearson correlations of Level Matrixes for P (Un) of Labour

•

Sig. (1-tailed)	Availability of	Competing	Wage rate	Trade union	Labour turnover	Labour	Relation be
``````````````````````````````````````	skilled labour	employers				dependency	water comp
Availability of skilled labour		.263	.008	.021	.044	.202	.126
Competing employers	.263		.063	.002	.131	000.	.230
Wage rate	.008	.063	•	.004	.001	000.	.028
Trade union	.021	.002	.004		.006	.011	.017
Labour turnover	.044	.131	.001	.006		.022	.052
Labour dependency	.202	000.	000.	.011	.022	-	.136
The relation between water companies	.126	.230	.028	.017	.052	.136	

			0	
Correlation	Availability of	Salary of	Competition for good	Staff turnover
	qualified managers	management	managers	
Availability of qualified managers	1.000	362**	.122	.312*
Salary of management	362**	1.000	.188	142
Competition for good managers	.122	.188	1.000	.147
Staff turnover	.312*	142	.147	1.000
*. Correlation is significant at the 0.05 level (1-tailed).				

Table A2.20: Pearson Correlation of Level Matrixes for P (CI) of Management

\*\*. Correlation is significant at the 0.01 level (1-tailed).

Table A2.20a: The Significance levels of the Pearson correlations of Level Matrixes for P (CI) of Management

Sig. (1-tailed)	Availability of	Salary of	Competition for good	Staff turnover
	qualified managers	management	managers	
Availability of qualified managers	•	.005	.200	.014
Salary of management	.005	•	.095	.162
Competition for good managers	.200	.095	•	.153
Staff turnover	.014	.162	.153	

n Correlation of Level	Matrixes for P (Un) of	Management	
Availability of	Salary of	Competition for good	Staff turnover
qualified managers	management	managers	
1.000	.511**	.269*	.510**
.511**	1.000	.356**	.362**
.269*	.356**	1.000	.305*
.510**	.362**	.305*	1.000
	n Correlation of Level Availability of qualified managers 1.000 511** 511** 510**	Availability of aualified managersSalary of Salary of management1.000.511**511**1.000511**.356**269*.362**	Availability of qualified managersCompetition for good managersqualified managersmanagers1.000.511**511**1.000511**1.000269*.356**510**.362**305*

Table A2.21a: The Significance levels of the Pearson correlations of Level Matrixes for P (Un) of Management

Sig. (1-tailed)	Availability of	Salary of	Competition for good	Staff turnover
	qualified managers	management	managers	
Availability of qualified managers	•	.000	.030	000.
Salary of management	.000	•	.006	.005
Competition for good managers	.030	.006	•	.016
Staff turnover	000.	.005	.016	

			0	
Correlation	Availability of fund	Sources of fund	Cost of finance	Staff turnover
Availability of fund	1.000	.093	053	.052
Sources of fund	.093	1.000	.128	077
Cost of finance	053	.128	1.000	087
Staff turnover	.052	077	087	1.000

Table A2.22: Pearson Correlation of Level Matrixes for P (CI) of Capital and Financing

\*. Correlation is significant at the 0.05 level (1-tailed).

\*\*. Correlation is significant at the 0.01 level (1-tailed).

Table A2.22a: The Significance levels of the Pearson correlations of Level Matrixes for P (CI) of Capital and Financing

			• .	)
Sig. (1-tailed)	Availability of fund	Sources of fund	Cost of finance	Staff turnover
Availability of fund	•	.259	.358	.359
Sources of fund	.259	•	.188	.298
Cost of finance	.358	.188		.274
Staff turnover	.052	.298	.274	•

			0	
Correlation	Availability of fund	Sources of fund	Cost of finance	Staff turnover
Availability of fund	1.000	.707**	.299*	.502**
Sources of fund	.707**	1.000	.188	.585**
Cost of finance	.299*	.188	1.000	.183
Staff turnover	.502**	.585**	.183	1.000

Table A2.23: Pearson Correlation of Level Matrixes for P (Un) of Capital and Financing

\*. Correlation is significant at the 0.05 level (1-tailed).

\*\*. Correlation is significant at the 0.01 level (1-tailed).

Table A2.23a: The Significance levels of the Pearson correlations of Level Matrixes for P (Un) of Capital and Financing

-		-		
Sig. (1-tailed)	Availability of fund	Sources of fund	Cost of finance	Staff turnover
Availability of fund	•	000.	.018	.000
Sources of fund	000.		.095	.000
Cost of finance	.018	.095	•	.102
Staff turnover	000.	000.	.102	•

							5			
	Availability	Equipment	Competitio	Delivery of	Equipment	Nature of	Nature of	Technology	Equipment	Capita
Correlation	of	suppliers	n for	equipment	failure	manufacturi	system	developme	obsolescenc	labour
	equipment		equipment			ng	technology	nt	e	substit
						technology				
Availability of equipment	1.000	.194	.120	.189	.243*	.227	.166	.048	.203	.139
Equipment suppliers	.194	1.000	.138	.443**	.154	.250*	.212	.042	149	017
Competition for equipment	.120	.138	1.000	039	.082	011	.244*	.160	040	016
Delivery of equipment	.189	.443**	039	1.000	.322*	.373**	.095	067	089	141
Equipment failure	.243*	.154	.082	.322*	1.000	.404**	.392**	019	.129	.204
Nature of manufacturing	.227	.250*	011	.373**	.404**	1.000	.332**	.159	112	156
technology									<u>.</u>	
Nature of system technology	.166	.212	.244*	.095	.392**	.332**	1.000	.221	.013	.020
Technology development	.048	.042	.160	067	019	.159	.221	1.000	.093	.199
Equipment obsolescence	.203	149	040	089	.129	122	.13	.093	1.000	.656
Capital labour substitution	.139	017	016	141	.204	156	.020	.199	.656**	1.00(

Table A2.24: Pearson Correlation of Level Matrixes for P (CI) of Technology

\*. Correlation is significant at the 0.05 level (1-tailed).

\*\*. Correlation is significant at the 0.01 level (1-tailed).

1 abiv 72:27a. 1 li	DIBUILICAIL		LILE I CALSUL	וו לעון לומעול		INTALLIACS			<b>1</b> 87	
	Availability	Equipment	Competitio	Delivery of	Equipment	Nature of	Nature of	Technology	Equipment	Capiti
Sig (1-failed)	of	suppliers	n for	equipment	failure	manufacturi	system	developme	obsolescenc	laboui
(nation - r) ·Stra	equipment		equipment			ng	technology	nt	ຍ	substi
						technology				
Availability of equipment	•	.088	.203	.095	.045	.056	.125	.371	.079	.168
Equipment suppliers	.088		.169	.001	.142	.040	.069	.387	.151	.454
Competition for equipment	.203	.169		.393	.287	.470	.044	.133	.390	.457
Delivery of equipment	.095	.001	.393		.011	.004	.255	.322	.269	.165
Equipment failure	.045	.142	.287	.011		.002	.002	.447	.187	.078
Nature of manufacturing	.056	.040	.470	.004	.002		600.	.135	.219	.139
technology										
Nature of system technology	.125	.069	.044	.255	.002	600.		.061	.466	.446
Technology development	.371	.387	.133	.322	.447	.135	.061		.261	.082
Equipment obsolescence	.079	.151	.390	.269	.187	.219	.466	.261		000.
	-									

Table A2.24a: The Significance levels of the Pearson correlations of Level Matrixes for P (CI) of Technology

519

•

000.

.082

.446

.139

.978

.165

.457

.454

.168

Capital labour substitution

	Availability	Equipment	Competitio	Delivery of	Equipment	Nature of	Nature of	Technology	Equipment	Capit
Correlation	of	suppliers	n for	equipment	failure	manufacturi	system	developme	obsolescenc	laboui
	equipment		equipment			ng	technology	nt	Ð	substi
,						technology				
Availability of equipment	1.000	.251*	.173	.248*	.227	.283*	.056	.052	423**	.021
Equipment suppliers	.251*	1.000	.128	.483**	.190	.348**	.304*	.348**	660.	.079
Competition for equipment	.173	.128	1.000	.337**	.367**	.360**	.611**	.437**	.121	.526
Delivery of equipment	.248*	.483**	.337**	1.000	.243*	.595**	.327*	.224	001	.231
Equipment failure	.227	.190	.367**	.243*	1.000	.343**	.255*	.189	022	.271
Nature of manufacturing	.283*	.348**	.360**	.595**	.343**	1.000	.458**	.266**	.028	.368
technology						•				
Nature of system technology	.056	.304*	.611**	.327*	.255*	.458**	1.000	.504**	.222	.638
Technology development	.052	.348**	.437**	.224	.189	.266*	.504**	1.000	.357**	.380
Equipment obsolescence	.423**	660.	.121	001	022	.028	.222	.357**	1.000	.205
Capital labour substitution	.021	.079	.526**	.231	.271*	.368**	.638**	.380**	.205	1.00

Table A2.25: Pearson Correlation of Level Matrixes for P (Un) of Technology

\*. Correlation is significant at the 0.05 level (1-tailed).

\*\*. Correlation is significant at the 0.01 level (1-tailed).

Table A2.25a: The	Significanc	e levels of t	he Pearsor	a correlatio	ns of Leve	Matrixes	or P (Un)	of Technol	ogy	
	Availability	Equipment	Competitio	Delivery of	Equipment	Nature of	Nature of	Technology	Equipment	Capita
Sig (1-failed)	of	suppliers	n for	equipment	failure	manufacturi	system	developme	obsolescenc	labour
(name_r) .5m	equipment		equipment			ng	technology	nt	e	substi
						technology				
Availability of equipment		.039	.115	.041	.057	.023	.348	.359	.001	.442
Equipment suppliers	.039		.187	000.	.093	.007	.016	.007	.247	.292
Competition for equipment	.115	.187		.008	.004	.005	.000	.001	.201	000.
Delivery of equipment	.041	000.	.008		.045	.000	.010	.059	.496	.053
Equipment failure	.057	.093	.004	.045	•	.007	.037	.094	.441	.029
Nature of manufacturing	.023	.007	.005	000.	.007	•	000.	.031	.425	.004
technology .	-			•					•	
Nature of system technology	.348	.016	000.	.010	.037	000		.000	.061	000.
Technology development	.359	.007	.001	.059	.094	.031	.000		.005	.003
Equipment obsolescence	.001	.247	.201	.496	.441	.425	.061	.005		.076
Capital labour substitution	.442	.292	000.	.053	.029	.004	000.	.003	.076	.

.

			0	
Correlation	Regulators in the	Number of legislation	Nature of legislation	Effect of legislation
	industry			
Regulators in the industry	1.000	.656**	.031	.303
Number of legislation	.656**	1.000	.025	.268*
Nature of legislation	.031	.025	1.000	.147
Effect of legislation	.303*	.268*	.147	1.000
*. Correlation is significant at the 0.05 level (1-tailed).				
** Coundation is simulficant at the 0.01 loved (1 toiled)				

Table A2.26: Pearson Correlation of Level Matrixes for P (CI) of Legislation

\*\*. Correlation is significant at the 0.01 level (1-tailed).

Table A2.26a: The Significance levels of the Pearson correlations of Level Matrixes for P (CI) of Legislation

Sig. (1-tailed)	Regulators in the	Number of legislation	Nature of legislation	Effect of legislatic
	industry			
Regulators in the industry	•	000	.415	.016
Number of legislation	000.	•	.431	.030
Vature of legislation	.415	.431	•	.154
Effect of legislation	.016	.030	.154	•

			)	
Correlation	Regulators in the	Number of legislation	Nature of legislation	Effect of legislatic
	industry			
Regulators in the industry	1.000	**062.	.409**	.620**
Number of legislation	.790**	1.000	.377**	.705**
Nature of legislation	.409**	.377**	1.000	.442**
Effect of legislation	.620**	.705**	.442**	1.000
*. Correlation is significant at the 0.05 level (1-tailed).				

Table A2.27: Pearson Correlation of Level Matrixes for P (Un) of Legislation

\*\*. Correlation is significant at the 0.01 level (1-tailed).

Table A2.27a: The Significance levels of the Pearson correlations of Level Matrixes for P (Un) of Legislation

Sig. (1-tailed)	Regulators in the	Number of legislation	Nature of legislation	Effect of legislatic
	industry			
Regulators in the industry	•	000.	.002	.000
Number of legislation	.000	•	.004	000.
Nature of legislation	.002	.004		.001
Effect of legislation	.000	.000	.001	